MICROWAVE ECONDEDITION ENGINEERING

David M. Pozar

John Wiley & Sons, Inc. New York • Chichester • Weinheim • Brisbane Singapore • Toronto

http://www.wiley.com/college



Microwave Engineering

Microwave Engineering

Second Edition

David M. Pozar University of Massachusetts at Amherst



New York • Chichester • Weinheim Brisbane • Singapore • Toronto EXECUTIVE HOITOR Charny Robey EDITORIAL ASSISTANT Susanne Dwyer MARKETING MANAGER Harper Mooy SENIOR PRODUCTION MANAGER Lucitle Buonocore SENIOR PRODUCTION EDITOR Monique Calello COVER DESIGNER David Levy ILLUSTRATION STUDIOS Wellington & Vantage Art If I USTRATION STUDIOS Wellington & Vantage Art If I USTRATION COORDINATOR Gene Atellin MANUFACTURING MANAGER Monique Calello

This book was set in 10/12 Times Roman by ETP HARRISON and printed and bound by R.R. Donnelley & Sons Company, Crawfordsville. The cover was printed by The Lehigh Press, Inc.

Recognizing the importance of preserving what has been written, it is a policy of John Wiley & Sons, Inc. to have books of enduring value published in the United States printed on acid-free paper, and we exert our best efforts to that end.

The paper on this book was manufactured by a mill whose forest management programs include sustained yield harvesting of its timberlands. Sustained yield harvesting principles ensure that the number of trees cut each year does not exceed the amount of new growth.

Copyright (2) 1998, by John Wiley & Sons, Inc.

All rights reserved. Published simultaneously in Canada.

Reproduction or translation of any part of this work beyond that permitted by Sections 107 and 108 of the 1976 United States Copyright Act without the permission of the copyright owner is unlawful. Requests for permission or further information should be addressed to the Permissions Department. John Wiley & Sons, Inc.

Labrary of Congress Cataloging in Publication Data

Pozar, David M,
Microwave engineering / David M Pozar -- 2nd ed,
p. cm,
ISBN 0-471-17096-8 (cloth : alk_paper)
1. Microwaves, 2. Microwave devices, 3. Microwave encults,
J. Thie,
TK7876,P69 1998
621 381'3--dc20 97-20878

⊂Ć1₿

Printed in the United States of America

14 13 12 11



Because education should be the accumulation of understanding, not just an accumulation of facts, I have tried to write a textbook that emphasizes the fundamental concepts of Maxwell's equations, wave propagation, network analysis, and design principles as applied to modern microwave engineering. Although I have avoided the handbook approach, in which a large number of results are presented with little or no explanation or context, a considerable amount of material in this book is related to the design of specific microwave circuits and components, for both practical and motivational value. I have tried to present the analysis and logic behind these designs so that the reader can see and understand the process of applying fundamental concepts to arrive at useful results. The engineer who has a firm grasp of the basic concepts and principles of microwave engineering, and has seen how these can be applied toward a specific design objective, is the engineer who is most likely to be rewarded with a creative and productive career.

Modern microwave engineering involves predominantly circuit analysis and design, in contrast to the field theory orientation of a generation ago. This may come as a surprise, possibly mixed with regret, to those of us teachers for whom microwave engineering meant applying sophisticated mathematics to the solution of Maxwell's equations for waveguide components. But the majority of practicing microwave engineers now design planar components and monolithic integrated circuits with no direct recourse to field theory analysis. Microwave computer-aided design (CAD) software and the network analyzer are the essential tools of today's microwave engineer. Microwave engineering education must respond to this shift in emphasis to network analysis, planar circuits and components, and active circuit design. Microwave engineering will always involve electromagnetics (many of the more sophisticated microwave CAD packages implement rigorous field theory solutions), and students will still benefit from exposure to subjects such as waveguide modes and coupling through apertures, but the change in emphasis to microwave circuit analysis and design is clear.

There are other changes as well. Several generations of students were taught that open-circuit stubs should not be used for tuning because of spurious radiation, but this is not a problem with miniature microwave circuitry, in which open-circuit lines are often used for matching in transistor circuits. Technology advancements in planar transmission lines, bipolar and field effect transistors, dielectric resonators, low-noise amplifiers, transistor oscillators, PIN diode control circuits, and monolithic integrated circuits similarly require reconsideration of many of the assumptions underlying the traditional study of the subject. Applications of microwave engineering are also changing, with increasing emphasis on commercial use of microwave technology for personal communications systems, wireless local area networks, millimeter wave collision avoidance vehicle radars, radio frequency (RF) identification tagging, direct broadcast satellite television, and many other systems related to the information infrastructure. These developments suggest that there will be no shortage of challenging problems in RF and microwave engineering and a clear need for engineers having creativity and an understanding of the fundamentals of microwave engineering.

The success of the first edition of Microwave Engineering has been gratifying. Feedback from students and teachers clearly supported coverage of the analysis and design of impedance matching networks, resonators, filters, couplers, amplifiers, and oscillators but also supporting topics such as transmission line theory, microwave network analysis, and aperture coupling of waveguides. The most consistent call was for more coverage of active circuit design. Because of a constraint on the length of this edition, new matetial could not be added without deletion of some existing material-a task that is more difficult than might be imagined. To this end, and with the objective of not eliminating critical fundamental material. I have limited major deletions from the first edition to the topics of plane wave propagation in anisotropic media, the transmission line analogy for plane wave reflection, and transients on transmission lines. Other reductions include topics that could be useful but are not necessary for the study of the rest of the book, such as Mason's rule, Foster's reactance theorem, and parallel connected networks. A few other sections in the beginning of the book were combined or otherwise consolidated. Several examples were eliminated, but most of these have been converted to problems in an attempt to retain at least some mention of the topics that were cut. Although most teachers appreciated it, students seldom read the historical material in Chapter 1 of the first edition, so this chapter was eliminated and the material largely moved to the beginning of the relevant chapters and sections throughout the book. We have also expanded the coverage of active circuit design from one to two chapters in this edition. Chapter 10 covers noise, detectors, mixers, PIN diode circuits, microwave integrated circuits, and a review of sources. Chapter 11 is devoted to transistor amplifier and oscillator design, with new material on the physical construction of transistors and their small-signal equivalent circuit models, balanced amplifiers, distributed amplifiers, and dielectric resonator oscillators. We have also added a discussion of the Thru-Reflect-Line (TRL) method of network analyzer calibration. Other changes include revised analyses of the Wilkinson power divider and multisection quarter wave transformers, the addition of titles to the examples, new or modified examples and problems, and the correction of several typographical errors.

This text was written for use in a two-semester course in microwave engineering. for senior or first-year graduate students. If students have had a good course in undergraduate electromagnetics, the material in Chapters 1 and 2 can be reviewed fairly quickly. Students with less background should study this material in more detail. The chapters are organized in the sequence in which our course at the University of Massachusetts at Amherst is taught, but it is certainly possible to pick and choose from the text to suit the needs of a one-semester course or an advanced course in active circuit design or to cover topics in a different order. Two important things that will be included in a successful course on microwave engineering are a microwave laboratory experience and the use of computer-aided design (CAD) software for microwave circuit analysis. A hands-on laboratory is expensive to equip but provides the best way for students to develop an intuition and physical feeling for microwave phenomenon. A laboratory with the first semester of the course should cover the measurement of microwave power, frequency, standing wave ratio (SWR), impedance, and S-parameters, as well as the characterization of basic microwave components such as tuners, couplers, resonators, loads, circulators, and filters. Important practical knowledge about connectors, waveguides, and microwave test equipment will also be acquired in this way. If available, a more advanced laboratory session can consider topics such as noise figure, amplifier characterization, intermodulation products, and microwave mixers. Naturally, the type of experiments that can be offered is heavily dependent on the test equipment that is available.

There are several commercially available CAD packages for microwave circuit analysis, for both personal computers and mainframe computers. Providing students with access to such software allows them to verify the results of the design-oriented problems in the text, giving immediate feedback that builds confidence and makes the effort more rewarding. Because the drudgery of repetitive calculation is eliminated, students can easily try alternative approaches and explore problems in a more detailed way. The effect of losses on the response of a filter, for example, would be practically impossible to evaluate by hand calculation, but it is easy to do on the computer. And because CAD software is used extensively in the microwave industry, classroom experience with such tools will be useful upon graduation.

ACKNOWLEDGMENTS

I would like to thank many people for their help in completing this book, but my foremost appreciation goes to the many students who have used the first edition of *Microwave Engineering*. I also would like to thank my colleagues in microwave engineering at the University of Massachusetts for their support and collegiality through the years. In particular, Bob McIntosh and Keith Carver made many helpful suggestions based on their experiences with the book in their classes. I thank my friends in industry and universities for supplying photographs: Dr. Naresh Deo of Millitech Corp., Dr. John Bryant of the University of Michigan, Mr, Harry Syrigos of Alpha Industries. Professor Cal Swift of the University of Massachusetts, Dr. Mike Adlerstein of Raytheon Co., Mr. Hugo Vifian of Hewlett-Packard Co., Mr. Mark Russell of Raytheon Co., and Dr. M. Abouzahra of Lincoln Laboratory. Finally, I would like to thank the staff of John Wiley & Sons for their helpful efforts during this project. The cheetful professionalism of Monique Calello (Production Editor) was especially appreciated.

> David M, Pozar Amherst, MA

	Contents
1	ELECTROMAGNETIC THEORY 1
1.1	Introduction to Microwave Engineering 1 Applications of Microwave Engineering 2 • A Short History of Microwave Engineering 3
1.2	Maxwell's Equations 5
1.3	Fields in Media and Boundary Conditions9Fields at a General Material Interface12•Fields at a General Material Interface12•Interface14•Fields at the Interface with a Perfect Conductor (ElectricWall)14•The Magnetic Wall Boundary Condition15The Radiation Condition15
I.4	The Wave Equation and Basic Plane Wave Solutions 16 The Helmholtz Equation 16 • Plane Waves in a Lossless Medium 16 • Plane Waves in a General Lossy Medium 18 • Plane Waves in a Good Conductor 19
L5	General Plane Wave Solutions21Circularly Polarized Plane Waves25
1.6	Energy and Power 26 Power Absorbed by a Good Conductor 29
1.7	Plane Wave Reflection from a Media Interface30General Medium31•Lossless Medium32•GoodConductor34•Perfect Conductor36•The SurfaceImpedance Concept36
1.8	Oblique Incidence at a Dielectric Interface 39 Parallel Polarization 40 • Perpendicular Polarization 41 • Total Reflection and Surface Wayes 43
1.9	Some Useful Theorems45The Reciprocity Theorem45•The Uniqueness Theorem49

ix

Contents

2_____ TRANSMISSION LINE THEORY 56

- The Lumped-Element Circuit Model for a Transmission Line 56 2.1 Wave Propagation on a Transmission Line 58 The Lossless Line 59 59 2.2Field Analysis of Transmission Lines Transmission Line Parameters The Telegrapher Equations Derived 60 ٠ from Field Analysis of a Coaxial Line 63 Propagation Constant, Impedance, and Power Flow for the Lossless Coaxial Line 64 2.3The Terminated Lossless Line 65 Special Cases of Lossfess Terminated Lines 68 Point of Interest: Decibels and Nepers 72 2.4 The Smith Chart 73 The Combined Impedance-Admittance Smith Chart 76 The Slotted Line 79 83 2.5 The Quarter-Wave Transformer The Impedance Viewpoint ٠ The Multiple Reflection Viewpoint 83 85 87 Generator and Load Mismatches 2.6 Load Matched to Line 89 Generator Matched to Loaded . Line 89 Conjugate Matching 89 •
- 2.7 Lossy Transmission Lines 90 The Distortionless Line 92 The Low-Loss Line 90 The • • Terminated Lossy Line The Perturbation Method for Calculating 93 • Attenuation 94 The Wheeler Incremental Inductance Rule 96 ٠

3

TRANSMISSION LINES AND WAVEGUIDES 104

3.1	General Solu	tions for	г ТЕМ	, TE, an	id TM	Waves	-103	5		
	TEM Waves	107		TE V	Waves	109				
	TM Waves	110	•							
	Attenuation D	ue to Die	electric	Loss	111					
3.2	Parallel Plate TEM Modes	Waveg 112	uide •	112 ТМ М	Aodes	114	•	TE	Modes	117
3.3	Rectangular '	Wavegu	ide	120						
	TE Modes Loaded Wave	120 guide	• 130	TM Mo •	des Point	125 of Interest:	• Wavi	TE _{mo} eguide	Modes of Flanges	f a Partially 131
3.4	Circular Way TE Modes	eguide 133	13 •	2 TM Mo	des	137				
3.5	Coaxial Line TEM Modes Interest: Coas	14) [4] sial Conu	• ectors	Hight 146	er-Orde	er Modes	143	1	• Poi	nt of

Contents

3.6	Surface Waves on a Grounded Dielectric Slab 147
	TM Modes 147 • TE Modes 150 • Point of Interest:
	Root-Finding Algorithms 152
3.7	Stripline 153
	Formulas for Propagation Constant, Characteristic Impedance, and
	Attenuation 154 An Approximate Electrostatic Solution 157
3.8	Microstrip 160
	Formulas for Effective Dielectric Constant, Characteristic Impedance, and
	Attenuation 162 • An Approximate Electrostatic Solution 164
3.9	The Transverse Resonance Technique 167
	TM Modes for the Parallel Plate Waveguide 168 • TE _{6n} Modes of a
	Partially Loaded Rectangular Waveguide 169
3.10	Wave Velocities and Dispersion 170
	Group Velocity 170
3.11	Summary of Transmission Lines and Waveguides 173
	Other Types of Lines and Guides 174 • Point of Interest: Power Capacity.
	of Transmission Lines 176
	MICHOWAVE NETWORK ANALYSIS 182
41	Impedance and Equivalent Voltages and Currents 183
	Equivalent Voltages and Currents 183 • The Concept
	of Impedance 187 • Even and Odd Properties of $Z(\omega)$ and $\Gamma(\omega)$ 190
47	Impedance and Admutance Matrices 191
	Reciprocal Networks 193 • Lossless Networks 195
4 2	The Scattering Matrix 196
7.5	Reciprocal Networks and Lossless Notworks 100 • A Shift in Reference
	Planes 202 • Generalized Scattering Parameters 204 • Point of
	Interest: The Vector Network Analyzer 205
44	The Transmission (ARCD) Matrix 206
••••	Relation to Impedance Matrix 209 • Fourvalent Circuits for Two-Port
	Networks 210
4.5	Signal Flow Graphs 213

4

- Decomposition of Signal Flow Graphs 214 Application to TRL Network Analyzer Calibration 217 • Point of Interest: Computer-Aided Design for Microwave Circuits 222
- 4.6 Discontinuities and Modal Analysis 222
 Modal Analysis of an H-Plane Step in Rectangular Waveguide 225 Point of Interest: Microstrip Discontinuity Compensation 229
- 4.7 Excitation of Waveguides—Electric and Magnetic Currents
 230
 Current Sheets That Excite Only One Waveguide Mode
 230
 Mode
 Excitation from an Arbitrary Electric or Magnetic Current Source
 232

xi

xii	C	Contents
	4.8	Excitation of Waveguides—Aperture Coupling 237 Coupling Through an Aperture in a Transverse Waveguide Wall 240 • Coupling Through an Aperture in the Broad Wall of a Waveguide 243
5		IMPEDANCE MATCHING AND TUNING 251
	5.1	Matching with Lumped Elements (L. Networks)252Analytic Solutions253• Smith Chart Solutions254• Point ofInterest: Lumped Elements for Microwave Integrated Circuits257
	5.2	Single-Stub Tuning 258 Shunt Stubs 259 • Series Stubs 262
	5.3	Double-Stub Tuning 266 Smith Chart Solution 266 • Analytic Solution 270
	5.4	The Quarter-Wave Transformer 271
	5.5	The Theory of Small Reflections275Single-Section Transformer276•Multisection Transformer277

5.6 Binomial Multisection Matching Transformers 278

Chebyshev Multisection Matching Transformers 282 5.7 Chebyshev Polynomials 283 Design of Chebyshev Transformers 285 ٠ Tapered Lines 5.8288 Exponential Taper Triangular Taper Klopfenstein 29() 291 ٠ • Taper 192

The Bode-Fano Criterion 5.9 295

6

MICROWAVE RESONATORS 300

Series and Parallel Reso	nant Cir	300				
Series Resonant Circuit	300	٠	Parallel Resonant Circ	uit .	303	٠
Loaded and Unloaded Q	306					
Transmission Line Reso	nators	306	i			
Short-Circuited $\lambda/2$ Line	306		Short-Circuited $\lambda_1/4^{-1}$	Line	310	
Open-Circuited $\lambda/2$ Line	311					
Rectangular Waveguide	Cavities	5 3	13			
Resonant Frequencies	313	•	Q of the TE_{00} . Mode	315		
Circular Waveguide Cav	/ities	318				
Resonant Frequencies	318	•	Q of the TE _{nml} Mode	320		
Dielectric Resonators	323					
Resonant Frequencies of T	Eate Mo	de	324			
Fabry-Perot Resonators	328					
Stability of Open Resonate	urs 30	30				
	Series and Parallel Reso Series Resonant Circuit Loaded and Unloaded Q Transmission Line Reso Short-Circuited $\lambda/2$ Line Open-Circuited $\lambda/2$ Line Rectangular Waveguide Resonant Frequencies Circular Waveguide Cav Resonant Frequencies Dielectric Resonators Resonant Frequencies of T Fabry-Perot Resonators Stability of Open Resonators	Series and Parallel Resonant CirSeries Resonant Circuit300Loaded and Unloaded Q306Transmission Line ResonatorsShort-Circuited $\lambda/2$ Line306Open-Circuited $\lambda/2$ Line306Open-Circuited $\lambda/2$ Line311Rectangular Waveguide CavitiesResonant Frequencies313Circular Waveguide CavitiesResonant Frequencies318Dielectric Resonators323Resonant Frequencies of TE ₀₁₆ MoFabry-Perot Resonators328Stability of Open Resonators33	Series and Parallel Resonant CircuitsSeries Resonant Circuit 300 Loaded and Unloaded Q 306 Transmission Line Resonators 306 Short-Circuited $\lambda/2$ Line 306 Open-Circuited $\lambda/2$ Line 311 Rectangular Waveguide Cavities 313 Resonant Frequencies 313 Circular Waveguide Cavities 318 Dielectric Resonators 323 Resonant Frequencies of TEate ModeFabry-Perot Resonators 328 Stability of Open Resonators 330	Series and Parallel Resonant Circuits300Series Resonant Circuit300Parallel Resonant CircuiteLoaded and Unloaded Q306Transmission Line Resonators306Short-Circuited $\lambda/2$ Line306Short-Circuited $\lambda/2$ Line306Open-Circuited $\lambda/2$ Line311Rectangular Waveguide Cavities313Resonant Frequencies313Q of the TEque, ModeCircular Waveguide Cavities318Resonant Frequencies318Q of the TEque, ModeDielectric Resonators323Resonant Frequencies of TEole Mode324Fabry-Perot Resonators328Stability of Open Resonators330	Series and Parallel Resonant Circuits300Series Resonant Circuit300Parallel Resonant CircuitLoaded and Unloaded Q306Transmission Line Resonators306Short-Circuited $\lambda/2$ Line306Short-Circuited $\lambda/2$ Line306Open-Circuited $\lambda/2$ Line311Rectangular Waveguide Cavities313Resonant Frequencies313Q of the TEquencies318Resonant Frequencies318Q of the TEquencies318Resonant Frequencies318Q of the TEquencies323Resonant Frequencies of TEque Mode324Fabry-Perot Resonators328Stability of Open Resonators330	Series and Parallel Resonant Circuits300Series Resonant Circuit300Parallel Resonant Circuit303Loaded and Unloaded Q306306306306Transmission Line Resonators306Short-Circuited $\lambda/2$ Line306Short-Circuited $\lambda/2$ Line306Short-Circuited $\lambda/4$ Line310Open-Circuited $\lambda/2$ Line311Rectangular Waveguide Cavities313Resonant Frequencies313Q of the TE ₄₀ . Mode315Circular Waveguide Cavities318Q of the TE _{nm} Mode320Dielectric Resonators323324Resonant Frequencies of TE ₀₁₆ Mode324Fabry-Perot Resonators328Stability of Open Resonators330

x

•

6.7	Excitation of Resonator	ж З	32	
	Critical Coupling 332	<u> </u>	A Gap-Coupled Microstr	îp
	Resonator 334 •	An	Aperture-Coupled Cavity	337
6.8	Cavity Perturbations	340		
	Material Perturbations	340	 Shape Perturbations 	343

7 POWER DIVIDERS AND DIRECTIONAL COUPLERS 351

7.1	Basic Properties of Dividers and Couplers 351
	Three-Port Networks (T-junctions) 351 • Four-Port Networks (Directional
	Couplers) 354 • Point of Interest Measuring Coupler Directivity 357
7.2	The T-Junction Power Divider 359
	Lossless Divider 360 • Resistive Divider 361
7.3	The Wilkinson Power Divider 363
	Even-Odd Mode Analysis 363 • Unequal Power Division and N-Way Wilkinson Dividers 367
7.4	Waveguide Directional Couplers 368
	Bethe Hole Coupler 369 • Design of Multihole Couplers 374
7.5	The Quadrature (90°) Hybrid 379
	Even-Odd Mode Analysis 380
7.6	Coupled Line Directional Couplers 383
	Coupled Line Theory 384 • Design of Coupled Line
	Couplers 389 • Design of Multisection Coupled Line Couplers 394
7.7	The Lange Coupler 398
7.8	The 180 Hybrid 401
	Even-Odd Mode Analysis of the Ring Hybrid 403 • Even-Odd Mode
	Analysis of the Tapered Coupled Line Hybrid 407 • Waveguide
	Magic-T 411
9.9	Other Couplers 411
	Point of Interest: The Reflectometer 414

8 _____ MICROWAVE FILTERS 422

8.1	Periodic Structures 423			
	Analysis of Infinite Periodic Structures 424 • Terminated Periodic			
	Structures 427 • k-f Diagrams and Wave Velocines 428			
8.2	Filter Design by the Image Parameter Method 431			
	Image Impedances and Transfer Functions for Two-Port Networks 431 •			
	Constant-k Filter Sections 433 • m-Derived Filter Sections 436			
	Composite Filters 440			

xiv Contents

- 8.3 Filter Design by the Insertion Loss Method 443
 Characterization by Power Loss Ratio 444 Maximally Flat Low-Pass Filter
 Prototype 447 Equal-Ripple Low-Pass Filter Prototype 450 •
 Linear Phase Low-Pass Filter Prototypes 451
- 8.4 Filter Transformations 452
 Impedance and Frequency Scaling 454 Bandpass and Bandstop Transformations 457
- 8.5 Filter Implementation 462
 Richard's Transformation 462
 Kuroda's Identities 464
 Impedance and Admittance Inverters 468
- 8.6 Stepped-Impedance Low-Pass Filters 470
 Approximate Equivalent Circuits for Short Transmission Line Sections 470
- 8.7 Coupled Line Filters 474
 Filter Properties of a Coupled Line Section 474
 Design of Coupled Line Bandpass Filters 477
- 8.8 Filters Using Coupled Resonators 486
 Bandstop and Bandpass Filters Using Quarter-Wave Resonators 486
 Bandpass Filters Using Capacitively Coupled Resonators 490
 Direct-Coupled Waveguide Cavity Filters 493

9 THEORY AND DESIGN OF FERRIMAGNETIC COMPONENTS 497

9.1	Basic Properties of Ferrimagnetic Materials 498
	The Permeability Tensor 498 • Circularly Polarized Fields 504 •
	Effect of Loss 506 • Demagnetization Factors 508 • Point of
	Interest: Permanent Magnets 510
9.2	Plane Wave Propagation in a Ferrite Medium 511
	Propagation in Direction of Bias (Faraday Rotation) 512 • Propagation
	Transverse to Bias (Birefringence) 515
9.3	Propagation in a Ferrite-Loaded Rectangular Waveguide 518
	TEma Modes of Waveguide with a Single Ferrite Slab 518 • TEma Modes
	of Waveguide with Two Symmetrical Ferrite Slabs 521
9.4	Ferrite Isolators 523
	Resonance Isolators 523 • The Field Displacement Isolator 527
9.5	Ferrite Phase Shifters 530
	Nonreciprocal Latching Phase Shifter 530 • Other Types of Ferrite Phase
	Shifters 533 • The Gyrator 535
9,6	Ferrite Circulators 535
	Properties of a Mismatched Circulator 537 • Junction Circulator 537

χv

10 ACTIVE MICROWAVE CIRCUITS 547

	10.1	Noise in Microwave Circuits 548
		Dynamic Range and Sources of Noise 548 • Noise Power and Equivalent Noise Temperature 550 • Measurement of Noise Temperature by the
		Y-factor Method 553 • Noise Figure 555 • Noise Figure of a
		Cascaded System 557
	10.2	Detectors and Mixers 559
		Diode Rectifiers and Detectors 559 • Single-Ended Mixer 565 Balanced
		Mixer 568 • Other Types of Mixers 571 • Intermodulation
		Products 574 • Point of Interest: The Spectrum Analyzer 575
	10.3	PIN Diode Control Circuits 576
	10.4	Single-Pole Switches 577 • Pin Dioge Phase Shifters 580
	10,4	Microwave Integrated Circuits 583 Monolithic Microwave
		Integrated Circuits 584
	10.5	Overview of Microwave Sources 588
		Solid-State Sources 589 Microwave Tubes 593
11		
		DESIGN OF MICROWAVE AMPLIFIERS AND
		OSCILLATORS 600
	11.1	Characteristics of Microwave Transistors 601
		Microwave Field Effect Transistors (FETs) 601 • Microwave Bipolar
	11.0	Transistors 604
	11.2	Gain and Stability 600
	113	Single State Translator Amplifier Device 618
	11.3	Design for Maximum Gain (Conjugate Matching) 618 • Constant Gain
		Circles and Design for Specified Gain (Unilateral Device) 622 • Low-Noise
		Amplifier Design 628
	11.4	Broadband Transistor Amplifier Design 632
		Balanced Amplifiers 632
	11.5	Oscillator Design 641
		One-Port Negative Resistance Oscillators 641 • Transistor
		Oscillators 644 • Dielectric Resonator Oscillators 648
<u>12</u>		INTRODUCTION TO MICROWAVE SYSTEMS 655

12.1 System Aspects of Antennas 655 Definite of Important Antenna Parameters Basic Types of 655 • Antennas 656Antenna Pattern Characteristics 658 Antenna e • Efficiency, Gain, and Temperature 661

Contents xvi

12.2	Microwave Communication Systems662Types of Communication Systems662TheFormula663Microwave Transmitters andNoise Characterization of a Microwave Receiver667Frequency-Multiplexed Systems670	e Friis Po Receivers 7 •	wer Transn 666	nission •
12.3	Radar Systems672The Radar Equation673• Pulse RadarRadar677• Radar Cross Section678	675	 Dopp 	ler
12.4	Radiometry679Theory and Applications of Radiometry679Radiometer681•The Dicke Radiometer	Total : 684	Power	
12.5	Microwave Propagation685Atmospheric Effects685• Ground EffectsEfects688	68 7	• Pl	asma
12.6	Other Applications and Topics689Microwave Heating689Energy TransferWarfare691Biological Effects and Safety	690 694	• E	lectronic

_ APPENDICES 697

Α	Prefixes	698

- Vector Analysis В 698
- С **Bessel Functions** 700
- D Other Mathematical Results 703
- E Physical Constants 704
- F Conductivities for Some Materials 704
- G Dielectric Constants and Loss Tangents for Some Materials 705
- Properties of Some Microwave Ferrite Materials Н 705
- Standard Rectangular Waveguide Data 1 706
- J Standard Coaxial Cable Data 707

INDEX

709

CHAPTER 1

Electromagnetic Theory

We begin our study of microwave engineering with a brief historical overview of the field, followed by a review of topics in electromagnetics that we will need throughout the book.

1.1 INTRODUCTION TO MICROWAVE ENGINEERING

The term microwaves refers to alternating current signals with frequencies between 300 MHz (3 \times 10⁸ Hz) and 300 GHz (3 \times 10¹¹), with a corresponding electrical wavelength between $\lambda = c/f = 1$ m and $\lambda = 1$ mm, respectively. Signals with wavelengths on the order of millimeters are called millimeter waves. Figure 1.1 shows the location of the microwave frequency band in the electromagnetic spectrum. Because of the high frequencies (and short wavelengths), standard circuit theory generally cannot be used directly to solve microwave network problems. In a sense, standard circuit theory is an approximation or special use of the broader theory of electromagnetics as described by Maxwell's equations. This is due to the fact that, in general, the lumped circuit element approximations of circuit theory are not valid at microwave frequencies. Microwave components are often distributed elements, where the phase of a voltage or current changes significantly over the physical extent of the device, because the device dimensions are on the order of the microwave wavelength. At much lower frequencies, the wavelength is large enough that there is insignificant phase variation across the dimensions of a component. The other extreme of frequency can be identified as optical engineering, in which the wavelength is much shorter than the dimensions of the component. In this case Maxwell's equations can be simplified to the geometrical optics regime, and optical systems can be designed with the theory of geometrical optics. Such techniques are sometimes applicable to millimeter wave systems, where they are referred to as quasioptical.

In microwave engineering, then, one must often begin with Maxwell's equations and their solutions. It is in the nature of these equations that mathematical complexity arises, since Maxwell's equations involve vector differential or integral operations on vector field quantities, and these fields are functions of spatial coordinates. One of the goals of this book, however, is to try to reduce the complexity of a field theory solution to a result



FIGURE 1.1 The electromagnetic spectrum.

that can be expressed in terms of simpler circuit theory. A field theory solution generally provides a complete description of the electromagnetic field at every point in space, which is usually much more information than we really need for most practical purposes. We are typically more interested in terminal quantities such as power, impedance, voltage, and current, which can often be expressed in terms of circuit theory concepts. It is this complexity that adds to the challenge, as well as the rewards, of microwave engineering.

Applications of Microwave Engineering

Just as the high frequencies and short wavelengths of microwave energy make for difficulties in analysis and design of microwave components and systems, these same factors provide unique opportunities for the application of microwave systems. This is because of the following considerations:

- Antenna gain is proportional to the electrical size of the antenna. At higher frequencies, more antenna gain is therefore possible for a given physical antenna size, which has important consequences for implementing miniaturized microwave systems.
- More bandwidth (information-carrying capacity) can be realized at higher frequencies. A 1% bandwidth at 600 MHz is 6 MHz (the bandwidth of a single television channel), and at 60 GHz a 1% bandwidth is 600 MHz (100 television channels). Bandwidth is critically important because available frequency bands in the electromagnetic spectrum are being rapidly depleted.

- Microwave signals travel by line of sight and are not bent by the ionosphere as are lower frequency signals. Satellite and terrestrial communication links with very high capacities are thus possible, with frequency reuse at minimally distant locations.
- The effective reflection area (radar cross section) of a radar target is usually proportional to the target's electrical size. This fact, coupled with the frequency characteristics of antenna gain, generally makes microwave frequencies preferred for radar systems.
- Various molecular, atomic, and nuclear resonances occur at microwave frequencies, creating a variety of unique applications in the areas of basic science, remote sensing, medical diagnostics and treatment, and heating methods.

Today, the majority of applications of microwaves are related to radar and communication systems. Radar systems are used for detecting and locating air, ground, or seagoing targets and for air-traffic control systems, missile tracking radars, automobile collision-avoidance systems, weather prediction, motion detectors, and a wide variety of remote sensing systems. Microwave communication systems handle a large fraction of the world's international and other long-haul telephone, data, and television transmissions. And most of the currently developing wireless telecommunications systems, such as direct broadcast satellite (DBS) television, personal communications systems (PCSs), wireless local area computer networks (WLANS), cellular video (CV) systems, and global positioning satellite (GPS) systems, operate in the frequency range 1.5 to 94 GHz, and thus rely heavily on microwave technology.

A Short History of Microwave Engineering

The field of microwave engineering is often considered a fairly mature discipline because the fundamental concepts of electromagnetics were developed over 100 years ago, and prohably because radar, being the first major application of microwave technology, was intensively developed as far back as World War II. But even though microwave engineering had its beginnings in the last century, significant developments in high-frequency solid-state devices, microwave integrated circuits, and the ever-widening applications of modern microsystems have kept the field active and vibrant.

The foundations of modern electromagnetic theory were formulated in 1873 by James Clerk Maxwell [1], who hypothesized, solely from mathematical considerations, electromagnetic wave propagation and the notion that light was a form of electromagnetic energy. Maxwell's formulation was cast in its modern form by Oliver Heaviside, during the period from 1885 to 1887. Heaviside was a reclusive genius whose efforts removed many of the mathematical complexities of Maxwell's theory, introduced vector notation, and provided a foundation for practical applications of guided waves and transmission lines. Heinrich Hertz, a German professor of physics and a gifted experimentalist who also understood the theory published by Maxwell, carried out a set of experiments during the period 1887–1891 that completely validated Maxwell's theory of electromagnetic waves. Figure 1.2 shows a photograph of the original equipment used by Hertz in his experiments. It is interesting to observe that this is an instance of a discovery occurring after a prediction has been made on theoretical grounds—a characteristic of many of the



FIGURE 1.2 Original apparatus used by Hertz for his electromagnetics experiments. (1) 50 MHz transmitter spark gap and loaded dipole antenna. (2) Parallel wire grid for polarization experiments. (3) Vacuum apparatus for cathode ray experiments. (4) Hot-wire galvanometer. (5) Reiss or Knochenhauer spirals. (6) Rolled-paper galvanometer. (7) Metal sphere probe. (8) Reiss spark micrometer. (9) Coaxial transmission line. (10-12) Equipment to demonstrate dielectric polarization effects. (13) Mercury induction coil interrupter. (14) Meidinger cell. (15) Vacuum bell jar. (16) High-voltage induction coil. (17) Bunsen cells. (18) Large-area conductor for charge storage. (19) Circular loop receiving antenna. (20) Eightsided receiver detector. (21) Rotating mirror and mercury interrupter. (22) Square loop receiving antenna. (23) Equipment for refraction and dielectric constant measurement. (24) Two square loop receiving antennas. (25) Square loop receiving antenna. (26) Transmitter dipole, (27) High-voltage induction coil. (28) Coaxial line. (29) High-voltage discharger. (30) Cylindrical parabolic reflector/receiver. (31) Cylindrical parabolic reflector/transmitter. (32) Circular loop receiving antenna. (33) Planar reflector. (34, 35) Battery of accumulators. Photographed on October 1, 1913 at the Bavarian Academy of Science, Munich. Germany, with Hertz's assistant, Julius Amman.

Photograph and identification courtesy of J. H. Bryant, University of Michigan.

major discoveries throughout the history of science. All of the practical applications of electromagnetic theory, including radio, television, and radar, owe their existence to the theoretical work of Maxwell.

Because of the lack of reliable microwave sources and other components, the rapid growth of radio technology in the early 1900s occurred primarily in the high frequency (HF) to very high frequency (VHF) range. It was not until the 1940s and the advent of radar development during World War II that microwave theory and technology received substantial interest. In the United States, the Radiation Laboratory was established at the Massachusetts Institute of Technology (MIT) to develop radar theory and practice. A number of top scientists, including N. Marcuvitz, I. I. Rabi, J. S. Schwinger, H. A. Bethe, E. M. Purcell, C. G. Montgomery, and R. H. Dicke, among others, were gathered for what turned out to be a very intensive period of development in the microwave field. Their work included the theoretical and experimental treatment of waveguide components, microwave antennas, small aperture coupling theory, and the beginnings of microwave network theory. Many of these researchers were physicists who went back to physics research after the war (many later received Nobel Prizes), but their microwave work is summarized in the classic 28-volume Radiation Laboratory Series of books that still finds application today.

Communications systems using microwave technology began to be developed soon after the birth of radar, benefitting from much of the work that was originally done for radar systems. The advantages offered by microwave systems, including wide bandwidths and line-of-sight propagation, have proved to be critical for both terrestrial and satellite communications systems and have thus provided an impetus for the continuing development of low-cost miniaturized microwave components. We refer the interested reader to the special Centennial Issue of the *IEEE Transactions on Microwave Theory and Techniques* [2] for further historical perspectives on the field of microwave engineering.

1.2 MAXWELL'S EQUATIONS

Electric and magnetic phenomena at the macroscopic level are described by Maxwell's equations, as published by Maxwell in 1873 [1]. This work summarized the state of electromagnetic science at that time and hypothesized from theoretical considerations the existence of the electrical displacement current, which led to the discovery by Hertz and Marconi of electromagnetic wave propagation. Maxwell's work was based on a large body of empirical and theoretical knowledge developed by Gauss. Antpere, Faraday, and others. A first course in electromagnetics usually follows this historical (or deductive) approach, and it is assumed that the reader has had such a course as a prerequisite to the present material. Several books are available, [3]-[9], that provide a good treatment of electromagnetic theory at the undergraduate or graduate level.

This chapter will outline the fundamental concepts of electromagnetic theory that we will require for the rest of the book. Maxwell's equations will be presented, and boundary conditions and the effect of dielectric and magnetic materials will be discussed. Wave phenomenon is of essential importance in microwave engineering, so much of the chapter is spent on plane wave topics. Plane waves are the simplest form of electromagnetic waves and so serve to illustrate a number of basic properties associated with wave propagation. Although it is assumed that the reader has studied plane waves before, the present material should help to reinforce many of the basic principles in the reader's mind and perhaps to introduce some concepts that the reader has not seen previously. This material will also serve as a useful reference for later chapters.

Chapter 1: Electromagnetic Theory

With an awareness of the historical perspective, it is usually advantageous from a pedagogical point of view to present electromagnetic theory from the "inductive," or axiomatic, approach by beginning with Maxwell's equations. The general form of time-varying Maxwell equations, then, can be written in "point," or differential, form as

$$abla imes ar{\mathcal{E}} = rac{-\partial ar{\mathcal{B}}}{\partial t} - ar{\mathcal{M}},$$
 1.1a

$$abla imes ar{\mathcal{H}} = rac{\partial ar{\mathcal{D}}}{\partial t} + ar{\mathcal{J}}.$$
 1.18

$$abla \cdot ar{\mathcal{D}} =
ho,$$
 I.le

$$\nabla \cdot \bar{\mathcal{B}} = 0.$$
 1.1d

The MKS system of units is used throughout this book. The script quantities represent time-varying vector fields and are real functions of spatial coordinates x,y,z, and the time coordinate t. These quantities are defined as follows:

- $\ddot{\mathcal{E}}$ is the electric field intensity, in V/m.
- $\bar{\mathcal{H}}$ is the magnetic field intensity, in A/m.
- $\bar{\mathcal{D}}$ is the electric flux density, in Coul/m².
- $\vec{\mathcal{B}}$ is the magnetic flux density, in Wb/m².
- $\overline{\mathcal{M}}$ is the (fictitious) magnetic current density, in V/m².
- $\bar{\mathcal{J}}$ is the electric current density, in A/m².
- ρ is the electric charge density, in Coul/m³.

The sources of the electromagnetic field are the currents $\tilde{\mathcal{M}}$ and $\tilde{\mathcal{J}}$, and the electric charge ρ . The magnetic current $\tilde{\mathcal{M}}$ is a fictitious source in the sense that it is only a mathematical convenience: the real source of a magnetic current is always a loop of electric current or some similar type of magnetic dipole, as opposed to the flow of an actual magnetic charge (magnetic monopole charges are not known to exist). The magnetic current is included here for completeness, as we will have occasion to use it in Chapter 4 when dealing with apertures. Since electric current is really the flow of charge, it can be said that the electric charge density ρ is the ultimate source of the electromagnetic field,

In free-space, the following simple relations hold between the electric and magnetic field intensities and flux densities:

$$\bar{\mathcal{B}} = \mu_0 \bar{\mathcal{H}},$$
 1.2a

$$\bar{\mathcal{D}} = \epsilon_0 \bar{\mathcal{E}}.$$
 1.26

where $\mu_0 = 4\pi \times 10^{-7}$ henry/m is the permeability of free-space, and $\epsilon_0 = 8.854 \times 10^{-12}$ farad/m is the permittivity of free-space. We will see in the next section how media other than free-space affect these constitutive relations.

Equations (1.1a)-(1.1d) are linear but are not independent of each other. For instance, consider the divergence of (1.1a). Since the divergence of the curl of any vector is zero

[vector identity (B.12), from Appendix B], we have

$$\nabla \cdot \nabla \times \bar{\mathcal{E}} = 0 = -\frac{\partial}{\partial t} (\nabla \cdot \bar{\mathcal{B}}) - \nabla \cdot \tilde{\mathcal{M}},$$

Since there is no free magnetic charge, $\nabla \cdot \tilde{\mathcal{M}} = 0$, which leads to $\nabla \cdot \tilde{\mathcal{B}} = 0$, or (1.1d). The continuity equation can be similarly derived by taking the divergence of (1.1b), giving

$$\nabla \cdot \bar{\mathcal{J}} + \frac{\partial \rho}{\partial t} = 0.$$
 1.3

where (1.1c) was used. This equation states that charge is conserved, or that current is continuous, since $\nabla \cdot \overline{\mathcal{J}}$ represents the outflow of current at a point, and $\partial \rho / \partial t$ represents the charge buildup with time at the same point. It is this result that led Maxwell to the conclusion that the displacement current density $\partial \overline{\mathcal{D}} / \partial t$ was necessary in (1.1b), which can be seen by taking the divergence of this equation.

The foregoing differential equations can be converted to integral form through the use of various vector integral theorems. Thus, applying the divergence theorem (B.15) to (1.1c) and (1.1d) yields

$$\oint_{S} \bar{\mathcal{D}} \cdot d\bar{s} = \int_{V} \rho \, dv = Q. \tag{1.4}$$

$$\oint_{S} \vec{B} \cdot d\bar{s} = 0, \qquad 1.5$$

where Q in (1.4) represents the total charge contained in the closed volume V (enclosed by a closed surface S). Applying Stokes' theorem (B.16) to (1.1a) gives

$$\oint_{C} \tilde{\mathcal{E}} \cdot d\bar{l} = -\frac{\partial}{\partial t} \int_{S} \hat{\mathcal{B}} \cdot d\bar{s} - \int_{S} \tilde{\mathcal{M}} \cdot d\bar{s}, \qquad 1.6$$

which, without the $\tilde{\mathcal{M}}$ term, is the usual form of Faraday's law and forms the basis for Kirchhoff's voltage law. In (1.6), C represents a closed contour around the surface S, as shown in Figure 1.3. Ampere's law can be derived by applying Stokes' theorem to (1.1b):

$$\oint_{C} \bar{\mathcal{H}} \cdot d\bar{l} = \frac{\partial}{\partial l} \int_{S} \bar{\mathcal{D}} \cdot d\bar{s} + \int_{S} \bar{\mathcal{J}} \cdot d\bar{s} = \frac{\partial}{\partial t} \int_{S} \bar{\mathcal{D}} \cdot d\bar{s} + \mathcal{I}, \qquad 1.7$$

where $\mathcal{I} = \int_{S} \overline{\mathcal{J}} \cdot d\overline{s}$ is the total electric current flow through the surface S. Equations (1.4)-(1.7) constitute the integral forms of Maxwell's equations.

The foregoing equations are valid for arbitrary time dependence, but most of our work will be involved with fields having a sinusoidal, or harmonic, time dependence, with steady-state conditions assumed. In this case phasor notation is very convenient, and so all field quantities will be assumed to be complex vectors with an implied $e^{j\omega t}$ time dependence and written with roman (rather than script) letters. Thus, a sinusoidal electric field in the \hat{x} direction of the form

$$\mathcal{E}(\mathbf{x}, \mathbf{y}, z, t) = \hat{\mathbf{x}} A(x, y, z) \cos(\omega t + \phi), \qquad 1.8$$



FIGURE 1.3 The closed contour C and surface S associated with Faraday's law.

where A is the (real) amplitude, ω is the radian frequency, and ϕ is the phase reference of the wave at t = 0, has the phasor form

$$\bar{E}(x,y,z) = \hat{x}A(x,y,z)e^{j\phi}.$$
1.9

We will assume cosine-based phasors in this book, so the conversion from phasor quantities to real time-varying quantities is accomplished by multiplying the phasor by $e^{j\omega t}$ and taking the real part:

$$\bar{\mathcal{E}}(x, y, z, t) = \operatorname{Re}[\bar{E}(x, y, z)e^{j\omega t}], \qquad 1.10$$

as substituting (1.9) into (1.10) to obtain (1.8) demonstrates. When working in phasor notation, it is customary to suppress the common $e^{j\omega t}$ factor on all terms.

When dealing with power and energy, we will often be interested in the time average of a quadratic quantity. This can be found very easily for time harmonic fields. For example, the average of the square of the magnitude of an electric field given by

$$\hat{\mathcal{E}} = \hat{x}E_1 \cos(\omega t + \phi_1) + \hat{y}E_2 \cos(\omega t + \phi_2) + \hat{z}E_2 \cos(\omega t + \phi_3), \qquad 1.11$$

which has the phasor form

$$\bar{E} = \hat{x}E_1e^{j\phi_1} + \hat{y}E_2e^{j\phi_2} + \hat{z}E_3e^{j\phi_3}, \qquad 1.12$$

can be calculated as

$$\begin{split} |\bar{E}|_{3v}^{2} &= \frac{1}{T} \int_{0}^{T} \bar{E} \cdot \bar{E} \, dt \\ &= \frac{1}{T} \int_{0}^{T} [E_{1}^{2} \cos^{2}(\omega t + \phi_{1}) + E_{2}^{2} \cos^{2}(\omega t + \phi_{2}) + E_{3}^{2} \cos^{2}(\omega t + \phi_{3})] \, dt \\ &= \frac{1}{2} (E_{1}^{2} + E_{2}^{2} + E_{3}^{2}) = \frac{1}{2} |\bar{E}|^{2} = \frac{1}{2} \bar{E} \cdot \bar{E}^{*}. \end{split}$$

$$1.13$$

Then the root-mean-square (rms) value is $|\bar{E}|_{\rm rms} = |\bar{E}|/\sqrt{2}$.

Assuming an $e^{j\omega t}$ time dependence, the time derivatives in (1.1a)-(1.1d) can be replaced by $j\omega$. Maxwell's equations in phasor form then become

$$abla imes ar{E} = -j\omegaar{B} - ar{M},$$
 1.14a

$$\nabla \times \bar{H} = j\omega \bar{D} + \bar{J}, \qquad 1.14b$$

$$\nabla \cdot \tilde{D} = \rho, \qquad \qquad 1.14c$$

$$\nabla \cdot \bar{B} = 0. \tag{1.14d}$$

The Fourier transform can be used to convert a solution to Maxwell's equations for an arbitrary frequency ω to a solution for arbitrary time dependence.

The electric and magnetic current sources. \overline{J} and \widetilde{M} , in (1.14) are volume current densities with units A/m² and V/m², respectively. In many cases, however, the actual currents will be in the form of a current sheet, a line current, or an infinitesimal dipole current. These special types of current distributions can always be written as volume current densities through the use of delta functions. Figure 1.4 shows examples of this procedure for electric and magnetic currents.

1.3 ____ FIELDS IN MEDIA AND BOUNDARY CONDITIONS

In the preceding section it was assumed that the electric and magnetic fields were in free-space, with no material bodies present. In practice, material bodies are often present; this complicates the analysis but also allows the useful application of material properties to microwave components. When electromagnetic fields exist in material media, the field vectors are related to each other by the constitutive relations.

For a dielectric material, an applied electric field \bar{E} causes the polarization of the atoms or molecules of the material to create electric dipole moments that augment the total displacement flux, \bar{D} . This additional polarization vector is called \bar{P}_{e} , the electric polarization, where

$$\bar{D} = \epsilon_0 \bar{E} + \bar{P}_e. \tag{1.15}$$

In a linear medium, the electric polarization is linearly related to the applied electric field as

$$\bar{P}_{e} = \epsilon_{0} \chi_{e} \bar{E}.$$
 1.16

where χ_e , which may be complex, is called the electric susceptibility. Then,

$$\bar{D} = \epsilon_0 \bar{E} + \bar{P}_e = \epsilon_0 (1 + \chi_e) \bar{E} = \epsilon \bar{E}, \qquad 1.17$$

where

$$\epsilon = \epsilon' - j\epsilon'' = \epsilon_0 (1 + \chi_e)$$
 1.18

is the complex permittivity of the medium. The imaginary part of ϵ accounts for loss in the medium (heat) due to damping of the vibrating dipole moments. (Free-space, having a real ϵ , is lossless.) Due to energy conservation, as we will see in Section 1.6, the imaginary part of ϵ must be negative (ϵ'' positive). The loss of a dielectric material may also be considered as an equivalent conductor loss. In a material with conductivity σ , a conduction current density will exist:

$$\bar{I} = \sigma \bar{E},$$
 1.19





which is Ohm's law from an electromagnetic field's point of view. Maxwell's curl equation for \overline{H} (1.14b) then becomes

$$abla imes ar{H} = j\omega ar{D} + ar{J}$$

= $j\omega \epsilon ar{E} + \sigma ar{E}$

1.3 Fields in Media and Boundary Conditions

$$= j\omega\epsilon'\vec{E} + (\omega\epsilon'' + \sigma)\vec{E}$$
$$= j\omega\left(\epsilon' - j\epsilon'' - j\frac{\sigma}{\omega}\right)\vec{E}, \qquad (.20)$$

where it is seen that loss due to dielectric damping $(\omega \epsilon'')$ is indistinguishable from conductivity loss (σ). The term $\omega \epsilon'' + \sigma$ can then be considered as the total effective conductivity. A related quantity of interest is the loss tangent, defined as

$$\tan b = \frac{\omega \epsilon'' + \sigma}{\omega \epsilon'}, \qquad 1.21$$

which is seen to be the ratio of the real to the imaginary part of the total displacement current. Microwave materials are usually characterized by specifying the real permittivity, $\epsilon' = \epsilon_r \epsilon_0$, and the loss tangent at a certain frequency. These constants are listed in Appendix G for several types of materials. It is useful to note that, after a problem has been solved assuming a lossless dielectric, loss can easily be introduced by replacing the real ϵ with a complex $\epsilon = \epsilon' - j\epsilon'' = \epsilon'(1 - j \tan \delta)$.

In the preceding discussion it was assumed that \bar{P}_c was a vector in the same direction as \bar{E} . Such materials are called isotropic materials, but not all materials have this property. Some materials are anisotropic and are characterized by a more complicated relation between \bar{P}_r and \bar{E} , or \bar{D} and \bar{E} . The most general linear relation between these vectors takes the form of a tensor of rank two (a dyad), which can be written in matrix form as

$$\begin{bmatrix} D_x \\ D_y \\ D_z \end{bmatrix} = \begin{bmatrix} \epsilon_{xx} & \epsilon_{xy} & \epsilon_{xz} \\ \epsilon_{yx} & \epsilon_{yy} & \epsilon_{yz} \\ \epsilon_{zx} & \epsilon_{zy} & \epsilon_{zz} \end{bmatrix} \begin{bmatrix} E_y \\ E_y \\ E_z \end{bmatrix} = [\epsilon] \begin{bmatrix} E_x \\ E_y \\ E_z \end{bmatrix}.$$
 1.22

It is thus seen that a given vector component of \overline{E} gives rise, in general, to three components of \overline{D} . Crystal structures and ionized gases are examples of anisotropic dielectrics. For a linear isotropic material, the matrix of (1.22) would reduce to a diagonal matrix with elements, ϵ_{i} .

An analogous situation occurs for magnetic materials. An applied magnetic field may align magnetic dipole moments in a magnetic material to produce a magnetic polarization (or magnetization) \bar{P}_m . Then,

$$\hat{\boldsymbol{B}} = \mu_0 (\hat{H} + \hat{P}_m). \tag{1.23}$$

For a linear magnetic material, \bar{P}_m is linearly related to \bar{H} as

$$\bar{P}_m = \chi_m \bar{H}, \qquad 1.24$$

where γ_m is a complex magnetic susceptibility. From (1.23) and (1.24),

$$\ddot{B} = \mu_0 (1 + \chi_m) \bar{H} = \mu \bar{H},$$
 1.25

where $\mu = \mu_0(1 + \chi_m) = \mu' - j\mu''$ is the permeability of the medium. Again, the imaginary part of χ_m or μ accounts for loss due to damping forces; there is no magnetic conductivity, since there is no real magnetic current. As in the electric case, magnetic

materials may be anisotropic, in which case a tensor permeability can be written as

$$\begin{bmatrix} B_x \\ B_y \\ B_z \end{bmatrix} = \begin{bmatrix} \mu_{xx} & \mu_{xy} & \mu_{xz} \\ \mu_{yx} & \mu_{yy} & \mu_{yz} \\ \mu_{zx} & \mu_{zy} & \mu_{zz} \end{bmatrix} \begin{bmatrix} H_x \\ H_y \\ H_z \end{bmatrix} \Rightarrow [\mu] \begin{bmatrix} H_x \\ H_y \\ H_z \end{bmatrix}.$$
1.26

An important example of anisotropic magnetic materials in microwave engineering is the class of ferrimagnetic materials known as ferrites: these materials and their applications will be discussed further in Chapter 9.

If linear media are assumed (ϵ, μ not depending on \tilde{E} or \tilde{H}), then Maxwell's equations can be written in phasor form as

$$abla imes ar{E} = -j\omega\muar{H} - ar{M},$$
 1.27a

$$\nabla \times \bar{H} = j\omega\epsilon\bar{E} + \bar{J}, \qquad 1.27b$$

$$\nabla \cdot \bar{D} = \rho. \tag{1.27c}$$

$$\nabla \cdot \bar{B} = 0. \qquad 1.27d$$

The constitutive relations are

$$\bar{D} = \epsilon \bar{E},$$
 1.28a

$$\bar{B} = \mu \bar{H},$$
 1.28b

where ϵ and μ may be complex and may be tensors. Note that relations like (1.28a) and (1.28b) generally cannot be written in time domain form, even for linear media, because of the possible phase shift between \overline{D} and \overline{E} , or \overline{B} and \overline{H} . The phasor representation accounts for this phase shift by the complex form of ϵ and μ .

Maxwell's equations (1.27a)-(1.27d) in differential form require known boundary values for a complete and unique solution. A general method used throughout this book is to solve the source-free Maxwell's equations in a certain region to obtain solutions with unknown coefficients, and then apply boundary conditions to solve for these coefficients. A number of specific cases of boundary conditions arise, as discussed in the following.

Fields at a General Material Interface

Consider a plane interface between two media, as shown in Figure 1.5. Maxwell's equations in integral form can be used to deduce conditions involving the normal and tangential fields at this interface. The time-harmonic version of (1.4), where S is the closed "pillbox"-shaped surface shown in Figure 1.6, can be written as

$$\oint_{S} \bar{D} \cdot d\bar{s} = \int_{V} \rho \, dv.$$
 1.29

In the limit as $h \rightarrow 0$, the contribution of D_{tan} through the sidewalls goes to zero, so (1.29) reduces to

$$\Delta SD_{2n} - \Delta SD_{1n} = \Delta S\rho_s,$$

$$D_{2n} - D_{1n} = \rho_s,$$

1.30







FIGURE 1.6 Closed surface 5 for equation (1.29).

where ρ_{s} is the surface charge density on the interface. In vector form, we can write

$$\tilde{n} \cdot (\tilde{D}_2 - \tilde{D}_1) = \rho_s. \tag{1.31}$$

A similar argument for \hat{B} leads to the result that

$$\hat{n} \cdot \hat{B}_2 = \hat{n} \cdot \hat{B}_1, \qquad 1.32$$

since there is no free magnetic charge.

For the tangential components of the electric field we use the phasor form of (1.6),

$$\oint_{C} \vec{E} \cdot d\vec{l} = -j\omega \int_{R} \vec{B} \cdot d\vec{s} - \int_{Z} \vec{M} \cdot d\vec{s}_{s}$$
(.33)

in connection with the closed contour C shown in Figure 1.7. In the limit as $h \to 0$, the surface integral of \bar{B} vanishes (since $S = h \Delta t$ vanishes). The contribution from the surface integral of \bar{M} , however, may be nonzero if a magnetic surface current density



FIGURE 1.7 Closed contour C for equation (1.33).

 \bar{M}_s exists on the surface. The Dirac delta function can then be used to write

$$ar{M} = ar{M}_s \delta(h),$$
 1.34

where h is a coordinate measured normal from the interface. Equation (1.33) then gives

$$\Delta \ell E_{t1} - \Delta \ell E_{t2} = -\Delta \ell M_s,$$

$$E_{t1} - E_{t2} = -M_s.$$
1.35

OF

which can be generalized in vector form as

$$(\bar{E}_2 - \bar{E}_1) \times \hat{n} = \bar{M}_s.$$
 1.36

A similar argument for the magnetic field leads to

$$\hat{n} \times (\bar{H}_2 - \bar{H}_1) = \bar{J}_s,$$
 1.37

where \bar{J}_s is an electric surface current density that may possibly exist at the interface. Equations (1.31), (1.32), (1.36), and (1.37) are the most general expressions for the boundary conditions at an arbitrary interface of materials and/or surface currents.

Fields at a Dielectric Interface

At an interface between two lossless dielectric materials, no charge or surface current densities will ordinarily exist. Equations (1.31), (1.32), (1.36), and (1.37) then reduce to

$$\hat{n}\cdotar{D}_1=\hat{n}\cdotar{D}_2.$$
 1.38a

$$\hat{n} \cdot \bar{B}_1 = \hat{n} \cdot \bar{B}_2, \qquad 1.38b$$

$$\hat{n} imes \vec{E}_1 = \hat{n} imes \vec{E}_2,$$
 1.38c

$$\hat{n} \times \hat{H}_1 = \hat{n} \times \hat{H}_2. \tag{1.38d}$$

In words, these equations state that the normal components of \overline{D} and \overline{B} are continuous across the interface, and the tangential components of \overline{E} and \overline{H} are equal across the interface. Because Maxwell's equations are not all linearly independent, the six boundary conditions contained in the above equations are not all linearly independent. Thus, the enforcement of (1.38c) and (1.38d) for the four tangential field components, for example, will automatically force the satisfaction of the equations for the continuity of the normal components.

Fields at the Interface with a Perfect Conductor (Electric Wall)

Many problems in microwave engineering involve boundaries with good conductors (e.g., metals), which can often be assumed as lossless ($\sigma \to \infty$). In this case of a perfect conductor, all field components must be zero inside the conducting region. This result can be seen by considering a conductor with finite conductivity ($\sigma < \infty$) and noting that the skin depth (the depth to which most of the microwave power penetrates) goes to zero as $\sigma \to \infty$. (Such an analysis will be performed in Section 1.7.) If we also assume here that $M_{\sigma} = 0$, which would be the case if the perfect conductor filled all the space on

one side of the boundary, then (1.31), (1.32), (1.36), and (1.37) reduce to the following:

$$\hat{n} \cdot \hat{D} = \rho_s. \qquad \qquad 1.39a$$

$$\hat{n} \cdot \bar{B} = 0. \qquad 1.39b$$

$$\hat{n} imes \bar{E} = 0,$$
 1.39 c

$$\hat{n} imes \bar{H} = \bar{J}_s,$$
 1.39a

where ρ_s and \hat{J}_s are the electric surface charge density and current density, respectively, on the interface, and \hat{n} is the normal unit vector pointing out of the perfect conductor. Such a boundary is also known as an *electric wall*, because the tangential components of \bar{E} are "shorted out," as seen from (1.39c), and must vanish at the surface of the conductor.

The Magnetic Wall Boundary Condition

Dual to the preceding boundary condition is the magnetic wall boundary condition, where the tangential components of \tilde{H} must vanish. Such a boundary does not really exist in practice, but may be approximated by a corrugated surface, or in certain planar transmission line problems. In addition, the idealization that $\hat{n} \times \tilde{H} = 0$ at an interface is often a convenient simplification, as we will see in later chapters. We will also see that the magnetic wall boundary condition is analogous to the relations between the voltage and current at the end of an open-circuited transmission line, while the electric wall boundary condition, the voltage and current at the end of a short-circuited transmission line. The magnetic wall condition, then, provides a degree of completeness in our formulation of boundary conditions and is a useful approximation in several cases of practical interest.

The fields at a magnetic wall satisfy the following conditions:

$$\hat{n} \cdot \bar{D} = 0, \qquad 1.40a$$

$$\hat{n} \cdot \vec{B} = 0. \qquad 1.40b$$

$$\hat{n} \times \bar{E} = -\bar{M}_s,$$
 1.40c

$$\hat{n} imes \bar{H} = 0,$$
 1.40d

where \hat{n} is the normal unit vector pointing out of the magnetic wall region.

The Radiation Condition

When dealing with problems that have one or more infinite boundaries, such as plane waves in an infinite medium, or infinitely long transmission lines, a condition on the fields at infinity must be enforced. This boundary condition is known as the *radiation condition*, and is essentially a statement of energy conservation. It states that, at an infinite distance from a source, the fields must either be vanishingly small (i.e., zero) or propagating in an outward direction. This result can easily be seen by allowing the infinite medium to contain a small loss factor (as any physical medium would have).

Incoming waves (from infinity) of finite amplitude would then require an infinite source at infinity, and so are disallowed.

1.4 THE WAVE EQUATION AND BASIC PLANE WAVE SOLUTIONS

The Helmholtz Equation

In a source-free, linear, isotropic, homogeneous region, Maxwell's curl equations in phasor form are

$$abla imes ar{E} = -j\omega\muar{H},$$
 1.41a

$$\nabla \times \bar{H} = j\omega\epsilon \bar{E}, \qquad 1.41b$$

and constitute two equations for the two unknowns, \vec{E} and \vec{H} . As such, they can be solved for either \vec{E} or \vec{H} . Thus, taking the curl of (1.41a) and using (1.41b) gives

$$abla imes
abla imes ar{E} = -j\omega\mu
abla imes ar{H} = \omega^2 \mu \epsilon ar{E},$$

which is an equation for \overline{E} . This result can be simplified through the use of vector identity (B.14), $\nabla \times \nabla \times \overline{A} = \nabla(\nabla \cdot \overline{A}) - \nabla^2 \overline{A}$, which is valid for the rectangular components of an arbitrary vector \overline{A} . Then,

$$\nabla^2 \bar{E} + \omega^2 \mu \epsilon \bar{E} = 0, \qquad 1.42$$

since $\nabla \cdot \overline{E} = 0$ in a source-free region. Equation (1.42) is the wave equation, or Helmholtz equation, for \overline{E} . An identical equation for \overline{H} can be derived in the same manner:

$$\nabla^2 \bar{H} + \omega^2 \mu \epsilon \bar{H} = 0.$$
 1.43

A constant $k = \omega \sqrt{\mu \epsilon}$ is defined and called the wavenumber, or propagation constant, of the medium; its units are 1/m.

As a way of introducing wave behavior, we will next study the solutions to the above wave equations in their simplest forms, first for a lossless medium and then for a lossy (conducting) medium.

Plane Waves in a Lossless Medium

In a lossless medium, ϵ and μ are real numbers, so k is real. A basic plane wave solution to the above wave equations can be found by considering an electric field with only an \hat{x} component and uniform (no variation) in the x and y directions. Then, $\partial/\partial x = \partial/\partial y = 0$, and the Helmholtz equation of (1.42) reduces to

$$\frac{\partial^2 E_x}{\partial z^2} + k^2 E_x = 0. 1.44$$

The solutions to this equation are easily seen, by substitution, to be of the form

$$E_x(z) = E^+ e^{-jkz} + E^- e^{jkz}.$$
 1.45

where E^+ and E^- are arbitrary amplitude constants.

The above solution is for the time harmonic case at frequency ω . In the time domain, this result is written as

$$\mathcal{E}_{x}(z,t) = E^{+}\cos(\omega t - kz) + E^{-}\cos(\omega t + kz), \qquad 1.46$$

where we have assumed that E^+ and E^- are real constants. Consider the first term in (1.46). This term represents a wave traveling in the +z direction, since, to maintain a fixed point on the wave ($\omega t - kz = \text{constant}$), one must move in the +z direction as time increases. Similarly, the second term in (1.46) represents a wave traveling in the negative z direction; hence the notation E^+ and E^- for these wave amplitudes. The velocity of the wave in this sense is called the phase velocity, because it is the velocity at which a fixed phase point on the wave travels, and it is given by

$$p_{\mathbf{p}} = \frac{dz}{dt} \Rightarrow \frac{d}{dt} \left(\frac{\omega t - \text{constant}}{k}\right) = \frac{\omega}{k} = \frac{1}{\sqrt{\mu\epsilon}}$$
 1.47

In free-space, we have $v_p = 1/\sqrt{\mu_0 \epsilon_0} = c = 2.998 \times 10^8$ m/sec, which is the speed of light.

The wavelength, λ , is defined as the distance between two successive maxima (or minima, or any other reference points) on the wave, at a fixed instant of time. Thus,

$$[\omega l - kz] - [\omega l - k(z + \lambda)] = 2\pi,$$

$$\lambda = \frac{2\pi}{k} = \frac{2\pi v_p}{\omega} = \frac{v_p}{f}.$$
 1.48

so,

A complete specification of the plane wave electromagnetic field must include the magnetic field. In general, whenever E or \tilde{H} is known, the other field vector can be readily found by using one of Maxwell's curl equations. Thus, applying (1.41a) to the electric field of (1.45) gives $H_x = H_z = 0$, and

$$H_{y} = \frac{1}{\eta} [E^{+} r^{-jkz} - E^{-} e^{jkz}], \qquad 1.49$$

where $\eta = \omega \mu / k = \sqrt{\mu / \epsilon}$ is the wave impedance for the plane wave, defined as the ratio of the \bar{E} and \bar{H} fields. For plane waves, this impedance is also the intrinsic impedance of the medium. In free-space we have $\eta_0 = \sqrt{\mu_0 / \epsilon_0} = 377 \ \Omega$. Note that the \bar{E} and \bar{H} vectors are orthogonal to each other and orthogonal to the direction of propagation $(\pm \hat{z})$; this is a characteristic of transverse electromagnetic (TEM) waves.



EXAMPLE 1.1 Basic Plane Wave Parameters

A plane wave with a frequency of 3 GHz is propagating in an unbounded material with $\epsilon_r = 7$ and $\mu_r = 3$. Compute the wavelength, phase velocity, and wave impedance for this wave.

Solution From (1.47) the phase velocity is

$$v_p = \frac{1}{\sqrt{\mu\epsilon}} = \frac{c}{\sqrt{\mu_r\epsilon_r}} = \frac{3 \times 10^8}{\sqrt{(7)(3)}} = 6.55 \times 10^7 \text{ m/sec.}$$

This is slower than the speed of light in free-space by a factor of $\sqrt{21} = 4.58$. From (1.48) the wavelength is

$$\lambda = \frac{v_p}{f} = \frac{6.55 \times 10^7}{3 \times 10^9} = 0.0218 \text{ m}$$

The wave impedance is

$$\eta = \sqrt{\frac{\mu}{\epsilon}} = \eta_0 \sqrt{\frac{\mu_r}{\epsilon_r}} = 377 \sqrt{\frac{3}{7}} = 246.8 \,\Omega.$$

Plane Waves in a General Lossy Medium

Now consider the effect of a lossy medium. If the medium is conductive, with a conductivity, σ , Maxwell's curl equations can be written, from (1.41a) and (1.20) as

$$\nabla \times \bar{E} = -j\omega\mu\bar{H}, \qquad 1.50a$$

$$\nabla \times \bar{H} = j\omega\epsilon \bar{E} + \sigma \bar{E}, \qquad 1.50b$$

The resulting wave equation for \bar{E} then becomes

$$\nabla^2 \bar{E} + \omega^2 \mu \epsilon \left(1 - j \frac{\sigma}{\omega \epsilon}\right) \bar{E} = 0, \qquad 1.51$$

where we see a similarity with (1.42), the wave equation for \overline{E} in the lossless case. The difference is that the wavenumber $k^2 = \omega^2 \mu \epsilon$ of (1.42) is replaced by $\omega^2 \mu \epsilon [1 - j(\sigma/\omega \epsilon)]$ in (1.51). We then define a complex propagation constant for the medium as

$$\gamma = \alpha + j\beta = j\omega\sqrt{\mu\epsilon}\sqrt{1 - j\frac{\sigma}{\omega\epsilon}}$$
 1.52

If we again assume an electric field with only an \hat{x} component and uniform in x and y, the wave equation of (1.51) reduces to

$$\frac{\partial^2 E_x}{\partial z^2} - \gamma^2 E_x = 0, \qquad 1.53$$

which has solutions

$$E_x(z) = E^+ e^{-\gamma z} + E^- e^{\gamma z}.$$
 1.54

The positive traveling wave then has a propagation factor of the form

$$e^{-\gamma z}=e^{-\alpha z}e^{-j\beta z},$$

which in the time domain is of the form

$$e^{-\alpha z} \cos(\omega t - \beta z).$$

We see then that this represents a wave traveling in the +z direction with a phase velocity $v_p = \omega/\beta$, a wavelength $\lambda = 2\pi/\beta$, and an exponential damping factor. The rate of decay with distance is given by the attenuation constant, α . The negative traveling wave term of (1.54) is similarly damped along the -z axis. If the loss is removed, $\sigma = 0$, and we have $\gamma = jk$ and $\alpha = 0$, $\beta = k$.

As discussed in Section 1.3, loss can also be treated through the use of a complex permittivity. From (1.20) with $\sigma = 0$ but $\epsilon = \epsilon' - j\epsilon''$ complex, we have that

$$\gamma = j\omega\sqrt{\mu\epsilon} = jk = j\omega\sqrt{\mu\epsilon'(1-j\,\tan\,\delta)}, \qquad 1.55$$

where $\tan \delta = \epsilon''/\epsilon'$ is the loss tangent of the material.

Next, the associated magnetic field can be calculated as

$$H_{y} = \frac{j}{\omega\mu} \frac{\partial E_{x}}{\partial z} = \frac{-j\gamma}{\omega\mu} (E^{+}e^{-\gamma z} - E^{-}e^{\gamma z}). \qquad 1.56$$

As with the lossless case, a wave impedance can be defined to relate the electric and magnetic fields:

$$\eta = \frac{j\omega\mu}{\gamma}.$$
 1.57

Then (1.56) can be rewritten as

$$H_{y} = \frac{1}{\eta} (E^{+} e^{-\gamma z} - E^{-} e^{\gamma z}), \qquad 1.58$$

Note that η is. in general, complex and reduces to the lossless case of $\eta = \sqrt{\mu/\epsilon}$ when $\gamma = jk = j\omega\sqrt{\mu\epsilon}$.

Plane Waves in a Good Conductor

Many problems of practical interest involve loss or attenuation due to good (but not perfect) conductors. A good conductor is a special case of the preceding analysis, where the conductive current is much greater than the displacement current, which means $\sigma \gg \omega \epsilon$. Most metals can be categorized as good conductors. In terms of a complex ϵ , rather than conductivity, this condition is equivalent to $\epsilon'' \gg \epsilon'$. The propagation constant of (1.52) can then be adequately approximated by ignoring the displacement current term, to give

$$\gamma = \alpha + j\beta \simeq j\omega\sqrt{\mu\epsilon}\sqrt{\frac{\sigma}{j\omega\epsilon}} = (1+j)\sqrt{\frac{\omega\mu\sigma}{2}}.$$
 (.59)

The skin depth, or characteristic depth of penetration, is defined as

$$\delta_s = \frac{1}{\alpha} = \sqrt{\frac{2}{\omega\mu\sigma}}.$$
 1.60

Then the amplitude of the fields in the conductor decay by an amount 1/e or 36.8%, after traveling a distance of one skin depth, since $e^{-\alpha z} = e^{-\alpha b_{\alpha}} = e^{-1}$. At microwave frequencies, for a good conductor, this distance is very small. The practical importance of this result is that only a thin plating of a good conductor (e.g., silver or gold) is necessary for low-loss microwave components.

P

EXAMPLE 1.2 Skin Depth at Microwave Frequencies

Compute the skin depth of aluminum, copper, gold, and silver at a frequency of 10 GHz.

Solution

The conductivities for these metals are listed in Appendix F. Equation (1.60) gives the skin depths as

$$\delta_{s} = \sqrt{\frac{2}{\omega\mu\sigma}} = \sqrt{\frac{1}{\pi f\mu_{0}\sigma}} = \sqrt{\frac{1}{\pi(10^{10})(4\pi \times 10^{-7})}} \sqrt{\frac{1}{\sigma}}$$
$$= 5.03 \times 10^{-3} \sqrt{\frac{1}{\sigma}}.$$

For aluminum: $\delta_s = 5.03 \times 10^{-3} \sqrt{\frac{1}{3.816 \times 10^7}} = 8.14 \times 10^{-7} \text{ m.}$ For copper: $\delta_s = 5.03 \times 10^{-3} \sqrt{\frac{1}{5.813 \times 10^7}} = 6.60 \times 10^{-7} \text{ m.}$ For gold: $\delta_s = 5.03 \times 10^{-3} \sqrt{\frac{1}{4.098 \times 10^7}} = 7.86 \times 10^{-7} \text{ m.}$ For silver: $\delta_s = 5.03 \times 10^{-3} \sqrt{\frac{1}{6.173 \times 10^7}} = 6.40 \times 10^{-7} \text{ m.}$

These results show that most of the current flow in a good conductor occurs in an extremely thin region near the surface of the conductor.

The wave impedance inside a good conductor can be obtained from (1.57) and (1.59). The result is

$$\eta = \frac{j\omega\mu}{\gamma} \simeq (1+j)\sqrt{\frac{\omega\mu}{2\sigma}} = (1+j)\frac{1}{\sigma\delta_s}.$$
 I.61

Notice that the phase angle of this impedance is 45° , a characteristic of good conductors. The phase angle of the impedance for a lossless material is 0° , and the phase angle of the impedance of an arbitrary lossy medium is somewhere between 0° and 45° .

Table 1.1 summarizes the results for plane wave propagation in lossless and lossy homogeneous media.
		Type of Medium	
Quantity	$Lossless (\epsilon'' = \sigma = 0)$	General Lossy	Good Conductor $\varepsilon'' > > \epsilon' \text{ or } \sigma > > \omega \epsilon'$
Complex propagation constant	$\gamma = j\omega\sqrt{\mu\epsilon}$	$\gamma = j \omega \sqrt{\mu \epsilon}$	$\gamma = (1+j)\sqrt{\omega\mu\sigma/2}$
		$= j \omega \sqrt{\mu \epsilon'} \sqrt{(1-j\sigma)/\omega \epsilon'}$	
Phase constant (wavenumber)	$d = k = \omega \sqrt{\mu k}$	$\vartheta = \operatorname{Im}(\gamma)$	$\beta = \operatorname{Im}(\gamma) = \sqrt{\omega\mu\sigma/2}$
Attenuation constant	$\alpha = 0$	$\alpha = \operatorname{Re}(\gamma)$	$\alpha = \operatorname{Re}(\gamma) = \sqrt{\omega\mu\sigma/2}$
Impedance	$\eta = \sqrt{\mu/\epsilon} = \omega \mu/k$	$\eta = j \omega \mu / \gamma$	$\eta = (1+j)\sqrt{\omega\mu/2a}$
Skin depth	$\delta_s = \infty$	$\delta_s = 1/lpha$	$\delta_s=\sqrt{2/\omega\mu\sigma}$
Wavelength	$\lambda=2\pi/eta$	$\lambda=2\pi/\beta$	$\lambda = 2\pi/\vartheta$
Phase velocity	$v_p = \omega/\beta$	$v_p \Rightarrow \omega/\beta$	$v_p = \omega/\beta$

TABLE 1.1 Summary of Results for Plane Wave Propagation in Various Media

1.5 GENERAL PLANE WAVE SOLUTIONS

5

Some specific features of plane waves were discussed in Section 1.4. Here we will look at plane waves again, from a more general point of view, and solve the wave equation by the method of separation of variables. This technique will find application in succeeding chapters. We will also discuss circularly polarized plane waves, which will be important for the discussion of ferrites in Chapter 9.

In free space, the Helmholtz equation for \tilde{E} can be written as

$$\nabla^2 \bar{E} + k_0^2 \bar{E} = \frac{\partial^2 \bar{E}}{\partial x^2} + \frac{\partial^2 \bar{E}}{\partial y^2} + \frac{\partial^2 \bar{E}}{\partial z^2} + k_0^2 \bar{E} = 0, \qquad 1.62$$

and this vector wave equation holds for each rectangular component of E:

$$\frac{\partial^2 E_i}{\partial x^2} + \frac{\partial^2 E_i}{\partial y^2} + \frac{\partial^2 E_i}{\partial z^2} + k_0^2 E_i = 0.$$
 1.63

where the index i = x, y, or z. This equation will now be solved by the method of separation of variables, a standard technique for treating such partial differential equations. The method begins by assuming that the solution to (1.63) for, say E_x , can be written as a product of three functions for each of the three coordinates:

$$E_x(x, y, z) = f(x) g(y) h(z).$$
 1.64

Substituting this form into (1.63) and dividing by fgh gives

$$\frac{f''}{f} + \frac{g''}{g} + \frac{h''}{h} + k_0^2 = 0, \qquad 1.65$$

21

where the double primes denote the second derivative. Now the key step in the argument is to recognize that each of the terms in (1.65) must be equal to a constant, since they are independent of each other. That is, f''/f is only a function of x, and the remaining terms in (1.65) do not depend on x, so f''/f must be a constant, and similarly for the other terms in (1.65). Thus, we define three separation constants, k_x , k_y and k_z , such that

$$f''/f = -k_x^2; \qquad g''/g = -k_y^2; \qquad h''/h = -k_z^2;$$
$$\frac{d^2f}{dx^2} + k_x^2f = 0; \qquad \frac{d^2g}{dy^2} + k_y^2g = 0; \qquad \frac{d^2h}{dz^2} + k_z^2h = 0. \qquad 1.66$$

or

Combining (1.65) and (1.66) shows that

$$k_x^2 + k_y^2 + k_z^2 = k_0^2. 1.67$$

The partial differential equation of (1.63) has now been reduced to three separate ordinary differential equations in (1.66). Solutions to these equations are of the form $e^{\pm_j k_x x}$, $e^{\pm_j k_y y}$, and $e^{\pm_j k_z z}$, respectively. As we have seen in the previous section, the terms with + signs result in waves traveling in the negative x, y, or z direction, while the terms with - signs result in waves traveling in the positive direction. Both solutions are possible and are valid; the amount to which these various terms are excited is dependent on the source of the fields. For our present discussion, we will select a plane wave traveling in the positive direction for each coordinate, and write the complete solution for E_x as

$$E_x(x, y, z) = A e^{-j(k_x x + k_y y + k_z z)}, \qquad 1.68$$

where A is an arbitrary amplitude constant. Now define a wavenumber vector \bar{k} as

$$\bar{k} = k_x \hat{x} + k_y \hat{y} + k_z \hat{z} = k_0 \hat{n}.$$
 1.69

Then from (1.67) $|\bar{k}| = k_0$, and so \hat{n} is a unit vector in the direction of propagation. Also define a position vector as

$$\ddot{r} = x\hat{x} + y\hat{y} + z\hat{z}; \qquad 1.70$$

then (1.68) can be written as

$$E_x(x, y, z) = Ae^{-jk\cdot\dot{r}}.$$
1.71

Solutions to (1.63) for E_y and E_z are, of course, similar in form to E_x of (1.71), but with different amplitude constants:

$$E_{\boldsymbol{y}}(\boldsymbol{x},\boldsymbol{y},\boldsymbol{z}) = Be^{-j\boldsymbol{k}\cdot\boldsymbol{r}}, \qquad 1.72$$

$$E_z(x, y, z) = C e^{-jk\cdot t}.$$

The x, y, and z dependences of the three components of \overline{E} in (1.71)-(1.73) must be the same (same k_x, k_y, k_z), because the divergence condition that

$$abla \cdot ar{E} = rac{\partial E_x}{\partial x} + rac{\partial E_y}{\partial y} + rac{\partial E_z}{\partial z} = 0$$

must also be applied in order to satisfy Maxwell's equations, which implies that $E_{xx}E_{yx}$ and E_z must each have the same variation in x, y, and z. (Note that the solutions in the preceding section automatically satisfied the divergence condition, since E_x was the only component of \bar{E} , and E_x did not vary with x.) This condition also imposes a constraint on the amplitudes A, B, and C, since if

$$\bar{E}_0 = A\hat{x} + B\hat{y} + C\hat{z},$$

 $\bar{E} = \bar{E}_0 e^{-j\bar{k}\cdot\bar{r}}.$

we have

and

$$\nabla \cdot \bar{E} = \nabla \cdot (\bar{E}_0 e^{-j\bar{k}\cdot\bar{r}}) = \bar{E}_0 \cdot \nabla e^{-j\bar{k}\cdot\bar{r}} = -j\bar{k}\cdot \bar{E}_0 e^{-j\bar{k}\cdot\bar{r}} = 0,$$

where vector identity (B.7) was used. Thus, we must have

$$\bar{k} \cdot \bar{E}_0 = 0, \qquad 1.74$$

which means that the electric field amplitude vector \overline{E}_0 must be perpendicular to the direction of propagation, \overline{k} . This condition is a general result for plane waves and implies that only two of the three amplitude constants, A, B, and C, can be chosen independently.

The magnetic field can be found from Maxwell's equation,

$$\nabla \times \vec{E} = -j\omega\mu_0 \vec{H}, \qquad 1.75$$

to give

$$\begin{split} \vec{H} &= \frac{j}{\omega \mu_0} \nabla \times \vec{E} = \frac{j}{\omega \mu_0} \nabla \times (\vec{E}_0 e^{-j\vec{k} \cdot \vec{r}}) \\ &= \frac{-j}{\omega \mu_0} \vec{E}_0 \times \nabla e^{-j\vec{k} \cdot \vec{r}} \\ &= \frac{-j}{\omega \mu_0} \vec{E}_0 \times (-j\vec{k}) e^{-j\vec{k} \cdot \vec{r}} \\ &= \frac{k_0}{\omega \mu_0} \vec{n} \times \vec{E}_0 e^{-j\vec{k} \cdot \vec{r}} \\ &= \frac{1}{\eta_0} \vec{n} \times \vec{E}_0 e^{-j\vec{k} \cdot \vec{r}} \\ &= \frac{1}{\eta_0} \vec{n} \times \vec{E}_s \end{split}$$
 1.76

where vector identity (B.9) was used in obtaining the second line. This result shows that the magnetic field intensity vector \bar{H} lies in a plane normal to \bar{k} , the direction of propagation, and that \bar{H} is perpendicular to \bar{E} . See Figure 1.8 for an illustration of these vector relations. The quantity $\eta_0 = \sqrt{\mu_0/\epsilon_0} = 377 \ \Omega$ in (1.76) is the intrinsic impedance of free-space.



FIGURE 1.8 Orientation of the \bar{E} , \bar{H} , and $\bar{k} = k_0 \hat{n}$ vectors for a plane wave.

The time-domain expression for the electric field can be found as

$$\bar{\mathcal{E}}(x, y, z, t) = \operatorname{Re}\{\bar{E}(x, y, z)e^{j\omega t}\}$$

$$= \operatorname{Re}\{\bar{E}_0 e^{-j\vec{k}\cdot\vec{r}}e^{j\omega t}\}$$

$$= \bar{E}_0 \cos(\bar{k}\cdot\vec{r} - \omega t), \qquad 1.77$$

assuming that the amplitude constants A, B, and C contained in \overline{E}_0 are real. If these constants are not real, their phases should be included inside the cosine term of (1.77).

From (1.77) we see that a wavelength, defined as the distance the wave must travel to undergo a phase shift of 2π , can be found as

$$|ar{k}|\lambda_0=k_0\lambda_0=2\pi$$

or $\lambda_0 = 2\pi/k_0$, which is identical to the result obtained in Section 1.4. The phase velocity, or the speed at which one would have to travel to maintain a constant phase point on the wave, can be found from the condition that

 $\bar{k}\cdot\bar{r}-\omega t=$ constant.

Taking the derivative with respect to time gives

$$k_0 \frac{dr}{dt} - \omega = 0,$$

$$k_0 v_p = \omega,$$

or

so that

$$v_p = \omega/k_0 = \frac{1}{\sqrt{\mu_0 \epsilon_0}} = c = 2.998 \times 10^8 \text{ m/sec},$$

as in Section 1.4.



EXAMPLE 1.3 Current Sheets as Sources of Plane Waves

An infinite sheet of surface current can be considered as a source for plane waves. If an electric surface current density $\bar{J}_s = J_o \hat{x}$ exists on the z = 0 plane in free-space, find the resulting fields by assuming plane waves on either side of the current sheet and enforcing the boundary conditions.

24

Solution

Since the source does not vary with x or y, the fields will not vary with x or y but will propagate away from the source in the $\pm z$ direction. The boundary conditions to be satisfied at z = 0 are

$$\hat{\boldsymbol{n}} \times (\bar{\boldsymbol{E}}_2 - \boldsymbol{E}_1) = \hat{\boldsymbol{z}} \times (\boldsymbol{E}_2 - \boldsymbol{E}_1) = 0.$$
$$\hat{\boldsymbol{n}} \times (\bar{\boldsymbol{H}}_2 - \bar{\boldsymbol{H}}_1) = \hat{\boldsymbol{z}} \times (\bar{\boldsymbol{H}}_2 - \bar{\boldsymbol{H}}_1) = J_{\boldsymbol{a}} \hat{\boldsymbol{x}},$$

where \vec{E}_1 , \vec{H}_1 are the fields for z < 0, and \vec{E}_2 , \vec{H}_2 are the fields for z > 0. To satisfy the second condition, \vec{H} must have a \hat{y} component. Then for \vec{E} to be orthogonal to \vec{H} and \hat{z} , \vec{E} must have an \hat{r} component. Thus the fields will have the following form:

for
$$z < 0$$
, $\bar{E}_1 = \hat{x} A \eta_0 e^{j k_0 z}$,
 $\bar{H}_1 = -\hat{y} A e^{j k_0 z}$,
for $z > 0$. $\bar{E}_2 = \hat{x} B \eta_0 e^{-j k_0 z}$,
 $\bar{H}_2 = \hat{y} B e^{-j k_0 z}$.

where A and B are arbitrary amplitude constants. The first boundary condition, that E_x is continuous at z = 0, yields A = B, while the boundary condition for \overline{H} yields the equation

$$-B - A = J_o,$$

Solving for A. B gives

$$A = B = -J_o/2.$$

which completes the solution.

Circularly Polarized Plane Waves

The plane waves discussed above all had their electric field vector pointing in a fixed direction and so are called linearly polarized waves. In general, the polarization of a plane wave refers to the orientation of the electric field vector, which may be in a fixed direction or may change with time.

Consider the superposition of an \hat{x} linearly polarized wave with amplitude E_1 , and a \hat{y} linearly polarized wave with amplitude E_2 , both traveling in the positive \hat{z} direction. The total electric field can be written as

$$\bar{E} = (E_1 \hat{x} + E_2 \hat{y}) e^{-jk_0 z}.$$
1.78

Ò

A number of possibilities now arise. If $E_1 \neq 0$ and $E_2 = 0$, we have a plane wave linearly polarized in the \hat{x} direction. Similarly, if $E_1 = 0$ and $E_2 \neq 0$, we have a plane wave linearly polarized in the \hat{y} direction. If E_1 and E_2 are both real and nonzero, we have a plane wave linearly polarized at the angle

$$\phi = \tan^{-1} \frac{E_2}{E_1}$$

For example, if $E_1 = E_2 = E_0$, we have

$$ar{E}=E_0(\hat{x}+\hat{y})e^{-jk_0z}$$
 ,

which represents an electric field vector at a 45° angle from the x-axis.

Now consider the case in which $E_1 = jE_2 = E_0$, where E_0 is real, so that

$$\bar{E} = E_0(\hat{x} - j\hat{y})e^{-jk_0z}.$$
1.79

The time domain form of this field is

$$\bar{\mathcal{E}}(z,t) = E_0\{\hat{x}\,\cos(\omega t - k_0 z) + \hat{y}\,\cos(\omega t - k_0 z - \pi/2)\}.$$
1.80

This expression shows that the electric field vector changes with time or, equivalently, with distance along the z-axis. To see this, pick a fixed position, say z = 0. Equation (1.80) then reduces to

$$\vec{\mathcal{E}}(0,t) = E_0\{\hat{x}\,\cos\,\omega t + \hat{y}\,\sin\,\omega t\},\qquad 1.81$$

so as ωt increases from zero, the electric field vector rotates counterclockwise from the x-axis. The resulting angle from the x-axis of the electric field vector at time t, at z = 0, is then

$$\phi = \tan^{-1}\left(\frac{\sin \,\omega t}{\cos \,\omega t}\right) = \omega t,$$

which shows that the polarization rotates at the uniform angular velocity ω . Since the fingers of the right hand point in the direction of rotation when the thumb points in the direction of propagation, this type of wave is referred to as a right hand circularly polarized (RHCP) wave. Similarly, a field of the form

$$\tilde{E} = E_0(\hat{x} + j\hat{y})e^{-jk_0z}$$
 1.82

constitutes a left hand circularly polarized (LHCP) wave, where the electric field vector rotates in the opposite direction. See Figure 1.9 for a sketch of the polarization vectors for RHCP and LHCP plane waves.

The magnetic field associated with a circularly polarized wave may be found from Maxwell's equations, or by using the wave impedance applied to each component of the electric field. For example, applying (1.76) to the electric field of a RHCP wave as given in (1.79) yields

$$\bar{H} = \frac{E_0}{\eta_0} \hat{z} \times (\hat{x} - j\hat{y}) e^{-jk_0 z} = \frac{E_0}{\eta_0} (\hat{y} + j\hat{x}) e^{-jk_0 z} = \frac{jE_0}{\eta_0} (\hat{x} - j\hat{y}) e^{-jk_0 z},$$

which is also seen to be a circularly polarized wave with a polarization vector of the RHCP sense.

1.6 ENERGY AND POWER

In general, a source of electromagnetic energy sets up fields that store electric and magnetic energy and carry power that may be transmitted or dissipated as loss. In the



FIGURE 1.9 Electric field polarization for (a) RHCP and (b) LHCP plane waves.

sinusoidal steady-state case, the time-average stored electric energy in a volume V is given by.

$$W_e = \frac{1}{4} \operatorname{Re} \int_V \vec{E} \cdot \vec{D}^* \, dv. \qquad 1.83$$

which in the case of simple lossless isotropic, homogeneous, linear media, where ϵ is a real scalar constant, reduces to

$$W_{e} = \frac{\epsilon}{4} \int_{\boldsymbol{V}} \tilde{E} \cdot \tilde{E}^{*} dv. \qquad 1.84$$

Similarly, the time-average magnetic energy stored in the volume V is

$$W_m = \frac{1}{4} \operatorname{Re} \int_{V} \tilde{H} \cdot \tilde{B}^* \, dv, \qquad 1.85$$

which becomes

$$W_m = \frac{\mu}{4} \int_V \bar{H} \cdot \bar{H}^* \, dv, \qquad 1.86$$

for a real, constant, scalar μ .

We can now derive Poynting's theorem, which leads to energy conservation for electromagnetic fields and sources. If we have an electric source current, \bar{J}_s , and a conduction current $\sigma \bar{E}$, as defined in (1.19), then the total electric current density is $\bar{J} = \bar{J}_s + \sigma \bar{E}$. Then multiplying (1.27a) by \bar{H}^* , and multiplying the conjugate of (1.27b) by \bar{E} , yields

$$\bar{H}^* \cdot (\nabla \times \bar{E}) = -j\omega\mu |\bar{H}|^2 - \bar{H}^* \cdot \bar{M}_s.$$
$$\bar{E} \cdot (\nabla \times \bar{H}^*) = \bar{E} \cdot \bar{J}^* - j\omega\epsilon^* |\bar{E}|^2 = \bar{E} \cdot \bar{J}^*_s + \sigma |\bar{E}|^2 - j\omega\epsilon^* |\bar{E}|^2$$

where \bar{M}_s is the magnetic source current. Using these two results in vector identity (B.8)

gives

$$\begin{aligned} \nabla \cdot (\bar{E} \times \bar{H}^*) &= \bar{H}^* \cdot (\nabla \times \bar{E}) - \bar{E} \cdot (\nabla \times \bar{H}^*) \\ &= -\sigma |\bar{E}|^2 + j\omega (\epsilon^* |\bar{E}|^2 - \mu |\bar{H}|^2) - (\bar{E} \cdot \bar{J}^*_s + \bar{H}^* \cdot \bar{M}_s). \end{aligned}$$

Now integrate over a volume V and use the divergence theorem:

$$\int_{V} \nabla \cdot (\bar{E} \times \bar{H}^{*}) dv = \oint_{S} \bar{E} \times \bar{H}^{*} \cdot d\bar{s}$$
$$= -\sigma \int_{V} |\bar{E}|^{2} dv + j\omega \int_{V} (\epsilon^{*} |\bar{E}|^{2} - \mu |\bar{H}|^{2}) dv - \int_{V} (\bar{E} \cdot \bar{J}_{s}^{*} + \bar{H}^{*} \cdot \bar{M}_{s}) dv, \quad 1.87$$

where S is a closed surface enclosing the volume V, as shown in Figure 1.10. Allowing $\epsilon = \epsilon' - j\epsilon''$ and $\mu = \mu' - j\mu''$ to be complex to allow for loss, and rewriting (1.87) gives

$$-\frac{1}{2}\int_{V}(\hat{E}\cdot\bar{J}_{s}^{*}+\hat{H}^{*}\cdot\bar{M}_{s})\,dv = \frac{1}{2}\oint_{S}\bar{E}\times\bar{H}^{*}\cdot d\bar{s} +\frac{\sigma}{2}\int_{V}|\bar{E}|^{2}\,dv + \frac{\omega}{2}\int_{V}(\epsilon''|\bar{E}|^{2}+\mu''|\bar{H}|^{2})\,dv + j\frac{\omega}{2}\int_{V}(\mu'|\bar{H}|^{2}-\epsilon'|\bar{E}|^{2})\,dv.$$
1.88

This result is known as Poynting's theorem, after the physicist J. H. Poynting (1852–1914), and is basically a power balance equation. Thus, the integral on the left-hand side represents the complex power, P_s , delivered by the sources \bar{J}_s and \bar{M}_s , inside S:

$$P_{s} = -\frac{1}{2} \int_{V} (\bar{E} \cdot \bar{J}_{s}^{*} + \bar{H}^{*} \cdot \bar{M}_{s}) \, dv. \qquad 1.89$$

The first integral on the right-hand side of (1.88) represents complex power flow out of the closed surface S. If we define a quantity called the Poynting vector, \vec{S} , as

$$\tilde{S} = \tilde{E} \times \bar{H}^*, \qquad 1.90$$

then this power can be expressed as

$$P_o = \frac{1}{2} \oint_S \bar{E} \times \bar{H}^* \cdot d\bar{s} = \frac{1}{2} \oint_S \bar{S} \cdot d\bar{s}.$$
 1.91



FIGURE 1.10 A volume V, enclosed by the closed surface S, containing fields \vec{E} , \vec{H} , and current sources \vec{J}_s , \vec{M}_s .

The surface S in (1.91) must be a closed surface in order for this interpretation to be valid. The real parts of P_s and P_p in (1.89) and (1.91) represent time-average powers.

The second and third integrals in (1.88) are real quantities representing the timeaverage power dissipated in the volume V due to conductivity, dielectric, and magnetic losses. If we define this power as P_t we have that

$$P_{t} = \frac{\sigma}{2} \int_{V} |\hat{E}|^{2} dv + \frac{\omega}{2} \int_{V} (\epsilon'' |\bar{E}|^{2} + \mu'' |\bar{H}|^{2}) dv, \qquad 1.92$$

which is sometimes referred to as Joule's law. The last integral in (1.88) can be seen to be related to the stored electric and magnetic energies, as defined in (1.84) and (1.86),

With the above definitions, Poynting's theorem can be rewritten as

$$P_{s} = P_{o} + P_{f} + 2j\omega(W_{m} - W_{c}), \qquad 1.93$$

In words, this complex power balance equation states that the power delivered by the sources (P_s) is equal to the sum of the power transmitted through the surface (P_a) , the power lost to heat in the volume (P_t) , and 2ω times the net reactive energy stored in the volume.

Power Absorbed by a Good Conductor

To calculate attenuation and loss due to an imperfect conductor, one must find the power dissipated in the conductor. We will show that this can be done using only the fields at the surface of the conductor, which is a very helpful simplification when calculating attenuation.

Consider the geometry of Figure 1.11, which shows the interface between a tossless medium and a good conductor. We assume that a field is incident from z < 0 and that the field penetrates into the conducting region z > 0. The real average power entering the conductor volume defined by the cross-sectional surface S_0 at the interface and the



FIGURE 1.11

An interface between a lossless medium and a good conductor with a closed surface $S_0 + S$ for computing the power dissipated in the conductor.

surface S is given from (1.91) as

$$P_{\rm av} = \frac{1}{2} \operatorname{Re} \int_{S_0+S} \tilde{E} \times \tilde{H}^* \cdot \hat{n} \, ds, \qquad 1.94$$

where \hat{n} is a unit normal vector pointing into the closed surface $S_0 + S$, and \bar{E} . \bar{H} are the fields at this surface. The contribution to the integral in (1.94) from the surface S can be made zero by proper selection of this surface. For example, if the field is a normally incident plane wave, the Poynting vector $\bar{S} = \bar{E} \times \bar{H}^*$ will be in the \hat{z} direction, and so tangential to the top, bottom, front, and back of S, if these walls are made parallel to the z-axis. If the wave is obliquely incident, these walls can be slanted to obtain the same result. And, if the conductor is good, the decay of the fields from the interface at z = 0 will be very rapid, so that the right-hand end of S can be made far enough away from z = 0 so that there is negligible contribution to the integral from this part of the surface S. The power entering the conductor through S_0 can then be written as

$$P_{\rm av} = \frac{1}{2} \operatorname{Re} \, \int_{S_0} \bar{E} \times \bar{H}^* \cdot \hat{z} \, ds. \qquad 1.95$$

From vector identity (B.3) we have

$$\hat{z} \cdot (\bar{E} \times \bar{H}^*) = (\hat{z} \times \bar{E}) \cdot \bar{H}^* = \eta \bar{H} \cdot \bar{H}^*,$$
 1.96

since $\hat{H} = \hat{n} \times \tilde{E}/\eta$, as generalized from (1.76) for conductive media, where η is the intrinsic wave impedance of the conductor. Equation (1.95) can then be written as

$$P_{\rm av} = \frac{R_s}{2} \int_{S_0} |\tilde{H}|^2 ds, \qquad 1.97$$

where

$$R_s = \operatorname{Re}(\eta) = \operatorname{Re}\left[(1+j)\sqrt{\frac{\omega\mu}{2\sigma}}\right] = \sqrt{\frac{\omega\mu}{2\sigma}} = \frac{1}{\sigma\delta_s}$$

is called the surface resistivity of the conductor. The magnetic field \bar{H} in (1.97) is tangential to the conductor surface and needs only to be evaluated at the surface of the conductor; since H_t is continuous at z = 0, it doesn't matter whether this field is evaluated just outside the conductor or just inside the conductor. In the next section we will show how (1.97) can be evaluated in terms of a surface current density flowing on the surface of the conductor, where the conductor can be assumed to be perfect.

1.7 PLANE WAVE REFLECTION FROM A MEDIA INTERFACE

A number of problems to be considered in later chapters involve the behavior of electromagnetic fields at the interface of a lossy or conducting medium, and so it is beneficial at this time to study the reflection of a plane wave normally incident from free-space onto the surface of a conducting half-space. The geometry is shown in Figure 1.12 where the lossy half-space z > 0 is characterized by the parameters ϵ, μ , and σ .



FIGURE 1.12 Plane wave reflection from a lossy medium; normal incidence.

General Medium

With no loss of generality, we can assume that the incident plane wave has an electric field vector oriented along the x-axis and is propagating along the positive z-axis. The incident fields can then be written, for z < 0, as

$$\hat{E}_i = \hat{x} E_0 e^{-jk_0 x}. \tag{1.99a}$$

$$\bar{H}_{i} = \hat{y} \frac{1}{320} E_{0} e^{-jk_{0}z}, \qquad 1.99b$$

where η_0 is the wave impedance of free-space, and E_0 is an arbitrary amplitude. Also in the region z < 0, a reflected wave may exist with the form

$$\tilde{E}_r = \hat{x} \Gamma E_0 e^{+jk_0 z}, \qquad 1.100a$$

$$\bar{H}_{r} = -\hat{y} \frac{\Gamma}{\eta_{0}} E_{0} e^{+\hat{y} k_{0} x}, \qquad 1.100b$$

where Γ is the unknown reflection coefficient of the electric field. Note that in (1.100), the sign in the exponential terms has been chosen as positive, to represent waves traveling in the $-\hat{z}$ direction of propagation, as derived in (1.49). This is also consistent with the Poynting vector $\bar{S}_r = \bar{E}_r \times \bar{H}_r^* = -|\Gamma|^2 |E_0|^2 \hat{z}/\eta_0$, which shows power to be traveling in the $-\hat{z}$ direction.

As shown in Section 1.4, from equations (1.54) and (1.58), the transmitted fields for z > 0 in the lossy medium can be written as

$$\bar{E}_t = \hat{z} T E_0 e^{-\gamma z}, \qquad 1.101a$$

$$\bar{H}_t = \frac{\hat{y}TE_0}{\eta}e^{-\gamma z}, \qquad 1.101b$$

where T is the transmission coefficient of the electric field and η is the intrinsic impedance

of the lossy medium in the region z > 0. From (1.57) and (1.52) the intrinsic impedance is

$$\eta = \frac{j\omega\mu}{\gamma}, \qquad 1.102$$

and the propagation constant is

$$\gamma = \alpha + j\beta = j\omega\sqrt{\mu\epsilon}\sqrt{1 - j\sigma/\omega\epsilon}.$$
 1.103

We now have a boundary value problem where the general form of the fields are known via (1.99)-(1.101) on either side of the material discontinuity at z = 0. The two unknown constants, Γ and T, are found by applying two boundary conditions on E_x and H_y at z = 0. Since these tangential field components must be continuous at z = 0, we arrive at the following two equations:

$$(1+1) = T, 1.104a$$

$$\frac{1-\Gamma}{\eta_0} = \frac{T}{\eta}.$$
 1.104b

Solving these equations for the reflection and transmission coefficients gives

$$\Gamma = \frac{\eta - \eta_0}{\eta + \eta_0}, \qquad \qquad 1.105a$$

$$T = 1 + \Gamma = \frac{2\eta}{\eta + \eta_0}.$$
 1.105b

This is a general solution for reflection and transmission of a normally incident wave at the interface of a lossy material, where η is the impedance of the material. We now consider three special cases of the above result.

Lossless Medium

If the region for z > 0 is a lossless dielectric, then $\sigma = 0$, and μ and ϵ are real quantities. The propagation constant in this case is purely imaginary and can be written as

$$\gamma = j\beta = j\omega\sqrt{\mu\epsilon} = jk_0\sqrt{\mu_r\epsilon_r}, \qquad 1.106$$

where $k_0 = \omega \sqrt{\mu_0 \epsilon_0}$ is the wavenumber of a plane wave in free-space. The wavelength in the dielectric is

$$\lambda = \frac{2\pi}{\beta} = \frac{2\pi}{\omega\sqrt{\mu\epsilon}} = \frac{\lambda_0}{\sqrt{\mu_r\epsilon_r}},\qquad 1.107$$

which is seen to be shorter than the wavelength in free-space (λ_0). The corresponding phase velocity is

$$v_p = \frac{\omega}{\beta} = \frac{1}{\sqrt{\mu\epsilon}} = \frac{c}{\sqrt{\mu_r\epsilon_r}},$$
 1.108

which is slower than the speed of light in free-space (c). The wave impedance of the dielectric is

$$\eta = \frac{j\omega\mu}{\gamma} = \sqrt{\frac{\mu}{\epsilon}} = \eta_0 \sqrt{\frac{\mu_\tau}{\epsilon_r}}.$$
 1.109

which may be greater or lesser than the impedance of free-space (η_0) , depending on whether μ_r is greater or lesser than ϵ_r . In the lossless case, η is real, so both Γ and T from (1.105) are real, and \vec{E} and \vec{H} are in phase with each other in both mediums.

Power conservation for the incident, reflected, and transmitted waves can be demonstrated by computing the Poynting vectors in the two regions. Thus, for z < 0, the complex Poynting vector is

$$\begin{split} \bar{S}^{-} &= \bar{E} \times \bar{H}^{*} = (\bar{E}_{i} + \bar{E}_{r}) \times (\bar{H}_{i} + \bar{H}_{r}) \\ &= \hat{z} |E_{0}|^{2} \frac{1}{\eta_{0}} (e^{-jk_{0}z} + \Gamma e^{jk_{0}z}) (e^{-jk_{0}z} - \Gamma e^{jk_{0}z})^{*} \\ &= \hat{z} |E_{0}|^{2} \frac{1}{\eta_{0}} (1 - |\Gamma|^{2} + \Gamma e^{2jk_{0}z} - \Gamma^{*} e^{-2jk_{0}z}) \\ &= \hat{z} |E_{0}|^{2} \frac{1}{\eta_{0}} (1 - |\Gamma|^{2} + 2j\Gamma \sin 2k_{0}z), \end{split}$$
1.110a

since Γ is real. For z > 0, the complex Poynting vector is

$$\bar{S}^+ = \bar{E}_t \times \bar{H}_t^- = \hat{z} \frac{|E_0|^2 |T|^2}{\eta}.$$

which can be rewritten, using (1.105), as

$$\bar{S}^{+} = \hat{z} |E_0|^2 \frac{4\eta}{(\eta + \eta_0)^2} = \hat{z} |E_0|^2 \frac{1}{\eta_0} (1 - |\Gamma|^2).$$
 1.110b

Now observe that at z = 0, $\overline{S}^{-} = \overline{S}^{+}$, so that complex power flow is conserved across the interface. Now consider the time-average power flow in the two regions. For z < 0, the time-average power flow through a 1-m^2 cross section is

$$P^{-} = \frac{1}{2} \operatorname{Re}(\tilde{S}^{-} \cdot \tilde{z}) = \frac{1}{2} |E_0|^2 \frac{1}{\eta_0} (1 - |\Gamma|^2), \qquad (.) ||a|$$

and for z > 0, the time-average power flow through a 1-m² cross section is

$$P^{+} = \frac{1}{2} \operatorname{Re}(\bar{S}^{-} \cdot \hat{z}) = \frac{1}{2} |E_0|^2 \frac{1}{\eta_0} (1 - |\Gamma|^2) = P^{-}, \qquad 1.111b$$

so real power flow is conserved.

We now note a subtle point. When computing the complex Poynting vector for z < 0 in (1.110a), we used the total \overline{E} and \overline{H} fields. If we compute separately the

Poynting vectors for the incident and reflected waves, we obtain

$$ar{S}_i = ar{E}_i imes ar{H}_i^* = ar{z} rac{|E_0|^2}{\eta_0}, ext{ I.112a}$$

$$ar{S}_r = ar{E}_r imes ar{H}_r^* = - \hat{z} rac{|E_0|^2 |\Gamma|^2}{\eta_0}.$$
 1.112b

and we see that $\bar{S}_i + \bar{S}_r \neq \bar{S}^-$ of (1.110a). The missing cross-product terms account for stored reactive energy in the standing wave in the z < 0 region. Thus, the decomposition of a Poynting vector into incident and reflected components is not, in general, meaningful. Some books define a time-average Poynting vector as $(1/2)\operatorname{Re}(\bar{E} \times \bar{H}^*)$, and in this case such a definition applied to the individual incident and reflected components will give the correct result, since $P_i = (1/2)\operatorname{Re}|\bar{E}_0|^2/\eta_0$, and $P_r = -(1/2)|E_0|^2|\Gamma|^2/\eta_0$, so $P_i + P_r = P^-$. But even this definition will fail to provide meaningful results when the medium for z < 0 is lossy.

Good Conductor

If the region for z > 0 is a good (but not perfect) conductor, the propagation constant can be written as discussed in Section 1.4:

$$\gamma = \alpha + j\beta = (1+j)\sqrt{\frac{\omega\mu\sigma}{2}} = (1+j)\frac{1}{\delta_s}.$$
 1.113

Similarly, the intrinsic impedance of the conductor simplifies to

$$\eta = (1+j)\sqrt{\frac{\omega\mu}{2\sigma}} = (1+j)\frac{1}{\sigma\delta_s}.$$
 1.114

Now the impedance is complex, with a phase angle of 45°, so \overline{E} and \overline{H} will be 45° out of phase, and Γ and T will be complex. In (1.113) and (1.114), $\delta_s = 1/\alpha$ is the skin depth, as defined in (1.60).

Let us check the complex power balance in this case. For z < 0, the complex Poynting vector is

$$\begin{split} \bar{S}^{-} &= \bar{E} \times \bar{H}^{*} = (\bar{E}_{i} + \bar{E}_{r}) \times (\bar{H}_{i} + \bar{H}_{r})^{*} \\ &= \hat{z} |E_{0}|^{2} \frac{1}{\eta_{0}} [1 - |\Gamma|^{2} + 2j \operatorname{Im}(\Gamma e^{2jk_{0}z})], \end{split}$$
 1.115a

which can be evaluated at z = 0 to give

$$\bar{S}^{-}(z=0) = \hat{z}|E_0|^2 \frac{1}{\eta_0}(1-|\Gamma|^2+\Gamma-\Gamma^*).$$

For z > 0, the complex Poynting vector is

$$ar{S}^+ = ar{E}_t imes ar{H}_t^* = \hat{z} |E_0|^2 |T|^2 rac{1}{\eta^*} e^{-2 lpha \hat{z}},$$

and using (1.105) for T and Γ gives

$$\bar{S}^{+} = \hat{z} |E_0|^2 \frac{4\eta}{|\eta + \eta_0|^2} e^{-2\alpha z} = \hat{z} |E_0|^2 \frac{1}{\eta_0} (1 - |\Gamma|^2 + \Gamma - \Gamma^*) e^{-2\alpha z}.$$
 1.115b

So at the interface at z = 0, $\bar{S}^- = \bar{S}^+$, and complex power is conserved.

Observe that if we were to compute the separate incident and reflected Poynting vectors for z < 0 as

$$\bar{S}_{t} = \bar{E}_{s} \times \bar{H}_{s}^{\star} = \hat{z} \frac{|E_{0}|^{2}}{\eta_{0}}, \qquad 1.116a$$

-

$$ar{S}_r = ar{E}_r imes ar{H}_r^* = - \hat{z} rac{|E_0|^2 |\Gamma|^2}{\eta_0},$$
 1.116b

we do not obtain $\hat{S}_r + \hat{S}_r = \hat{S}^-$ of (1.115a), even for z = 0. It is possible, however, to consider real power flow in terms of the individual traveling wave components. Thus, the time-average power flows through a 1-m² cross section are

$$P^{-} = \frac{1}{2} \operatorname{Re}(\bar{S}^{-} \cdot \hat{z}) = \frac{1}{2} |E_0|^2 \frac{1}{\eta_0} (1 - |\Gamma|^2), \qquad 1.117a$$

$$P^{+} = \frac{1}{2} \operatorname{Re}(\bar{S}^{+} \cdot \hat{z}) = \frac{1}{2} |E_0|^2 \frac{1}{\eta_0} (1 - |\Gamma|^2) e^{-2\alpha z}.$$
 1.117b

which shows power balance at z = 0. In addition, $P_i = |E_0|^2/2\eta_0$, and $P_r = -|E_0|^2$ $|\Gamma|^2/2\eta_0$ so that $P_i + P_r = P_i$, showing that the real power flow for z < 0 can be decomposed into incident and reflected wave components.

Now notice that \bar{S}^- , the power density in the lossy conductor, decays exponentially according to the $e^{-2\alpha z}$ attenuation factor. This means that power is being dissipated in the lossy material as the wave propagates into the medium in the +z direction. The power, and also the fields, decay to a negligibly small value within a few skin depths of the material, which for a reasonably good conductor is an extremely small distance at microwave frequencies.

The electric volume current density flowing in the conducting region is given as

$$\tilde{J} = \sigma \tilde{E}_t = \hat{x} \sigma E_0 T e^{-\gamma z} \mathsf{A/m}^2, \qquad 1.118$$

and so the average power dissipated in (or transmitted into) a 1 m^2 cross-sectional volume of the conductor can be calculated from the conductor loss term of (1,92) (Joule's law) as

$$P^{t} = \frac{1}{2} \int_{V} \bar{E} \cdot \bar{J}^{*} dv = \frac{1}{2} \int_{x=0}^{1} \int_{y=0}^{1} \int_{z=0}^{\infty} (\hat{x} E_{0} T e^{-\gamma z}) \cdot (\hat{x} \sigma E_{0} T e^{-\gamma z})^{*} dz dy dx$$
$$= \frac{1}{2} \sigma |E_{0}|^{2} |T|^{2} \int_{z=0}^{\infty} e^{-2\alpha z} dz = \frac{\sigma |E_{0}|^{2} |T|^{2}}{4\alpha}.$$
 1.119

Since $1/\eta = \sigma \delta_s/(1+j) = (\sigma/2\alpha)(1-j)$, the real power entering the conductor through a 1-m² cross section (as given by $(1/2)\operatorname{Re}(\bar{S}^+ \cdot \hat{z})$ at z = 0) can be expressed using (1.115b) as $P^t = |E_0|^2 |T|^2 (\sigma/4\alpha)$, which is in agreement with (1.119).

Perfect Conductor

Now assume that the region z > 0 contains a perfect conductor. The above results can be specialized to this case by allowing $\sigma \to \infty$. Then, from (1.113) $\alpha \to \infty$; from (1.114) $\eta \to 0$; from (1.60) $\delta_s \to 0$; and from (1.105a,b) $T \to 0$, and $\Gamma \to -1$. The fields for z > 0 thus decay infinitely fast, and are identically zero in the perfect conductor. The perfect conductor can be thought of as "shorting out" the incident electric field. For z < 0, from (1.99) and (1.100), the total \bar{E} and \bar{H} fields are, since $\Gamma = -1$,

$$\bar{E} = \bar{E}_i + \bar{E}_r = \hat{x} E_0 (e^{-jk_0 z} - e^{jk_0 z}) = -\hat{x} 2j E_0 \sin k_0 z, \qquad 1.120a$$

$$\bar{H} = \bar{H}_i + \bar{H}_r = \hat{y} \frac{1}{\eta_0} E_0 (e^{-jk_0 z} + e^{jk_0 z}) = \hat{y} \frac{2}{\eta_0} E_0 \cos k_0 z. \qquad 1.120b$$

Observe that at z = 0, $\overline{E} = 0$ and $\overline{H} = \hat{y}(2/\eta_0)E_0$. The Poynting vector for z < 0 is

$$ar{S}^- = ar{E} imes ar{H}^* = ar{z} j rac{4}{\eta_0} |E_0|^2 \sin k_0 z \cos k_0 z_i$$
 1.121

which has a zero real part and thus indicates that no real power is delivered to the perfect conductor.

The volume current density of (1.118) for the lossy conductor reduces to an infinitely thin sheet of surface current in the limit of infinite conductivity:

$$\bar{J}_{s} = \hat{n} \times \bar{H} = -\hat{z} \times \left(\hat{y}\frac{2}{\eta_{0}}E_{0} \cos k_{0}z\right)|_{z=0} = \hat{x}\frac{2}{\eta_{0}}E_{0} \text{ A/m.}$$
 1.122

The Surface Impedance Concept

In many problems, particularly those in which the effect of attenuation or conductor loss is needed, the presence of an imperfect conductor must be taken into account. The surface impedance concept allows us to do this in a very convenient way. We will develop this method from the theory presented in the previous sections.

Consider a good conductor in the region z > 0. As we have seen, a plane wave normally incident on this conductor is mostly reflected, and the power that is transmitted into the conductor is dissipated as heat within a very short distance from the surface. There are three ways to compute this power.

First, we can use Joule's law, as in (1.119):

$$P^{t} = \frac{1}{2} \int_{V} \bar{E} \cdot \bar{J}^{*} dv = \frac{1}{2\sigma} \int_{V} |\bar{J}|^{2} dv. \qquad 1.123$$

÷

For a 1 m^2 area of conductor surface, the power transmitted through this surface and dissipated as heat is, from (1.119),

$$P^t = \frac{\sigma |E_0|^2 |T|^2}{4\alpha}.$$

Using (1.150b) for T, (1.114) for η , and the fact that $\alpha = 1/\delta_s$, gives the following

result:

$$rac{\sigma |T|^2}{lpha} = rac{\sigma \delta_s 4 |\eta|^2}{|\eta+\eta_0|^2} \simeq rac{8}{\sigma \delta_s \eta_0^2} = rac{8 R_s}{\eta_0^2},$$

where we have assumed $\eta \ll \eta_0$, which is true for a good conductor. Then the above power can be written as

$$P' = \frac{2 E_0 |^2 R_s}{\eta_0^2}.$$
 1.124

$$R_s = \operatorname{Re}(\eta) = \operatorname{Re}\left(\frac{1+j}{\sigma\delta_s}\right) = \frac{1}{\sigma\delta_s} = \sqrt{\frac{\omega\mu}{2\sigma}}$$
 1.125

where

is the surface resistance of the metal.

Another way to find the power loss is to compute the power flow into the conductor using the Poynting vector, since all power entering the conductor at z = 0 is dissipated. As in (1.115b), we have

$$P^{t} = rac{1}{2} \operatorname{Re}(\ddot{S}^{+} \cdot \hat{z})_{z=0} = rac{2|E_{0}|^{2} \operatorname{Re}(\eta)}{|\eta + \eta_{0}|^{2}},$$

which for large conductivity becomes, since $\eta \ll \eta_0$.

$$P^{t} = \frac{2|E_{0}|^{2}R_{s}}{\eta_{0}^{2}}.$$
 1.126

which agrees with (1.124).

A third method uses an effective surface current density and the surface impedance, without the need for the fields inside the conductor. From (1.118), the volume current density in the conductor is

$$\bar{J} = \dot{x}\sigma T E_0 e^{-\gamma z} \mathbf{A}/\mathbf{m}^2, \qquad 1.127$$

so the total current flow per unit width in the x direction is

$$\bar{J}_s = \int_0^\infty \bar{J} \, dz = \hat{r} \sigma T E_0 \, \int_0^\infty e^{-\gamma_E} \, dz = \frac{\hat{r} \sigma T E_0}{\gamma} \, \mathrm{A/m},$$

and taking the limit of $\sigma T/\gamma$ for large σ gives

$$\frac{\sigma T}{\gamma} = \frac{\sigma \xi_s}{(1+j)} \frac{2\eta}{(\eta+\eta_0)} \simeq \frac{\sigma \delta_s}{(1+j)} \frac{2(1+j)}{\sigma \delta_s \eta_0} = \frac{2}{\eta_0},$$
$$\hat{J}_s = \hat{x} \frac{2E_0}{\eta_0} \quad \text{A/m.} \qquad 1.128$$

SÓ

If the conductivity were infinite, a true surface current density of

$$\bar{J}_s = \hat{n} \times \bar{H}|_{z=0} = -\hat{z} \times (\bar{H}_i + \bar{H}_r)|_{z=0} = \hat{x} E_0 \frac{1}{\eta_0} (1 - \Gamma) = \hat{x} \frac{2E_0}{\eta_0} \text{ A/m}$$

would flow, which is identical to the total current in (1.128).

Now replace the exponentially decaying volume current of (1.127) in Joule's law of (1.123) with a uniform volume current extending a distance of one skin depth. Thus, let

$$\vec{J} = \begin{cases} \vec{J}_s / \delta_s & \text{for } 0 < z < \delta_s \\ 0 & \text{for } z > \delta_s, \end{cases}$$
1.129

so that the total current flow is the same. Then use (1.123) to find the power lost:

$$P^{t} = \frac{1}{2\sigma} \int_{S} \int_{z=0}^{\delta_{s}} \frac{|\bar{J}_{s}|^{2}}{\delta_{s}^{2}} dz \, ds = \frac{R_{s}}{2} \int_{S} |\bar{J}_{s}|^{2} \, ds = \frac{2|E_{0}|^{2}R_{s}}{\eta_{0}^{2}}, \qquad 1.130$$

where \int_S denotes a surface integral over the conductor surface, in this case chosen as 1 m^2 . The result of (1.130) agrees with our previous results for P^t in (1.126) and (1.124), and shows that the power loss can be calculated as

$$P^{t} = \frac{R_{s}}{2} \int_{S} |\bar{J}_{s}|^{2} ds = \frac{R_{s}}{2} \int_{S} |\bar{H}_{t}|^{2} ds, \qquad 1.131$$

in terms of the surface resistance R_s and the surface current \bar{J}_s , or tangential magnetic field \bar{H}_t . It is important to realize that the surface current can be found from $\bar{J}_s = \hat{n} \times \bar{H}$, as if the metal were a perfect conductor. This method is very general, applying to fields other than plane waves, and to conductors of arbitrary shape, as long as bends or corners have radii on the order of a skin depth or larger. The method is also quite accurate, as the only approximation in the above was that $\eta \ll \eta_0$, which is a good approximation. As an example, copper at 1 GHz has $|\eta| = 0.012 \Omega$, which is indeed much less than $\eta_0 = 377 \Omega$.

(

EXAMPLE 1.4 Plane Wave Reflection from a Conductor

Consider a plane wave normally incident on a half-space of copper. If f = 1 GHz, compute the propagation constant, impedance, and skin depth for the conductor. Also compute the reflection and transmission coefficients.

Solution

For copper, $\sigma = 5.813 \times 10^7 \text{S/m}$, so from (1.60) the skin depth is

$$\delta_s = \sqrt{\frac{2}{\omega\mu\sigma}} = 2.088 \times 10^{-6} \mathrm{m},$$

and the propagation constant is, from (1.113),

$$\gamma = \frac{1+j}{\delta_s} = (4.789 + j4.789) \times 10^5 \text{ m}^{-1}.$$

The intrinsic impedance is, from (1.114),

$$\eta = \frac{1+j}{\sigma \delta_s} = (8.239 + j 8.239) \times 10^{-3} \ \Omega.$$

which is quite small relative to the impedance of free-space ($\eta_0 = 377 \Omega$). The reflection coefficient is then

$$\Gamma = \frac{\eta - \eta_0}{\eta + \eta_0} = 1 \underline{/179.99}^{q}$$

(practically that of an ideal short circuit), and the transmission coefficient is

$$T = \frac{2\eta}{\eta + \eta_0} = 6.181 \times 10^{-5} \underline{.45}^{\circ}.$$

1.8 OBLIQUE INCIDENCE AT A DIELECTRIC INTERFACE

We continue our discussion of plane waves by considering the problem of a plane wave obliquely incident on a plane interface between two lossless dielectric regions, as shown in Figure 1.13. There are two canonical cases of this problem: the electric field is either in the xz plane (parallel polarization), or normal to the xz plane (perpendicular polarization). An arbitrary incident plane wave, of course, may have a polarization that is neither of these, but it can be expressed as a linear combination of these two individual cases.

The general method of solution is similar to the problem of normal incidence: we will write expressions for the incident, reflected, and transmitted fields in each region and match boundary conditions to find the unknown amplitude coefficients and angles.





Geometry for a plane wave obliquely incident at the interface between two dielectric regions.

Parallel Polarization

In this case, the electric field vector lies in the xz plane, and the incident fields can be written as

$$\bar{E}_i = E_0(\hat{x} \cos \theta_i - \hat{z} \sin \theta_i) e^{-jk_1(x\sin \theta_i + z\cos \theta_i)}, \qquad 1.132a$$

$$\bar{H}_{i} = \frac{E_{0}}{\eta_{1}} \hat{y} e^{-jk_{1}(x\sin\theta_{1}+x\cos\theta_{1})}, \qquad 1.132b$$

where $k_1 = \omega \sqrt{\mu_0 \epsilon_1}$, and $\eta_1 = \sqrt{\mu_0 / \epsilon_1}$ are the wavenumber and wave impedance of region 1. The reflected and transmitted fields can be written as

$$\hat{E}_r = E_0 \Gamma(\hat{x} \cos \theta_r + \hat{z} \sin \theta_r) e^{-jk_1 (x \sin \theta_r - z \cos \theta_r)}, \qquad 1.133a$$

$$\bar{H}_r = \frac{-E_0\Gamma}{\eta_1} \hat{y} e^{-jk_1(x\sin\theta_r - z\cos\theta_r)}, \qquad 1.133b$$

$$\tilde{E}_t = E_0 T(\hat{x}\cos\theta_t - \hat{z}\sin\theta_t) e^{-jk_2(x\sin\theta_t + z\cos\theta_t)}, \qquad 1.134a$$

$$\hat{H}_t = \frac{E_0 T}{\eta_2} \hat{y} e^{-jk_2 (x \sin \theta_t + z \cos \theta_t)}.$$
1.134b

In the above, Γ and T are the reflection and transmission coefficients, and k_2 , η_2 are the wavenumber and wave impedance of region 2, defined as

$$k_2=\omega\sqrt{\mu_0\epsilon_2},\qquad \eta_2=\sqrt{\mu_0/\epsilon_2}.$$

At this point, we have Γ, T, θ_r , and θ_t as unknowns.

We can obtain two complex equations for these unknowns by enforcing the continuity of E_x and H_y , the tangential field components, at the interface at z = 0. We then obtain

$$\cos \theta_i e^{-jk_1 x \sin \theta_1} + \Gamma \cos \theta_r e^{-jk_1 x \sin \theta_r} = T \cos \theta_t e^{-jk_2 x \sin \theta_t}, \qquad 1.135a$$

$$\frac{1}{\eta_1}e^{-jk_1x\sin\theta_i} - \frac{\Gamma}{\eta_1}e^{-jk_1x\sin\theta_r} = \frac{T}{\eta_2}e^{-jk_2x\sin\theta_t}.$$
 1.135b

Both sides of (1.135a) and (1.135b) are functions of the coordinate x. If E_x and H_y are to be continuous at the interface z = 0 for all x, then this x variation must be the same on both sides of the equations, leading to the following condition:

$$k_1 \sin \theta_i = k_1 \sin \theta_r = k_2 \sin \theta_t$$

which results in the well-known Snell's laws of reflection and refraction:

$$\theta_i = \theta_r,$$
 1.136*a*

$$k_1 \sin \theta_i = k_2 \sin \theta_t. \tag{1.136b}$$

The above argument ensures that the phase terms in (1.135) vary with x at the same rate on both sides of the interface, and so is often called the *phase matching condition*.

ŧ.

Using (1.136) in (1.135) allows us to solve for the reflection and transmission coefficients as

$$\Gamma = \frac{\eta_2 \cos \theta_t - \eta_1 \cos \theta_i}{\eta_2 \cos \theta_t + \eta_1 \cos \theta_i}, \qquad 1.137a$$

$$T = \frac{2\eta_2 \cos \theta_i}{\eta_2 \cos \theta_i + \eta_1 \cos \theta_i}.$$
 1.137b

Observe that for normal incidence, we have $\theta_t = \theta_t = \theta_t = 0$, so then

$$\Gamma = \frac{\eta_2 - \eta_1}{\eta_2 + \eta_1}$$
 and $T = \frac{2\eta_2}{\eta_2 + \eta_1}$,

in agreement with the results of Section 1.7.

For this polarization, a special angle of incidence, θ_b , called the Brewster angle, exists where $\Gamma = 0$. This occurs when the numerator of (1.137a) goes to zero ($\theta_i = \theta_b$):

$$\eta_2 \cos \theta_l = \eta_1 \cos \theta_b$$
,

which can be reduced using

$$\cos \theta_t = \sqrt{1 - \sin^2 \theta_t} = \sqrt{1 - \frac{k_1^2}{k_2^2} \sin^2 \theta_b},$$

to give

$$\sin \theta_b = \frac{1}{\sqrt{1 + \epsilon_1/\epsilon_2}}.$$
 1.138

Perpendicular Polarization

In this case, the electric field vector is perpendicular to the xz plane. The incident field can be written as

$$\bar{E}_{i} = E_{0}\hat{y}e^{-jk_{i}(x\sin\theta_{i}+x\cos\theta_{i})}, \qquad 1.139a$$

$$\bar{H}_i = \frac{E_0}{\eta_1} (-\hat{x}\cos\theta_i + \hat{z}\sin\theta_i)e^{-jk_1(x\sin\theta_i + x\cos\theta_i)}, \qquad 1.139b$$

where $k_1 = \omega \sqrt{\mu_0 \epsilon_1}$ and $\eta_1 = \sqrt{\mu_0/\epsilon_1}$ are the wavenumber and wave impedance for region 1, as before. The reflected and transmitted fields can be expressed as

$$\bar{E}_r = E_0 \Gamma \hat{y} e^{-jk \left((x \sin \theta_r - x \cos \theta_r)\right)}, \qquad 1.140a$$

$$\bar{H}_r = \frac{E_0 \Gamma}{\eta_1} (\hat{x} \cos \theta_r + \hat{z} \sin \theta_r) e^{-jk_1 (x \sin \theta_r - x \cos \theta_r)}, \qquad 1.140b$$

$$\tilde{E}_t = E_0 T \hat{y} e^{-y k_2 (x \sin \theta_t + z \cos \theta_t)}, \qquad 1.141a$$

$$\bar{H}_t = \frac{E_0 T}{\eta_2} (-\hat{x} \cos \theta_t + \hat{z} \sin \theta_t) e^{-jk_2(x \sin \theta_t + z \cos \theta_t)}, \qquad 1.141b$$

with $k_2 = \omega \sqrt{\mu_0 \epsilon_2}$ and $\eta_2 = \sqrt{\mu_0/\epsilon_2}$ being the wavenumber and wave impedance in region 2.

Equating the tangential field components E_y and H_x at z = 0 gives

$$e^{-jk_1x\sin\theta_i} + \Gamma e^{-jk_1x\sin\theta_r} = T e^{-jk_2x\sin\theta_i}, \qquad 1.142a$$

$$\frac{-1}{\eta_1}\cos\theta_i e^{-jk_1x\sin\theta_r} + \frac{\Gamma}{\eta_1}\cos\theta_r e^{-jk_2x\sin\theta_r} = \frac{-T}{\eta_2}\cos\theta_t e^{-jk_2x\sin\theta_t}.$$
 1.142b

By the same phase matching argument that was used in the parallel case, we obtain Snell's laws

$$k_1 \sin heta_i = k_1 \sin heta_r = k_2 \sin heta_t$$

identical to (1.136).

Using (1.136) in (1.142) allows us to solve for the reflection and transmission coefficients as

$$\Gamma = \frac{\eta_2 \cos \theta_i - \eta_1 \cos \theta_t}{\eta_2 \cos \theta_7 + \eta_1 \cos \theta_t}, \qquad 1.143a$$

$$T = \frac{2\eta_2 \cos \theta_i}{\eta_2 \cos \theta_i + \eta_1 \cos \theta_t}.$$
 1.143*b*

Again, for the normally incident case, these results reduce to those of Section 1.7.

For this polarization no Brewster angle exists where $\Gamma = 0$, as we can see by examining the numerator of (1.143a).

$$\eta_2 \cos \theta_i = \eta_1 \cos \theta_t,$$

and using Snell's law to give

$$k_2^2(\eta_2^2 - \eta_1^2) = (k_2^2\eta_2^2 - k_1^2\eta_1^2)\sin^2 \theta_1$$

But this leads to a contradiction, since the term in parentheses on the right-hand side is identically zero for dielectric media. Thus, no Brewster angle exists for perpendicular polarization, for dielectric media.

F H

EXAMPLE 1.5 Oblique Reflection from a Dielectric Interface

Plot the reflection coefficients for parallel and perpendicular polarized plane waves incident from free-space onto a dielectric region with $\epsilon_r = 2.55$, versus incidence angle.

Solution The wave impedances are

$$\eta_1 = 377 \,\Omega,$$

 $\eta_2 = \frac{\eta_0}{\sqrt{\epsilon_r}} = \frac{377}{\sqrt{2.55}} = 236 \,\Omega.$



FIGURE 1.14 Reflection coefficient magnitude for parallel and perpendicular polarizations of a plane wave obliquely incident on a dielectric half-space.

We then evaluate (1.137a) and (1.143a) versus incidence angle: the results are shown in Figure 1.14.

Total Reflection and Surface Waves

Snell's law of (1.136b) can be rewritten as

Now consider the case (for either parallel or perpendicular polarization), where $\epsilon_1 > \epsilon_2$. As θ_i increases from 0° to 90°, the refraction angle θ_i will increase from 0° to 90°, but at a faster rate than θ_i increases. The incidence angle θ_i for which $\theta_i = 90^\circ$ is called the critical angle, θ_c , thus

$$\sin\theta_c = \sqrt{\frac{\epsilon_2}{\epsilon_1}}.$$
 1.145

At this angle and beyond, the incident wave will be totally reflected, as the transmitted wave is not propagating into region 2. Let us look at this situation more closely, for the case of $\theta_i > \theta_c$ with parallel polarization.

When $\theta_i > \theta_c$ (1.144) shows that $\sin \theta_t > 1$, so that $\cos \theta_t = \sqrt{1 - \sin^2 \theta_t}$ must be imaginary, and so the angle θ_t loses its physical significance. At this point, it is better

to replace the expressions for the transmitted fields in region 2 with the following:

$$\bar{E}_t = E_0 T \left(\frac{j\alpha}{k_2} \hat{x} - \frac{\beta}{k_2} \hat{z} \right) e^{-j\beta x} e^{-\alpha z}, \qquad 1.146a$$

$$\tilde{H}_t = \frac{E_0 T}{\eta_2} \hat{y} e^{-j\beta x} e^{-\alpha x}.$$
 1.146b

The form of these fields are derived from (1.134) after noting that $-jk_2 \sin \theta_t$ is still imaginary for $\sin \theta_t > 1$, but $-jk_2 \cos \theta_t$ is real, so we can replace $\sin \theta_t$ by β/k_2 , and $\cos \theta_t$ by $j\alpha/k_2$. Substituting (1.146b) into the Helmholtz wave equation for \overline{H} gives

$$-\beta^2 + \alpha^2 + k_2^2 = 0. 1.147$$

Matching E_x and H_y of (1.146) with the \hat{x} and \hat{y} components of the incident and reflected fields of (1.132) and (1.133) at z = 0 gives

$$\cos \theta_i e^{-jk_1 x \sin \theta_i} + \Gamma \cos \theta_\tau e^{-jk_1 x \sin \theta_\tau} = T \frac{j\alpha}{k_2} e^{-j\beta x}, \qquad 1.148a$$

$$\frac{1}{\eta_1}e^{-jk_1x\sin\theta_1} - \frac{\Gamma}{\eta_1}e^{-jk_1x\sin\theta_r} = \frac{T}{\eta_2}e^{-j\beta x}.$$
 1.148b

To obtain phase matching at the z = 0 boundary, we must have

$$k_1 \sin \theta_r = k_1 \sin \theta_r = \beta,$$

which leads again to Snell's law for reflection, $\theta_i = \theta_r$, and to $\beta = k_i \sin \theta_i$. Then α is determined from (1.147) as

$$\alpha = \sqrt{k_2^2 + \beta^2} = \sqrt{k_2^2 - k_1^2 \sin^2 \theta_i}, \qquad 1.149$$

which is seen to be a positive real number. The reflection and transmission coefficients can be obtained from (1.148) as

$$\Gamma = \frac{(j\alpha/k_2)\eta_2 - \eta_1 \cos \theta_i}{(j\alpha/k_2)\eta_2 + \eta_1 \cos \theta_i},$$
 1.150*a*

$$T = \frac{2\eta_2 \cos \theta_i}{(j\alpha/k_2)\eta_2 + \eta_1 \cos \theta_i}.$$
 1.150b

Since Γ is of the form (a - jb)/(a + jb), its magnitude is unity, indicating that all incident power is reflected.

The transmitted fields of (1.146) show propagation in the x direction, along the interface, but exponential decay in the z direction. Such a field is known as a *surface* wave,* since it is tightly bound to the interface. A surface wave is an example of a

* Some authors argue that the term "surface wave" should not be used for a field of this type, since it exists only when plane wave fields exist in the z < 0 region, and so prefer to call it a "surface wave-like" field, or a "forced surface wave."

nonuniform plane wave, so called because it has an amplitude variation in the z direction, apart from the propagation factor in the x direction.

Finally, it is of interest to calculate the complex Poynting vector for the surface wave fields of (1.146):

$$\bar{S}_t = \bar{E}_t \times \bar{H}_t^* = \frac{|E_0|^2 |T|^2}{m_2} \left(\hat{z} \frac{j\alpha}{k_2} + \hat{x} \frac{it}{k_2} \right) e^{-2\alpha z}.$$
 1.151

This shows that no real power flow occurs in the z direction. The real power flow in the x direction is that of the surface wave field, and decays exponentially with distance into region 2. So even though no real power is transmitted into region 2, a nonzero field does exist there, in order to satisfy the boundary conditions at the interface.

1.9 SOME USEFUL THEOREMS

Finally, we discuss several theorems in electromagnetics that we will find useful for later discussions.

The Reciprocity Theorem

Reciprocity is a general concept that occurs in many areas of physics and engineering, and the reader may already be familiar with the reciprocity theorem of circuit theory. Here we will derive the Lorentz reciprocity theorem for electromagnetic fields in two different forms. This theorem will be used later in the book to obtain general properties of network matrices representing inicrowave circuits and to evaluate the coupling of waveguides from current probes and loops, and the coupling of waveguides through apertures. There are a number of other important uses of this powerful concept.

Consider the two separate sets of sources. \bar{J}_1 . \bar{M}_1 and \bar{J}_2 . \bar{M}_2 , which generate the fields \bar{E}_1 , \bar{H}_1 , and \bar{E}_2 , \bar{H}_2 , respectively, in the volume V enclosed by the closed surface S, as shown in Figure 1.15. Maxwell's equations are satisfied individually for these two



FIGURE 1.15 Geometry for the Lorentz reciprocity theorem.

sets of sources and fields, so we can write

$$abla imes ar{E}_1 = -j\omega\muar{H}_1 - ar{M}_1,$$
 1.152a

$$\nabla \times \bar{H}_1 = j\omega\epsilon \bar{E}_1 + \bar{J}_1, \qquad 1.152b$$

$$abla imes ar{E}_2 = -j\omega\muar{H}_2 - ar{M}_2,$$
 1.153a

$$\nabla \times \bar{H}_2 = j\omega\epsilon \bar{E}_2 + \bar{J}_2. \qquad 1.153b$$

Now consider the quantity $\nabla \cdot (\bar{E}_1 \times \bar{H}_2 - \bar{E}_2 \times \bar{H}_1)$, which can be expanded using vector identity (B.8) to give

$$\nabla \cdot (\bar{E}_1 \times \bar{H}_2 - \bar{E}_2 \times \bar{H}_1) = \bar{J}_1 \cdot \bar{E}_2 - \bar{J}_2 \cdot \bar{E}_1 + \bar{M}_2 \cdot \bar{H}_1 - \bar{M}_1 \cdot \bar{H}_2.$$
 1.154

Integrating over the volume V, and applying the divergence theorem (B.15), gives

$$\int_{V} \nabla \cdot (\hat{E}_{1} \times \bar{H}_{2} - \bar{E}_{2} \times \bar{H}_{1}) \, dv = \oint_{S} (\bar{E}_{1} \times \bar{H}_{2} - \bar{E}_{2} \times \bar{H}_{1}) \cdot ds \qquad 1.155$$
$$= \int_{V} (\bar{E}_{2} \cdot \bar{J}_{1} - \bar{E}_{1} \cdot \bar{J}_{2} + \bar{H}_{1} \cdot \bar{M}_{2} - \bar{H}_{2} \cdot \bar{M}_{1}) dv$$

Equation (1.155) represents a general form of the reciprocity theorem, but in practice a number of special situations often occur leading to some simplification. We will consider three cases.

S encloses no sources. Then $\tilde{J}_1 = \tilde{J}_2 = \tilde{M}_1 = \tilde{M}_2 = 0$, and the fields \tilde{E}_1, \tilde{H}_1 , and \tilde{E}_2, \tilde{H}_2 are source-free fields. In this case, the right-hand side of (1.155) vanishes with the result that

$$\oint_{S} \bar{E}_{1} \times \bar{H}_{2} \cdot d\bar{s} = \oint_{S} \bar{E}_{2} \times \bar{H}_{1} \cdot d\bar{s}.$$
1.156

This result will be used in Chapter 4, when demonstrating the symmetry of the impedance matrix for a reciprocal microwave network.

S bounds a perfect conductor. For example, S may be the inner surface of a closed, perfectly conducting cavity. Then the surface integral of (1.155) vanishes, since $\bar{E}_1 \times \bar{H}_1 \cdot \hat{n} = (\hat{n} \times \bar{E}_1) \cdot \bar{H}_2$ (by vector identity B.3), and $\hat{n} \times \bar{E}_1$ is zero on the surface of a perfect conductor (similarly for \bar{E}_2). The result is

$$\int_{V} (\bar{E}_{1} \cdot \bar{J}_{2} - \bar{H}_{1} \cdot \bar{M}_{2}) \, dv = \int_{V} (\bar{E}_{2} \cdot \bar{J}_{1} - \bar{H}_{2} \cdot \bar{M}_{1}) \, dv. \qquad 1.157$$

This result is analogous to the reciprocity theorem of circuit theory. In words, this result states that the system response \vec{E}_1 or \vec{E}_2 is not changed when the source and observation points are interchanged. That is, \vec{E}_2 (caused by \vec{J}_2) at \vec{J}_1 is the same as \vec{E}_1 (caused by \vec{J}_2) at \vec{J}_2 .

S is a sphere at infinity. In this case, the fields evaluated on *S* are very far from the sources and so can be considered locally as plane waves. Then the impedance relation $\tilde{H} = \hat{n} \times \bar{E}/\eta$ applies to (1.155) to give

$$(ar{E}_1 imes ar{H}_2 - ar{E}_2 imes ar{H}_1) \cdot ar{h} = (ar{n} imes ar{E}_1) \cdot ar{H}_2 - (ar{n} imes ar{E}_2) \cdot ar{H}_1$$

= $rac{1}{\eta} ar{H}_1 \cdot ar{H}_2 - rac{1}{\eta} ar{H}_2 \cdot ar{H}_1 = 0,$

so that the result of (1.157) is again obtained. This result can also be obtained for the case of a closed surface S where the surface impedance boundary condition applies.

Image Theory

In many problems a current source is located in the vicinity of a conducting ground plane. Image theory permits the removal of the ground plane by placing a virtual image source of the other side of the ground plane. The reader should be familiar with this concept from electrostatics, so we will prove the result for an infinite current sheet next to an infinite ground plane and then summarize the other possible cases.

Consider the surface current density $\hat{J}_s = J_{s0}\hat{x}$ parallel to a ground plane, as shown in Figure 1.16a. Because the current source is of infinite extent and is uniform in the x, y directions, it will excite plane waves traveling outward from it. The negatively



FIGURE 1.16

Illustration of image theory as applied to an electric current source next to a ground plane. (a) An electric surface current density parallel to a ground plane. (b) The ground plane of (a) replaced with image current at z = -d.

traveling wave will reflect from the ground plane at z = 0, and then travel in the positive direction. Thus, there will be a standing wave field in the region 0 < z < d and a positively traveling wave for z > d. The forms of the fields in these two regions can thus be written as

$$E_x^s = A(e^{jk_0z} - e^{-jk_0z}), \qquad \text{for } 0 < z < d.$$
 1.158a

$$H_y^s = \frac{-A}{\eta_0} (e^{jk_0 z} + e^{-jk_0 z}), \quad \text{for } 0 < z < d, \quad 1.158b$$

$$E_x^+ = Be^{-jk_0 z},$$
 for $z > d,$ 1.159a

$$H_y^+ = \frac{B}{\eta_0} e^{-jk_0 z}, \qquad \text{for } z > d,$$
 1.159b

where η_0 is the wave impedance of free-space. Note that the standing wave fields of (1.158) have been constructed to satisfy the boundary condition that $E_x = 0$ at z = 0. The remaining boundary conditions to satisfy are continuity of \bar{E} at z = d, and the discontinuity in the \bar{H} field at z = d due to the current sheet. From (1.36), since $\bar{M}_s = 0$,

$$E_x^s = E_x^+|_{z=0},$$
 1.160a

while from (1.37) we have

$$\bar{J}_s = \hat{z} \times \hat{y} (H_y^+ - H_y^s)|_{z=0}.$$
 1.160b

Using (1.158) and (1.159) then gives

$$2jA \sin k_0 d = Be^{-jk_0 d}$$
$$J_{s0} = -\frac{B}{\eta_0}e^{-jk_0 d} - \frac{2A}{\eta_0} \cos k_0 d,$$

and

which can be solved for
$$A$$
 and B :

$$A = \frac{-J_{s0}\eta_0}{2} e^{-jk_0 d},$$
$$B = -jJ_{s0}\eta_0 \sin k_0 d.$$

So the total fields are

$$E_x^s = -jJ_{s0}\eta_0 e^{-jk_0 d} \sin k_0 z, \qquad \text{for } 0 < z < d, \qquad 1.161a$$

$$H_y^s = J_{s0} e^{-jk_0 d} \cos k_0 z, \qquad \text{for } 0 < z < d, \qquad 1.161 b$$

$$E_x^+ = -j J_{s0} \eta_0 \sin k_0 de^{-jk_0 z}, \qquad \text{for } z > d, \qquad 1.162a$$

$$H_{y}^{+} = -jJ_{s0}\sin k_{0}de^{-jk_{0}z}. \qquad \text{for } z > d. \qquad 1.162b$$

Now consider the application of image theory to this problem. As shown in Figure 1.16b, the ground plane is removed and an image source of $-J_s$ is placed at z = -d.

By superposition, the total fields for z > 0 can be found by combining the fields from the two sources individually. These fields can be derived by a procedure similar to that above, with the following results:

Fields due to source at z = d:

$$E_{x} = \begin{cases} \frac{-J_{s0}\eta_{0}}{2}e^{-jk_{0}(z-d)} & \text{for } z > d \\ \frac{-J_{s0}\eta_{0}}{2}e^{jk_{0}(z-d)} & \text{for } z < d, \end{cases}$$
1.163a

$$H_{y} = \begin{cases} \frac{-J_{s(t)}}{2} e^{-jk_{y}(z-d)} & \text{for } z > d \\ \frac{J_{s0}}{2} e^{jk_{y}(z-d)} & \text{for } z < d. \end{cases}$$
1.163b

Fields due to source at z = -d:

$$E_{z} = \begin{cases} \frac{J_{z0}\eta_{0}}{2} e^{-jk_{0}(z+d)} & \text{for } z > -d \\ \frac{J_{z0}\eta_{0}}{2} e^{jk_{0}(z+d)} & \text{for } z < -d, \end{cases}$$

$$H_{y} = \begin{cases} \frac{J_{z0}}{2} e^{-jk_{0}(z+d)} & \text{for } z > -d \\ \frac{-J_{z0}}{2} e^{jk_{0}(z+d)} & \text{for } z < -d, \end{cases}$$
1.164*b*

The reader can verify that the solution is identical to that of (1.161) for 0 < z < d, and to (1.162) for z > d, thus verifying the validity of the image theory solution. Note that image theory only gives the correct fields to the right of the conducting plane. Figure 1.17 shows more general image theory results for electric and magnetic dipoles.

The Uniqueness Theorem

Once we have found a solution to Maxwell's equations and the appropriate boundary conditions, the uniqueness theorem assures us that, under the proper conditions, this solution is the only possible solution. This is a particularly useful result when, as in Chapter 4, we can find the fields due to a source by postulating the form of the fields, and then enforce boundary conditions by adjusting some amplitude constants. The uniqueness theorem then guarantees that this procedure gives the correct and unique solution.

Although it can be expressed in various ways, we will state the uniqueness theorem in the following form: In a region bounded by a closed surface S and completely filled with dissipative media, the field \bar{E} , \bar{H} is uniquely determined by the source currents in the region and the tangential components of \bar{E} or \bar{H} on S.

This result can be proved by assuming two solutions to Maxwell's equations, \vec{E}^a , \hat{H}^a , and \vec{E}^b , \hat{H}^b , and showing that they must be identical. Thus, if \vec{E}^a , \vec{H}^a and \vec{E}^b , \vec{H}^b satisfy



FIGURE 1.17

Electric and magnetic current images. (a) An electric current parallel to a ground plane. (b) An electric current normal to a ground plane. (c) A magnetic current parallel to a ground plane. (d) A magnetic current normal to a ground plane.

Maxwell's equations in S, then the difference fields, $\tilde{E}^a - \bar{E}^b$, $\bar{H}^a - \bar{H}^b$, must also satisfy Maxwell's equations inside S. Furthermore, these difference fields must be source-free fields, as substitution into Maxwell's equations (1.27a,b) readily shows. Then Poynting's theorem of (1.88) (with $\sigma = 0$) gives for these difference fields the following result:

$$\oint_{S} (\bar{E}^{a} - \bar{E}^{b}) \times (\bar{H}^{a} - \bar{H}^{b})^{*} \cdot d\bar{s} + j\omega \int_{V} (\mu |\bar{H}^{a} - \bar{H}^{b}|^{2} - \epsilon^{*} |\bar{E}^{a} - \bar{E}^{b}|^{2}) dv = 0. \quad 1.165$$

Now, if \overline{E} and/or \overline{H} is prescribed on the surface S in any of the following ways, or in any combination of these conditions, then the first integral in (1.165) will vanish:

- 1. $\hat{n} \times \hat{E} = 0$, electric walls.
- 2. $\hat{n} \times \hat{H} = 0$, magnetic walls.

- 3. $\hat{n} \times \bar{E} = \bar{E}_t$, a fixed tangential electric field.
- 4. $\hat{n} \times \bar{H} = \bar{H}_t$, a fixed tangential magnetic field.

Now assume a small loss in the medium, so that $\epsilon = \epsilon' - j\epsilon''$ and $\mu = \mu' - j\mu''$ are complex. Then the real part of (1.165) becomes

$$\int_{V} (\mu'' |\bar{H}^{a} - \bar{H}^{b}|^{2} + \epsilon'' |\bar{E}^{a} - \bar{E}^{b}|^{2}) dn = 0.$$
 1.166

Since all of these terms are nonnegative, the equation can be satisfied only if $\bar{E}^a = \bar{E}^b$ and $\bar{H}^a = \bar{H}^b$, which shows that only one solution is possible.

Note that it was necessary to introduce loss to achieve this result; loss can always be introduced into a problem by making ϵ and/or μ complex. This result also suggests that problems involving lossless materials may not have unique solutions. This is indeed the case as the following examples point out.

Resonant modes in a lossless cavity. Such source-free fields are defined as having equal time-average electric and magnetic stored energies, as given by (1.84) and (1.86). If there is no loss, μ and ϵ are real, and the second integral in (1.165) is identically zero for a difference field equal to the field of such a resonant mode. Such source-free fields would rapidly dissipate in the presence of loss, however.

Plane wave incident on a lossless dielectric slab. In Problem 1.7 we will solve the problem of plane wave transmission and reflection from a dielectric slab of finite thickness. This solution will satisfy Maxwell's equations and the boundary conditions, but it may not be the only solution. This is because a surface wave field can also exist on the dielectric slab. This field is source-free, and also satisfies Maxwell's equations and the boundary conditions. For a lossy slab, such a field would quickly be dissipated.

Plane waves in free-space. In infinite lossless free-space, source-free plane wave solutions are possible for any polarization and any direction of propagation. If loss is introduced, such fields would quickly decay to zero.

REFERENCES

- [1] J. C. Maxwell, A Treatise on Electricity and Magnetism, Dover, N.Y., 1954.
- [2] A. A. Oliner. "Historical Perspectives on Microwave Field Theory." IEEE Trans. Microwave Theory and Techniques, vol. MTT-32, pp. 1022–1045. September 1984 (this special issue contains other articles on the history of microwave engineering).
- [3] S. V. Marshall and G. G. Skitek. Electromagnetic Concepts and Applications. Third Edition, Prentice-Hall, N.J., 1990.

- [4] C. A. Balanis, Advanced Engineering Electromagnetics, John Wiley & Sons, N.Y., 1989.
- [5] R. E. Collin, Foundations for Microwave Engineering, Second Edition, McGraw-Hill, N.Y., 1992.
- [6] D. K. Cheng, Field and Wave Electromagnetics, Second Edition, Addison-Wesley, Reading, Mass., 1989.
- [7] S. Ramo, T. R. Whinnery, and T. van Duzer, Fields and Waves in Communication Electronics, Third Edition, John Wiley & Sons, N.Y., 1994.
- [8] C. G. Montgomery, R. H. Dicke, and E. M. Purcell, *Principles of Microwave Circuits*, vol. 8 of MIT Rad. Lab. Series, McGraw-Hill, N.Y., 1948.
- [9] R.F. Harrington. Time-Harmonic Electromagnetic Fields, McGraw-Hill, N.Y., 1961.

PROBLEMS

1.1 Assume that an infinite sheet of electric surface current density $\bar{J}_s = J_a \hat{x} A/m$ is placed on the z = 0 plane between free-space for z < 0, and a dielectric with $\epsilon = \epsilon_r \epsilon_0$ for z > 0, as shown below. Find the resulting \bar{E} and \bar{H} fields in the two regions. HINT: Assume plane wave solutions propagating away from the current sheet, and match boundary conditions to find the amplitudes, as in Example 1.3.



- **1.2** Let $\bar{E} = E_{\rho}\hat{\rho} \div E_{\phi}\hat{\phi} + E_{z}\hat{z}$ be an electric field vector in cylindrical coordinates. Demonstrate that it is incorrect to interpret the expression $\nabla^{2}\hat{E}$ in cylindrical coordinates as $\hat{\rho}\nabla^{2}E_{\rho} + \hat{\phi}\nabla^{2}E_{\phi} + \hat{z}\nabla^{2}E_{z}$ by evaluating both sides of the vector identity $\nabla \times \nabla \times \tilde{E} = \nabla (\nabla \cdot \bar{E}) \nabla^{2}\bar{E}$ for the given electric field.
- 1.3 An anisotropic material has a tensor permittivity $[\epsilon]$ as given below, and a permeability of $4\mu_0$. At a certain point in the material, the electric field is known to be $\tilde{E} = 3\hat{x} + 4\hat{y} + 6\hat{z}$. What is \tilde{D} at this point?

$$[\epsilon] = \epsilon_0 \begin{bmatrix} 1 & -2j & 0\\ 2j & 3 & 0\\ 0 & 0 & 4 \end{bmatrix}$$

1.4 Consider a permanent magnet with a steady magnetic field $\overline{H} = H_0 \hat{y}$, and a parallel plate capacitor with an electric field $\overline{E} = E_0 \hat{x}$, arranged as shown on the next page. Calculate the Poynting vector at a point between both the magnet poles and the capacitor plates. This nonzero result seems to imply real power flow in the z direction, but clearly there is no wave propagation or power delivered from the sources. How do you explain this apparent paradox?



- **1.5** Show that a linearly polarized plane wave of the form $\bar{E} = E_0(\hat{x} + 2\hat{y})e^{-jk_0x}$ can be represented as the sum of an RHCP and an LHCP wave.
- 1.6 Compute the Poynting vector for the general plane wave field of (1.76).
- **1.7** A plane wave is normally incident on a dielectric slab of permittivity ϵ_r and thickness d, where $d = \lambda_0/(4\sqrt{\epsilon_r})$, and λ_0 is the free-space wavelength of the incident wave, as shown below. If free-space exists on both sides of the slab, find the reflection coefficient of the wave reflected from the front of the slab.



1.8 Consider an RHCP plane wave normally incident from free-space (z < 0) onto a half-space (z > 0) consisting of a good conductor. Let the incident electric field be of the form

$$ar{E}_i = E_0(\hat{x} - j\hat{y})e^{-j\kappa_0oldsymbol{z}},$$

and find the electric and magnetic fields in the region z > 0. Compute the Poynting vectors for z < 0 and z > 0, and show that complex power is conserved. What is the polarization of the reflected wave?

1.9 Consider a plane wave propagating in a lossy dielectric medium for z < 0, with a perfectly conducting plate at z = 0. Assume that the lossy medium is characterized by $\epsilon = (5 - j2)\epsilon_0$, $\mu = \mu_0$, and that the frequency of the plane wave is 1.0 GHz, and let the amplitude of the incident electric field be 4 V/m at z = 0. Find the reflected electric field for z < 0, and plot the magnitude of the total electric field for $-0.5 \le z \le 0$.

53

- **1.10** A plane wave in free-space is normally incident on a thin copper sheet of thickness t. What is the approximate required thickness if the copper sheet is to be used as a shield to reduce the level of the transmitted electric field by 90 dB? Do this calculation for f = 1 MHz, 1 GHz, and 100 GHz. HINT: Simplify the problem by ignoring reflections at the interfaces.
- 1.11 A uniform lossy medium with $\epsilon_r = 3.0$, tan $\delta = 0.1$, and $\mu = \mu_0$ fills the region between z = 0 and z = 20 cm, with a ground plane at z = 20 cm, as shown below. An incident plane wave with an electric field,

$$ar{E}_i = \hat{x} 100 e^{-\gamma z}$$
 V/m.

- is present at z = 0 and propagates in the +z direction. The frequency is f = 3.0 GHz.
- (a) Compute P_t , the power density of the incident wave, and P_r , the power density of the reflected wave, at z = 0.
- (b) Compute the input power density, P_{in} , at z = 0, from the total fields at z = 0. Does $P_{in} = P_i P_r$?



- **1.12** Redo Problem 1.1, but with an electric surface current density of $J_s = J_o \hat{x} e^{-j\beta x} A/m$, where $\beta < k_0$.
- **1.13** A parallel polarized plane wave is obliquely incident from free-space onto a magnetic material with permittivity ϵ_0 and permeability $\mu_0\mu_r$. Find the reflection and transmission coefficients. Does a Brewster angle exist for this case, where the reflection coefficient vanishes for a particular angle of incidence?
- 1.14 Repeat Problem 1.13 for the perpendicularly polarized case.
- 1.15 Consider the gyrotropic permittivity tensor shown below;

$$[\boldsymbol{\epsilon}] = \boldsymbol{\epsilon}_{\mathbf{0}} \begin{bmatrix} \boldsymbol{\epsilon}_{r} & j\boldsymbol{\kappa} & \boldsymbol{0} \\ -j\boldsymbol{\kappa} & \boldsymbol{\epsilon}_{r} & \boldsymbol{0} \\ \boldsymbol{0} & \boldsymbol{0} & \boldsymbol{1} \end{bmatrix}.$$

The \overline{D} and \overline{E} fields are related as

$$\begin{bmatrix} D_x \\ D_y \\ D_z \end{bmatrix} = [\epsilon] \begin{bmatrix} E_x \\ E_y, \\ E_z \end{bmatrix}.$$

Show that the transformations

$$E_+ = E_x - jE_y,$$
 $D_+ = D_x - jD_y,$
 $E_- = E_x + jE_y,$ $D_- = D_x + jD_y,$

allow the relation between \bar{E} and \bar{D} to be written as

$$\begin{bmatrix} D_{+} \\ D_{-} \\ D_{z} \end{bmatrix} = [\epsilon'] \begin{bmatrix} E_{+} \\ E_{-} \\ E_{z} \end{bmatrix},$$

where $[\epsilon']$ is now a diagonal matrix. What are the elements of $[\epsilon']$? Using this result, derive wave equations for E_{+} and E_{-} , and find the resulting propagation constants.

- **1.16** Show that the reciprocity theorem expressed in (1.157) also applies to a region enclosed by a closed surface *S*, where a surface impedance boundary condition applies.
- 1.17 Consider an electric surface current density of $\bar{J}_s = \hat{y} J_0 e^{-\beta x} A/m$, located on the z = d plane. If a perfectly conducting ground plane is placed at z = 0, use image theory to find the total fields for z > 0.

Transmission Line Theory

In many ways transmission line theory bridges the gap between field analysis and basic circuit theory, and so is of significant importance in microwave network analysis. As we will see, the phenomenon of wave propagation on transmission lines can be approached from an extension of circuit theory or from a specialization of Maxwell's equations; we shall present both viewpoints and show how this wave propagation is described by equations very similar to those used in Chapter 1 for plane wave propagation.

2.1 THE LUMPED-ELEMENT CIRCUIT MODEL FOR A TRANSMISSION LINE

The key difference between circuit theory and transmission line theory is electrical size. Circuit analysis assumes that the physical dimensions of a network are much smaller than the electrical wavelength, while transmission lines may be a considerable fraction of a wavelength, or many wavelengths, in size. Thus a transmission line is a distributed-parameter network, where voltages and currents can vary in magnitude and phase over its length.

As shown in Figure 2.1a, a transmission line is often schematically represented as a two-wire line, since transmission lines (for TEM wave propagation) always have at least two conductors. The short piece of line of length Δz of Figure 2.1a can be modeled as a lumped-element circuit, as shown in Figure 2.1b, where R, L, G, C are per unit length quantities defined as follows:

R = series resistance per unit length, for both conductors, in Ω/m .

L = series inductance per unit length, for both conductors, in H/m.

G = shunt conductance per unit length, in S/m.

C = shunt capacitance per unit length, in F/m.

The series inductance L represents the total self-inductance of the two conductors, and the shunt capacitance C is due to the close proximity of the two conductors. The series resistance R represents the resistance due to the finite conductivity of the conductors, and the shunt conductance G is due to dielectric loss in the material between the


FIGURE 2.1 Voltage and current definitions and equivalent circuit for an incremental length of transmission line. (a) Voltage and current definitions. (b) Lumped-element equivalent circuit.

conductors. R and G, therefore, represent loss. A finite length of transmission line can be viewed as a cascade of sections of the form of Figure 2.1b.

From the circuit of Figure 2.1b. Kirchhoff's voltage law can be applied to give

$$v(z,t) - R\Delta z i(z,t) - L\Delta z \frac{\partial i(z,t)}{\partial t} - v(z + \Delta z,t) = 0.$$
 2.4a

and Kirchhoff's current law leads to

$$i(z,t) - G\Delta zv(z + \Delta z, t) - C\Delta z \frac{\partial v(z + \Delta z, t)}{\partial t} - i(z + \Delta z, t) = 0.$$
 2.1b

Dividing (2.1a) and (2.1b) by Δz and taking the limit as $\Delta z \rightarrow 0$ gives the following differential equations:

$$\frac{\partial v(z,t)}{\partial z} = -Ri(z,t) - L\frac{\partial i(z,t)}{\partial t}, \qquad 2.2a$$

$$\frac{\partial i(z,t)}{\partial z} = -Gv(z,t) - C\frac{\partial v(z,t)}{\partial t}.$$
 2.25

These equations are the time-domain form of the transmission line, or telegrapher, equations,

For the sinusoidal steady-state condition, with cosine-based phasors, (2.2) simplify to

$$\frac{dV(z)}{dz} = -(R + j\omega L)I(z), \qquad 2.3a$$

$$\frac{dI(z)}{dz} = -(G + j\omega C)V(z).$$
 2.3b

Note the similarity in the form of (2.3) and Maxwell's curl equations of (1.41a) and (1.41b).

Wave Propagation on a Transmission Line

The two equations of (2.3) can be solved simultaneously to give wave equations for V(z) and I(z):

$$\frac{d^2 V(z)}{dz^2} - \gamma^2 V(z) = 0, \qquad 2.4a$$

$$\frac{d^2 I(z)}{dz^2} - \gamma^2 I(z) = 0, \qquad 2.4b$$

where

$$\gamma = \alpha + j\beta = \sqrt{(R + j\omega L)(G + j\omega C)}$$
 2.5

is the complex propagation constant, which is a function of frequency. Traveling wave solutions to (2.4) can be found as

$$V(z) = V_o^+ e^{-\gamma z} + V_o^- e^{\gamma z}, \qquad 2.6a$$

$$I(z) = I_o^+ e^{-\gamma z} + I_o^- e^{\gamma z},$$
 2.6b

where the $e^{-\gamma z}$ term represents wave propagation in the +z direction, and the $e^{\gamma z}$ term represents wave propagation in the -z direction. Applying (2.3a) to the voltage of (2.6a) gives the current on the line:

$$I(z) = \frac{\gamma}{R+j\omega L} \left[V_o^+ e^{-\gamma z} - V_o^- e^{\gamma z} \right].$$

Comparison with (2.6b) shows that a characteristic impedance, Z_0 , can be defined as

$$Z_0 = \frac{R + j\omega L}{\gamma} \approx \sqrt{\frac{R + j\omega L}{G + j\omega C}},$$
 2.7

to relate the voltage and current on the line as

$$\frac{V_o^+}{I_o^+} = Z_0 = \frac{-V_o^-}{I_o^-}.$$

Then (2.6b) can be rewritten in the following form:

$$I(z) = \frac{V_o^+}{Z_0} e^{-\gamma z} - \frac{V_o^-}{Z_0} e^{\gamma z}.$$
 2.8

Converting back to the time domain, the voltage waveform can be expressed as

where ϕ^{\pm} is the phase angle of the complex voltage V_o^{\pm} . Using arguments similar to those in Section 1.4, we find that the wavelength on the line is

$$\lambda = \frac{2\pi}{\beta}, \qquad 2.10$$

and the phase velocity is

$$v_p = \frac{\omega}{\beta} = \lambda f. \tag{2.11}$$

The Lossless Line

The above solution was for a general transmission line, including loss effects, and it was seen that the propagation constant and characteristic impedance were complex. In many practical cases, however, the loss of the line is very small and so can be neglected, resulting in a simplification of the above results. Setting R = G = 0 in (2.5) gives the propagation constant as

$$\gamma = \alpha + j\beta = j\omega\sqrt{LC},$$

 $\beta = \omega\sqrt{LC},$ 2.12a

OF

$$\alpha = 0. \qquad \qquad 2.12b$$

As expected for the lossless case, the attenuation constant α is zero. The characteristic impedance of (2.7) reduces to

$$Z_0 = \sqrt{\frac{L}{C}}, \qquad 2.13$$

which is now a real number. The general solutions for voltage and current on a lossless transmission line can then be written as

$$V(z) = V_o^+ e^{-j\beta z} + V_o^- e^{j\beta z}$$
. 2.14a

$$I(z) = \frac{V_o^+}{Z_0} e^{-j\beta z} - \frac{V_o^-}{Z_0} e^{j\beta z}.$$
 2.14b

The wavelength is

$$\lambda = \frac{2\pi}{\beta} = \frac{2\pi}{\omega\sqrt{LC}},$$
 2.15

and the phase velocity is

$$v_p = \frac{\omega}{\beta} = \frac{1}{\sqrt{LC}}.$$
 2.16

2.2 FIELD ANALYSIS OF TRANSMISSION LINES

In this section we will rederive the time-harmonic form of the telegrapher's equations, starting with Maxwell's equations. We will begin by deriving the transmission line

parameters (R, L, G, C) in terms of the electric and magnetic fields of the transmission line and then derive the telegrapher equations using these parameters for the specific case of a coaxial line.

Transmission Line Parameters

Consider a 1 m section of a uniform transmission line with fields \overline{E} and \overline{H} , as shown in Figure 2.2, where S is the cross-sectional surface area of the line. Let the voltage between the conductors be $V_{\sigma}e^{\pm j\beta z}$ and the current be $I_{\sigma}e^{\pm j\beta z}$. The time-average stored magnetic energy for this 1 m section of line can be written, from (1.86), as

$$W_m = \frac{\mu}{4} \int_S \bar{H} \cdot \bar{H}^* \, ds,$$

and circuit theory gives $W_m = L|I_o|^2/4$, in terms of the current on the line. We can thus identify the self-inductance per unit length as

$$L = \frac{\mu}{|I_o|^2} \int_S \bar{H} \cdot \bar{H}^* \, ds \, \mathrm{H/m.}$$
 2.17

Similarly, the time-average stored electric energy per unit length can be found from (1.84) as

$$W_{\epsilon} = rac{\epsilon}{4} \, \int_{S} \, \hat{E} \cdot \hat{E}^{*} \, ds,$$

and circuit theory gives $W_e = C|V_o|^2/4$, resulting in the following expression for the capacitance per unit length:

$$C = \frac{\epsilon}{|V_o|^2} \int_S \bar{E} \cdot \bar{E}^* \, ds \, \mathrm{F/m}.$$
 2.18

From (1.130), the power loss per unit length due to the finite conductivity of the metallic conductors is

$$P_{\rm c} = \frac{R_s}{2} \int_{C_1 + C_2} \tilde{H} \cdot \tilde{H}^* \, d\ell$$



FIGURE 2.2 Field lines on an arbitrary TEM transmission line.

(assuming \bar{H} is tangential to S), and circuit theory gives $P_c = R|I_o|^2/2$, so the series resistance R per unit length of line is

$$R = \frac{R_s}{|I_o|^2} \int_{C_1 + C_2} \bar{\Pi} \cdot \bar{H}^* dl \ \Omega/\mathrm{m}.$$
 2.19

In (2.19), $R_s = 1/\sigma \delta_s$ is the surface resistance of the conductors, and $C_1 + \tilde{C}_2$ represent integration paths over the conductor boundaries. From (1.92), the time-average power dissipated per unit length in a lossy dielectric is

$$P_{\vec{n}} = \frac{\omega \epsilon''}{2} \int_{S} \bar{E} \cdot \bar{E}^* \, ds.$$

where ϵ'' is the imaginary part of the complex dielectric constant $\epsilon = \epsilon' - j\epsilon'' = \epsilon'(1-j\tan \delta)$. Circuit theory gives $P_d = G|V_{oj}|^2/2$, so the shunt conductance per unit length can be written as

$$G = \frac{\omega t''}{|V_o|^2} \int_S \tilde{E} \cdot \tilde{E}^* \, ds \, \mathsf{S/m}.$$
 2.20



EXAMPLE 2.1 Transmission Line Parameters of a Coaxial Line

The fields of a traveling TEM wave inside the coaxial line shown in Figure 2.3 can be expressed as

$$\begin{split} \bar{E} &= \frac{V_o \hat{\rho}}{\rho \ln b/a} e^{-\gamma z}, \\ \bar{E} &= \frac{I_o \hat{\phi}}{2\pi \rho} e^{-\gamma z}, \end{split}$$

where γ is the propagation constant of the line. The conductors are assumed to have a surface resistivity R_{s} , and the material filling the space between



FIGURE 2.3 Geometry of a coaxial line with surface resistance R_s nn the inner and outer conductors.

the conductors is assumed to have a complex permittivity $\epsilon = \epsilon' - j\epsilon''$ and a permeability $\mu = \mu_0 \mu_r$. Determine the transmission line parameters.

Solution

From (2.17)-(2.20) and the above fields the parameters of the coaxial line can be calculated as

$$\begin{split} \mathcal{L} &= \frac{\mu}{(2\pi)^2} \int_{\phi=\phi}^{2\pi} \int_{\rho=a}^{b} \frac{1}{\rho^2} \rho \, d\rho \, d\phi = \frac{\mu}{2\pi} \ln b/a \quad \mathrm{H/m}, \\ C &= \frac{\epsilon'}{(\ln b/a)^2} \int_{\phi=0}^{2\pi} \int_{\rho=a}^{b} \frac{1}{\rho^2} \rho \, d\rho \, d\phi = \frac{2\pi\epsilon'}{\ln b/a} \quad \mathrm{F/m}, \\ R &= \frac{R_s}{(2\pi)^2} \left\{ \int_{\phi=0}^{2\pi} \frac{1}{a^2} a \, d\phi + \int_{\phi=0}^{2\pi} \frac{1}{b^2} b \, d\phi \right\} = \frac{R_s}{2\pi} \left(\frac{1}{a} + \frac{1}{b} \right) \, \Omega/\mathrm{m}, \\ G &= \frac{\omega\epsilon''}{(\ln b/a)^2} \int_{\phi=0}^{2\pi} \int_{\rho=a}^{b} \frac{1}{\rho^2} \rho \, d\rho \, d\phi = \frac{2\pi\omega\epsilon''}{\ln b/a} \, \mathrm{S/m}. \end{split}$$

Table 2.1 summarizes the parameters for coaxial, two-wire, and parallel plate lines. As we will see in the next chapter, the propagation constant, characteristic impedance, and attenuation of most transmission lines are derived directly from a field theory solution; the approach here of first finding the equivalent circuit parameters (L, C, R, G) is useful only for relatively simple lines. Nevertheless, it provides a helpful intuitive concept, and relates a line to its equivalent circuit model.

TABLE 2.1 Transmission Line Parameters for Some Common Lines

	COAX (a) b	TWO-WIRE T_{D} T_{d} T_{d} T_{d} T_{d} T_{d} T_{d}		_
 Ŀ	$\frac{\mu}{2\pi}\ln\frac{b}{a}$	$\frac{\mu}{\pi}\cosh^{-1}\left(\frac{D}{2a}\right)$	$\frac{\mu d}{w}$	
Ċ	<u>Žařeš</u> Inb/a	$\frac{\pi\epsilon'}{\cosh^{-1}(D/2a)}$	<u>e'#</u>	
Ŗ	$\frac{R_s}{2\pi}\left(\frac{1}{a}+\frac{1}{b}\right)$	$\frac{R_s}{\pi a}$	2R. W	
G	<u>2πωε"</u> Inb/a	$\frac{\pi\omega\epsilon^{n}}{\cosh^{-1}(D/2a)}$	$\frac{\omega \epsilon^{"} w}{d}$	

The Telegrapher Equations Derived from Field Analysis of a Coaxial Line

We now show that the telegrapher equations of (2.3), derived using circuit theory, can also be obtained from Maxwell's equations. We will consider the specific geometry of the coaxial line of Figure 2.3. Although we will treat TEM wave propagation more generally in the next chapter, the present discussion should provide some insight into the relationship of circuit and field quantities.

A TEM wave on the coaxial line of Figure 2.3 will be characterized by $E_z = H_z = 0$; furthermore, due to azimuthal symmetry, the fields will have no ϕ -variation, and so $\partial/\partial \phi = 0$. The fields inside the coaxial line will satisfy Maxwell's curl equations,

$$abla imes ar{E} = -j\omega\muar{H},$$
 2.21a

$$abla imes ar{H} = j\omega\epsilonar{E},$$
 2.21b

where $\epsilon = \epsilon' - j\epsilon''$ may be complex to allow for a lossy dielectric filling. Conductor loss will be ignored here. A rigorous field analysis of conductor loss can be carried out, but at this point would tend to obscure our purpose; the interested reader is referred to Ramo, Winnery, and Van Duzer [1] or Stratton [2].

Expanding (2.21a) and (2.21b) then gives the following vector equations:

$$-\hat{\rho}\frac{\partial E_{\varphi}}{\partial z} + \hat{\varphi}\frac{\partial E_{\rho}}{\partial z} + \hat{z}\frac{1}{\rho}\frac{\partial}{\partial\rho}\left(\rho E_{\varphi}\right) = -j\omega\mu\left(\hat{\rho}H_{\rho} + \hat{\phi}H_{\phi}\right), \qquad 2.22a$$

$$-\hat{\rho}\frac{\partial H_{\phi}}{\partial z} + \hat{\phi}\frac{\partial H_{\rho}}{\partial z} + \hat{z}\frac{1}{\rho}\frac{\partial}{\partial \rho}\left(\rho H_{\phi}\right) = j\omega\epsilon\left(\hat{\rho}E_{\rho} + \hat{\phi}E_{\phi}\right).$$
 2.22b

Since the \hat{z} components of these two equations must vanish, it is seen that E_{ϕ} and H_{ϕ} must have the forms

$$E_{\varphi} = \frac{f(z)}{\rho}, \qquad 2.23a$$

$$H_{\phi} = \frac{g(z)}{\rho}.$$
 2.23b

To satisfy the boundary condition that $E_{\phi} = 0$ at $\rho = a, b$, we must have $E_{\phi} = \hat{0}$ everywhere, due to the form of E_{ϕ} in (2.23a). Then from the $\hat{\rho}$ component of (2.22a), it is seen that $H_{\rho} = 0$. With these results, (2.22) can be reduced to

$$\frac{\partial E_{\rho}}{\partial z_{o}} = -j\omega\mu H_{\phi}, \qquad 2.24a$$

$$\frac{\partial H_{\phi}}{\partial z} = -j\omega\epsilon E_{\rho}.$$
 2.24b

From the form of H_{ϕ} in (2.23b) and (2.24a), E_{ρ} must be of the form

$$E_{\rho} = \frac{h(z)}{\rho}.$$
 2.25

Using (2.23b) and (2.25) in (2.24) gives

$$\frac{\partial h(z)}{\partial z} = -j\omega\mu g(z), \qquad 2.26a$$

$$\frac{\partial g(z)}{\partial z} = -j\omega\epsilon h(z).$$
 2.26b

Now the voltage between the two conductors can be evaluated as

$$V(z) = \int_{\rho=a}^{b} E_{\rho}(\rho, z) \, d\rho = h(z) \int_{\rho=a}^{b} \frac{d\rho}{\rho} = h(z) \ln \frac{b}{a}.$$
 2.27a

and the total current on the inner conductor at $\rho = a$ can be evaluated using (2.23b) as

$$I(z) = \int_{\phi=0}^{2\pi} H_{\phi}(a, z) a \, d\phi = 2\pi g(z). \qquad 2.27b$$

Then h(z) and g(z) can be eliminated from (2.26) by using (2.27) to give

$$\frac{\partial V(z)}{\partial z} = -j \frac{\omega \mu \ln b/a}{2\pi} I(z),$$
$$\frac{\partial I(z)}{\partial z} = -j \omega (\epsilon' - j \epsilon'') \frac{2\pi V(z)}{\ln b/a}$$

Finally, using the results for L, G, and C for a coaxial line as derived above, we obtain the telegrapher equations as

$$\frac{\partial V(z)}{\partial z} = -j\omega LI(z), \qquad 2.28a$$

$$\frac{\partial I(z)}{\partial z} = -(G + j\omega C)V(z) \qquad 2.28b$$

(excluding R, the series resistance, since the conductors were assumed to have perfect conductivity). A similar analysis can be carried out for other simple transmission lines.

Propagation Constant, Impedance, and Power Flow for the Lossless Coaxial Line

Equations (2.24a) and (2.24b) for E_{ρ} and H_{ϕ} can be simultaneously solved to yield a wave equation for E_{ρ} (or H_{ϕ}):

$$\frac{\partial^2 E_{\rho}}{\partial z^2} + \omega^2 \mu \epsilon E_{\rho} = 0, \qquad 2.29$$

from which it is seen that the propagation constant is $\gamma^2 = -\omega^2 \mu \epsilon$, which, for lossless media, reduces to

$$\beta = \omega \sqrt{\mu \epsilon} = \omega \sqrt{LC}, \qquad 2.30$$

where the last result is from (2,12). Observe that this propagation constant is of the same form as that for plane waves in a lossless dielectric medium. This is a general result for TEM transmission lines.

The wave impedance is defined as $Z_u = E_{\mu}/H_{\phi}$, which can be calculated from (2.24a) assuming an e^{-jH_z} dependence to give

$$Z_{w} = \frac{E_{\rho}}{H_{\phi}} = \frac{\omega\mu}{\beta} = \sqrt{\mu/\epsilon} = \eta.$$
 2.31

This wave impedance is then seen to be identical to the intrinsic impedance of the medium. η , and again is a general result for TEM transmission lines.

The characteristic impedance of the coaxial line is defined as

$$\dot{Z}_{0} = \frac{V_{o}}{I_{p}} = \frac{E_{\mu}}{2\pi H_{o}} \frac{\ln b/a}{2\pi} = \frac{\eta \ln b/a}{2\pi} = \sqrt{\frac{\mu}{\epsilon}} \frac{\ln b/a}{2\pi}.$$
 2.32

where the forms for E_{μ} and H_{ϕ} from Example 2.1 have been used. The characteristic impedance is geometry dependent and with be different for other transmission line configurations.

Finally, the power flow (in the z direction) on the coaxial line may be computed from the Poynting vector as

$$P = \frac{1}{2} \int_{s} \tilde{E} \times \bar{H}^{*} \cdot d\bar{s} = \frac{1}{2} \int_{\phi=0}^{2\pi} \int_{\rho=a}^{b} \frac{V_{a}I_{a}^{*}}{2\pi\rho^{2} \ln b/a} \rho \, d\rho d\phi = \frac{1}{2} V_{a}I_{a}^{*}, \qquad 2.33$$

a result that is in clear agreement with circuit theory. This shows that the flow of power in a transmission line takes place entirely via the electric and magnetic fields between the two conductors; power is not transmitted through the conductors themselves. As we will see later, for the case of finite conductivity, power may enter the conductors, but this power is then lost as heat and is not delivered to the load.

2.3 THE TERMINATED LOSSLESS TRANSMISSION LINE

Figure 2.4 shows a lossless transmission line terminated in an arbitrary load impedance Z_L . This problem will illustrate wave reflection on transmission lines, a fundamental property of distributed systems.

Assume that an incident wave of the form $V_{\alpha}^{+}e^{-j\beta z}$ is generated from a source at z < 0. We have seen that the ratio of voltage to current for such a traveling wave is





 Z_0 , the characteristic impedance. But when the line is terminated in an arbitrary load $Z_L \neq Z_0$, the ratio of voltage to current at the load must be Z_L . Thus, a reflected wave must be excited with the appropriate amplitude to satisfy this condition. The total voltage on the line can then be written as in (2.14a), as a sum of incident and reflected waves:

$$V(z) = V_o^+ e^{-j\beta z} + V_o^- e^{j\beta z}.$$
 2.34a

Similarly, the total current on the line is described by (2.14b):

$$I(z) = \frac{V_o^+}{Z_0} e^{-j\beta z} - \frac{V_o^-}{Z_0} e^{j\beta z}.$$
 2.34b

The total voltage and current at the load are related by the load impedance, so at z = 0 we must have

$$Z_L = \frac{V(0)}{I(0)} = \frac{V_o^+ + V_o^-}{V_o^+ - V_o^-} Z_0.$$

Solving for V_o^- gives

$$V_{o}^{-} = \frac{Z_{L} - Z_{0}}{Z_{L} + Z_{0}} V_{o}^{+}.$$

The amplitude of the reflected voltage wave normalized to the amplitude of the incident voltage wave is known as the voltage reflection coefficient, Γ :

$$\Gamma = \frac{V_o^-}{V_o^+} = \frac{Z_L - Z_0}{Z_L + Z_0}.$$
 2,35

A current reflection coefficient, giving the normalized amplitude of the reflected current wave. can also be defined. But because such a current reflection coefficient is just the negative of the voltage reflection coefficient (as seen from (2.34)), we will avoid confusion by using only the voltage reflection coefficient in this book.

The total voltage and current waves on the line can then be written as

$$V(z) = V_o^+ \left[e^{-j\beta z} + \Gamma e^{j\beta z} \right], \qquad 2.36a$$

$$I(z) = \frac{V_o^+}{Z_0} \left[e^{-j\beta z} - \Gamma e^{j\beta z} \right].$$
 2.36b

From these equations it is seen that the voltage and current on the line consist of a superposition of an incident and reflected wave; such waves are called standing waves. Only when $\Gamma = 0$ is there no reflected wave. To obtain $\Gamma = 0$, the load impedance Z_L must be equal to the characteristic impedance Z_0 of the transmission line, as seen from (2.35). Such a load is then said to be *matched* to the line, since there is no reflection of the incident wave.

Now consider the time-average power flow along the line at the point z:

$$P_{av} = \frac{1}{2} \operatorname{Re} \left[V(z) I(z)^* \right] = \frac{1}{2} \frac{|V_a^+|^2}{Z_0} \operatorname{Re} \left\{ 1 - \Gamma^* e^{-2j\beta z} + \Gamma e^{2j\beta z} - |\Gamma|^2 \right\},$$

where (2.36) has been used. The middle two terms in the brackets are of the form $A - A^* = 2j \text{Im}(A)$ and so are purely imaginary. This simplifies the result to

$$P_{\mathbf{a}^*} = \frac{1}{2} \frac{|V_o^+|^2}{Z_0} \left(1 - |\Gamma|^2\right).$$
 2.37

which shows that the average power flow is constant at any point on the line, and that the total power delivered to the load (P_{av}) is equal to the incident power $(|V_o|^2/2Z_0)$, minus the reflected power $(|V_o|^2 |\Gamma|^2/2Z_0)$. If $\Gamma = 0$, maximum power is delivered to the load, while no power is delivered for $|\Gamma|^2 = 1$. The above discussion assumes that the generator is matched, so that there is no rereflection of the reflected wave from z < 0,

When the load is mismatched, then, not all of the available power from the generator is delivered to the load. This "loss" is called *return loss* (RL), and is defined (in dB) as

$$\mathbf{RL} = -20\log|\mathbf{I}'| \, \mathbf{dB}, \qquad 2.38$$

so that a matched load ($\Gamma = 0$) has a return loss of ∞ dB (no reflected power), whereas a total reflection ($|\Gamma| = 1$) has a return loss of 0 dB (all incident power is reflected).

If the load is matched to the line, $\Gamma=0$ and the magnitude of the voltage on the line is $|V(z)| = |V_o^+|$, which is a constant. Such a line is sometimes said to be "flat." When the load is mismatched, however, the presence of a reflected wave leads to standing waves where the magnitude of the voltage on the line is not constant. Thus, from (2.36a).

$$|V(z)| = |V_o^+||1 + \Gamma e^{2j\beta z}| = |V_o^+||1 + \Gamma e^{-2j\beta \ell}|$$

= $|V_o^+||1 + |\Gamma|e^{j(\theta - 2\beta \ell)}|,$ 2.39

where $\ell = -z$ is the positive distance measured from the load at z = 0, and θ is the phase of the reflection coefficient ($\Gamma = |\Gamma|e^{\theta}$). This result shows that the voltage magnitude oscillates with position z along the line. The maximum value occurs when the phase term $e^{j(\theta - 2\beta \ell)} = 1$, and is given by

$$V_{\max} = |V_{\alpha}^{\perp}|(1 + |\Gamma|),$$
 2.40a

The minimum value occurs when the phase term $e^{j(\theta - 2\beta t)} = -1$, and is given by

$$V_{\min} = |V_o^+|(1 - |\Gamma|).$$
 2.40b

As $|\Gamma|$ increases, the ratio of V_{max} to V_{max} increases, so a measure of the mismatch of a line, called the *standing wave ratio* (SWR), can be defined as

SWR =
$$\frac{V_{\text{max}}}{V_{\text{min}}} = \frac{1 + |\Gamma|}{1 - |\Gamma|}$$
. 2.41

This quantity is also known as the voltage standing wave ratio, and is sometimes identified as VSWR. From (2.41) it is seen that SWR is a real number such that $1 \le SWR \le \infty$, where SWR = 1 implies a matched load.

From (2.39), it is seen that the distance between two successive voltage maxima (or minima) is $\ell = 2\pi/2\beta = \pi\lambda/2\pi = \lambda/2$, while the distance between a maximum and a minimum is $\ell = \pi/2\beta = \lambda/4$, where λ is the wavelength on the transmission line.

The reflection coefficient of (2.35) was defined as the ratio of the reflected to the incident voltage wave amplitudes at the load ($\ell = 0$), but this quantity can be generalized to any point ℓ on the line as follows. From (2.34a), with $z = -\ell$, the ratio of the reflected component to the incident component is

$$\Gamma(\ell) = \frac{V_o^- e^{-j\beta\ell}}{V_o^+ e^{j\beta\ell}} = \Gamma(0)e^{-2j\beta\ell},$$
2.42

where $\Gamma(0)$ is the reflection coefficient at z = 0, as given by (2.35). This form is useful when transforming the effect of a load mismatch down the line.

We have seen that the real power flow on the line is a constant but that the voltage amplitude, at least for a mismatched line, is oscillatory with position on the line. The perceptive reader may therefore have concluded that the impedance seen looking into the line must vary with position, and this is indeed the case. At a distance $\ell = -z$ from the load, the input impedance seen looking toward the load is

$$Z_{in} = \frac{V(-\ell)}{I(-\ell)} = \frac{V_o^+ \left[e^{j\beta\ell} + \Gamma e^{-j\beta\ell}\right]}{V_o^+ \left[e^{j\beta\ell} - \Gamma e^{-j\beta\ell}\right]} Z_0 = \frac{1 + \Gamma e^{-2j\beta\ell}}{1 - \Gamma e^{-2j\beta\ell}} Z_0, \qquad 2.43$$

where (2.36a,b) have been used for V(z) and I(z). A more usable form may be obtained by using (2.35) for Γ in (2.43):

$$Z_{\rm in} = Z_0 \frac{(Z_L + Z_0)e^{j\beta\ell} + (Z_L - Z_0)e^{-j\beta\ell}}{(Z_L + Z_0)e^{j\beta\ell} - (Z_L - Z_0)e^{-j\beta\ell}}$$
$$= Z_0 \frac{Z_L \cos \beta\ell + jZ_0 \sin \beta\ell}{Z_0 \cos \beta\ell + jZ_L \sin \beta\ell}$$
$$= Z_0 \frac{Z_L + jZ_0 \tan \beta\ell}{Z_0 + jZ_L \tan \beta\ell} .$$
 2.44

This is an important result giving the input impedance of a length of transmission line with an arbitrary load impedance. We will refer to this result as the transmission line impedance equation; some special cases will be considered next.

Special Cases of Lossless Terminated Lines

A number of special cases of lossless terminated transmission lines will frequently appear in our work, so it is appropriate to consider the properties of such cases here.

Consider first the transmission line circuit shown in Figure 2.5, where a line is terminated in a short circuit, $Z_L = 0$. From (2.35) it is seen that the reflection coefficient for a short circuit load is $\Gamma = -1$; it then follows from (2.41) that the standing wave ratio is infinite. From (2.36) the voltage and current on the line are

$$V(z) = V_o^+ \left[e^{-j\beta z} - e^{j\beta z} \right] = -2jV_o^+ \sin \beta z, \qquad 2.45a$$

$$I(z) = \frac{V_o^+}{Z_0} \left[e^{-j\beta z} + e^{j\beta z} \right] = \frac{2V_o^+}{Z_0} \cos \beta z, \qquad 2.45b$$



FIGURE 2.5 A transmission line terminated in a short circuit.

which shows that V = 0 at the load (as it should, for a short circuit), while the current is a maximum there. From (2.44), or the ratio $V(-\ell)/I(-\ell)$, the input impedance is

$$Z_{\rm in} = j Z_0 \tan \beta \ell. \qquad 2.45c$$

which is seen to be purely imaginary for any length, ℓ , and to take on all values between $+j\infty$ and $-j\infty$. For example, when $\ell = 0$ we have $Z_{in} = 0$, but for $\ell = \lambda/4$ we have $Z_{in} = \infty$ (open circuit). Equation (2.45c) also shows that the impedance is periodic in ℓ , repeating for multiples of $\lambda/2$. The voltage, current, and input reactance for the short-circuited line are plotted in Figure 2.6.

Next consider the open-circuited line shown in Figure 2.7, where $Z_L = \infty$. Dividing the numerator and denominator of (2.35) by Z_L and allowing $Z_L \to \infty$ shows that the reflection coefficient for this case is $\Gamma = 1$, and the standing wave ratio is again infinite. From (2.36) the voltage and current on the line are

$$V(z) = V_0^+ \left[e^{-g \beta z} + e^{g \beta z} \right] = 2V_0^+ \cos \beta z, \qquad 2.46a$$

$$I(z) = \frac{V_o^+}{Z_0} \left[e^{-j\beta z} - e^{j\beta \bar{z}} \right] = \frac{-2jV_o^+}{Z_0} \sin \beta z. \qquad 2.46b$$

which shows that now I = 0 at the load, as expected for an open circuit, while the voltage is a maximum. The input impedance is

$$Z_{\rm in} = -jZ_0 \,\cot\,\,\beta\ell. \qquad \qquad 2.46c$$

which is also purely imaginary for any length, ℓ . The voltage, current, and input reactance of the open-circuited line are plotted in Figure 2.8.

Now consider terminated transmission lines with some special lengths. If $\ell = \lambda/2$, (2.44) shows that

$$Z_{\rm in} = Z_L, \qquad \qquad 2.47$$

meaning that a half-wavelength line (or any multiple of $\lambda/2$) does not alter or transform the load impedance, regardless of the characteristic impedance.

If the line is a quarter-wavelength long or, more generally, $\ell = \lambda/4 + n\lambda/2$, for $n = 1, 2, 3, \dots$ (2.44) shows that the input impedance is given by

$$Z_{\rm m}=\frac{Z_0^2}{Z_L}.$$



FIGURE 2.6 (a) Voltage, (b) current, and (c) impedance $(R_{in} = 0 \text{ or } \infty)$ variation along a short-circuited transmission line.

Such a line is known as a *quarter-wave transformer* because it has the effect of transforming the load impedance, in an inverse manner, depending on the characteristic impedance of the line. We will study this case more thoroughly in Section 2.5.

Now consider a transmission line of characteristic impedance Z_0 feeding a line of different characteristic impedance, Z_1 , as shown in Figure 2.9. If the load line is



FIGURE 2.7 A transmission line terminated in an open circuit.



FIGURE 2.8 (a) Voltage, (b) surrent, and (c) impedance $(R_{in} = 0 \text{ or } \infty)$ variation along an open-circuited transmission line.

infinitely long, or if it is terminated in its own characteristic impedance, so that there are no reflections from its end, then the input impedance seen by the feed line is Z_i , so that the reflection coefficient Γ is

$$\Gamma = \frac{Z_1 - Z_0}{Z_1 + Z_0}.$$
 2.49

Not all of the incident wave is reflected; some of it is transmitted onto the second line with a voltage amplitude given by a transmission coefficient, T.

From (2.36a), the voltage for z < 0 is

$$V(z) = V_{c}^{+}(e^{-j\beta z} + f^{*}e^{j\beta z}), \qquad z < 0, \qquad 2.50a$$

where V_o^+ is the amplitude of the incident voltage wave on the feed line. The voltage wave for z > 0, in the absence of reflections, is outgoing only, and can be written as

$$V(z) = V_o^+ T e^{-j\beta z}$$
, for $z > 0$, 2.50b



FIGURE 2.9 Reflection and transmission at the junction of two transmission lines with different characteristic impedances.

Equating these voltages at z = 0 gives the transmission coefficient, T, as

$$T = 1 + \Gamma = 1 + \frac{Z_1 - Z_0}{Z_1 + Z_0} = \frac{2Z_1}{Z_1 + Z_0}.$$
 2.51

The transmission coefficient between two points in a circuit is often expressed in dB as the *insertion loss*, *IL*,

$$IL = -20 \log |T| \, dB.$$
 2.52

POINT OF INTEREST: Decibels and Nepers Often the ratio of two power levels, P_1 and P_2 , in a microwave system is expressed in decibels (dB) as

$$10 \log \frac{P_1}{P_2} dB$$

Thus, a power ratio of 2 is equivalent to 3 dB, while a power ratio of 0.1 is equivalent to -10 dB. Using power ratios in dB makes it easy to calculate power loss or gain through a series of components, since multiplicative loss or gain factors can be accounted for by adding the loss or gain in dB for each stage. For example, a signal passing through a 6 dB attenuator followed by a 23 dB amplifier will have an overall gain of 23 - 6 = 17 dB.

Decibels are used only to represent power ratios, but if $P_1 = V_1^2/R_1$ and $P_2 = V_2^2/R_2$, then the result in terms of voltage ratios is

10 log
$$\frac{V_1^2 R_2}{V_2^2 R_1} = 20 \log \frac{V_1}{V_2} \sqrt{\frac{R_2}{R_1}} \, \mathrm{dB} \; ,$$

where R_1 , R_2 are the load resistances and V_1 , V_2 are the voltages appearing across these loads. If the load resistances are equal, then this formula simplifies to

$$20 \log \frac{V_1}{V_2} dB$$

The ratio of voltages across equal load resistances can also be expressed in terms of nepers (Np) as

$$\ln \frac{V_1}{V_2}$$
 Np.

The corresponding expression in terms of powers is

$$\frac{1}{2} \ln \frac{P_1}{P_2} \operatorname{Np}$$

since voltage is proportional to the square root of power. Transmission line attenuation is often expressed in nepers. Since 1 Np corresponds to a power ratio of ϵ^2 , the conversion between nepers and decibels is

$$1Np = 10 \log \epsilon^2 = 8.686 dB.$$

Absolute powers can also be expressed in decibel notation if a reference power level is assumed. If we let $P_2 = 1$ mW, then the power P_1 can be expressed in dBm as

$$10 \log \frac{P_i}{1 \text{ mW}} \text{ dBm.}$$

Thus a power of 1 mW is 0 dBm, while a power of 1W is 30 dBm, etc.

2.4 THE SMITH CHART

The Smith chart, shown in Figure 2.10, is a graphical aid that is very useful when solving transmission line problems. Although there are a number of other impedance and reflection coefficient charts that can be used for such problems [3], the Smith chart is probably the best known and most widely used. It was developed in 1939 by P, Smith at the Bell Telephone Laboratories [4]. The reader may feel that, in this day of scientific calculators and powerful computers, graphical solutions have no place in modern engineering. The Smith chart, however, is more than just a graphical technique. Besides being an integral part of much of the current computer-aided design (CAD) software and test equipment for microwave design, the Smith chart provides an extremely useful way of visualizing transmission line phenomenon and so is also important for pedagogical reasons. A microwave engineer can develop intuition about transmission line and impedance-matching problems by learning to think in terms of the Smith chart.

At first glance the Smith chart may seem intimidating, but the key to its understanding is to realize that it is essentially a polar plot of the voltage reflection coefficient, Γ . Let the reflection coefficient be expressed in magnitude and phase (polar) form as $\Gamma = |\Gamma|e^{j\theta}$. Then the magnitude ${}_{1}\Gamma|$ is plotted as a radius ($|\Gamma| \le 1$) from the center of the chart, and the angle $\theta(-180^{\circ} \le \theta \le 180^{\circ})$ is measured from the right-hand side of the horizontal diameter. Any passively realizable ($|\Gamma| \le 1$) reflection coefficient can then be plotted as a unique point on the Smith chart.

The real utility of the Smith chart, however, lies in the fact that it can be used to convert from reflection coefficients to normalized impedances (or admittances), and vice versa, using the impedance (or admittance) circles printed on the chart. When dealing with impedances on a Smith chart, normalized quantitites are generally used, which we will denote by lowercase letters. The normalization constant is usually the characteristic impedance of the line. Thus, $z = Z/Z_0$ represents the normalized version of the impedance Z.



FIGURE 2.10 The Smith chart.

If a lossless line of characteristic impedance Z_0 is terminated with a load impedance Z_L , the reflection coefficient at the load can be written from (2.35) as

$$\Gamma = \frac{z_L - 1}{z_L + 1} = |\Gamma| e^{j\theta}, \qquad 2.53$$

where $z_L = Z_L/Z_0$ is the normalized load impedance. This relation can be solved for z_L in terms of Γ to give (or, from (2.43) with $\ell = 0$)

$$z_L = \frac{1 + |\Gamma|e^{j\theta}}{1 - |\Gamma|e^{j\theta}}.$$
 2.54

This complex equation can be reduced to two real equations by writing Γ and z_L in terms of their real and imaginary parts. Let $\Gamma = \Gamma_r + j\Gamma_1$, and $z_L = r_L + jx_L$. Then,

$$r_L + jx_L = \frac{(1+\Gamma_r) + j\Gamma_i}{(1-\Gamma_r) - j\Gamma_i}$$

The real and imaginary parts of this equation can be found by multiplying the numerator and denominator by the complex conjugate of the denominator to give

$$r_L = \frac{1 - \Gamma_r^2 - \Gamma_r^2}{(1 - \Gamma_r)^2 + \Gamma_s^2},$$
 2.55a

$$x_L = \frac{2\Gamma_i}{\left(1 - \Gamma_r\right)^2 + \Gamma_i^2}.$$
 2.55b

Rearranging (2.55) gives

$$\left(\Gamma_r - \frac{r_L}{1 + r_L}\right)^2 + \Gamma_i^2 = \left(\frac{1}{1 + r_L}\right)^2.$$
 2.56a

$$(\Gamma_r - 1)^2 + \left(\Gamma_r - \frac{1}{x_L}\right)^2 = \left(\frac{1}{x_L}\right)^2,$$
 2.56b

which are seen to represent two families of circles in the Γ_r , Γ_i plane. Resistance circles are defined by (2.56a), and reactance circles are defined by (2.56b). For example, the $r_L = 1$ circle has its center at $\Gamma_r = 0.5$, $\Gamma_i = 0$, and has a radius of 0.5, and so passes through the center of the Smith chart. All of the resistance circles of (2.56a) have centers on the horizontal $\Gamma_r = 0$ axis, and pass through the $\Gamma = 1$ point on the right-hand side of the chart. The centers of all of the reactance circles of (2.56b) lie on the vertical $\Gamma_r = 1$ line (off the chart), and these circles also pass through the $\Gamma = 1$ point. The resistance and reactance circles are orthogonal.

The Smith chart can also be used to graphically solve the transmission line impedance equation of (2.44), since this can be written in terms of the generalized reflection coefficient as

$$Z_{\rm in} = Z_0 \frac{1 + \Gamma e^{-2j\beta\ell}}{1 - \Gamma e^{-2j\beta\ell}}, \qquad 2.53$$

where Γ is the reflection coefficient at the load, and ℓ is the (positive) length of transmission line. We then see that (2.57) is of the same form as (2.54), differing only by the phase angles of the Γ terms. Thus, if we have plotted the reflection coefficient $|\Gamma|e^{j\theta}$ at the load, the normalized input impedance seen looking into a length ℓ of transmission line terminated with z_L can be found by rotating the point clockwise an amount $2\beta\ell$ (subtracting $2\beta\ell$ from θ) around the center of the chart. The radius stays the same, since the magnitude of Γ does not change with position along the line.

To facilitate such rotations, the Smith chart has scales around its periphery calibrated in electrical wavelengths, toward and away from the "generator" (which just means the direction away from the load). These scales are relative, so only the difference in wavelengths between two points on the Smith chart is meaningful. The scales cover a range of 0 to 0.5 wavelengths, which reflects the fact that the Smith chart automatically includes the periodicity of transmission line phenomenon. Thus, a line of length $\lambda/2$ (or any multiple) requires a rotation of $2\beta \ell = 2\pi$ around the center of the chart, bringing the point back to its original position, showing that the input impedance of a load seen through a $\lambda/2$ line is unchanged. We will now illustrate the use of the Smith chart for a variety of typical transmission line problems through examples.

EXAMPLE : A load impe

EXAMPLE 2.2 Basic Smith Chart Operations

A load impedance of $130 + j90 \Omega$ terminates a 50 Ω transmission line that is 0.3λ long. Find the reflection coefficient at the load, the reflection coefficient at the input to the line, the input impedance, the SWR on the line, and the return loss.

Solution

The normalized load impedance is

$$z_L = \frac{\vec{Z}_L}{\vec{Z}_0} = 2.60 + j1.80,$$

which can be plotted on the Smith chart as shown in Figure 2.11. Using a compass and the voltage coefficient scale below the chart, the reflection coefficient magnitude at the load can be read as |T| = 0.60. This same compass setting can then be applied to the standing wave ratio (SWR) scale to read SWR = 3.98, and to the return loss (in dB) scale to read RL = 4.4 dB. Now draw a radial line through the load impedance point, and read the angle of the reflection coefficient at the load from the outer scale of the chart as 21.8°.

We now draw an SWR circle through the load impedance point. Reading the reference position of the load on the wavelengths-toward-generator (WTG) scale gives a value of 0.220 λ . Moving down the line 0.3 λ toward the generator brings us to 0.520 λ on the WTG scale, which is equivalent to 0.020 λ . Drawing a radial line at this position gives the normalized input impedance at the intersection with SWR circle of $z_{in} = 0.255 + j0.117$. Then the input impedance of the line is

$$Z_{\rm in} = Z_0 z_{\rm in} = 12.7 + j5.8 \ \Omega.$$

The reflection coefficient at the load still has a magnitude of $|\Gamma| = 0.60$; the phase is read from the radial line at the input and the phase scale as 165.8° . \bigcirc

The Combined Impedance-Admittance Smith Chart

The Smith chart can be used for normalized admittance in the same way that it is used for normalized impedances, and it can be used to convert between impedance and admittance. The latter technique is based on the fact that, in normalized form, the input impedance of a load z_L connected to a $\lambda/4$ line is, from (2.44),

$$z_{in} = l/z_L,$$

which has the effect of converting a normalized impedance to a normalized admittance.

Since a complete revolution around the Smith chart corresponds to a length of $\lambda/2$, a $\lambda/4$ transformation is equivalent to rotating the chart by 180°; this is also equivalent to



FIGURE 2.11 Smith chart for Example 2.2.

imaging a given impedance (or admittance) point across the center of the chart to obtain the corresponding admittance (or impedance) point.

Thus, the same Smith chart can be used for both impedance and admittance calculations during the solution of a given problem. At different stages of the solution, then, the chart may be either an *impedance Smith chart* or an *admittance Smith chart*. This procedure can be made less confusing by using a Smith chart that has a superposition of the scales for a regular Smith chart and the scales of a Smith chart which has been rotated 180°, as shown in Figure 2.12. Such a chart is referred to as an *impedance and admittance Smith chart* and usually has different-colored scales for impedance and admittance.

F A

EXAMPLE 2.3 Smith Chart Operations Using Admittances

A load of $Z_L = 100 - j50 \ \Omega$ terminates a 50 Ω line. What are the load admittance and the input admittance if the line is $0.15\lambda \log ?$

Solution

The normalized load impedance is $z_L = 2 + j1$. A standard Smith chart can be used for this problem by initially considering it as an impedance chart and plotting z_L and the SWR circle. Conversion to admittance can be accomplished with a $\lambda/4$ rotation of z_L (easily obtained by drawing a straight line through z_L and the center of the chart to intersect the SWR circle). The chart can now be considered as an admittance chart, and the input admittance can be found by rotating 0.15λ from y_L .

Alternatively, we can use the combined zy chart of Figure 2.12, where conversion between impedance and admittance is accomplished merely by reading the appropriate scales. Plotting z_L on the impedances scales and reading the admittance scales at this same point give $y_L = 0.40 - j0.20$. The actual load



FIGURE 2.12 ZY Smith chart with solution for Example 2.3.

admittance is then

$$Y_L = y_L Y_0 = \frac{y_L}{Z_0} = 0.0080 - j0.0040$$
 S

Then, on the WTG scale, the load admittance is seen to have a reference position of 0.214 λ . Moving 0.15 λ past this point brings us to 0.364 λ . A radial line at this point on the WTG scale intersects the SWR circle at an admittance of y = 0.61 + j0.66. The actual input admittance is then V = 0.0122 + j0.0132 S.

The Slotted Line

A slotted line is a transmission line configuration (usually waveguide or coax) that allows the sampling of the electric field amplitude of a standing wave on a terminated line. With this device the SWR and the distance of the first voltage minimum from the load can be measured, and from this data the load impedance can be determined. Note that because the load impedance is in general a complex number (with two degrees of freedom), two distinct quantities must be measured with the slotted line to uniquely determine this impedance. A typical waveguide slotted line is shown in Figure 2.13.

Although the slotted line used to be the principal way of measuring an unknown impedance at microwave frequencies, it has been largely superseded by the modern vector



FIGURE 2.13

An X-band waveguide slotted line. Courtesy of Hewlett-Package Company, Santa Rosa, Culif. network analyzer in terms of accuracy, versatility, and convenience. The slotted line is still of some use, however, in certain applications such as high-millimeter wave frequencies, or where it is desired to avoid connector mismatches by connecting the unknown load directly to the slotted line, thus avoiding the use of imperfect transitions. Another reason for studying the slotted line is that it provides an unexcelled tool for learning basic concepts of standing waves and mismatched transmission lines. We will derive expressions for finding the unknown load impedance from slotted line measurements and also show how the Smith chart can be used for the same purpose.

Assume that, for a certain terminated line, we have measured the SWR on the line and ℓ_{\min} , the distance from the load to the first voltage minimum on the line. The load impedance Z_L can then be determined as follows. From (2.41) the magnitude of the reflection coefficient on the line is found from the standing wave ratio as

$$|\Gamma| = \frac{\mathrm{SWR} - 1}{\mathrm{SWR} + 1}.$$
 2.58

From Section 2.3, we know that a voltage minimum occurs when $e^{j(\theta-2\theta)} = -1$, where θ is the phase angle of the reflection coefficient, $\Gamma = |\Gamma|e^{j\theta}$. The phase of the reflection coefficient is then

$$\theta = \pi + 2\beta \ell_{\min},$$
 2.59

where ℓ_{\min} is the distance from the load to the first voltage minimum. Actually, since the voltage minimums repeat every $\lambda/2$, where λ is the wavelength on the line, any multiple of $\lambda/2$ can be added to ℓ_{\min} without changing the result in (2.59), because this just amounts to adding $2\beta n\lambda/2 = 2\pi n$ to θ , which will not change Γ . Thus, the two quantities SWR and ℓ_{\min} can be used to find the complex reflection coefficient Γ at the load. It is then straightforward to use (2.43) with $\ell = 0$ to find the load impedance from Γ :

$$Z_L = Z_0 \frac{1+\Gamma}{1-\Gamma}.$$
 2.60

The use of the Smith chart in solving this problem is best illustrated by an example.

EXAMPLE 2.4 Impedance Measurement with a Slotted Line

The following two-step procedure has been carried out with a 50 Ω slotted line to determine an unknown load impedance:

1. A short circuit is placed at the load plane, resulting in a standing wave on the line with infinite SWR, and sharply defined voltage minima, as shown in Figure 2.14a. On the arbitrarily positioned scale on the slotted line, voltage minima are recorded at

$$z = 0.2 \text{ cm}, 2.2 \text{ cm}, 4.2 \text{ cm}.$$



FIGURE 2.14 Voltage standing wave patterns for Example 2.4. (a) Standing wave for short-circuit load. (b) Standing wave for unknown load,

2. The short circuit is removed, and replaced with the unknown load. The standing wave ratio is measured as SWR = 1.5, and voltage minima, which are not as sharply defined as those in step 1, are recorded at

$$z = 0.72$$
 cm, 2.72 cm, 4.72 cm,

as shown in Figure 2.14b. Find the load impedance.

Solution

Knowing that voltage minima repeat every $\lambda/2$, we have from the data of step 1 above that $\lambda = 4.0$ cm. In addition, because the reflection coefficient and input impedance also repeat every $\lambda/2$, we can consider the load terminals to be effectively located at any of the voltage minima locations listed in step 1. Thus, if we say the load is at 4.2 cm, then the data from step 2 shows that the next voltage minimum away from the load occurs at 2.72 cm, giving $t_{min} =$ $4.2 - 2.72 = 1.48 \text{ cm} = 0.37\lambda.$

Applying (2.58)-(2.60) to this data gives

$$|\Gamma| = \frac{1.5 - 1}{1.5 + 1} = 0.2,$$

$$\theta = \pi + \frac{4\pi}{4.0}(1.48) = 86.4^{\circ}.$$

$$\Gamma = 0.2e^{j86.4^{\circ}} = 0.0126 + j0.1996.$$

SO

0+

The load impedance is then

$$Z_L = 50\left(\frac{1+\Gamma}{1-\Gamma}\right) = 47.3 + j19.7 \ \Omega.$$

For the Smith chart version of the solution, we begin by drawing the SWR circle for SWR = 1.5, as shown in Figure 2.15; the unknown normalized load impedance must lie on this circle. The reference that we have is that the load is 0.37λ away from the first voltage minimum. On the Smith chart, the position of a voltage minimum corresponds to the minimum impedance point (minimum voltage, maximum current), which is the horizontal axis (zero reactance) to the left of the origin. Thus, we begin at the voltage minimum point and move 0.37λ toward the load (counterclockwise), to the normalized load impedance point, $z_L = 0.95 + j0.4$, as shown in Figure 2.15. The actual load impedance is then $Z_L = 47.5 + j20 \Omega$, in close agreement with the above result using the equations.

Note that, in principle, voltage maxima locations could be used as well as voltage minima positions, but that voltage minima are more sharply defined than voltage maxima, and so usually result in greater accuracy.



2.5 THE QUARTER-WAVE TRANSFORMER

The quarter-wave transformer is a useful and practical circuit for impedance matching and also provides a simple transmission line circuit that further illustrates the properties of standing waves on a mismatched line. Although we will study the design and performance of quarter-wave matching transformers more extensively in Chapter 5, the main purpose here is the application of the previously developed transmission line theory to a basic transmission line circuit. We will first approach the problem from the impedance viewpoint, and then show how this result can also be interpreted in terms of an infinite set of multiple reflections on the matching section.

The Impedance Viewpoint

Figure 2.16 shows a circuit employing a quarter-wave transformer. The load resistance R_L , and the feedline characteristic impedance Z_0 , are both real and assumed to be given. These two components are connected with a lossless piece of transmission line of (unknown) characteristic impedance Z_1 and length $\lambda/4$. It is desired to match the load to the Z_0 line, by using the $\lambda/4$ piece of line, and so make $\Gamma = 0$ looking into the $\lambda/4$ matching section. From (2.44) the input impedance Z_{in} can be found as

$$Z_{\rm in} = Z_1 \frac{R_L + jZ_1 \tan \beta \ell}{Z_1 + jR_L \tan \beta \ell}.$$
 2.61

To evaluate this for $\mathcal{H} = (2\pi/\lambda)(\lambda/4) = \pi/2$, we can divide the numerator and denominator by tan $\beta \ell$ and take the limit as $\mathcal{H} \to \pi/2$ to get

$$Z_{\rm in} = \frac{\mathbb{Z}_1^2}{R_L}.$$

In order for $\Gamma = 0$, we must have $Z_{in} = Z_0$, which yields the characteristic impedance Z_1 as

$$Z_1 = \sqrt{Z_0 R_L}, \qquad 2.63$$

the geometric mean of the load and source impedances. Then there will be no standing waves on the feedline (SWR = 1), although there will be standing waves on the $\lambda/4$ matching section. Also, the above condition applies only when the length of the matching



____ FIGURE 2.16 The quarter-wave matching transformer.

section is $\lambda/4$, or an odd multiple (2n + 1) of $\lambda/4$ long, so that a perfect match may be achieved at one frequency, but mismatch will occur at other frequencies.



EXAMPLE 2.5 Frequency Response of a Quarter-Wave Transformer

Consider a load resistance $R_L = 100 \Omega$, to be matched to a 50 Ω line with a quarter-wave transformer. Find the characteristic impedance of the matching section and plot the magnitude of the reflection coefficient versus normalized frequency, f/f_o , where f_o is the frequency at which the line is $\lambda/4$ long.

Solution From (2.63), the necessary characteristic impedance is

$$Z_1 = \sqrt{(50)(100)} = 70.71 \ \Omega.$$

The reflection coefficient magnitude is given as

$$|\Gamma| = \left| rac{Z_{\mathrm{in}} - Z_0}{Z_{\mathrm{in}} + Z_0}
ight|,$$

where the input impedance Z_{in} is a function of frequency as given by (2.44). The frequency dependence in (2.44) comes from the $\beta \ell$ term, which can be written in terms of f/f_o as

$$\beta \ell = \left(\frac{2\pi}{\lambda}\right) \left(\frac{\lambda_0}{4}\right) = \left(\frac{2\pi f}{v_p}\right) \left(\frac{v_p}{4f_o}\right) = \frac{\pi f}{2f_o}.$$

where it is seen that $\beta \ell = \pi/2$ for $f = f_o$, as expected. For higher frequencies the line looks electrically longer, and for lower frequencies it looks shorter. The magnitude of the reflection coefficient is plotted versus f/f_o in Figure 2.17. \bigcirc





Reflection coefficient versus normalized frequency for the quarter-wave transformer of Example 2.5.

This method of impedance matching is limited to real load impedances, although a complex load impedance can easily be made real, at a single frequency, by transformation through an appropriate length of line.

The above analysis shows how useful the impedance concept can be when solving transmission line problems, and this method is probably the preferred method in practice. It may aid our understanding of the quarter-wave transformer (and other transmission line circuits), however, if we now look at it from the viewpoint of multiple reflections.

The Multiple Reflection Viewpoint

Figure 2.18 shows the quarter-wave transformer circuit with reflection and transmission coefficients defined as follows:

- Γ = overall, or total, reflection coefficient of a wave incident on the $\lambda/4$ -transformer (same as Γ in Example 2.5).
- Γ_1 = partial reflection coefficient of a wave incident on a load Z_1 , from the Z_0 line.
- Γ_2 = partial reflection coefficient of a wave incident on a load Z_0 , from the Z_1 line.
- Γ_3 = partial reflection coefficient of a wave incident on a load R_L , from the Z_1 line.
- T_1 = partial transmission coefficient of a wave from the \geq_0 line into the Z_1 line.
- T_3 = partial transmission coefficient of a wave from the \gtrsim_1 line into the Z_0 line.



___FIGURE 2.18

Multiple reflection analysis of the quarter-wave transformer.

These coefficients can then be expressed as

$$\Gamma_1 = \frac{Z_1 - Z_0}{Z_1 + Z_0}, \qquad 2.64a$$

$$\Gamma_2 = \frac{Z_0 - Z_1}{Z_0 + Z_1} = -\Gamma_1, \qquad 2.64b$$

$$\Gamma_3 = \frac{R_L - Z_1}{R_L + Z_1}, \qquad 2.64c$$

$$T_1 = \frac{2Z_1}{Z_1 + Z_0}, 2.64d$$

$$T_2 = \frac{2Z_0}{Z_1 + Z_0}.$$
 2,64e

Now think of the quarter-wave transformer of Figure 2.18 in the time domain, and imagine a wave traveling down the Z_0 feedline toward the transformer. When the wave first hits the junction with the Z_1 line, it sees only an impedance Z_1 since it has not yet traveled to the load R_L and can't see that effect. Part of the wave is reflected with a coefficient Γ_1 , and part is transmitted onto the Z_1 line with a coefficient T_1 . The transmitted wave then travels $\lambda/4$ to the load, is reflected with a coefficient Γ_3 , and travels another $\lambda/4$ back to the junction with the Z_0 line. Part of this wave is transmitted through (to the left) to the Z_0 line, with coefficient T_2 , and part is reflected back toward the load with coefficient Γ_2 . Clearly, this process continues with an infinite number of bouncing waves, and the total reflection coefficient. Γ , is the sum of all of these partial reflections. Since each round trip path up and down the $\lambda/4$ transformer section results in a 180° phase shift (90° up and 90° down), the total reflection coefficient can be expressed as

$$\Gamma = \Gamma_1 - T_1 T_2 \Gamma_3 + T_1 T_2 \Gamma_2 \Gamma_3^2 - T_1 T_2 \Gamma_2^2 \Gamma_3^3 + \cdots$$

= $\Gamma_1 - T_1 T_2 \Gamma_3 \sum_{n=0}^{\infty} (-\Gamma_2 \Gamma_3)^n$.
2.65

Since $|\Gamma_3| < 1$ and $|\Gamma_2| < 1$, the infinite series in (2.65) can be summed using the geometric series result that

$$\sum_{n=0}^{\infty} x^n = \frac{1}{1-x}, \qquad \text{for } |x| < 1,$$

to give

$$\Gamma = \Gamma_1 - \frac{T_1 T_2 \Gamma_3}{1 + \Gamma_2 \Gamma_3} = \frac{\Gamma_1 + \Gamma_1 \Gamma_2 \Gamma_3 - T_1 T_2 \Gamma_3}{1 + \Gamma_2 \Gamma_3}.$$
 2.66

The numerator of this expression can be simplified using (2.64) to give

$$\begin{split} \Gamma_1 - \Gamma_3 \left(\Gamma_1^2 + T_1 T_2 \right) &= \Gamma_1 - \Gamma_3 \left[\frac{(Z_1 - Z_0)^2}{(Z_1 + Z_0)^2} + \frac{4Z_1 Z_0}{(Z_1 + Z_0)^2} \right] \\ &= \Gamma_1 - \Gamma_3 = \frac{(Z_1 - Z_0)(R_L + Z_1) - (R_L - Z_1)(Z_1 + Z_0)}{(Z_1 + Z_0)(R_L + Z_1)} \\ &= \frac{2(Z_1^2 - Z_0 R_L)}{(Z_1 + Z_0)(R_L + Z_1)}, \end{split}$$

which is seen to vanish if we choose $Z_1 = \sqrt{Z_0 R_L}$, as in (2.63). Then T of (2.66) is zero, and the line is matched. This analysis shows that the matching property of the quarter-wave transformer comes about by properly selecting the characteristic impedance and length of the matching section so that the superposition of all the partial reflections add to zero. Under steady-state conditions, an infinite sum of waves traveling in the same direction with the same phase velocity can be combined into a single traveling wave. Thus, the infinite set of waves traveling in the forward and reverse directions on the matching section can be reduced to two waves, traveling in opposite directions. See Problem 2.24.

2.6 GENERATOR AND LOAD MISMATCHES

In Section 2.3 we treated the terminated (mismatched) transmission line assuming that the generator was matched, so that no reflections occurred at the generator. In general, however, both generator and load may present mismatched impedances to the transmission line. We will study this case, and also see that the condition for maximum power transfer from the generator to the load may, in some situations, require a standing wave on the line.

Figure 2.19 shows a transmission line circuit with arbitrary generator and load impedances, Z_g and Z_ℓ , which may be complex. The transmission line is assumed to be lossless, with a length ℓ and characteristic impedance Z_0 . This circuit is general enough to model most passive and active networks that occur in practice.

Because both the generator and load are mismatched, multiple reflections can occur on the line, as in the problem of the quarter-wave transformer. The present circuit





Transmission line circuit for mismatched load and generator.

could thus be analyzed using an infinite series to represent the multiple bounces, as in Section 2.5, but we will use the easier and more useful method of impedance transformation. The input impedance looking into the terminated transmission line from the generator end is, from (2.43) and (2.44),

$$Z_{\rm in} = Z_0 \frac{1 + \Gamma_\ell e^{-2j\beta\ell}}{1 - \Gamma_\ell e^{-2j\beta\ell}} = Z_0 \frac{Z_\ell + jZ_0 \tan\beta\ell}{Z_0 + jZ_\ell \tan\beta\ell}, \qquad 2.67$$

where Γ_t is the reflection coefficient of the load:

$$\Gamma_{\ell} = rac{Z_{\ell} - Z_0}{Z_{\ell} + Z_0} \; .$$

The voltage on the line can be written as

$$V(z) = V_o^+ \left(e^{-j\beta z} + \Gamma_\ell e^{j\beta z} \right), \qquad 2.69$$

and we can find V_o^+ from the voltage at the generator end of the line, where $z = -\ell$:

$$V(-\ell) = V_g \frac{Z_{\rm in}}{Z_{\rm in} + Z_g} = V_o^+ \left(e^{j\beta\ell} + \Gamma_\ell e^{-j\beta\ell} \right),$$

so that

$$V_{o}^{+} = V_{g} \frac{Z_{in}}{Z_{in} + Z_{g}} \frac{1}{(e^{j\beta\ell} + \Gamma_{\ell} e^{-j\beta\ell})}.$$
 2.70

This can be rewritten, using (2.67), as

$$V_{o}^{+} = V_{g} \frac{Z_{0}}{Z_{0} + Z_{g}} \frac{e^{-j\beta\ell}}{(1 - \Gamma_{\ell}\Gamma_{g}e^{-2j\beta\ell})},$$
 2.71

where Γ_g is the reflection coefficient seen looking into the generator:

$$\Gamma_g = \frac{Z_g - Z_0}{Z_g + Z_0}.$$
 2.72

The standing wave ratio on the line is then

$$SWR = \frac{1 + |\Gamma_{\ell}|}{1 - |\Gamma_{\ell}|}.$$
 2.73

The power delivered to the load is

$$P = \frac{1}{2} \operatorname{Re}\{V_{\text{in}}I_{\text{in}}^*\} = \frac{1}{2}|V_{\text{in}}|^2 \operatorname{Re}\left\{\frac{1}{Z_{\text{in}}}\right\} = \frac{1}{2}|V_g|^2 \left|\frac{Z_{\text{in}}}{Z_{\text{in}} + Z_g}\right|^2 \operatorname{Re}\left\{\frac{1}{Z_{\text{in}}}\right\}.$$
 2.74

Now let $Z_{in} = R_{in} + jX_{in}$ and $Z_g = R_g + jX_g$; then (2.74) can be reduced to

$$P = \frac{1}{2} |V_g|^2 \frac{R_{\rm in}}{(R_{\rm in} + R_g)^2 + (X_{\rm in} + X_g)^2}.$$
 2.75

We now assume that the generator impedance, Z_g , is fixed, and consider three cases of load impedance.

Load Matched to Line

In this case we have $Z_l = Z_0$, so $\Gamma_l = 0$, and SWR = 1, from (2.68) and (2.73). Then the input impedance is $Z_{in} = Z_0$, and the power delivered to the load is, from (2.75),

$$P = \frac{1}{2} |V_g|^2 \frac{Z_0}{(Z_0 + R_g)^2 + X_g^2}.$$
 2.76

Generator Matched to Loaded Line

In this case the load impedance Z_{ℓ} and/or the transmission line parameters $\partial \ell$, Z_0 are chosen to make the input impedance $Z_{in} = Z_g$, so that the generator is matched to the load presented by the terminated transmission line. Then the overall reflection coefficient, Γ , is zero:

$$\Gamma = \frac{Z_{\rm in} - Z_g}{Z_{\rm in} + Z_g} = 0.$$
 2.77

There may, however, be a standing wave on the line since Γ_{ℓ} may not be zero. The power delivered to the load is

$$P = \frac{1}{2} |V_g|^2 \frac{R_g}{4(R_g^2 + X_g^2)}.$$
 2.78

Now observe that even though the loaded line is matched to the generator, the power delivered to the load may be less than the power delivered to the load from (2.76), where the loaded line was not necessarily matched to the generator. Thus, we are led to the question of what is the optimum load impedance, or equivalently, what is the optimum input impedance, to achieve maximum power transfer to the load for a given generator impedance.

Conjugate Matching

Assuming that the generator series impedance, Z_g , is fixed, we may vary the input impedance Z_{in} until we achieve the maximum power delivered to the load. Knowing Z_{in} , it is then easy to find the corresponding load impedance Z_l via an impedance transformation along the line. To maximize P, we differentiate with respect to the real and imaginary parts of Z_{in} . Using (2.75) gives

$$\frac{\partial P}{\partial R_{\rm in}} = 0 \rightarrow \frac{1}{(R_{\rm in} + R_g)^2 + (X_{\rm in} + X_g)^2} + \frac{-2R_{\rm in}(R_{\rm in} + R_g)}{\left[(R_{\rm in} + R_g)^2 + (X_{\rm in} + X_g)^2\right]^2} = 0.$$

or,

$$R_{g}^{2} - R_{in}^{2} + (X_{in} + X_{g})^{2} = 0, \qquad 2.79a$$

$$\frac{\partial P}{\partial X_{in}} = 0 \rightarrow \frac{-2X_{in}(X_{in} + X_{g})}{\left[(R_{in} + R_{g})^{2} + (X_{in} + X_{g})^{2}\right]^{2}} = 0, \qquad 2.79b$$

οг,

Solving (2.79a,b) simultaneously for R_{in} and X_{in} gives

$$R_{
m in}=R_g, \qquad X_{
m in}=-X_g,$$

 $Z_{
m in}=Z_g^*$. 2.80

ÖГ

This condition is known as conjugate matching, and results in maximum power transfer to the load, for a fixed generator impedance. The power delivered is, from (2.75) and (2.80),

$$P = \frac{1}{2} |V_g|^2 \frac{1}{4R_g}, \qquad 2.81$$

which is seen to be greater than or equal to the powers of (2.76) or (2.78). Also note that the reflection coefficients Γ_{ℓ} , Γ_g , and Γ may be nonzero. Physically, this means that in some cases the power in the multiple reflections on a mismatched line may add in phase to deliver more power to the load than would be delivered if the line were flat (no reflections). If the generator impedance is real ($X_g = 0$), then the last two cases reduce to the same result, which is that maximum power is delivered to the load when the loaded line is matched to the generator ($R_{in} = R_g$, with $X_{in} = X_g = 0$).

Finally, note that neither matching for zero reflection $(Z_{\ell} = Z_0)$ or conjugate matching $(Z_{in} = Z_g^*)$ necessarily yields a system with the best efficiency. For example, if $Z_g = Z_{\ell} = Z_0$ then both load and generator are matched (no reflections), but only half the power produced by the generator is delivered to the load (half is lost in Z_g), for a transmission efficiency of 50%. This efficiency can only be improved by making Z_g as small as possible.

2.7 LOSSY TRANSMISSION LINES

In practice, all transmission lines have loss due to finite conductivity and/or lossy dielectric, but these losses are usually small. In many practical problems, then, loss may be neglected, but at times the effect of loss may be of interest. Such is the case when dealing with the attenuation of a transmission line, or the Q of a resonant cavity, for example. In this section we will study the effects of loss on transmission line behavior and show how the attenuation constant can be calculated.

The Low-Loss Line

In most practical microwave transmission lines the loss is small—if this were not the case, the line would be of little practical value. When the loss is small, some approximations can be made that simplify the expressions for the general transmission line parameters of $\gamma = \alpha + j\beta$ and Z_0 .

The general expression for the complex propagation constant is, from (2.5),

$$\gamma = \sqrt{(R + j\omega L)(G + j\omega C)}, \qquad 2.82$$

which can be rearranged as

$$\gamma = \sqrt{(j\omega L)(j\omega C)\left(1 + \frac{R}{j\omega L}\right)\left(1 + \frac{G}{j\omega C}\right)}$$
$$= j\omega\sqrt{LC}\sqrt{1 - j\left(\frac{R}{\omega L} + \frac{G}{\omega C}\right) - \frac{RG}{\omega^2 LC}}.$$
 2.83

If the line is low-loss we can assume that $R \ll \omega L$ and $G \ll \omega C$, which means that both the conductor loss and dielectric loss are small. Then, $RG \ll \omega^2 LC$, and (2.83) reduces to

$$\gamma = j\omega\sqrt{LC}\sqrt{1-j\left(\frac{R}{\omega L}+\frac{G}{\omega C}\right)}.$$
 2.84

If we were to ignore the $(R/\omega L + G/\omega C)$ term, we would obtain the result that γ was purely imaginary (no loss), so we will instead use the first two terms of the Taylor series expansion for $\sqrt{1+x} \simeq 1 + x/2 + \cdots$, to give the first higher order real term for γ :

$$\gamma \simeq j\omega\sqrt{LC}\left[1-\frac{j}{2}\left(\frac{R}{\omega L}+\frac{G}{\omega C}\right)\right].$$

so that

$$\boldsymbol{\alpha} \simeq \frac{1}{2} \left(R \sqrt{\frac{C}{L}} + G \sqrt{\frac{L}{C}} \right) = \frac{1}{2} \left(\frac{R}{Z_0} + G Z_0 \right).$$
 2.85*a*

$$\beta \simeq \omega \sqrt{LC},$$
 2.85b

where $Z_0 = \sqrt{L/C}$ is the characteristic impedance of the line in the absence of loss. Note from (2.85b) that the propagation constant β is the same as the lossless case of (2.12). By the same order of approximation, the characteristic impedance Z_0 can be approximated as a real quantity:

$$Z_0 = \sqrt{\frac{R + j\omega L}{G + j\omega C}} \simeq \sqrt{\frac{L}{C}} .$$
 2.86

Equations (2.85)–(2.86) are known as the high-frequency, low-loss approximations for transmission lines, and are important because they show that the propagation constant and characteristic impedance for a low-loss line can be closely approximated by considering the line as lossless.



EXAMPLE 2.6 Attenuation Constant of the Coaxial Line

In Example 2.1 the L, C, R, and G parameters were derived for a lossy coaxial line. Assuming the loss is small, calculate the attenuation constant from (2.85a) and the results of Example 2.1.

Solution From (2.85a),

$$\alpha = \frac{1}{2} \left(R \sqrt{\frac{C}{L}} + G \sqrt{\frac{L}{C}} \right).$$

Using the results derived in Example 2.1 gives

$$\alpha = \frac{1}{2} \left[\frac{R_s}{\eta \ln b/a} \left(\frac{1}{a} + \frac{1}{b} \right) + \omega \epsilon'' \eta \right].$$

where $\eta = \sqrt{\mu/\epsilon'}$ is the intrinsic impedance of the dielectric material filling the coaxial line. Also, $\beta = \omega\sqrt{LC} = \omega\sqrt{\mu\epsilon'}$, and $Z_0 = \sqrt{L/C} = (\eta/2\pi) \ln b/a$.

The above method for the calculation of attenuation requires that the line parameters L, C, R, and G be known. These can often be derived using the formulas of (2.17)–(2.20), but a more direct and versatile procedure is to use the perturbation method, to be discussed shortly.

The Distortionless Line

As can be seen from the exact equations (2.82) and (2.83) for the propagation constant of a lossy line, the phase term β is generally a complicated function of frequency, ω , when loss is present. In particular, we note that β is generally not exactly a linear function of frequency, as in (2.85b), unless the line is lossless. If β is not a linear function of frequency (of the form $\beta = a\omega$), then the phase velocity $v_p = \omega/\beta$ will be different for different frequencies ω . The implication is that the various frequency components of a wideband signal will travel with different phase velocities, and so arrive at the receiver end of the transmission line at slightly different times. This will lead to *dispersion*, a distortion of the signal, and is generally an undesirable effect. Granted, as we have argued above, the departure of β from a linear function may be quite small, but the effect can be significant if the line is very long. This effect leads to the concept of group velocity, which we will address in detail in Section 3.10.

There is a special case, however, of a lossy line that has a linear phase factor as a function of frequency. Such a line is called a distortionless line, and is characterized by line parameters that satisfy the relation

$$\frac{R}{L} = \frac{G}{C}.$$
 2.87
From (2.83) the exact complex propagation constant, under the condition specified by (2.87), reduces to

$$\begin{aligned} \gamma &= j\omega\sqrt{LC}\sqrt{1-2j\frac{R}{\omega L}-\frac{R^2}{\omega^2 L^2}} \\ &= j\omega\sqrt{LC}\left(1-j\frac{R}{\omega L}\right) \\ &= R\sqrt{\frac{C}{L}}+j\omega\sqrt{LC}=\alpha+j\beta, \end{aligned}$$
2.88

which shows that $\beta = \omega \sqrt{LC}$ is a linear function of frequency. Equation (2.88) also shows that the attenuation constant, $\alpha = R \sqrt{C/L}$, is not a function of frequency, so that all frequency components will be attenuated by the same amount (actually, R is usually a weak function of frequency). Thus, the distortionless line is not loss free, but is capable of passing a pulse or modulation envelope without distortion. To obtain a transmission line with parameters that satisfy (2.87) often requires that L be increased by adding series loading coils spaced periodically along the line.

The above theory for the distortionless line was first developed by Oliver Heaviside (1850–1925), a reclusive genius who, with no formal education, solved many problems in transmission line theory and worked Maxwell's original theory of electromagnetism into the modern and more usable version that we are familiar with today [5].

The Terminated Lossy Line

Figure 2.20 shows a length *i* of a lossy transmission line terminated in a load impedance Z_L . Thus, $\gamma = \alpha + j\beta$ is complex, but we assume the loss is small so that Z_0 is approximately real, as in (2.86).

In (2.36), expressions for the voltage and current wave on a lossless line are given. The analogous expressions for the lossy case are

$$V(z) = V_{\sigma}^{+} \left[e^{-\gamma z} + \Gamma e^{\gamma z} \right]_{\ell}$$
 2.89*a*

$$I(z) = \frac{V_o^+}{Z_0} \left[e^{-\gamma z} - \zeta e^{\gamma z} \right], \qquad 2.89b$$

where Γ is the reflection coefficient of the load, as given in (2.35), and V_{e}^{\pm} is the incident voltage amplitude referenced at z = 0. From (2.42), the reflection coefficient at



FIGURE 2.20

0 A lossy transmission line terminated in the impedance $Z_{L_{i}}$

a distance ℓ from the load is

$$\Gamma(\ell) = \Gamma e^{-2j\beta\ell} e^{-2\alpha\ell} = \Gamma e^{-2\gamma\ell}.$$
 2.90

The input impedance Z_{in} at a distance ℓ from the load is then

$$Z_{in} = \frac{V(-\ell)}{I(-\ell)} = Z_0 \frac{Z_L + Z_0 \tanh \gamma \ell}{Z_0 + Z_L \tanh \gamma \ell}.$$
 2.91

We can compute the power delivered to the input of the terminated line at $z = -\ell$ as

$$P_{\rm in} = \frac{1}{2} \operatorname{Re} \left\{ V(-\ell) I^*(-\ell) \right\} = \frac{|V_o^+|^2}{2Z_0} \left[e^{2\alpha \ell} - |\Gamma|^2 e^{-2\alpha \ell} \right]$$

= $\frac{|V_o^+|^2}{2Z_0} \left[1 - |\Gamma(\ell)|^2 \right] e^{2\alpha \ell},$ (2.92)

where (2.89) have been used for $V(-\ell)$ and $I(-\ell)$. The power actually delivered to the load is

$$P_L = \frac{1}{2} \operatorname{Re} \left\{ V(0) I^*(0) \right\} = \frac{|V_o^+|^2}{2Z_0} (1 - |\Gamma|^2).$$
 2.93

The difference in these powers corresponds to the power lost in the line:

$$P_{\text{loss}} = P_{\text{in}} - P_L = \frac{|V_o^+|^2}{2Z_0} \left[\left(e^{2\alpha \ell} - 1 \right) + |\Gamma|^2 \left(1 - e^{-2\alpha \ell} \right) \right].$$
 2.94

The first term in (2.94) accounts for the power loss of the incident wave, while the second term accounts for the power loss of the reflected wave; note that both terms increase as α increases.

The Perturbation Method for Calculating Attenuation

Here we derive a useful and standard technique for finding the attenuation constant of a low-loss line. The method avoids the use of the transmission line parameters L, C, R, and G, and instead uses the fields of the lossless line. with the assumption that the fields of the lossy line are not greatly different from the fields of the lossless line—hence the term. *perturbation*.

We have seen that the power flow along a lossy transmission line, in the absence of reflections, is of the form

$$P(z) = P_o e^{-2\alpha z}, \qquad 2.95$$

where P_{α} is the power at the z = 0 plane, and α is the attenuation constant we wish to determine. Now define the power loss per unit length along the line as

$$P_{\ell} = \frac{-\partial P}{\partial z} = 2\alpha P_o e^{-2\alpha z} = 2\alpha P(z),$$

where the negative sign on the derivative was chosen so that P_t would be a positive quantity. From this, the attenuation constant can be determined as

$$\alpha = \frac{P_{\ell}(z)}{2P(z)} = \frac{P_{\ell}(z=0)}{2P_{o}}.$$
 2.96

This equation states that α can be determined from P_{α} , the power on the line, and $P_{\ell_{\alpha}}$ the power loss per unit length of line. It is important to realize that P_{ℓ} can be computed from the fields of the lossless line, and can account for both conductor loss (using 1.131) and dielectric loss (using 1.92).

EXAMPLE 2.7 Using the Perturbation Method to Find the Attenuation Constant

Use the perturbation method to find the attenuation constant of a coaxial line having a lossy dielectric and lossy conductors.

Solution

From Example 2.1 and (2.32), the fields of the lossless coaxial line are, for $a < \rho < b$.

$$ar{E} = rac{V_o \hat{p}}{
ho \ln b/a} e^{-jeta z},$$
 $ar{H} = rac{V_o \hat{\phi}}{2\pi
ho Z_0} e^{-jeta z}.$

where $Z_0 = (\eta/2\pi) \ln b/a$ is the characteristic impedance of the coaxial line and V_{α} is the voltage across the line at z = 0. The first step is to find P_{α} , the power flowing on the lossless line:

$$P_{\rho} = \frac{1}{2} \operatorname{Re} \, \int_{S} \bar{E} \times \bar{H}^{*} \cdot d\bar{s} = \frac{|V_{\sigma}|^{2}}{2Z_{0}} \, \int_{\rho=a}^{b} \, \int_{\phi=0}^{2\pi} \frac{\rho \, d\rho \, d\phi}{2\pi \rho^{2} \ln b/a} = \frac{|V_{\sigma}|^{2}}{2Z_{0}}.$$

as expected from basic circuit theory.

The loss per unit length, P_{t} , comes from conductor loss (P_{tr}) and dielectric loss (P_{td}). From (1.131), the conductor loss in a 1 m length of line can be found as

$$P_{\ell \sigma} = \frac{R_s}{2} \int_S |\bar{H}_t|^2 ds = \frac{R_s}{2} \int_{z=0}^1 \left\{ \int_{\phi=0}^{2\pi} |H_{\phi}(\rho = a)|^2 a \, d\phi + \int_{\phi=0}^{2\pi} |H_{\phi}(\rho = b)|^2 b \, d\phi \right\} dz$$
$$= \frac{R_s |V_0|^2}{4\pi Z_0^2} \left(\frac{1}{a} + \frac{1}{b}\right).$$

The dielectric loss in a 1 m length of line is, from (1.92),

$$P_{\ell d} = \frac{\omega \epsilon''}{2} \int_{V} |\bar{E}|^2 ds = \frac{\omega \epsilon''}{2} \int_{\rho = a}^{b} \int_{\phi = 0}^{2\pi} \int_{z=0}^{1} |E_{\rho}|^2 \rho d\rho d\phi dz = \frac{\pi \omega \epsilon''}{\ln b/a} |V_{\sigma}|^2,$$

where ϵ'' is the imaginary part of the complex dielectric constant, $\epsilon = \epsilon' - j\epsilon''$. Finally, applying (2.96) gives

$$\alpha = \frac{P_{\ell c} + P_{\ell d}}{2P_o} = \frac{R_s}{4\pi Z_0} \left(\frac{1}{a} + \frac{1}{b}\right) + \frac{\pi \omega \epsilon'' Z_0}{\ln b/a}$$
$$= \frac{R_s}{2\eta \ln b/a} \left(\frac{1}{a} + \frac{1}{b}\right) + \frac{\omega \epsilon'' \eta}{2},$$

where $\eta = \sqrt{\mu/\epsilon'}$. This result is seen to agree with that of Example 2.6.

The Wheeler Incremental Inductance Rule

Another useful technique for the practical evaluation of attenuation due to conductor loss for TEM or quasi-TEM lines is the Wheeler incremental inductance rule [6]. This method is based on the similarity of the equations for the inductance per unit length and resistance per unit length of a transmission line, as given by (2.17) and (2.19), respectively. In other words, the conductor loss of a line is due to current flow inside the conductor which, as was shown in Section 1.7, is related to the tangential magnetic field at the surface of the conductor, and thus to the inductance of the line.

From (1.131), the power loss into a cross section S of a good (but not perfect) conductor is

$$P_{\ell} = \frac{R_s}{2} \int_{S} |\bar{J}_s|^2 ds = \frac{R_s}{2} \int_{S} |\bar{H}_t|^2 ds \ W/m^2, \qquad 2.97$$

so the power loss per unit length of a uniform transmission line is

$$P_{\ell} = \frac{R_s}{2} \int_C |\bar{H}_t|^2 \, d\ell \, \mathbf{W/m}, \qquad 2.98$$

where the line integral of (2.98) is over the cross-sectional contours of both conductors. Now, from (2.17), the inductance per unit length of the line is

$$L = \frac{\mu}{|I|^2} \int_S |\bar{H}|^2 \, \mathrm{d}s, \qquad 2.99$$

which is computed assuming the conductors are lossless. When the conductors have a small loss, the \bar{H} field in the conductor is no longer zero, and this field contributes a small additional "incremental" inductance, ΔL , to that of (2.99). As discussed in Chapter 1, the fields inside the conductor decay exponentially so that the integration into the conductor dimension can be evaluated as

$$\Delta L = \frac{\mu_0 \delta_s}{2 |I|^2} \int_C |\bar{H}_t|^2 \, d\ell.$$
 2.100

since $\int_0^\infty e^{-2z/\delta_s} dz = \delta_s/2$. (The skin depth is $\delta_s = \sqrt{2/\omega\mu\sigma}$.) Then P_{ℓ} from (2.98) can be written in terms of ΔL as

$$P_{\ell} = \frac{R_s |I|^2 \Delta L}{\mu_0 \delta_s} = \frac{|I|^2 \Delta L}{\sigma \mu_0 \delta_s^2} = \frac{|I|^2 \omega \Delta L}{2}$$
 W/m, 2.101

since $R_s = \sqrt{(\omega\mu_0)/2\sigma} = 1/(\sigma\delta_s)$. Then from (2.96) the attenuation due to conductor loss can be evaluated as

$$\alpha_{c} = \frac{P_{\ell}}{2P_{o}} = \frac{\omega \Delta L}{2Z_{0}},$$
 2.102

since P_{σ} , the total power flow down the line, is $P_{\sigma} = |I|^2 Z_0/2$, where Z_0 is the characteristic impedance of the line. In (2.102), ΔL is evaluated as the change in inductance when all conductor walls are receded by an amount $\delta_s/2$.

Equation (2.102) can also be written in terms of the change in characteristic impedance, since

$$Z_0 = \sqrt{\frac{L}{C}} = \frac{L}{\sqrt{LC}} = Lr_p, \qquad 2.103$$

so that

$$\alpha_c = \frac{\beta \Delta Z_0}{2Z_0}.$$
 2.104

where ΔZ_0 is the change in characteristic impedance when all conductor walls are receded by an amount $\delta_s/2$. Yet another form of the incremental inductance rule can be obtained by using the first two terms of a Taylor series expansion for Z_0 . Thus,

$$Z_0\left(\frac{\delta_s}{2}\right) \simeq Z_0 + \frac{\delta_s}{2} \frac{dZ_0}{d\ell}, \qquad 2.105$$

so that

$$\Delta Z_0 = Z_0 \left(\frac{\delta_s}{2}\right) \quad Z_0 = \frac{\delta_s}{2} \frac{dZ_0}{d\ell},$$

where $Z_0(\delta_s/2)$ refers to the characteristic impedance of the line when the walls are receded by $\delta_s/2$, and t refers to a distance into the conductors. Then (2.104) can be written as

$$q_{c} = \frac{3\delta_{s}}{4Z_{0}}\frac{dZ_{0}}{d\ell} = \frac{R_{s}}{2Z_{0}\eta}\frac{dZ_{0}}{d\ell}.$$
 2.106

where $\eta = \sqrt{\mu_0/\epsilon}$ is the intrinsic impedance of the dielectric, and R_s is the surface resistivity of the conductor. Equation (2.106) is one of the most practical forms of the incremental inductance rule, because the characteristic impedance is known for a wide variety of transmission lines.

EXAMPLE 2.8 Using the Wheeler Incremental Inductance Rule to Find the Attenuation Constant

Calculate the attenuation due to conductor loss of a coaxial line using the incremental inductance rule.

Solution

From (2.32) the characteristic impedance of the coaxial line is

$$Z_0=rac{\eta}{2\pi}\,\ln\,rac{b}{a}.$$

Then, using the incremental inductance rule of the form in (2.106), the attenuation due to conductor loss is

$$\alpha_c = \frac{R_s}{2Z_0\eta} \frac{dZ_0}{d\ell} = \frac{R_s}{4\pi Z_0} \left\{ \frac{d \ln b/a}{db} - \frac{d \ln b/a}{da} \right\} = \frac{R_s}{4\pi Z_0} \left(\frac{1}{b} + \frac{1}{a} \right),$$

which is seen to be in agreement with the result of Example 2.7. The negative sign on the second differentiation in the above equation is because the derivative for the inner conductor is in the $-\rho$ direction (receding wall).

Regardless of how attenuation is calculated, measured attenuation constants for practical lines are usually higher. The main reason for this discrepancy is the fact that realistic transmission lines have metallic surfaces that are somewhat rough, which increases the loss, while our theoretical calculations assume perfectly smooth conductors. A quasiempirical formula that can be used to correct for surface roughness for any transmission line is [7]

$$\ddot{\alpha}_{c}' = \alpha_{c} \left[1 + \frac{2}{\pi} \tan^{-1} 1.4 \left(\frac{\Delta}{\delta_{s}} \right)^{2} \right]. \qquad 2.107$$

where α_c is the attenuation due to perfectly smooth conductors, α'_c is the attenuation corrected for surface roughness, Δ is the rms surface roughness, and δ_s is the skin depth of the conductors.

REFERENCES

- S. Ramo, J. R. Winnery, and T. Van Duzer, Fields and Waves in Communication Electronics, Third Edition, John Wiley & Sons, N.Y., 1994.
- [2] J. A. Stratton, Electromagnetic Theory, McGraw-Hill, N.Y., 1941.
- [3] H. A. Wheeler, "Reflection Charts Relating to Impedance Matching," IEEE Trans. Microwave Theory and Techniques, vol. MTT-32, pp. 1008-1021, September 1984.
- [4] P. H. Smith, "Transmission Line Calculator," Electronics, vol. 12, No. 1, pp. 29-31, January 1939.
- [5] P. J. Nahin, Oliver Heaviside: Sage in Solitude, IEEE Press, N.Y., 1988.
- [6] H. A. Wheeler, "Formulas for the Skin Effect," Proc. IRE, vol. 30, pp. 412-424, September 1942.
- [7] T. C. Edwards, Foundations for Microstrip Circuit Design, John Wiley & Sons, N.Y., 1987.

PROBLEMS

2.1 A transmission line has the following per unit length parameters: $L = 0.2 \ \mu$ H/m, $C = 300 \ \text{pF/m}$, $R = 5 \ \Omega/\text{m}$, and $G = 0.01 \ \text{S/m}$. Calculate the propagation constant and characteristic impedance of this line at 500 MHz. Recalculate these quantities in the absence of loss (R = G = 0).

2.2 Show that the following *T*-model of a transmission line also yields the telegrapher equations derived in Section 2.1.



2.3 For the parallel plate line shown below, derive the R, L, G, and C parameters. Assume $\underline{w} \ge d$,



- 2.4 For the parallel plate line of Problem 2.3, derive the telegrapher equations using the field theory approach.
- **2.5** A certain coaxial line has copper conductors with an inner conductor diameter of 1 mm and an outer conductor diameter of 3 mm. The dielectric filling has $\epsilon_r = 2.8$ with a loss tangent of 0.005. Compute the *R*, *L*, *G*, and *C* parameters of this line at 3 GHz and the characteristic impedance and phase velocity.
- **2.6** Compute and plot the attenuation of the coaxial line of Problem 2.5. in dB/m, over a frequency range of 1 MHz to 10 GHz. Use log-log graph paper.
- 2.7 A lossless transmission line of electrical length $t = 0.3\lambda$ is terminated with a complex load impedance as shown below. Find the reflection coefficient at the load, the SWR on the line, and the input impedance to the line.



2.8 A lossless transmission line is terminated with a 100 Ω load. If the SWR on the line is 1.5, find the two possible values for the characteristic impedance of the line.

- **2.9** A radio transmitter is connected to an antenna having an impedance $80 + j40 \Omega$ with a 50 Ω coaxial cable. If the 50 Ω transmitter can deliver 30 W when connected to a 50 Ω load, how much power is delivered to the antenna?
- **2.10** A 75 Ω coaxial transmission line has a length of 2.0 cm and is terminated with a load impedance of 37.5 + j75 Ω . If the dielectric constant of the line is 2.56 and the frequency is 3.0 GHz, find the input impedance to the line, the reflection coefficient at the load, the reflection coefficient at the input, and the SWR on the line,
- 2.11 Calculate SWR, reflection coefficient magnitude, and return loss values to complete the entries in the following table:

SWR	$ \Gamma $	<i>RL</i> (d B)
1.00	0.00	∞
1.01		_
	0.01	_
1.05		
		30.0
1.10		
1.20		
	0.10	—
1.50		_
_		10.0
2.00	-	
2.50	—	

2.12 The transmission line circuit on the next page has V_g = 15 V rms, Z_g = 75 Ω, Z₀ = 75 Ω, Z_L = 60 - j40 Ω, and l = 0.7λ. Compute the power delivered to the load using three different techniques:
(a) find Γ and compute

$$P_L = \left(rac{V_g}{2}
ight)^2 rac{1}{Z_0} \left(1 - |\Gamma|^2
ight);$$

(b) find Z_{in} and compute

٠

$$P_L = \left| \frac{V_g}{Z_g + Z_{in}} \right|^2 \operatorname{Re}\left(Z_{in}\right); \text{ and}$$

(c) find V_L and compute

$$P_L = \left|\frac{V_L}{Z_L}\right|^2 \operatorname{Re}(Z_L).$$

Discuss the rationale for each of these methods. Which of these methods can be used if the line is not lossless?

_



- **2.13** For a purely reactive load impedance of the form $Z_L = {}_J X$, show that the reflection coefficient magnitude $|\Gamma|$ is always unity. Assume the characteristic impedance Z_0 is real.
- **2.14** Consider the transmission line circuit shown below. Compute the incident power, the reflected power, and the power transmitted into the infinite 75 Ω line. Show that power conservation is satisfied.



2.15 A generator is connected to a transmission line as shown below. Find the voltage as a function of z along the transmission line. Plot the magnitude of this voltage for $-\ell \le z \le 0$.



- **2.16** A load impedance of $Z_L = 80 + \sqrt{20} \Omega$ is to be matched to a $Z_0 = 100 \Omega$ line using a length t of lossless line of characteristic impedance Z_1 . Find the required Z_1 (real) and t.
- 2.17 Use the Smith chart to find the following quantities for the transmission line circuit below:
 - (a) The SWR on the line.
 - (b) The reflection coefficient at the load.
 - (c) The load admittance.
 - (d) The input impedance of the line,
 - (e) The distance from the load to the first voltage minimum.
 - (f) The distance from the load to the first voltage maximum.



- **2.18** Repeat problem 2.17 for $Z_L = 40 j30 \ \Omega$.
- **2.19** Repeat problem 2.17 for $\ell = 1.8\lambda$.
- **2.20** Use the Smith chart to find the shortest lengths of a short-circuited 75 Ω line to give the following input impedance:
 - (a) $Z_{in} = 0.$
 - **(b)** $Z_{in} = \infty$.
 - (c) $Z_{ln} = j75 \ \Omega$.
 - (d) $Z_{in} = -j50 \ \Omega$.
 - (e) $Z_{in} = j10 \ \Omega$.
- **2.21** Repeat Problem 2.20 for an open-circuited length of 75 Ω line.
- **2.22** A slotted-line experiment is performed with the following results: distance between succesive minima = 2.1 cm; distance of first voltage minimum from load = 0.9 cm; SWR of load = 2.5. If $Z_0 = 50 \Omega$, find the load impedance.
- **2.23** Design a quarter-wave matching transformer to match a 40 Ω load to a 75 Ω line. Plot the SWR for $0.5 \le f/f_o \le 2.0$, where f_o is the frequency at which the line is $\lambda/4$ long.
- **2.24** Consider the quarter-wave matching transformer circuit shown below. Derive expressions for V^+ and V^- , the amplitudes of the forward and reverse traveling waves on the quarter-wave line section, in terms of V^+ , the incident voltage amplitude.



- 2.25 Derive equation (2.71) from (2.70).
- 2.26 In Example 2.7, the attenuation of a coaxial line due to finite conductivity is

$$\alpha_c = \frac{R_s}{2\eta \ln b/a} \left(\frac{1}{a} + \frac{1}{b}\right).$$

Show that α_c is minimized for conductor radii such that $x \ln x = 1 + x$, where x = b/a. Solve this equation for x, and show that the corresponding characteristic impedance for $\epsilon_r = 1$ is 77 Ω .

- 2.27 Compute and plot the factor by which attenuation is increased due to surface roughness, for rms roughness ranging from zero to 0.01 mm. Assume copper conductors at 10 GHz.
- **2.28** A 50 Ω transmission line is matched to a 10 V source and feeds a load $Z_L = 100 \Omega$. If the line is 2.3 λ long and has an attenuation constant $\alpha = 0.5 \text{ dB}/\lambda$, find the powers that are delivered by the source, lost in the line, and delivered to the load.
- **2.29** Consider a nonreciprocal transmission line having different propagation constants, β^+ and β^- , for propagation in the forward and reverse directions, with corresponding characteristic impedances

 Z_0^+ and Z_0^- . (An example of such a line could be a microstrip transmission line on a magnetized ferrite substrate.) If the line is terminated as shown below, derive expressions for the reflection coefficient and impedance seen at the input of the line.



Transmission Lines and Waveguides

One of the early milestones in microwave engineering was the development of waveguide and other transmission lines for the low-loss transision of microwave power. Although Heaviside considered the possibility of propagation of electromagnetic waves inside a closed hollow tube in 1893, he rejected the idea because he believed that two conductors were necessary for the transfer of electromagnetic energy [1]. In 1897, Lord Rayleigh (John William Strutt) [2] mathematically proved that wave propagation in waveguides was possible, for both circular and rectangular cross sections. Rayleigh also noted the infinite set of modes of the TE and TM type that were possible and the existence of a cutoff frequency, but no experimental verification was made at the time. The waveguide was essentially forgotten until it was rediscovered independently in 1936 by two men [3]. After preliminary experiments in 1932. George C. Southworth of the AT&T Company in New York presented a paper on the waveguide in 1936. At the same meeting, W. L. Barrow of MTT presented a paper on the circular waveguide, with experimental confirmation of propagation.

Early microwave systems relied on waveguide and coaxial lines for transmission line media. Waveguide has the advantage of high power-handling capability and low loss but is bulky and expensive. Coaxial line has very high bandwidth and is convenient for test applications, but is a difficult medium in which to fabricate complex microwave components. Planar transmission lines provide an alternative, in the form of stripline, microstrip, slotline, coplanar waveguide, and many other types of related geometries. Such transmission lines are compact, low in cost, and are capable of being easily integrated with active devices such as diodes and transistors to form microwave integrated circuits. The first planar transmission line may have been a flat-strip coaxial line, similar to stripline, used in a production power divider network in World War II [4]. But planar lines did not receive intensive development until the 1950s. Microstrip line was developed at ITT laboratories [5] and was a competitor of stripline. The first microstrip lines used a relatively thick dielectric substrate, which accentuated the non-TEM mode behavior and frequency dispersion of the line. This characteristic made it less desirable than stripline until the 1960s, when much thinner substrates began to be used. This reduced the frequency dependence of the line, and now microstrip is often the preferred medium for microwave integrated circuits.

In this chapter we will study the properties of several types of transmission lines and waveguides that are in common use today. As we know from Chapter 2, a transmission

104

line is characterized by a propagation constant and a characteristic impedance; if the line is lossy, attenuation is also of interest. These quantities will be derived by a field theory analysis for the various lines and waveguides treated here.

We will begin with a general discussion of the different types of wave propagation and modes that can exist on transmission lines and waveguides. Transmission lines that consist of two or more conductors may support transverse electromagnetic (TEM) waves, characterized by the lack of longitudinal field components. TEM waves have a uniquely defined voltage, current, and characteristic impedance. Waveguides, often consisting of a single conductor, support transverse electric (TE) and/or transverse magnetic (TM) waves, characterized by the presence of longitudinal magnetic or electric, respectively, field components. As we will see in Chapter 4, a unique definition of characteristic impedance is not possible for such waves, although definitions can be chosen so that the characteristic impedance concept can be used for waveguides with meaningful results.

3.1 GENERAL SOLUTIONS FOR TEM, TE, AND TM WAVES

In this section we will find general solutions to Maxwell's equations for the specific cases of TEM. TE, and TM wave propagation in cylindrical transmission lines or waveguides. The geometry of an arbitrary transmission line or waveguide is shown in Figure 3.1, and is characterized by conductor boundaries that are parallel to the z-axis. These structures are assumed to be uniform in the z direction and infinitely long. The conductors will initially be assumed to be perfectly conducting, but attenuation can be found by the perturbation method discussed in Chapter 2.

We assume time-harmonic fields with an e^{j+t} dependence and wave propagation along the z-axis. The electric and magnetic fields can then be written as

$$\bar{E}(x, y, z) = [\bar{e}(x, y) + \hat{z}e_z(x, y)]e^{-jBz}, \qquad 3.1a$$

$$\bar{H}(x,y,z) = [\bar{h}(x,y) + \hat{z}h_{z}(x,y)]e^{-j\beta z}, \qquad 3.1t$$



FIGURE 3.1 (a) General two-conductor transmission line and (b) closed waveguide.

where $\bar{e}(x, y)$ and $\bar{h}(x, y)$ represent the transverse (\hat{x}, \hat{y}) electric and magnetic field components, while e_z and h_z are the longitudinal electric and magnetic field components. In the above, the wave is propagating in the +z direction; -z propagation can be obtained by replacing β by $-\beta$. Also, if conductor or dielectric loss is present, the propagation constant will be complex; $j\beta$ should then be replaced with $\gamma = \alpha + j\beta$.

Assuming that the transmission line or waveguide region is source free. Maxwell's equations can be written as

$$abla imes ar{E} = -j\omega\muar{H},$$
 3.2a

$$abla imes \bar{H} = j\omega\epsilon \bar{E}.$$
 3.2b

With an $e^{-j\beta z} z$ dependence, the three components of each of the above vector equations can be reduced to the following:

$$rac{\partial E_z}{\partial y} + jeta E_y = -j\omega\mu H_x, \qquad 3.3a$$

$$-j\beta E_x - \frac{\partial E_z}{\partial x} = -j\omega\mu H_y, \qquad 3.3b$$

$$\frac{\partial E_y}{\partial x} - \frac{\partial E_x}{\partial y} = -j\omega\mu H_z, \qquad 3.3c$$

$$\frac{\partial H_z}{\partial y} + j\beta H_y = j\omega\epsilon E_x, \qquad 3.4a$$

$$-j\beta H_x - \frac{\partial H_z}{\partial x} = j\omega\epsilon E_y, \qquad 3.4b$$

$$\frac{\partial H_y}{\partial x} - \frac{\partial H_x}{\partial y} = j\omega\epsilon E_z. \qquad 3.4c$$

The above six equations can be solved for the four transverse field components in terms of E_z and H_z (for example, H_x can be derived by eliminating E_y from (3.3a) and (3.4b)) as follows:

$$H_x = \frac{j}{k_c^2} \left(\omega \epsilon \frac{\partial E_z}{\partial y} - \beta \frac{\partial H_z}{\partial x} \right), \qquad 3.5a$$

$$H_y = \frac{-j}{k_c^2} \left(\omega \epsilon \frac{\partial E_z}{\partial x} + \beta \frac{\partial H_z}{\partial y} \right), \qquad 3.5b$$

$$E_x = \frac{-j}{k_c^2} \left(\beta \frac{\partial E_z}{\partial x} + \omega \mu \frac{\partial H_z}{\partial y} \right), \qquad 3.5c$$

$$E_y = \frac{j}{k_c^2} \left(-\beta \frac{\partial E_z}{\partial y} + \omega \mu \frac{\partial H_z}{\partial x} \right), \qquad 3.5d$$

where

$$k_c^2 = k^2 - \beta^2, 3.6$$

has been defined as the cutoff wavenumber; the reason for this terminology will become

.

clear later. As in previous chapters,

$$k = \omega \sqrt{\mu \epsilon} = 2\pi / \lambda$$
 3.7

is the wavenumber of the material filling the transmission line or waveguide region. If dielectric loss is present, ϵ can be made complex by using $\epsilon = \epsilon_0 \epsilon_r (1 - j \tan \delta)$, where $\tan \delta$ is the loss tangent of the material.

Equations (3.5a-d) are very useful general results that can be applied to a variety of waveguiding systems. We will now specialize these results to specific wave types.

TEM Waves

or

Transverse electromagnetic (TEM) waves are characterized by $E_z = H_z = 0$. Observe from (3.5) that if $E_z = H_z \approx 0$, then the transverse fields are also all zero, unless $k_c^2 = 0$ ($k^2 = \beta^2$), in which case we have an indeterminate result. Thus, we can return to (3.3)-(3.4) and apply the condition that $E_z = H_z = 0$. Then from (3.3a) and (3.4b), we can eliminate H_x to obtain

$$eta^2 E_y = \omega^2 \mu \epsilon E_y, \ eta = \omega \sqrt{\mu \epsilon} = k,$$
 3.8

as noted earlier. (This result can also be obtained from (3.3b) and (3.4a).) The cutoff wavenumber. $k_c = \sqrt{k^2 - b^2}$, is thus zero for TEM waves.

Now the Helmholtz wave equation for E_x is, from (1.42).

but for $e^{-j\beta z}$ dependence, $(\partial^2/\partial z^2)E_x = -\beta^2 E_x = -k^2 E_x$, so (3.9) reduces to

$$\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2}\right) E_x = 0.$$
 \$.10

A similar result also applies to E_y , so using the form of \bar{E} assumed in (3.1a) we can write

$$\nabla_t^2 \bar{e}(x, y) = 0. \tag{3.11}$$

where $\nabla_t^2 = \partial^2 / \partial x^2 + \partial^2 / \partial y^2$ is the Laplacian operator in the two transverse dimensions.

The result of (3.11) shows that the transverse electric fields, $\bar{e}(x, y)$, of a TEM wave satisfy Laplace's equation. It is easy to show in the same way that the transverse magnetic fields also satisfy Laplace's equation:

$$\nabla_t^2 \tilde{h}(x, y) = 0. \tag{3.12}$$

The transverse fields of a TEM wave are thus the same as the static fields that can exist between the conductors. In the electrostatic case, we know that the electric field can be expressed as the gradient of a scalar potential, $\Phi(x, y)$:

$$\overline{e}(x,y) = -\nabla_t \Phi(x,y), \qquad 3.13$$

where $\nabla_t = \hat{x}(\partial/\partial x) + \hat{y}(\partial/\partial y)$ is the transverse gradient operator in two dimensions. In order for the relation in (3.13) to be valid, the curl of \bar{e} must vanish, and this is indeed the case here since

$$abla_t imes ilde{e} = -j\omega\mu h_z oldsymbol{\hat{z}} = 0.$$

Using the fact that $\nabla \cdot \overline{D} = \epsilon \nabla_t \cdot \overline{e} = 0$ with (3.13) shows that $\Phi(x, y)$ also satisfies Laplace's equation,

$$\nabla_t^2 \Phi(x, y) = 0, \qquad 3.14$$

as expected from electrostatics. The voltage between two conductors can be found as

$$V_{12} = \Phi_1 - \Phi_2 = \int_1^2 \vec{E} \cdot d\vec{\ell},$$
 3.15

where Φ_1 and Φ_2 represent the potential at conductors 1 and 2, respectively. The current flow on a conductor can be found from Ampere's law as

$$I = \oint_C \bar{H} \cdot d\bar{\ell}, \qquad 3.16$$

where C is the cross-sectional contour of the conductor.

TEM waves can exist when two or more conductors are present. Plane waves are also examples of TEM waves, since there are no field components in the direction of propagation: in this case the transmission line conductors may be considered to be two infinitely large plates separated to infinity. The above results show that a closed conductor (such as a rectangular waveguide) cannot support TEM waves, since the corresponding static potential in such a region would be zero (or possibly a constant). leading to $\tilde{e} = 0$.

The wave impedance of a TEM mode can be found as the ratio of the transverse electric and magnetic fields:

$$Z_{\text{TEM}} = \frac{E_x}{H_y} = \frac{\omega\mu}{\beta} = \sqrt{\frac{\mu}{\epsilon}} = \eta, \qquad \qquad 3.17a$$

where (3.4a) was used. The other pair of transverse field components, from (3.3a), give

$$Z_{\text{TEM}} = \frac{-E_y}{H_x} = \sqrt{\frac{\mu}{\epsilon}} = \eta.$$
 3.17b

Combining the results of (3.17a) and (3.17b) gives a general expression for the transverse fields as

$$\bar{h}(x,y) = \frac{1}{Z_{\text{TEM}}} \hat{z} \times \bar{e}(x,y). \qquad 3.18$$

Note that the wave impedance is the same as that for a plane wave in a lossless medium, as derived in Chapter 1: the reader should not confuse this impedance with the characteristic impedance. Z_0 , of a transmission line. The latter relates an incident voltage and current and is a function of the line geometry as well as the material filling the line, while the wave impedance relates transverse field components and is dependent only on the material

constants. From (2.32), the characteristic impedance of the TEM line is $Z_0 = V/I$, where V and I are the amplitudes of the incident voltage and current waves.

The procedure for analyzing a TEM line can be summarized as follows:

- 1. Solve Laplace's equation, (3.14), for $\Phi(x, y)$. The solution will contain several unknown constants.
- Find these constants by applying the boundary conditions for the known voltages on the conductors.
- **3.** Compute $\bar{\epsilon}$ and \bar{E} from (3.13), (3.1a), Compute \bar{h} , \bar{H} from (3.18), (3.1b),
- **4.** Compute V from (3.15), I from (3.16).
- 5. The propagation constant is given by (3.8), and the characteristic impedance is given by $Z_0 = V/I$.

TE Waves

Transverse electric (TE) waves, (also referred to as *H*-waves) are characterized by $E_x = 0$ and $H_{iz} \neq 0$. Equations (3.5) then reduce to

$$H_x = \frac{-j\beta}{k_c^2} \frac{\partial H_z}{\partial x}, \qquad 3.19a$$

$$\boldsymbol{H}_{\boldsymbol{y}} = \frac{-j\beta}{k_c^2} \frac{\partial H_z}{\partial y}.$$
 3.19b

$$E_x = \frac{-j\omega\mu}{k_c^2} \frac{\partial H_z}{\partial y}, \qquad \qquad 3.19c$$

$$E_y = \frac{j\omega\mu}{k_c^2} \frac{\partial H_z}{\partial r}.$$
 3.19d

In this case, $k_c \neq 0$, and the propagation constant $\beta = \sqrt{k^2 - k_c^2}$ is generally a function of frequency and the geometry of the line or guide. To apply (3.19), one must first find H_z from the Helmholtz wave equation.

$$\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + \frac{\partial^2}{\partial z^2} + k^2\right) H_{-} = 0.$$
 3.20

which, since $H_z(x, y, z) = h_z(x, y)e^{-j\beta z}$, can be reduced to a two-dimensional wave equation for h_z :

$$\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + k_c^2\right)h_z = 0, \qquad 3.21$$

since $k_c^2 = k^2 - \beta^2$. This equation must be solved subject to the boundary conditions of the specific guide geometry.

The TE wave impedance can be found as

$$Z_{\rm TE} = \frac{E_x}{H_y} = \frac{-E_y}{H_x} = \frac{\omega\mu}{\beta} = \frac{k\eta}{\beta}, \qquad 3.22$$

which is seen to be frequency dependent. TE waves can be supported inside closed conductors, as well as between two or more conductors.

TM Waves

Transverse magnetic (TM) waves (also referred to as *E*-waves) are characterized by $E_z \neq 0$ and $H_z = 0$. Equations (3.5) then reduce to

$$H_x = \frac{j\omega\epsilon}{k_c^2} \frac{\partial E_z}{\partial y},$$
 3.23a

$$H_y = \frac{-j\omega\epsilon}{k_c^2} \frac{\partial E_z}{\partial x},$$
 3.23b

$$E_x = rac{-jeta}{k_{
m c}^2}rac{\partial E_z}{\partial x},$$
 3.23c

$$E_y = rac{-jeta}{k_c^2}rac{\partial E_z}{\partial y}.$$
 3.23d

As in the TE case, $k_c \neq 0$, and the propagation constant $\beta = \sqrt{k^2 - k_c^2}$ is a function of frequency and the geometry of the line or guide. E_z is found from the Helmholtz wave equation,

$$\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + \frac{\partial^2}{\partial z^2} + k^2\right) E_z = 0, \qquad 3.24$$

which, since $E_z(x, y, z) = e_z(x, y)e^{-j\beta z}$, can be reduced to a two-dimensional wave equation for e_z :

$$\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + k_c^2\right)e_z = 0, \qquad 3.25$$

since $k_c^2 = k^2 - \beta^2$. This equation must be solved subject to the boundary conditions of the specific guide geometry.

The TM wave impedance can be found as

$$Z_{\rm TM} = \frac{E_x}{H_y} = \frac{-E_y}{H_x} = \frac{\beta}{\omega\epsilon} = \frac{\beta\eta}{k}, \qquad 3.26$$

which is frequency dependent. As for TE waves, TM waves can be supported inside closed conductors, as well as between two or more conductors.

The procedure for analyzing TE and TM waveguides can be summarized as follows:

- 1. Solve the reduced Helmholtz equation, (3.21) or (3.25), for h_z or e_z . The solution will contain several unknown constants, and the unknown cutoff wavenumber, k_c .
- 2. Use (3.19) or (3.23) to find the transverse fields from h_z or e_z .

- 3. Apply the boundary conditions to the appropriate field components to find the unknown constants and k_c .
- 4. The propagation constant is given by (3.6), and the wave impedance by (3.22) or (3.26).

Attenuation Due to Dielectric Loss

Attenuation in a transmission line or waveguide can be caused by either dielectric loss or conductor loss. If α_d is the attenuation constant due to dielectric loss, and α_c is the attenuation constant due to conductor loss, then the total attenuation constant is $\alpha \neq \alpha_d + \alpha_c$.

Attenuation caused by conductor loss can be calculated using the perturbation method of Section 2.7: this loss depends on the field distribution in the guide and so must be evaluated separately for each type of transmission line or waveguide. But if the line or guide is completely filled with a homogeneous dielectric, the attenuation due to lossy dielectric can be calculated from the propagation constant, and this result will apply to any guide or line with a homogeneous dielectric tilling.

Thus, using the complex dielectric constant allows the complex propagation constant to be written as

$$\gamma = \alpha_d + j\beta = \sqrt{k_c^2 - k^2}$$
$$= \sqrt{k_c^2 - \omega^2 \mu_0 \epsilon_0 \epsilon_r (1 - j \tan \delta)}.$$
 3.27

In practice, most dielectric materials have a very small loss (tan $\delta << 1$), so this expression can be simplified by using the first two terms of the Taylot expansion.

$$\sqrt{a^2 + x^2} \simeq a + \frac{1}{2} \left(\frac{x^2}{a}\right)$$
. for $x \ll a$.

Then (3.27) reduces to

$$\gamma = \sqrt{k_c^2 - k^2 + jk^2 \tan \delta}$$

$$\approx \sqrt{k_c^2 - k^2} + \frac{jk^2 \tan \delta}{2\sqrt{k_c^2 - k^2}}$$

$$= \frac{k^2 \tan \delta}{2\beta} + j\vec{\beta};$$
3.28

since $\sqrt{k_c^2 - k^2} = j\beta$. In these results, $k^2 = \omega^2 \mu_0 \epsilon_0 \epsilon_\tau$ is the (real) wavenumber in the absence of loss. Equation (3.28) shows that when the loss is small the phase constant, β , is unchanged, while the attenuation constant due to dielectric loss is given by

$$\alpha_d = \frac{k^2 \tan b}{2\beta}$$
 Np/m (TE or TM waves). 3.29

This result applies to any TE or TM wave, as long as the guide is completely filled with the dielectric. It can also be used for TEM lines, where $k_c = 0$, by letting $\beta = k$:

$$\alpha_d = \frac{k \tan \delta}{2} \text{ Np/m (TEM waves).} \qquad 3.30$$

3.2 PARALLEL PLATE WAVEGUIDE

The parallel plate waveguide is probably the simplest type of guide that can support TM and TE modes; it can also support a TEM mode, since it is formed from two flat plates, or strips, as shown in Figure 3.2. Although an idealization, this guide is also important for practical reasons, since its operation is quite similar to that of a variety of other waveguides, and models the propagation of higher order modes in stripline.

In the geometry of the parallel plate waveguide in Figure 3.2, the strip width W is assumed to be much greater than the separation, d, so that fringing fields and any x variation can be ignored. A material with permittivity ϵ and permeability μ is assumed to fill the region between the two plates. We will discuss solutions for TEM, TM, and TE waves.

TEM Modes

As discussed in Section 3.1, the TEM mode solution can be obtained by solving Laplace's equation. (3.14), for the electrostatic potential $\Phi(x, y)$ between the two plates. Thus,

$$\nabla_t^2 \Phi(x, y) = 0, \qquad \text{for } 0 \le x \le W, \quad 0 \le y \le d.$$
3.31

If we assume that the bottom plate is at ground (zero) potential and the top plate at a potential of V_a , then the boundary conditions for $\Phi(x, y)$ are

$$\Phi(x,0)=0, \qquad \qquad 3.32a$$

$$\Phi(x,d) = V_o. \qquad 3.32b$$



FIGURE 3.2 Geometry of a parallel plate waveguide.

Since there is no variation in x, the general solution to (3.31) for $\Phi(x, y)$ is

$$\Phi(x,y) = A + By.$$

and the constants A, B can be evaluated from the boundary conditions of (3.32) to give the final solution as

$$\Phi(x,y) = V_0 y/d, \qquad 3.33$$

The transverse electric field is, from (3.13).

$$\bar{e}(x,y) = -\nabla_t \Phi(x,y) = -\hat{y} \frac{V_o}{d}, \qquad 3.34$$

so that the total electric field is

$$\bar{E}(x,y,z) = \tilde{e}(x,y)e^{-jkz} = -\tilde{y}\frac{V_0}{d}e^{-jkz}, \qquad 3.35$$

where $k = \omega \sqrt{\mu \epsilon}$ is the propagation constant of the TEM wave, as in (3.8). The magnetic field, from (3.18), is

$$\bar{H}(x,y,z) = \frac{1}{\eta} \hat{z} \times \bar{E}(x,y,z) = \hat{x} \frac{V_o}{\eta d} e^{-jkz}, \qquad 3.36$$

where $\eta = \sqrt{\mu/\epsilon}$ is the intrinsic impedance of the medium between the parallel plates. Note that $E_{z} = H_{z} = 0$ and that the fields are similar in form to a plane wave in a homogeneous region.

The voltage of the top plate with respect to the bottom plate can be calculated from (3.15) and (3.35) as

$$V = -\int_{y=0}^{d} E_y dy = V_0 e^{-jkz}, \qquad 3.37$$

as expected. The total current on the top plate can be found from Ampere's law or the surface current density:

$$I = \int_{x=0}^{w} \bar{J}_s \cdot \hat{z} dx = \int_{x=0}^{w} (-\hat{y} \times \bar{H}) \cdot \hat{z} dx = \int_{x=0}^{w} H_s dx = \frac{u V_0}{\eta d} e^{-jkz}, \qquad 3.38$$

Thus the characteristic impedance can be found as

$$Z_0 = \frac{V}{I} = \frac{\eta d}{w}.$$
 3.39

which is seen to be a constant dependent only on the geometry and material parameters of the guide. The phase velocity is also a constant:

$$v_{p} = \frac{\omega}{\beta} = \frac{1}{\sqrt{\mu\epsilon}}.$$
 3.40

which is the speed of light in the material medium.

Attenuation due to dielectric loss is given by (3.30). The formula for conductor attenuation will be derived in the next subsection, as a special case of TM mode attenuation,

TM Modes

As discussed in Section 3.1, TM waves are characterized by $H_z = 0$ and a nonzero E_z field that satisfies the reduced wave equation of (3.25), with $\partial/\partial x = 0$:

$$\left(\frac{\partial^2}{\partial y^2} + k_c^2\right) e_z(x, y) = 0.$$
 3.41

where $k_c^2 = k^2 - \beta^2$ is the cutoff wavenumber, and $E_z(x, y, z) = e_z(x, y)e^{-i\beta z}$. The general solution to (3.41) is of the form

$$e_z(x,y) = A \sin k_c y + B \cos k_c y, \qquad 3.42$$

subject to the boundary conditions that

$$e_z(x,y) = 0,$$
 at $y = 0, d.$ 3.43

This implies that B = 0 and $k_c d = n\pi$, for $n = 0, 1, 2, 3, \dots$, or

$$k_c = \frac{n\pi}{d}$$
. $n = 0, 1, 2, 3...$ 3.44

Thus the cutoff wavenumber k_c is constrained to discrete values as given by (3.44); this implies that the propagation constant β is given by

$$\beta = \sqrt{k^2 - k_c^2} = \sqrt{k^2 - (n\pi/d)^2}.$$
 3.45

The solution for $e_x(x, y)$ is then

$$e_z(x,y) = A_n \sin \frac{n\pi y}{d}, \qquad 3.46$$

thus,

$$E_z(x, y, z) = A_n \sin \frac{n\pi y}{d} e^{-j\beta z}.$$
 3.47

The transverse field components can be found, using (3.23), to be

$$H_x = \frac{j\omega\epsilon}{k_c} A_n \cos \frac{n\pi y}{d} e^{-j\beta z}, \qquad 3.48a$$

$$E_y = \frac{-\hat{j}\beta}{\cdot k_c} A_n \cos \frac{n\pi y}{d} e^{-j\beta z}, \qquad 3.48b$$

$$E_x = H_y = 0. \tag{3.48c}$$

Observe that for n = 0, $\beta = k = \omega \sqrt{\mu \epsilon}$, and that $E_z = 0$. The E_y and H_x fields are then constant in y, so that the TM_o mode is actually identical to the TEM mode. For $n \ge 1$, however, the situation is different. Each value of n corresponds to a different TM mode, denoted as the TM_n mode, and each mode has its own propagation constant given by (3.45), and field expressions as given by (3.48).

From (3.45) it can be seen that β is real only when $k > k_c$. Since $k = \omega \sqrt{\mu \epsilon}$ is proportional to frequency, the TM_n modes (for n > 0) exhibit a cutoff phenomenon, whereby no propagation will occur until the frequency is such that $k > k_c$. The cutoff

frequency of the TM_{μ} mode can then be deduced as

$$f_c = \frac{k_c}{2\pi\sqrt{\mu\epsilon}} = \frac{n}{2d\sqrt{\mu\epsilon}},$$
 3.49

Thus, the TM mode that propagates at the lowest frequency is the TM₁ mode, with a cutoff frequency of $f_c = 1/2d\sqrt{\mu\epsilon}$; the TM₂ mode has a cutoff frequency equal to twice this value, and so on. At frequencies below the cutoff frequency of a given mode, the propagation constant is purely imaginary, corresponding to a rapid exponential decay of the fields. Such modes are referred to as cutoff, or evanescent, modes. TM_n mode propagation is analogous to a high-pass filter response.

The wave impedance of the TM modes, from (3.26), is a function of frequency;

$$Z_{\rm TM} = \frac{-E_y}{H_x} = \frac{\beta}{\omega\epsilon} = \frac{\beta\eta}{k}.$$
 3.50

which we see is pure real for $f > f_c$, but pure imaginary for $f < f_c$. The phase velocity is also a function of frequency:

$$v_p = \frac{\omega}{\beta}, \qquad \qquad 3.51$$

and is seen to be greater than $1/\sqrt{\mu\epsilon} = \omega/k$, the speed of light in the medium, since $\beta < k$. The guide wavelength is defined as

$$\lambda_g = \frac{2\pi}{3}, \qquad \qquad \mathbf{3.52}$$

and is the distance between equiphase planes along the z-axis. Note that $\lambda_g > \lambda = 2\pi/k$, the wavelength of a plane wave in the material. The phase velocity and guide wavelength are defined only for a propagating mode, for which β is real. One may also define a cutoff wavelength as

$$\lambda_e = 2d. \qquad \qquad 3.53$$

It is instructive to compute the Poynting vector to see how power propagates in the TM_n mode. From (1.91), the time-average power passing a transverse cross section of the parallel plate guide is

$$P_{o} = \frac{1}{2} \operatorname{Re} \int_{x=0}^{w} \int_{y=0}^{d} \tilde{E} \times \tilde{H}^{*} \cdot 2dy \, dx = -\frac{1}{2} \operatorname{Re} \int_{x=0}^{w} \int_{y=0}^{d} E_{y} H_{x}^{*} \, dy \, dx$$
$$= \frac{w \operatorname{Re}(\vartheta) \omega \epsilon}{2k_{c}^{2}} |A_{n}|^{2} \int_{y=0}^{d} \cos^{2} \frac{n \pi y}{d} \, dy = \begin{cases} \frac{w \operatorname{Re}(\vartheta) \omega \epsilon d}{4k_{c}^{2}} |A_{n}|^{2} & \text{for } n > 0\\ \frac{w \operatorname{Re}(\vartheta) \omega \epsilon d}{2k_{c}^{2}} |A_{n}|^{2} & \text{for } n = 0 \end{cases}$$

where (3.48,a,b) were used for E_y , H_x . Thus, P_a is positive and nonzero when β is real, which occurs for $f > f_c$. When the mode is below cutoff, β is imaginary and so $P_o = 0$.

The TM (or TE) waveguide mode propagation has an interesting interpretation when viewed as a pair of bouncing plane waves. For example, consider the dominant TM_1 mode, which has a propagation constant,

$$\beta_1 = \sqrt{k^2 - (\pi/d)^2}.$$
 3.55

and E_z field,

$$E_z = A_1 \, \sin \, rac{\pi y}{d} e^{-jeta_1 z}$$

which can be rewritten as

$$E_{z} = \frac{A_{1}}{2j} \{ e^{j(\pi y/d - \beta_{1}z)} - e^{-j(\pi y/d + \beta_{1}z)} \}.$$
 3.56

This result is in the form of two plane waves traveling obliquely. in the -y, +z and +y, +z directions, respectively, as shown in Figure 3.3. By comparison with the phase factor of (1.132), the angle θ that each plane wave makes with the z-axis satisfies the relations

$$k\sin\theta = \frac{\pi}{d},\qquad \qquad 3.57a$$

$$k\cos\theta = \beta_1, \qquad \qquad 3.57b$$

so that $(\pi/d)^2 + \beta_1^2 = k^2$, as in (3.55). For $f > f_c, \beta$ is real and less than k_1 , so θ is some angle between 0° and 90° , and the mode can be thought of as two plane waves alternately bouncing off of the top and bottom plates.

The phase velocity of each plane wave along its direction of propagation (θ direction) is $\omega/k = 1/\sqrt{\mu\epsilon}$, which is the speed of light in the material filling the guide. But the phase velocity of the plane waves in the z direction is $\omega/\beta_1 = 1/\sqrt{\mu\epsilon}\cos\theta$, which is greater than the speed of light in the material. (This situation is analogous to ocean waves hitting a shoreline: the intersection point of the shore and an obliquely incident wave crest moves faster than the wave crest itself.) The superposition of the two plane wave fields is such that complete cancellation occurs at y = 0 and y = d, to satisfy the boundary condition that $E_z = 0$ at these planes. As f decreases to f_c , β_1 approaches zero so that, by (3.57b), θ approaches 90°. The two plane waves are then bouncing up and down with no motion in the +z direction, and no real power flow occurs in the z direction.

Attenuation due to dielectric loss can be found from (3.29). Conductor loss can be treated using the perturbation method. Thus,

$$\alpha_c = \frac{P_\ell}{2P_g}, \qquad \qquad 3.58$$





where P_o is the power flow down the guide in the absence of conductor loss, as given by (3.54). P_{ℓ} is the power dissipated per unit length in the two lossy conductors and can be found from (2.97) as

$$P_{\ell} = 2\left(\frac{R_s}{2}\right) \int_{x=0}^{w} |\bar{J}_s|^2 dx = \frac{\omega^2 \epsilon^2 R_s w}{k_c^2} |A_n|^2, \qquad 3.59$$

where R_s is the surface resistivity of the conductors. Using (3.54) and (3.59) in (3.58) gives the attenuation due to conductor loss as

$$\boldsymbol{\alpha}_n = \frac{2\omega\epsilon R_s}{\beta d} = \frac{2kR_s}{\beta\eta d}$$
 Np/m. for $n > 0$. 3.60

As discussed previously, the TEM mode is identical to the TM_0 mode for the parallel plate waveguide, so the above attenuation results for the TM_n mode can be used to obtain the TEM mode attenuation by letting n = 0. For attenuation due to conductor loss for the TEM mode, the n = 0 result of (3.54) must be used in (3.58), to obtain

$$\mathbf{\alpha}_{c} = \frac{R_{\star}}{\eta d} \text{ Np/m.}$$
 3.61

TE Modes

TE modes, characterized by $E_z = 0$, can also propagate on the parallel plate waveguide. From (3.21), with $\partial/\partial x = 0$, H_z must satisfy the reduced wave equation.

$$\left(\frac{\partial^2}{\partial y^2} + k_c^2\right) h_z(x, y) = 0.$$
3.62

where $k_c^2 = k^2 - \beta^2$ is the cutoff wavenumber and $H_z(x, y, z) = h_z(x, y)e^{-\beta\beta z}$. The general solution to (3.62) is

$$h_x(x, y) = A\sin k_c y + B\cos k_c y. \qquad 3.63$$

The boundary conditions are that $E_x = 0$ at y = 0, d; E_z is identically zero for TE modes. From (3.19c), we have

$$E_x = \frac{-j\omega\mu}{k_c} \left[A\cos k_c y - B\sin k_c y\right] e^{-j\beta z}, \qquad 3.64$$

and applying the boundary conditions shows that A = 0 and

$$k_{\rm c} = \frac{n\pi}{d}, \qquad n = 1, 2, 3...,$$
 3.65

as for the TM case. The final solution for H_z is then

$$H_{z}(x, y) = B_{n} \cos \frac{n\pi y}{d} e^{-j\beta z}.$$
 3.66

The transverse fields can be computed from (3.19) as

$$E_x = \frac{j\omega\mu}{k_c} B_n \sin\frac{n\pi y}{d} e^{-j\beta x}, \qquad 3.67a$$

$$H_y = \frac{j\beta}{k_c} B_n \sin \frac{n\pi y}{d} e^{-j\beta z}, \qquad 3.67b$$

$$E_y = H_x = 0. \tag{3.67c}$$

The propagation constant of the TE_n mode is thus,

$$\beta = \sqrt{k^2 - \left(\frac{n\pi}{d}\right)^2},$$
3.68

which is the same as the propagation constant of the TM_n mode. The cutoff frequency of the TE_n mode is

$$f_c = \frac{n}{2d\sqrt{\mu\epsilon}},$$
 3.69

which is also identical to that of the TM_n mode. The wave impedance of the TE_n mode is, from (3.22),

$$Z_{\rm TE} = \frac{E_x}{H_y} = \frac{\omega\mu}{\beta} = \frac{k\eta}{\beta}, \qquad 3.70$$

which is seen to be real for propagating modes and imaginary for nonpropagating, or cutoff, modes. The phase velocity, guide wavelength, and cutoff wavelength are similar to the results for the TM modes.

The power flow down the guide for a TE_n mode can be calculated as

$$P_{\sigma} = \frac{1}{2} \operatorname{Re} \int_{x=0}^{w} \int_{y=0}^{d} \bar{E} \times \bar{H}^{*} \cdot \hat{z} \, dy \, dx = \frac{1}{2} \operatorname{Re} \int_{x=0}^{w} \int_{y=0}^{d} E_{x} H_{y}^{*} \, dy \, dx$$

$$= \frac{\omega \mu dw}{4k_{c}^{2}} |B_{n}|^{2} \operatorname{Re}(\beta), \quad \text{for } n > 0,$$

3.71

which is zero if the operating frequency is below the cutoff frequency (β imaginary).

Note that if n = 0, then $E_x = H_y = 0$ from (3.67), and thus $P_o = 0$, implying that there is no TE_o mode.

Attenuation can be calculated in the same way as for the TM modes. The attenuation due to dielectric loss is given by (3.29). It is left as a problem to show that the attenuation due to conductor loss for the TE modes is given by

$$lpha_c = rac{2k_c^2 R_s}{\omega\mueta d} = rac{2k_c^2 R_s}{keta \eta d}$$
 Np/m. 3.72

Figure 3.4 shows the attenuation due to conductor loss for the TEM, TM₁, and TE₁ modes. Observe that $\alpha_c \rightarrow \infty$ as cutoff is approached for the TM and TE modes.



FIGURE 3.4 Attenuation due to conductor loss for the TEM, TM_1 , and TE_1 modes of a parallel plate waveguide.

Table 3.1 summarizes a number of useful results for TEM, TM, and TE mode propagation on parallel plate waveguides. Field lines for the TEM, TM_1 , and TE_1 modes are shown in Figure 3.5.

Quantity	TEM Mode	TM ₁ Mode	TE_n Mode
k	$\omega \sqrt{\mu \epsilon}$	ω_ <u>με</u>	$\omega\sqrt{\mu\epsilon}$
k _c	0	$n\pi/d$	$n\pi/d$
β	$k=\omega\sqrt{\mu\epsilon}$	$\sqrt{k^2-k_{e}^2}$	$\sqrt{k^2-k_c^2}$
λ_{e}	20	$2\pi/k_c=2d/n$	$2\pi/k_{ m c}=2d/n$
λ_g	$2\pi/k$	$2\pi/\beta$	2 π/β
v_p	$\omega/k=1/\sqrt{\mu\epsilon}$	w/B	ω / B
ad	$(k an \delta)/2$	$(k^2 an \delta)/2B$	$(k^2 \tan \delta)/2\beta$
α_{a}	$R_s/\eta d$	$2kR_s/\beta\eta d$	$2k_s^2 R_s/k\beta\eta d$
E_z	0	$A_n \sin{(n\pi y/d)}e^{-jeta z}$	0
H _z	0	Ö	$B_n \cos{(n\pi y/d)}e^{-j\beta z}$
E_x	Q	0	$(j\omega\mu/k_{\rm m})B_{\rm m}\sin{(n\pi y/d)}e^{-j\dot{\mu}}$
E_y	$(-V_o/d)e^{-j\beta z}$	$(-j_l \beta/k_c) A_n \cos{(n\pi y/d)} e^{-j\Im s}$	Ø
H_{x}	$(V_{lpha}/\eta d)e^{-jeta_{ar{x}}}$	$(j\omega\epsilon/k_c)A_n\cos{(n\pi y/d)}e^{-jeta_2}$	Ť.
H_y	0	0	$(jeta/k_r)B_n\sin{(n\pi y/d)}e^{-jeta_2}$
Z	$Z_{{ m TEM}}=\eta d/w$	$Z_{ m TM}=eta\eta/k$	$Z_{ ext{TE}} = k \eta / eta$

TABLE 3.1 Summary of Results for Parallel Plate Waveguide



FIGURE 3.5 Field lines for the (a) TEM, (b) TM_1 , and (c) TE_1 modes of a parallel plate waveguide. There is no variation across the width of the waveguide.

3.3 RECTANGULAR WAVEGUIDE

Rectangular waveguides were one of the earliest types of transmission lines used to transport microwave signals and are still used today for many applications. A large variety of components such as couplers, detectors, isolators, attenuators, and slotted lines are commercially available for various standard waveguide bands from 1 GHz to over 220 GHz. Figure 3.6 shows some of the standard rectangular waveguide components that are available. Because of the recent trend toward miniaturization and integration, a lot of microwave circuitry is currently fabricated using planar transmission lines, such as microstrip and stripline, rather than waveguide. There is, however, still a need for waveguides in many applications such as high-power systems, millimeter wave systems, and in some precision test applications.

The hollow rectangular waveguide can propagate TM and TE modes, but not TEM waves, since only one conductor is present. We will see that the TM and TE modes of a rectangular waveguide have cutoff frequencies below which propagation is not possible, similar to the TM and TE modes of the parallel plate guide.

TE Modes

The geometry of a rectangular waveguide is shown in Figure 3.7, where it is assumed that the guide is filled with a material of permittivity ϵ and permeability μ . It is standard convention to have the longest side of the waveguide along the x-axis, so that a > b.

The TE modes are characterized by fields with $E_z = 0$, while H_z must satisfy the reduced wave equation of (3.21):

$$\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + k_c^2\right) h_z(x, y) = 0.$$
 3.73





Courtesy of Hewlett-Packard Company, Santa Rosa, Calif.

with $H_z(x, y, z) = h_z(x, y)e^{-jt^3 z}$, and $k_r^2 = k^2 - \beta^2$ is the cutoff wavenumber. The partial differential equation of (3.73) can be solved by the method of separation of variables by letting

$$h_z(x, y) = X(x)Y(y),$$
 3.74

and substituting into (3.73) to obtain

$$\frac{1}{X}\frac{d^2X}{dx^2} + \frac{1}{Y}\frac{d^2Y}{dy^2} - k_c^2 = 0.$$
3.75

Then, by the usual separation of variables argument, each of the terms in (3.75) must be equal to a constant, so we define separation constants k_x and k_y , such that

$$\frac{d^2 X}{dx^2} + k_x^2 X = 0. 3.76a$$

$$\frac{d^2Y}{dy^2} + k_y^2 Y = 0, 3.76b$$





anđ

$$k_x^2 + k_y^2 = k_e^2. 3.77$$

The general solution for h_z can then be written as

$$h_{*}(x,y) = (A\cos k_{x}x + B\sin k_{x}x)(C\cos k_{y}y + D\sin k_{y}y). \qquad 3.78$$

To evaluate the constants in (3.78) we must apply the boundary conditions on the electric field components tangential to the waveguide walls. That is,

$$e_x(x,y) = 0,$$
 at $y = 0, b,$ 3.79a

$$e_y(x,y) = 0,$$
 at $x = 0, a.$ 3.79b

We thus cannot use h_z of (3.78) directly, but must first use (3.19c) and (3.19d) to find e_x and e_y from h_z :

$$e_x = \frac{-j\omega\mu}{k_c^2} k_y (A\cos k_x x + B\sin k_x x) (-C\sin k_y y + D\cos k_y y), \qquad 3.80a$$

$$e_y = \frac{-j\omega\mu}{k_c^2} k_x (-A\sin k_x x + B\cos k_x x) (C\cos k_y y + D\sin k_y y). \qquad 3.80b$$

Then from (3.79a) and (3.80a), we see that D = 0, and $k_y = n\pi/b$ for n = 0, 1, 2,...From (3.79b) and (3.80b) we have that B = 0 and $k_x = m\pi/a$ for m = 0, 1, 2,... The final solution for H_z is then

$$H_z(x, y, z) = A_{mn} \cos \frac{m\pi x}{a} \cos \frac{n\pi y}{b} e^{-j\beta z}, \qquad 3.81$$

where A_{mn} is an arbitrary amplitude constant composed of the remaining constants A and C of (3.78).

The transverse field components of the TE_{mn} mode can then be found using (3.19) and (3.81):

$$E_x = \frac{j\omega\mu n\pi}{k_c^2 b} A_{mn} \cos\frac{m\pi x}{a} \sin\frac{n\pi y}{b} e^{-j\beta x}, \qquad 3.82a$$

$$E_{y} = \frac{-j\omega\mu m\pi}{k_{c}^{2}a} A_{mn} \sin \frac{m\pi x}{a} \cos \frac{n\pi y}{b} e^{-j\beta z}, \qquad 3.82b$$

3.3 Rectangular Waveguide

$$H_x = \frac{j\beta m\pi}{k_c^2 a} A_{mn} \sin \frac{m\pi x}{a} \cos \frac{n\pi y}{b} e^{-j\beta x}, \qquad 3.82c$$

$$H_y = \frac{j\beta n\pi}{k_c^2 b} A_{mn} \cos \frac{m\pi x}{a} \sin \frac{n\pi y}{b} e^{-j\beta \varphi}.$$
 3.82d

The propagation constant is

$$\beta = \sqrt{k^2 - k_c^2} = \sqrt{k^2 - \left(\frac{m\pi}{a}\right)^2 - \left(\frac{n\pi}{b}\right)^2},$$
 3.83

which is seen to be real, corresponding to a propagating mode, when

$$k > k_c = \sqrt{\left(\frac{m\pi}{a}\right)^2 + \left(\frac{n\pi}{b}\right)^2}.$$

Each mode (combination of m and n) thus has a cutoff frequency $f_{c_{max}}$ given by

$$f_{e_{mn}} = \frac{k_c}{2\pi\sqrt{\mu\epsilon}} = \frac{1}{2\pi\sqrt{\mu\epsilon}}\sqrt{\left(\frac{m\pi}{a}\right)^2 + \left(\frac{n\pi}{b}\right)^2}.$$
 3.84

The mode with the lowest cutoff frequency is called the dominant mode; since we have assumed a > b, the lowest f_c occurs for the TE₁₀ (m = 1, n = 0) mode;

$$f_{e_{10}} = \frac{1}{2a\sqrt{\mu\epsilon}}.$$
 3.85

Thus the TE_{10} mode is the dominant TE mode and, as we will see, the overall dominant mode of the rectangular waveguide. Observe that the field expressions for \bar{E} and \bar{H} in (3.82) are all zero if both m = n = 0; thus there is no TE₀₀ mode.

At a given operating frequency f, only those modes having $f_c < f$ will propagate; modes with $f_c > f$ will lead to an imaginary β (or real α), meaning that all field components will decay exponentially away from the source of excitation. Such modes are referred to as cutoff, or evanescent, modes. If more than one mode is propagating, the waveguide is said to be *overmoded*.

From (3.22) the wave impedance that relates the transverse electric and magnetic fields is

$$Z_{\rm TE} = \frac{E_x}{H_y} = \frac{-E_y}{H_x} \approx \frac{k\eta}{\beta}.$$
 3.86

where $\eta = \sqrt{\mu/\epsilon}$ is the intrinsic impedance of the material filling the waveguide. Note that Z_{TE} is real when β is real (a propagating mode), but is imaginary when β is imaginary (an evanescent mode).

The guide wavelength is defined as the distance between two equal phase planes along the waveguide, and is equal to

$$\lambda_g = \frac{2\pi}{\beta} > \frac{2\pi}{k} = \lambda, \qquad 3.87$$

which is thus greater than λ , the wavelength of a plane wave in the filling medium. The phase velocity is

$$v_p = \frac{\omega}{\beta} > \frac{\omega}{k} = 1/\sqrt{\mu\epsilon}.$$
 3.88

which is greater than $1/\sqrt{\mu\epsilon}$, the speed of light (plane wave) in the filling material.

In the vast majority of applications the operating frequency and guide dimensions are chosen so that only the dominant TE_{10} mode will propagate. Because of the practical importance of the TE_{10} mode, then, we will list the field components and derive the attenuation due to conductor loss for this case.

Specializing (3.81) and (3.82) to the m = 1, n = 0 case gives the following results for the TE₁₀ mode fields:

$$H_z = A_{10} \cos \frac{\pi x}{a} e^{-j\beta z}, \qquad 3.89a$$

$$E_y = \frac{-j\omega\mu a}{\pi} A_{10} \sin\frac{\pi x}{a} e^{-j\beta x}, \qquad 3.89b$$

$$H_x = \frac{j\beta a}{\pi} A_{10} \sin \frac{\pi x}{a} e^{-j\beta z}, \qquad 3.89c$$

$$E_x = E_z = H_y = 0. \tag{3.89d}$$

In addition, for the TE_{10} mode,

$$k_c = \pi/a, \qquad \qquad 3.90$$

3.91

$$\beta = \sqrt{k^2 - (\pi/a)^2}.$$

and

The power flow down the guide for the TE_{10} mode is calculated as

$$P_{10} = \frac{1}{2} \operatorname{Re} \int_{x=0}^{a} \int_{y=0}^{b} \bar{E} \times \bar{H}^{*} \cdot \hat{z} \, dy \, dx$$

$$= \frac{1}{2} \operatorname{Re} \int_{x=0}^{a} \int_{y=0}^{b} E_{y} H_{x}^{*} \, dy \, dx$$

$$= \frac{\omega \mu a^{2}}{2\pi^{2}} \operatorname{Re}(\beta) |A_{10}|^{2} \int_{x=0}^{a} \int_{y=0}^{b} \sin^{2} \frac{\pi x}{a} \, dy \, dx$$

$$= \frac{\omega \mu a^{3} |A_{10}|^{2} b}{4\pi^{2}} \operatorname{Re}(\beta).$$

3.92

Note that this result gives nonzero real power only when β is real, corresponding to a propagating mode.

Attenuation in a rectangular waveguide can occur because of dielectric loss or conductor loss. Dielectric loss can be treated by making ϵ complex and using a Taylor series approximation, with the general result given in (3.29).

Conductor loss is best treated using the perturbation method. The power lost per unit length due to finite wall conductivity is, from (1.131),

$$P_{\ell} = \frac{R_s}{2} \int_{\mathcal{O}} |\bar{J}_s|^2 d\ell.$$

$$3.93$$

where R_s is the wall surface resistance, and the integration contour C encloses the perimeter of the guide walls. There are surface currents on all four walls, but from symmetry the currents on the top and bottom walls are identical, as are the currents on the left and right side walls. So we can compute the power lost in the walls at x = 0and y = 0 and double their sum to obtain the total power loss. The surface current on the x = 0 (left) wall is

$$\tilde{J}_{s} = h \times \bar{H}|_{x=0} = \hat{x} \times \hat{z} H_{z}|_{z=0} = -\hat{y} H_{z}|_{x=0} = -\hat{y} A_{10} e^{-j\hat{z}_{z}}, \qquad 3.94a$$

while the surface current on the y = 0 (bottom) wall is

$$\begin{split} \bar{J}_s &= \hat{n} \times \bar{H}|_{y=0} = \tilde{y} \times (\tilde{x} H_x|_{y=0} + \hat{z} H_z|_{y=0}) \\ &= -\hat{z} \frac{j\beta u}{\pi} A_{10} \sin \frac{\pi x}{a} e^{-j\beta z} + \hat{x} \dot{A}_{10} \cos \frac{\pi x}{a} e^{-j\beta z}, \end{split}$$

$$3.94b$$

Substituting (3.94) into (3.93) gives

$$P_{\ell} = R_{s} \int_{y=0}^{b} |J_{sy}|^{2} dy + R_{s} \int_{\pi=0}^{a} \left[|J_{\delta x}|^{2} + |J_{s2}|^{2} \right] dx$$

$$= R_{s} |A_{10}|^{2} \left(b + \frac{a}{2} + \frac{\beta^{2} a^{3}}{2\pi^{2}} \right).$$

$$3.95$$

The attenuation due to conductor loss for the TE_{10} mode is then

$$\alpha_{c} = \frac{P_{l}}{2P_{10}} = \frac{2\pi^{2}R_{s}[b + a/2 + (\theta^{2}a^{3})/2\pi^{2}]}{\omega\mu a^{3}b\beta}$$
$$= \frac{R_{s}}{a^{3}b\beta k\eta} (2b\pi^{2} + a^{3}k^{2}) \operatorname{Np}/m.$$

_ _

TM Modes

The TM modes are characterized by fields with $H_z = 0$, while E_z must satisfy the reduced wave equation of (3.25):

$$\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + k_c^2\right) e_z(x, y) = 0, \qquad \qquad 3.97$$

with $E_z(x, y, z) = e_z(x, y)e^{-\beta\beta z}$ and $k_c^2 = k^2 - \beta^2$. Equation (3.97) can be solved by the separation of variables procedure that was used for the TE modes. The general solution is then

$$e_{z}(x, y) = (A \cos k_{x} x + B \sin k_{x} x)(C \cos k_{y} y + D \sin k_{y} y).$$
 3.98

The boundary conditions can be applied directly to e_x :

$$e_{z}(x, y) = 0,$$
 at $x = 0, a,$ 3.99a

$$e_x(x, y) = 0.$$
 at $y = 0, b.$ 3.99b

We will see that satisfaction of the above conditions on e_z will lead to satisfaction of the boundary conditions by e_x and e_y .

Applying (3.99a) to (3.98) shows that A = 0 and $k_x = m\pi/a$, for m = 1, 2, 3,...Similarly, applying (3.99b) to (3.98) shows that C = 0 and $k_y = n\pi/b$, for n = 1, 2, 3,...The solution for E_z then reduces to

$$E_z(x, y, z) = B_{mn} \sin \frac{m\pi x}{a} \sin \frac{n\pi y}{b} e^{-j\beta z}, \qquad 3.100$$

where B_{mn} is an arbitrary amplitude constant.

The transverse field components for the TM_{mn} mode can be computed from (3.23) and (3.100) as

$$E_x = \frac{-j\beta m\pi}{ak_c^2} B_{mn} \cos \frac{m\pi x}{a} \sin \frac{n\pi y}{b} e^{-j\beta z}, \qquad 3.101a$$

$$E_y = \frac{-j\beta n\pi}{bk_a^2} B_{mn} \sin \frac{m\pi x}{a} \cos \frac{n\pi y}{b} e^{-j\beta z}, \qquad 3.101b$$

$$H_x = \frac{j\omega\epsilon n\pi}{bk_c^2} B_{mn} \sin\frac{m\pi x}{a} \cos\frac{n\pi y}{b} e^{-j\beta z}, \qquad 3.101c$$

$$H_y = \frac{-j\omega\epsilon m\pi}{ak_c^2} B_{mn} \cos\frac{m\pi x}{a} \sin\frac{n\pi y}{b} e^{-j\beta z}.$$
 3.101d

As for the TE modes, the propagation constant is

$$\beta = \sqrt{k^2 - k_c^2} = \sqrt{k^2 - \left(\frac{m\pi}{a}\right)^2 - \left(\frac{n\pi}{b}\right)^2}.$$
 3.102

and is real for propagating modes, and imaginary for evanescent modes. The cutoff frequency for the TM_{mn} modes is also the same as that of the TE_{mn} modes, as given in (3.84). The guide wavelength and phase velocity for TM modes are also the same as those for TE modes.

Observe that the field expressions for \tilde{E} and \tilde{H} in (3.101) are identically zero if either *m* or *n* is zero. Thus there are no TM₀₀, TM₀₁, or TM₁₀ modes, and the lowest order TM mode to propagate (lowest f_c) is the TM₁₁ mode, having a cutoff frequency of

$$f_{c_{11}} = \frac{1}{2\pi\sqrt{\mu\epsilon}}\sqrt{\left(\frac{\pi}{a}\right)^2 + \left(\frac{\pi}{b}\right)^2}.$$
 3.103

which is seen to be larger than $f_{c_{10}}$ for the cutoff frequency of the TE₁₀ mode.

The wave impedance relating the transverse electric and magnetic fields is, from (3.26),

$$Z_{\mathsf{TM}} = \frac{E_x}{H_y} = \frac{-E_y}{H_x} = \frac{\beta\eta}{k}.$$
 3.104





Attenuation due to dielectric loss is computed in the same way as for the TE modes, with the same result. The calculation of attenuation due to conductor loss is left as a problem: Figure 3.8 shows the attenuation versus frequency for some TE and TM modes in a rectangular waveguide. Table 3.2 summarizes results for TE and TM wave propagation in rectangular waveguides, and Figure 3.9 shows the field lines for several of the lowest order TE and TM modes.

EXAMPLE 3.1 Characterístics of a Rectangular Waveguide

Consider a length of air-filled copper X-band waveguide, with dimensions a = 2.286 cm, b = 1.016 cm. Find the cutoff frequencies of the first four propagating modes. What is the attenuation in dB of a 1 m length of this guide when operating at f = 10 GHz?

Solution

From (3.84), the cutoff frequencies are given by

$$f_{c_{m\pi}} = \frac{c}{2\pi} \sqrt{\left(\frac{m\pi}{a}\right)^2 + \left(\frac{n\pi}{b}\right)^2}.$$

Computing f_c for the first few values of m and n gives:

Mode	m	n _	$fe_{mn}(\text{GHz})$
TE	i	0	6.562
ŤΒ	2	0	13.123
TE	0	I	14.7 6 4
TE, TM	l	1	16,156
TE, TM	I	2	30,248
TE, TM	2	1	19.753

Quantity	TEmn Mode	TM _{nin} Mode
Å	ω	w VIIE
ka	$\sqrt{(m\pi/a)^2 + (n\pi/b)^2}$	$\sqrt{(m\pi/a)^2 + (n\pi/b)^2}$
ß	$\sqrt{k^2-k_c^2}$	$\sqrt{k^2-k_a^2}$
λ_{e}	$\frac{2\pi}{k_e}$	$\frac{2\pi}{k_c}$
λ_g	2 41 <u>B</u>	$\frac{2\pi}{\beta}$
v_p	$\frac{\omega}{\beta}$	$\frac{\omega}{\beta}$
(Jag	$\frac{k^2 \tan \delta}{2\beta}$	$\frac{k^2 \tan \delta}{2\beta}$
E_{z}	0	$B_{mn}\sin{\frac{m\pi x}{a}}\sin{\frac{n\pi y}{b}}e^{-j\beta x}$
H _z	$A_{mn}\cosrac{m\pi x}{a}\cosrac{n\pi y}{\delta}e^{-\jmatheta_z}$	а 0 0
E_{x}	$rac{j\omega\mu n\pi}{k_c^2 b}A_{mn}\cosrac{m\pi x}{a}\sinrac{n\pi y}{b}e^{-jeta z}$	$\frac{-j\beta m\pi}{k_c^2 a} B_{mn} \cos \frac{m\pi x}{a} \sin \frac{n\pi y}{b} e^{-j\beta z}$
E_y	$\frac{-j\omega\mu m\pi}{k_c^2a}A_{mn}\sin\frac{m\pi x}{a}\cos\frac{n\pi y}{b}e^{-j\beta x}$	$rac{-jeta n\pi}{k_c^2 b}B_{mn}\sinrac{m\pi x}{a}\cosrac{n\pi y}{b}\mathrm{e}^{-jeta z}$
H_{x}	$\frac{j\beta m\pi}{k_c^2 a} A_{mn} \sin \frac{m\pi x}{a} \cos \frac{n\pi y}{b} e^{-j\beta z}$	$rac{j\omega\epsilon n\pi}{k_c^2 b}B_{mn}\sinrac{m\pi x}{a}\cosrac{n\pi y}{b}\mathrm{e}^{-jeta z}$
Ħ _{y.}	$\frac{j\beta n\pi}{k_c^2 b} A_{mn} \cos \frac{m\pi x}{a} \sin \frac{n\pi y}{b} e^{-j\beta x}$	$-\frac{-j\omega\epsilon m\pi}{k_c^2 a}B_{mn}\cos\frac{m\pi x}{a}\sin\frac{n\pi y}{b}e^{-j\beta z}$
2	$Z_{ ext{TE}} = rac{k\eta}{eta}$	$oldsymbol{Z_{TM}}=rac{eta\eta}{k}$

TABLE 3.2 Summary of Results for Rectangular Waveguide

Thus the TE_{10} , TE_{20} , TE_{01} , and TE_{11} modes will be the first four modes to propagate (the TE_{11} and TM_{11} modes have the same cutoff frequency).

At 10 GHz, $k = 209.44 \text{ m}^{-1}$, and the propagation constant of the TE₁₀ mode (the only propagating mode) is

$$\beta = \sqrt{k^2 - \left(\frac{\pi}{a}\right)^2} = \sqrt{\left(\frac{2\pi f}{c}\right)^2 - \left(\frac{\pi}{a}\right)^2} = 158.05 \text{ m}^{-1}.$$

The surface resistivity of the copper walls is ($\sigma = 5.8 \times 10^7 \text{ S/m}$)

$$R_s = \sqrt{rac{\omega\mu}{2\sigma}} = 0.026 \ \Omega,$$

so the attenuation constant, from (3.96), is

$$\alpha_c = \frac{R_s}{a^3 b \beta k \eta} (2b\pi^2 + a^3 k^2) = 0.0125 \text{ Np/m},$$

$$\alpha_c (dB) = -20 \log e^{-\alpha_c} = 0.11 \text{ dB/m},$$
nery, and T. Van Duzer. Copyright (2) 1965 by John Wiley & Sons, Inc. Table 8.02.

Reprinted with pertifyion from Fields and Waves in Communication Electronics, S. Ramo, J.R. Whin-





FIGURE 3.9

TE_{mo} Modes of a Partially Loaded Waveguide

The above results also apply for a rectangular waveguide filled with a homogeneous dielectric or magnetic material, but in many cases of practical interest (such as impedance matching or phase-shifting sections) a waveguide is used with only a partial filling. Then an additional set of boundary conditions are introduced at the material interface, necessitating a new analysis. To illustrate the technique we will consider the TE_{mo} modes of a rectangular waveguide that is partially loaded with a dielectric slab, as shown in Figure 3.10. The analysis still follows the basic procedure outlined at the end of Section 3.1.

Since the geometry is uniform in the y direction and n = 0, the TE_{mo} modes have no y dependence. Then the wave equation of (3.21) for h_z can be written separately for the dielectric and air regions as

$$\left(rac{\partial^2}{\partial x^2}+k_d^2
ight)h_z=0, \qquad ext{for } 0\leq x\leq t, \qquad 3.105a$$

$$\left(\frac{\partial^2}{\partial x^2} + k_a^2\right)h_z = 0, \quad \text{for } t \le x \le a, \quad 3.105b$$

where k_d and k_a are the cutoff wavenumbers for the dielectric and air regions, defined as follows:

$$\beta = \sqrt{\epsilon_r k_0^2 - k_d^2}, \qquad 3.106a$$

$$\boldsymbol{\beta} = \sqrt{k_0^2 - k_a^2}.$$
 3.106b

These relations incorporate the fact that the propagation constant, β , must be the same in both regions to ensure phase matching of the fields along the interface at x = t. The solutions to (3.105) can be written as

$$h_{z} = \begin{cases} A\cos k_{d}x + B\sin k_{d}x & \text{for } 0 \le x \le t \\ C\cos k_{a}(a-x) + D\sin k_{a}(a-x) & \text{for } t \le x \le a, \end{cases}$$
3.107

where the form of the solution for t < x < a was chosen to simplify the evaluation of boundary conditions at x = a.



FIGURE 3.10 Geometry of a partially loaded rectangular waveguide.

Now we need \hat{y} and \hat{z} field components to apply the boundary conditions at x = 0, t, and a. $E_z = 0$ for TE modes, and $H_y = 0$ since $\partial/\partial y = 0$. E_y is found from (3.19d) as

$$\boldsymbol{e}_{y} = \begin{cases} \frac{j\omega\mu_{0}}{k_{d}} [-A\sin k_{d}x + B\cos k_{d}x] & \text{for } 0 \le x \le t \\ \frac{j\omega\mu_{0}}{k_{a}} [C\sin k_{a}(a-x) - D\cos k_{a}(a-x)] & \text{for } t \le x \le a. \end{cases}$$
3.108

To satisfy the boundary conditions that $E_y = 0$ at x = 0 and x = a requires that B = D = 0. Next, we must enforce continuity of tangential fields (E_y, H_z) at x = t. Equations (3.107) and (3.108) then give the following:

$$\frac{-A}{k_d}\sin k_d t = \frac{C}{k_a}\sin k_a(a-t),$$
$$A\cos k_d t = C\cos k_a(a-t).$$

Since this is a homogeneous set of equations, the determinant must vanish in order to have a nontrivial solution. Thus,

$$k_a \tan k_d t + k_d \tan k_d (a - t) = 0.$$
 3.109

Using (3.106) allows k_a and k_d to be expressed in terms of \mathcal{J} , sp (3.109) can be solved numerically for \mathcal{A} . There is an infinite number of solutions to (3.109), corresponding to the propagation constants of the $T \to m_a$ modes.

This technique can be applied to many other waveguide geométries involving dielectric or magnetic inhomogeneities, such as the surface waveguide of Section 3.6 or the ferrite-loaded waveguide of Section 9.3. In some cases, however, it will be impossible to satisfy all the necessary boundary conditions with only TE- or TM-type modes, and a hybrid combination of both types of modes will be required.

POINT OF INTEREST: Waveguide Flanges

There are two commonly used waveguide flanges: the cover flange and the choke flange. As shown in the figure, two waveguides with cover-type flanges can be bolted together to form a contacting joint. To avoid reflections and resistive loss at this joint, it is necessary that the contacting surfaces be smooth, clean, and square, because RF currents must flow across this discontinuity. In high-power applications voltage breakdown may occur at this joint. Otherwise, the simplicity of the cover-to-cover connection makes it preferable for general use. The SWR from such a joint is typically less than 1.03.

An alternative waveguide connection uses a cover flange against a choke flange, as shown in the figure. The choke flange is machined to form an effective radial transmission line in the narrow gap between the two flanges; this line is approximately $\lambda_q/4$ in length between the guide and the point of contact for the two flanges. Another $\lambda_q/4$ line is formed by a circular axial groove in the choke flange. So the short circuit at the right-hand end of this grouve is transformed to an open circuit at the contact point of the flanges. Any resistance in this contact is in series with an inlinite (or very high) impedance and thus has little effect. Then this high impedance is transformed back to a short circuit (nr very low impedance) at the edges of the waveguides, to provide an effective low-resistance path for current flow across the joint. Since there is a negligible voltage drop across the ohmic contact between the flanges, voltage breakdown is avoided. Thus, the cover-to-choke connection can be useful for high-power applications. The SWR for this joint is typically less than 1.05, but is more frequency dependent than the cover-to-cover joint.



Reference: C. G. Monigomery, R. H. Dicke, and E. M. Purcell. Principles of Microwave Circuits, McGraw-Hill, New York, 1948.

3.4 ____ CIRCULAR WAVEGUIDE

A hollow metal tube of circular cross section also supports TE and TM waveguide modes. Figure 3.11 shows the cross-section geometry of such a circular waveguide of inner radius a. Since a cylindrical geometry is involved, it is appropriate to employ cylindrical coordinates. As in the rectangular coordinate case, the transverse fields in cylindrical coordinates can be derived from E_z or H_z field components, for TM and



TE modes, respectively. Paralleling the development of Section 3.1, the cylindrical components of the transverse fields can be derived from the longitudinal components as

$$E_{\mu} = \frac{-j}{k_{c}^{2}} \left(\beta \frac{\partial E_{z}}{\partial \rho} + \frac{\omega \mu}{\rho} \frac{\partial H_{z}}{\partial \phi} \right).$$
 3.110*a*

$$E_{\Phi} = \frac{-j}{k_c^2} \left(\frac{\beta}{\rho} \frac{\partial E_z}{\partial \phi} - \omega \mu \frac{\partial H_z}{\partial \rho} \right), \qquad 3.110b$$

$$H_{\rho} = \frac{j}{k_{c}^{2}} \left(\frac{\omega \epsilon}{\rho} \frac{\partial E_{z}}{\partial \phi} - \beta \frac{\partial H_{\varepsilon}}{\partial \rho} \right).$$
 3.110c

$$H_{\phi} = \frac{-j}{k_{\epsilon}^2} \left(\omega \epsilon \frac{\partial E_{\epsilon}}{\partial \rho} + \frac{\beta}{\rho} \frac{\partial H_{\epsilon}}{\partial \phi} \right).$$
 3.110d

where $k_c^2 = k^2 - \beta^2$, and $e^{-j\beta z}$ propagation has been assumed. For $e^{+j\beta z}$ propagation, replace β with $-\beta$ in all expressions,

TE Modes

For TE modes, $E_z = 0$, and H_z is a solution to the wave equation,

$$\nabla^2 H_z + k^2 H_z = 0.$$
3.111

If $H_z(\rho, \phi, z) = h_z(\rho, \phi) e^{-j\beta z}$, (3.111) can be expressed in cylindrical coordinates as

$$\left(\frac{\partial^2}{\partial\rho^2} + \frac{1}{\rho}\frac{\partial}{\partial\rho} + \frac{1}{\rho^2}\frac{\partial^2}{\partial\phi^2} + k_c^2\right)h_z(\rho,\phi) = 0.$$
 3.112

Again, a solution can be derived using the method of separation of variables. Thus, we let

$$h_z(\rho,\phi) = R(\rho)P(\phi), \qquad 3.113$$

and substitute into (3.112) to obtain

$$\frac{1}{R}\frac{d^2R}{d\rho^2} + \frac{1}{\rho R}\frac{dR}{d\rho} + \frac{1}{\rho^2 P}\frac{d^2P}{d\phi^2} + k_e^2 = 0.$$

$$\frac{\rho^2}{R}\frac{d^2R}{d\rho^2} + \frac{\rho}{R}\frac{dR}{d\rho} + \rho^2 k_e^2 = \frac{-1}{P}\frac{d^2P}{d\phi^2}.$$
3.114

or

The left side of this equation depends on p (not ϕ), while the right side depends only on ϕ . Thus, each side must be equal to a constant, which we will call k_{ω}^2 . Then,

$$\frac{-1}{P}\frac{d^2P}{d\phi^2} = k_{\phi}^2,$$
$$\frac{d^2P}{d\phi^2} + k_{\phi}^2P = 0.$$
 3.115

or

Also,

$$\rho^2 \frac{d^2 R}{d\rho^2} + \rho \frac{dR}{d\rho} + \left(\rho^2 k_a^2 - k_{\phi}^2\right) R = 0.$$
 3.116

The general solution to (3.115) is

$$P(\phi) = A \sin k_{\phi} \phi + B \cos k_{\phi} \phi. \qquad 3.117$$

Since the solution to h_z must be periodic in ϕ (that is, $h_z(\rho, \phi) = h_z(\rho, \phi \pm 2m\pi)$), k_{ϕ} must be an integer, n. Thus (3.117) becomes

$$P(\phi) = A\sin n\phi + B\cos n\phi, \qquad 3.118$$

while (3.116) becomes

$$\rho^2 \frac{d^2 R}{d\rho^2} + \rho \frac{dR}{d\rho} + \left(\rho^2 k_c^2 - n^2\right) R = 0, \qquad 3.119$$

which is recognized as Bessel's differential equation. The solution is

$$R(\rho) = CJ_n(k_c\rho) + DY_n(k_c\rho), \qquad 3.120$$

where $J_n(x)$ and $Y_n(x)$ are the Bessel functions of first and second kinds, respectively. Since $Y_n(k_c\rho)$ becomes infinite at $\rho = 0$, this term is physically unacceptable for the circular waveguide problem, so that D = 0. The solution for h_z can then be written as

$$h_z(\rho,\phi) = (A\sin n\phi + B\cos n\phi)J_n(k_c\rho), \qquad 3.121$$

where the constant C of (3.120) has been absorbed into the constants A and B of (3.121). We must still determine the cutoff wavenumber k_c , which we can do by enforcing the boundary condition that $E_{tan} = 0$ on the waveguide wall. Since $E_z = 0$, we must have that

$$E_{\phi}(\rho,\phi) = 0,$$
 at $\rho = a.$ 3.122

From (3.110b), we find E_{ϕ} from H_z as

$$E_{\phi}(\rho,\phi,z) = \frac{j\omega\mu}{k_c} (A\sin n\phi + B\cos n\phi) J'_n(k_c\rho) e^{-j\beta z}, \qquad 3.123$$

where the notation $J'_n(k_c\rho)$ refers to the derivative of J_n with respect to its argument. For E_{ϕ} to vanish at $\rho = a$, we must have

$$J'_n(k_c a) = 0. 3.124$$

If the roots of $J'_n(x)$ are defined as p'_{nm} , so that $J'_n(p'_{nm}) = 0$, where p'_{nm} is the *m*th root of J'_n , then k_c must have the value

$$k_{c_{nm}} = \frac{p'_{nm}}{a}.$$
 3.125

Values of p'_{nm} are given in mathematical tables; the first few values are listed in Table 3.3.

The TE_{nm} modes are thus defined by the cutoff wavenumber, $k_{c_{nm}} = p'_{nm}/a$, where *n* refers to the number of circumferential (ϕ) variations, and *m* refers to the number of radial (ρ) variations. The propagation constant of the TE_{nm} mode is

$$\beta_{nm} = \sqrt{k^2 - k_c^2} = \sqrt{k^2 - \left(\frac{p'_{nm}}{a}\right)^2},$$
 3.126

n	p'_{n1}	p'_{n2}	p'_{n3}
Ū.	3.832	7.016	10.174
I	1.841	5.331	8.536
2	3.054	6.706	9.970

TABLE 3.3 Values of p'am for TE Modes of a Circular Waveguide

with a cutoff frequency of

$$f_{c_{nm}} = \frac{k_c}{2\pi\sqrt{\mu\epsilon}} = \frac{p'_{nm}}{2\pi a\sqrt{\mu\epsilon}}.$$
 3.127

The first TE mode to propagate is the mode with the smallest p'_{nm} , which from Table 3.3 is seen to be the TE₁₁ mode. This mode is then the dominant circular waveguide mode, and the one most frequently used. Because $m \ge 1$, there is no TE₁₀ mode, but there is a TE₀₁ mode.

The transverse field components are, from (3.110) and (3.121),

$$E_{\rho} = \frac{-j\omega\mu n}{k_c^2 \rho} \left(A\cos n\phi - B\sin n\phi\right) J_n\left(k_c\rho\right) e^{-j\beta z}, \qquad 3.128a$$

$$E_{\phi} = \frac{j\omega\mu}{k_c} \left(A\sin n\phi + B\cos n\phi\right) J_n^{\gamma}(k_c\rho) e^{-j\beta k_c}.$$
 3.128b

$$H_{\rho} = \frac{-jn}{k_c} (A\sin n\phi + B\cos n\phi) J'_n(k_c\rho) e^{-j\beta z}, \qquad 3.128c$$

$$H_{\phi} = \frac{-j\beta n}{k_c^2 \rho} (A\cos n\phi - B\sin n\phi) J_n(k_c\rho) e^{-j\beta s}.$$
 3.128d

The wave impedance is

$$Z_{\text{TF}} = \frac{E_{\rho}}{H_{\phi}} = \frac{-E_{\phi}}{H_{\rho}} = \frac{\eta k}{\beta}.$$
 3.129

In the above solutions there are two remaining arbitrary amplitude constants, A and B. These constants control the amplitude of the sin $n\phi$ and $\cos n\phi$ terms, which are independent. That is, because of the azimuthal symmetry of the circular waveguide, both the sin $n\phi$ and $\cos n\phi$ terms are valid solutions, and can be present in a specific problem to any degree. The actual amplitudes of these terms will be dependent on the excitation of the waveguide. From a different viewpoint, the coordinate system can be rotated about the z-axis to obtain an h_z with eather A = 0 or B = 0.

Now consider the dominant TE_{11} mode with an excitation such that B = 0. The fields can be written as

$$H_z = A\sin\phi J_1(k_c \rho)e^{-j\mathscr{G}z},\qquad 3.130a$$

$$E_{\rho} = \frac{-j\omega\mu}{k_c^2\rho} A\cos\phi J_1(k_c\rho)e^{-j\mathcal{A}z}.$$
 3.130b

$$E_{\phi} = \frac{j\omega\mu}{k_c} A \sin\phi J_1'(k_c\rho) e^{-j\beta x}, \qquad 3.130c$$

$$H_{\rho} = \frac{-j\beta}{k_{c}} A \sin \phi J_{1}'(k_{c}\rho) e^{-j\beta z}, \qquad 3.130d$$

$$H_{\varphi} = \frac{-j\beta}{k_c^2 \rho} A \cos \phi J_1(k_c \rho) e^{-j\beta z}, \qquad 3.130e$$

$$E_z = 0. \qquad 3.130f$$

The power flow down the guide can be computed as

$$P_{o} = \frac{1}{2} \operatorname{Re} \int_{\rho=0}^{a} \int_{\phi=0}^{2\pi} \tilde{E} \times \bar{H}^{*} \cdot \hat{z} \rho \, d\phi \, d\rho$$

$$= \frac{1}{2} \operatorname{Re} \int_{\rho=0}^{a} \int_{\phi=0}^{2\pi} \left[E_{\rho} H_{\phi}^{*} - E_{\phi} H_{\rho}^{*} \right] \rho \, d\phi \, d\rho$$

$$= \frac{\omega \mu |A|^{2} \operatorname{Re}(\beta)}{2k_{c}^{4}} \int_{\rho=0}^{a} \int_{\omega=0}^{2\pi} \left[\frac{1}{\rho^{2}} \cos^{2} \phi J_{1}^{2}(k_{c}\rho) + k_{c}^{2} \sin^{2} \phi J_{1}^{'2}(k_{c}\rho) \right] \rho \, d\phi \, d\rho$$

$$= \frac{\pi \omega \mu |A|^{2} \operatorname{Re}(\beta)}{2k_{c}^{4}} \int_{\rho=0}^{a} \left[\frac{1}{\rho} J_{1}^{2}(k_{c}\rho) + \rho k_{c}^{2} J_{1}^{'2}(k_{c}\rho) \right] d\rho$$

$$= \frac{\pi \omega \mu |A|^{2} \operatorname{Re}(\beta)}{4k_{c}^{4}} \left(p_{11}^{'2} - 1 \right) J_{1}^{2}(k_{c}a), \qquad 3.131$$

which is seen to be nonzero only when β is real, corresponding to a propagating mode. (The required integral for this result is given in Appendix C.)

Attenuation due to dielectric loss is given by (3.29). The attenuation due to a lossy waveguide conductor can be found by computing the power loss per unit length of guide:

$$P_{\ell} = \frac{R_s}{2} \int_{\phi=0}^{2\pi} |J_s|^2 a \, d\phi$$

= $\frac{R_s}{2} \int_{\phi=0}^{2\pi} \left[|H_{\phi}|^2 + |H_z|^2 \right] a \, d\phi$
= $\frac{|A|^2 R_s}{2} \int_{\phi=0}^{2\pi} \left[\frac{\beta^2}{k_c^4 a^2} \cos^2 \phi + \sin^2 \phi \right] J_1^2(k_c a) a \, d\phi$
= $\frac{\pi |A|^2 R_s a}{2} \left(1 + \frac{\beta^2}{k_c^4 a^2} \right) J_1^2(k_c a).$ 3.132

The attenuation constant is then

$$\alpha_{c} = \frac{P_{\ell}}{2P_{\rho}} = \frac{R_{s} \left(k_{c}^{4} a^{2} + \beta^{2}\right)}{\eta k \beta a(p_{11}^{\prime 2} - 1)}$$
$$= \frac{R_{s}}{a k \eta \beta} \left(k_{c}^{2} + \frac{k^{2}}{p_{11}^{\prime 2} - 1}\right) Np/m. \qquad 3.133$$

.

TM Modes

For the TM modes of the circular waveguide, we must solve for E_z from the wave equation in cylindrical coordinates:

$$\left(\frac{\partial^2}{\partial\rho^2} + \frac{1}{\rho}\frac{\partial}{\partial\rho} + \frac{1}{\rho^2}\frac{\partial^2}{\partial\phi^2} + k_c^2\right)\epsilon_z = 0, \qquad 3.134$$

where $E_x(\rho, \phi, z) = e_x(\rho, \phi)e^{-j\beta z}$, and $k_c^2 = k^2 - \beta^2$. Since this equation is identical to (3.107), the general solutions are the same. Thus, from (3.121),

$$e_{\mathbf{z}}(\rho,\phi) = (A\sin n\phi + B\cos n\phi)J_n(k_c\rho), \qquad 3.135$$

The difference between the TE solution and the present solution is that the boundary conditions can now be applied directly to e_z of (3.135), since

$$E_z(\rho, \phi) = 0,$$
 at $\rho = a.$ 3.136

Thus, we must have

$$J_n(k_c a) = 0, \qquad \qquad 3.137$$

$$k_c = p_{am}/a, \qquad 3.138$$

where p_{nm} is the *m*th root of $J_n(x)$; that is, $J_n(p_{nm}) = 0$. Values of p_{nm} are given in mathematical tables; the first few values are listed in Table 3.4.

The propagation constant of the TM_{nm} mode is

$$\theta_{nm} = \sqrt{k^2 - k_c^2} = \sqrt{k^2 - (p_{nm}/a)^2}.$$
 3.139

The cutoff frequency is

$$f_{c_{mn}} = \frac{k_c}{2\pi\sqrt{\mu\epsilon}} = \frac{p_{nm}}{2\pi\sigma\sqrt{\mu\epsilon}}.$$
 3.140

Thus, the first TM mode to propagate is the TM_{01} mode, with $p_{01} = 2.405$. Since this is greater than $p'_{11} = 1.841$ of the lowest order TE_{11} mode, the TE_{11} mode is the dominant mode of the circular waveguide. As with the TE modes, $m \ge 1$, so there is no TM_{10} mode.

From (3.110), the transverse fields can be derived as

$$E_{\rho} = \frac{-j\beta}{k_{c}} (A\sin n\phi + B\cos n\phi) J'_{\mu}(k_{c}\rho) e^{-j\beta z}.$$
 3.141q

TABLE 3.4 Values of p_{nm} for TM Modes of a Circular Waveguide

n	p_{ni}	p_{n2}	p_{n3}
0	2.405	5.520	8.654
l	3.832	7.016	10,174
2	5.135	8.417	11.620



FIGURE 3.12 Attenuation of various modes in a circular copper waveguide with a = 2.54 cm.

$$E_{\phi} = \frac{-j\beta n}{k_c^2 \rho} (A\cos n\phi - B\sin n\phi) J_n(k_c \rho) e^{-j\beta z}, \qquad 3.141b$$

$$H_{\rho} = \frac{j\omega\epsilon n}{k_c^2\rho} (A\cos n\phi - B\sin n\phi) J_n(k_c\rho) e^{-j\beta z}, \qquad 3.141c$$

$$H_{\phi} = \frac{-j\omega\epsilon}{k_c} (A\sin n\phi + B\cos n\phi) J'_n(k_c\rho) e^{-j\beta z}.$$
 3.141d

The wave impedance is

$$Z_{\rm TM} = \frac{E_{\rho}}{H_{\phi}} = \frac{-E_{\phi}}{H_{\rho}} = \frac{\eta\beta}{k}.$$
 3.142

Calculation of the attenuation for TM modes is left as a problem. Figure 3.12 shows the attenuation due to conductor loss versus frequency for various modes of a circular waveguide. Observe that the attenuation of the TE_{01} mode decreases to a very small value with increasing frequency. This property makes the TE_{01} mode of interest for low-loss transmission over long distances. Unfortunately, this mode is not the dominant mode of the circular waveguide, so in practice power can be lost from the TE_{01} mode to lower-order propagating modes.

Figure 3.13 shows the relative cutoff frequencies of the TE and TM modes, and Table 3.5 summarizes results for wave propagation in circular waveguide. Field lines for some of the lowest order TE and TM modes are shown in Figure 3.14.

EXAMPLE 3.2 Characteristics of a Circular Waveguide

Find the cutoff frequencies of the first two propagating modes of a circular waveguide with a = 0.5 cm and $\epsilon_r = 2.25$. If the guide is silver plated and



FIGURE 3.13 Cutoff frequencies of the first few TE and TM modes of a circular waveguide, relative to the cutoff frequency of the dominant TE_{11} mode.

the dielectric loss tangent is 0.001, calculate the attenuation in dB for a 50 cm length of guide operating at 13.0 GHz.

Solution

From Figure 3.13, the first two propagating modes of a circular waveguide are the TE_{11} and TM_{01} modes. The cutoff frequencies can be found using

Quantity	$TE_{\eta,m}$ Mode	TM., m Mode
k		- $ -$
k_c	$\frac{p'_{nm}}{a}$	$\frac{p_{nm}}{a}$
β	$\sqrt{k^2-k_e^2}$	$\sqrt{k^2-k_r^2}$
λ_c	$\frac{2\pi}{k_{r}}$	$\frac{2\pi}{k_{\pi}}$
λ_g	$\frac{2\pi}{3}$	$\frac{2\pi}{\beta^3}$
v_p	50 13	レ 山 済
$lpha_d$	$\frac{k^2 \tan \delta}{2 \sqrt{k}}$	$\frac{k^2 \tan \delta}{2k}$
E_{z}	0	$(A\sin n\phi + B\cos n\phi)J_n(k_n\rho)e^{-j\beta z}$
H_z	$(A\sin n\phi + B\cos n\phi)J_n(k_n\rho)e^{-jeta_x}$	0
$E_{ ho}$	$\frac{-j\omega\mu n}{k_c^2\rho}(A\cos n\phi - B\sin n\phi)J_n(k_c\rho)e^{-j\beta z}$	$\frac{-j\beta}{k_c}(A\sin n\phi + B\cos n\phi)J'_n(k_c\rho)e^{-j\vartheta_{\overline{\sigma}}}$
E_{ϕ}	$\frac{j\omega\mu}{k_c}(A\sin n\phi + B\cos n\phi)J'_n(k_c\rho)e^{-j\beta x}$	$\frac{-j;n}{k_{c,\rho}^2}(A\cos n\phi - B\sin n\phi)J_n(k_c\rho)e^{-j\beta m}$
H_{ρ}	$\frac{-j\beta}{k_c}(A\sin n\phi + B\cos n\phi)J'_n(k_c\rho)e^{-j\beta x}$	$\frac{J\omega\epsilon n}{k^2\rho}(A\cos n\phi - B\sin n\phi)J_n(k_{\epsilon}\rho)e^{-j\dot{\theta}\phi}$
H_{c_2}	$\frac{-j\partial n}{k_c^2\rho}(\mathbf{A}\cos n\phi - B\sin n\phi)\mathcal{J}_n(\mathbf{k}_c\rho)e^{-j\partial x}$	$\frac{-j\omega\epsilon}{k_c}(A\sin n\phi + B\cos n\phi)J'_c(k_c\rho)e^{-i\theta x}$
Z	, $Z_{ extsf{TE}} = rac{k\eta}{eta}$	$Z_{\rm TM} = \frac{\beta \eta}{k}$

TABLE 3.5 Summary of Results for Circular Waveguide



FIGURE 3.14 Field lines for some of the lower order modes of a circular waveguide.

Reprinted with permission from Fields and Waves in Communication Electronics, S. Ramo, J.R. Whinnery, and T. Van Duzer. Copyright © 1965 by John Wiley & Sons, Inc. Table 8.04.

(3.127) and (3.140):

$$TE_{11}: f_{c} = \frac{p_{11}'c}{2\pi a\sqrt{\epsilon_{r}}} = \frac{1.841(3 \times 10^{8})}{2\pi(0.005)\sqrt{2.25}} = 11.72 \text{ GHz}.$$

$$TM_{01}: f_{c} = \frac{p_{01}c}{2\pi a\sqrt{\epsilon_{r}}} = \frac{2.405(3 \times 10^{8})}{2\pi(0.005)\sqrt{2.25}} = 15.31 \text{ GHz}.$$

So only the TE_{11} mode is propagating at 13.0 GHz. The wavenumber is

$$k = \frac{2\pi f \sqrt{\epsilon_r}}{c} = \frac{2\pi (13 \times 10^9) \sqrt{2.25}}{3 \times 10^8} = 408.4 \text{ m}^{-1}$$

and the propagation constant of the TE₁₁ mode is

$$\beta = \sqrt{k^2 - \left(\frac{p_{11}'}{a}\right)^2} = \sqrt{(408.4)^2 - \left(\frac{1.841}{0.005}\right)^2} = 176.7 \text{ m}^{-1}.$$

The attenuation due to dielectric loss is calculated from (3.29) as

$$\alpha_d = \frac{k^2 \tan \delta}{2\beta} = \frac{(408.4)^2(0.001)}{2(176.7)} = 0.47$$
 Np/m.

The conductivity of silver is $\sigma = 6.17 \times 10^7$ S/m, so the surface resistance is

$$R_s = \sqrt{rac{\omega \mu_0}{2\sigma}} = 0.029 \,\Omega.$$

Then from (3.133) the attenuation due to metallic loss is

$$\mathbf{a}_{c} = \frac{R_{s}}{ak\eta\beta} \left(k_{c}^{2} + \frac{k^{2}}{p_{11}^{\prime 2} - 1} \right) = 0.066 \text{ Np/m}$$

So the total attenuation factor is

$$\alpha = \alpha_c + \alpha_d = 0.54$$
 Np/m.

Note that the dielectric loss dominates this result. The attenuation in the 50 cm long guide is

attenuation (dB) =
$$-20 \log e^{-\alpha t} = -20 \log r^{-(0.537)(0.5)} = 2.38 \text{ dB}.$$
 ()

. COAXIAL LINE TEM Modes

3.5

Although we have already discussed TEM mode propagation on a coaxial line in Chapter 2, we will briefly reconsider it here in the context of the general framework that was briefly developed in the previous section.

The coaxial line geometry is shown in Figure 3.15, where the inner conductor is at a potential of V_{ϕ} volts and the outer conductor is at zero volts. From Section 3.1, we know that the fields can be derived from a scalar potential function, $\Phi(\rho, \phi)$, which is a solution to Laplace's equation (3.14): in cylindrical coordinates Laplace's equation takes





the form

$$\frac{1}{\rho} \frac{\partial}{\partial \rho} \left(\rho \frac{\partial \Phi(\rho, \phi)}{\partial \rho} \right) + \frac{1}{\rho^2} \frac{\partial^2 \Phi(\rho, \phi)}{\partial \phi^2} = 0, \qquad 3.143$$

This equation must be solved for $\Phi(\rho, \phi)$ subject to the boundary conditions that

$$\Phi(a,\phi) = V_o, \qquad \qquad 3.144a$$

$$\Phi(b,\phi) = 0. \qquad \qquad 3.144b$$

Using the method of separation of variables, we let $\Phi(\rho, \phi)$ be expressed in product form as

$$\Phi(\rho,\phi) = R(\rho)P(\phi). \qquad 3.145$$

Substitution of (3.145) into (3.143) gives

$$\frac{\rho}{R}\frac{\partial}{\partial\rho}\left(\rho\frac{dR}{d\rho}\right) + \frac{1}{\rho}\frac{d^2P}{d\phi^2} = 0.$$
3.146

By the usual separation of variables argument, the two terms in (3.146) must be equal to constants, so that

$$\frac{\rho}{R}\frac{\partial}{\partial\rho}\left(\rho\frac{dR}{d\rho}\right) = -k_{\rho}^{2}, \qquad 3.147$$

$$\frac{1}{P}\frac{d^2P}{d\phi^2} = -k_{\phi}^2.$$
 3.148

and

$$k_{\mu}^2 + k_{\phi}^2 = 0. \qquad 3.149$$

The general solution to (3.148) is

$$P(\phi) = A\cos n\phi + B\sin n\phi, \qquad 3.150$$

where $k_{\phi} = n$ must be an integer, since increasing ϕ by a multiple of 2π should not change the result. Now, because of the fact that the boundary conditions of (3.144) do not vary with ϕ , the potential $\Phi(\rho, \phi)$ should not vary with ϕ . Thus, *n* must be zero. By (3.149), this implies that k_{ρ} must also be zero, so that the equation for $R(\rho)$ in (3.147) reduces to

$$\frac{\partial}{\partial \rho} \left(\rho \frac{dR}{d\rho} \right) = 0.$$

The solution for $R(\rho)$ is then

$$R(p) = C \ln \rho + D,$$

$$\Phi(\rho, \phi) = C \ln \rho + D.$$
3.151

and so

Applying the boundary conditions of (3.144) gives two equations for the constants C and D:

$$\Phi(a,\phi) = V_o = C \ln a + D, \qquad 3.152a$$

$$\Phi(b;\phi) = 0 = C \ln b + D.$$
 3.152b

After solving for C and D, the final solution for $\Phi(\rho, \phi)$ can be written as

$$\Phi(\rho,\phi) = \frac{V_o \ln b/\rho}{\ln b/q}.$$
 3.153

The \overline{E} and \overline{H} fields can then be found using (3.13) and (3.18). Then the voltage, current, and characteristic impedance can be determined as in Chapter 2. Attenuation due to dielectric or conductor loss has already been treated in Chapter 2 and will not be repeated here.

Higher Order Modes

The coaxial line, like the parallel plate waveguide. can also support TE and TM waveguide modes in addition to a TEM mode. In practice, these modes are usually cutoff (evanescent), and so have only a reactive effect near discontinuities or sources, where they are excited. It is important in practice, however, to be aware of the cutoff frequency of the lowest order waveguide-type modes, to avoid the propagation of these modes. Deleterious effects may otherwise occur, due to the superposition of two or more propagating modes with different propagation constants. Avoiding the propagation of higher order modes sets an upper limit on the size of a coaxial cable: this ultimately limits the power handling capacity of a coaxial line (see the Point of Interest on power capacity of transmission lines).

We will derive the solution for the TE modes of the coaxial line: the TE₁₁ mode is the dominant waveguide mode of the coaxial line, and so is of primary importance.

For TE modes, $E_z = 0$, and H_z satisfies the wave equation of (3.112):

$$\left(\frac{\partial^2}{\partial\rho^2} + \frac{1}{\rho}\frac{\partial}{\partial\rho} + \frac{1}{\rho^2}\frac{\partial^2}{\partial\phi^2} + k_c^2\right)h_z(\rho,\phi) = 0, \qquad 3.154$$

where $H_z(\rho, \phi, z) = h_z(\rho, \phi)e^{-j\beta z}$, and $k_c^2 = k^2 - \beta^2$. The general solution to this equation, as derived in Section 3.4, is given by the product of (3.118) and (3.120):

$$h_{z}(\rho,\phi) = (A\sin n\phi + B\cos n\phi)(CJ_{n}(k_{c}\rho) + DY_{n}(k_{c}\rho)). \qquad 3.155$$

In this case, $a \le p \le b$, so we have no reason to discard the Y_n term. The boundary conditions are that

$$E_{\phi}(\rho, \phi) = 0.$$
 for $\rho = a, b.$ 3.156

Using (3.110b) to find E_{ϕ} from H_z gives

$$E_{\phi} = \frac{j\omega\mu}{k_c} (A\sin n\phi + B\cos n\phi) (CJ'_n(k_c\rho) + DY'_n(k_c\rho)) e^{-j\beta\pi}.$$
 3.157

Applying (3.156) to (3.157) gives two equations:

$$CJ'_{n}(k_{c}a) + DY'_{n}(k_{c}a) = 0,$$
 3.158a

$$CJ'_{n}(k_{c}b) + DY'_{n}(k_{c}b) = 0.$$
 3.158b

Since this is a homogeneous set of equations, the only nontrivial $(C \neq 0, D \neq 0)$ solution occurs when the determinant is zero. Thus we must have

$$J'_{n}(k_{c}a)Y'_{n}(k_{c}b) = J'_{n}(k_{c}b)Y'_{n}(k_{c}a).$$
3.159

This is a characteristic (or eigenvalue) equation for k_c . The values of k_c that satisfy (3.159) then define the TE_{am} modes of the coaxial line.

Equation (3.159) is a transcendental equation, which must be solved numerically for k_c . Figure 3.16 shows the result of such a solution for n = 1, for various b/a ratios. An approximate solution that is often used in practice is

$$k_c = \frac{2}{a+b}$$

Once k_c is known, the propagation constant or cutoff frequency can be determined. Solutions for the TM modes can be found in a similar manner; the required determinantal equation is the same as (3.159), except for the derivatives. Field lines for the TEM and TE₁₁ modes of the coaxial line are shown in Figure 3.17.

EXAMPLE 3.3 Higher Order Mode of a Coaxial Line

Consider a piece of RG-142 coaxial cable, with a = 0.035" and b = 0.116", and a dielectric with $\epsilon_r = 2.2$. What is the highest usable frequency, before the TE₁₁ waveguide mode starts to propagate?



FIGURE 3.16 Normalized eutoff frequency of the dominant TE₁₁ waveguide mode for a coaxial line.

Solution We have

$$\frac{b}{a} = \frac{0.116}{0.035} = 3.3.$$

From Figure 3.16 this value of b/a gives $k_e a = 0.47$ (the approximate result is $k_e a = 2/(1 + b/a) = 0.465$). Thus, the cutoff frequency of the TE₁₁ mode is

$$f_r = \frac{ck_r}{2\pi\sqrt{\epsilon_r}} - 17 \text{ GHz}$$

In practice, a 5% safety margin is usually recommended, so $f_{\text{max}} = 0.95(17 \text{ GHz}) = 16 \text{ GHz}$.



FIGURE 3.17 Field lines for the (a) TEM and (b) TE11 modes of a coaxial line.



POINT OF INTEREST: Coaxial Connectors

Most coaxial cables and connectors in common use have a 50 Ω characteristic impedance, with an exception being the 75 Ω coax used in television systems. The reasoning behind these choices is that an air-filled coaxial line has minimum attenuation for a characteristic impedance of 77 Ω (Problem 2.26), while maximum power capacity occurs for a characteristic impedance of 30 Ω (Problem 3.27). A 50 Ω characteristic impedance thus represents a compromise between minimum attenuation and maximum power capacity. Requirements for coaxial connectors include low SWR, higher-order-mode-free operation at a high frequency, high repeatability after a connect-disconnect cycle, and mechanical strength. Connectors are used in pairs, with a male end and a female end (or plug and jack). Below we describe some of the most common microwave coaxial connectors.

Type-N connector. This connector was developed in 1942 and named after P. Neill, who worked on its design at Bell Labs. The male and female connectors thread together; the outer diameter of the female connector is about 0.625 in., so this is a relatively large connector. The recommended upper operating frequency ranges from 11 to 18 GHz, depending on cable size. The SWR for a mated connector pair is typically less than 1.07.

SMA connector. The need for a smaller and lighter connector led to the development of the subminiature SMA connector in the early 1960s. The outer diameter of the female end of the SMA connector is about 0.250 in. It can be used up to 25 GHz, and is probably the most frequently used microwave connector today. Increasing demand for millimeter wave components has led to the development of two popular variations of the SMA connector: the K-connector (usable to 40 GHz) and the 2.4 mm connector (usable to 50 GHz).

SSMA connector. The SSMA (scaled SMA) connector is similar in design to the SMA connector, but smaller in size. The outer diameter of the female end is about 0.192 in., and the maximum operating frequency is about 38 GHz.

APC-7 connector. This is a precision connector (Amphenol precision connector) that can repeatably achieve an SWR less than 1.04 at frequencies up to 18 GHz. The connectors are "sexless," with butt contact between both the inner conductors and the outer conductors. Reference J. H. Bryant, "Coaxial Transmission Lines, Related Two-Conductor Transmission Lines, Connectors, and Components: A U.S. Historical Perspective," *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-32, pp. 970–983, September 1984.

3.6 ____ SURFACE WAVES ON A GROUNDED DIELECTRIC SLAB

We briefly discussed surface waves in Chapter 1, in connection with the field of a plane wave totally reflected from a dielectric interface. In general, surface waves can exist in a variety of geometries involving dielectric interfaces. Here we consider the TM and TE surface waves that can be excited along a grounded dielectric slab. Other geometries that can be used as surface waveguides include an ungrounded dielectric slab, a dielectric rod, a corrugated conductor, or a dielectric coated conducting rod.

Surface waves are typified by a field that decays exponentially away from the dielectric surface, with most of the field contained in or near the dielectric. At higher frequencies the field generally becomes more tightly bound to the dielectric, making such waveguides practical. Because of the presence of the dielectric, the phase velocity of a surface wave is less than the velocity of light in a vacuum. Another reason for studying surface waves is that they may be excited on some types of planar transmission lines, such as microstrip and slotline.

TM Modes

Figure 3.18 shows the geometry of a grounded dielectric slab waveguide. The dielectric slab, of thickness d and relative dielectric constant ϵ_r , is assumed to be of infinite extent in the y and z directions. We will assume propagation in the +z direction with an $e^{-j\beta z}$ propagation factor, and no variation in the y direction $(\partial/\partial y = 0)$.

Because there are two distinct regions, with and without a dielectric, we must separately consider the field in these regions, and then match tangential fields across the





interface. E_z must satisfy the wave equation of (3.25) in each region:

$$\left(\frac{\partial^2}{\partial x^2} + \epsilon_r k_0^2 - \beta^2\right) e_z(x, y) = 0, \quad \text{for } 0 \le x \le d, \quad 3.160a$$

$$\left(\frac{\partial^2}{\partial x^2}+k_0^2-\beta^2\right)e_z(x,y)=0, \quad \text{for } d\leq x<\infty, \quad 3.160b$$

where $E_z(x, y, z) = e_z(x, y)e^{-j\beta z}$.

Now define the cutoff wavenumbers for the two regions as

$$k_c^2 = \epsilon_r k_0^2 - \beta^2, \qquad \qquad 3.161a$$

$$h^2 = \beta^2 - k_0^2, \qquad 3.161b$$

where the sign on h^2 has been selected in anticipation of an exponentially decaying result for x > d. Observe that the same propagation constant β has been used for both regions. This must be the case to achieve phase matching of the tangential fields at the x = dinterface for all values of z.

The general solutions to (3.160) are then

$$e_{\mathbf{z}}(x,y) = A \sin k_{e} x + B \cos k_{e} x, \quad \text{for } 0 \le x \le d, \quad 3.162a$$

$$e_x(x,y) = Ce^{hx} + De^{-hx},$$
 for $d \le x < \infty$. 3.162b

Note that these solutions are valid for k_c and h either real or imaginary; it will turn out that both k_c and h are real, because of the choice of definitions in (3.161).

The boundary conditions that must be satisfied are

$$E_{z}(x, y, z) = 0,$$
 at $x = 0,$ 3.163a

$$E_x(x,y,z) < \infty,$$
 as $x \to \infty,$ $3.163b$

$$E_z(x, y, z)$$
 continuous, at $x = d$, 3.163c

$$H_y(x, y, z)$$
 continuous, at $x = d$. 3.163d

From (3.23), $H_x = E_y = H_z = 0$. Condition (3.163a) implies that B = 0 in (3.162a). Condition (3.163b) comes about as a requirement for finite fields (and energy) infinitely far away from a source, and implies that C = 0. The continuity of E_z leads to

$$A\sin k_c d = De^{-hd}, \qquad 3.164a$$

while (3.23b) must be used to apply continuity to H_y , to obtain

$$\frac{\epsilon_r A}{k_c} \cos k_c d = \frac{D}{h} e^{-hd}.$$
 3.164b

For a nontrivial solution, the determinant of the two equations of (3.164) must vanish, leading to

$$k_c \tan k_c d = \epsilon_r h. \tag{3.165}$$



FIGURE 3.19 Graphical solution of the transcendental equation for the cutoff frequency of a TM surface wave mode of the grounded dielectric slab.

Eliminating β from (3.161a) and (3.161b) gives

$$k_c^2 + h^2 = (\epsilon_\tau - 1)k_{0*}^2 3.166$$

Equations (3.165) and (3.166) constitute a set of simultaneous transcendental equations that must be solved for the propagation constants k_c and h, given k_a and ϵ_v . These equations are best solved numerically, but Figure 3.19 shows a graphical representation of the solutions. Multiplying both sides of (3.166) by d^2 gives

$$(k_r d)^2 + (hd)^2 = (\epsilon_r - 1)(k_0 d)^2.$$

which is the equation of a circle in the $k_c d$, h d plane, as shown in Figure 3.19. The radius of the circle is $\sqrt{\epsilon_r - 1}k_0 d$, which is proportional to the electrical thickness of the dielectric slab. Multiplying (3.165) by d gives

$$k_c d \tan k_c d = \epsilon_r h d$$
.

which is also plotted in Figure 3.19. The intersection of these curves implies a solution to both (3.165) and (3.166). Observe that k_c may be positive or negative; from (3.162a) this is seen to merely change the sign of the constant A. As $\sqrt{\epsilon_r - 1}k_0d$ becomes larger, the circle may intersect more than one branch of the tangent function, implying that more than one TM mode can propagate. Solutions for negative h, however, must be excluded since we assumed h was positive real when applying boundary condition (3.163b).

For any nonzero thickness slab, with a permittivity greater than unity, there is at least one propagating TM mode, which we will call the TM_0 mode. This is the dominant mode of the dielectric slab waveguide, and has a zero cutoff frequency. (Although for $k_0 = 0$, $k_c = h = 0$ and all fields vanish.) From Figure 3.19, it can be seen that the next TM mode, the TM_1 mode, will not turn on until the radius of the circle becomes greater than π . The cutoff frequency of the TM_n mode can then be derived as

$$f_c = \frac{nc}{2d\sqrt{\epsilon_r - 1}}, \qquad n = 0, 1, 2, \dots,$$
 3.167

Once k_c and h have been found for a particular surface wave mode, the field expressions can be found as,

$$E_{z}(x, y, z) = \begin{cases} A \sin k_{c} x e^{-j\beta z} & \text{for } 0 \leq x \leq d \\ A \sin k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } 0 \leq x \leq d \\ \text{for } d \leq x < \infty, \end{cases}$$

$$E_{x}(x, y, z) = \begin{cases} \frac{-j\beta}{k_{c}} A \cos k_{c} x e^{-j\beta z} & \text{for } 0 \leq x \leq d \\ \frac{-j\beta}{h} A \sin k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } d \leq x < \infty, \end{cases}$$

$$H_{y}(x, y, z) = \begin{cases} \frac{-j\omega\epsilon_{0}\epsilon_{r}}{k_{c}} A \cos k_{c} x e^{-j\beta z} & \text{for } 0 \leq x \leq d \\ \frac{-j\omega\epsilon_{0}}{h} A \sin k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } 0 \leq x \leq d \\ \frac{-j\omega\epsilon_{0}}{h} A \sin k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } 0 \leq x \leq d \\ \frac{-j\omega\epsilon_{0}}{h} A \sin k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } d \leq x < \infty. \end{cases}$$
3.168a

TE Modes

TE modes can also be supported by the grounded dielectric slab. The H_z field satisfies the wave equations

$$\left(\frac{\partial^2}{\partial x^2}+k_c^2\right)h_s(x,y)=0, \qquad ext{for } 0\leq x\leq d, \qquad 3.169a$$

$$\left(\frac{\partial^2}{\partial x^2}-h^2\right)h_z(x,y)=0,$$
 for $\tilde{a}\leq x<\infty,$ 3.169b

with $H_x(x, y, z) = h_z(x, y)e^{-j\beta z}$, and k_c^2 and h^2 defined in (3.161a) and (3.161b). As for the TM modes, the general solutions to (3.169) are

$$h_{z}(x, y) = A \sin k_{c} x + B \cos k_{c} x, \qquad 3.170a$$

$$h_x(x,y) = Ce^{hx} + De^{-hx}.$$
 3.170b

To satisfy the radiation condition, C = 0. Using (3.19d) to find E_y from H_z leads to A = 0 for $E_y = 0$ at x = 0, and to the equation

$$\frac{-B}{k_c}\sin k_c d = \frac{D}{h}e^{-hd},$$
3.171a

for continuity of E_y at x = d. Continuity of H_z at x = d gives

$$B\cos k_c d = De^{-hd}.$$
 3.171b

Simultaneously solving (3.171a) and (3.171b) leads to the determinantal equation

$$-k_c \cot k_c d = h. \qquad 3.172$$

From (3.161a) and (3.161b) we also have that

$$k_c^2 + h^2 = (\epsilon_r - 1)k_0^2.$$
 3.173



FIGURE 3.20 Graphical solution of the transcendental equation for the cutoff frequency of a TE surface wave mode. Figure depicts a mode below cutoff.

Equations (3.172) and (3.173) must be solved simultaneously for the variables k_v and h. Equation (3.173) again represents circles in the $k_c d$. hd plane, while (3.172) can be rewritten as

$$-k_{c}d \cot k_{c}d = hd.$$

and plotted as a family of curves in the $k_e d_e h d$ plane, as shown in Figure 3.20. Since negative values of h must be excluded, we see from Figure 3.20 that the first TE mode does not start to propagate until the radius of the circle, $\sqrt{\epsilon_r - 1} k_0 d$, becomes greater than $\pi/2$. The cutoff frequency of the TE_n modes can then be found as

$$f_c = \frac{(2n-1)c}{4d\sqrt{c_c-1}},$$
 for $n = 1, 2, 3, ...,$ 3.174

Comparing with (3.167) shows that the order of propagation for the TM_n and TE_n modes is, TM_0 , TE_1 , TM_1 , TE_2 , TM_2 , ...,

After finding the constants k_i and b_i the field expressions can be derived as

$$H_{z}(x, y, z) = \begin{cases} B \cos k_{c} x e^{-j\beta \frac{\pi}{2}} & \text{for } 0 \le x \le d \\ B \cos k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } 0 \le x \le d \\ \text{for } d \le x < \infty. \end{cases}$$

$$H_{x}(x, y, z) = \begin{cases} \frac{j\beta B}{k_{c}} \sin k_{c} x e^{-j\beta z} & \text{for } 0 \le x \le d \\ \frac{-j\beta B}{k_{c}} \cos k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } 0 \le x \le d \\ \frac{-j\beta B}{k_{c}} \cos k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } 0 \le x \le d \\ \frac{j\omega \mu_{o} B}{k_{c}} \sin k_{c} x e^{-j\beta z} & \text{for } 0 \le x \le d \\ \frac{j\omega \mu_{o} B}{k_{c}} \cos k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } 0 \le x \le d \\ \frac{j\omega \mu_{o} B}{k_{c}} \cos k_{c} d e^{-h(x-d)} e^{-j\beta z} & \text{for } 0 \le x \le d \end{cases}$$
3.175c





EXAMPLE 3.4 Surface Wave Propagation Constants

Calculate and plot the propagation constants of the first three propagating surface wave modes of a grounded dielectric sheet with $\epsilon_r = 2.55$, for $d/\lambda_0 = 0$ to 1.2.

Solution

The first three propagating surface wave modes are the TM_0 , TE_1 , and TM_1 modes. The cutoff frequencies for these modes can be found from (3.167) and (3.174) as

$$TM_0: f_c = 0 \implies \frac{d}{\lambda_0} = 0,$$

$$TE_1: f_c = \frac{c}{4d\sqrt{\epsilon_r - 1}} \implies \frac{d}{\lambda_0} = \frac{1}{(4\sqrt{\epsilon_r - 1})},$$

$$TM_1: f_c = \frac{c}{2d\sqrt{\epsilon_r - 1}} \implies \frac{d}{\lambda_0} = \frac{1}{(2\sqrt{\epsilon_r - 1})}.$$

The propagation constants must be found from the numerical solution of (3.165) and (3.166) for the TM modes, and (3.172) and (3.173) for the TE modes. This can be done with a relatively simple root-finding algorithm (see the Point of Interest on root-finding algorithms); the results are shown in Figure 3.21.



POINT OF INTEREST: Root-Finding Algorithms

In several examples throughout this book we will need to numerically find the root of a transcendental equation, so it may be useful to review two relatively simple but effective algorithms for doing this. Both methods can be easily programmed.

In the interval-halving method the root of f(x) = 0 is first bracketed between the values x_1 and x_2 . These values can often be estimated from the problem under consideration. If a single root lies between x_1 and x_2 , then $f(x_1)f(x_2) < 0$. An estimate, x_3 , of the root is made by halving the interval between x_1 and x_2 . Thus,

$$x_3 \coloneqq \frac{x_1+x_2}{2}.$$

If $f(x_1)f(x_3) < 0$, then the root must lie in the interval $x_1 < x < x_3$; if $f(x_3)f(x_2) < 0$, then the root must be in the interval $x_2 < x < x_3$. A new estimate, x_4 , can be made by halving the appropriate interval, and this process repeated until the location of the root has been determined with the desired accuracy. The figure below illustrates this algorithm for several iterations.

The Newton-Raphson method begins with an estimate, x_1 , of the roat of f(x) = 0. Then a new estimate, x_2 , is obtained from the formula

$$x_2 = x_1 - \frac{f(x_1)}{f'(x_1)}.$$

where $f'(x_1)$ is the derivative of f(x) at x_1 . This result is easily derived from a two-term Taylor series expansion of f(x) near $x = x_1:f(x) = f(x_1) + (x - x_1)f'(x_1)$. It can also be interpreted geometrically as fitting a straight line at $x = x_1$ with the same slope as f(x) at this point: this line then intercepts the x-axis at $x = x_2$, as shown in the figure below. Reapplying the above formula gives improved estimates of the root. Convergence is generally much faster than with the interval halving method, but a disadvantage is that the derivative of f(x) is required: this can often be computed numerically. The Newton-Raphson technique can easily be applied to the case where the root is complex (a situation that occurs, for example, when finding the propagation constant of a line or guide with loss).



Reference: R. W. Hornbeck, Numerical Methods, Quantum Publishers, New York, 1975,

3.7 STRIPLINE

We now consider stripline, a planar-type of transmission line that lends itself well to microwave integrated circuitry and photolithographic fabrication. The geometry of a stripline is shown in Figure 3.22a. A thin conducting strip of width W is centered between two wide conducting ground planes of separation h, and the entire region between the ground planes is filled with a dielectric. In practice, stripline is usually constructed by etching the center conductor on a grounded substrate of thickness b/2, and then covering



FIGURE 3.22 Stripline transmission line. (a) Geometry. (b) Electric and magnetic field lines.

with another grounded substrate of the same thickness. An example of a stripline circuit is shown in Figure 3.23.

Since stripline has two conductors and a homogeneous dielectric, it can support a TEM wave, and this is the usual mode of operation. Like the parallel plate guide and coaxial lines, however, the stripline can also support higher order TM and TE modes, but these are usually avoided in practice (such modes can be suppressed with shorting screws between the ground planes and by restricting the ground plane spacing to less than $\lambda/4$). Intuitively, one can think of stripline as a sort of "flattened out" coax-both have a center conductor completely enclosed by an outer conductor and are uniformly filled with a dielectric medium. A sketch of the field lines for stripline is shown in Figure 3.22b. The main difficulty we will have with stripline is that it does not lend itself to a simple analysis, as did the transmission lines and waveguides that we have previously discussed. Since we will be concerned primarily with the TEM mode of the stripline, an electrostatic analysis is sufficient to give the propagation constant and characteristic impedance. An exact solution of Laplace's equation is possible by a conformal mapping approach [6], but the procedure and results are cumbersome. Thus, we will present closed-form expressions that give good approximations to the exact results and then discuss an approximate numerical technique for solving Laplace's equation for a geometry similar to stripline; this technique will also be applied to microstrip line in the following section.

Formulas for Propagation Constant, Characteristic Impedance, and Attenuation

From Section 3.1 we know that the phase velocity of a TEM mode is given by

$$v_p = 1/\sqrt{\mu_0} \overline{\epsilon_0 \epsilon_r} = c/\sqrt{\epsilon_r}, \qquad 3.176$$



FIGURE 3.23

B Photograph of a stripline circuit assembly, showing four quadrature hybrids, open-circuit tuning stubs, and coaxial transitions. Countesy of Harlan Howe, Jr., M/A-COM Inc.

thus the propagation constant of the stripline is

$$\beta = \frac{\omega}{v_p} = \omega \sqrt{\mu_0 \epsilon_0 \epsilon_r} = \sqrt{\epsilon_r} k_0. \qquad 3.177$$

In (3.176), $c = 3 \times 10^8$ m/sec is the speed of light in free-space. The characteristic impedance of a transmission line is given by

$$Z_0 = \sqrt{\frac{L}{C}} = \frac{\sqrt{LC}}{C} = \frac{1}{v_p C}.$$
 3.178

where L and C are the inductance and capacitance per unit length of the line. Thus, we can find Z_0 if we know C. As mentioned above, Laplace's equation can be solved by conformal mapping to find the capacitance per unit length of the stripline. The resulting solution, however, involves complicated special functions [6], so for practical computations simple formulas have been developed by curve fitting to the exact solution [6], [7]. The resulting formula for characteristic impedance is

$$Z_0 = \frac{30\pi}{\sqrt{\epsilon_r}} \frac{b}{W_e + 0.441b}, \qquad 3.179a$$

where W_e is the effective width of the center conductor given by

$$\frac{W_{\epsilon}}{b} = \frac{W}{b} - \begin{cases} 0 & \text{for } \frac{W}{b} > 0.35\\ (0.35 - W/b)^2 & \text{for } \frac{W}{b} < 0.35. \end{cases}$$
3.179b

These formulas assume a zero strip thickness, and are quoted as being accurate to about 1% of the exact results. It is seen from (3.179) that the characteristic impedance decreases as the strip width W increases,

When designing stripline circuits, one usually needs to find the strip width, given the characteristic impedance (and height b and permittivity ϵ_r), which requires the inverse of the formulas in (3.179). Such formulas have been derived as

$$\frac{W}{b} = \begin{cases} x & \text{for } \sqrt{\epsilon_r} \ Z_0 < 120\\ 0.85 - \sqrt{0.6 - x} & \text{for } \sqrt{\epsilon_r} \ Z_0 > 120, \end{cases}$$
 3.180*a*

where

$$x = \frac{30\pi}{\sqrt{\epsilon_r Z_0}} - 0.441.$$
 3.180b

Since stripline is a TEM type of line, the attenuation due to dielectric loss is of the same form as that for other TEM lines and is given in (3.30). The attenuation due to conductor loss can be found by the perturbation method or Wheeler's incremental inductance rule. An approximate result is

$$\alpha_{c} = \begin{cases} \frac{2.7 \times 10^{-3} R_{s} \epsilon_{r} Z_{0}}{30\pi (b-t)} A & \text{for } \sqrt{\epsilon_{r}} Z_{0} < 120 \\ \frac{0.16 R_{s}}{Z_{0} b} B & \text{for } \sqrt{\epsilon_{r}} Z_{0} > 120 \end{cases}$$

$$A = 1 + \frac{2W}{b-t} + \frac{1}{\pi} \frac{b+t}{b-t} \ln\left(\frac{2b-t}{t}\right),$$

$$B = 1 + \frac{b}{(0.5W+0.7t)} \left(0.5 + \frac{0.414t}{W} + \frac{1}{2\pi} \ln\frac{4\pi W}{t}\right),$$

with

B

٦

EXAMPLE 3.5 Stripline Design

Find the width for a 50 Ω copper stripline conductor, with b = 0.32 cm and $\epsilon_r = 2.20$. If the dielectric loss tangent is 0.001 and the operating frequency

is 10 GHz, calculate the attenuation in dB/ λ . Assume a conductor thickness of t = 0.01 mm.

Solution

Since $\sqrt{\epsilon_r}Z_0 = \sqrt{2.2}(50) = 74.2 < 120$, and $x = 30\pi/(\sqrt{\epsilon_r}Z_0) - 0.441 = 0.830$. (3.180) gives the width as W = bx = (0.32)(0.830) = 0.266 cm. At 10 GHz, the wavenumber is

$$k = \frac{2\pi f \sqrt{\epsilon_r}}{r} = 310.6 \text{ m}^{-1}.$$

From (3.30) the dielectric attenuation is

$$\alpha_d = \frac{k \tan \delta}{2} = \frac{(310.6)(0.001)}{2} = 0.155 \text{ Np/m}.$$

The surface resistance of copper at 10 GHz is $R_s = 0.026 \Omega$. Then from (3.181) the conductor attenuation is

$$a_c = \frac{2.7 \times 10^{-3} R_s c_c Z_0 A}{30\pi (b-t)} = 0.122 \text{ Np/m}.$$

since A = 4.74. The total attenuation constant is

$$\alpha = \alpha_d + \alpha_r = 0.277$$
 Np/m.

In dB,

$$r_1(dB) = 20 \log c^{\alpha} = 2.41 \, dB/m.$$

At 10 GHz, the wavelength on the stripline is

$$\lambda = \frac{c}{\sqrt{\epsilon_y} f} = 2.02 \text{ cm.}$$

so in terms of wavelength the attenuation is

$$\alpha(dB) = (2.41)(0.0202) = 0.049 \ dB/\lambda,$$

An Approximate Electrostatic Solution

Many practical problems in microwave engineering are very complicated and do not lend themselves to straightforward analytic solutions, but require some sort of numerical approach. Thus it is useful for the student to become aware of such techniques; we will introduce such methods when appropriate throughout this book, beginning with a numerical solution for the characteristic impedance of stripline.

We know that the fields of the TEM mode on a stripline must satisfy Laplace's equation, (3.11), in the region between the two parallel plates. The actual stripline geometry of Figure 3.22a extends to $\pm\infty$, which makes the analysis more difficult. Since we suspect, from the field line drawing of Figure 3.22b, that the field lines do not extend very far away from the center conductor, we can simplify the geometry by fruncating the plates beyond some distance, say |x| > a/2, and placing metal walls on the sides. Thus, the geometry we will analyze looks like that shown in Figure 3.24,



FIGURE 3.24 Geometry of enclosed stripline.

where a >> b so that the fields around the center conductor are not perturbed by the sidewalls. We then have a closed, finite region in which the potential $\Phi(x, y)$ satisfies Laplace's equation,

$$abla_t^2 \Phi(x, y) = 0, \quad \text{for } |x| \le a/2, \quad 0 \le y \le b, \quad 3.182$$

with the boundary conditions that

$$\Phi(x, y) = 0,$$
 at $x = \pm a/2,$ 3.183a

$$\Phi(x, y) = 0,$$
 at $y = 0, b.$ 3.183b

Laplace's equation can be solved by the method of separation of variables. Since the center conductor at y = b/2 will contain a surface charge density, the potential $\Phi(x, y)$ will have a slope discontinuity there, because $\overline{D} = -\epsilon_0 \epsilon_r \nabla_t \Phi$ is discontinuous at y = b/2. So separate solutions for $\Phi(x, y)$ must be found for 0 < y < b/2, and b/2 < y < b. The general solutions for $\Phi(x, y)$ in these two regions can be written as

$$\Phi(x,y) = \begin{cases}
\sum_{\substack{n=1 \\ \text{odd}}}^{\infty} A_n \cos \frac{n\pi x}{a} \sinh \frac{n\pi y}{a} & \text{for } 0 \le y \le b/2 \\
\sum_{\substack{n=1 \\ \text{odd}}}^{\infty} B_n \cos \frac{n\pi x}{a} \sinh \frac{n\pi}{a} (b-y) & \text{for } b/2 \le y \le b.
\end{cases}$$
3.184

In this solution, only the odd-*n* terms are needed because the solution is an even function of x. The reader can verify by substitution that (3.184) satisfies Laplace's equation in the two regions and satisfies the boundary conditions of (3.183).

Now, the potential must be continuous at y = b/2, which from (3.184) leads to

$$A_n = \mathcal{B}_n. \tag{3.185}$$

The remaining set of constants, A_n , can be found by solving for the charge density on the center strip. Since $E_y = -\partial \Phi / \partial y$, we have

$$E_{y} = \begin{cases} -\sum_{\substack{n=1\\\text{odd}}}^{\infty} A_{n}\left(\frac{n\pi}{a}\right)\cos\frac{n\pi x}{a}\cosh\frac{n\pi y}{a} & \text{for } 0 \le y \le b/2 \\ \sum_{\substack{n=1\\\text{odd}}}^{\infty} A_{n}\left(\frac{n\pi}{a}\right)\cos\frac{n\pi x}{a}\cosh\frac{n\pi}{a}(b-y) & \text{for } b/2 \le y \le b. \end{cases}$$
3.186

The surface charge density on the strip at y = b/2 is

$$p_{s} = D_{y}(x, y) = b/2^{+}) - D_{y}(x, y) = b/2^{-})$$

$$= \epsilon_{0}\epsilon_{r}[E_{y}(x, y) = b/2^{+}) - E_{y}(x, y) = b/2^{-}]$$

$$= 2\epsilon_{0}\epsilon_{r}\sum_{\substack{n=1\\ n \neq d}}^{\infty} A_{n}\left(\frac{n\pi}{a}\right)\cos\frac{n\pi s}{y}\cosh\frac{n\pi b}{2a}.$$
3.187

which is seen to be a Fourier series in x for the surface charge density, p_s . If we know the surface charge density, we could easily find the unknown constants, A_n , and then the capacitance. We do not know the exact surface charge density, but we can make a good guess by approximating it as a constant over the width of the strip

$$\varphi_{s}(\boldsymbol{x}) = \begin{cases}
1 & \text{for } |\boldsymbol{x}| < W/2 \\
0 & \text{for } |\boldsymbol{x}| > W/2.
\end{cases}$$
3.188

Equating this to (3.187) and using the orthogonality properties of the $\cos(n\pi x/a)$ functions gives the constants A_n as

$$A_m = \frac{2a\sin(n\pi W/2a)}{(n\pi)^2 \epsilon_0 \epsilon_n \cosh(n\pi b/2a)},$$
 3.189

The voltage of the center strip relative to the bottom conductor is

$$V = -\int_0^{b/2} E_y(x = 0, y) \, dy = \sum_{\substack{n=1 \\ n \neq d}}^{\infty} A_n \, \sinh \, \frac{n\pi b}{2a}.$$
 3.190

The total charge, per unit length, on the center conductor is

$$Q = \int_{-W/2}^{W/2} \rho_s(x) \, dx = W \, C/m, \qquad 3.191$$

so that the capacitance per unit length of the stripline is

$$C = \frac{Q}{V} = \frac{W}{\sum_{\substack{n=1\\ \text{with}}}^{\infty} \frac{2\alpha \sin(n\pi W/2\alpha) \sinh(n\pi h/2\alpha)}{(n\pi)^2 \epsilon_0 \epsilon_r \cosh(n\pi h/2\alpha)}} \quad \text{Fd/m.} \qquad 3.192$$

The characteristic impedance is then found as

$$\dot{Z}_0 = \sqrt{\frac{L}{C}} = \frac{\sqrt{LC}}{C} = \frac{1}{v_p C} = \frac{\sqrt{\epsilon_v}}{eC},$$

where $c = 3 \times 10^8$ m/sec.



EXAMPLE 3.6 Numerical Calculation of Stripline Parameters

Evaluate the above expressions for a stripline having $\epsilon_v = 2.55$ and a = 100b, to find the characteristic impedance for W'/b = 0.25 to 5.0. Compare with the results from (3.179).

Solution

A short BASIC computer program was written to evaluate (3.192). The series was truncated after 500 terms, and the results are shown below.

	Numerical	Formula
W/b	Eq. (3.192)	Eq. (3.179)
0.25	98.8 Ω	86.6 Ω
0.50	73.3	62.7
1.0	49.0	41.0
2.0	28.4	24.2
3.5	16.8	15.0
5.0	11.8	10.8

We see that the results are in reasonable agreement with the closed-form equations of (3.179), particularly for wider strips. Better results could be obtained if more sophisticated estimates were used for the charge density, ρ_s . \bigcirc

3.8 MICROSTRIP

Microstrip line is one of the most popular types of planar transmission lines, primarily because it can be fabricated by photolithographic processes and is easily integrated with other passive and active microwave devices. The geometry of a microstrip line is shown in Figure 3.25a. A conductor of width W is printed on a thin, grounded dielectric substrate of thickness d and relative permittivity ϵ_r ; a sketch of the field lines is shown in Figure 3.25b.

If the dielectric were not present ($c_r = 1$), we could think of the line as a two-wire line consisting of two flat strip conductors of width W, separated by a distance 2d (the ground plane can be removed via image theory). In this case we would have a simple TEM transmission line, with $v_{\mu} = c$ and $\beta = k_0$.

The presence of the dielectric, and particularly the fact that the dielectric does not fill the air region above the strip (y > d), complicates the behavior and analysis of microstrip line. Unlike stripline, where all the fields are contained within a homogeneous dielectric region, microstrip has some (usually most) of its field lines in the dielectric region, concentrated between the strip conductor and the ground plane, and some fraction in the air region above the substrate. For this reason the microstrip line cannot support a pure TEM wave, since the phase velocity of TEM fields in the dielectric region would be $c/\sqrt{\epsilon_r}$, but the phase velocity of TEM fields in the air region would be c. Thus, a phase match at the dielectric-air interface would be impossible to attain for a TEM-type wave.



FIGURE 3.25 Microstrip transmission line. (a) Geometry. (b) Electric and magnetic field lines.

In actuality, the exact fields of a microstrip line constitute a hybrid TM-TE wave, and require more advanced analysis techniques than we are prepared to deal with here. In most practical applications, however, the dielectric substrate is electrically very thin, $(d \ll \lambda)$, and so the fields are quasi-TEM. In other words, the fields are essentially the same as those of the static case. Thus, good approximations for the phase velocity, propagation constant, and characteristic impedance can be obtained from static or quasi-static solutions. Then the phase velocity and propagation constant can be expressed as

$$v_p = \frac{c}{\sqrt{\epsilon_p}}, \qquad \qquad 3.193$$

$$\beta = k_0 \sqrt{\epsilon_v}. \qquad \qquad 3.194$$

where ϵ_r is the effective dielectric constant of the microstrip line. Since some of the field lines are in the dielectric region and some are in air, the effective dielectric constant satisfies the relation.

and is dependent on the substrate thickness. d. and conductor width, W.

We will first present design formulas for the effective dielectric constant and characteristic impedance of microstrip line; these results are curve-fit approximations to rigorous quasi-static solutions [8], [9]. Then we will outline a numerical method of solution (similar to that used in the previous section for stripline) for the capacitance per unit length of microstrip line.

Formulas for Effective Dielectric Constant, Characteristic Impedance, and Attenuation

The effective dielectric constant of a microstrip line is given approximately by

$$\epsilon_e = \frac{\epsilon_r + 1}{2} + \frac{\epsilon_r - 1}{2} \frac{1}{\sqrt{1 + 12d/W}}$$
 3.195

The effective dielectric constant can be interpreted as the dielectric constant of a homogeneous medium that replaces the air and dielectric regions of the microstrip, as shown in Figure 3.26. The phase velocity and propagation constant are then given by (3.193) and (3.194).

Given the dimensions of the microstrip line, the characteristic impedance can be calculated a_8

$$Z_{0} = \begin{cases} \frac{60}{\sqrt{\epsilon_{e}}} \ln\left(\frac{8d}{W} + \frac{W}{4d}\right) & \text{for } W/d \le 1\\ \frac{120\pi}{\sqrt{\epsilon_{e}}\left[W/d + 1.393 + 0.667\ln\left(W/d + 1.444\right)\right]} & \text{for } W/d \ge 1. \end{cases}$$

For a given characteristic impedance Z_0 and dielectric constant ϵ_r , the W/d ratio can be found as

$$\frac{W}{d} = \begin{cases} \frac{8e^{A}}{e^{2A} - 2} & \text{for } W/d < 2\\ \frac{2}{\pi} \left[B - 1 - \ln(2B - 1) + \frac{\epsilon_{r} - 1}{2\epsilon_{r}} \left\{ \ln(B - 1) + 0.39 - \frac{0.61}{\epsilon_{r}} \right\} \right] & \text{for } W/d > 2,\\ 3.197\end{cases}$$

where

$$A = \frac{Z_0}{60} \sqrt{\frac{\epsilon_r + 1}{2}} + \frac{\epsilon_r - 1}{\epsilon_r + 1} \left(0.23 + \frac{0.11}{\epsilon_r} \right)$$

$$377\pi$$

$$B = \frac{377\pi}{2Z_0\sqrt{\epsilon_r}}$$





Considering microstrip as a quasi-TEM line, the attenuation due to dielectric loss can be determined as

$$\alpha_d = \frac{k_0 \epsilon_r (\epsilon_e - 1) \tan \delta}{2\sqrt{\epsilon_r} (\epsilon_r - 1)} \text{ Np/m.}$$
3.198

where $\tan \delta$ is the loss tangent of the dielectric. This result is derived from (3.30) by multiplying by a "filling factor,"

$$\frac{\epsilon_r(\epsilon_e-1)}{\epsilon_e(\epsilon_r-1)}.$$

which accounts for the fact that the fields around the microstrip line are partly in air (lossless) and partly in the dielectric. The attenuation due to conductor loss is given approximately by [8]

$$\alpha_c = \frac{R_s}{Z_0 W} \text{ Np/m.} \qquad 3.199$$

where $R_s = \sqrt{\omega\mu_0/2\sigma}$ is the surface resistivity of the conductor. For most microstrip substrates, conductor loss is much more significant than dielectric loss; exceptions may occur with some semiconductor substrates, however,



EXAMPLE 3.7 Microstrip Design

Calculate the width and length of a microstrip line for a 50 Ω characteristic impedance and a 90° phase shift at 2.5 GHz. The substrate thickness is d = 0.127 cm, with $\epsilon_r = 2.20$.

Solution

We first find W/d for $Z_0 = 50 \ \Omega$, and initially guess that W/d > 2. From (3.197),

$$B = 7.985$$
, $W/d = 3.081$.

So W/d > 2: otherwise we would use the expression for W/d < 2. Then W = 3.081d = 0.391 cm. From (3.195) the effective dielectric constant is

c_e 1.87.

The line length, ℓ , for a 90° phase shift is found as

$$\varphi = 90^{\circ} = \beta \ell = \sqrt{\epsilon_c} k_0 \ell.$$

$$k_0 = \frac{2\pi f}{c} = 52.35 \text{ m}^{-1}.$$

$$\ell = \frac{90^{\circ} (\pi/180^{\circ})}{\sqrt{\epsilon_c} k_0} = 2.19 \text{ cm.}$$

An Approximate Electrostatic Solution

We now look at an approximate quasi-static solution for the microstrip line, so that the appearance of design equations like those of (3.195)-(3.197) is not a complete mystery. This analysis is very similar to that carried out for stripline in the previous section. Like that analysis, it is again convenient to place conducting sidewalls on the microstrip line, as shown in Figure 3.27. The sidewalls are placed at $x = \pm a/2$, where a >> d, so that the walls should not perturb the field lines localized around the strip conductor. We then can solve Laplace's equation in the region between the sidewalls:

$$abla_t^2 \Phi(x, y) = 0, \qquad \text{for } |x| \le a/2, \quad 0 \le y < \infty, \qquad \qquad 3.200$$

with boundary conditions,

$$\Phi(x,y) = 0,$$
 at $x = \pm a/2,$ 3.201a

$$\Phi(x, y) = 0,$$
 at $y = 0, \infty.$ 3.201b

Since there are two regions defined by the air/dielectric interface, with a charge discontinuity on the strip, we will have separate expressions for $\Phi(x, y)$ in these regions. Solving (3.200) by the method of separation of variables and applying the boundary conditions of (3.201a,b) gives the general solutions as

$$\Phi(x,y) = \begin{cases}
\sum_{\substack{n=1\\\text{odd}}}^{\infty} A_n \cos \frac{n\pi x}{a} \sinh \frac{n\pi y}{a} & \text{for } 0 \le y \le d \\
\sum_{\substack{n=1\\\text{odd}}}^{\infty} B_n \cos \frac{n\pi x}{a} e^{-n\pi y/a} & \text{for } d \le y < \infty.
\end{cases}$$
3.202

Now the potential must be continuous at y = d, so from (3.202) we have that

$$A_n \sinh \frac{n\pi x}{a} = B_n e^{-n\pi d/a}.$$
 3.203



FIGURE 3.27 Geometry of a microstrip line with conducting sidewalls.
so $\Phi(x, y)$ can be written as

$$\Phi(x,y) = \begin{cases} \sum_{\substack{n=1\\\text{odd}}}^{\infty} A_n \cos \frac{n\pi x}{a} \sinh \frac{n\pi y}{a} & \text{for } 0 \le y \le d \\ \sum_{\substack{n=1\\\text{odd}}}^{\infty} A_n \cos \frac{n\pi x}{a} \sinh \frac{n\pi d}{a} e^{-n\pi(y-d)/a} & \text{for } d \le y < \infty. \end{cases}$$
3.204

The remaining constants, A_n , can be found by considering the surface charge density on the strip. We first find $E_y = -\partial \Phi / \partial y$:

$$E_{y} = \begin{cases} -\sum_{\substack{n=1\\\text{odd}}}^{\infty} A_{n}\left(\frac{n\pi}{a}\right)\cos\frac{n\pi x}{a}\cosh\frac{n\pi y}{a} & \text{for } 0 \le y < d\\ \sum_{\substack{n=1\\\text{odd}}}^{\infty} A_{n}\left(\frac{n\pi}{a}\right)\cos\frac{n\pi x}{a}\sinh\frac{n\pi d}{a}e^{-n\pi(y-d)/a} & \text{for } d \le y < \infty. \end{cases}$$
3.205

Then the surface charge density on the strip at y = d is

$$\rho_s = D_y(x, y = d^+) - D_y(x, y = d^+)$$

$$= \epsilon_0 E_y(x, y = d^+) - \epsilon_0 \epsilon_\tau E_y(x, y = d^-)$$

$$= \epsilon_0 \sum_{\substack{n=1 \ \text{odd}}}^{\infty} A_n\left(\frac{n\pi}{a}\right) \cos\frac{n\pi x}{a} \left[\sinh\frac{n\pi d}{a} + \epsilon_\tau \cosh\frac{n\pi d}{a}\right], \qquad 3.206$$

which is seen to be a Fourier series in x for the surface charge density, ρ_s . As for the stripline case, we can approximate the charge density on the microstrip line by a uniform distribution:

$$\rho_s(x) = \begin{cases} 1 & \text{for } |x| < W/2 \\ 0 & \text{for } |x| > W/2. \end{cases}$$
 3.207

Equating (3.207) to (3.206) and using the orthogonality of the $\cos n\pi x/a$ functions gives the constants A_n as

$$A_n = \frac{4a \sin n\pi W/2a}{(n\pi)^2 \epsilon_0 [\sinh(n\pi d/a) + \epsilon_r \cosh(n\pi d/a)]}.$$
 3.208

The voltage of the strip relative to the ground plane is

$$V = -\int_{0}^{d} E_{y}(x=0,y) \, dy = \sum_{\substack{n=1 \ \text{odd}}}^{\infty} A_{n} \, \sinh \, \frac{n\pi d}{\alpha}.$$
 3.209

The total charge, per unit length, on the center strip is

$$Q = \int_{-W/2}^{W/2} \rho_s(x) \, dx = W \, \mathrm{C/m}.$$
 3.210

so the static capacitance per unit length of the microstrip line is

$$C = \frac{Q}{V} = \frac{1}{\sum_{\substack{n=1 \ \text{odd}}}^{\infty} \frac{4a \sin(n\pi W/2a) \sinh(n\pi d/a)}{(n\pi)^2 W \epsilon_0 [\sinh(n\pi d/a) + \epsilon_r \cosh(n\pi d/a)}}.$$
 3.211

Now to find the effective dielectric constant, we consider two cases of capacitance;

- Let C = capacitance per unit length of the microstrip line with a dielectric substrate $(\epsilon_{\tau} \neq 1)$
- Let $C_o = \text{capacitance per unit length of the microstrip line with an air dielectric } (\epsilon_r = 1)$

Since capacitance is proportional to the dielectric constant of the material homogeneously filling the region around the conductors, we have that

$$\epsilon_e = \frac{C}{C_o}.$$
 3.212

So (3.212) can be evaluated by computing (3.211) twice; once with ϵ_r equal to the dielectric constant of the substrate (for C), and then with $\epsilon_r = 1$ (for C_o). The characteristic impedance is then

$$Z_0 = \frac{1}{v_p C} = \frac{\sqrt{\epsilon_c}}{cC},$$
 3.213

where $c = 3 \times 10^8$ m/sec.



EXAMPLE 3.8 Numerical Calculation of Microstrip Parameters

Evaluate the above expressions for a microstrip line on a substrate with $\epsilon_{\rm p} = 2.55$. Calculate the effective dielectric constant and characteristic impedance for W/d = 0.5 to 10.0, and compare with the results from (3.195) and (3.196). Let a = 100d.

Solution

A computer program was written in BASIC to evaluate (3.211) for $\epsilon = \epsilon_0$ and then $\epsilon = \epsilon_r \epsilon_0$. Then (3.212) was used to evaluate the effective dielectric constant, ϵ_e , and (3.213) to evaluate the characteristic impedance, Z_0 . The series was truncated after 50 terms, and the results are shown in the following table.

	Numerical Solutions		Formulas		
W/d	 Fr	$Z_0(\Omega)$	۴,	$Z_0(\Omega)$	
0.5	1.977	100.9	1.938		
1.0	1.989	94.9	1.990	89.8	
2.0	2.036	75.8	2.068	62.2	
4,0	2.179	45.0	2.163	39.3	
7.0	2.287	29.5	2.245	25.6	
10.0	2.351	21.7	2.198	19.1	

The comparison is reasonably good, although better results could be \mathfrak{ob} -tained from the approximate numerical solution by using a better estimate of the charge density on the strip.

3.9 THE TRANSVERSE RESONANCE TECHNIQUE

According to the general solutions to Maxwell's equations for TE or TM waves given in Section 3.1. a uniform waveguide structure always has a propagation constant of the form

$$\mathcal{J} = \sqrt{k^2 - k_r^2} = \sqrt{k^2 - k_r^2 - k_y^2}.$$
 3.214

where $k_c = \sqrt{k_x^2 + k_y^2}$ is the cutoff wavenumber of the guide and, for a given mode, is a fixed function of the cross-sectional geometry of the guide. Thus, if we know k_c we can determine the propagation constant of the guide. In previous sections we determined k_c by solving the wave equation in the guide, subject to the appropriate boundary conditions; this technique is very powerful and general, but can be complicated for complex waveguides, especially if dielectric layers are present. In addition, the wave equation solution gives a complete field description inside the waveguide, which is much more information than we really need If we are only interested in the propagation constant of the guide. The transverse resonance technique employs a transmission line model of the transverse cross section of the waveguide, and gives a much simpler and more direct solution for the cutoff frequency. This is another example where circuit and transmission line theory can be used to simplify the field theory solution.

The transverse resonance procedure is based on the fact that in a waveguide at cutoff, the fields form standing waves in the transverse plane of the guide, as can be inferred from the "bouncing plane wave" interpretation of waveguide modes discussed in Section 3.2. This situation can be modeled with an equivalent transmission line circuit operating at resonance. One of the conditions of such a resonant line is the fact that, at any point on the line, the sum of the input impedances seen looking to either side must be zero. That is,

$$Z_{in}^{r}(x) + Z_{in}^{\ell}(x) = 0,$$
 for all x . 3.215

where $Z_{in}^{r}(x)$ and $Z_{in}^{f}(x)$ are the input impedances seen looking to the right and left, respectively, at the point x on the resonant line.

The transverse resonance technique only gives results for the cutoff frequency of the guide. If fields or attenuation due to conductor loss are needed, the complete field theory solution will be required. The procedure will now be illustrated with several examples.

TM Modes for the Parallel Plate Waveguide

We will initially demonstrate the transverse resonance technique by re-solving the problem of Section 3.2 for the TM modes of the parallel plate waveguide. The geometry is shown in Figure 3.28. At cutoff, $k = k_c$, and there is no propagation down the guide in the z direction ($\beta = 0$). The fields thus form a standing wave along the y dimension of the guide. The equivalent circuit is a transmission line of length d (the height of the guide), shorted at both ends (representing the parallel plates at y = 0, d), as shown in Figure 3.28. The propagation constant for this line is k_y , and is to be determined. Because of the uniformity in the x direction, $k_x = 0$, so the cutoff wavenumber will be given by $k_c = k_y$. The characteristic impedance of the equivalent transmission line is taken as the wave impedance seen by a TM wave (3.26), with propagation constant k_y :

$$Z_0 = Z_{\rm TM} = \eta k_y / k, \qquad 3.216$$

where $\eta = \sqrt{\mu/\epsilon}$ is the intrinsic impedance of the material filling the guide, and $k = \omega\sqrt{\mu\epsilon}$ is the wavenumber.

At any point, $0 \le y \le d$, along the line we have

$$Z_{in}^{r}(y) = j Z_{\text{TM}} \tan k_y (d-y),$$
 3.217*a*

$$Z_{\rm in}^{\ell}(y) = j Z_{\rm TM} \tan k_y y. \qquad 3.217b$$

Using these results in (3.215) gives the condition for transverse resonance as

$$jZ_{\text{TM}}[\tan k_y(d-y) + \tan k_y y] = 0,$$

$$jZ_{\text{TM}}\frac{\sin k_y d}{\cos k_y(d-y)\cos k_y y} = 0.$$







This determines the cutoff wavenumbers as

$$k_c = k_y = \frac{n\pi}{d}$$
, for $n = 0, 1, 2, ..., 3.218$

The propagation constant is then found from (3.214). This is the same result as obtained in Section 3.2. For TE modes, we change the characteristic impedance of the line to $Z_{\text{TE}} = \eta k/k_y$, but the same cutoff wavenumber is obtained.

The above procedure can be simplified by noting that condition (3.215) must be valid for any value of x (or y), so we can select a certain point along the transmission line to simplify the evaluation of Z_{in}^r or Z_{in}^f . For example, in the present case we could choose y = 0; then $Z_{in}^r(0) = 0$ and $Z_{in}^r(0) = jZ_{TM} \tan k_y d_y$, which yields $k_y = n\pi/d$ more directly.

TEon Modes of a Partially Loaded Rectangular Waveguide

The transverse resonance technique is particularly useful when the guide contains dielectric layers because the boundary conditions at the dielectric interfaces, which require the solution of simultaneous algebraic equations in the field theory approach, can be easily handled as junctions of different transmission lines. As an example, consider the rectangular waveguide partially filled with dielectric, as shown in Figure 3.29. To find the cutoff frequencies for the TE_{tin} modes, the equivalent transverse resonance circuit shown in the figure can be used. The line for 0 < y < t represents the dielectric-filled part of the guide, and has a transverse propagation constant k_{yd} and a characteristic impedance for TE modes given by

$$Z_d = \frac{k\eta}{k_{yd}} = \frac{k_0\eta_0}{k_{yd}},$$
 3.219a

where $k_0 = \omega \sqrt{\mu_0 \epsilon_0}$, $\eta_0 = \sqrt{\mu_0 / \epsilon_0}$. For t < y < b, the guide is air filled and has a transverse propagation constant k_{η_0} and an equivalent characteristic impedance given by

$$Z_{\rm ct} = \frac{k_0 \eta_0}{k_{ya}}.$$
 3.219b





A rectangular waveguide partially filled with dielectric and the transverse resonance equivalent circuit.

Applying condition (3.215) yields

$$k_{ya} \tan k_{yd} t + k_{yd} \tan k_{ya} (b-t) = 0. \qquad 3.220$$

This equation contains two unknowns, k_{yn} and k_{yd} . An additional equation is obtained from the fact that the propagation constant, β , must be the same in both regions, for phase matching of the tangential fields at the dielectric interface. Thus, with $k_x = 0$,

$$\beta = \sqrt{\epsilon_r k_0^2 - k_{yd}^2} = \sqrt{k_0^2 - k_{yd}^2},$$

$$\epsilon_r k_0^2 - k_{yd}^2 = k_0^2 - k_{ya}^2.$$
 3.221

Equations (3.220) and (3.221) can then be solved (numerically or graphically) to obtain k_{yd} and k_{ya} . There will be an infinite number of solutions, corresponding to the *n* dependence (number of variations in *y*) of the TE_{on} mode.

3.10 WAVE VELOCITIES AND DISPERSION

So far, we have encountered two types of velocities related to the propagation of electromagnetic waves:

- The speed of light in a medium $(1/\sqrt{\mu\epsilon})$
- The phase velocity $(v_p = \omega/\beta)$

The speed of light in a medium is the velocity at which a plane wave would propagate in that medium, while the phase velocity is the speed at which a constant phase point travels. For a TEM plane wave, these two velocities are identical, but for other types of guided wave propagation the phase velocity may be greater or less than the speed of light.

If the phase velocity and attenuation of a line or guide are constants that do not change with frequency, then the phase of a signal that contains more than one frequency component will not be distorted. If the phase velocity is different for different frequencies, then the individual frequency components will not maintain their original phase relationships as they propagate down the transmission line or waveguide, and signal distortion will occur. Such an effect is called *dispersion*, since different phase velocities allow the "faster" waves to lead in phase relative to the "slower" waves, and the original phase relationships will gradually be dispersed as the signal propagates down the line. In such a case, there is no single phase velocity that can be attributed to the signal as a whole. However, if the bandwidth of the signal is relatively small, or if the dispersion is not too severe. a *group velocity* can be defined in a meaningful way. This velocity then can be used to describe the speed at which the signal propagates.

Group Velocity

As discussed above, the physical interpretation of group velocity is the velocity at which a narrow band signal propagates. We will derive the relation of group velocity to the propagation constant by considering a signal f(t) in the time domain. The Fourier

or

transform of this signal is defined as

$$F(\omega) = \int_{-\infty}^{\infty} f(t) e^{-j\omega t} dt, \qquad 3.222a$$

and the inverse transform is then

$$f(t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} F(\omega) e^{j\omega t} d\omega.$$
 3.222b

Now consider the transmission line or waveguide on which the signal f(t) is propagating as a linear system, with a transfer function $Z(\omega)$ that relates the output. $F_{\omega}(\omega)$, of the line to the input, $F(\omega)$, of the line, as shown in Figure 3.30. Thus,

$$F_{\alpha}(\omega) = Z(\omega)F(\omega). \qquad \qquad 3.223$$

For a lossless, matched transmission line or waveguide, the transfer function $Z(\omega)$ can be expressed as

$$Z(\omega) = Ae^{-j\omega t_z} = |Z(\omega)|e^{-j\psi}, \qquad 3.224$$

where A is a constant and B is the propagation constant of the line or guide. The time-domain representation of the output signal, $f_{\alpha}(t)$, can then be written as

$$f_o(t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} F(\omega) |Z(\omega)| e^{j(\omega t - \omega)} d\omega. \qquad (3.225)$$

Now if $|Z(\omega)| = A$ is a constant, and the phase ψ of $Z(\omega)$ is a linear function of ω , say $\psi = a\omega$, then the output can be expressed as

$$f_{\theta}(t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} AF(\omega) e^{j\omega(t-\alpha)} d\omega = Af(t-\alpha), \qquad 3.226$$

which is seen to be a replica of f(t), except for an amplitude factor, A, and time shift, a. Thus, a transfer function of the form $Z(\omega) = Ae^{-j\omega a}$ does not distort the input signal. A lossless TEM wave has a propagation constant $\beta = \omega/c$, which is of this form, so a TEM line is dispersionless, and does not lead to signal distortion. If the TEM line is lossy, however, the attenuation may be a function of frequency, which could lead to signal distortion.

Now consider a narrowband input signal of the form

$$s(t) = f(t)\cos\omega_{\alpha}t = \operatorname{Re}\left\{f(t)e^{j\omega_{\alpha}t}\right\},\qquad 3.227$$



FIGURE 3.30

A transmission line or waveguide represented as a linear system with transfer function $Z(\omega)$.

which represents an amplitude modulated carrier wave of frequency ω_o . Assume that the highest frequency component of f(t) is ω_m , where $\omega_m \ll \omega_o$. The Fourier transform, $S(\omega)$, of s(t), is

$$S(\omega) = \int_{-\infty}^{\infty} f(t)e^{-j\omega_o t}e^{j\omega t}dt = F(\omega - \omega_o), \qquad 3.228$$

where we have used the complex form of the input signal as expressed in (3.227). We will then need to take the real part of the output inverse transform to obtain the time-domain output signal. The spectrums of $F(\omega)$ and $S(\omega)$ are depicted in Figure 3.31.

The output signal spectrum is

$$S_o(\omega) = AF(\omega - \omega_o)e^{-j\beta z}, \qquad 3.229$$

and in the time domain,

$$s_{o}(t) = \frac{1}{2\pi} \operatorname{Re} \int_{-\infty}^{\infty} S_{o}(\omega) e^{j\omega t} d\omega$$

= $\frac{1}{2\pi} \operatorname{Re} \int_{\omega_{o}-\omega_{m}}^{\omega_{o}+\omega_{m}} AF(\omega-\omega_{o}) e^{j(\omega t-\beta_{2})} d\omega.$ 3.230

In general, the propagation constant β may be a complicated function of ω . But if $F(\omega)$ is narrow band ($\omega_m \ll \omega_o$), then β can be linearized by using a Taylor series expansion about ω_o :

$$\beta(\omega) = \beta(\omega_o) + \frac{d\beta}{d\omega} \bigg|_{\omega = \omega_o} (\omega - \omega_o) + \frac{1}{2} \frac{d^2\beta}{d\omega^2} \bigg|_{\omega = \omega_o} (\omega - \omega_o)^2 + \cdots$$
 3.231

Retaining the first two terms gives



FIGURE 3.31 Fourier spectrums of the signals (a) f(t) and (b) s(t).

where

Then after a change of variables to $y = \omega - \omega_o$, the expression for $s_o(t)$ becomes

$$s_{\theta}(t) = \frac{A}{2\pi} \operatorname{Re} \left\{ e^{f(\omega_{\theta}t - \beta_{\theta}z)} \int_{-\omega_{m}}^{\omega_{m}} F(y) e^{j(t - \beta_{\theta}'z)y} \, dy \right\}$$
$$= A \operatorname{Re} \left\{ f(t - \beta_{n}'z) e^{j(\omega_{\theta}t - \beta_{\theta}z)} \right\}$$
$$= A f(t - \beta_{\theta}'z) \cos(\omega_{\theta}t - \beta_{\theta}z). \qquad 3.233$$

which is a time-shifted replica of the original modulation envelope. f(t), of (3.227). The velocity of this envelope is the group velocity, v_g :

$$v_g = \frac{1}{\beta'_p} = \left(\frac{d\beta}{d\omega}\right)^{-1} \Big|_{\omega = \omega_p}.$$
 3.234



EXAMPLE 3.9 Waveguide Wave Velocities

Calculate the group velocity for a waveguide mode propagating in an air-filled guide. Compare this velocity to the phase velocity and speed of light.

Solution

The propagation constant for a mode in an air-filled waveguide is

$$\beta = \sqrt{k_0^2 - k_c^2} = \sqrt{(\omega/c)^2 - k_c^2}.$$

Taking the derivative with respect to frequency gives

$$\frac{d\mathcal{A}}{d\omega} = \frac{\omega/c^2}{\sqrt{(\omega/c)^2 - k_c^2}} = \frac{k_o}{c\vartheta}.$$

so from (3.234) the group velocity is

$$v_g = \left(\frac{d\beta}{d\omega}\right)^{-1} = \frac{r\beta}{k_0}.$$

The phase velocity is $v_p = \omega/\beta = (k_0 c)/\beta$.

Since $\beta < k_0$, we have that $v_g < c < v_p$, which indicates that the phase velocity of a waveguide mode may be greater than the speed of light, but the group velocity (the velocity of a narrowband signal) will be less than the speed of light.

3.11 SUMMARY OF TRANSMISSION LINES AND WAVEGUIDES

In this chapter we have discussed a variety of transmission lines and waveguides; here we will summarize some of the basic properties of these transmission media and their relative advantages in a broader context.

In the beginning of this chapter we made the distinction between TEM. TM. and TE waves and saw that transmission lines and waveguides can be categorized according to

Characteristic	Coax	Waveguide	Stripline	Microstrip
Modes: Preferred	TEM _	TE _{i0}	TEM	Quasi-TEM
Other	TM,TE	TM,TE	TM,TE	Hybrid TM,TE
Dispersion	None	Medium	None	Low
Bandwidth	High	Low	High	High
Loss	Medium	Low	High	High
Power capacity	Medium	High	Low	Low
Physical size	Large	Large	Medium	Small
Ease of fabrication	Medium	Medium	Easy	Easy
Integration with other components	Hard	Hard	Fair	Easy

TABLE 3.6 Comparison of Common Transmission Lines and Waveguides

which type of waves they can support. We have seen that TEM waves are nondispersive, with no cutoff frequency, whereas TM and TE waves exhibit dispersion and generally have nonzero cutoff frequencies. Other electrical considerations include bandwidth, attenuation, and power handling capacity. Mechanical factors are also very important, however, and include such considerations as physical size (volume and weight), ease of fabrication (cost), and the ability to be integrated with other devices (active or passive). Table 3.6 compares several types of transmission media with regard to the above considerations: this table only gives general guidelines, as specific cases may give better or worse results than those indicated.

Other Types of Lines and Guides

While we have discussed the most common types of waveguides and transmission lines, there are many other guides and lines (and variations) that we have not discussed. A few of the more popular types are briefly mentioned here.

Ridge waveguide. The bandwidth of a rectangular waveguide is, for practical purposes, less than an octave (a 2:1 frequency range). This is because the TE_{20} mode begins to propagate at a frequency equal to twice the cutoff frequency of the TE_{10} mode. The ridge waveguide, shown in Figure 3.32, consists of a rectangular waveguide loaded with







FIGURE 3.33 Dielectric waveguide geometry.

conducting ridges on the top and/or bottom walls. This loading tends to lower the cutoff frequency of the dominant mode, leading to increased bandwidth and better impedance characteristics. Such a guide is often used for impedance matching purposes, where the ridge may be tapered along the length of the guide. The presence of the ridge, however, reduces the power-handling capacity of the waveguide.

Dielectric waveguide. As we have seen from our study of surface waves, metallic conductors are not necessary to confine and support a propagating electromagnetic field. The dielectric waveguide shown in Figure 3.33 is another example of such a guide, where ϵ_{r2} , the dielectric constant of the ridge, is usually greater than ϵ_{r1} , the dielectric constant of the substrate. The fields are thus mostly confined to the area around the dielectric ridge. This type of guide supports TM and TE modes, and is convenient for integration with active devices. Its small size makes it useful for millimeter wave to optical frequencies, although it can be very lossy at bends or junctions in the ridge line. Many variations in this basic geometry are possible.

Slotline. Of the many types of planar lines that have been proposed, slotline probably ranks next, behind microstrip and stripline, in terms of popularity. The geometry of a slotline is shown in Figure 3.34. It consists of a thin slot in the ground plane on one side of a dielectric substrate. Thus, like microstrip, the two conductors of slotline lead to a quasi-TEM type of mode. Changing the width of the slot changes the characteristic impedance of the line.

Coplanar waveguide. A structure similar to slotline is coplanar waveguide, shown in Figure 3.35. Coplanar waveguide can be thought of as a slotline with a third conductor centered in the slot region. Because of the presence of this additional conductor, this type of line can support even or odd quasi-TEM modes, depending on whether the \bar{E} -fields in the two slots are in the opposite direction, or the same direction. Coplanar waveguide



FIGURE 3.34 Geometry of a printed slotline.



FIGURE 3.35 Coplanar waveguide geometry.

is particularly useful for fabricating active circuitry, due to the presence of the center conductor and the close proximity of the ground planes.

Covered microstrip. Many variations of the basic microstrip geometry are possible, but one of the more common is covered microstrip, shown in Figure 3.36. The metallic cover plate is often used for electrical shielding and physical protection of the microstrip circuit and is usually situated several substrate thicknesses away from the circuit. Its presence can, however, perturb the operation of the circuit enough so that its effect must be taken into account during design.

POINT OF INTEREST: Power Capacity of Transmission Lines

The power handling capacity of an air-filled transmission line or waveguide is limited by voltage breakdown, which occurs at a field strength of about $E_d = 3 \times 10^6$ V/m for room temperature air at sea level pressure.

In an air-filled coaxial line, the electric field varies as $E_{\rho} = V_{o}/(\rho \ln b/a)$, which has a maximum at $\rho = a_{\tau}$. Thus the maximum voltage before breakdown is

$$V_{\text{max}} = E_d a \ln \frac{b}{a},$$
 (peak-to-peak),

and the maximum power capacity is then

$$P_{\max} = rac{V_{\max}^2}{2Z_0} = rac{\pi a^2 E_d^2}{\eta_0} \ln rac{b}{a}.$$

As might be expected, this result shows that power capacity can be increased by using a larger coaxial cable (larger a, b with fixed b/a for the same characteristic impedance). But propagation of higher order modes limits the maximum operating frequency for a given cable size. Thus, there is an upper limit on the power capacity of a coaxial line for a given maximum operating frequency.





 f_{max} , which can be shown to be given by

$$P_{\rm max} = \frac{0.025}{\eta_0} \left(\frac{cE_d}{f_{\rm max}}\right)^2 = 5.8 \times 10^{12} \left(\frac{E_d}{f_{\rm max}}\right)^2.$$

As an example, at 10 GHz the maximum peak power capacity of any coaxial line with no higher order modes is about 520 kW.

In an air-filled rectangular waveguide, the electric field varies as $E_y = E_0 \sin(\pi x/a)$, which has a maximum value of E_0 at $x = a_f 2$. Thus the maximum power capacity before breakdown is

$$P_{\rm run} = \frac{abE_0^2}{4Z_w} = \frac{abE_0^2}{4Z_w}$$

which shows that power capacity increases with guide size. For most waveguides, $b \simeq 2a$. To avoid propagation of the TE₂₀ mode, we must have $a < c_c/f_{max}$, where f_{max} is the maximum operating frequency. Then the maximum power capacity of the guide can be shown to be

$$P_{\max} = \frac{0.11}{\eta_0} \left(\frac{cE_d}{f_{\max}}\right)^2 = 2.6 \times 10^{14} \left(\frac{E_d}{f_{\max}}\right)^2.$$

As an example, at 10 GHz the maximum peak power capacity of a rectangular waveguide operating in the TE₁₀ mode is about 2300 kW, which is considerably higher than the power capacity of a coaxial cable at the same frequency.

Because arcing and voltage breakdown are very high-speed effects, the above voltage and power limits are peak quantities. In addition, it is good engineering practice to provide a safety factor of at least two, so the maximum powers which can be safely transmitted should be limited to about half of the above values. If there are reflections on the line or guide, the power capacity is further reduced. In the worst case, a reflection coefficient magnitude of unity will double the maximum voltage on the line, so the power capacity will be reduced by a factor of four.

The power capacity of a line can be increased by pressurizing the line with air or an inert gas, or by using a dielectric. The dielectric strength (E_d) of most dielectrics is greater than that of air, but the power capacity may be primarily limited by the heating of the dielectric due to ohmic loss.

Reference: P. A. Rizzi, Microwave Engineering-Passive Circuits, Prentice-Hall, New Jersey, 1988.

REFERENCES

- [1] O. Heaviside, *Electromagnetic Theory*, vol. 1, 1893. Reprinted by Dover, New York, 1950.
- [2] Lord Rayleigh. "On the Passage of Electric Waves Through Tubes," *Philos. Mag.*, vol. 43, pp. 125–132, 1897. Reprinted in *Collected Papers*. Cambridge Univ. Press, 1903.
- [3] K. S. Packard, "The Origin of Waveguides: A Case of Multiple Rediscovery," *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-32, pp. 961–969. September 1984.
- [4] R. M. Barrett, "Microwave Printed Circuits—An Historical Perspective," IEEE Trans. Microwave Theory and Techniques, vol. MTT-32, pp. 983–990, September 1984.
- [5] D. D. Grieg and H. F. Englemann, "Microstrip—A New Transmission Technique for the Kilomegacycle Range," Proc. IRE, vol. 40, pp. 1644–1650, December 1952.

- [6] H. Howe, Jr., Stripline Circuit Design. Artech House, Dedham, Mass., 1974.
- [7] I. J. Bahl and R. Garg, "A Designer's Guide to Stripline Circuits," *Microwaves*, January 1978, pp. 90–96.
- [8] I. J. Bahl and D. K. Trivedi, "A Designer's Guide to Microstrip Line." Microwaves, May 1977, pp. 174–182.
- [9] K. C. Gopta, R. Garg, and I. J. Bahl, Microstrip Lines and Stotlines, Artech House, Dedham, Mass., 1979.

PROBLEMS

- 3.1 Derive equations (3.5a)-(3.5d) from equations (3.3) and (3.4).
- 3.2 Calculate the attenuation due to conductor loss for the TE_n mode of a parallel plate waveguide.
- **3.3** Consider a section of K-band waveguide. From the dimensions given in Appendix I, determine the cutoff frequencies of the first two propagating modes. From the recommended operating range given in Appendix I for this guide, determine the percentage reduction in bandwidth that this operating range represents, relative to the theoretical bandwidth for a single propagating mode.
- 3.4 Compute the TE₁₀ mode attenuation, in dB/m, for a length of K-band waveguide operating at f = 20 GHz. The waveguide is made from brass, and is filled with a dielectric material having $\epsilon_r = 2.6$ and tan $\delta = 0.001$.
- **3.5** An attenuator can be made using a section of waveguide operating below cutoff, as shown below. If a = 2.286 cm and the operating frequency is 12 GHz, determine the required length of the below-cutoff section of waveguide to achieve an attenuation of 100 dB between the input and output guides. Ignore the effect of reflections at the step discontinuities.



- **3.6** Find expressions for the electric surface current density on the walls of a rectangular waveguide for a TE₁₀ mode. Why can a narrow slot be cut along the centerline of the broad wall of a rectangular waveguide without perturbing the operation of the guide? (Such a slot is often used in a slotted line for a probe to sample the standing wave field inside the guide.)
- 3.7 Derive the expression for the attenuation of the TM_{mn} mode of a rectangular waveguide, due to imperfectly conducting walls.
- 3.8 For the partially loaded rectangular waveguide shown on the next page, solve (3.109) with $\beta = 0$ to find the cutoff frequency of the TE₁₀ mode. Assume a = 2.286 cm, t = a/2, and $\epsilon_r = 2.25$.



3.9 Consider the partially filled parallel plate waveguide shown below. Derive the solution (fields and cutoff frequency) for the lowest order TE mode of this structure. Assume the metal plates are infinitely wide. Can a TEM wave propagate on this structure?



3.10 Consider the partially filled parallel plate waveguide shown below. Derive the solution (fields and cutoff frequency) for the TE modes. Can a TEM wave exist in this structure? Ignore fring fields at the sides, and assume no x dependence.



- **3.11** Derive equations (3.110a-d) for the transverse field components in terms of longitudinal field^{5, in} cylindrical subtributes.
- **3.12** Derive the expression for the attenuation of the TM_{nm} mode in a circular waveguide with finite conductivity.
- **3.13** Consider a circular waveguide with a = 0.8 cm. Compute the cutoff frequencies and identify the first four propagating modes.
- **3.14** Derive the \hat{E} and \bar{H} fields of a coaxial line from the expression for the potential given in (3.153). Also find expressions for the voltage and current on the line and the characteristic impedance.
- **3.15** Derive a transcendental equation for the cutoff frequency of the TM modes of a coaxial waveguide. Using table 5, obtain an approximate value of k_{cu} for the TM₀₁ mode, if b/u = 2.
- **3.16** Derive an expression for the attenuation of a TE surface wave on a grounded dielectric slab, when the ground plane has finite conductivity.
- 3.17 Consider the geometry shown on the next page. Derive a solution for the TM surface waves that can propagate on this structure.



3.18 Consider the partially filled coaxial line shown below. Can a TEM wave propagate on this line? Derive the solution for the $TM_{\alpha\alpha}$ (no azimuthal variation) modes of this geometry.



- **3.19** Design a stripline transmission line for a 100 Ω characteristic impedance. The ground plane separation is 0.316 cm, and the dielectric constant of the filling material is 2.20. What is the guide wavelength on this transmission line if the frequency is 4.0 GHz?
- **3.20** Design a microstrip transmission line for a 100 Ω characteristic impedance. The substrate thickness is 0.158 cm, with $\epsilon_r = 2.20$. What is the guide wavelength on this transmission line if the frequency is 4.0 GHz?
- **3.21** A microwave antenna feed network operating at 5 GHz requires a 50 Ω printed transmission line that is 16 λ long. Possible choices are (1) copper microstrip, with d = 0.16 cm. $\epsilon_r = 2.20$, and tan b = 0.001. or (2) copper stripline, with b = 0.32 cm. $\epsilon_r = 2.20, t = 0.01$ mm, and tan $\delta = 0.001$. Which line should be used, if attenuation is to be minimized?
- **3.22** Consider the TE modes of an arbitrary uniform waveguiding structure, where the transverse fields are related to H_z as in (3.19). If H_z is of the form $H_z(x, y, z) = h_z(x, y)e^{-y\beta z}$, where $h_z(x, y)$ is a real function, compute the Poynting vector and show that real power flow occurs only in the z direction. Assume that β is real, corresponding to a propagating mode.
- **3.23** A piece of rectangular waveguide is air filled for z < 0 and dielectric filled for z > 0. Assume that both regions can support only the dominant TE_{10} mode, and that a TE_{10} mode is incident on the interface from z < 0. Using a field analysis, write general expressions for the transverse field components of the incident, reflected, and transmitted waves in the two regions, and enforce the boundary conditions at the dielectric interface to find the reflection and transmission coefficients. Compare these results to those obtained with an impedance approach, using Z_{TE} for each region.
- **3.24** Use the transverse resonance technique to derive a transcendental equation for the propagation constant of the TM modes of a rectangular waveguide that is air filled for 0 < x < d and dielectric filled for d < x < a.
- 3.25 Apply the transverse resonance technique to find the propagation constants for the TE surface waves that can be supported by the structure of Problem 3.17.

- **3.26** An X-band waveguide filled with Teflon is operating at 9.5 GHz. Calculate the speed of light in this material and the phase and group velocities in the waveguide.
- 3.27 As discussed in the Point of Interest on the power handling capacity of transmission lines, the maximum power capacity of a coaxial line is limited by voltage breakdown, and is given by

$$P_{\rm max} = \frac{\pi a^2 E_d^2}{\eta_0} \ln \frac{b}{a}$$

where E_d is the field strength at breakdown. Find the value of b/a that maximizes the maximum power capacity and show that the corresponding characteristic impedance is about 30 Ω .

Microwave Network Analysis

Circuits operating at low frequencies, for which the circuit dimensions are small relative to the wavelength, can be treated as an interconnection of lumped passive or active components with unique voltages and currents defined at any point in the circuit. In this situation the circuit dimensions are small enough so that there is negligible phase change from one point in the circuit to another. In addition, the fields can be considered as TEM fields supported by two or more conductors. This leads to a quasi-static type of solution to Maxwell's equations, and to the well-known Kirchhoff voltage and current laws and impedance concepts of circuit theory [1]. As the reader is aware, there exists a powerful and useful set of techniques for analyzing low-frequency circuits. In general, these techniques cannot be directly applied to microwave circuits. It is the purpose of the present chapter, however, to show how circuit and network concepts can be extended to handle many microwave analysis and design problems of practical interest.

The main reason for doing this is that it is usually much easier to apply the simple and intuitive ideas of circuit analysis to a microwave problem than it is to solve Maxwell's equations for the same problem. In a way, field analysis gives us much more information about the particular problem under consideration than we really want or need. That is, because the solution to Maxwell's equations for a given problem is complete, it gives the electric and magnetic fields at all points in space. But usually we are interested in only the voltage or current at a set of terminals, the power flow through a device, or some other type of "global" quantity, as opposed to a minute description of the response at all points in space. Another reason for using circuit or network analysis is that it is then very easy to modify the original problem, or combine several elements together and find the response, without having to analyze in detail the behavior of each element in combination with its neighbors. A field analysis using Maxwell's equations for such problems would be hopelessly difficult. There are situations, however, where such circuit analysis techniques are an oversimplification, leading to erroneous results. In such cases one must resort to a field analysis approach, using Maxwell's equations. It is part of the education of a microwave engineer to be able to determine when circuit analysis concepts apply, and when they should be cast aside.

The basic procedure for microwave network analysis is as follows. We first treat a set of basic, canonical problems rigorously, using field analysis and Maxwell's equations. (As we have done in Chapters 2 and 3, for a variety of transmission line and waveguide problems.) When so doing, we try to obtain quantities that can be directly related to a circuit or transmission line parameter. For example, when we treated various transmission lines and waveguides in Chapter 3 we derived the propagation constant and characteristic impedance of the line. This allowed the transmission line or waveguide to be treated as a distributed component characterized by its length, propagation constant, and characteristic impedance. At this point, we can interconnect various components and use network and/or transmission line theory to analyze the behavior of the entire system of components, including effects such as multiple reflections, loss, impedance transformations, and transitions from one type of transmission medium to another (e.g., coax to microstrip). As we will see, transitions between different transmission lines, or discontinuities on a transmission line, generally cannot be treated as a simple junction between two transmission lines, but must be augmented with some type of equivalent circuit to account for reactances associated with the transition or discontinuity.

Microwave network theory was originally developed in the service of radar system and component development at the MIT Radiation Lab in the 1940s. This work was continued and extended at the Microwave Research Institute, which was organized at the Polytechnic Institute of Brooklyn in 1942. The researchers there included E. Weber, N. Marcuvitz, A. A. Oliner, L. B. Felsen, A. Hessel, and others [2].

4.1 IMPEDANCE AND EQUIVALENT VOLTAGES AND CURRENTS Equivalent Voltages and Currents

At microwave frequencies the measurement of voltage or current is difficult (or impossible), unless a clearly defined terminal pair is available. Such a terminal pair may be present in the case of TEM-type lines (such as coaxial cable, microstrip, or stripline), but does not exist for non-TEM lines (such as rectangular, circular, or surface waveguides).

Figure 4.1 shows the electric and magnetic field lines for an arbitrary two-conductor TEM transmission line. As in Chapter 3, the voltage, V, of the + conductor relative to the - conductor can be found as

$$V = \int_{-}^{-} \vec{E} \cdot d\vec{\ell}, \qquad 4.1$$

where the integration path begins on the + conductor and ends on the - conductor, it is important to realize that, because of the electrostatic nature of the transverse fields between the two conductors, the voltage defined in (4.1) is unique and does not depend on the shape of the integration path. The total current flowing on the - conductor can be determined from an application of Ampere's law as

$$I = \oint_{C^+} \bar{H} \cdot d\bar{\ell}.$$
 4.2

where the integration contour is any closed path enclosing the + conductor (but not the - conductor). A characteristic impedance Z_0 can then be defined for traveling waves as

$$Z_0 = \frac{V}{I}.$$
 4.3



FIGURE 4.1 Electric and magnetic field lines for an arbitrary two-conductor TEM line.

At this point, after having defined and determined a voltage, current, and characteristic impedance (and assuming we know the propagation constant for the line), we can proceed to apply the circuit theory for transmission lines developed in Chapter 2 to characterize this line as a circuit element.

The situation is more difficult for waveguides. To see why, we will look at the case of a rectangular waveguide, as shown in Figure 4.2. For the dominant TE_{10} mode, the transverse fields can be written, from Table 3.2, as

$$E_{\boldsymbol{y}}(\boldsymbol{x},\boldsymbol{y},\boldsymbol{z}) = \frac{j\omega\mu a}{\pi}A\sin\frac{\pi x}{a}e^{-j\beta z} = Ae_{\boldsymbol{y}}(\boldsymbol{x},\boldsymbol{y})e^{-j\beta z}, \qquad 4.4a$$

$$H_x(x, y, z) = \frac{j\beta a}{\pi} A \sin \frac{\pi x}{a} e^{-j\beta z} = Ah_x(x, y)e^{-j\beta z}.$$
4.4b

Applying (4.1) to the electric field of (4.4a) gives

$$V = \frac{-j\omega\mu a}{\pi} A \sin\frac{\pi x}{a} e^{-j\beta z} \int_{y} dy.$$
 4.5



FIGURE 4.2 Electric field lines for the TE_{10} mode of a rectangular waveguide.

Thus it is seen that this voltage depends on the position. x, as well as the length of the integration contour along the y direction. Integrating from y = 0 to b for r = a/2 gives a voltage that is quite different from that obtained by integrating from y = 0 to b for r = 0, for example. What, then, is the correct voltage? The answer is that there is no "correct" voltage in the sense of being unique or pertinent for all applications. A similar problem arises with current, and also impedance. We will now show how we can define voltages, currents, and impedances that are useful for non-TEM lines.

There are many ways to define equivalent voltage, current, and impedance for waveguides, since these quantities are not unique for non-TEM lines, but the following considerations usually lead to the most useful results [1], [3], [4]:

- Voltage and current are defined only for a particular waveguide mode, and are defined so that the voltage is proportional to the transverse electric field, and the current is proportional to the transverse magnetic field.
- In order to be used in a manner similar to voltages and currents of circuit theory, the equivalent voltages and currents should be defined so that their product gives also power flow of the morge.
- The ratio of the voltage to the current for a single traveling wave should be equal to the characteristic impedance of the line. This impedance may be chosen arbitrarily, but is usually selected as equal to the wave impedance of the line, or else normalized to unity.

For an arbitrary waveguide mode with both positively and negatively traveling waves, the transverse fields can be written as

$$\bar{E}_t(x,y,z) = \bar{e}(x,y)(A^+e^{-j\beta z} + A^-e^{j\beta z}) = \frac{\bar{e}(x,y)}{C_1}(V^+e^{-j\beta z} + V^-e^{j\beta z}), \quad 4.6a$$

$$\bar{H}_{t}(x,y,z) = \bar{h}(x,y)(A^{+}e^{-j\beta z} - A^{-}e^{j\beta z}) = \frac{\bar{h}(x,y)}{C_{2}}(I^{+}e^{-j\beta z} - I^{+}e^{j\beta z}), \quad 4.6b$$

where \bar{e} and \bar{h} are the transverse field variations of the mode, and A^+ , A^- are the field amplitudes of the traveling waves. Since \bar{E}_t and \bar{H}_t are related by the wave impedance. Z_w , according to (3.22) or (3.26), we also have that

$$\bar{h}(x,y) = \frac{\hat{z} \times \bar{c}(x,y)}{Z_w},$$
4.7

Equation (4.6) also defines equivalent voltage and current waves as

$$V(z) = V^{+} e^{-j\beta z} + V^{-} e^{j\beta z}, \qquad 4.8a$$

$$I(z) = I^{+} e^{-j\beta \dot{z}} - I^{-} e^{j\beta \dot{z}}.$$
4.8b

with $V^+/I^- = V^-/I^- = Z_0$. This definition embodies the idea of making the equivalent voltage and current proportional to the transverse electric and magnetic fields, respectively. The proportionality constants for this relationship are $C_1 = V^+/A^+ = V^-/A^-$ and $C_2 = I^+/A^+ = I^-/A^-$, and can be determined from the remaining two conditions for power and impedance.

The complex power flow for the incident wave is given by

$$P^{+} = \frac{1}{2} |A^{+}|^{2} \iint_{\mathcal{S}} \bar{e} \times \bar{h}^{*} \cdot \hat{z} \, ds = \frac{V^{+} I^{+*}}{2C_{1}C_{2}^{*}} \iint_{\mathcal{S}} \bar{e} \times \bar{h}^{*} \cdot \hat{z} \, ds.$$

$$4.9$$

Since we want this power to be equal to $(1/2)V^+I^{+*}$, we have the result that

$$C_1 C_2^* = \iint_{\mathbf{S}} \bar{e} \times \bar{h}^* \cdot \hat{z} \, ds, \qquad 4.10$$

where the surface integration is over the cross section of the waveguide. The characteristic impedance is

$$Z_0 = \frac{V^+}{I^+} = \frac{V^-}{I^-} = \frac{C_1}{C_2},$$
4.11

since $V^+ = C_1 A$ and $I^+ = C_2 A$, from (4.6a,b). If it is desired to have $Z_0 = Z_w$, the wave impedance (Z_{TE} or Z_{TM}) of the mode, then

$$\frac{C_1}{C_2} = Z_w \ (Z_{\text{TE}} \text{ or } Z_{\text{TM}}). \tag{4.12a}$$

Alternatively, it may be desirable to normalize the characteristic impedance to unity $(Z_0 = 1)$, in which case we have

$$\frac{C_1}{C_2} = 1.$$
 4.12b

So for a given waveguide mode, (4.10) and (4.12) can be solved for the constants, C_1 and C_2 , and equivalent voltages and currents defined. Higher order modes can be treated in the same way, so that a general field in a waveguide can be expressed in the following form:

$$\bar{E}_{t}(x,y,z) = \sum_{n=1}^{N} \left(\frac{V_{n}^{+}}{C_{1n}} e^{-j\beta_{n}z} + \frac{V_{n}^{-}}{C_{1n}} e^{j\beta_{n}z} \right) \hat{e}_{n}(x,y), \qquad 4.13a$$

$$\bar{H}_t(x,y,z) = \sum_{n=1}^N \left(\frac{I_n^+}{C_{2n}} e^{-j\beta_n z} - \frac{I_n^-}{C_{2n}} e^{j\beta_n z} \right) \bar{h}_n(x,y), \qquad 4.13b$$

where V_n^{\pm} and I_n^{\pm} are the equivalent voltages and currents for the *n*th mode, and C_{1n} and C_{2n} are the proportionality constants for each mode.

EXAMPLE 4.1 Equivalent Voltage and Current for a Rectangular Waveguide

Find the equivalent voltages and currents for a TE_{10} mode in a rectangular waveguide.

Solution

The transverse field components and power flow of the TE_{10} rectangular waveguide mode and the equivalent transmission line model of this mode can be written as follows:

Waveguide Fields	Transmission Line Model		
$\overline{E_u} = (A^+ e^{-j\beta z} + A^- e^{j\beta z}) \sin(\pi x/a)$	$V(z) = V^+ e^{-j\beta z} + V^- e^{j\beta z}$		
$H_x = \frac{-1}{2\pi} (A^+ e^{-j\beta z} - A^- e^{j\beta z}) \sin(\pi x/a)$	$I(z) = I^+ e^{-j\beta z} - I^- e^{j\beta z}$		
	$= \frac{1}{Z_0} (V^+ e^{-j/3z} - V^- e^{j\beta z})$		
$P^+ = -(1/2)\int_s E_y H_x^* dx dy$	$P = \frac{1}{2} V^{\tau+} I^{+*}$		
$=rac{ab}{4Z_{TE}} \mathcal{A} ^2$			

We now find the constants C_1 and C_2 that relate the equivalent voltage V^+ and current I^- to the field amplitude, A. Equating incident powers gives

$$\frac{\mathfrak{gl}[\mathbf{A}]^2}{4Z_{\text{Tr}}} = \frac{1}{2}V^+ I^{-*} = \frac{1}{2}|\mathbf{A}|^2 C_1 C_2^*.$$

If we choose $Z_0 = Z_{\text{TE}}$, then we also have that

$$\frac{V^+}{I^+} = \frac{C_1}{C_2} = Z_{\rm RF}.$$

Solving for C_1, C_2 gives

$$C_1 = \sqrt{\frac{ab}{2}},$$
$$C_2 = \frac{1}{Z_{\text{TE}}} \sqrt{\frac{ab}{2}}.$$

which completes the transmission line equivalence for the TE₁₀ mode.

The Concept of Impedance

We have used the idea of impedance in several different applications, so it may be useful at this point to discuss the concept of impedance in more general terms. The term impedance was first used by Oliver Heaviside in the nineteenth century to describe the complex ratio V/I in AC circuits consisting of resistors, inductors, and capacitors: the impedance concept quickly became indispensable in the analysis of AC circuits. It was then applied to transmission lines, in terms of lumped-element equivalent circuits and the distributed series impedance and shunt admittance of the line. In the 1930s, Schelkunoff recognized that the impedance concept could be extended to electromagnetic fields in a systematic way, and noted that impedance should be regarded as characteristic of the type of field, as well as the medium [2]. And, in relation to the analogy between transmission lines and plane wave propagation, impedance may even be dependent on direction. The concept of impedance, then, forms an important link between field theory and transmission line or circuit theory.

Ο

Below we summarize the various types of impedance we have used so far and their notation:

- $\eta = \sqrt{\mu/\epsilon}$ = intrinsic impedance of the medium. This impedance is dependent only on the material parameters of the medium, but is equal to the wave impedance for plane waves.
- $Z_w = E_t/H_t = 1/Y_w$ = wave impedance. This impedance is a characteristic of the particular type of wave. TEM, TM, and TE waves each have different wave impedances ($Z_{\text{TEM}}, Z_{\text{TM}}, Z_{\text{TE}}$), which may depend on the type of line or guide, the material, and the operating frequency.
- $Z_0 = 1/Y_0 = \sqrt{L/C}$ = characteristic impedance. Characteristic impedance is the ratio of voltage to current for a traveling wave. Since voltage and current are uniquely defined for TEM waves, the characteristic impedance of a TEM wave is unique. TE and TM waves, however, do not have a uniquely defined voltage and current, so the characteristic impedance for such waves may be defined in various ways.

Ø

EXAMPLE 4.2 Application of Waveguide Impedance

Consider a rectangular waveguide with a = 3.485 cm and b = 1.580 cm (Cband guide), air filled for z < 0 and dielectric filled ($\epsilon_r = 2.56$) for z > 0, as shown in Figure 4.3. If the operating frequency is 4.5 GHz, use an equivalent transmission line model to compute the reflection coefficient of a TE₁₀ wave incident on the interface from z < 0.

Solution

The propagation constants in the air (z < 0) and the dielectric (z > 0) regions are

$$\beta_a = \sqrt{k_0^2 - \left(\frac{\pi}{a}\right)^2} = 27.50 \text{ m}^{-1},$$
$$\beta_d = \sqrt{\epsilon_r k_0^2 - \left(\frac{\pi}{a}\right)^2} = 120.89 \text{ m}^{-1}$$

where $k_0 = 94.25 \text{ m}^{-1}$.





The reader may verify that the TE_{10} mode is the only propagating mode in either waveguide region. Now we can set up an equivalent transmission line for the TE_{10} mode in each waveguide and treat the problem as the reflection of an incident voltage wave at the junction of two infinite transmission lines.

By Example 4.1 and Table 3.2, the equivalent characteristic impedances for the two lines are

$$Z_{0_d} = \frac{k_0 \eta_0}{\beta_a} = \frac{(94.25)(377.)}{27.50} = 1292.1 \ \Omega.$$
$$Z_{0_d} = \frac{k\eta}{\beta_d} - \frac{k_0 \eta_0}{\beta_d} = \frac{(94.25)(377.)}{120.89} = 293.9 \ \Omega.$$

The reflection coefficient seen looking into the dielectric filled region is then

$$\Gamma = \frac{Z_{0_d} - Z_{0_d}}{Z_{0_d} + Z_{0_d}} = -0.629.$$

With this result, expressions for the incident, reflected, and transmitted waves can be written in terms of fields, or in terms of equivalent voltages and currents.

We now consider the arbitrary one-port network shown in Figure 4.4, and derive a general relation between its impedance properties and electromagnetic energy stored in, and the power dissipated by, the network. The complex power delivered to this network is given by (1.91):

$$P = \frac{1}{2} \oint_{S} \bar{E} \times \bar{H}^* \cdot d\bar{s} = P_l + 2j\omega(W_m - W_c).$$

$$4.14$$

where P_{ℓ} is real and represents the average power dissipated by the network, and W_m and W_e represent the stored magnetic and electric energy, respectively. Note that the unit normal vector in Figure 4.4 is pointing into the volume.



FIGURE 4.4 An arbitrary one-port network.

If we define real transverse modal fields, \bar{e} and \bar{h} , over the terminal plane of the network such that

$$\bar{E}_t(x, y, z) = V(z)\bar{e}(x, y)e^{-j\beta z}, \qquad 4.15a$$

$$\hat{H}_t(x,y,z) = I(z)\bar{h}(x,y)e^{-j\beta z}, \qquad 4.15b$$

with a normalization such that

$$\int_{S} \bar{e} \times \bar{h} \cdot d\bar{s} = 1,$$

then (4.14) can be expressed in terms of the terminal voltage and current:

$$P = \frac{1}{2} \int_{S} V I^* \tilde{e} \times \tilde{h} \cdot d\bar{s} = \frac{1}{2} V I^*.$$

$$4.16$$

Then the input impedance is

$$Z_{is} = R + jX = \frac{V}{I} = \frac{VI^*}{|I|^2} = \frac{P}{\frac{1}{2}|I|^2} = \frac{P_{\ell} + 2j\omega(W_m - W_e)}{\frac{1}{2}|I|^2}.$$
 4.17

Thus we see that the real part, R, of the input impedance is related to the dissipated power, while the imaginary part, X, is related to the net energy stored in the network. If the network is lossless, then $P_{\ell} = 0$ and R = 0. Then $Z_{\rm un}$ is purely imaginary, with a reactance

$$X = \frac{4\omega(W_m - W_e)}{|I|^2},$$
 4.18

which is positive for an inductive load $(W_m > W_e)$, and negative for a capacitive load $(W_m < W_e)$.

Even and Odd Properties of $Z(\omega)$ and $\Gamma(\omega)$

Consider the driving point impedance. $Z(\omega)$, at the input port of an electrical network. The voltage and current at this port are related as $V(\omega) = Z(\omega)I(\omega)$. For an arbitrary frequency dependence, we can find the time-domain voltage by taking the inverse Fourier transform of $V(\omega)$:

$$v(t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} V(\omega) e^{j\omega t} d\omega. \qquad 4.19$$

Since v(t) must be real, we have that $v(t) = v^*(t)$, or

$$\int_{-\infty}^{\infty} V(\omega) e^{j\omega t} \, d\omega = \int_{-\infty}^{\infty} V^*(\omega) e^{-j\omega t} \, d\omega = \int_{-\infty}^{\infty} V^*(-\omega) e^{j\omega t} \, d\omega,$$

where the last term was obtained by a change of variable from ω to $-\omega$. This shows that $V(\omega)$ must satisfy the relation

$$V(-\omega) = V^*(\omega), \qquad 4.20$$

which means that $\operatorname{Re}\{V(\omega)\}\$ is even in ω , while $\operatorname{Im}\{V(\omega)\}\$ is odd in ω . Similar results hold for $I(\omega)$, and for $Z(\omega)$ since

$$V^*(-\omega)=Z^*(-\omega)I^*(-\omega)=Z^*(-\omega)I(\omega)=V(\omega)=Z(\omega)I(\omega),$$

Thus, if $Z(\omega) = R(\omega) + jX(\omega)$, then $R(\omega)$ is even in ω and $X(\omega)$ is odd in ω . These results can also be inferred from (4.17).

Now consider the reflection coefficient at the input port;

$$\Gamma(\omega) = \frac{Z(\omega)}{Z(\omega)} - \frac{Z_0}{Z_0} = \frac{R(\omega) - Z_0 + jX(\omega)}{R(\omega) + Z_0 + jX(\omega)}.$$
(4.2)

Then,

$$\Gamma(-\omega) = \frac{R(\omega) - Z_0 - jX(\omega)}{R(\omega) + Z_0 - jX(\omega)} = \Gamma^*(\omega), \qquad 4.22$$

which shows that the real and imaginary parts of $\Gamma(\omega)$ are even and odd, respectively, in ω . Finally, the magnitude of the reflection coefficient is

$$|\Gamma(\omega)|^{2} = \Gamma(\omega)\Gamma^{*}(\omega) = \Gamma(\omega)\Gamma(-\omega) = |\Gamma(-\omega)|^{2}, \qquad 4.23$$

which shows that $|\Gamma(\omega)|^2$ and $|\Gamma(\omega)|$ are even functions of ω . This result implies that only even series of the form $a + b\omega^2 + c\omega^4 + \cdots$ can be used to represent $|\Gamma(\omega)|$ or $|\Gamma(\omega)|^2$.

4.2 IMPEDANCE AND ADMITTANCE MATRICES

In the previous section we have seen how equivalent voltages and currents can be defined for TEM and non-TEM waves. Once such voltages and currents have been defined at various points in a microwave network, we can use the impedance and/or admittance matrices of circuit theory to relate these terminal or "port" quantities to each other, and thus to essentially arrive at a matrix description of the network. This type of representation lends itself to the development of equivalent circuits of arbitrary networks, which will be quite useful when we discuss the design of passive components such as couplers and filters.

We begin by considering an arbitrary N-port microwave network, as depicted in Figure 4.5. The ports in Figure 4.5 may be any type of transmission line or transmission line equivalent of a single propagating waveguide mode. (The term *port* was introduced by H. A. Wheeler in the 1950s to replace the less descriptive and more cumbersome phrase, "two-terminal pair" [3], [2].) If one of the physical ports of the network is a waveguide supporting more than one propagating mode, additional electrical ports can be added to account for these modes. At a specific point on the *n*th port, a terminal plane, t_n , is defined along with equivalent voltages and currents for the incident (V_n^+, I_n^+) and reflected (V_n^-, I_n) waves. The terminal planes are important in providing a phase teference for the voltage and current phasors. Now at the *n*th terminal plane, the total voltage and current is given by

$$I_n = I_n^+ - I_n^- \,. \tag{4.24b}$$

as seen from (4.8) when z = 0,



FIGURE 4.5 An arbitrary N-port microwave network.

The impedance matrix [Z] of the microwave network then relates these voltages and currents:

$$\begin{bmatrix} V_1 \\ V_2 \\ \vdots \\ V_N \end{bmatrix} = \begin{bmatrix} Z_{11} & Z_{12} & \cdots & Z_{1N} \\ Z_{21} & & \vdots \\ \vdots & & & \vdots \\ Z_{N1} & \cdots & \cdots & Z_{NN} \end{bmatrix} \begin{bmatrix} I_1 \\ I_2 \\ \vdots \\ I_N \end{bmatrix},$$

or in matrix form as

$$[V] = [Z][I].$$
 4.25

Similarly, we can define an admittance matrix [Y] as

$$\begin{bmatrix} I_1 \\ I_2 \\ \vdots \\ I_N \end{bmatrix} = \begin{bmatrix} Y_{11} & Y_{12} & \cdots & Y_{1N} \\ Y_{21} & & \vdots \\ \vdots & & & \vdots \\ Y_{N1} & \cdots & \cdots & Y_{NN} \end{bmatrix} \begin{bmatrix} V_1 \\ V_2 \\ \vdots \\ V_N \end{bmatrix}$$

or in matrix form as

$$[I] = [Y][V]. 4.26$$

Of course, the [Z] and [Y] matrices are the inverses of each other:

$$[Y] = [Z]^{-1}.$$
 4.27

Note that both the [Z] and [Y] matrices relate the total port voltages and currents.

From (4.25), we see that Z_{ij} can be found as

$$Z_{ij} = \frac{V_i}{I_j} \Big|_{I_k = 0 \text{ for } k \neq j}.$$
4.28

In words, (4.28) states that Z_{ij} can be found by driving port j with the current I_j , opencircuiting all other ports (so $I_k = 0$ for $k \neq j$), and measuring the open-circuit voltage at port i. Thus, Z_{ij} is the input impedance seen looking into port i when all other ports are open-circuited, and Z_{ij} is the transfer impedance between ports i and j when all other ports are open-circuited.

Similarly, from (4.26), Y_{ij} can be found as

$$Y_{ij} = \frac{I_i}{V_j} \bigg|_{V_k = \emptyset \text{ for } k \neq j},$$
4.29

which states that Y_{ij} can be determined by driving port j with the voltage V_j , shortcircuiting all other ports (so $V_k = 0$ for $k \neq j$), and measuring the short-circuit current at port i.

In general, each Z_{ij} or Y_{ij} element may be complex. For an N-port network, the impedance and admittance matrices are $N \times N$ in size, so there are $2N^2$ independent quantities or degrees of freedom for an arbitrary N-port network. In practice, however, many networks are either reciprocal or lossless, or both. If the network is reciprocal (not containing any nonreciprocal media such as ferrites or plasmas, or active devices), we will show that the impedance and admittance matrices are symmetric, so that $Z_{ij} = Z_{ji}$, and $Y_{ij} = Y_{ji}$. If the network is lossless, we can show that all the Z_{ij} or Y_{ij} elements are purely imaginary. Either of these special cases serve to reduce the number of independent quantities or degrees of freedom that an N-port network may have. We now derive the above characteristics for reciprocal and lossless networks.

Reciprocal Networks

Consider the arbitrary network of Figure 4.5 to be reciprocal (no active devices, ferrites, or plasmas), with short circuits placed at all terminal planes except those of ports 1 and 2. Now let \bar{E}_{a} , \bar{H}_{a} and \bar{E}_{b} , \bar{H}_{h} be the fields anywhere in the network due to two independent sources, a and b, located somewhere in the network. Then the reciprocity theorem of (1.156) states that

$$\oint_{S} \bar{E}_{a} \times \bar{H}_{b} \cdot d\bar{s} = \oint_{S} \bar{E}_{b} \times \bar{H}_{a} \cdot d\bar{s}, \qquad 4.30$$

where we will take S as the closed surface along the boundaries of the network and through the terminal planes of the ports. If the boundary walls of the network and transmission lines are metal, then $\bar{E}_{tan} = 0$ on these walls (assuming perfect conductors). If the network or the transmission lines are open structures, like microstrip or slotline, the boundaries of the network can be taken arbitrarily far from the lines so that \bar{E}_{tan} is negligible. Then the only nonzero contribution to the integrals of (4.30) come from the cross-sectional areas of ports 1 and 2.

Now from Section 4.1, the fields due to sources a and b can be evaluated at the terminal planes t_1 and t_2 as

$$\begin{split} \bar{E}_{1a} &= V_{1a}\bar{e}_1 \qquad H_{1a} = I_{1a}h_1 \\ \bar{E}_{1b} &= V_{1b}\bar{e}_1 \qquad \bar{H}_{1b} = I_{1b}\bar{h}_1 \\ \bar{E}_{2a} &= V_{2a}\bar{e}_2 \qquad \bar{H}_{2a} = I_{2a}\bar{h}_2 \\ \bar{E}_{2b} &= V_{2b}\bar{e}_2 \qquad \bar{H}_{2b} = I_{2b}\bar{h}_2, \end{split}$$

where \bar{e}_1 , \bar{h}_1 and \bar{e}_2 , \bar{h}_2 are the transverse modal fields of ports 1 and 2, respectively, and the Vs and Is are the equivalent total voltages and currents. (For instance, \bar{E}_{1b} is the transverse electric field at terminal plane t_1 of port 1 due to source b.) Substituting the fields of (4.31) into (4.30) gives

$$(V_{1a}I_{1b} - V_{1b}I_{1a}) \int_{S_{2}} \hat{e}_{1} \times \bar{h}_{1} \cdot d\bar{s} + (V_{2a}I_{2b} - V_{2b}I_{2a}) \int_{S_{2}} \bar{e}_{2} \times \bar{h}_{2} \cdot d\bar{s} = 0.$$
 4.32

where S_1, S_2 are the cross-sectional areas at the terminal planes of ports 1 and 2.

As in Section 4.1, the equivalent voltages and currents have been defined so that the power through a given port can be expressed as $VI^*/2$; then comparing (4.31) to (4.6) implies that $C_1 = C_2 = 1$ for each port, so that

$$\int_{S_1} \bar{e}_1 \times \bar{h}_1 \cdot d\bar{s} = \int_{S_2} \bar{e}_2 \times \bar{h}_2 \cdot d\bar{s} = 1.$$

$$4.33$$

This reduces (4.32) to

$$V_{1a}I_{1b} - V_{1b}I_{1a} + V_{2a}I_{2b} - V_{2b}I_{2a} = 0. 4.34$$

Now use the 2×2 admittance matrix of the (effectively) two-port network to eliminate the *Is*:

$$I_1 = Y_{11}V_1 + Y_{12}V_2,$$

 $I_2 = Y_{21}V_1 + Y_{22}V_2.$

Substitution into (4.34) gives

$$(V_{1a}V_{2b} - V_{1b}V_{2a})(Y_{12} - Y_{21}) = 0. 4.35$$

But since the sources a and b are independent, the voltages V_{1a} , V_{1b} , V_{2a} , and V_{2b} can take on arbitrary values. So in order for (4.35) to be satisfied for any choice of sources, we must have $Y_{12} = Y_{21}$, and since the choice of which ports are labeled as 1 and 2 is arbitrary, we have the general result that

$$Y_{ij} = Y_{jj}.$$
 4.36

Then if [Y] is a symmetric matrix, its inverse, [Z], is also symmetric.

Lossless Networks

Now consider a reciprocal lossless N-port junction; we will show that the elements of the impedance and admittance matrices must be pure imaginary. If the network is lossless, then the net real power delivered to the network must be zero. Thus, $\text{Re}\{P_{av}\} = 0$, where

$$P_{\mu\nu} = \frac{1}{2} [I^{\dagger}]' [I]^{*} + \frac{1}{2} ([Z]][I])' [I]^{*} = \frac{1}{2} [I^{\dagger}]' [Z][I]^{*}$$

$$= \frac{1}{2} (I_{1}Z_{11}I_{1}^{*} + I_{1}Z_{12}I_{2}^{*} + I_{2}Z_{21}I_{1}^{*} + \cdots)$$

$$= \frac{1}{2} \sum_{n=1}^{N} \sum_{m=1}^{N} I_{m}Z_{mn}I_{n}^{*}.$$
4.37

(We have used the result from matrix algebra that ([A][B])' = [B]'[A]'.) Since the I_n s are independent, we must have the real part of each self term $(I_n Z_{nn} I_n^*)$ equal to zero, since we could set all port currents equal to zero except for the *n*th current. So,

 $\operatorname{Re}\{I_n Z_{nn} I_n^*\} = |I_n|^2 \operatorname{Re}\{\{Z_{nn}\} = 0,$

 $\operatorname{Re}\{Z_{nn}\}=0.$

Now let all port currents be zero except for I_m and I_n . Then (4.37) reduces to

$$\operatorname{Re}\{(I_n I_m^* + I_m I_n^*)Z_{mn}\} = 0.$$

since $Z_{mn} = Z_{nm}$. But $(I_n I_m^* + I_m I_n^*)$ is a purely real quantity which is, in general, nonzero. Thus we must have that

$$\operatorname{Re}\{Z_{mn}\} = 0. \tag{4.39}$$

Then (4.38) and (4.39) imply that $\operatorname{Re}\{Z_{mn}\}=0$ for any *m*. *n*. The reader can verify that this also leads to an imaginary $\{Y\}$ matrix.

EXAMPLE 4.3 Evaluation of Impedance Parameters

Find the Z parameters of the two-port T-network shown in Figure 4.6.

Solution

From (4.28), Z_{11} can be found as the input impedance of port 1 when port 2 is **open-circuited**:

$$Z_{11} = \frac{V_1}{I_1}\Big|_{I_2=0} = Z_A + Z_C.$$

The transfer impedance Z_{12} can be found measuring the open-circuit voltage at port 1 when a current I_2 is applied at port 2. By voltage division.

$$Z_{12} = \frac{V_1}{I_2}\Big|_{I_1=0} = \frac{V_2}{I_2} \frac{Z_C}{Z_B + Z_C} = Z_C.$$



4.38



FIGURE 4.6 A two-port T-network.

The reader can verify that $Z_{21} = Z_{12}$, indicating that the circuit is reciprocal. Finally, Z_{22} is found as

$$Z_{22} = \frac{V_2}{I_2} \bigg|_{I_1=0} = Z_B + Z_C.$$

4.3 THE SCATTERING MATRIX

We have already discussed the difficulty in defining voltages and currents for non-TEM lines. In addition, a practical problem exists when trying to measure voltages and currents at microwave frequencies because direct measurements usually involve the magnitude (inferred from power) and phase of a wave traveling in a given direction, or of a standing wave. Thus, equivalent voltages and currents, and the related impedance and admittance matrices, become somewhat of an abstraction when dealing with highfrequency networks. A representation more in accord with direct measurements, and with the ideas of incident, reflected, and transmitted waves, is given by the scattering matrix.

Like the impedance or admittance matrix for an N-port network, the scattering matrix provides a complete description of the network as seen at its N ports. While the impedance and admittance matrices relate the total voltages and currents at the ports, the scattering matrix relates the voltage waves incident on the ports to those reflected from the ports. For some components and circuits, the scattering parameters can be calculated using network analysis techniques. Otherwise, the scattering parameters can be measured directly with a vector network analyzer; a photograph of a modern network analyzer is shown in Figure 4.7. Once the scattering parameters of the network are known, conversion to other matrix parameters can be performed, if needed.

Consider the N-port network shown in Figure 4.5, where V_n^+ is the amplitude of the voltage wave incident on port *n*, and V_n^- is the amplitude of the voltage wave reflected from port *n*. The scattering matrix, or [S] matrix, is defined in relation to these incident and reflected voltage waves as

$$\begin{bmatrix} V_1^-\\ V_2^-\\ \vdots\\ V_N^- \end{bmatrix} = \begin{bmatrix} S_{11} & S_{12} & \cdots & S_{1N} \\ S_{21} & & \vdots\\ \vdots\\ S_{N1} & \cdots & S_{NN} \end{bmatrix} \begin{bmatrix} V_1^+\\ V_2^+\\ \vdots\\ V_N^+ \end{bmatrix}.$$

$$[V^-] = [S][V^+].$$
4.40



FIGURE 4.7

A photograph of the Hewfett-Packard HP8510B Network Analyzer. This test instrument is used to measure the scattering parameters (magnitude and phase) of a one- or two-port microwave network from 0.05 GHz to 26.5 GHz. Built-in microprocessors provide error correction, a high degree of accuracy, and a wide choice of display formats. This analyzer can also perform a fast Fourier transform of the frequency domain data to provide a time domain response of the network under test.

Courtesy of Hewlent-Packard Company, Santa Rosa, Calif,

A specific element of the [S] matrix can be determined as

$$S_{ij} = rac{V_{i}^{+}}{V_{j}^{+}} \Big|_{V_{k}^{+} = 0 \, \, {
m for} \, k \neq j}.$$
 4.41

In words, (4.41) says that S_{ij} is found by driving port j with an incident wave of voltage V_j^{\pm} , and measuring the reflected wave amplitude, V_i^{\pm} , coming out of port i. The incident waves on all ports except the *j*th port are set to zero, which means that all ports should

be terminated in matched loads to avoid reflections. Thus, S_{ii} is the reflection coefficient seen looking into port *i* when all other ports are terminated in matched loads, and S_{ij} is the transmission coefficient from port *j* to port *i* when all other ports are terminated in matched loads.

EXAMPLE 4.4 Evaluation of Scattering Parameters

Find the S parameters of the 3 dB attenuator circuit shown in Figure 4.8.

Solution

From (4.41), S_{11} can be found as the reflection coefficient seen at port 1 when port 2 is terminated in a matched load ($Z_0 = 50 \Omega$):

$$S_{11} = \frac{V_1^-}{V_1^+}\Big|_{V_2^+=0} = \Gamma^{(1)}\Big|_{V_2^+=0} = \frac{Z_{\text{in}}^{(1)} - Z_0}{Z_{\text{in}}^{(1)} + Z_0}\Big|_{Z_0 \text{ on port } 2}$$

but, $Z_{in}^{(1)} = 8.56 + [141.8(8.56 + 50)]/(141.8 + 8.56 + 50) = 50 \Omega$, so $S_{11} = 0$. Because of the symmetry of the circuit, $S_{22} = 0$.

 S_{21} can be found by applying an incident wave at port 1, V_1^+ , and measuring the outcoming wave at port 2, V_2^- . This is equivalent to the transmission coefficient from port 1 to port 2:

$$S_{21} = \frac{V_2^-}{V_1^+} \bigg|_{V_2^+ = 0}.$$

From the fact that $S_{11} = S_{22} = 0$, we know that $V_1^- = 0$ when port 2 is terminated in $Z_0 = 50 \ \Omega$, and that $V_2^+ = 0$. In this case we then have that $V_1^+ = V_1$ and $V_2^- = V_2$. So by applying a voltage V_1 at port 1 and using voltage division twice we find $V_2^- = V_2$ as the voltage across the 50 Ω load resistor at port 2:

$$V_2^- = V_2 = V_1 \left(\frac{41.44}{41.44 + 8.56}\right) \left(\frac{50}{50 + 8.56}\right) = 0.707V_1,$$

where 41.44 = 141.8(58.56)/(141.8 + 58.56) is the resistance of the parallel combination of the 50 Ω load and the 8.56 Ω resistor with the 141.8 Ω resistor. Thus, $S_{12} = S_{21} = 0.707$.



FIGURE 4.8 A matched 3 dB attenuator with a 50 Ω characteristic impedance (Example 4.4).

If the input power is $|V_1^+|^2/2Z_0$, then the output power is $|V_2^-|^2/2Z_0 = |S_{21}V_1^+|^2/2Z_0 = |S_{21}|^2/2Z_0|V_1^+|^2 = |V_1^+|^2/4Z_0$, which is one-half (-3 dB) of the input power.

We now show how the [S] matrix can be determined from the |Z| (or |Y|) matrix, and vice versa. First, we must assume that the characteristic impedances, Z_{0n} , of all the ports are identical. (This restriction will be removed when we discuss generalized scattering parameters.) Then for convenience, we can set $Z_{0n} = 1$. From (4.24) the total voltage and current at the *n*th port can be written as

$$V_n = V_n^{\pm} + V_n^{\pm}, \qquad \qquad 4.42a$$

$$I_n = I_n^+ - I_n^- = V_n^+ - \tilde{V}_n^-.$$
 4.42b

Using the definition of [Z] from (4.25) with (4.42) gives

$$|Z|[I] = [Z|[V^{-1}] - [Z|[V^{-1}] = [V]] = [V^{-1}] + |V^{-1}].$$

which can be rewritten as

$$([Z] + [U])[V] = ([Z] - [U])[V^+].$$
 4.43

where [U] is the unit, or identity, matrix defined as

$$[U] = \begin{bmatrix} 1 & 0 & \cdots & 0 \\ 0 & 1 & & \vdots \\ \vdots & & \ddots & \\ 0 & & \cdots & 1 \end{bmatrix}.$$

Comparing (4.43) to (4.40) suggests that

$$[S] = ([Z] + [U])^{-1} ([Z] - [U]), \qquad 4.44$$

giving the scattering matrix in terms of the impedance matrix. Note that for a one-port network (4.44) reduces to

$$S_{11} = \frac{z_{11} - 1}{z_{11} + 1}.$$

in agreement with the result for the reflection coefficient seen looking into a load with a normalized input impedance of z_{11} .

To find [Z] in terms of (S), rewrite (4.44) as [Z][S] + [U][S] = [Z] - [U], and solve for [Z] to give

$$[Z] = ([U] - [S])^{-1} ([U] + [S]), \qquad 4.45$$

Reciprocal Networks and Lossless Networks

As we discussed in Section 4.2, the impedance and admittance matrices are symmetric for reciprocal networks, and purely imaginary for lossless networks. Similarly, the scattering matrices for these types of networks have special properties. We will show that the |S| matrix for a reciprocal network is symmetric, and that the |S| matrix for a lossless network is unitary.

By adding (4.42a) and (4.42b) we obtain

$$V_n^+ = \frac{1}{2}(V_n + I_n).$$

[V⁺] = $\frac{1}{2}([Z] + [U])[I].$ 4.46a

OF

By subtracting (4.42a) and (4.42b) we obtain

$$V_n^- = \frac{\mathbf{i}}{2} (V_n - I_n),$$

[V⁻] = $\frac{1}{2} ([Z] - [U])[I].$ 4.46b

or

Eliminating [I] from (4.46a) and (4.46b) gives

$$[V^{-}] = ([Z] - [U])([Z] + [U])^{-1}[V^{+}],$$

[S] = ([Z] - [U])([Z] + [U])^{-1}. 4.47

so that

Taking the transpose of (4.47) gives

$$[S]^{t} = \{ ([Z] + [U])^{-1} \}^{t} ([Z] - [U])^{t}.$$

Now [U] is diagonal, so $[U]^t = [U]$, and if the network is reciprocal, [Z] is symmetric so that [Z]' = [Z]. The above then reduces to

$$[S]^{t} = ([Z] + [U])^{-1}([Z] - [U]),$$

which is equivalent to (4.44). We have thus shown that,

$$[S] = [S]^t, \qquad 4.48$$

for reciprocal networks,

If the network is lossless, then no real power can be delivered to the network. Thus, if the characteristic impedances of all the ports are identical and assumed to be unity, the average power delivered to the network is

$$P_{av} = \frac{1}{2} \operatorname{Re}\{[V]^{t}[I]^{*}\} = \frac{1}{2} \operatorname{Re}\{([V^{+}]^{t} + [V^{-}]^{t})([V^{+}]^{*} - [V^{-}]^{*})\}$$

$$= \frac{1}{2} \operatorname{Re}\{[V^{+}]^{t}[V^{+}]^{*} - [V^{+}]^{t}[V^{-}]^{*} + [V^{-}]^{t}[V^{+}]^{*} - [V^{-}]^{t}[V^{-}]^{*}\}$$

$$= \frac{1}{2} [V^{+}]^{t}[V^{+}]^{*} - \frac{1}{2} [V^{-}]^{t}[V^{-}]^{*} = 0.$$

$$4.49$$

since the terms $-[V^+]^t[V^-]^* + [V^-]^t[V^+]^*$ are of the form $A - A^*$, and so are purely imaginary. Of the remaining terms in (4.49), $(1/2)[V^+]^t[V^+]^*$ represents the total incident power, while $(1/2)[V^-]^t[V^-]^*$ represents the total reflected power. So for a
lossless junction, we have the intuitive result that the incident and reflected powers are equal:

$$[V^+]^t [V^-]^* = [V^-]^t [V^-]^*.$$
4.50

Using $[V^-] = [S][V^+]$ in (4.50) gives

$$[\mathcal{V}^{-}]^{i}[\mathcal{V}^{+}]^{*} = [\mathcal{V}^{+}]^{i}[\mathcal{S}]^{i}[\mathcal{S}]^{*}[\mathcal{V}^{+}]^{*}.$$

so that, for nonzero $[V^+]$.

$$[S]^{t}[S]^{*} = [U],$$

$$[S]^{*} = \{[S]^{t}\}^{-1}.$$

4.51

or

A matrix that satisfies the condition of (4.51) is called a unitary matrix.

The matrix equation of (4.51) can be written in summation form as

$$\sum_{k=1}^{N} S_{ki} S_{kj}^* = \delta_{ij}, \quad \text{for all } i_{\tau, j}, \qquad 4.52$$

where $\delta_{ij} = i$ if i = j and $\delta_{ij} = 0$ if $i \neq j$ is the Kronecker delta symbol. Thus, if i = j (4.52) reduces to

$$\sum_{k=1}^{N} S_{ki} S_{ki}^{*} = 1.$$
 4.53*a*

while if $i \neq j$ (4.52) reduces to

$$\sum_{k=1}^{N} S_{ki} S_{kj}^{*} = 0, \quad \text{for } i \neq j.$$
4.53b

In words, (4.53a) states that the dot product of any column of [S] with the conjugate of that column gives unity, while (4.53b) states that the dot product of any column with the conjugate of a different column gives zero (orthogonal). If the tetwork is reciprocal, then [S] is symmetric, and the same statements can be made about the rows of the scattering matrix.

EXAMPLE 4.5 Application of Scattering Parameters

A certain two-port network is measured and the following scattering matrix is obtained:

$$[S] = \begin{bmatrix} 0.1 \underline{-0} & 0.8 \underline{-90}^{\alpha} \\ 0.8 \underline{-90}^{\alpha} & 0.2 \underline{-0} \end{bmatrix}.$$

From this data, determine whether the network is reciprocal or lossless. If a shortcircuit is placed on port 2, what will be the resulting return loss at port 1?

Solution

Since [S] is symmetric, the network is reciprocal. To be lossless, the S parameters must satisfy (4.53). Taking the first row (i = 1 in (4.53a)) gives

$$|S_{11}|^2 + |S_{12}|^2 = (0.1)^2 + (0.8)^2 = 0.65 \neq 1.$$

Thus, the network is not lossless.

The reflection coefficient, Γ , at port 1 when port 2 is shorted can be calculated as follows. From the definition of the scattering matrix and the fact that $V_2^+ = -V_2^-$ (for a short circuit at port 2), we can write

$$V_1^- = S_{11}V_1^+ + S_{12}V_2^+ = S_{11}V_1^+ - S_{12}V_2^-,$$

$$V_2^- = S_{21}V_1^+ + S_{22}V_2^+ = S_{21}V_1^+ - S_{22}V_2^-.$$

The second equation gives

$$V_2^- = \frac{S_{21}}{1 + S_{22}} V_1^+.$$

Dividing the first equation by V_1^+ , and using the above result, gives the input reflection coefficient as

$$\Gamma = \frac{V_1^-}{V_1^+} = S_{11} - S_{12} \frac{V_2^-}{V_1^+} = S_{11} - \frac{S_{12}S_{22}}{1 + S_{22}}$$
$$= 0.1 - \frac{(j0.8)(j0.8)}{1 + 0.2} = 0.633.$$

So the return loss is

$$RL = -20 \log |\Gamma| = 3.97 \text{ dB}.$$

An important point to understand about S parameters is that the reflection coefficient looking into port n is not equal to S_{nn} , unless all other ports are matched (this is illustrated in the above example). Similarly, the transmission coefficient from port m to port n is not equal to S_{nm} , unless all other ports are matched. The S parameters of a network are properties only of the network itself (assuming the network is linear), and are defined under the condition that all ports are matched. Changing the terminations or excitations of a network does not change its S parameters, but may change the reflection coefficient seen at a given port, or the transmission coefficient between two ports.

A Shift in Reference Planes

Because the S parameters relate amplitudes (magnitude and phase) of traveling waves incident on and reflected from a microwave network, phase reference planes must be specified for each port of the network. We now show how the S parameters are transformed when the reference planes are moved from their original locations.

Consider the N-port microwave network shown in Figure 4.9, where the original terminal planes are assumed to be located at $z_n = 0$ for the *n*th port, and where z_n is



FIGURE 4.9 Shifting reference planes for an N-port network.

an arbitrary coordinate measured along the transmission line feeding the *n*th port. The scattering matrix for the network with this set of terminal planes is denoted by [S]. Now consider a new set of reference planes defined at $z_n = \ell_{ns}$ for the *n*th port, and let the new scattering matrix be denoted as [S⁴]. Then in terms of the incident and reflected port voltages we have that

$$[V^{-}] = [S][V^{+}], 4.54a$$

$$[V'^{-}] = [S'][V'^{+}]. 4.54b$$

where the unprimed quantities are referenced to the original terminal planes at $z_n = 0$, and the primed quantities are referenced to the new terminal planes at $z_n = \ell_n$.

Now from the theory of traveling waves on lossless transmission lines we can relate the new wave amplitudes to the original ones as

$$V_{n}^{\prime+} = V_{n}^{\pm} e^{j\theta_{n}}.$$

$$4.55a$$

$$V_n^{\prime-} = V_n^- e^{-j\theta_n}, \qquad 4.55b$$

where $\theta_n = \beta_n \ell_n$ is the electrical length of the outward shift of the reference plane of port *n*. Writing (4.55) in matrix form and substituting into (4.54a) gives

$$\begin{bmatrix} e^{j\theta_1} & \mathbf{0} \\ & e^{j\theta_2} & \\ & \ddots & \\ \mathbf{0} & & e^{j\theta_N} \end{bmatrix} [V'^-] = [S] \begin{bmatrix} e^{-j\theta_1} & \mathbf{0} \\ & e^{-j\theta_2} & \\ & \ddots & \\ \mathbf{0} & & e^{-j\theta_N} \end{bmatrix} [V'^+].$$

Multiplying by the inverse of the first matrix on the left gives

$$[V'^{-}] = \begin{bmatrix} e^{-j\theta_{1}} & 0 \\ & e^{-j\theta_{2}} \\ & \ddots \\ 0 & e^{-j\theta_{N}} \end{bmatrix} [S] \begin{bmatrix} e^{-j\theta_{1}} & 0 \\ & e^{-j\theta_{2}} \\ & \ddots \\ 0 & e^{-j\theta_{N}} \end{bmatrix} [V'^{+}].$$

Comparing with (4.54b) shows that

$$[S'] = \begin{bmatrix} e^{-j\theta_1} & 0 \\ & e^{-j\theta_2} \\ & \ddots \\ 0 & e^{-j\theta_N} \end{bmatrix} [S] \begin{bmatrix} e^{-j\theta_1} & 0 \\ & e^{-j\theta_2} \\ & \ddots \\ 0 & e^{-j\theta_N} \end{bmatrix}, \quad 4.56$$

which is the desired result. Note that $S'_{nn} = e^{-2j\theta_n}S_{nn}$, meaning that the phase of S_{nn} is shifted by twice the electrical length of the shift in terminal plane *n*, because the wave travels twice over this length upon incidence and reflection.

Generalized Scattering Parameters

So far we have considered the scattering parameters for networks with the same characteristic impedance for all ports. This is the case in many practical situations, where the characteristic impedance is often 50 Ω . In other cases, however, the characteristic impedances of a multiport network may be different, which requires a generalization of the scattering parameters as defined up to this point.

Consider the N-port network shown in Figure 4.10, where Z_{0n} is the (real) characteristic impedance of the *n*th port, and V_n^+ and V_n^- , respectively, represent the incident and reflected voltage waves at port *n*. In order to obtain physically meaningful power relations in terms of wave amplitudes, we must define a new set of wave amplitudes as

$$a_n = V_n^+ / \sqrt{Z_{0n}},$$
 4.57a



FIGURE 4,10 An N-port network with different characteristic impedances.

4.3 The Scattering Matrix

$$b_n = V_n^+ / \sqrt{Z_{0n}}. \tag{4.57}$$

where a_n represents an incident wave at the *n*th port, and b_n represents a reflected wave from that port [1], [5]. Then from (4.42a,b) we have that

$$V_n = V_n^+ + V_n^- = \sqrt{Z_{0n}}(a_n + b_n), \qquad 4.58a$$

$$I_n = \frac{1}{Z_{0n}} (V_n^+ - V_n^-) = \frac{1}{\sqrt{Z_{0n}}} (a_n - b_n).$$
 4.58b

Now the average power delivered to the nth port is

$$P_n = \frac{1}{2} \operatorname{Re}\{V_n I_n^*\} = \frac{1}{2} \operatorname{Re}\{|a_n|^2 - |b_n|^2 + (b_n a_n^* - b_n^* a_n)\} = \frac{1}{2} |a_n|^2 - \frac{1}{2} |b_n|^2. \quad 4.59$$

since the quantity $(b_n a_n^* - b_n^* a_n)$ is purely imaginary. This is a physically satisfying result, since it says that the average power delivered through port *n* is equal to the power in the incident wave minus the power in the reflected wave. If expressed in terms of V_n^- and V_n^- , the corresponding result would be dependent on the characteristic impedance of the *n*th port

A generalized scattering matrix can then be used to relate the incident and reflected waves defined in (4.57).

$$[b] = [S][a], \qquad 4.60$$

where the i, jth element of the scattering matrix is given by

$$S_{ij} = \frac{\langle \mathbf{x}_j \rangle}{\alpha_j} \bigg|_{\alpha_k = 0 \text{ for } k \neq j}, \qquad 4.61$$

and is analogous to the result of (4.41) for networks with identical characteristic impedance at all ports. Using (4.57) in (4.61) gives

$$S_{ij} = \frac{V_i^{-} \sqrt{Z_{0j}}}{V_j^{+} \sqrt{Z_{0i}}} \bigg|_{V_k^{+} = 0 \text{ for } k \neq j},$$

$$4.62$$

which shows how the S parameters of a network with equal characteristic impedance $(V_j^+/V_j^+ \text{ with } V_k^+ = 0 \text{ for } k \neq j)$ can be converted to a network connected to transmission lines with unequal characteristic impedances.

POINT OF INTEREST: The Vector Network Analyzer

The S parameters of passive and active networks can be measured with a vector network analyzer, which is a two- (or four-) channel microwave receiver designed to process the magnitude and phase of the transmitted and reflected waves from the network. A simplified block diagram of a network analyzer similar to the HP8510 system is shown below. In operation, the RF source is usually set to sweep over a specified bandwidth. A four-port reflectometer samples the incident, reflected, and transmitted RF waves, a switch allows the network to be driven from either port 1 or port 2. Four dual-conversion channels convert these signals to 100 kHz IF frequencies, which are then detected and converted to digital form. A powerful internal computer is used to calculate



and display the magnitude and phase of the S parameters, or other quantities that can be derived from the S parameters, such as SWR, return loss, group delay, impedance, etc. An important feature of this network analyzer is the substantial improvement in accuracy made possible with error correcting software. Errors caused by directional coupler mismatch, imperfect directivity, loss, and variations in the frequency response of the analyzer system are accounted for by using a twelve-term error model and a calibration procedure. Another useful feature is the capability to determine the time domain response of the network by calculating the inverse Fourier transform of the frequency domain data.

4.4 ____ THE TRANSMISSION (ABCD) MATRIX

The Z, Y, and S parameter representations can be used to characterize a microwave network with an arbitrary number of ports, but in practice many microwave networks consist of a cascade connection of two or more two-port networks. In this case it is convenient to define a 2×2 transmission, or ABCD matrix, for each two-port network. We will then see that the ABCD matrix of the cascade connection of two or more two-port networks can be easily found by multiplying the ABCD matrices of the individual two-ports.

The ABCD matrix is defined for a two-port network in terms of the total voltages and currents as shown in Figure 4.11a and the following:

$$V_1 = AV_2 + BI_2,$$

$$I_1 = CV_2 + DI_2,$$

or in matrix form as

$$\begin{bmatrix} V_1 \\ I_1 \end{bmatrix} = \begin{bmatrix} A & B \\ C & D \end{bmatrix} \begin{bmatrix} V_2 \\ I_2 \end{bmatrix}.$$
4.63

It is important to note from Figure 4.11a that a change in the sign convention of I_2 has been made from our previous definitions, which had I_2 as the current flowing *into* port 2. The convention that I_2 flows *out* of port 2 will be used when dealing with *ABCD* matrices so that in a cascade network I_2 will be the same current that flows *into* the adjacent network, as shown in Figure 4.11b. (Alternatively, I_2 in (4.43) could be replaced by $-I_2$, so that the sign convention would not have to be changed [1], [5].) Then the left-hand side of (4.63) represents the voltage and current at port 1 of the network, while the right-hand side of (4.63) represents the voltage and current at port 2.

In the cascade connection of two two-port networks shown in Figure 4.11b, we have that

$$\begin{bmatrix} V_1 \\ I_1 \end{bmatrix} = \begin{bmatrix} A_1 & B_1 \\ C_1 & D_1 \end{bmatrix} \begin{bmatrix} V_2 \\ I_2 \end{bmatrix}.$$
 4.64a

$$\begin{bmatrix} V_2 \\ I_2 \end{bmatrix} = \begin{bmatrix} A_2 & B_2 \\ C_2 & D_2 \end{bmatrix} \begin{bmatrix} V_3 \\ I_3 \end{bmatrix}.$$
 4.64b

Substituting (4.64b) into (4.64a) gives

$$\begin{bmatrix} V_1 \\ I_1 \end{bmatrix} \approx \begin{bmatrix} A_1 & B_1 \\ C_1 & C_1 \end{bmatrix} \begin{bmatrix} A_2 & B_2 \\ C_2 & D_2 \end{bmatrix} \begin{bmatrix} V_3 \\ I_3 \end{bmatrix},$$
4.65

which shows that the ABCD matrix of the cascade connection of the two networks is equal to the product of the ABCD matrices representing the individual two-ports. Note that the order of multiplication of the matrix must be the same as the order in which the networks are arranged, since matrix multiplication is not, in general, commutative,



____ FIGURE 4.11

(a) A two-port network: (b) a cascade connection of two-port networks.

The usefulness of the ABCD matrix representation lies in the fact that a library of ABCD matrices for elementary two-port networks can be built up, and applied in building-block fashion to more complicated microwave networks that consist of cascades of these simpler two-ports. Table 4.1 lists a number of useful two-port networks and their ABCD matrices.



EXAMPLE 4.6 Evaluation of ABCD Parameters

Find the ABCD parameters of a two-port network consisting of a series impedance Z between ports 1 and 2 (the first entry in Table 4.1).

Circuit	ABCD Parame	eters
•Z•	A = 1 $C = 0$	B = Z $D = 1$
оо о	A = 1 $C = Y$	B = 0 $D = 1$
	$A = \cos \beta l$ $C = jY_0 \sin \beta l$	$B = jZ_0 \sin \beta l$ $D = \cos \beta l$
	A = N $C = 0$	$B = 0$ $D = \frac{1}{N}$
$\begin{array}{c} \bullet \\ \hline Y_1 \\ \bullet \\ \hline \end{array}$	$A = 1 + \frac{Y_2}{Y_3}$ $C = Y_1 + Y_2 + \frac{Y_1 Y_2}{Y_3}$	$B = \frac{1}{Y_3}$ $D = 1 + \frac{Y_1}{Y_3}$
	$A = 1 + \frac{Z_1}{Z_3}$ $C = \frac{1}{Z_3}$	$B = Z_1 + Z_2 + \frac{Z_1 Z_2}{Z_3}$ $D = \frac{1}{2} + \frac{Z_2}{Z_3}$

TABLE 4.1 The ABCD Parameters of Some Useful Two-Port Circuits

Solution

From the defining relations of (4.63), we have that

$$A = \frac{V_1}{V_2}\bigg|_{I_2 \to 0},$$

which indicates that A is found by applying a voltage V_1 at port 1, and measuring the open-circuit voltage V_2 at port 2. Thus, A = 1. Similarly,

$$B = \frac{V_1}{I_2}\Big|_{V_2=0} = \frac{V_1}{V_1/Z} = Z,$$

$$C = \frac{I_1}{V_2}\Big|_{I_2=0} = 0,$$

$$D = \frac{I_1}{I_2}\Big|_{V_2=0} = \frac{I_1}{I_1} = 1.$$

Relation to Impedance Matrix

Knowing the Z parameters of a network, one can determine the ABCD parameters. Thus, from the definition of the ABCD parameters in (4.63), and from the defining relations for the Z parameters of (4.25) for a two-port network with I_2 to be consistent with the sign convention used with ABCD parameters,

we have that

$$A = \frac{V_1}{V_2}\Big|_{I_2=0} = \frac{I_1 Z_{11}}{I_1 Z_{21}} = Z_{11}/Z_{21}.$$

$$B = \frac{V_1}{I_2}\Big|_{V_2=0} = \frac{I_1 Z_{11} - I_2 Z_{12}}{I_2}\Big|_{V_2=0} = Z_{11} \frac{I_1}{I_2}\Big|_{V_2=0} - Z_{12}$$
4.67*a*

$$= Z_{11} \frac{I_1 Z_{22}}{I_1 Z_{21}} - Z_{12} = \frac{Z_{11} Z_{22} - Z_{12} Z_{21}}{Z_{21}},$$
4.67b

$$\tilde{C} = \frac{I_1}{V_2}\Big|_{I_2=0} = \frac{I_1}{I_1 Z_{21}} = 1/Z_{21}.$$
 4.67c

$$D = \frac{I_1}{I_2} \bigg|_{V_2 = 0} = \frac{I_2 Z_{22} / Z_{21}}{I_2} = Z_{22} / Z_{21}.$$
 4.67d

If the network is reciprocal, then $Z_{12} = Z_{21}$ and (4.67) can be used to show that AD - BC = 1.

Equivalent Circuits for Two-Port Networks

The special case of a two-port microwave network occurs so frequently in practice that it deserves further attention. Here we will discuss the use of equivalent circuits to represent an arbitrary two-port network. Useful conversions for two-port network parameters are given in Table 4.2.

Figure 4.12a shows a transition between a coaxial line, and a microstrip line, and serves as an example of a two-port network. Terminal planes can be defined at arbitrary points on the two transmission lines: a convenient choice might be as shown in the figure. But because of the physical discontinuity in the transition from a coaxial line to a microstrip line, electric and/or magnetic energy can be stored in the vicinity of the junction, leading to reactive effects. Characterization of such effects can be obtained by measurement or by theoretical analysis (although such an analysis may be quite complicated), and represented by the two-port "black box" shown in Figure 4.12b. The properties of the transition can then be expressed in terms of the network parameters (Z, Z)Y, S, or ABCD) of the two-port network. This type of treatment can be applied to a variety of two-port junctions, such as transitions from one type of transmission line to another, transmission line discontinuities such as step changes in width, or bends, etc. When modeling a microwave junction in this way, it is often useful to replace the two-port "black box" with an equivalent tircuit containing a few idealized components, as shown in Figure 4.12c. (This is particularly useful if the component values can be related to some physical features of the actual junction.) There is an unlimited number of ways in which such equivalent circuits can be defined; we will discuss some of the most common and useful types below.

As we have seen from the previous sections, an arbitrary two-port network can be described in terms of impedance parameters as

$$V_1 = Z_{1_1}I_1 + Z_{12}I_2,$$

$$V_2 = Z_{2_1}I_1 + Z_{22}I_2,$$

4.68a

or in terms of admittance parameters as

$$\begin{split} I_1 &= Y_{14}V_1 + Y_{12}V_2, \\ I_2 &= Y_{24}V_1 + Y_{22}V_2. \end{split}$$
 4.68b

If the network is reciprocal, then $Z_{12} = Z_{21}$ and $Y_{12} = Y_{21}$. These representations lead naturally to the T and π equivalent Grouts shown in Figure 4.13a and 4.13b. The relations in Table 4.2 can be used to relate the component values to other network parameters.

Other equivalent circuits can also be used to represent a two-port network. If the network is reciprocal, there are six degrees of freedom (the real and imaginary parts of three matrix elements), so the equivalent circuit should have six independent parameters. A nonreciprocal network cannot be represented by a passive equivalent circuit using reciprocal elements.

	TABLE 4.2 Conversions Between 7	wo-Port Network Parameters		
	s,	2	Y	ABCD
Sil	S_{11}	$rac{(Z_{11}-Z_0)(Z_{22}+Z_0)-Z_{12}Z_{21}}{\Delta Z}$	$\frac{(Y_0 - Y_{11})(Y_0 + Y_{22}) + Y_{12}Y_{23}}{\Delta Y}$	$\frac{A+B/Z_0-CZ_0-D}{A+B/Z_0+CZ_0+D}$
S_{12}	Sız	$rac{2Z_{12}Z_0}{\Delta Z}$	$\frac{-2Y_{12}Y_{6}}{\Delta Y}$	$\frac{2(AD - BC)}{A + B/Z_0 + CZ_0 + D}$
521		$\frac{2Z_{21}Z_0}{\Delta Z}$	$\frac{-2Y_{21}Y_0}{\Delta Y}$	$\frac{1}{2}$
S22	- Se	$rac{(Z_{11}+Z_0)(Z_{22}-Z_0)-Z_{12}Z_{24}}{\Delta Z}$	$\frac{(Y_{11} + Y_{11})(Y_{11} - Y_{22}) + Y_{12}Y_{21}}{\Delta Y}$	$\frac{-A+B/Z_0-CZ_0+D}{A+B/Z_0+CZ_0+D}$
21	$Z_0 \frac{(1+S_{11})(1-S_{22})+S_{12}S_{21}}{(1-S_{11})(1-S_{22})-S_{22}S_{22}}$	×11	T.	512
Z_{12}	$Z_{0} \frac{2S_{12}}{(1-S_{11})(1-S_{22})-S_{12}S_{23}}$	Z_{12}	- <u>Y</u> 12	$\frac{AD - BC}{C}$
221	$Z_0 \frac{2S_{21}}{(1-S_{11})(1-S_{22})-S_{12}S_{21}}$	盛	- X ₂₁	- ¢
Z17	$\mathbb{Z}_{0}^{1}\frac{(1-S_{11}\chi_{1}+S_{22})+S_{12}\chi_{2}}{(1-S_{11})(1-S_{22})+S_{12}S_{21}}$	Ř	**	مالا
	$Y_{11} \frac{(1-S_{11})(1+S_{22})+S_{12}S_{21}}{(1+S_{11})(1+S_{22})-S_{12}S_{21}}$	Z	Ā	1218
Y_{12}	$Y_{11} - \frac{-2S_{12}}{(1+S_{11})(1+S_{22}) - S_{12}S_{21}}$	<u>-Zr</u>	Y.	<u>IIC' - AU</u> B
Y ₂₁	$\gamma_{6}^{\gamma_{6}} \frac{-2S_{21}}{(1+S_{11})(1+S_{22})-S_{12}S_{21}}$	$\frac{-2_{24}}{ 2 }$	Ч ^й	7 📾
$\chi_{\rm u}$	$y_{0} \frac{(1+S_{11})(1-S_{22})+S_{12}N_{2}}{(1+S_{11})(1+S_{22})-S_{12}N_{2}}$	<u>Z</u>	Y ₁₂	* 3
V	$\frac{(1+S_{11})(1-S_{22})+S_{12}S_{21}}{2S_{23}}$	2	$-V_{22}$	¥
13	$Z_0 \frac{(1+S_{11})(1+S_{22})-S_{12}S_{21}}{2S_{21}}$	Za Za	- <u>-</u>	B
ひ	$\frac{1}{Z_0} \frac{(1-S_{11})(1-S_{22})-S_{12}S_{21}}{2S_{21}}$	- [2]	<u> </u> <u>k</u> <u>k</u>	0
Q	$\frac{(1-S_{11})(1+S_{22})+S_{12}S_{21}}{2(S_{21})}$	$\frac{Z_{21}}{Z_{21}}$	<u>-Y.</u> <u>F.</u>	n
$\frac{ }{ z }$	$Z_{11}Z_{22} - Z_{12}Z_{21}$ Y = Y ₁₁ Y ₂₂ - Y ₁	$(Y_{2i}, \Delta Y = (Y_{11} + Y_0)(Y_{22} + Y_0)$	$-Y_{12}Y_{21}; \Delta Z = (Z_{11} + Z_{01})(Z_{22} - Z_{12})$	$-Z_{\rm fl}$) - $Z_{12}Z_{21}$; $Y_0 = 1/Z_0$

÷ ĝ 4 ÷ Ż Å F ġ , đ ę ĥ

211



FIGURE 4.12

A coax-to-microstrip transition and equivalent circuit representations. (a) Geometry of the transition. (b) Representation of the transition by a "black box." (c) A possible equivalent circuit for the transition [6].



FIGURE 4.13

³ Equivalent circuits for a reciprocal two-port network. (a) T equivalent. (b) π equivalent.

If the network is lossless, which is a good approximation for many practical two-port junctions, some simplifications can be made in the equivalent circuit. As was shown in Section 4.2, the impedance or admittance matrix elements are purely imaginary for a lossless network. This reduces the degrees of freedom for such a network to three, and implies that the T and π equivalent circuits of Figure 4.13 can be constructed from purely reactive elements.

4.5 SIGNAL FLOW GRAPHS

We have seen how transmitted and reflected waves can be represented by scattering parameters, and how the interconnection of sources, networks, and loads can be treated with various matrix representations. In this section we discuss the signal flow graph, which is an additional technique that is very useful for the analysis of microwave networks in terms of transmitted and reflected waves. We first discuss the features and the construction of the flow graph itself, and then present a technique for the reduction, or solution, of the flow graph.

The primary components of a signal flow graph are nodes and branches:

- Nodes: Each port, *i*, of a microwave network has two nodes, a_i and b_i . Node a_i is identified with a wave entering port *i*, while node b_i is identified with a wave reflected from port *i*.
- Branches: A branch is a directed path between an *a*-node and a *b*-node, representing signal flow from node *a* to node *b*. Every branch has an associated *S* parameter or reflection coefficient.

At this point it is useful to consider the flow graph of an arbitrary two-port network, as shown in Figure 4.14. Figure 4.14a shows a two-port network with incident and reflected waves at each port, and Figure 4.14b shows the corresponding signal flow



FIGURE 4.14

The signal flow graph representation of a two-port network. (a) Definition of incident and reflected waves. (b) Signal flow graph.

graph representation. The flow graph gives an intuitive graphical illustration of the network behavior.

For example, a wave of amplitude a_1 incident at port 1 is split, with part going through S_{11} and out port 1 as a reflected wave and part transmitted through S_{21} to node b_{22} . At node b_2 , the wave goes out port 2; if a load with nonzero reflection coefficient is connected at port 2, this wave will be at least partly reflected and reenter the two-port network at node a_2 . Part of the wave can be reflected back out port 2 via S_{22} , and part can be transmitted out port 1 through S_{12} .

Two other special networks, a one-port network and a voltage source, are shown in Figure 4.15 along with their signal flow graph representations. Once a microwave network has been represented in signal flow graph form, it is a relatively easy matter to solve for the ratio of any combination of wave amplitudes. We will discuss how this can be done using four basic decomposition rules, but the same results can also be obtained using Mason's rule from control system theory.

Decomposition of Signal Flow Graphs

A signal flow graph can be reduced to a single branch between two nodes using the four basic decomposition rules below, to obtain any desired wave amplitude ratio,

• Rule 1 (Series Rule). Two branches, whose common node has only one incoming and one outgoing wave (branches in series), may be combined to form a single branch whose coefficient is the product of the coefficients of the original branches. Figure 4.16a shows the flow graphs for this rule. Its derivation follows from the basic relation that

$$V_3 = S_{32}V_2 = S_{32}S_{21}V_1. \tag{4.69}$$



(b)



The signal flow graph representations of a one-port network and a source. (a) A one-port network and its flow graph. (b) A source and its flow graph,



FIGURE 4.16

Decomposition rules. (a) Series rule. (b) Parallel rule. (c) Self-loop rule. (d) Splitting rule.

• Rule 2 (Parallel Rule). Two branches from one common node to another common node (branches in parallel) may be combined into a single branch whose coefficient is the sum of the coefficients of the original branches. Figure 4.16b shows the flow graphs for this rule. The derivation follows from the obvious relation that

$$V_2 = S_a V_1 + S_b V_1 = (S_a + S_b) V_1.$$
4.70

• Rule 3 (Self-Loop Rule). When a node has a self-loop (a branch that begins and ends on the same node) of coefficient S, the self-loop can be eliminated by multiplying coefficients of the branches feeding that node by 1/(1 - S).

Figure 4.16c shows the flow graphs for this rule, which can be derived as follows. From the original network we have that

$$V_2 = S_{21}V_1 + S_{22}V_2. 4.71a$$

$$V_3 = S_{32}V_2. 4.71b$$





Eliminating V_2 gives

$$V_3 = \frac{S_{32}S_{21}}{1 - S_{22}}V_1, \qquad 4.72$$

which is seen to be the transfer function for the reduced graph of Figure 4.16c.

• Rule 4 (Splitting Rule). A node may be split into two separate nodes as long as the resulting flow graph contains, once and only once, each combination of separate (not self loops) input and output branches that connect to the original node.

This rule is illustrated in Figure 4.16d, and follows from the observation that

$$V_4 = S_{42}V_2 = S_{21}S_{42}V_1, 4.73$$

in both the original flow graph and the flow graph with the split node.

We now illustrate the use of each of these rules with an example.



EXAMPLE 4.7 Application of Signal Flow Graph

Derive the expression for Γ_{in} for the terminated two-port network shown in Figure 4.17 using signal flow graphs and the above decomposition rules.

Solution

The signal flow graph for the circuit of Figure 4.17 is shown in Figure 4.18. We wish to find $\Gamma_{in} = b_1/a_1$. Figure 4.19 shows the four steps in the decomposition of the flow graphs, with the final result that

$$\Gamma_{\rm in} = \frac{b_1}{a_1} = S_{11} + \frac{S_{12}S_{21}\Gamma_{\ell}}{1 - S_{22}\Gamma_{\ell}},$$



FIGURE 4.18

Signal flow path for the two-port network with general source and load impedances of Figure 4.17.



FIGURE 4.19

Decomposition of the flow graph of Figure 4.18 to find $\Gamma_{\rm m} = b_1/a_1$. (a) Using Rule 4 on node a_2 . (b) Using Rule 3 for the self-loop. (c) Using Rule 1. (d) Using Rule 2.

Application to TRL Network Analyzer Calibration

As a further application of signal flow graphs we consider the calibration of a network analyzer using the Thru-Reflect Line (TRL) technique [7]. The general problem is shown in Figure 4.20, where it is intended to measure the S-parameters of a two-port device at the indicated reference planes. As discussed in the previous Point of Interest, a network analyzer measures S-parameters as ratios of complex voltage amplitudes. The primary reference plane for such measurements is generally at some point within the analyzer itself, so the measurement will include losses and phase delays caused by the effects of the connectors, cables, and transitions that must be used to connect the device under test (DUT) to the analyzer. In the block diagram of Figure 4.20 these effects are lumped together in a two-port *error box* placed at each port between the actual measurement





reference plane and the desired reference plane for the two-port DUT. A calibration procedure is used to characterize the error boxes before measurement of the DUT; then the actual *error-corrected S*-parameters of the DUT can be calculated from the measured data. Measurement of a one-port network can be considered as a reduced case of the two-port case.

The simplest way to calibrate a network analyzer is to use three or more known loads, such as shorts, opens, and matched loads. The problem with this approach is that such standards are always imperfect to some degree, and therefore introduce errors into the measurement. These errors become increasingly significant at higher frequencies and as the quality of the measurement system improves. The TRL calibration scheme does not rely on known standard loads, but uses three simple connections to allow the error boxes to be characterized completely. These three connections are shown in Figure 4.21. The *Thru* connection is made by directly connecting port 1 to port 2, at the desired reference planes. The *Reflect* connection uses a load having a large reflection coefficient, Γ_L , such as a nominal open or short. It is not necessary to know the exact value of Γ_L , as this will be determined by the TRL calibration procedure. The *Line* connection involves connecting ports 1 and 3 together through a length of matched transmission line. It is not necessary to know the length of the line, and it is not required that the line be lossless; these parameters will be determined by the TRL procedure.

We can use signal flow graphs to derive the set of equations necessary to find the S-parameters for the error boxes in the TRL calibration procedure. With reference to Figure 4.20, we will apply the *Thru*, *Reflect*, and *Line* connections at the reference plane for the DUT, and measure the S-parameters for these three cases at the measurement planes. For simplicity, we assume the same characteristic impedance for ports 1 and 2, and that the error boxes are reciprocal and identical for both ports. The error boxes are characterized by the *S*-matrix [*S*], and alternatively by the *ABCD* matrix. Thus we have



FIGURE 4.21a Block di

Block diagram and signal flow graph for the Thru connection.



FIGURE 4.21b Block diagram and signal flow graph for the Reflect connection.

 $S_{21} = S_{12}$ for both error boxes, and an inverse relation between the ABCD matrices of the error boxes for ports 1 and 2, since they are symmetrically connected as shown in the figure. To avoid confusion in notation we will denote the measured S-parameters for the Thru, Reflect. and Line connections as the [T], [R], and [L] matrices, respectively.

Figure 4.21a shows the arrangement for the *Thru* connection and the corresponding signal flow graph. Observe that we have made use of the fact that $S_{21} = S_{12}$ and that the error boxes are identical and symmetrically arranged. The signal flow graph can be



FIGURE 4.21c Block diagram and signal flow graph for the Line connection,

easily reduced using the decomposition rules to give the measured S-parameters at the measurement planes in terms of the S-parameters of the error boxes as

$$T_{11} = \frac{b_1}{a_1}\Big|_{a_2=0} = S_{11} + \frac{S_{22}S_{12}^2}{1 - S_{22}^2}$$
 4.74a

$$T_{12} = \frac{b_1}{a_2} \bigg|_{a_1 = 0} = \frac{S_{12}^2}{1 - S_{22}^2}$$
 4.74b

By symmetry we have $T_{22} = T_{11}$, and by reciprocity we have $T_{21} = T_{12}$.

The Reflect connection is shown in Figure 4.21b, with the corresponding signal flow graph. Note that this arrangement effectively decouples the two measurement ports, so $R_{12} = R_{21} = 0$. The signal flow graph can be easily reduced to show that,

$$R_{11} = \left. \frac{b_1}{a_1} \right|_{a_2 = 0} = S_{11} + \frac{S_{12}^2 \Gamma_L}{1 - S_{22} \Gamma_L}$$

$$4.75$$

By symmetry we have $R_{22} \approx R_{11}$.

The Line connection is shown in Figure 4.21c, with its corresponding signal flow graph. A reduction similar to that used for the Thru case gives,

$$L_{11} = \frac{b_1}{a_1} \bigg|_{a_2 = 0} = S_{11} + \frac{S_{22} S_{12}^2 e^{-2\gamma \ell}}{1 - S_{22}^2 e^{-2\gamma \ell}}$$

$$4.76a$$

$$L_{12} = \left. \frac{b_1}{a_2} \right|_{a_1=0} = \frac{S_{12}^2 e^{-\gamma \ell}}{1 - S_{22}^2 e^{-2\gamma \ell}}$$

$$4.76b$$

By symmetry and reciprocity we have $L_{22} = L_{11}$ and $L_{21} = L_{12}$.

We now have five equations (4.74)-(4.76) for the five unknowns S_{11} , S_{12} , S_{22} , Γ_L , and $e^{-\gamma \ell}$; the solution is straightforward but lengthy. Since (4.75) is the only equation that contains Γ_L , we can first solve the four equations in (4.74) and (4.76) for the other four unknowns. Equation (4.74b) can be used to eliminate S_{12} from (4.74a) and (4.76), and then S_{11} can be eliminated from (4.74a) and (4.76a). This leaves two equations for S_{22} and $e^{-\gamma \ell}$:

$$L_{12}e^{-2\gamma\ell} - L_{12}S_{22}^2 = T_{12}e^{-\gamma\ell} - T_{12}S_{22}^2e^{-\gamma\ell}$$

$$4.77a$$

$$e^{-2\gamma\ell}\left(T_{11}-S_{22}T_{12}\right)-T_{11}S_{22}^2=L_{11}\left(e^{-2\gamma\ell}-S_{22}^2\right)-S_{22}T_{12}.$$
4.77b

Equation (4.77a) can now be solved for S_{22} and substituted into (4.77b) to give a quadratic equation for $e^{-\gamma \ell}$. Application of the quadratic formula then gives the solution for $e^{-\gamma \ell}$ in terms of the measured TRL S-parameters as

$$e^{-\gamma\ell} = \frac{L_{12}^2 + T_{12}^2 - (T_{11} - L_{11})^2 \pm \sqrt{\left[L_{12}^2 + T_{12}^2 - (T_{11} - L_{11})^2\right]^2 - 4L_{12}^2 T_{12}^2}}{2L_{12}T_{12}} \quad 4.78$$

The choice of sign can be determined by the requirement that the real and imaginary parts of γ be positive, or by knowing the phase of Γ_{ℓ} (as determined from (4.83)) to within 180°.

Next we multiply (4.74b) by S_{22} and subtract from (4.74a) to get

$$T_{11} = S_{11} + S_{22} T_{12}, \qquad 4.79a$$

and similarly multiply (4.76b) by S_{23} and subtract from (4.76a) to get

$$L_{11} = S_{11} + S_{22}L_{12}e^{-\gamma\ell}.$$
 4.79b

Eliminating S_{11} from these two equations gives S_{22} in terms of $e^{-\gamma t}$ as

$$S_{22} = \frac{T_{11} - L_{11}}{T_{12} - L_{12}e^{-\gamma t}}.$$

Solving (4.79a) for S_{11} gives

$$S_{11} = T_{11} - S_{22} T_{12}, \qquad \qquad 4.81$$

and solving (4.74b) for S_{12} gives

$$S_{12}^2 = T_{12}(1 - S_{22}^2).$$
 4.82

Finally, (4.75) can be solved for Γ_L to give

$$\Gamma_{L} = \frac{R_{11} - S_{11}}{S_{12}^2 + S_{22} (R_{11} - S_{11})}.$$
4.83

Equations (4.78) and (4.80)–(4.83) give the S-parameters for the error boxes, as well as the unknown reflection coefficient, Γ_i (to within the sign), and the propagation factor, $e^{-\gamma\ell}$. This completes the calibration procedure for the TRL method.

The S-parameters of the DUT can now be measured at the measurement reference planes shown in Figure 4.20, and corrected using the above TRL error box parameters to give the S-parameters at the reference planes of the DUT. Since we are now working with a cascade of three two-port networks, it is convenient to use ABCD parameters. Thus, we convert the error box S-parameters to the corresponding ABCD parameters, and convert the measured S-parameters of the cascade to the corresponding $A^m B^m C^m D^m$ parameters. If we use A'B'C'D' to denote the parameters for the DUT, then we have that

$$\begin{bmatrix} A^m & B^m \\ C^m & D^m \end{bmatrix} \approx \begin{bmatrix} A & B \\ C & D \end{bmatrix} \begin{bmatrix} A' & B' \\ C' & D' \end{bmatrix} \begin{bmatrix} A & B \\ C & D \end{bmatrix}^{-1}$$

from which we can determine the ABCD parameters for the DUT as

$$\begin{bmatrix} A' & B' \\ C' & D' \end{bmatrix} \begin{bmatrix} A & B \\ C & D \end{bmatrix}^{-1} \begin{bmatrix} A^m & B^m \\ C^m & D^m \end{bmatrix} \begin{bmatrix} A & B \\ C & D \end{bmatrix}.$$
4.84



POINT OF INTEREST: Computer-Aided Design for Microwave Circuits

A computer-aided design (CAD) software package for microwave circuit analysis and optimization can be a very useful tool for the microwave engineer. Several microwave CAD programs are commercially available, such as SUPERCOMPACT[®] and TOUCHSTONE[®], with the capability of analyzing microwave circuits consisting of transmission lines, lumped elements, active devices, coupled lines, waveguides, and other components. Although such computer programs can be fast, powerful, and accurate, they cannot serve as a substitute for an experienced engineer with a good understanding of microwave design.

A typical design process will usually begin with specifications or design goals for the circuit. Based on previous designs and his own experience, the engineer can develop an initial design, including specific components and a circuit layout. CAD can then be used to model and analyze the design, using data for each of the components and including effects such as loss and discontinuities. The CAD program can be used to optimize the design by adjusting some of the circuit parameters to achieve the best performance. If the specifications are not met, the design may have to be revised. The CAD analysis can also be used to study the effects of component tolerances and errors, to improve circuit reliability and robustness. When the design meets the specifications, an engineering prototype can be built and tested. If the measured results satisfy the specifications, the design process is completed. Otherwise the design will need to be revised, and the procedure repeated.

Without CAD tools, the design process would require the construction and measurement of a laboratory prototype at each iteration, which would be expensive and time consuming. Thus, CAD can greatly decrease the time and cost of a design, while enhancing its quality. The simulation and optimization process is especially important for monolithic microwave integrated circuits (MMICs) because these circuits cannot easily be tuned or trimmed after fabrication.

CAD techniques are not without limitations, however. Of primary importance is the fact that a computer model is only an approximation to a "real-world" circuit, and cannot completely account for the inevitable effects of component and fabricational tolerances, surface roughness, spurious coupling, higher order modes, and junction discontinuities. These limitations generally become most serious at frequencies above 10 GHz.

4.6 DISCONTINUITIES AND MODAL ANALYSIS

By either necessity or design, microwave networks often consist of transmission lines with various types of transmission line discontinuities. In some cases discontinuities are an unavoidable result of mechanical or electrical transitions from one medium to another (e.g., a junction between two waveguides, or a coax-to-microstrip transition), and the discontinuity effect is unwanted but may be significant enough to warrant characterization. In other cases discontinuities may be deliberately introduced into the circuit to perform a certain electrical function (e.g., reactive diaphragms in waveguide or stubs in microstrip line for matching or filter circuits). In any event, a transmission line discontinuity can be represented as an equivalent circuit at some point on the transmission line. Depending on the type of discontinuity, the equivalent circuit may be a simple shunt or series element across the line or, in the more general case, a T- or π -equivalent circuit may be required. The component values of an equivalent circuit depend on the parameters of the line and the discontinuity, as well as the frequency of operation. In some cases the equivalent

[®] Registered trademarks of Compact Software Corp. and EEsof, Inc., respectively.

circuit involves a shift in the phase reference planes on the transmission lines. Once the equivalent circuit of a given discontinuity is known, its effect can be incorporated into the analysis or design of the network using the theory developed previously in this chapter.

The purpose of the present section is to discuss how equivalent circuits are obtained for transmission line discontinuities; we will see that the basic procedure is to start with a field theory solution to a canonical discontinuity problem and develop a circuit model, with component values. This is thus another example of our objective of replacing complicated field analyses with circuit concepts.

Figures 4.22 and 4.23 show some common transmission line discontinuities and their equivalent circuits. As shown in Figures 4.22a–c, thin metallic diaphragms (or "irises")





Rectangular waveguide discontinuities.



FIGURE 4.23

Some common microstrip discontinuities. (a) Open-ended microstrip. (b) Gap in microstrip. (c) Change in width. (d) T-junction. (e) Coax-to-microstrip junction.

can be placed in the cross section of a waveguide to yield equivalent shunt inductance, capacitance, or a resonant combination. Similar effects occur with step changes in the height or width of the waveguide, as shown in Figures 4.22d,e. Similar discontinuities can also be made in circular waveguide. The best reference for waveguide discontinuities and their equivalent circuits is *The Waveguide Handbook* [8].

Some typical microstrip discontinuities and transitions are shown in Figure 4.23; similar geometries exist for stripline and other printed transmission lines such as slotline, covered microstrip, coplanar waveguide, etc. Since printed transmission lines are newer, relative to waveguide, and much more difficult to analyze, more research work is needed to accurately characterize printed transmission line discontinuities; some approximate results are given in reference [9].

Modal Analysis of an H-Plane Step in Rectangular Waveguide

The field analysis of most discontinuity problems is very difficult, and beyond the scope of this book. The technique of modal analysis, however, is relatively straight-forward and similar in principle to the reflection/transmission problems which were discussed in Chapters 1 and 2. In addition, modal analysis is a rigorous and versatile technique that can be applied to many coax, waveguide, and planar transmission line discontinuity problems, and lends itself well to computer implementation. We will present the technique of modal analysis by applying it to the problem of finding the equivalent circuit of an H-plane step (change in width) in rectangular waveguide.

The geometry of the *H*-plane step is shown in Figure 4.24. It is assumed that only the dominant TE_{10} mode is propagating in guide 1 (z < 0), and that such a mode is incident on the junction from z < 0. It is also assumed that no modes are propagating in guide 2, although the analysis to follow is still valid if propagation can occur in guide 2. From Section 3.3, the transverse components of the incident TE_{10} mode can then be written, for z < 0,

$$E_y^4 = \sin\frac{\pi x}{a}e^{-j\theta_1^a x},\qquad 4.85a$$

$$H_{x}^{2} = \frac{-1}{Z_{1}^{o}} \sin \frac{\pi x}{a} e^{-i\beta_{1}^{a} z}, \qquad 4.85b$$



FIGURE 4.24 Geometry of an H-plane step (change in width) in rectangular waveguide.

where

$$\beta_n^a = \sqrt{k_0^2 - \left(\frac{n\pi}{a}\right)^2} \tag{4.86}$$

is the propagation constant of the TE_{n0} mode in guide 1 (of width a), and

$$Z_n^a = \frac{k_0 \eta_0}{\beta_n^a} \tag{4.87}$$

is the wave impedance of the TE_{n0} mode in guide 1. Because of the discontinuity at z = 0 there will be reflected and transmitted waves in both guides, consisting of infinite sets of TE_{n0} modes in guides 1 and 2. Only the TE_{10} mode will propagate in guide 1, but the higher-order modes are also important in this problem because they account for stored energy, localized near z = 0. Because there is no y variation introduced by this discontinuity, TE_{nm} modes for $m \neq 0$ are not excited, nor are any TM modes. A more general discontinuity, however, may excite such modes.

The reflected modes in guide 1 may then be written, for z < 0, as

$$E_y^r = \sum_{n=1}^{\infty} A_n \sin \frac{n\pi x}{a} e^{j\beta_n^a z}, \qquad 4.88a$$

$$H_x^r = \sum_{n=1}^{\infty} \frac{A_n}{Z_n^a} \sin \frac{n\pi x}{a} e^{j\beta_n^a z}, \qquad 4.88b$$

where A_n is the unknown amplitude coefficient of the reflected TE_{n0} mode in guide 1. The reflection coefficient of the incident TE₁₀ mode is then A_1 . Similarly, the transmitted modes into guide 2 can be written, for z > 0, as

$$E_y^t = \sum_{n=1}^{\infty} B_n \sin \frac{n\pi x}{c} e^{-j\beta_n^c z}, \qquad 4.89a$$

$$H_{x}^{t} = -\sum_{n=1}^{\infty} \frac{B_{n}}{Z_{n}^{c}} \sin \frac{n\pi x}{c} e^{-j\beta_{n}^{c}z}.$$
 4.89b

where the propagation constant in guide 2 is

$$\beta_n^c = \sqrt{k_0^2 - \left(\frac{n\pi}{c}\right)^2},\tag{4.90}$$

and the wave impedance in guide 2 is

$$Z_n^c = \frac{k_0 \eta_0}{\beta_n^c}.$$
 4.91

Now at z = 0, the transverse fields (E_y, H_x) must be continuous for 0 < x < c; in addition, E_y must be zero for c < x < a because of the step. Enforcing these boundary conditions leads to the following equations:

$$E_{y} = \sin \frac{\pi x}{a} + \sum_{n=1}^{\infty} A_{n} \sin \frac{n\pi x}{a} = \begin{cases} \sum_{n=1}^{\infty} B_{n} \sin \frac{n\pi x}{c} & \text{for } 0 < x < c, \\ 0 & \text{for } c < x < a, \end{cases}$$
4.92a

$$H_x = \frac{-1}{Z_1^o} \sin \frac{\pi x}{a} + \sum_{n=1}^{\infty} \frac{A_n}{Z_n^n} \sin \frac{n\pi x}{a} = -\sum_{n=1}^{\infty} \frac{B_n}{Z_n^c} \sin \frac{n\pi x}{c} \qquad \text{for } 0 < x < c. \quad 4.92b$$

Equations (4.92a) and (4.92b) constitute a doubly infinite set of linear equations for the modal coefficients A_n and B_n . We will first eliminate the B_n s, and then truncate the resulting equation to a finite number of terms and solve for the A_n s.

Multiplying (4.92a) by $\sin(n\pi x/a)$, integrating from x = 0 to u, and using the orthogonality relations from Appendix D yields

$$\frac{a}{2}\delta_{m1} + \frac{a}{2}A_{m} = \sum_{n=1}^{\infty} B_n I_{nm} = \sum_{k=1}^{\infty} B_k I_{km},$$
4.93

where

$$I_{mn} = \int_{x=0}^{c} \sin \frac{m\pi x}{c} \sin \frac{n\pi x}{a} dx \qquad 4.94$$

is an integral that can be easily evaluated, and

$$\delta_{mn} = \begin{cases} 1 & \text{if } m = n \\ 0 & \text{if } m \neq n \end{cases}$$
4.95

is the Kronecker delta symbol. Now solve (4.92b) for B_k by multiplying (4.92b) by $\sin(k\pi x/c)$ and integrating from x = 0 to c. After using the orthogonality relations, we obtain

$$\frac{-1}{Z_1^a}I_{k1} + \sum_{n=1}^{\infty} \frac{A_n}{Z_n^a}I_{kn} = \frac{-cB_k}{2Z_k^e}.$$
4.96

Substituting B_k from (4.96) into (4.93) gives an infinite set of linear equations for the A_n s, where m = 1, 2, ..., n

$$\frac{a}{2}A_m + \sum_{n=1}^{\infty} \sum_{k=1}^{\infty} \frac{2Z_k^c I_{km} I_{kn} A_n}{cZ_n^a} = \sum_{k=1}^{\infty} \frac{2Z_k^c I_{km} I_{k1}}{cZ_1^a} - \frac{a}{2}\delta_{m1}.$$
4.97

For numerical calculation we can truncate the above summations to N terms, which will result in N linear equations for the first N coefficients, A_n . For example, let N = 1. Then (4.97) reduces to

$$\frac{a}{2}A_1 + \frac{2Z_1^c I_{11}^2}{aZ_1^a}A_1 = \frac{2Z_1^c I_{11}^2}{cZ_1^a} - \frac{a}{2}.$$
4.98

Solving for A_1 (the reflection coefficient of the incident TE₁₀ mode) gives

$$A_{\rm I} = \frac{Z_{\rm f} - Z_{\rm I}^{\rm a}}{Z_{\rm f} + Z_{\rm I}^{\rm a}}, \qquad {\rm for } N = 1.$$
 4.99

where $Z_t = 4Z_1^t l_{11}^2/ac$, which looks like an effective load impedance to guide 1. Accuracy is improved by using larger values of N, and leads to a set of equations which can be written in matrix form as

$$[Q][A] = [P], 4.100$$

where [Q] is a square $N \times N$ matrix of coefficients,

$$Q_{mn} = \frac{a}{2}\delta_{mn} + \sum_{k=1}^{N} \frac{2Z_k^c I_{km} I_{kn}}{cZ_n^a},$$
(4.10)

[P] is an $N \times 1$ column vector of coefficients given by

$$P_{m} = \sum_{k=1}^{N} \frac{2Z_{k}^{c} I_{km} I_{k1}}{cZ_{1}^{a}} - \frac{a}{2} \delta_{m1}, \qquad 4.102$$

and [A] is an $N \times 1$ column vector of the coefficients A_n . After the A_n s are found, the B_n s can be calculated from (4.96), if desired. Equations (4.100)–(4.102) lend themselves well to computer implementation.

Figure 4.25 shows the results of such a calculation. If the width, c, of guide 2 is such that all modes are cutoff (evanescent), then no real power can be transmitted into guide 2, and all the incident power is reflected back into guide 1. The evanescent fields on both sides of the discontinuity store reactive power, however, which implies that the step discontinuity and guide 2 beyond the discontinuity look like a reactance (in this case an inductive reactance) to an incident TE_{10} mode in guide 1. Thus the equivalent circuit of the *H*-plane step looks like an inductor at the z = 0 plane of guide 1, as shown in Figure 4.22e. The equivalent reactance can be found from the reflection coefficient A_1 (after solving (4.100)) as

$$X = -jZ_1^a \frac{1+A_1}{1-A_1}.$$
4.103

Figure 4.25 shows the normalized equivalent inductance versus the ratio of the guide widths, c/a, for a free-space wavelength $\lambda = 1.4o$ and for N = 1, 2, and 10 equations. The modal analysis results are compared to calculated data from reference [8]. Note



FIGURE 4.25 Equivalent inductance of an H-plane asymmetric step.

that the solution converges very quickly (because of the fast exponential decay of the higher-order evanescent modes), and that the result using just two modes is very close to the data of reference [8].

The fact that the equivalent circuit of the *H*-plane step looks inductive is a result of the actual value of the reflection coefficient, A_1 , but we can verify this result by computing the complex power flow into the evanescent modes on either side of the discontinuity. For example, the complex power flow into guide 2 can be found as

$$P = \int_{x=0}^{c} \int_{y=0}^{b} \overline{E} \times \widehat{H}^{*} \Big|_{x=0^{+}} \cdot \hat{z} \, dx \, dy$$

$$= -b \int_{x=0}^{c} E_{y} H_{x}^{*} \, dx$$

$$= -b \int_{x=0}^{c} \left[\sum_{n=1}^{\infty} B_{n} \sin \frac{n\pi x}{c} \right] \left[-\sum_{m=1}^{\infty} \frac{B_{m}^{*}}{Z_{m}^{c*}} \sin \frac{m\pi x}{c} \right] dx$$

$$= \frac{bc}{2} \sum_{n=1}^{\infty} \frac{|B_{n}|^{2}}{Z_{n}^{c*}}$$

$$= \frac{jbc}{2k_{0}\eta_{0}} \sum_{n=1}^{\infty} |B_{n}|^{2} |\beta_{n}^{*}|, \qquad 4.104$$

where the orthogonality property of the sine functions was used, as well as (4.89)-(4.91). Equation (4.104) shows that the complex power flow into guide 2 is purely inductive. A similar result can be derived for the evanescent modes in guide 1; this is left as a problem.

POINT OF INTEREST: Microstrip Discontinuity Compensation

Because a microstrip circuit is easy to fabricate and allows the convenient integration of passive and active components, many types of microwave circuits and subsystems are made in microstrip form. One problem with microstrip circuits (and other planar circuits), however, is that the inevitable discontinuities at bends, step changes in widths, and junctions can cause a degradation in circuit performance. This is because such discontinuities introduce parasitic reactances that can lead to phase and amplitude errors, input and output mismatch, and possibly sptrious coupling. One approach for eliminating such effects is to construct an equivalent circuit for the discontinuity (perhaps by measurement), including it in the design of the circuit, and compensating for its effect by adjusting other circuit parameters (such as line lengths and characteristic impedances, or tuning stubs). Another approach is to minimize the effect of a discontinuity by compensating the discontinuity directly, often by chamfering or mitering the conductor.

Consider the case of a bend in a microstrip line. The straightforward right-angle bend shown below has a parasitic discontinuity capacitance caused by the increased conductor area near the bend. This effect could be eliminated by making a smooth, "swept" bend with a radius $r \ge 3W$, but this takes up more space. Alternatively, the right-angle bend can be compensated by mitering the corner, which has the effect of reducing the excess capacitance at the bend. As shown below, this technique can be applied to bends of arbitrary angle. The optimum value of the miter length, a, depends on the characteristic impedance and the bend angle, but a value of a = 1.8W is offea used in practice.

The technique of mitering can also be used to compensate step and T-junction discontinuities, as shown below.



Reference: T. C. Edwards, Foundations for Microwave Circuit Design, Wiley, New York, 1981.

4.7 EXCITATION OF WAVEGUIDES-ELECTRIC AND MAGNETIC CURRENTS

So far we have considered the propagation, reflection, and transmission of guided waves in the absence of sources, but obviously the waveguide or transmission line must be coupled to a generator or some other source of power. For TEM or quasi-TEM lines, there is usually only one propagating mode that can be excited by a given source, although there may be reactance (stored energy) associated with a given feed. In the waveguide case, it may be possible for several propagating modes to be excited, along with evanescent modes that store energy. In this section we will develop a formalism for determining the excitation of a given waveguide mode due to an arbitrary electric or magnetic current source. This theory can then be used to find the excitation and input impedance of probe and loop feeds and, in the next section, to determine the excitation of waveguides by apertures.

Current Sheets That Excite Only One Waveguide Mode

Consider an infinitely long rectangular waveguide with a transverse sheet of electric surface current density at z = 0, as shown in Figure 4.26. First assume that this current has \hat{x} and \hat{y} components given as

$$\bar{J}_{s}^{TE}(x,y) = -\hat{x}\frac{2A_{mn}^{+}n\pi}{b}\cos\frac{m\pi x}{a}\sin\frac{n\pi y}{b} + \hat{y}\frac{2A_{mn}^{+}m\pi}{a}\sin\frac{m\pi x}{a}\cos\frac{n\pi y}{b}, \quad 4.105$$



FIGURE 4.26 An infinitely long rectangular waveguide with surface current densities at z = 0.

We will show that such a current excites a TE_{mn} waveguide mode traveling away from the current source in both the -z and -z directions.

From Table 3.2, the transverse fields for positive and negative traveling TE_{mm} waveguide modes can be written as

$$E_x^{\pm} = Z_{\text{TE}} \left(\frac{n\pi}{b}\right) A_{mn}^{\pm} \cos \frac{m\pi x}{a} \sin \frac{n\pi y}{b} e^{\mp j \beta x}, \qquad 4.106n$$

$$E_y^{\pm} = -Z_{\text{TE}}\left(\frac{m\pi}{a}\right) A_{mn}^{\pm} \sin \frac{m\pi x}{a} \cos \frac{n\pi y}{b} e^{\pm j\beta z}, \qquad 4.106b$$

$$H_x^{\pm} = \pm \left(\frac{m\pi}{a}\right) A_{mn}^{\pm} \sin \frac{m\pi x}{a} \cos \frac{n\pi y}{b} e^{\mp j\beta x}, \qquad 4.106c$$

$$H_y^{\pm} = \pm \left(\frac{n\pi}{b}\right) A_{mn}^{\pm} \cos \frac{m\pi x}{a} \sin \frac{n\pi y}{b} e^{\pm j\beta z}.$$
 4.106d

where the \pm notation refers to waves traveling in the +z direction or -z direction, with amplitude coefficients A_{mn}^+ and A_{mn}^- , respectively.

From (1.36) and (1.37), the following boundary conditions must be satisfied at z = 0:

$$(\dot{E}^+ - \bar{E}^-) \times \dot{z} = 0.$$
 4.107*a*

$$(\bar{H}^+ - \bar{H}^-) = \tilde{J}_{\mu}$$
. 4.107b

Equation (4.107a) states that the transverse components of the electric field must be continuous at z = 0, which when applied to (4.106a) and (4.106b) gives

$$A_{mn}^{+} = A_{mn}^{-}.$$
 4.108

Equation (4.107b) states that the discontinuity in the transverse magnetic field is equal to the electric surface current density. Thus, the surface current density at z = 0 must be

$$\hat{J}_{s} = \hat{y}(H_{x}^{+} - H_{x}^{-}) - \hat{x}(H_{y}^{+} - H_{y}^{-})$$

$$= -\hat{x}\frac{2A_{mn}^{+}n\pi}{b}\cos\frac{m\pi x}{a}\sin\frac{n\pi y}{b} + \hat{y}\frac{2A_{mn}^{-}m\pi}{a}\sin\frac{m\pi x}{a}\cos\frac{n\pi y}{b}, \quad 4.109$$

where (4.108) was used. This current is seen to be the same as the current of (4.105), which shows, by the uniqueness theorem, that such a current will excite only the TE_{mn} mode propagating in each direction, since Maxwell's equations and all boundary conditions are satisfied.

The analogous electric current that excites only the TM_{mn} mode can be shown to be

$$\bar{J}_{s}^{\text{TM}}(x,y) = \hat{x} \frac{2B_{mn}^{+}m\pi}{a} \cos \frac{m\pi x}{a} \sin \frac{n\pi y}{b} + \hat{y} \frac{2B_{mn}^{+}n\pi}{b} \sin \frac{m\pi x}{a} \cos \frac{n\pi y}{b}.$$
 4.110

It is left as a problem to verify that this current excites TM_{mn} modes that satisfy the appropriate boundary conditions.

Similar results can be derived for magnetic surface current sheets. From (1.36) and (1.37) the appropriate boundary conditions are

$$(\bar{E}^+ - \bar{E}^-) \times \hat{z} = \bar{M}_s, \qquad 4.111a$$

$$\hat{z} \times (\bar{H}^+ - \bar{H}^-) = 0.$$
 4.111b

For a magnetic current sheet at z = 0, the TE_{mn} waveguide mode fields of (4.106) must now have continuous H_x and H_y field components, due to (4.111b). This results in the condition that

$$A_{mn}^{+} = -A_{mn}^{-}.$$
 4.112

Then applying (4.111a) gives the source current as

$$\bar{M}_{s}^{\text{TE}} = \frac{-\hat{x}2Z_{\text{TE}}A_{mn}^{+}m\pi}{a}\sin\frac{m\pi x}{a}\cos\frac{n\pi y}{b} - \hat{y}\frac{2Z_{\text{TE}}A_{mn}^{+}n\pi}{b}\cos\frac{m\pi x}{a}\sin\frac{n\pi y}{b}.$$
 4.113

The corresponding magnetic surface current that excites only the TM_{mn} mode can be shown to be

$$\bar{M}_{s}^{\text{TM}} = \frac{-\hat{x}2B_{mn}^{+}n\pi}{b} \sin \frac{m\pi x}{a} \cos \frac{n\pi y}{b} + \frac{\hat{y}2B_{mn}^{+}m\pi}{a} \cos \frac{m\pi x}{a} \sin \frac{n\pi y}{b}.$$
 4.114

These results show that a single waveguide mode can be selectively excited, to the exclusion of all other modes, by either an electric or magnetic current sheet of the appropriate form. In practice, however, such currents are very difficult to generate, and are usually only approximated with one or two probes or loops. In this case many modes may be excited, but usually most of these modes are evanescent.

Mode Excitation from an Arbitrary Electric or Magnetic Current Source

We now consider the excitation of waveguide modes by an arbitrary electric or magnetic current source [4]. With reference to Figure 4.27, first consider an electric





current source \bar{J} located between two transverse planes at z_1 and z_2 , which generates the fields \bar{E}^+ , \bar{H}^+ traveling in the +z direction, and the fields \bar{E}^- , \bar{H}^- traveling in the -z direction. These fields can be expressed in terms of the waveguide modes as follows:

$$\bar{E}^{+} = \sum_{n} A_{n}^{+} \bar{E}_{n}^{+} = \sum_{n} A_{n}^{+} (\bar{c}_{n} + \hat{z} e_{zn}) e^{-j \#_{n} \hat{z}}, \qquad z > z_{2}, \qquad 4.115a$$

$$\bar{H}^{+} = \sum_{n} A_{n}^{+} \bar{H}_{n}^{+} = \sum_{n} A_{n}^{+} (\bar{h}_{n} + \hat{z} h_{zn}) e^{-j \beta_{n} z}, \qquad z > z_{3} \qquad 4.115 h$$

$$\vec{E}^{-} = \sum_{n} A_{n}^{-} \bar{E}_{n} = \sum_{n} A_{n} (\bar{e}_{n} - \hat{z} e_{zn}) e^{j\beta_{n} z}, \qquad \dot{z} < z_{b} \qquad 4.115c$$

$$\hat{H}^{-} = \sum_{n} A_{n}^{-} \hat{H}_{n}^{-} = \sum_{n} A_{n}^{-} (-\tilde{h}_{n} + \hat{z} h_{xn}) e^{j \mathcal{B}_{n} z}, \qquad z < z_{b}, \qquad 4.115d$$

where the single index n is used to represent any possible TE or TM mode. For a given current \hat{J} , we can determine the unknown amplitude A_0^+ by using the Lorentz reciprocity theorem of (1.155) with $\hat{M}_1 = \hat{M}_2 = 0$ (since here we are only considering an electric current source),

$$\oint_{S} (\bar{E}_1 \times \bar{H}_2 - \bar{E}_2 \times \bar{H}_1) \cdot d\bar{s} = \int_{V} (\bar{E}_2 \cdot \bar{J}_1 - \bar{E}_1 \cdot \bar{J}_2) dv,$$

where S is a closed surface enclosing the volume V, and \vec{E}_i , \vec{H}_i are the fields due to the current source \vec{J}_i (for i = 1 or 2).

To apply the reciprocity theorem to the present problem, we let the volume V be the region between the waveguide walls and the transverse cross-section planes at z_1 and z_2 . Then let $\bar{E}_1 = \bar{E}^{\pm}$ and $\bar{H}_1 = \bar{H}^{\pm}$, depending on whether $z \ge z_2$, or $z \le z_1$, and let \bar{E}_{21}, \bar{H}_2 be the *n*th waveguide mode traveling in the negative z direction:

$$\begin{split} \bar{E}_2 &= \bar{E}_n^+ = (\bar{e}_n - \hat{z} e_{zn}) e^{j\beta_n z}, \\ \bar{H}_2 &= \bar{H}_n^- = (-\bar{h}_n + \hat{z} h_{zn}) e^{j\beta_n z} \end{split}$$

Substitution into the above form of the reciprocity theorem gives, with $\bar{J}_1 = \bar{J}$ and $\tilde{J}_2 = 0$,

$$\oint_{S} (\bar{E}^{\pm} \times \bar{H}_{n}^{-} - \bar{E}_{n}^{-} \times \bar{H}^{\pm}) \cdot d\bar{s} = \int_{V} \bar{E}_{n}^{-} \cdot \tilde{J} dv. \qquad 4.116$$

The portion of the surface integral over the waveguide walls vanishes because the tangential electric field is zero there: that is, $\overline{E} \times \overline{H} \cdot \hat{z} = \overline{H} \cdot (\hat{z} \times \overline{E}) = 0$ on the waveguide walls. This reduces the integration to the guide cross section, S_0 , at the planes z_1 and z_2 . In addition, the waveguide modes are orthogonal over the guide cross section:

$$\int_{S_0} \bar{E}_m^{\pm} \times \bar{H}_n^{\pm} \cdot d\bar{s} = \int_{S_0} (\bar{e}_m \pm \hat{z} e_{zn}) \times (\pm \bar{h}_n + \hat{z} h_{zn}) \cdot \hat{z} \, ds$$
$$= \pm \int_{S_0} \bar{e}_m \times \bar{h}_n \cdot \hat{z} \, ds = 0. \quad \text{for } m \neq n. \quad 4.117$$

Using (4.115) and (4.117) then reduces (4.116) to

$$\begin{aligned} A_n^+ \int_{z_2} (\bar{E}_n^+ \times \bar{H}_n^- - \bar{E}_n^- \times \bar{H}_n^+) \cdot d\bar{s} + A_n^- \int_{z_1}^{z_1} (\bar{E}_n^- \times \bar{H}_n^- - \bar{E}_n^- \times \bar{H}_n^-) \cdot d\bar{s} \\ &= \int_V \bar{E}_n^- \cdot \bar{J} dv. \end{aligned}$$

Since the second integral vanishes, this further reduces to

$$\begin{aligned} A_n^+ \int_{z_2} [(\bar{e}_n + \hat{z}e_{zn}) \times (-\bar{h}_n + \hat{z}h_{zn}) - (\bar{e}_n - \hat{z}e_{zn}) \times (\bar{h}_n + \hat{z}h_{zn})] \cdot \hat{z} \, ds \\ &= -2A_n^+ \int_{z_2} \bar{e}_n \times \bar{h}_n \cdot \hat{z} \, ds = \int_V \bar{E}_n^- \cdot \bar{J} \, dv, \end{aligned}$$

$$A_n^+ = \frac{-1}{P_n} \int_V \bar{E}_n^- \cdot \bar{J} dv = \frac{-1}{P_n} \int_V (\bar{e}_n - \hat{z} e_{zn}) \cdot \bar{J} e^{j\beta_n z} dv, \qquad 4.118$$

where

Øf

$$P_n = 2 \int_{S_0} \bar{e}_n \times \bar{h}_n \cdot \hat{z} \, ds. \tag{4.119}$$

is a normalization constant proportional to the power flow of the nth mode.

By repeating the above procedure with $\bar{E}_2 = \bar{E}_n^+$ and $\bar{H}_2 = \bar{H}_n^+$, the amplitude of the negatively traveling waves can be derived as

$$A_{n}^{-} = \frac{-1}{P_{n}} \int_{V} \bar{E}_{n}^{+} \cdot \hat{J} \, dv = \frac{-1}{P_{n}} \int_{V} (\bar{e}_{n} + \hat{e}_{zn}) \cdot \bar{J} e^{-j\beta_{n}z} \, dv.$$
 4.120

The above results are quite general, being applicable to any type of waveguide (including planar lines such as stripline and microstrip), where modal fields can be defined. Example 4.8 applies this theory to the problem of a probe-fed rectangular waveguide.



EXAMPLE 4.8 Probe-Fed Rectangular Waveguide

For the probe-fed rectangular waveguide shown in Figure 4.28, determine the amplitudes of the forward and backward traveling TE_{10} modes, and the input resistance seen by the probe. Assume that the TE_{10} mode is the only propagating mode.

Solution

If the current probe is assumed to have an infinitesimal diameter, the source volume current density \overline{J} can be written as

$$\bar{J}(x,y,z) = I_0 \delta\left(x-\frac{a}{2}\right) \delta(z)\hat{y}, \quad \text{ for } 0 \leq y \leq b.$$



FIGURE 4.28 A uniform current probe in a rectangular waveguide.

From Chapter 3 the TE₁₀ modal fields can be written as

$$ar{e}_1 = \hat{y}\sinrac{\pi x}{a},$$

 $ar{h}_1 = rac{-\hat{x}}{Z_1}\sinrac{\pi x}{a}.$

where $Z_1 = k_0 \eta_0 / \beta_1$ is the TE₁₀ wave impedance. From (4.119) the normalization constant P_1 is,

$$P_{1} = \frac{2}{Z_{1}} \int_{x=0}^{a} \int_{y=0}^{b} \sin^{2} \frac{\pi x}{a} dx \, dy = \frac{ab}{Z_{1}}.$$

Then from (4.118) the amplitude A_1^+ is

$$A_1^{\pm} = \frac{-1}{P_1} \int_V \sin \frac{\pi x}{a} e^{i\beta_1 z} I_0 \delta\left(x - \frac{a}{2}\right) \delta(z) dx \, dy \, dz = \frac{-I_0 b}{P_1} = \frac{-Z_1 I_0}{a}.$$

Similarly.

$$A_{\perp}^{-}=\frac{-Z_{\perp}I_{\emptyset}}{u},$$

If the TE₁₀ mode is the only propagating mode in the waveguide, then this mode carries all of the average power, which can be calculated for real Z_1 as

$$P = \frac{1}{2} \int_{S_0} \bar{E}^+ \times \bar{H}^{+*} \cdot d\bar{s} + \frac{1}{2} \int_{S_0} \bar{E}^- \times \bar{H}^{-*} \cdot d\bar{s}$$

= $\int_{S_0} \bar{E}^+ \times \bar{H}^{+*} \cdot d\bar{s}$
= $\int_{x=0}^a \int_{y=0}^b \frac{|A_1^+|^2}{Z_1} \sin^2 \frac{\pi x}{a} \, dx \, dy$
= $\frac{ab|A_1^+|^2}{2Z_1}$.

If the input resistance seen looking into the probe is R_{in} , and the terminal current is I_0 , then $P = I_0^2 R_{in}/2$, so that the input resistance is

$$R_{\rm in} = \frac{2P}{I_0^2} = \frac{ab|A_1^+|^2}{I_0^2 Z_1} = \frac{bZ_1}{a},$$

which is real for real Z_1 (corresponding to a propagating TE₁₀ mode).

A similar derivation can be carried out for a magnetic current source \overline{M} . This source will also generate positively and negatively traveling waves which can be expressed as a superposition of waveguide modes, as in (4.115). For $\overline{J}_1 = \overline{J}_2 = 0$, the reciprocity theorem of (1.155) reduces to

$$\oint_{S} (\bar{E}_1 \times \bar{H}_2 - \bar{E}_2 \times \bar{H}_1) \cdot d\bar{s} = \int_{V} (\bar{H}_1 \cdot \bar{M}_2 - \bar{H}_2 \cdot \bar{M}_1) dv.$$

$$4.121$$

By following the same procedure as for the electric current case, the excitation coefficients of the nth waveguide mode can be derived as

$$A_{n}^{+} = \frac{1}{P_{n}} \int_{V} \bar{H}_{n}^{-} \cdot \bar{M} \, dv = \frac{1}{P_{n}} \int_{V} (-\bar{h}_{n} + \hat{z}h_{zn}) \cdot \bar{M} e^{j\beta_{n}z} \, dv, \qquad 4.122$$

$$A_{n}^{-} = \frac{1}{P_{n}} \int_{V} \bar{H}_{n}^{+} \cdot \bar{M} \, dv = \frac{1}{P_{n}} \int_{V} (\bar{h}_{n} + \hat{z}h_{zn}) \cdot \bar{M}e^{-j\beta_{n}z} \, dv.$$
 4.123

where P_n is defined in (4.119).

EXAMPLE 4.9 Loop-Fed Rectangular Waveguide

Find the excitation coefficient of the forward traveling TE_{10} mode generated by the loop in the end wall of the waveguide shown in Figure 4.29a.

Solution

By image theory, the half-loop of current I_0 on the end wall of the waveguide can be replaced by a full loop of current I_0 , without the end wall, as shown in Figure 4.29b. Assuming that the current loop is very small, it is equivalent to a magnetic dipole moment.

$$\bar{P}_m = \hat{x} I_0 \pi r_0^2 \delta\left(x - \frac{a}{2}\right) \delta\left(y - \frac{b}{2}\right) \delta(z).$$

Now since $\nabla \times \bar{E} = -j\omega\bar{B} - \bar{M} = -j\omega\mu_0\bar{H} - j\omega\mu_0\bar{P}_m - \bar{M}$, a magnetic polarization current \bar{P}_m can be related to an equivalent magnetic current density \bar{M} as

$$\bar{M} = j\omega\mu_0\bar{P}_m.$$

Thus, the loop can be represented as a magnetic current density:

$$\bar{M} = \hat{x} j \omega \mu_0 I_0 \pi \tau_0^2 \delta \left(x - \frac{a}{2} \right) \delta \left(y - \frac{b}{2} \right) \delta(z) \, \mathrm{V/m^2}.$$


FIGURE 4.29 Application of image theory to a loop in the end wall of a rectangular waveguide. (a) Original geometry. (b) Using image theory to replace the end wall with the image of the half-loop.

If we define the modal \tilde{h}_1 field as

$$\vec{h}_{\perp} = \frac{-\vec{x}}{Z_1} \sin \frac{\pi x}{a}.$$

then (4.122) gives the forward wave excitation coefficient A_1^+ as

$$A_{1}^{+} = \frac{1}{P_{1}} \int_{V} -\bar{h}_{1} \cdot \bar{M} \, dv = \frac{jk_{0}\eta_{0}I_{0}\pi r_{0}^{2}}{ab}, \qquad \qquad \bigcirc$$

4.8

EXCITATION OF WAVEGUIDES—APERTURE COUPLING

Besides the probe and loop feeds of the previous section, waveguides and other transmission lines can also be coupled through small apertures. One common application of such coupling is in directional couplers and power dividers, where power from one guide is coupled to another guide through small apertures in a common wall. Figure 4.30 shows a variety of waveguide and other transmission line configurations where aperture coupling can be employed. We will first develop an intuitive explanation for the fact that a small aperture can be represented as an infinitesimal electric and/or an infinitesimal magnetic dipole, then we will use the results of Section 4.7 to find the fields generated by these equivalent currents. Our analysis will be somewhat phenomenological [4], [10]; a more advanced theory of aperture coupling based on the equivalence theorem can be found in reference [11].

Consider Figure 4.31a, which shows the normal electric field lines near a conducting wall (the tangential electric field is zero near the wall). If a small aperture is cut into the



FIGURE 4.30 Various waveguide and other transmission line configurations using aperture coupling. (a) Coupling between two waveguides via an aperture in the common broad wall. (b) Coupling to a waveguide cavity via an aperture in a transverse wall. (c) Coupling between two microstrip lines via an aperture in the common ground plane. (d) Coupling from a waveguide to a stripline via an aperture.





Illustrating the development of equivalent electric and magnetic polarization currents at an aperture in a conducting wall. (a) Normal electric field at a conducting wall. (b) Electric field lines around an aperture in a conducting wall. (c) Electric field lines around electric polarization currents normal to a conducting wall. (d) Magnetic field lines near a conducting wall. (e) Magnetic field lines near an aperture in a conducting wall. (f) Magnetic field lines near magnetic polarization currents parallel to a conducting wall.

conductor the electric field lines will fringe through and around the aperture as shown in Figure 4.31b. Now consider Figure 4.31c, which shows the fringing field lines around two infinitesimal electric polarization currents, \tilde{P}_{e} , normal to a conducting wall (without an aperture). The similarity of the field lines of Figures 4.31c and 4.31b suggests that an aperture excited by a normal electric field can be represented by two oppositely directed infinitesimal electric polarization currents. \tilde{P}_{e} , normal to the closed conducting wall. The strength of this polarization current is proportional to the normal electric field, thus,

$$P_{e} = \epsilon_{0} \alpha_{e} \hat{n} E_{n} \delta(x - x_{0}) \delta(y - y_{0}) \delta(z - z_{0}), \qquad 4.124$$

where the proportionality constant α_r is defined as the *electric polarizability* of the aperture, and (x_0, y_0, z_0) are the coordinates of the center of the aperture.

Similarly, Figure 4.31e shows the fringing of tangential magnetic field lines (the normal magnetic field is zero at the conductor) near a small aperture. Since these field lines are similar to those produced by two magnetic polarization currents located parallel to the conducting wall, (as shown in Figure 4.31f), we can conclude that the aperture can be replaced by two oppositely directed infinitesimal polarization currents, \bar{P}_{in} , where

$$\bar{P}_m = -\alpha_m \bar{H}_t \delta(x - x_0) \delta(y - y_0) \delta(z - z_0). \qquad 4.125$$

In (4.125), α_m is defined as the *mugnetic polarizability* of the aperture.

The electric and magnetic polarizabilities are constants that depend on the size and shape of the aperture, and have been derived for a variety of simple shapes [3], [10], [11]. The polarizabilities for circular and rectangular apertures, which are probably the most commonly used shapes, are given in Table 4.3.

We now show that the electric and magnetic polarization currents, \tilde{P}_e and \tilde{P}_m , can be related to electric and magnetic current sources, \tilde{J} and \tilde{M} , respectively. From Maxwell's equations (1.27a) and (1.27b) we have

$$\nabla \times \tilde{E} = -j\omega\mu \bar{H} - \bar{M},$$
 4.126*a*

$$\nabla \times \bar{H} = j\omega\epsilon \bar{E} + \bar{J}.$$
 4.126b

Then using (1.15) and (1.23), which define \bar{P}_e and \bar{P}_m , we obtain

$$\nabla \times \bar{E} = -j\omega\mu_0\bar{H} - j\omega\mu_0\bar{P}_m - \bar{M}. \qquad 4.127a$$

$$\nabla \times \bar{H} = j\omega\epsilon_0 \bar{E} + j\omega \bar{P}_e + \bar{J}.$$
4.127b

TABLE 4.3	Electric	and	Magnetic	Polarizations
-----------	----------	-----	----------	----------------------

Арепште Shape	α_c	Qт
Round hole	$\frac{2r_0^3}{3}$	$\frac{4r_0^3}{3}$
Rectangular slot	$\frac{\pi \hat{t} d^2}{16}$	$\frac{\pi \ell d^2}{16}$
($ar{H}$ across slot)		10

Thus, since \tilde{M} has the same role in these equations as $j\omega\mu_0 \tilde{P}_m$, and \tilde{J} has the same role as $j\omega \tilde{P}_e$, we can define equivalent currents as

$$\bar{J} = j\omega \bar{P}_e, \qquad 4.128a$$

$$ar{M} = j\omega\mu_0ar{P}_m.$$
 4.1286

These results then allow us to use the formulas of (4.118), (4.120), (4.122), and (4.123) to compute the fields from these currents.

The above theory is approximate because of various assumptions involved in the evaluation of the polarizabilities, but generally gives reasonable results for apertures which are small (where the term *small* implies small relative to an electrical wavelength), and not located too close to edges or corners of the guide. In addition, it is important to realize that the equivalent dipoles given by (4.124) and (4.125) radiate in the presence of the conducting wall to give the fields transmitted through the aperture. The fields on the input side of the conducting wall are also affected by the presence of the aperture, and this effect is accounted for by the equivalent dipoles on the incident side of the conductor (which are the negative of those on the output side). In this way, continuity of tangential fields is preserved across the aperture. In both cases, the presence of the (closed) conducting wall can be accounted for by using image theory to remove the wall and double the strength of the dipoles. These details will be clarified by applying this theory to apertures in transverse and broad walls of waveguides.

Coupling Through an Aperture in a Transverse Waveguide Wall

Consider a small circular aperture centered in the transverse wall of a waveguide, as shown in Figure 4.32a. Assume that only the TE_{10} mode propagates in the guide, and that such a mode is incident on the transverse wall from z < 0. Then, if the aperture is assumed to be closed, as in Figure 4.32b, the standing wave fields in the region z < 0 can be written as

$$E_y = A(e^{-j\beta z} - e^{j\beta z})\sin\frac{\pi x}{a}, \qquad 4.129a$$

$$H_{x} = \frac{-A}{Z_{10}} (e^{-j\beta z} + e^{j\beta z}) \sin \frac{\pi x}{a}, \qquad 4.129b$$

where β and Z_{10} are the propagation constant and wave impedance of the TE₁₀ mode. From (4.124) and (4.125) we can determine the equivalent electric and magnetic polarization currents from the above fields as

$$\begin{split} \vec{P}_{e} &= \hat{z}\epsilon_{0}\alpha_{e}E_{z}\delta\left(x-\frac{a}{2}\right)\delta\left(y-\frac{b}{2}\right)\delta(z) = 0, \quad 4.130a\\ \vec{P}_{m} &= -\hat{x}\alpha_{m}H_{x}\delta(x-\frac{a}{2})\delta\left(y-\frac{b}{2}\right)\delta(z)\\ &= \hat{x}\frac{2A\alpha_{m}}{Z_{10}}\delta\left(x-\frac{a}{2}\right)\delta\left(y-\frac{b}{2}\right)\delta(z), \quad 4.130b \end{split}$$



FIGURE 4.32 Applying small-hole coupling theory and image theory to the problem of an aperture in the transverse wall of a waveguide. (a) Geometry of a circular aperture in the transverse wall of a waveguide, (b) Fields with aperture closed, (c) Fields with aperture open. (d) Fields with aperture closed and replaced with equivalent dipoles. (e) Fields radiated by equivalent dipoles for z < 0; wall removed by image theory. (f) Fields radiated by equivalent dipoles for z > 0; wall removed by image theory.

since $E_z = 0$ for a TE mode. Now, by (4.128b), the magnetic polarization current \vec{P}_m is equivalent to a magnetic current density,

$$\bar{M} = j\omega\mu_0\bar{P}_m = \hat{x}\frac{2j\omega\mu_0A\alpha_m}{Z_{10}}\delta\left(x-\frac{a}{2}\right)\delta\left(y-\frac{b}{2}\right)\delta(z), \qquad 4.131$$

242 Chapter 4: Microwave Network Analysis

As shown in Figure 4.32d, the fields scattered by the aperture are considered as being produced by the equivalent currents \bar{P}_m and $-\bar{P}_m$ on either side of the closed wall. The presence of the conducting wall is easily accounted for using image theory, which has the effect of doubling the dipole strengths and removing the wall, as depicted in Figure 4.32e (for z < 0) and Figure 4.32f (for z > 0). Thus the coefficients of the transmitted and reflected waves caused by the equivalent aperture currents can be found by using (4.131) in (4.122) and (4.123) to give

$$A_{10}^{+} = \frac{-1}{P_{10}} \int \bar{h}_{10} \cdot (2j\omega\mu_0 \bar{P}_m) dv = \frac{4jA\omega\mu_0\alpha_m}{abZ_{10}} = \frac{4jA\beta\alpha_m}{ab}, \qquad 4.132a$$

$$A_{10}^{-} = \frac{-1}{P_{10}} \int \bar{h}_{10} \cdot (-2j\omega\mu_0 \bar{P}_m) dv = \frac{4jA\omega\mu_0\alpha_m}{abZ_{10}} = \frac{4jA\beta\alpha_m}{ab}, \qquad 4.132b$$

since $\bar{h}_{10} = (-\hat{x}/Z_{10}) \sin(\pi x/a)$, and $P_{10} = ab/Z_{10}$. The magnetic polarizability α_m is given in Table 4.3. The complete fields can now be written as

$$E_y = [Ae^{-j\beta z} + (A_{10}^- - A)e^{j\beta z}]\sin\frac{\pi x}{a}, \qquad \text{for } z < 0. \qquad 4.133a$$

$$H_x = \frac{1}{Z_{10}} \left[-Ae^{-j\beta z} + (A_{10}^- - A)e^{j\beta z} \right] \sin \frac{\pi x}{a}, \quad \text{for } z < 0, \quad 4.133b$$

and

$$E_y = A_{10}^+ e^{-j\beta z} \sin \frac{\pi x}{a},$$
 for $z > 0,$ 4.134a

$$H_x = \frac{-A_{10}^+}{Z_{10}} e^{-j\beta z} \sin \frac{\pi x}{a}, \quad \text{for } z > 0. \quad 4.134b$$

Then the reflection and transmission coefficients can be found as

$$\Gamma = \frac{A_{10}^{\sim} - A}{A} = \frac{4j\beta\alpha_m}{ab} - 1, \qquad 4.135a$$

$$T = \frac{A_{10}^+}{A} = \frac{4j\beta\alpha_m}{ab}.$$
 4.135b

since $Z_{10} = k_0 \eta_0 / \beta$. Note that $|\Gamma| > 1$; this physically unrealizable result (for a passive network) is an artifact of the approximations used in the above theory. An equivalent circuit for this problem can be obtained by comparing the reflection coefficient of (4.135a) with that of the transmission line with a normalized shunt susceptance, *jB*, shown in Figure 4.33. The reflection coefficient seen looking into this line is

$$\Gamma = \frac{1 - y_{\rm m}}{1 + y_{\rm in}} = \frac{1 - (1 + jB)}{1 + (1 + jB)} = \frac{-jB}{2 + jB}.$$

If the shunt susceptance is very large (low impedance), Γ can be approximated as

$$\Gamma = \frac{-1}{1+(2/jB)} \simeq -1 - j\frac{2}{B}$$



FIGURE 4.33 Equivalent circuit of the aperture in a transverse waveguide wall.

Comparison with (4.135a) suggests that the aperture is equivalent to a normalized inductive susceptance,

$$B = rac{-ab}{2etalpha_m},$$

Coupling Through an Aperture in the Broad Wall of a Waveguide

Another configuration for aperture coupling is shown in Figure 4.34, where two parallel waveguides share a common broad wall and are coupled with a small centered aperture. We will assume that a TE_{10} mode is incident from z < 0 in the lower guide (guide 1), and compute the fields coupled to the upper guide. The incident fields can be written as

$$E_y = A \sin \frac{\pi x}{a} e^{-y\beta \hat{z}}, \qquad 4.136a$$

$$H_{\#} = \frac{-A}{Z_{10}} \sin \frac{\pi x}{a} e^{-j\beta z},$$
 4.136b

The excitation field at the center of the aperture at (x = a/2, y = b, z = 0) is then

$$E_y = A. 4.137a$$

$$H_x = \frac{-A}{Z_{10}}.$$
 (4.137b)

(If the aperture were not centered at x = a/2, the H_{z} field would be nonzero and would have to be included.)



FIGURE 4.34 Two parallel wavegoides coupled through an aperture in a common broad wall.

Now from (4.124), (4.125), and (4.128), the equivalent electric and magnetic dipoles for coupling to the fields in the upper guide are

$$J_{y} = j\omega\epsilon_{0}\alpha_{e}A\delta\left(x-\frac{a}{2}\right)\delta(y-b)\delta(z), \qquad 4.138a$$

$$M_x = \frac{j\omega\mu_0\alpha_m A}{Z_{10}}\delta\left(x - \frac{a}{2}\right)\delta(y - b)\delta(z).$$
4.138b

Note that in this case we have excited both an electric and a magnetic dipole. Now let the fields in the upper guide be expressed as

$$E_{\mathbf{y}}^{-} = A^{-} \sin \frac{\pi x}{a} e^{+j\beta z}, \qquad \text{for } z < 0, \qquad \qquad 4.139a$$

$$H_x^- = \frac{A^-}{Z_{10}} \sin \frac{\pi x}{a} e^{+j\beta z}, \qquad \text{for } z < 0.$$
 4.139b

$$E_{y}^{+} = A^{+} \sin \frac{\pi x}{a} e^{-j\beta z},$$
 for $z > 0,$ 4.140 a

$$H_x^+ = \frac{-A^+}{Z_{10}} \sin \frac{\pi x}{\alpha} e^{-j\beta z}, \quad \text{for } z > 0, \qquad 4.140b$$

where A^+ . A^- are the unknown amplitudes of the forward and backward traveling waves in the upper guide, respectively.

By superposition, the total fields in the upper guide due to the electric and magnetic currents of (4.138) can be found from (4.118) and (4.122) for the forward wave as

$$A^{+} = \frac{-1}{P_{10}} \int_{V} (E_{y}^{-} J_{y} - H_{x}^{-} M_{x}) dx = \frac{-j\omega A}{P_{10}} \left(\epsilon_{0} \alpha_{e} - \frac{\mu_{0} \alpha_{m}}{Z_{10}^{2}} \right), \qquad 4.141a$$

and from (4.120) and (4.123) for the backward wave as

$$A^{-} = \frac{-1}{P_{10}} \int_{V} (E_{y}^{+} J_{y} - H_{x}^{+} M_{x}) dv = \frac{-j\omega A}{P_{10}} \left(\epsilon_{0} \alpha_{e} + \frac{\mu_{0} \alpha_{m}}{Z_{10}^{2}} \right), \qquad 4.141b$$

where $P_{10} = ab/Z_{10}$. Note that the electric dipole excites the same fields in both directions, but the magnetic dipole excites oppositely polarized fields in the forward and backward directions.

REFERENCES

- [1] S. Ramo, T. R. Whinnery, and T. van Duzer, Fields and Waves in Communication Electronics, John Wiley & Sons, N.Y., 1965.
- [2] A. A. Oliner, "Historical Perspectives on Microwave Field Theory," IEEE Trans. Microwave Theory and Techniques, vol. MTT-32, pp. 1022–1045, September 1984.
- [3] C. G. Montgomery, R. H. Dicke, and E. M. Purcell, Principles of Microwave Circuits, vol. 8 of MIT Rad. Lab. Series, McGraw-Hill, N.Y., 1948.
- [4] R. E. Collin, Foundations for Microwave Engineering, Second Edition, McGraw-Hill, N.Y., 1992.
- [5] G. Gonzalez, Microwave Transistor Amplifiers, Prentice-Hall, N.J., 1984.

- [6] J. S. Wright, O. P. Jain, W. J. Chudobiak, and V. Makios, "Equivalent Circuits of Microstrip Impedance Discontinuities and Launchers," *IEEE Trans, Microwave Theory and Techniques*, vol. MTT-22, pp. 48–52, January 1974.
- [7] G. F. Engen and C. A. Hoer, "Thru-Reflect-Line: An Improved Technique for Calibrating the Dual Six-Port Automatic Network Analyzer," *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-27, pp. 987–998, December 1979.
- [8] N. Marcuvitz, Waveguide Handbook, vol. 10 of MIT Rad. Lab. Series, McGraw-Hill, N.Y., 1948.
- [9] K. C. Gupta, R. Garg, and I. J. Bahl, Microstrip Lines and Stotlines, Artech House, Dedham, Mass., 1979.
- [10] G. Matthaei, L. Young, and E. M. T. Jones, Microwave Filters, Impedance-Matching Networks, and Coupling Structures, Chapter 5, Artech House, Dedham, Mass., 1980.
- [11] R. E. Collin. Field Theory of Guided Waves, McGraw-Hill, N.Y., 1960.

PROBLEMS

- 4.1 Solve the problem of Example 4.2 by writing expressions for the incident, reflected, and transmitted E_y and H_x fields for the regions z < 0 and z > 0, and applying the boundary conditions for these fields at the dielectric interface at z = 0.
- **4.2** Consider the reflection of a TE₁₀ mode, incident from z < 0, at a step change in the height of a rectangular waveguide, as shown below. Show that if the method of Example 4.2 is used, the result $\Gamma = 0$ is obtained. Do you think this is the correct solution? Why? (This problem shows that the one-mode impedance viewpoint does not always provide a correct analysis.)



- **4.3** Consider a series *RLC* circuit with a current, *L*. Calculate the power lost and the stored electric and magnetic energies, and show that the input impedance can be expressed as in (4.17).
- **4.4** Show that the input impedance. Z, of a parallel RLC circuit satisfies the condition that $Z(-\omega) = Z^*(\omega)$.
- 4.5 Show that the admittance matrix of a lossless N-port network has purely imaginary elements.
- 4.6 Does a nonreciprocal lossless network always have a purely imaginary impedance matrix?
- **4.7** Derive the [Z] and [Y] matrices for the two-port networks shown below.



4.8 Consider a two-port network, and let $Z_{SC}^{(1)}$, $Z_{SC}^{(2)}$, $Z_{OC}^{(1)}$, $Z_{OC}^{(2)}$ be the input impedance seen when port 2 is short-circuited, when port 1 is short-circuited, when port 2 is open-circuited, and when port 1 is open-circuited, respectively. Show that the impedance matrix elements are given by

 $Z_{11} = Z_{\text{OC}}^{(1)}$, $Z_{22} = Z_{\text{OC}}^{(2)}$, $Z_{12}^2 = Z_{21}^2 = (Z_{\text{OC}}^{(1)} - Z_{\text{SC}}^{(1)})Z_{\text{OC}}^{(2)}$

4.9 A two-port network is driven at both ports such that the port voltages and currents have the following values:

$$V_1 = 10 \angle 0,$$
 $I_1 = 0.1 \angle 30^\circ,$
 $V_2 = 12 \angle 90^\circ,$ $I_2 = 0.15 \angle 120^\circ.$

Determine the incident and reflected voltages at both ports, if the characteristic impedance is 50 Ω .

4.10 Derive the scattering matrix for each of the lossless transmission lines shown below, relative to a system impedance of Z_0 . Verify that each matrix is unitary.



4.11 Consider two two-port networks with individual scattering matrices, $[S^A]$ and $[S^B]$. Show that the overall S_{21} parameter of the cascade of these networks is given by

$$S_{21} = \frac{S_{21}^A S_{21}^B}{1 - S_{22}^A S_{11}^B}$$

- 4.12 Consider a lossless two-port network. (a) If the network is reciprocal, show that |S₂₁|² = 1 − |S₁₁|².
 (b) If the network is nonreciprocal, show that it is impossible to have unidirectional transmission, where S₁₂ = 0 and S₂₁ ≠ 0.
- **4.13** Show that it is impossible to construct a three-port network that is lossless, reciprocal, and matched at all ports. Is it possible to construct a nonreciprocal three-port network that is lossless and matched at all ports?
- **4.14** Prove the following *decoupling theorem*: For any lossless reciprocal three-port network, one port (say port 3) can be terminated in a reactance so that the other two ports (say ports 1 and 2) are decoupled (no power flow from port 1 to port 2, or from port 2 to port 1).
- **4.15** A certain three-port network is lossless and reciprocal, and has $S_{13} = S_{23}$ and $S_{11} = S_{22}$. Show that if port 2 is terminated with a matched load, then port 1 can be matched by placing an appropriate reactance at port 3.
- 4.16 A four-port network has the scattering matrix shown below.
 - (a) Is this network lossless?
 - (b) Is this network reciprocal?
 - (c) What is the return loss at port 1 when all other ports are matched?
 - (d) What is the insertion loss and phase between ports 2 and 4, when all other ports are matched?

(e) What is the reflection coefficient seen at port 1 if a short circuit is placed at the terminal plane of port 3, and all other ports are matched?

$$[S] = \begin{bmatrix} 0.1 \underline{..90^{\circ}} & 0.6 \underline{..-45^{\circ}} & 0.6 \underline{..45^{\circ}} & 0\\ 0.6 \underline{..-45^{\circ}} & 0 & 0 & 0.6 \underline{..45^{\circ}} \\ 0.6 \underline{..-45^{\circ}} & 0 & 0 & 0.6 \underline{..-45^{\circ}} \\ 0 & 0.6 \underline{..45^{\circ}} & 0.6 \underline{..-45^{\circ}} & 0 \end{bmatrix}.$$

4.17 A four-port network has the scattering matrix shown below. If ports 3 and 4 are connected with a lossless matched transmission line with an electrical length of 100°, find the resulting insertion loss and phase between ports 1 and 2.

$$[S] = \begin{bmatrix} 0.6 \angle 90^{\circ} & 0 & 0 & 0.8 \angle 0 \\ 0 & 0.7 \angle 45^{\circ} & 0.7 \angle -45^{\circ} & 0 \\ 0 & 0.7 \angle -45^{\circ} & 0.7 \angle 45^{\circ} & 0 \\ 0.8 \angle 0 & 0 & 0 & 0.6 \angle 90^{\circ} \end{bmatrix}$$

4.18 Consider a two-port network consisting of a junction of two transmission lines with characteristic impedances Z_{01} and Z_{02} , as shown below. Find the generalized scattering parameters of this network.



4.19 The scattering parameters of a certain two-port network were measured to be

 $S_{11} = 0.3 \pm j0.7,$ $S_{12} = S_{21} \pm j0.6,$ $S_{22} = 0.3 \pm j0.7.$

Find the equivalent impedance parameters for this network, if the characteristic impedance is 50Ω .

4.20 When normalized to a characteristic impedance Z_0 , a certain two-port network has scattering parameters S_{ij} . This network is now placed in a circuit as shown below. Find the new (generalized) scattering parameters S'_{ij} , relative to the characteristic impedances Z_{01} and Z_{02} , in terms of S_{ij} .



- **4.21** Find the impedance parameters of a section of transmission line with length *l*, characteristic impedance Z_0 , and propagation constant β .
- **4.22** The *ABCD* parameters of the first entry in Table 4.1 were derived in Example 4.6. Verify the *ABCD* parameters for the second, third, and fourth entries.

- 4.23 Derive expressions that give the impedance parameters in terms of the ABCD parameters,
- 4.24 Use ABCD matrices to find the voltage V_L across the load resistor in the circuit shown below.



4.25 Find the ABCD matrix for the circuit shown below by direct calculation using the definition of the ABCD matrix, and compare with the ABCD matrix of the appropriate cascade of canonical circuits from Table 4.1.



4.26 Show that the admittance matrix of the two parallel connected two-port π -networks shown below can be found by adding the admittance matrices of the individual two-ports. Apply this result to find the admittance matrix of the bridged-T circuit shows.



- **4.27** Derive the expressions for S parameters in terms of the ABCD parameters, as given in Table 4.2.
- **4.28** Find the S parameters for the series and shunt loads shown below. Show that $S_{12} = 1 S_{11}$ for the series case, and that $S_{12} = 1 + S_{11}$ for the shunt case. Assume a characteristic impedance Z_0 .



4.29 Find the signal flow graph for a matched length of lossless transmission line, with an electrical length $\beta \ell$.

4.30 Use signal flow graphs to find the power ratios P_2/P_1 and P_3/P_1 for the mismatched three-port network shown below.



- **4.31** For the *H*-plane step analysis of Section 4.6, compute the complex power flow in the reflected modes in guide 1, and show that the reactive power is inductive.
- **4.32** For the *H*-plane step of Section 4.6, assume that $\lambda = 4.2a$ and c = 0.8a, so that a TE₁₀ mode can propagate in each guide. Using N = 2 equations, compute the coefficients A_1 and A_2 from the modal analysis solution and draw the equivalent circuit of the discontinuity
- **4.33** Derive the modal analysis equations for the symmetric *H*-plane step shown below. (HINT: Because of symmetry, only the TE_{n0} modes, for *n* odd, will be excited.)



- **4.34** Find the transverse \overline{E} and \overline{H} fields excited by the current of (4.110) by postulating traveling TM_{mn} modes on either side of the source at z = 0, and applying the appropriate boundary conditions.
- **4.35** Show that the magnetic surface current density of (4.114) excites TM_{mn} waves traveling away from the source.
- **4.36** An infinitely long rectangular waveguide is fed with a probe of length d, as shown below. The current on this probe can be approximated as $I(q) = I_0 \sin k(d q)/\sin kd$. If the TE₁₀ mode is the only propagating mode in the waveguide, compute the input resistance seen at the probe terminals.



4.37 Consider the infinitely long waveguide fed with two probes driven 180° out of phase, as shown below. What are the resulting excitation coefficients for the TE_{10} and TE_{20} modes? What other modes can be excited by this feeding arrangement?



4.38 Consider a small current loop on the side wall of a rectangular waveguide, as shown below. Find the TE₁₀ fields excited by this loop, if the loop is of radius r_0 .



4.39 A rectangular waveguide is shorted at z = 0, and has an electric current sheet, J_{sy} , located at z = d, where

$$J_{xy} = \frac{2\pi A}{a} \sin \frac{\pi x}{a}$$

Find expressions for the fields generated by this current by assuming standing wave fields for 0 < z < d, and traveling wave fields for z > d, and applying boundary conditions at z = 0 and z = d. Now solve the problem using image theory, by placing a current sheet $-J_{sy}$ at z = -d, and removing the shorting wall at z = 0. Use the results of Section 4.7 and superposition to find the fields radiated by these two currents, which should be the same as the first results for z > 0.



Impedance Matching and Tuning

This chapter marks a turning point in that we now begin to apply the theory and techniques of the previous chapters to practical problems in microwave engineering. We begin with the topic of impedance matching, which is often a part of the larger design process for a microwave component or system. The basic idea of impedance matching is illustrated in Figure 5.1, which shows an impedance matching network placed between a load impedance and a transmission line. The matching network is ideally lossless, to avoid unnecessary loss of power, and is usually designed so that the impedance seen looking into the matching network is Z_0 . Then reflections are eliminated on the transmission line to the left of the matching network, although there will be multiple reflections between the matching network and the load. This procedure is also referred to as tuning. Impedance matching or tuning is important for the following reasons:

- Maximum power is delivered when the load is matched to the line (assuming the generator is matched), and power loss in the feed line is minimized.
- Impedance matching sensitive receiver components (antenna, low-noise amplifier, etc.) improves the signal-to-noise ratio of the system.
- Impedance matching in a power distribution network (such as an antenna array feed network) will reduce amplitude and phase errors.

As long as the load impedance, Z_L , has some nonzero real part, a matching network can always be found. Many choices are available, however, and we will discuss the design and performance of several types of practical matching networks. Factors that may be important in the selection of a particular matching network include the following:

• Complexity—As with most engineering solutions, the simplest design that satisfies the required specifications is generally the most preferable. A simpler matching



FIGURE 5.1 A lossless network matching an arbitrary load impedance to a transmission line. 251

network is usually cheaper, more reliable, and less lossy than a more complex design.

- Bandwidth—Any type of matching network can ideally give a perfect match (zero reflection) at a single frequency. In many applications, however, it is desirable to match a load over a band of frequencies. There are several ways of doing this with, of course, a corresponding increase in complexity.
- Implementation—Depending on the type of transmission line or waveguide being used, one type of matching network may be preferable compared to another. For example, tuning stubs are much easier to implement in waveguide than are multisection quarter-wave transformers.
- Adjustability—In some applications the matching network may require adjustment to match a variable load impedance. Some types of matching networks are more amenable than others in this regard.

5.1____ MATCHING WITH LUMPED ELEMENTS (L NETWORKS)

Probably the simplest type of matching network is the L section, which uses two reactive elements to match an arbitrary load impedance to a transmission line. There are two possible configurations for this network, as shown in Figure 5.2. If the normalized load impedance, $z_L = Z_L/Z_0$, is inside the 1 + jx circle on the Smith chart, then the circuit of Figure 5.2a should be used. If the normalized load impedance is outside the 1 + jx circle on the Smith chart, the circuit of Figure 5.2b should be used. The 1 + jxcircle is the resistance circle on the impedance Smith chart for which r = 1.

In either of the configurations of Figure 5.2, the reactive elements may be either inductors or capacitors, depending on the load impedance. Thus, there are eight distinct possibilities for the matching circuit for various load impedances. If the frequency is low enough and/or the circuit size is small enough, actual lumped-element capacitors and inductors can be used. This may be feasible for frequencies up to about 1 GHz or so, although modern microwave integrated circuits may be small enough so that lumped elements can be used at higher frequencies as well. There is, however, a large range of frequencies and circuit sizes where lumped elements may not be realizable. This is a limitation of the L section matching technique.



FIGURE 5.2 L section matching networks. (a) Network for z_L inside the 1+jx circle. (b) Network for z_L outside the 1 + jx circle.

We will now derive the analytic expressions for the matching network elements of the two cases in Figure 5.2, then illustrate an alternative design procedure using the Smith chart.

Analytic Solutions

Although we will discuss a simple graphical solution using the Smith chart, it may be useful to derive expressions for the L section matching network components. Such expressions would be useful in a computer-aided design program for L section matching, or when it is necessary to have more accuracy than the Smith chart can provide.

Consider first the circuit of Figure 5.2a, and let $Z_L = R_L + jX_L$. We stated that this circuit would be used when $z_L = Z_L/Z_0$ is inside the 1 + jx circle on the Smith chart, which implies that $R_L > Z_0$ for this case.

The impedance seen looking into the matching network followed by the load impedance must be equal to Z_0 , for a match:

$$Z_0 = jX + \frac{1}{jB + 1/(R_L - jX_L)}.$$
 5.1

Rearranging and separating into real and imaginary parts gives two equations for the two unknowns, X and B:

$$B(XR_L - X_L Z_0) = R_L - Z_0. 5.2a$$

$$X(1 - BX_L) = BZ_0R_L - X_L$$
, 5.2*h*

Solving (5.2a) for X and substituting into (5.2b) gives a quadratic equation for B. The solution is

$$B_{\rm c} = \frac{X_L \pm \sqrt{R_L/Z_0} \sqrt{R_L^2 + X_L^2 - Z_0 R_L}}{R_L^2 + X_L^2}.$$
 5.3*a*

Note that since $R_L > Z_0$, the argument of the second square root is always positive. Then the series reactance can be found as

$$X = \frac{1}{B} + \frac{X_L Z_0}{R_L} - \frac{Z_0}{BR_L}.$$
 5.3*b*

Equation (5.3a) indicates that two solutions are possible for B and X. Both of these solutions are physically realizable, since both positive and negative values of B and X are possible (positive X implies an inductor, negative X implies a capacitor, while positive B implies a capacitor and negative B implies an inductor.) One solution, however, may result in significantly smaller values for the reactive components, and may be the preferred solution if the bandwidth of the match is better, or the SWR on the line between the matching network and the load is smaller.

Now consider the circuit of Figure 5.2b. This circuit is to be used when z_L is outside the 1 + jx circle on the Smith chart, which implies that $R_L < Z_0$. The admittance seen looking into the matching network followed by the load impedance $Z_L = R_L + jX_L$

must be equal to $1/Z_0$, for a match:

$$\frac{1}{Z_0} = jB + \frac{1}{R_L + j(X + X_L)}.$$
 5.4

Rearranging and separating into real and imaginary parts gives two equations for the two unknowns, X and B:

$$BZ_0(X + X_L) = Z_0 - R_L, \qquad 5.5a$$

$$(X+X_L)=BZ_0R_L.$$
 5.5b

Solving for X and B gives

$$X = \pm \sqrt{R_L(Z_0 - R_L)} - X_L, \qquad 5.6a$$

$$B = \pm \frac{\sqrt{(Z_0 - R_L)/R_L}}{Z_0}.$$
 5.6b

Since $R_L < Z_0$, the arguments of the square roots are always positive. Again, note that two solutions are possible.

In order to match an arbitrary complex load to a line of characteristic impedance Z_0 , the real part of the input impedance to the matching network must be Z_0 , while the imaginary part must be zero. This implies that a general matching network must have at least two degrees of freedom; in the L section matching circuit these two degrees of freedom are provided by the values of the two reactive components.

Smith Chart Solutions

Instead of the above formulas, the Smith chart can be used to quickly and accurately design L section matching networks, a procedure best illustrated by an example.



EXAMPLE 5.1 L-Section Impedance Matching

Design an L section matching network to match a series RC load with an impedance $Z_L = 200 - j100 \Omega$, to a 100 Ω line, at a frequency of 500 MHz. Solution

The normalized load impedance is $z_L = 2 - j1$, which is plotted on the Smith chart of Figure 5.3a. This point is inside the 1 + jx circle, so we will use the matching circuit of Figure 5.2a. Since the first element from the load is a shunt susceptance, it makes sense to convert to admittance by drawing the SWR circle through the load, and a straight line from the load through the center of the chart, as shown in Figure 5.3a. Now, after we add the shunt susceptance and convert back to impedance, we want to be on the 1 + jx circle, so that we can add a series reactance to cancel the jx and match the load. This means that the shunt susceptance must move us from y_L to the 1 + jx circle on the *admittance* Smith chart. Thus, we construct the rotated 1+jx circle as shown in Figure 5.3a (center at 0.333). (A combined ZY chart is convenient to use here, if it is not too confusing.) Then we see that adding a susceptance of jb = j0.3will move us along a constant conductance circle to y = 0.4 + j0.5 (this choice



_____FIGURE 5.3 Solution to Example 5.1. (a) Smith chart for the L section matching networks.

is the shortest distance from y_L to the shifted 1 + jx circle). Converting back to impedance leaves us at z = 1 - j1.2, indicating that a series reactance x = j1.2 will bring us to the center of the chart. For comparison, the formulas of (5.3a,b) give the solution as b = 0.29, x = 1.22,

This matching circuit consists of a shunt capacitor and a series inductor, as shown in Figure 5.3b. For a frequency of f = 500 MHz, the capacitor has a value of

$$\tilde{C} = rac{b}{2\pi f Z_0} = 0.92 ext{ pF}.$$

and the inductor has a value of

$$L = \frac{xZ_0}{2\pi f} = 38.8 \text{ nH}.$$



FIGURE 5.3 Continued. (b) The two possible L section matching circuits. (c) Reflection coefficient magnitudes versus frequency for the matching circuits of (b).

It may also be interesting to look at the second solution to this matching problem. If instead of adding a shunt susceptance of b = 0.3, we use a shunt susceptance of b = -0.7, we will move to a point on the lower half of the shifted 1 + jx circle, to y = 0.4 - j0.5. Then converting to impedance and adding a series reactance of x = -1.2 leads to a match as well. The formulas of (5.3a,b) give this solution as b = -0.69, x = -1.22. This matching circuit is also shown in Figure 5.3b, and is seen to have the positions of the inductor and capacitor reversed from the first matching network. At a frequency of f = 500 MHz, the capacitor has a value of

$$C = \frac{-1}{2\pi f x Z_0} = 2.61 \text{ pF},$$

while the inductor has a value of

$$L = \frac{-Z_0}{2\pi fb} = 46.1$$
 nH.

Figure 5.3c shows the reflection coefficient magnitude versus frequency for these two matching networks, assuming that the load impedance of $Z_L = 200 - j100 \ \Omega$ at 500 MHz consists of a 200 Ω resistor and a 3.18 pF capacitor in series. There is not a substantial difference in bandwidth for these two solutions.

POINT OF INTEREST: Lumped Elements for Microwave Integrated Circuits

Lumped R. L, and C elements can be practically realized at microwave frequencies if the length, ℓ , of the component is very small relative to the operating wavelength. Over a limited range of values, such components can be used in hybrid and monolithic microwave integrated circuits (MICs) at frequencies up to 60 GHz, if the condition that $\ell < \lambda/10$ is satisfied. Usually, however, the characteristics of such an element are far from ideal, requiring that undesirable effects such as parasitic capacitance and/or inductance, spurious resonances, fringing fields, loss, and perturbations caused by a ground plane be incorporated in the design via a CAD model (see the Point of Interest concerning CAD).

Resistors are fabricated with thin films of lossy material such as nichrome, tantalum nitride, or doped semiconductor material. In monofithic circuits such films can be deposited or grown, while chip resistors made from a lossy film deposited on a ceramic chip can be bonded or soldered in a hybrid circuit. Low resistances are hard to obtain.

Small values of inductance can be realized with a short length or loop of transmission line, and larger values (up to about 10 nH) can be obtained with a spiral inductor, as shown in the following figures. Larger inductance values generally incur more loss, and more shunt capacitance; this leads to a resonance that limits the maximum operating frequency.

Capacitors can be fabricated in several ways. A short transmission line stub can provide a shunt capacitance in the range of 0 to 0.1 pF. A single gap or interdigital set of gaps in a transmission



line can provide a series capacitance up to about 0.5 pF. Greater values (up to about 25 pF) can be obtained using a metal-insulator-metal (MTM) sandwich, either in monolithic or chip (hybrid) form.

5.2 SINGLE-STUB TUNING

We next consider a matching technique that uses a single open-circuited or shortcircuited length of transmission line (a "stub"), connected either in parallel or in series with the transmission feed line at a certain distance from the load, as shown in Figure 5.4. Such a tuning circuit is convenient from a microwave fabrication aspect, since lumped elements are not required. The shunt tuning stub is especially easy to fabricate in microstrip or stripline form.

In single-stub tuning, the two adjustable parameters are the distance, d, from the load to the stub position, and the value of susceptance or reactance provided by the shunt or series stub. For the shunt-stub case, the basic idea is to select d so that the admittance, Y, seen looking into the line at distance d from the load is of the form $Y_0 + jB$. Then





the stub susceptance is chosen as -jB, resulting in a matched condition. For the series stub case, the distance d is selected so that the impedance. Z, seen looking into the line at a distance d from the load is of the form $Z_0 + jX$. Then the stub reactance is chosen as -jX, resulting in a matched condition.

As discussed in Chapter 2, the proper length of open or shorted transmission line can provide any desired value of reactance or susceptance. For a given susceptance or reactance, the difference in lengths of an open- or short-circuited stub is $\lambda/4$. For transmission line media such as microstrip or stripline, open-circuited stubs are easier to fabricate since a via hole through the substrate to the ground plane is not needed. For lines like coax or waveguide, however, short-circuited stubs are usually preferred, because the crosssectional area of such an open-circuited line may be large enough (electrically) to radiate, in which case the stub is no longer purely reactive.

Below we discuss both Smith chart and analytic solutions for shunt and series stub turning. The Smith chart solutions are fast, intuitive, and usually accurate enough in practice. The analytic expressions are more accurate, and useful for computer analysis.

Shunt Stubs

The single-stub shunt tuning circuit is shown in Figure 5.4a. We will first discuss an example illustrating the Smith chart solution, and then derive formulas for d and ℓ .

Ø

EXAMPLE 5.2 Single-Stub Shunt Tuning

For a load impedance $Z_L = 15 \pm 310 \Omega$, design two single-stub shunt tuning networks to match this load to a 50 Ω line. Assuming that the load is matched at 2 GHz, and that the load consists of a resistor and inductor in series, plot the reflection coefficient magnitude from 1 GHz to 3 GHz for each solution.

Solution

The first step is to plot the normalized load impedance $z_L = 0.3 + j0.2$, construct the appropriate SWR circle, and convert to the load admittance, y_L , as shown on the Smith chart in Figure 5.5a. For the remaining steps we consider the Smith chart as an admittance chart. Now notice that the SWR circle intersects the 1 + jb circle at two points, denoted as y_1 and y_2 in Figure 5.5a. Thus the distance d, from the load to the stub, is given by either of these two intersections. Reading the WTG scale, we obtain

$$d_1 = 0.328 - 0.284 = 0.044\lambda,$$

$$d_2 = (0.5 - 0.284) + 0.171 = 0.387\lambda$$

Actually, there is an infinite number of distances, d, on the SWR circle that intersect the 1 + jb circle. Usually, it is desired to keep the matching stub as close as possible to the load, to improve the bandwidth of the match and to reduce losses caused by a possibly large standing wave ratio on the line between the stub and the load.

Chapter 5: Impedance Matching and Tuning



FIGURE 5.5 Solution to Example 5.2. (a) Smith chart for the shant-stub tuners.

At the two intersection points, the normalized admittances are

$$y_1 = 1 - j1.33,$$

 $y_2 = 1 + j1.33.$

Thus, the first tuning solution requires a stub with a susceptance of j1.33. The length of an open-circuited stub that gives this susceptance can be found on the Smith chart by starting at y = 0 (the open circuit) and moving along the outer edge of the chart (g = 0) toward the generator to the j1.33 point. The length is then

$$\ell_1=0.147\lambda.$$

Similarly, the required open-circuit stub length for the second solution is

$$\ell_2 = 0.353\lambda_c$$

This completes the tuner designs.

To analyze the frequency dependence of these two designs, we need to know the load impedance as a function of frequency. The series-*RL* load impedance is $Z_L = 15 \pm j10 \Omega$ at 2 GHz, so $R = 15 \Omega$ and L = 0.796 nII. The two tuning circuits are shown in Figure 5.5b. Figure 5.5c shows the calculated reflection coefficient magnitudes for these two solutions. Observe that solution I has a significantly better bandwidth than solution 2; this is because both dand ℓ are shorter for solution 1, which reduces the frequency variation of the match.



FIGURE 5.5 Continued. (b) The two shunt-stub tuning solutions. (c) Reflection coefficient magnitudes versus frequency for the tuning circuits of (b).

To derive formulas for d and ℓ , let the load impedance be written as $Z_L = 1/Y_L = R_L + jX_L$. Then the impedance Z down a length, d, of line from the load is

$$Z = Z_0 \frac{(R_L + jX_L) + jZ_0 t}{Z_0 + j(R_L + jX_L)t},$$
 5.7

where $t = \tan \beta d$. The admittance at this point is

$$Y = G + jB = \frac{1}{Z},$$

$$G = \frac{R_L(1+t^2)}{R_L^2 + (X_L + Z_0 t)^2}.$$
5.8a

where

$$B = \frac{R_L^2 t - (Z_0 - X_L t)(X_L + Z_0 t)}{Z_0 [R_L^2 + (X_L + Z_0 t)^2]}.$$
 5.8b

Now d (which implies t) is chosen so that $G = Y_0 = 1/Z_0$. From (5.8a), this results in a quadratic equation for t:

$$Z_0(R_L - Z_0)t^2 - 2X_L Z_0 t + (R_L Z_0 - R_L^2 - X_L^2) = 0.$$

Solving for t gives

$$t = \frac{X_L \pm \sqrt{R_L [(Z_0 - R_L)^2 + X_L^2]/Z_0}}{R_L - Z_0}, \quad \text{for } R_L \neq Z_0. \quad 5.9$$

If $R_L = Z_0$, then $t = -X_L/2Z_0$. Thus, the two principal solutions for d are

$$\frac{d}{\lambda} = \begin{cases} \frac{1}{2\pi} \tan^{-1} t, & \text{for } t \ge 0\\ \frac{1}{2\pi} (\pi + \tan^{-1} t), & \text{for } t < 0. \end{cases}$$
5.10

To find the required stub lengths, first use t in (5.8b) to find the stub susceptance. $B_s = -B$. Then, for an open-circuited stub,

$$\frac{\boldsymbol{\ell}_o}{\lambda} = \frac{1}{2\pi} \tan^{-1} \left(\frac{B_s}{Y_0} \right) = \frac{-1}{2\pi} \tan^{-1} \left(\frac{B}{Y_0} \right).$$
 5.11*a*

while for a short-circuited stub,

$$\frac{\ell_s}{\lambda} = \frac{-1}{2\pi} \tan^{-1} \left(\frac{Y_0}{B_s} \right) = \frac{1}{2\pi} \tan^{-1} \left(\frac{Y_0}{B} \right).$$
 5.11b

If the length given by (5.11a) or (5.11b) is negative, $\lambda/2$ can be added to give a positive result.

Series Stubs

The series stub tuning circuit is shown in Figure 5.4b. We will illustrate the Smith chart solution by an example, and then derive expressions for d and ℓ .



EXAMPLE 5.3 Single-Stub Series Tuning

Match a load impedance of $Z_L = 100 + j80$ to a 50 Ω line using a single series open-circuit stub. Assuming that the load is matched at 2 GHz, and that the load consists of a resistor and inductor in series, plot the reflection coefficient magnitude from 1 GHz to 3 GHz.

Solution

The first step is to plot the normalized load impedance, $z_L = 2 + j1.6$, and draw the SWR circle. For the series-stub design, the chart is an impedance chart. Note that the SWR circle intersects the 1 + jx circle at two points, denoted as z_1 and z_2 in Figure 5.6a. The shortest distance, d_1 ; from the load to



Solution to Example 5.3. (a) Smith chart for the series-stud tuners,

the stub is, from the WTG scale,

$$d_1 = 0.328 - 0.208 = 0.120\lambda,$$

while the second distance is

$$d_2 = (0.5 - 0.208) + 0.172 = 0.463\lambda.$$

As in the shunt-stub case, additional rotations around the SWR circle lead to additional solutions, but these are usually not of practical interest.

The normalized impedances at the two intersection points are

$$z_1 = 1 - j1.33,$$

 $z_2 = 1 + j1.33.$

Thus, the first solution requires a stub with a reactance of j1.33. The length of an open-circuited stub that gives this reactance can be found on the Smith chart by starting at $z = \infty$ (open circuit), and moving along the outer edge of the chart (r = 0) toward the generator to the j1.33 point. This gives a stub length of

$$\ell_1 = 0.397\lambda$$
.

Similarly, the required open-circuited stub length for the second solution is

$$\ell_2 = 0.103\lambda$$

This completes the tuner designs.

If the load is a series resistor and inductor with $Z_L = 100 + j80 \Omega$ at 2 GHz, then $R = 100 \Omega$ and L = 6.37 nH. The two matching circuits are shown in Figure 5.6b. Figure 5.6c shows the calculated reflection coefficient magnitudes versus frequency for the two solutions.

To derive formulas for d and ℓ for the series-stub tuner, let the load admittance be written as $Y_L = 1/Z_L = G_L + jB_L$. Then the admittance Y down a length, d, of line from the load is

$$Y = Y_0 \frac{(G_L + jB_L) + jtY_0}{Y_0 + jt(G_L + jB_L)},$$
5.12

where $t = \tan \beta d$, and $Y_0 = 1/Z_0$. Then the impedance at this point is

$$Z = R + jX = \frac{1}{Y},$$

$$R = \frac{G_L(1 + t^2)}{G_L^2 + (B_L + Y_0 t)^2},$$
5.13a

where

$$X = \frac{G_L^2 t - (Y_0 - tB_L)(B_L + tY_0)}{Y_0[G_L^2 + (B_L + Y_0 t)^2]}.$$
 5.13b



FIGURE 5.6 Continued. (b) The two series-stub tuning solutions. (c) Reflection coefficient magnitudes versus frequency for the tuning circuits of (b).

Now d (which implies t) is chosen so that $R = Z_0 = 1/Y_0$. From (5.13a), this results in a quadratic equation for t:

$$Y_0(G_L - Y_0)t^2 - 2B_LY_0t + (G_LY_0 - G_L^2 - B_L^2) = 0.$$

Solving for t gives

$$t = \frac{B_L \pm \sqrt{G_L[(Y_0 - G_L)^2 + B_L^2]/Y_0}}{G_L - Y_0}, \quad \text{for } G_L \neq Y_0.$$
 \$14

If $G_L = Y_0$, then $t = -B_L/2Y_0$. Then the two principal solutions for d are

$$d/\lambda = \begin{cases} \frac{1}{2\pi} \tan^{-1} t & \text{for } t \ge 0\\ \frac{1}{2\pi} (\pi + \tan^{-1} t) & \text{for } t < 0. \end{cases}$$
(5.15)

The required stub lengths are determined by first using t in (5.13b) to find the reactance. X. This reactance is the negative of the necessary stub reactance, X_s . Thus, for a short-circuited stub,

$$\frac{\ell_s}{\lambda} \approx \frac{1}{2\pi} \tan^{-1} \left(\frac{X_s}{Z_0} \right) = \frac{-1}{2\pi} \tan^{-1} \left(\frac{X}{Z_0} \right). \qquad 5.16a$$

while for an open-circuited stub,

$$\frac{\ell_o}{\lambda} = \frac{-1}{2\pi} \tan^1\left(\frac{Z_0}{X_s}\right) = \frac{1}{2\pi} \tan^{-1}\left(\frac{Z_0}{X}\right).$$
 5.16b

If the length given by (5.16a) or (5.16b) is negative, $\lambda/2$ can be added to give a positive result.

5.3 DOUBLE-STUB TUNING

The single-stub tuners of the previous section are able to match any load impedance (as long as it has a nonzero real part) to a transmission line, but suffer from the disadvantage of requiring a variable length of line between the load and the stub. This may not be a problem for a fixed matching circuit, but would probably pose some difficulty if an adjustable tuner was desired. In this case, the double-stub tuner, which uses two tuning stubs in fixed positions, can be used. Such tuners are often fabricated in coaxial line, with adjustable stubs connected in parallel to the main coaxial line. We will see, however, that the double-stub tuner cannot match all load impedances.

The double-stub tuner circuit is shown in Figure 5.7a, where the load may be an arbitrary distance from the first stub. Although this is more representative of a practical situation, the circuit of Figure 5.7b, where the load Y'_L has been transformed back to the position of the first stub, is easier to deal with and does not lose any generality. The stubs shown in Figure 5.7 are shunt stubs, which are usually easier to implement in practice than are series stubs; the latter could be used just as well, in principle. In either case, the stubs can be open-circuited or short-circuited.

Smith Chart Solution

The Smith chart of Figure 5.8 illustrates the basic operation of the double-stub tuner. As in the case of the single-stub tuners, two solutions are possible. The susceptance of the first stub, b_1 (or b'_1 , for the second solution), moves the load admittance to y_1 (or y'_1). These points lie on the rotated 1 + jb circle; the amount of rotation is d wavelengths toward the load, where d is the electrical distance between the two stubs. Then transforming y_1 (or y'_1) toward the generator through a length, d, of line leaves us at the point y_2 (or y'_2), which must be on the 1 + jb circle. The second stub then adds a susceptance b_2 (or b'_2), which brings us to the center of the chart, and completes the match.

Notice from Figure 5.8 that if the load admittance, y_L , were inside the shaded region of the $g_0 + jb$ circle, no value of stub susceptance b_1 could ever bring the load point



FIGURE 5.7 Double-stub tuning. (a) Original circuit with the load an arbitrary distance from the first stub. (b) Equivalent circuit with load at the first stub.

to intersect the rotated 1 + jb circle. This shaded region thus forms a forbidden range of load admittances, which cannot be matched with this particular double-stub tuner. A simple way of reducing the forbidden range is to reduce the distance, d, between the stubs. This has the effect of swinging the rotated 1 - jb circle back toward the $y = \infty$ point, but d must be kept large enough for the practical purpose of fabricating the two separate stubs. In addition, stub spacings near 0 or $\lambda/2$ lead to matching networks that are very frequency sensitive. In practice, stub spacings are usually chosen as $\lambda/8$ or $3\lambda/8$. If the length of line between the load and the first stub can be adjusted, then the load admittance y_L can always be moved out of the forbidden region.

EXAMPLE 5.4 Double-Stub Tuning

Design a double-stub shunt tuner to match a load impedance $Z_L = 60 - j80 \Omega$ to a 50 Ω line. The stubs are to be short-circuited stubs, and are spaced $\lambda/8$ apart. Assuming that this load consists of a series resistor and capacitor, and that the match frequency is 2 GHz, plot the reflection coefficient magnitude versus frequency from 1 GHz to 3 GHz.



FIGURE 5.8 Smith chart diagram for the operation of a double-stub tuner.

Solution

The normalized load admittance is $y_L = 0.3 + j0.4$, which is plotted on the Smith chart of Figure 5.9a. Next we construct the rotated 1 + jb conductance circle, by moving every point on the g = 1 circle $\lambda/8$ toward the load. We then find the susceptance of the first stub, which can be one of two possible values:

or
$$b_1 = 1.314$$
,
 $b_1' = -0.114$.

We now transform through the $\lambda/8$ section of line by rotating along a constant radius (SWR) circle $\lambda/8$ toward the generator. This brings the two solutions to the following points:

or
$$y_2 = 1 - j3.38$$
,
 $y'_2 = 1 + j1.38$.

Then the susceptance of the second stub should be

$$b_2 = 3.38,$$

or $b_2' = -1.38.$

268



FIGURE 5.9 Solution to Example 5.4. (a) Smith chart for the double-stub tuners.

The lengths of the short-circuited stubs are then found as

$$\ell_1 = 0.396\lambda, \qquad \ell_2 = 0.454\lambda,$$

 $\ell'_1 = 0.232\lambda, \qquad \ell'_2 = 0.100\lambda.$

or

This completes both solutions for the double-stub tuner design.

Now if the resistor-capacitor load $Z_L = 60 - j80 \ \Omega$ at f = 2 GHz, then $R = 60 \ \Omega$ and C = 0.995 pF. The two tuning circuits are then shown in Figure 5.9b, and the reflection coefficient magnitudes are plotted versus frequency in Figure 5.9c. Note that the first solution has a much narrower bandwidth than the second (primed) solution, due to the fact that both stubs for the first solution are somewhat longer (and closer to $\lambda/2$) than the stubs of the second solution.



FIGURE 5.9 Continued. (b) The two double-stub tuning solutions. (c) Reflection coefficient magnitudes versus frequency for the tuning circuits of (b).

Analytic Solution

Just to the left of the first stub in Figure 5.7b, the admittance is

$$Y_1 = G_L + j(B_L + B_1), 5.17$$

where $Y_L = G_L + jB_L$ is the load admittance and B_1 is the susceptance of the first stub. After transforming through a length d of transmission line, the admittance just to the right of the second stub is

$$Y_2 = Y_0 \frac{G_L + j(B_L + B_1 + Y_0 t)}{Y_0 + jt(G_L + jB_L + jB_1)},$$
5.18

where $t = \tan \beta d$ and $Y_0 = 1/Z_0$. At this point, the real part of Y_2 must equal Y_0 , which leads to the equation

$$G_L^2 - G_L Y_0 \frac{1+t^2}{t^2} + \frac{(Y_0 - B_L t - B_1 t)^2}{t^2} = 0.$$
 5.19

5.4 The Quarter-Wave Transformer

Solving for G_L gives

$$G_L = Y_0 \frac{1+t^2}{2t^2} \left[1 \pm \sqrt{1 - \frac{4t^2(Y_0 - B_L t - B_1 t)^2}{Y_0^2(1+t^2)^2}} \right].$$
 5.20

Since G_L is real, the quantity within the square root must be nonnegative, and so

$$0 \le \frac{4t^2(Y_0 - B_L t - B_1 t)^2}{Y_0^2(1 + t^2)^2} \le 1.$$

This implies that

$$0 \le G_L \le Y_0 \frac{1+t^2}{t^2} = \frac{Y_0}{\sin^2 \beta d},$$
5.21

which gives the range on G_L that can be matched for a given stub spacing, d. After d has been fixed, the first stub susceptance can be determined from (5.19) as

$$B_1 = -B_L + \frac{Y_0 \pm \sqrt{(1+t^2)G_L Y_0 - G_L^2 t^2}}{t}.$$
 5.22

Then the second stub susceptance can be found from the negative of the imaginary part of (5.18) to be

$$B_2 = \frac{\pm Y_0 \sqrt{Y_0 G_L (1 + t^2) - G_L^2 t^2} + G_L Y_0}{G_L t}.$$
 5.23

The upper and lower signs in (5.22) and (5.23) correspond to the same solutions. The open-circuited stub length is found as

$$\frac{\ell_a}{\lambda} = \frac{1}{2\pi} \tan^{-1} \left(\frac{B}{Y_0} \right).$$
 5.24*a*

while the short-circuited stub length is found as

$$\frac{\ell_s}{\lambda} = \frac{-1}{2\pi} \tan^{-1}\left(\frac{Y_0}{B}\right).$$
 5.24b

where $B = B_1, B_2$.

<u>5.4</u>

- THE QUARTER-WAVE TRANSFORMER

As discussed in Section 2.5, the quarter-wave transformer is a simple and useful circuit for matching a real load impedance to a transmission line. An additional feature of the quarter-wave transformer is that it can be extended to multisection designs in a methodical manner, for broader bandwidth. If only a narrow band impedance match is required, a single-section transformer may suffice. But, as we will see in the next few

sections, multisection quarter-wave transformer designs can be synthesized to yield optimum matching characteristics over a desired frequency band. We will see in Chapter 8 that such networks are closely related to bandpass filters.

One drawback of the quarter-wave transformer is that it can only match a real load impedance. A complex load impedance can always be transformed to a real impedance, however, by using an appropriate length of transmission line between the load and the transformer, or an appropriate series or shunt reactive stub. These techniques will usually alter the frequency dependence of the equivalent load, which often has the effect of reducing the bandwidth of the match.

In Section 2.5 we analyzed the operation of the quarter-wave transformer from an impedance viewpoint and a multiple reflection viewpoint. Here we will concentrate on the bandwidth performance of the transformer, as a function of the load mismatch; this discussion will also serve as a prelude to the more general case of multisection transformers in the sections to follow.

The single-section quarter wave matching transformer circuit is shown in Figure 5.10. The characteristic impedance of the matching section is

$$Z_1 = \sqrt{Z_0 Z_L}.$$
 5.25

At the design frequency, f_0 , the electrical length of the matching section is $\lambda_0/4$, but at other frequencies the length is different, so a perfect match is no longer obtained. We will now derive an approximate expression for the mismatch versus frequency.

The input impedance seen looking into the matching section is

$$Z_{\rm in} = Z_1 \frac{Z_L + j Z_1 t}{Z_1 + j Z_L t},$$
 5.26

where $t = \tan \beta \ell = \tan \theta$, and $\beta \ell = \theta = \pi/2$ at the design frequency, f_0 . The reflection coefficient is then

$$\Gamma = \frac{Z_{\rm in} - Z_0}{Z_{\rm in} + Z_0} = \frac{Z_1(Z_L - Z_0) + jt(Z_1^2 - Z_0 Z_L)}{Z_1(Z_L + Z_0) + jt(Z_1^2 + Z_0 Z_L)}.$$
5.27

Since $Z_1^2 = Z_0 Z_L$, this reduces to

$$\Gamma = \frac{Z_L - Z_0}{Z_L + Z_0 + j2t\sqrt{Z_0 Z_L}}.$$
 5.28



FIGURE 5.10 A single-section quarter-wave matching transformer. $\ell = \lambda_0/4$ at the design frequency f_0 .
The reflection coefficient magnitude is

$$|\Gamma| = \frac{|Z_L - Z_0|}{\left[(Z_L + Z_0)^2 + 4t^2 Z_0 Z_L\right]^{1/2}}$$

= $\frac{1}{\left\{(Z_L + Z_0/Z_L - Z_0)^2 + [4t^2 Z_0 Z_L/(Z_L - Z_0)^2]\right\}^{1/2}}$
= $\frac{1}{\left\{1 + [4Z_0 Z_L/(Z_L - Z_0)^2] + [4Z_0 Z_L t^2/(Z_L - Z_0)^2]\right\}^{1/2}}$
= $\frac{1}{\left\{1 + [4Z_0 Z_L/(Z_L - Z_0)^2] \sec^2\theta\right\}^{1/2}},$ 5.29

since $1 + t^2 = 1 + \tan^2 \theta = \sec^2 \theta$.

Now if we assume that the frequency is near the design frequency, f_0 , then $\ell \simeq \lambda_0/4$ and $\theta \simeq \pi/2$. Then $\sec^2 \theta \gg 1$, and (5.29) simplifies to

$$|\Gamma| \simeq \frac{|Z_L - Z_0|}{2\sqrt{Z_0 Z_L}} |\cos\theta|, \quad \text{for } \theta \text{ near } \pi/2. \quad 5.30$$

This result gives the approximate mismatch of the quarter-wave transformer near the design frequency, as sketched in Figure 5.11.

If we set a maximum value, Γ_m , of the reflection coefficient magnitude that can be tolerated, then we can define the bandwidth of the matching transformer as

$$\Delta \theta = 2\left(\frac{\pi}{2} - \theta_m\right), \qquad 5.31$$

since the response of (5.29) is symmetric about $\theta = \pi/2$, and $\Gamma = \Gamma_m$ at $\theta = \theta_m$ and at $\theta = \pi - \theta_m$. Equating Γ_m to the exact expression for reflection coefficient magnitude



FIGURE 5.11 Approximate behavior of the reflection coefficient magnitude for a single-section quarter-wave transformer operating near its design frequency.

in (5.29) allows us to solve for θ_m :

$$\frac{1}{\Gamma_m^2} = 1 + \left(\frac{2\sqrt{Z_0 Z_L}}{Z_L - Z_0} \sec \theta_m\right)^2.$$
$$\cos \theta_m = \frac{\Gamma_m}{\sqrt{1 - \Gamma_m^2}} \frac{2\sqrt{Z_0 Z_L}}{|Z_L - Z_0|}.$$
5.32

от

If we assume TEM lines, then

$$\boldsymbol{\theta} = \boldsymbol{\beta}\boldsymbol{\ell} = \frac{2\pi f}{v_p} \frac{v_p}{4f_0} = \frac{\pi f}{2f_0},$$

therefore the frequency of the lower band edge at $\theta = \theta_m$ is

$$f_m \coloneqq rac{2 heta_m f_0}{\pi},$$

and the fractional bandwidth is, using (5.32),

$$\frac{\Delta f}{f_0} = \frac{2(f_0 - f_m)}{f_0} = 2 - \frac{2f_m}{f_0} = 2 - \frac{4\theta_m}{\pi}$$
$$= 2 - \frac{4}{\pi} \cos^{-1} \left[\frac{\Gamma_m}{\sqrt{1 - \Gamma_m^2}} \frac{2\sqrt{Z_0 Z_L}}{|Z_L - Z_0|} \right].$$
 5.33

The fractional bandwidth is usually expressed as a percentage, $100\Delta f/f_0$ %. Note that the bandwidth of the transformer increases as Z_L becomes closer to Z_0 (a less mismatched load).

The above results are strictly valid only for TEM lines. When non-TEM lines (such as waveguides) are used, the propagation constant is no longer a linear function of frequency, and the wave impedance will be frequency dependent. These factors serve to complicate the general behavior of guarter-wave transformers for non-TEM lines, but in practice the bandwidth of the transformer is often small enough so that these complications do not substantially affect the result. Another factor ignored in the above analysis is the effect of reactances associated with discontinuities when there is a step change in the dimensions of a transmission line. This can often be compensated for by making a small adjustment in the length of the matching section.

Figure 5.12 shows a plot of the reflection coefficient magnitude versus normalized frequency for various mismatched loads. Note the trend of increased bandwidth for smaller load mismatches.

P

EXAMPLE 5.5 Quarter-Wave Transformer Bandwidth

Design a single-section quarter-wave matching transformer to match a 10 Ω load to a 50 Ω line, at $f_0 = 3$ GHz. Determine the percent bandwidth for which the SWR ≤ 1.5 .



FIGURE 5.12 Reflection coefficient magnitude versus frequency for a single-section quarterwave matching transformer with various load mismatches.

Solution

From (5.25), the characteristic impedance of the matching section is

$$Z_1 = \sqrt{Z_0 Z_L} = \sqrt{(50)(10)} = 22.36 \ \Omega,$$

and the length of the matching section is $\lambda/4$ at 3 GHz. An SWR of 1.5 corresponds to a reflection coefficient magnitude of

$$\Gamma_m = \frac{\text{SWR} - 1}{\text{SWR} + 1} = \frac{1.5 - 1}{1.5 + 1} = 0.2.$$

The fractional bandwidth is computed from (5.33) as

$$\frac{\Delta f}{f_0} = 2 - \frac{4}{\pi} \cos^{-1} \left[\frac{\Gamma_m}{\sqrt{1 - \Gamma_m^2}} \frac{2\sqrt{Z_0 Z_L}}{|Z_L - Z_0|} \right]$$
$$= 2 - \frac{4}{\pi} \cos^{-1} \left[\frac{0.2}{\sqrt{1 - (0.2)^2}} \frac{2\sqrt{(50)(10)}}{|10 - 50|} \right]$$
$$= 0.29, \text{ or } 29\%.$$

5.5 THE THEORY OF SMALL REFLECTIONS

The quarter-wave transformer provides a simple means of matching any real load impedance to any line impedance. For applications requiring more bandwidth than a single quarter-wave section can provide, multisection transformers can be used. The design of such transformers is the subject of the next two sections, but prior to that material we need to derive some approximate results for the total reflection coefficient caused by the partial reflections from several small discontinuities. This topic is generally referred to as the theory of small reflections [1].



FIGURE 5.13 Partial reflections and transmissions on a single-section matching transformer.

Single-Section Transformer

Consider the single-section transformer shown in Figure 5.13; we will derive an approximate expression for the overall reflection coefficient Γ . The partial reflection and transmission coefficients are

$$\Gamma_1 = \frac{Z_2 - Z_1}{Z_2 + Z_1}, 5.34$$

$$\Gamma_2 = -\Gamma_1, \qquad 5.35$$

$$\Gamma_3 = \frac{Z_L - Z_2}{Z_L + Z_2}.$$
 5.36

$$T_{21} = 1 + \Gamma_1 = \frac{2Z_2}{Z_1 + Z_2},$$
 5.37

$$T_{12} = 1 + \Gamma_2 = \frac{2Z_1}{Z_1 + Z_2}.$$
 5.38

We can compute the total reflection, Γ , seen by the feed line by the impedance method or by the multiple reflection method, as discussed in Section 2.5. For our present purpose the latter technique is preferred, so we can express the total reflection as an infinite sum of partial reflections and transmissions as follows:

$$\Gamma = \Gamma_{1} + T_{12}T_{21}\Gamma_{3}e^{-2j\theta} + T_{12}T_{21}\Gamma_{3}^{2}\Gamma_{2}e^{-4j\theta} + \cdots$$

$$= \Gamma_{1} + T_{12}T_{21}\Gamma_{3}e^{-2j\theta}\sum_{n=0}^{\infty}\Gamma_{2}^{n}\Gamma_{3}^{n}e^{-2jn\theta}.$$
5.39

Using the geometric series,

$$\sum_{n=0}^{\infty} x^n = \frac{1}{1-x}, \quad \text{for } |x| < 1.$$

(5.39) can be expressed in closed form as

$$\Gamma = \Gamma_1 + \frac{T_{12}T_{21}\Gamma_3 e^{-2j\theta}}{1 - \Gamma_2\Gamma_3 e^{-2j\theta}}.$$
5.40

From (5.35), (5.37), and (5.38), we use $\Gamma_2 = -\Gamma_1$, $T_{21} = 1 + \Gamma_1$, and $T_{12} = 1 - \Gamma_1$ in (5.40) to give

$$\Gamma = \frac{\Gamma_1 + \Gamma_3 e^{-2j\theta}}{1 + \Gamma_1 \Gamma_3 e^{-2j\theta}}.$$
5.41

Now if the discontinuities between the impedances Z_1, Z_2 and Z_2, Z_L are small, then $|\Gamma_1\Gamma_3| \ll 1$, so we can approximate (5.41) as

$$\Gamma \simeq \Gamma_1 + \Gamma_3 e^{-2j\theta}.$$
 5.42

This result states the intuitive idea that the total reflection is dominated by the reflection from the initial discontinuity between Z_1 and Z_2 (Γ_1), and the first reflection from the discontinuity between Z_2 and Z_L ($\Gamma_3 e^{-2j\theta}$). The $e^{-2j\theta}$ term accounts for the phase delay when the incident wave travels up and down the line. The accuracy of this approximation is illustrated in Problem 5.14.

Multisection Transformer

Now consider the multisection transformer shown in Figure 5.14. This transformer consists of N equal-length (*commensurate*) sections of transmission lines. We will derive an approximate expression for the total reflection coefficient Γ .

Partial reflection coefficients can be defined at each junction, as follows:

$$\Gamma_0 = \frac{Z_1 - Z_0}{Z_1 + Z_0}.$$
 5.43*a*

$$\Gamma_n = \frac{Z_{n+1} - Z_n}{Z_{n+1} + Z_n},$$
 5.43b

$$\Gamma_N = \frac{Z_L - Z_N}{Z_L + Z_N}.$$
5.43c





We also assume that all Z_n increase or decrease monotonically across the transformer, and that Z_L is real. This implies that all Γ_n will be real, and of the same sign ($\Gamma_n > 0$ if $Z_L > Z_0$; $\Gamma_n < 0$ if $Z_L < Z_0$). Then using the results of the previous section, the overall reflection coefficient can be approximated as

$$\Gamma(\theta) = \Gamma_0 + \Gamma_1 e^{-2j\theta} + \Gamma_2 e^{-4j\theta} + \dots + \Gamma_N e^{-2jN\theta}.$$
 5.44

Further assume that the transformer can be made symmetrical, so that $\Gamma_0 = \Gamma_N, \Gamma_1 = \Gamma_{N-1}, \Gamma_2 = \Gamma_{N-2}$, etc. (Note that this does *not* imply that the Z_n s are symmetrical.) Then (5.44) can be written as

$$\Gamma(\theta) = e^{-jN\theta} \{ \Gamma_0[e^{jN\theta} + e^{-jN\theta}] + \Gamma_1[e^{j(N-2)\theta} + e^{-j(N-2)\theta}] + \cdots \}.$$
 5.45

If N is odd, the last term is $\Gamma_{(N-1)/2}(e^{j\theta} + e^{-j\theta})$, while if N is even the last term is $\Gamma_{N/2}$. Equation (5.45) is then seen to be of the form of a finite Fourier cosine series in θ , which can be written as

$$\Gamma(\theta) = 2e^{-jN\theta} \Big[\Gamma_0 \cos N\theta + \Gamma_1 \cos(N-2)\theta + \dots + \Gamma_n \cos(N-2n)\theta \\ + \dots + \frac{1}{2}\Gamma_{N/2} \Big], \quad \text{for } N \text{ even}, \qquad 5.46a$$

$$\Gamma(\theta) = 2e^{-jN\theta} [\Gamma_0 \cos N\theta + \Gamma_1 \cos(N-2)\theta + \dots + \Gamma_n \cos(N-2n)\theta \\ + \dots + \Gamma_{(N-1)/2} \cos \theta], \quad \text{for } N \text{ odd}. \qquad 5.46b$$

The importance of these results lies in the fact that we can synthesize any desired reflection coefficient response as a function of frequency (θ), by properly choosing the Γ_n s and using enough sections (N). This should be clear from the realization that a Fourier series can represent an arbitrary smooth function, if enough terms are used. In the next two sections we will show how to use this theory to design multisection transformers for two of the most commonly used passband responses: the binomial (maximally flat) response, and the Chebyshev (equal ripple) response.

5.6 BINOMIAL MULTISECTION MATCHING TRANSFORMERS

The passband response of a binomial matching transformer is optimum in the sense that, for a given number of sections, the response is as flat as possible near the design frequency. Thus, such a transformer is also known as maximally flat. This type of response is designed, for an N-section transformer, by setting the first N - 1 derivatives of $|\Gamma(\theta)|$ to zero, at the center frequency f_0 . Such a response can be obtained if we let

$$\Gamma(\theta) = A(1 + e^{-2j\theta})^N.$$
 5.47

Then the magnitude $|\Gamma(\theta)|$ is

$$|\Gamma(\theta)| = |A| |e^{-j\theta|^{N}} |e^{j\theta} + e^{-j\theta}|^{N}$$
$$= 2^{N} |A| |\cos \theta|^{N}$$
5.48

Note that $|\Gamma(\theta)| = 0$ for $\theta = \pi/2$, and that $(d^n |\Gamma(\theta)|)/d\theta^n = 0$ at $\theta = \pi/2$ for n = 1, 2, ..., N-1. ($\theta = \pi/2$ corresponds to the center frequency f_0 , for which $\ell = \lambda/4$ and $\theta = \beta \ell = \pi/2$.)

We can determine the constant A by letting $f \to 0$. Then $\theta = \beta \ell = 0$, and (5.47) reduces to

$$\Gamma(0) = 2^N A = \frac{Z_L - Z_0}{Z_L + Z_0}$$

since for f = 0 all sections are of zero electrical length. Thus the constant A can be written as

$$A = 2^{-N} \frac{Z_L - Z_0}{Z_L + Z_0}.$$
 5.49

Now expand $\Gamma(\theta)$ in (5.47) according to the binomial expansion:

$$\Gamma(\theta) = A(1 + e^{-2j\theta})^N = A \sum_{n=0}^N C_n^N e^{-2jn\theta},$$
 \$.50

$$C_n^N \approx \frac{N!}{(N-n)!n!},$$
5.51

where

are the binomial coefficients. Note that $C_n^N = C_{N-n}^N$, $C_0^N = 1$, and $C_1^N = N = C_{N-1}^N$. The key step is now to equate the desired passband response as given in (5.50), to the actual response as given (approximately) by (5.44):

$$\Gamma(\theta) = A \sum_{n=0}^{N} C_n^N e^{-2jn\theta} = \Gamma_0 + \Gamma_1 e^{-2j\theta} + \Gamma_2 e^{-4j\theta} + \dots + \Gamma_N e^{-2jN\theta}.$$

This shows that the Γ_n must be chosen as

$$\Gamma_n = AC_n^N.$$
 5.52

where A is given by (5.49), and C_n^N is a binomial coefficient.

At this point, the characteristic impedances Z_n can be found via (5.43), but a simpler solution can be obtained using the following approximation [1]. Since we assumed that the Γ_n are small, we can write

$$\Gamma_n = \frac{Z_{n+1} - Z_n}{Z_{n+1} + Z_n} \simeq \frac{1}{2} \ln \frac{Z_{n+1}}{Z_n},$$

since $\ln x \simeq 2(x-1)/(x+1)$. Then, using (5.52) and (5.49) gives

$$\ln \frac{Z_{n+1}}{Z_n} \simeq 2\Gamma_n = 2AC_n^N = 2(2^{-N})\frac{Z_L - Z_0}{Z_L + Z_0}C_n^N \simeq 2^{-N}C_n^N \ln \frac{Z_L}{Z_0}, \qquad 5.53$$

which can be used to find Z_{n+1} , starting with n = 0. This technique has the advantage of ensuring self-consistency, in that Z_{N+1} computed from (5.53) will be equal to Z_L , as it should.

Exact results, including the effect of multiple reflections in each section, can be found by using the transmission line equations for each section and numerically solving for the characteristic impedances [2]. The results of such calculations are listed in Table 5.1, which give the exact line impedances for N = 2, 3, 4, 5, and 6 section binomial matching transformers, for various ratios of load impedance, Z_L , to feed line impedance, Z_0 . The table gives results only for $Z_L/Z_0 > 1$; if $Z_L/Z_0 < 1$, the results for Z_0/Z_L should be used, but with Z_1 starting at the load end. This is because the solution is symmetric about $Z_L/Z_0 = 1$; the same transformer that matches Z_L to Z_0 can be reversed and used to match Z_0 to Z_L . More extensive tables can be found in reference [2].

The bandwidth of the binomial transformer can be evaluated as follows. As in Section 5.4, let Γ_m be the maximum value of reflection coefficient that can be tolerated over the passband. Then from (5.48).

$$\Gamma_m = 2^N |A| \cos^N \theta_m.$$

where $\theta_m < \pi/2$ is the lower edge of the passband, as shown in Figure 5.11. Thus,

$$\theta_m = \cos^{-1} \left[\frac{1}{2} \left(\frac{\Gamma_m}{|A|} \right)^{1/N} \right], \qquad 5.54$$

and using (5.33) gives the fractional bandwidth as

$$\frac{\Delta f}{f_0} = \frac{2(f_0 - f_m)}{f_0} = 2 - \frac{4\theta_m}{\pi}$$
$$= 2 - \frac{4}{\pi} \cos^{-1} \left[\frac{1}{2} \left(\frac{\Gamma_m}{|A|} \right)^{1/N} \right].$$
 5.55

Ø

EXAMPLE 5.6 Binomial Transformer Design

Design a three-section binomial transformer to match a 50 Ω load to a 100 Ω line, and calculate the bandwidth for $\Gamma_m = 0.05$. Plot the reflection coefficient magnitude versus normalized frequency for the exact designs using 1, 2, 3, 4, and 5 sections.

Solution

For $N \approx 3$, $Z_L = 50 \ \Omega$, $Z_0 \approx 100 \ \Omega$ we have, from (5.49) and (5.53),

$$A = 2^{-N} \frac{Z_L - Z_0}{Z_L + Z_0} \simeq \frac{1}{2^{N+1}} \ln \frac{Z_L}{Z_0} = -0.0433.$$

From (5.55) the bandwidth is

$$\frac{\Delta f}{f_0} = 2 - \frac{4}{\pi} \cos^{-1} \left[\frac{1}{2} \left(\frac{\Gamma_m}{|A|} \right)^{1/N} \right]$$
$$= 2 - \frac{4}{\pi} \cos^{-1} \left[\frac{1}{2} \left(\frac{0.05}{0.0433} \right)^{1/3} \right] = 0.70, \text{ or } 70\%.$$

	 	N =	4		N=3			*	N = 4		
Z_L/Z_0	2	1/20	Z_2/Z_0	Z1/2	$Z_0 = Z_2/Z_0$	Z_4/Z_{41}	$Z^{I}/2$	Zo Z2/2	0 Z3/2	0 ZA/	1 55
0.1	1''	0000	00007	1,000	X0 1.0000	1.0000	1.00(0001 00	0001 0	00.1	18
1.5		1067 1	.3554	1.052	20 1.2247	1.4259	1.02	57 1.135	1 1.321	5 1.46	24
2.0		1892 1	8189	360.1	17 1.4142	1.8337	1.04	44 1.242	1 1.610	16.1 2	8
3.0	_	3161 2	.2795	1.147	79 1.7321	2.6135	1.07	18 1.410	5 2.126	9 2.79	Q
4.0	· [4142 2	.8285	1.190	7 2.0000	3.3594	160'T	19 1.544	2 2.590	3 3.66	33
6.0		5651 3	,8336	1.254	14 2.4495	4.7832	I.12	15 1.755	3 3.418	2 5.351	Q
8.0	-	5818 4	.7568	1.302	22 2.8284	6.1434	1.14.	36 1.923	2 4.159	7 6.99	55
10.0		7783 5	.6233	1.340	3.1623	7.4577	1.16	13 2.065	1 4.842	4 8.61	10
			N = 5				:	N =	Q		I
Z_L/Z_0	Z_{1}/Z_{0}	Z_2/Z_0	Z_3/Z_0	Z_A/Z_0	Z_5/Z_0	Z_1/Z_0	Z_2/Z_0	Z_{3}/Z_{0}	Z_A/Z_0	Z_{5}/Z_{0}	Z_{0}/Z_{0}
1.0	1.0000	00001	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000
1.5	1.0128	1.0790	1.2247	1.3902	1.4810	L.0064	1.0454	1.1496	1.3048	1.4349	1.4905
2.0	1.0220	1.1391	1.4142	1.7558	1.9569	01101	1.0790	1.2693	1.5757	1.8536	1.9782
3.0	1.0354	1.2300	1.7321	2.4390	2.8974	1.0176	1.1288	1.4599	2.0549	2.6577	2.9481
4.0	1.0452	1.2995	2.0000	3.0781	3.8270	1.0225	1,1661	1.6129	2.4800	3.4302	3.9120
6.0	1.0596	1.4055	2.4495	4.2689	5.6625	1.0296	1.2219	1.8573	3.2305	4.9104	5.8275
8.0	1.0703	1.4870	2.8284	5.3800	7.4745	1.0349	1.2640	2.0539	3.8950	6.3291	7.7302
10.0	1.0789	1.5541	3.1623	6.4346	9.2687	1.0392	1.2982	2.2215	4.5015	7.7030	9.6228

TABLE 5.1 Binomial Transformer Design

281

The necessary binomial coefficients are

$$C_0^3 = \frac{3!}{3!0!} = 1,$$

$$C_1^3 = \frac{3!}{2!1!} = 3,$$

$$C_2^3 = \frac{3!}{1!2!} = 3.$$

Then using (5.53) gives the required characteristic impedances as

$$n = 0: \ln Z_1 = \ln Z_0 + 2^{-N} C_0^3 \ln \frac{Z_L}{Z_0}$$

= $\ln 100 + 2^{-3}(1) \ln \frac{50}{100} = 4.518$,
 $Z_1 = 91.7 \Omega$:
 $n = 1: \ln Z_2 = \ln Z_1 + 2^{-N} C_1^3 \ln \frac{Z_L}{Z_0}$
= $\ln 91.7 + 2^{-3}(3) \ln \frac{50}{100} = 4.26$.
 $Z_2 = 70.7 \Omega$;
 $n = 2: \ln Z_3 = \ln Z_2 + 2^{-N} C_2^3 \ln \frac{Z_L}{Z_0}$
= $\ln 70.7 + 2^{-3}(3) \ln \frac{50}{100} = 4.00$,
 $Z_3 = 54.5 \Omega$.

To use the data in Table 5.1, we reverse the source and load impedances and consider the problem of matching a 100 Ω load to a 50 Ω line. Then $Z_L/Z_0 = 2.0$, and we obtain the exact characteristic impedances as $Z_1 = 91.7 \ \Omega$, $Z_2 = 70.7 \ \Omega$, and $Z_3 = 54.5 \ \Omega$, which agree with the approximate results to three significant digits. Figure 5.15 shows the reflection coefficient magnitude versus frequency for exact designs using N = 1, 2, 3, 4, and 5 sections. Observe that greater bandwidth is obtained for transformers using more sections.

5.7 ____ CHEBYSHEV MULTISECTION MATCHING TRANSFORMERS

In contrast with the binomial matching transformer, the Chebyshev transformer optimizes bandwidth at the expense of passband ripple. If such a passband characteristic can be tolerated, the bandwidth of the Chebyshev transformer will be substantially better than that of the binomial transformer, for a given number of sections. The Chebyshev transformer is designed by equating $\Gamma(\theta)$ to a Chebyshev polynomial, which has the



FIGURE 5.15 Reflection coefficient magnitude versus frequency for multisection binomial matching transformers of Example 5.6. $Z_L = 50 \ \Omega$ and $Z_0 = 100 \ \Omega$.

optimum characteristics needed for this type of transformer. Thus we will first discuss the properties of the Chebyshev polynomials, and then derive a design procedure for Chebyshev matching transformers using the small reflection theory of Section 5.5.

Chebyshev Polynomials

The nth order Chebyshev polynomial is a polynomial of degree n, and is denoted by $T_n(x)$. The first four Chebyshev polynomials are

$$T_1(x) = x, \qquad 5.56a$$

$$T_2(x) = 2x^2 - 1, 5.56b$$

$$T_3(x) = 4x^3 - 3x, \qquad 5.56c$$

$$T_4(x) = 8x^4 - 8x^2 + 1. 5.56d$$

Higher-order polynomials can be found using the following recurrence formula:

$$T_n(x) = 2xT_{n-1}(x) - T_{n-2}(x).$$
 5.57

The first four Chebyshev polynomials are plotted in Figure 5.16, from which the following very useful properties of Chebyshev polynomials can be noted:

- For $-1 \le x \le 1$, $|T_n(x)| \le 1$. In this range, the Chebyshev polynomials oscillate between ± 1 . This is the equal ripple property, and this region will be mapped to the passband of the matching transformer.
- For |x| > 1, $|T_n(x)| > 1$. This region will map to the frequency range outside the passband.
- For |x| > 1, the $|T_n(x)|$ increases faster with x as n increases.



FIGURE 5.16 The first four Chebyshev polynomials, $T_n(x)$,

Now let $x = \cos \theta$ for |x| < 1. Then it can be shown that the Chebyshev polynomials can be expressed as

$$T_n(\cos \theta) = \cos n \theta,$$

or more generally as

$$T_n(x) = \cos(n\cos^{-1}x),$$
 for $|x| < 1,$ 5.58a

$$T_n(x) = \cosh(n\cosh^{-1}x), \quad \text{for } |x| > 1.$$
 5.58b

We desire equal ripple in the passband of the transformer, so it is necessary to map θ_m to x = 1 and $\pi - \theta_m$ to x = -1, where θ_m and $\pi - \theta_m$ are the lower and upper edges of the passband, as shown in Figure 5.11. This can be accomplished by replacing $\cos \theta$ in (5.58a) with $\cos \theta / \cos \theta_m$:

$$T_n\left(\frac{\cos\theta}{\cos\theta_m}\right) = T_n(\sec\theta_m\cos\theta) = \cos n \left[\cos^{-1}\left(\frac{\cos\theta}{\cos\theta_m}\right)\right].$$
 5.59

Then $|\sec \theta_m \cos \theta| \le 1$ for $\theta_m < \theta < \pi - \theta_m$, so $|T_n(\sec \theta_m \cos \theta)| \le 1$ over this same range.

Since $\cos^n \theta$ can be expanded into a sum of terms of the form $\cos(n - 2m)\theta$, the Chebyshev polynomials of (5.56) can be rewritten in the following useful form:

$$T_1(\sec\theta_m\cos\theta) = \sec\theta_m\cos\theta, \qquad 5.60a$$

$$T_2(\sec\theta_m\cos\theta) = \sec^2\theta_m(1+\cos 2\theta) - 1, \qquad 5.60b$$

$$T_{3}(\sec \theta_{m} \cos \theta) = \sec^{3} \theta_{m}(\cos 3\theta + 3\cos \theta) - 3\sec \theta_{m} \cos \theta, \qquad 5.60c$$
$$T_{4}(\sec \theta_{m} \cos \theta) = \sec^{4} \theta_{m}(\cos 4\theta + 4\cos 2\theta + 3)$$
$$- 4\sec^{2} \theta_{m}(\cos 2\theta + 1) + 1. \qquad 5.60d$$

The above results can be used to design matching transformers with up to four sections, and will also be used in later chapters for the design of directional couplers and filters.

Design of Chebyshev Transformers

We can now synthesize a Chebyshev equal-ripple passband by making $\Gamma(\theta)$ proportional to $T_N(\sec \theta_m \cos \theta)$, where N is the number of sections in the transformer. Thus, using (5.46).

$$\Gamma(\theta) = 2e^{-jN\theta} [\Gamma_0 \cos N\theta + \Gamma_1 \cos(N-2)\theta + \dots + \Gamma_n \cos(N-2n)\theta + \dots]$$

= $Ae^{-jN\theta} T_N (\sec \theta_m \cos \theta),$ 5.61

where the last term in the series of (5.61) is $(1/2)\Gamma_{N/2}$ for N even and $\Gamma_{(N-1)/2} \cos \theta$ for N odd. As in the binomial transformer case, we can find the constant A by letting $\theta = 0$, corresponding to zero frequency. Thus,

$$\Gamma(0) = \frac{Z_L - Z_0}{Z_L + Z_0} = AT_N(\sec\theta_m).$$

so we have

$$A = \frac{Z_L - Z_0}{Z_L + Z_0} \frac{1}{T_N(\sec\theta_m)}.$$
 5.62

Now if the maximum allowable reflection coefficient magnitude in the passband is Γ_m , then from (5.61) $\Gamma_m = |A|$, since the maximum value of $T_n(\sec \theta_m \cos \theta)$ in the passband is unity. Then, from (5.62) and the approximations introduced in Section 5.6, θ_m is determined as

$$T_N(\sec \theta_m) = \frac{1}{\Gamma_m} \left| \frac{Z_L - Z_0}{Z_L + Z_0} \right| \simeq \frac{1}{2\Gamma_m} \left| \ln \frac{Z_L}{Z_0} \right|,$$

or, using (5.58b),

$$\sec \theta_m = \cosh \left[\frac{1}{N} \cosh^{-1} \left(\frac{1}{\Gamma_m} \left| \frac{Z_L - Z_0}{Z_L + Z_0} \right| \right) \right]$$
$$\simeq \cosh \left[\frac{1}{N} \cosh^{-1} \left(\left| \frac{\ln Z_L / Z_0}{2\Gamma_m} \right| \right) \right]$$
5.63

Once θ_m is known, the fractional bandwidth can be calculated from (5.33) as

$$\frac{\Delta f}{f_0} = 2 - \frac{4\theta_m}{\pi}.$$
 5.64

From (5.61), the Γ_n can be determined using the results of (5.60) to expand $T_N(\sec\theta_m$ $\cos\theta$ and equating similar terms of the form $\cos(N-2n)\theta$. The characteristic impedances Z_n can then be found from (5.43) although, as in the case of the binomial transformer, accuracy can be improved and self-consistency can be achieved by using the approximation that

$$\Gamma_n \simeq \frac{1}{2} \ln \frac{Z_{n+1}}{Z_n}.$$

This procedure will be illustrated in Example 5.7.

The above results are approximate pecause of the reliance on small reflection theory, but are general enough to design transformers with an arbitrary ripple level, Γ_m . Table 5.2 gives exact results [2] for a few specific values of Γ_m , for N = 2, 3, and 4 sections; more extensive tables can be found in reference [2].

TABLE 5.2 Chebyshev Transformer Design

<u> </u>		= 2		= 3
	$\Gamma_m = 0.05$	$\Gamma_m = 0.20$	$\Gamma_m = 0.05$	$\Gamma_m = 0.20$
$Z_{ m L}/Z_0$	$\overline{Z_1/Z_0 - Z_2/Z_0}$	$\overline{Z_1/Z_0-Z_1/Z_0}$	$Z_1/Z_0 = Z_2/Z_0 = Z_3/Z_0$	$Z_1/Z_0 = Z_2/Z_0 = Z_3/2$
1.0	1.0000 1.0000	1.0000 1.0000	1.0000 1.0000 1.0000	1.0000 1.0000 1.0000
1.5	1.1347 1.3219	1.2247 1.2247	1.1029 1.2247 1.3601	1.2247 1.2247 1.2247
2 .0	1.2193 1.6402	1.3161 1.5197	1.1475 1.4142 1.7429	1.2855 1.4142 1.5558
3.0	1.3494 2.2232	1.4565 2.0598	1.2171 1.7321 2.4649	1.3743 1.7321 2.1829
4.0	1.4500 2.7585	1.5651 2.5558	1.2662 2 0000 3.1591	1.4333 2.0000 2.790
6.0	1 6047 3 7389	1.7321 3.4641	1.3383 2.4495 4.4833	1.5193 2.4495 3.9492
80	1 7244 4 6393	1.8612 4 2983	1.3944 2.8284 5.7372	1.5766 2.8284 5.0742
10.0	1.8233 5.4845	1.9680 5.0813	1.4385 3.1623 6.9517	1.6415 3.1623 6.0920

N = 4

				<u>N</u> ·	= 4			
		$\Gamma_m =$	= 0.05			$\Gamma_m =$	= 0.20	
Z_L/Z_0	$\overline{Z_1/Z_0}$	Z_2/Z_0	Z_{1}/Z_{0}	Z_4/Z_0	Z_1/Z_0	Z_{2}/Z_{0}	Z_3/Z_0	Z_4/Z_0
1.0	1.0000	1.0000	1.0000	J.0000	1.0000	0000.1	1.0000	1.0000
1.5	1.0892	1.1742	1.2775	1.3772	1.2247	l.2247	1.2247	1.2247
2.0	1.1201	1.2979	1.5409	1.7 85 5	1.2727	1.3634	1.4669	1.5715
3.0	1.1586	1.4876	2.0167	z.5893	1.4879	1.5819	1.8965	2.0163
4.0	1.1906	1.6414	2.4369	3.359 7	1.3692	1.7490	2.2870	2.9214
6 .0	1.2290	1.8773	3.1961	4.8820	1.4415	2.0231	2.9657	4.1623
8.0	1.2583	2,0657	3.8728	6 .3578	1.4914	2.2428	3.5670	5.3641
10.0	1.2832	2.2268	4.4907	7.7930	1.5163	2.4210	4.1305	6.5950

EXAMPLE 5.7 Chebyshev Transformer Design

Design a three-section Chebyshev transformer to match a 100 Ω load to a 50 Ω line, with $\Gamma_m = 0.05$, using the above theory. Plot the reflection coefficient magnitude versus normalized frequency for exact designs using 1, 2, 3, and 4 sections.

Solution From (5.61) with N = 3,

$$\Gamma(\theta) = 2e^{-j3\theta} [\Gamma_0 \cos 3\theta + \Gamma_1 \cos \theta] = Ae^{-j3\theta} T_3(\sec \theta_m \cos \theta).$$

Then, $A = \Gamma_m = 0.05$, and from (5.63),

$$\sec \theta_m = \cosh \left[\frac{1}{N} \cosh^{-1} \left(\frac{\ln Z_L / Z_0}{2\Gamma_m} \right) \right]$$
$$= \cosh \left[\frac{1}{3} \cosh^{-1} \left(\frac{\ln(100/50)}{2(0.05)} \right) \right]$$
$$= 1.408,$$

so, $\theta_m = 44.7^\circ$.

Using (5.60c) for T_3 gives

 $2[\Gamma_0 \cos 3\theta + \Gamma_1 \cos \theta] = A \sec^3 \theta_m (\cos 3\theta + 3\cos \theta) - 3A \sec \theta_m \cos \theta.$

Equating similar terms in $\cos n\theta$ gives the following results:

$$\cos 3\theta: \quad 2\Gamma_0 = A \sec^3 \theta_m;$$

$$\Gamma_0 = 0.0698;$$

$$\cos \theta: \quad 2\Gamma_1 = 3A(\sec^3 \theta_m - \sec \theta_m);$$

$$\Gamma_1 = 0.1037.$$

From symmetry we also have that

$$\Gamma_3=\Gamma_0=0.0698,$$
 and
$$\Gamma_2=\Gamma_1=0.1037.$$

Then the characteristic impendances are:

$$n = 0: \quad \ln Z_1 = \ln Z_0 + 2\Gamma_0$$

= $\ln 50 + 2(0.0698) = 4.051$
 $Z_1 = 57.5 \ \Omega$
 $n = 1: \quad \ln Z_2 = \ln Z_1 + 2\Gamma_1$
= $\ln 57.5 + 2(0.1037) = 4.259$
 $Z_2 = 70.7 \ \Omega$



FIGURE 5.17 Reflection coefficient magnitude versus frequency for the multisection matching transformers of Example 5.7.

$$n = 2$$
: $\ln Z_3 = \ln Z_2 + 2\Gamma_2$
= $\ln 70.7 + 2(0.1037) = 4.466$
 $Z_3 = 87.0 \ \Omega$

These values can be compared to the exact values from Table 5.2 of $Z_1 = 57.37 \ \Omega$, $Z_2 = 70.71 \ \Omega$, and $Z_3 = 87.15 \ \Omega$. The bandwidth, from (5.64), is

$$\frac{\Delta f}{f_0} = 2 - \frac{4\theta_m}{\pi} = 2 - 4\left(\frac{44.7^\circ}{180^\circ}\right) = 1.01,$$

or 101%. This is significantly greater than the bandwidth of the binomial transformer of Example 5.6 (70%), which was for the same type of mismatch. The trade-off, of course, is a nonzero ripple in the passband of the Chebyshev transformer.

Figure 5.17 shows reflection coefficient magnitudes versus frequency for the exact designs from Table 5.2 for N = 1, 2, 3 and 4 sections.

5.8 TAPERED LINES

In the preceding sections we discussed how an arbitrary real load impedance could be matched to a line over a desired bandwidth by using multisection matching transformers. As the number, N, of discrete sections increases, the step changes in characteristic impedance between the sections become smaller. Thus, in the limit of an infinite number of sections, we approach a continuously tapered line. In practice, of course, a matching transformer must be of finite length, often no more than a few sections long. But instead of discrete sections, the line can be continuously tapered, as suggested in Figure 5.18a. Then by changing the type of taper, we can obtain different passband characteristics.



FIGURE 5.18 A tapered transmission line matching section and the model for an incremental length of tapered line. (a) The tapered transmission line matching section. (b) Model for an incremental step change in impedance of the tapered line.

In this section we will derive an approximate theory, based on the theory of small reflections, to predict the reflection coefficient response as a function of the impedance taper, Z(z). We will then apply these results to a few common types of tapers.

Consider the continuously tapered line of Figure 5.18a as being made up of a number of incremental sections of length Δz , with an impedance change $\Delta Z(z)$ from one section to the next, as shown in Figure 5.18b. Then the incremental reflection coefficient from the step at z is given by

$$\Delta\Gamma = \frac{(Z + \Delta Z) - Z}{(Z + \Delta Z) + Z} \simeq \frac{\Delta Z}{2Z}.$$
 5.65

In the limit as $\Delta z \rightarrow 0$, we have an exact differential:

$$d\Gamma = \frac{dZ}{2Z} = \frac{1}{2} \frac{d(\ln Z/Z_0)}{dz} dz,$$

$$\frac{d(\ln f(z))}{dz} = \frac{1}{f} \frac{df(z)}{dz}.$$
5.66

since

Then, by using the theory of small reflections, the total reflection coefficient at z = 0can be found by summing all the partial reflections with their appropriate phase shifts:

$$\Gamma(\theta) = \frac{1}{2} \int_{z=0}^{L} e^{-2j\beta z} \frac{d}{dz} \ln\left(\frac{Z}{Z_0}\right) dz, \qquad 5.67$$

where $\theta = 2\beta \ell$. So if Z(z) is known, $\Gamma(\theta)$ can be found as a function of frequency. Alternatively, if $\Gamma(\theta)$ is specified, then in principle Z(z) can be found. This latter procedure is difficult, and is generally avoided in practice; the reader is referred to references [1], [4] for further discussion along these lines. Here we will consider three special cases of Z(z) impedance tapers, and evaluate the resulting responses.

Exponential Taper

290

Consider first an exponential taper, where

$$Z(z) = Z_0 e^{az}, \quad \text{for } 0 < z < L,$$
 5.68

as indicated in Figure 5.19a. At z = 0. $Z(0) = Z_0$, as desired. At z = L, we wish to have $Z(L) = Z_L \approx Z_0 e^{aL}$, which determines the constant a as

$$a = \frac{1}{L} \ln\left(\frac{Z_L}{Z_0}\right).$$
 5.69

We now find $\Gamma(\theta)$ by using (5.68) and (5.69) in (5.67):

$$\Gamma = \frac{1}{2} \int_0^L e^{-2j\beta z} \frac{d}{dz} (\ln e^{\alpha z}) dz$$
$$= \frac{\ln Z_L/Z_0}{2L} \int_0^L e^{-2j\beta z} dz$$
$$= \frac{\ln Z_L/Z_0}{2} e^{-j\beta L} \frac{\sin \beta L}{\beta L}.$$
 5.70





A matching section with an exponential impedance taper. (a) Variation of impedance. (b) Resulting reflection coefficient magnitude response.

Observe that this derivation assumes that β , the propagation constant of the tapered line, is not a function of z—an assumption which is generally valid only for TEM lines.

The magnitude of the reflection coefficient in (5.70) is sketched in Figure 5.19b; note that the peaks in $|\Gamma|$ decrease with increasing length, as one might expect, and that the length should be greater than $\lambda/2$ ($\beta L > \pi$) to minimize the mismatch at low frequencies.

Triangular Taper

Next consider a triangular taper for $(d \ln Z/Z_0)/dz$, that is,

$$Z(z) = \begin{cases} Z_0 e^{2(z/L)^2 \ln Z_L/Z_0} & \text{for } 0 \le z \le L/2 \\ Z_0 e^{(4z/L - 2z^2/L^2 - 1) \ln Z_L/Z_0} & \text{for } L/2 \le z \le L. \end{cases}$$
5.71

Then,

$$\frac{d(\ln Z/Z_0)}{dz} = \begin{cases} 4z/L^2 \ln Z_L/Z_0 & \text{for } 0 \le z \le L/2\\ (4/L - 4z/L^2) \ln Z_L/Z_0 & \text{for } L/2 \le z \le L. \end{cases}$$
5.72

Z(z) is plotted in Figure 5.20a. Evaluating Γ from (5.67) gives

$$\Gamma(\theta) = \frac{1}{2} e^{-j\beta L} \ln\left(\frac{Z_L}{Z_0}\right) \left[\frac{\sin(\beta L/2)}{\beta L/2}\right]^2.$$
 5.73

The magnitude of this result is sketched in Figure 5.20b. Note that, for $\beta L > 2\pi$, the peaks of the triangular taper are lower than the corresponding peaks of the exponential case. But the first null for the triangular taper occurs at $\beta L = 2\pi$, whereas for the exponential taper it occurs at $\beta L = \pi$.

Klopfenstein Taper

Considering the fact that there is an infinite number of possibilities for choosing an impedance matching taper, it is logical to ask if there is a design which is "best." For a given taper length (greater than a critical value), the Klopfenstein impedance taper [4], [5] has been shown to be optimum in the sense that the reflection coefficient is minimum over the passband. Alternatively, for a maximum reflection coefficient specification in the passband, the Klopfenstein taper yields the shortest matching section.

The Klopfenstein taper is derived from a stepped Chebyshev transformer as the number of sections increases to infinity, and is analogous to the Taylor distribution of antenna array theory. We will not present the details of this derivation, which can be found in references [1], [4]; only the necessary results for the design of Klopfenstein tapers are given below.

The logarithm of the characteristic impedance variation for the Klopfenstein taper is given by

$$\ln Z(z) = \frac{1}{2} \ln Z_0 Z_L + \frac{\Gamma_0}{\cosh A} A^2 \phi(2z/L - 1, A), \quad \text{for } 0 \le z \le L, \quad 5.74$$



FIGURE 5.20 A matching section with a triangular taper for $d(\ln Z/Z_0)/dz$. (a) Variation of impedance. (b) Resulting reflection coefficient magnitude response.

where the function $\phi(x, A)$ is defined as

$$\phi(x,A) = -\phi(-x,A) = \int_0^x \frac{I_1(A\sqrt{1-y^2})}{A\sqrt{1-y^2}} dy, \quad \text{for } |x| \le 1, \quad 5.75$$

where $I_1(x)$ is the modified Bessel function. This function takes the following special values:

$$\phi(0, A) = 0$$

$$\phi(x, 0) = \frac{x}{2}$$

$$\phi(1, A) = \frac{\cosh A - 1}{A^2}$$

but otherwise must be calculated numerically. A very simple and efficient method for doing this is available [6].

The resulting reflection coefficient is given by

$$\Gamma(\theta) = \Gamma_0 e^{-j\beta L} \frac{\cos\sqrt{(\beta L)^2 - A^2}}{\cosh A}, \quad \text{for } \beta L > A, \quad 5.76$$

If $\beta L < A$, the $\cos \sqrt{(\beta L)^2 - A^2}$ term becomes $\cosh \sqrt{A^2 - (\beta L)^2}$.

In (5.74) and (5.76), Γ_0 is the reflection coefficient at zero frequency, given as

$$\Gamma_0 = \frac{Z_L - Z_0}{Z_L + Z_0} \simeq \frac{1}{2} \ln\left(\frac{Z_L}{Z_0}\right).$$
 5.77

The passband is defined as $\beta L \ge A$, and so the maximum ripple in the passband is

$$\Gamma_m = \frac{\Gamma_0}{\cosh A},$$
5.78

because $\Gamma(\theta)$ oscillates between $\pm \Gamma_0/\cosh A$ for $\beta L > A$.

It is interesting to note that the impedance taper of (5.74) has steps at z = 0 and L (the ends of the tapered section), and so does not smoothly join the source and load impedances. A typical Klopfenstein impedance taper and its response are given in the following example.

EXAMPLE 5.8 Design of Tapered Matching Sections

Design a triangular taper, an exponential taper, and a Klopfenstein taper (with $\Gamma_m = 0.02$) to match a 50 Ω load to a 100 Ω line. Plot the impedance variations and resulting reflection coefficient magnitudes versus βL .

Solution

Triangular taper: From (5.71) the impedance variation is

$$Z(z) = Z_0 \begin{cases} e^{2(z/L)^2 \ln Z_L/Z_0} & \text{for } 0 \le z \le L/2 \\ e^{(4z/L - 2z^2/L^2 - 1) \ln Z_L/Z_0} & \text{for } L/2 \le z \le L, \end{cases}$$

with $Z_0 = 100 \ \Omega$ and $Z_L = 50 \ \Omega$. The resulting reflection coefficient response is given by (5.73):

$$|\Gamma(\theta)| = \frac{1}{2} \ln\left(\frac{Z_L}{Z_0}\right) \left[\frac{\sin(\beta L/2)}{\beta L/2}\right]^2$$

Exponential taper: From (5.68) the impedance variation is

$$Z(z) = Z_0 e^{az}, \qquad \text{for } 0 < z < L$$

with $a = (1/L) \ln Z_L/Z_0 = 0.693/L$. The reflection coefficient response is, from (5.70),

$$|\Gamma(\theta)| = \frac{1}{2} \ln\left(\frac{Z_L}{Z_0}\right) \frac{\sin\beta L}{\beta L}.$$

Klopfenstein taper: Using (5.77) gives Γ_0 as

$$\Gamma_0 = \frac{1}{2} \ln \left(\frac{Z_L}{Z_0} \right) = 0.346,$$

and (5.78) gives A as

$$A = \cosh^{-1}\left(\frac{\Gamma_0}{\Gamma_m}\right) = \cosh^{-1}\left(\frac{0.346}{0.002}\right) = 3.543.$$

The impedance taper must be numerically evaluated from (5.74). The reflection coefficient magnitude is given by (5.76):

$$|\Gamma(\theta)| = \Gamma_0 \frac{\cos\sqrt{(\beta L)^2 - A^2}}{\cosh A}$$

The passband for the Klopfenstein taper is defined as $\beta L > A = 3.543 = 1.13\pi$.

Figure 5.21a,b shows the impedance variations (versus z/L), and the resulting reflection coefficient magnitude (versus βL) for the three types of tapers.





Solution to Example 5.8. (a) Impedance variations for the triangular, exponential, and Klopfenstein tapers. (b) Resulting reflection coefficient magnitude versus frequency for the tapers of (a).

The Klopfenstein taper is seen to give the desired response of $|\Gamma| \leq \Gamma_m = 0.02$ for $\beta L \geq 1.13\pi$, which is lower than either the triangular or exponential taper responses. Also note that, like the stepped-Chebyshev matching transformer, the response of the Klopfenstein taper has equal-ripple lobes versus frequency in its passband.

5.9____ THE BODE-FANO CRITERION

In this chapter we discussed several techniques for matching an arbitrary load at a single frequency, using lumped elements, tuning stubs, and single-section quarter-wave transformers. We then presented multisection matching transformers and tapered lines as a means of obtaining broader bandwidths, with various passband characteristics. We will now close our study of impedance matching with a somewhat qualitative discussion of the theoretical limits that constrain the performance of an impedance matching network.

We limit our discussion to the circuit of Figure 5.1, where a lossless network is used to match an arbitrary complex load, generally over a nonzero bandwidth. From a very general perspective, we might raise the following questions in regard to this problem:

- Can we achieve a perfect match (zero reflection) over a specified bandwidth?
- If not, how well can we do? What is the trade-off between Γ_m , the maximum allowable reflection in the passband, and the bandwidth?
- How complex must the matching network be for a given specification?

These questions can be answered by the Bode-Fano criterion [7], [8] which gives, for certain canonical types of load impedances, a theoretical limit on the minimum reflection coefficient magnitude that can be obtained with an arbitrary matching network. The Bode-Fano criterion thus represents the optimum result that can be ideally achieved, even though such a result may only be approximated in practice. Such optimal results are always important, however, because they give us the upper limit of performance, and provide a benchmark against which a practical design can be compared.

Figure 5.22a shows a lossless network used to match a parallel RC load impedance. The Bode-Fano criterion states that

$$\int_0^\infty \ln \frac{1}{|\Gamma(\omega)|} d\omega \le \frac{\pi}{RC},$$
5.79

where $\Gamma(\omega)$ is the reflection coefficient seen looking into the arbitrary lossless matching network. The derivation of this result is beyond the scope of this text (the interested reader is referred to references [7] and [8]), but our goal here is to discuss the implications of the above result.

Assume that we desire to synthesize a matching network with a reflection coefficient response like that shown in Figure 5.23a. Applying (5.79) to this function gives

$$\int_0^\infty \ln \frac{1}{|\Gamma|} d\omega = \int_{\Delta\omega} \ln \frac{1}{\Gamma_m} d\omega = \Delta\omega \ln \frac{1}{\Gamma_m} \le \frac{\pi}{RC}, \qquad 5.80$$

which leads to the following conclusions:



FIGURE 5.22 The Bode-Fano limits for RC and RL loads matched with passive and lossless networks (ω_0 is the center frequency of the matching bandwidth). (a) Parallel RC. (b) Series RC. (c) Parallel RL. (d) Series RL.

- For a given load (fixed RC product), a broader bandwidth $(\Delta \omega)$ can be achieved only at the expense of a higher reflection coefficient in the passband (Γ_m) .
- The passband reflection coefficient Γ_m cannot be zero unless $\Delta \omega = 0$. Thus a perfect match can be achieved only at a finite number of frequencies, as illustrated in Figure 5.23b.
- As R and/or C increases, the quality of the match ($\Delta \omega$ and/or $1/\Gamma_m$) must decrease. Thus, higher-Q circuits are intrinsically harder to match than are lower-Q circuits.

Since $\ln 1/|\Gamma|$ is proportional to the return loss (in dB) at the input of the matching network, (5.79) can be interpreted as requiring that the area between the return loss curve and the $|\Gamma| = 1$ (RL = 0 dB) axis must be less than or equal to a constant. Optimization

297



FIGURE 5.23 Illustrating the Bode-Fano criterion. (a) A possible reflection coefficient response. (b) Nonrealizable and realizable reflection coefficient responses.

then implies that the return loss curve be adjusted so that $|\Gamma| = \Gamma_m$ over the passband and $|\Gamma| = 1$ elsewhere, as in Figure 5.23a. In this way, no area under the return loss curve is wasted outside the passband, or lost in regions within the passband for which $|\Gamma| < \Gamma_m$. The square-shaped response of Figure 5.23a is thus the optimum response, but cannot be realized in practice because it would require an infinite number of elements in the matching network. It can be approximated, however, with a reasonably small number of elements, as described in reference [8]. Finally, note that the Chebyshev matching transformer can be considered as a close approximation to the ideal passband of Figure 5.23a, when the ripple of the Chebyshev response is made equal to Γ_m . Figure 5.22 lists the Bode-Fano limits for other types of RC and RL loads.

REFERENCES

- [1] R. E. Collin, Foundations for Microwave Engineering, Second Edition, McGraw-Hill, N.Y., 1992.
- [2] G. L. Matthaei, L. Young, and E. M. T. Jones. Microwave Filters, Impedance-Matching Networks, and Coupling Structures, Artech House Books, Dedham, Mass. 1980.
- [3] P. Bhartia and I. J. Bahl, Millimeter Wave Engineering and Applications, Wiley Interscience, N.Y., 1984.
- ^[4] R. E. Collin, "The Optimum Tapered Transmission Line Matching Section." *Proc. IRE*, vol. 44, pp. 539–548, April 1956.
- [5] R. W. Klopfenstein, "A Transmission Line Taper of Improved Design," Proc. IRE, vol. 44, pp. 31-15, January 1956.
- [6] M. A. Grossberg, "Extremely Rapid Computation of the Klopfenstein Impedance Taper," Proc. IEEE, vol. 56, pp. 1629-1630, September 1968.

- [7] H. W. Bode, Network Analysis and Feedback Amplifier Design. Van Nostrand, N.Y., 1945,
- [8] R. M. Fano, "Theoretical Limitations on the Broad-Band Matching of Arbitrary Impedances," *Journal of the Franklin Institute*, vol. 249, pp. 57–83, January 1950, and pp. 139–154, February 1950.

PROBLEMS

5.1 Design lossless L section matching networks for the following normalized load impedances:

(a)
$$z_L = 1.4 + j2.0$$
 (c) $z_L = 0.5 + j0.9$

- (b) $z_L = 0.2 + j0.3$ (d) $z_L = 1.6 j0.3$
- 5.2 We have seen that the matching of an arbitrary load impedance requires a network with at least two degrees of freedom. Determine the types of load impedances/admittances that can be matched with the two single-element networks shown below.



- 5.3 A load impedance $Z_L = 200 + j160 \Omega$ is to be matched to a 100 Ω line using a single shunt-stub tuner. Find two solutions using open-circuited stubs.
- 5.4 Repeat Problem 5.3 using short-circuited stubs.
- 5.5 A load impedance $Z_L = 20 j60 \ \Omega$ is to be matched to a 50 Ω line using a single series stub tuner. Find two solutions using open-circuited stubs.
- 5.6 Repeat Problem 5.5 using short-circuited stubs.
- 5.7 In the circuit shown below a $Z_L = 200 + j100 \ \Omega$ load is to be matched to a 40 Ω line, using a length, ℓ , of lossless transmission line of characteristic impedance, Z_1 . Find ℓ and Z_1 . Determine, in general, what type of load impedances can be matched using such a circuit.



- 5.8 An open-circuit tuning stub is to be made from a lossy transmission line with an attenuation constant α . What is the maximum value of normalized reactance that can be obtained with this stub? What is the maximum value of normalized reactance that can be obtained with a shorted stub of the same type of transmission line? Assume $\alpha \ell$ is small.
- 5.9 Design a double-stub tuner using open-circuited stubs with a $\lambda/8$ spacing to match a load admittance $Y_L = (1.4 + j_2)Y_0$.
- 5.10 Repeat Problem 5.9 using a double-stub tuner with short-circuited stubs and a $3\lambda/8$ spacing.
- 5.11 Derive the design equations for a double-stub tuner using two series stubs, spaced a distance d apart. Assume the load impedance is $Z_L = R_L + jX_L$.

- 5.12 Consider matching a load $Z_L = 200 \ \Omega$ to a 100 Ω line, using single shunt-stub, single series stub, and double shunt-stub tuners, with short-circuited stubs. Which tuner will give the best bandwidth? Justify your answer by calculating the reflection coefficient for all *six* solutions at 1.1 f_0 , where f_0 is the match frequency, or use CAD to plot the reflection coefficient versus frequency.
- 5.13 Design a single-section quarter-wave matching transformer to match a 350 Ω load to a 100 Ω line. What is the percent bandwidth of this transformer, for SWR ≤ 2 ? If the design frequency is 4 GHz, sketch the layout of a microstrip circuit, including dimensions, to implement this matching transformer. Assume the substrate is 0.159 cm thick, with a dielectric constant of 2.2.
- 5.14 Consider the quarter-wave transformer of Figure 5.13, with $Z_1 = 100 \ \Omega$. $Z_2 = 150 \ \Omega$, and $Z_L = 225 \ \Omega$. Evaluate the worst-case percent error in computing $|\Gamma|$ from the approximate expression of (5.42), compared to the exact result.
- 5.15 A waveguide load with an equivalent TE_{10} wave impedance of 377 Ω must be matched to an air-filled X-band rectangular guide at 10 GHz. A quarter-wave matching transformer is to be used, and is to consist of a section of guide filled with dielectric. Find the required dielectric constant and physical length of the matching section. What restrictions on the load impedance apply to this technique?
- 5.16 Design a four-section binomial matching transformer to match a 10 Ω load to a 50 Ω line. What is the bandwidth of this transformer, for $\Gamma_{m} = 0.05$? Use CAD to plot the input reflection coefficient versus frequency.
- 5.17 Derive the exact characteristic impedance for a two-section binomial matching transformer, for a normalized load impedance $Z_L/Z_0 = 1.5$. Check your results with Table 5.1.
- 5.18 Calculate and plot the percent bandwidth for a N = 1, 2, and 4 section binomial matching transformer, versus $Z_L/Z_0 = 1.5$ to 6 for $\Gamma_m = 0.2$.
- 5.19 Using (5.56) and trigonometric identities, verify the results of (5.60).
- 5.20 Design a four-section Chebyshev matching transformer to match a 40 Ω line to a 60 Ω load. The maximum permissible SWR over the passband is 1.2. What is the resulting bandwidth? Use the approximate theory developed in the text, as opposed to the tables. Use CAD to plot the input reflection coefficient versus frequency.
- 5.21 Derive the exact characteristic impedances for a two-section Chebyshev matching transformer, for a normalized load impedance $Z_L/Z_0 = 1.5$. Check your results with Table 5.2 for $\Gamma_m = 0.05$.
- **5.22** A load of $Z_L/Z_0 = 1.5$ is to be matched to a feed line using a multisection transformer, and it is desired to have a passband response with $|\Gamma(\theta)| = A(0.1 + \cos^2 \theta)$, for $0 \le \theta \le \pi$. Use the approximate theory for multisection transformers to design a two-section transformer.
- 5.23 A tapered matching section has $d(\ln Z/Z_0)/dz = A \sin \pi z/L$. Find the constant A so that $Z(0) = Z_0$ and $Z(L) = Z_L$. Compute Γ , and plot $|\Gamma|$ versus βL .
- 5.24 Design an exponentially tapered matching transformer to match a 100 Ω load to a 50 Ω line. Plot $|\Gamma|$ versus βL , and find the length of the matching section (at the center frequency) required to obtain $|\Gamma| \leq 0.05$ over a 100% bandwidth. How many sections would be required if a Chebyshev matching transformer were used to achieve the same specifications?
- 5.25 A parallel RC load with $R = 100 \Omega$ and C = 1.5 pF is to be matched to a 50 Ω line over a frequency band from 2.0 to 10.0 GHz. What is the best return loss over this band that can be obtained with an optimum matching network?
- 5.26 Consider a series RL load with $R = 80 \Omega$ and L = 5 nH. Design a lumped-element L section matching network to match this load to a 50 Ω line at 2 GHz. Plot $|\Gamma|$ versus frequency for this network to determine the bandwidth for which $|\Gamma| \leq \Gamma_m = 0.1$. Compare this with the maximum possible bandwidth for this load, as given by the Bode-Fano criterion. (Assume a square reflection coefficient response like that of Figure 5.23a.)

Microwave Resonators

Microwave resonators are used in a variety of applications, including filters, oscillators, frequency meters, and tuned amplifiers. Since the operation of microwave resonators is very similar to that of the lumped-element resonators of circuit theory, we will begin by reviewing the basic characteristics of series and parallel *RLC* resonant circuits. We will then discuss various implementations of resonators at microwave frequencies using distributed elements such as transmission lines, rectangular and circular waveguide, and dielectric cavities. We will also discuss open resonators of the Fabry-Perot type, and the excitation of resonators using apertures and current sheets.

6.1 SERIES AND PARALLEL RESONANT CIRCUITS

Near resonance, a microwave resonator can usually be modeled by either a series or parallel RLC lumped-element equivalent circuit, and so we will derive some of the basic properties of such circuits below.

Series Resonant Circuit

A series RLC lumped-element resonant circuit is shown in Figure 6.1a. The input impedance is

$$Z_{\rm in} = R + j\omega L - j\frac{1}{\omega C}, \qquad 6.1$$

and the complex power delivered to the resonator is

$$P_{\rm in} = \frac{1}{2} V I^* = \frac{1}{2} Z_{\rm in} |I|^2 = \frac{1}{2} Z_{\rm in} \left| \frac{V}{Z_{\rm in}} \right|^2$$
$$= \frac{1}{2} |I|^2 \left(R + j\omega L - j\frac{1}{\omega C} \right).$$
6.2

The power dissipated by the resistor, R, is

$$P_{\rm loss} = \frac{1}{2} |I|^2 R, \qquad 6.3a$$

300



FIGURE 6.1 A series *RLC* resonator and its response. (a) The series *RLC* circuit. (b) The input impedance magnitude versus frequency.

the average magnetic energy stored in the inductor, L, is

$$W_m = \frac{1}{4} |I|^2 L, \qquad 6.3b$$

and the average electric energy stored in the capacitor, C_{*} is

$$W_e = \frac{1}{4} |V_e|^2 C = \frac{1}{4} |I|^2 \frac{1}{\omega^2 C},$$
 6.3c

where V_c is the voltage across the capacitor. Then the complex power of (6.2) can be rewritten as

$$P_{\rm in} = P_{\rm loss} + 2j\omega(W_m - W_e), \qquad 6.4$$

and the input impedance of (6.1) can be rewritten as

$$Z_{\rm in} = \frac{2P_{\rm in}}{|I|^2} = \frac{P_{\rm loss} + 2j\omega(W_m - W_e)}{|I|^2/2}.$$
 6.5

Resonance occurs when the average stored magnetic and electric energies are equal. or $W_m = W_e$. Then from (6.5) and (6.3a), the input impedance at resonance is

$$Z_{\rm in} = \frac{P_{\rm loss}}{|I|^2/2} = R,$$

which is a purely real impedance. From (6.3b,c), $W_m = W_e$ implies that the resonant frequency, ω_0 , must be defined as

$$\omega_0 = \frac{1}{\sqrt{L\bar{C}}}.$$
 6.6

Another important parameter of a resonant circuit is its Q, or quality factor, which is defined as

$$Q = \omega \frac{(\text{average energy stored})}{(\text{energy loss/second})}$$
$$= \omega \frac{W_m + W_e}{P_\ell}.$$
 6.7

Thus Q is a measure of the loss of a resonant circuit—lower loss implies a higher Q. For the series resonant circuit of Figure 6.1a, the Q can be evaluated from (6.7) using (6.3), and the fact that $W_m = W_e$ at resonance, to give

$$Q = \omega_0 \frac{2W_m}{P_{\text{loss}}} = \frac{\omega_0 L}{R} = \frac{1}{\omega_0 RC},$$
6.8

which shows that Q increases as R decreases.

Now consider the behavior of the input impedance of this resonator near its resonant frequency [1]. We let $\omega = \omega_0 + \Delta \omega$, where $\Delta \omega$ is small. The input impedance can then be rewritten from (6.1) as

$$egin{split} Z_{
m in} &= R + j\omega L \left(1 - rac{1}{\omega^2 LC}
ight) \ &= R + j\omega L \left(rac{\omega^2 - \omega_0^2}{\omega^2}
ight), \end{split}$$

since $\omega_0^2 = 1/LC$. Now $\omega^2 - \omega_0^2 = (\omega - \omega_0)(\omega + \omega_0) = \Delta\omega(2\omega - \Delta\omega) \simeq 2\omega\Delta\omega$ for small $\Delta\omega$. Thus,

$$Z_{\rm in} \simeq R + j2L\Delta\omega$$

$$\simeq R + j\frac{2RQ\Delta\omega}{\omega_0}.$$
 6.9

This form will be useful for identifying equivalent circuits with distributed element resonators.

Alternatively, a resonator with loss can be treated as a lossless resonator whose resonant frequency ω_0 has been replaced with a complex effective resonant frequency:

$$\omega_0 \longleftarrow \omega_0 \left(1 + \frac{j}{2Q} \right). \tag{6.10}$$

This can be seen by considering the input impedance of a series resonator with no loss, as given by (6.9) with R = 0:

$$\mathbf{Z}_{\rm in}=j2L(\omega-\omega_0).$$

Then substituting the complex frequency of (6.10) for ω_0 to give

$$egin{split} Z_{ ext{in}} &= j2L\left(\omega-\omega_0-jrac{\omega_0}{2Q}
ight) \ &= rac{\omega_0L}{Q} + j2L(\omega-\omega_0) = R + j2L\Delta\omega, \end{split}$$

which is identical to (6.9). This is a useful procedure because for most practical resonators the loss is very small, so the Q can be found using the perturbation method, beginning with the solution for the lossless case. Then the effect of loss can be added to the input impedance by replacing ω_0 with the complex resonant frequency given in (6.10).

Finally, consider the half-power fractional bandwidth of the resonator. Figure 6.1b shows the variation of the magnitude of the input impedance versus frequency. When the frequency is such that $|Z_{in}|^2 = 2R^2$, then by (6.2) the average (real) power delivered to the circuit is one-half that delivered at resonance. If BW is the fractional bandwidth, then $\Delta\omega/\omega_0 = BW/2$ at the upper band edge. Then using (6.9) gives

$$|R + jRQ(BW)|^2 = 2R^2,$$

BW = $\frac{1}{Q}$. 6.11

OT

Parallel Resonant Circuit

The parallel RLC resonant circuit, shown in Figure 6.2a, is the dual of the series RLC circuit. The input impedance is

$$Z_{\rm in} = \left(\frac{1}{R} + \frac{1}{j\omega L} + j\omega C\right)^{-1}, \qquad 6.12$$

and the complex power delivered to the resonator is

$$P_{\rm in} = \frac{1}{2} V I^* = \frac{1}{2} Z_{\rm in} |I|^2 = \frac{1}{2} |V|^2 \frac{1}{Z_{\rm in}^*}$$
$$= \frac{1}{2} |V|^2 \left(\frac{1}{R} + \frac{j}{\omega L} - j\omega C \right).$$
 6.13

The power dissipated by the resistor, R, is

$$P_{\text{loss}} = \frac{1}{2} \frac{|V|^2}{R}.$$
 6.14a

the average electric energy stored in the capacitor, C, is

$$W_{\rm e} = \frac{1}{4} |V|^2 C,$$
 6.14b

and the average magnetic energy stored in the inductor, L, is

$$W_m = \frac{1}{4} |I_L|^2 L = \frac{1}{4} |V|^2 \frac{1}{\omega^2 L},$$
 6.14c



FIGURE 6.2 A parallel *RLC* resonator and its response. (a) The parallel *RLC* circuit. (b) The input impedance magnitude versus frequency.

where I_L is the current through the inductor. Then the complex power of (6.13) can be rewritten as

$$P_{\rm in} = P_{\rm loss} + 2j\omega(W_m - W_e), \qquad 6.15$$

which is identical to (6.4). Similarly, the input impedance can be expressed as

$$Z_{\rm in} = \frac{2P_{\rm in}}{|I|^2} = \frac{P_{\rm Ioss} + 2j\omega(W_m - W_e)}{|I|^2/2}, \qquad 6.16$$

which is identical to (6.5),

As in the series case, resonance occurs when $W_m = W_e$. Then from (6.16) and (6.14a) the input impedance at resonance is

$$Z_{\rm in}=\frac{P_{\rm loss}}{|I|^2/2}=R,$$

which is a purely real impedance. From (6.14b, c), $W_m = W_e$ implies that the resonant frequency, ω_0 , should be defined as

$$\omega_0 \approx \frac{1}{\sqrt{LC}}, \qquad \qquad 6.17$$

which again is identical to the series resonant circuit case.

From the definition of (6.7), and the results in (6.14), the Q of the parallel resonant circuit can be expressed as

$$Q = \omega_0 \frac{2W_{\pi^*}}{P_{\text{loss}}} = \frac{R}{\omega_0 L} = \omega_0 RC, \qquad 6.18$$

since $W_m = W_e$ at resonance. This result shows that the Q of the parallel resonant circuit increases as R increases.

Near resonance, the input impedance of (6.12) can be simplified using the result that

$$\frac{1}{1+x} \simeq 1-x+\cdots.$$

Letting $\omega = \omega_0 + \Delta \omega$, where $\Delta \omega$ is small. (6.12) can be rewritten as [1]

$$Z_{\rm in} \simeq \left(\frac{1}{R} + \frac{1 - \Delta\omega/\omega_0}{j\omega_0 L} + j\omega_0 C + j\Delta\omega C\right)^{-1}$$
$$\simeq \left(\frac{1}{R} + j\frac{\Delta\omega}{\omega_0^2 L} + j\Delta\omega C\right)^{-1}$$
$$\simeq \left(\frac{1}{R} + 2j\Delta\omega C\right)^{-1}$$
$$\simeq \frac{R}{1 + 2j\Delta\omega RC} = \frac{R}{1 + 2j\Delta\omega/\omega_0}.$$
 6.19

since $\omega_0^2 = 1/LC$. When $R = \infty$ (6.19) reduces to

$$Z_{
m in}=rac{1}{j2C(\omega-\omega_0)}.$$

As in the series resonator case, the effect of loss can be accounted for by replacing ω_0 in this expression with a complex effective resonant frequency:

$$\omega_0 \longleftarrow \omega_0 \left(1 + \frac{j}{2Q} \right)$$
 6.20

Figure 6.2b shows the behavior of the magnitude of the input impedance versus frequency. The half-power bandwidth edges occur at frequencies ($\Delta\omega/\omega_0 = BW/2$), such that

$$|Z_{\rm in}|^2 = \frac{R^2}{2},$$

which, from (6.19), implies that

$$\mathbf{B}\mathbf{W} = \frac{\mathbf{I}}{Q},\tag{6.21}$$

as in the series resonance case.



FIGURE 6.3 A resonant circuit connected to an external load, R_L .

Loaded and Unioaded Q

The Q defined in the preceding sections is a characteristic of the resonant circuit itself, in the absence of any loading effects caused by external circuitry, and so is called the unloaded Q. In practice, however, a resonant circuit is invariably coupled to other circuitry, which will always have the effect of lowering the overall, or loaded Q. Q_L , of the circuit. Figure 6.3 depicts a resonator coupled to an external load resistor, R_L . If the resonator is a series RLC circuit, the load resistor R_L adds in series with R so that the effective resistance in (6.8) is $R \div R_L$. If the resonator is a parallel RLC circuit, the load resistor R_L combines in parallel with R so that the effective resistance in (6.18) is $RR_L/(R + R_L)$. If we define an external Q, Q_{ϵ} , as

$$Q_e = \begin{cases} \frac{\omega_0 L}{R_L} & \text{for series circuits} \\ \frac{R_L}{\omega_0 L} & \text{for parallel circuits,} \end{cases}$$
6.22

then the loaded Q can be expressed as

$$\frac{1}{Q_L} = \frac{1}{Q_e} + \frac{1}{Q}.$$
6.23

Table 6.1 summarizes the above results for series and parallel resonant circuits.

6.2 TRANSMISSION LINE RESONATORS

As we have seen, ideal lumped elements are usually unattainable at microwave frequencies, so distributed elements are more commonly used. In this section we will study the use of transmission line sections with various lengths and terminations (usually open or short circuited) to form resonators. Since we will be interested in the Q of these resonators, we must consider lossy transmission lines.

Short-Circuited $\lambda/2$ Line

Consider a length of lossy transmission line, short circuited at one end, as shown in Figure 6.4. The line has a characteristic impedance Z_0 , propagation constant β , and attenuation constant α . At the frequency $\omega = \omega_0$, the length of the line is $\ell = \lambda/2$, where $\lambda = 2\pi/\beta$. From (2.91), the input impedance is,

$$Z_{
m in} = Z_0 \tanh(lpha + jeta)\ell.$$

Quantity	Series Resonator	Parallel Resonator
Input	$Z_{\rm in} = R + j\omega L - j\frac{1}{\omega C}$	$Y_{\rm m} = \frac{1}{R} + j\omega C - j\frac{1}{\omega L}$
Impedance/adminance	$\simeq R + j \frac{2RQ\Delta\omega}{\omega_0}$	$\simeq rac{1}{R} + \jmath rac{2Q\Delta\omega}{R\omega_0}$
Power loss	$P_{ m hoss}=rac{1}{2} I ^2R$	$P_{\rm loss} = \frac{1}{2} \frac{ V ^2}{R}$
Stored magnetic energy	$W_m = \frac{1}{4} I ^2 L$	$W_m=rac{1}{4} V ^2rac{1}{\omega^2 L}$
Stored electric energy	$W_e = \frac{1}{4} I ^2 \frac{1}{\omega^2 C}$	$W_{arepsilon}=rac{1}{4} V ^{2}C$
Resonant frequency	$\omega_0 = rac{1}{\sqrt{LC}}$	$\omega_0 = rac{1}{\sqrt{LC}}$
Unloaded Q	$Q = \frac{\omega_0 L}{R} = \frac{1}{\omega_0 RC}$	$Q = \omega_0 RC = rac{R}{\omega_0 L}$
External Q	$Q_{e}=rac{\omega_{0}L}{R_{L}}$	$Q_{ m e}=rac{R_L}{\omega_0 L}$

TABLE 6.1 Summary of Results for Series and Parallel Resonators

Using an identity for the hyperbolic tangent gives

$$Z_{\rm in} = Z_0 \frac{\tanh \alpha \ell + j \tan \beta \ell}{1 + j \tan \beta \ell \tanh \alpha \ell}.$$
 6.24

Observe that $Z_{in} = jZ_0 \tan \beta \ell$ if $\alpha = 0$ (no loss).





In practice, most transmission lines have small loss, so we can assume that $\alpha \ell \ll 1$, and so $\tanh \alpha \ell \simeq \alpha \ell$. Now let $\omega = \omega_0 + \Delta \omega$, where $\Delta \omega$ is small. Then, assuming a TEM line,

$$eta \ell = rac{\omega \ell}{v_p} = rac{\omega_0 \ell}{v_p} + rac{\Delta \omega \ell}{v_p},$$

where v_p is the phase velocity of the transmission line. Since $\ell = \lambda/2 = \pi v_p/\omega_0$ for $\omega = \omega_0$, we have

$$\beta \ell = \pi + \frac{\Delta \omega \pi}{\omega_0},$$

and then

$$\tan \beta \ell = \tan \left(\pi + \frac{\Delta \omega \pi}{\omega_0} \right) = \tan \frac{\Delta \omega \pi}{\omega_0} \simeq \frac{\Delta \omega \pi}{\omega_0}$$

Using these results in (6.24) gives

$$Z_{ip} \simeq Z_0 \frac{\alpha \ell + j(\Delta \omega \pi / \omega_0)}{1 + j(\Delta \omega \pi / \omega_0) \alpha \ell} \simeq Z_0 \left(\alpha \ell + j \frac{\Delta \omega \pi}{\omega_0} \right),$$
 6.25

since $\Delta \omega \alpha \ell / \omega_0 \ll 1$.

Equation (6.25) is of the form

$$Z_{\rm in} = R + 2jL\Delta\omega,$$

which is the input impedance of a series RLC resonant circuit, as given by (6.9). We can then identify the resistance of the equivalent circuit as

$$\mathcal{R} = Z_0 \alpha \ell, \qquad \qquad \mathbf{6.26a}$$

and the inductance of the equivalent circuit as

$$L = \frac{Z_0 \pi}{2\omega_0}.$$
 6.26b

The capacitance of the equivalent circuit can be found from (6.6) as

$$C = \frac{1}{\omega_0^2 L} \tag{6.26c}$$

The resonator of Figure 6.4 thus resonates for $\Delta \omega \approx 0$ ($\ell \approx \lambda/2$), and its input impedance at this frequency is $Z_{in} = R = Z_0 \alpha \ell$. Resonance also occurs for $\ell = n\lambda/2$, n = 1, 2, 3, ... The voltage distributions for the n = 1 and n = 2 resonant modes are shown in Figure 6.4.

The Q of this resonator can be found from (6.8) and (6.26) as

$$Q = \frac{\omega_0 L}{R} = \frac{\pi}{2\alpha\ell} = \frac{\beta}{2\alpha}, \qquad 6.27$$
since $\beta \ell = \pi$ at the first resonance. This result shows that the Q decreases as the attenuation of the line increases, as expected.

Ø

EXAMPLE 6.1 Q of Half-Wave Coaxial Line Resonators

A $\lambda/2$ resonator is to be made from a piece of copper coaxial line, with an inner conductor radius of 1 mm and an outer conductor radius of 4 mm. If the resonant frequency is 5 GHz, compare the Q of an air-filled coaxial line resonator to that of a Teflon-filled coaxial line resonator.

Solution

We must first compute the attenuation of the coaxial line, which can be done using the results of Example 2.6 or 2.7. From Appendix F, the conductivity of copper is $\sigma = 5.813 \times 10^7$ S/m. Then the surface resistivity is

$$R_{\rm s} = \sqrt{\frac{\omega\mu_0}{2\sigma}} = 1.84 \times 10^{-2} \ \Omega.$$

and the attenuation due to conductor loss for the air-filled line is

$$\alpha_{c} = \frac{R_{s}}{2\eta \ln b/a} \left(\frac{1}{a} + \frac{1}{b}\right)$$
$$= \frac{1.84 \times 10^{-2}}{2(377)\ln(0.004/0.001)} \left(\frac{1}{0.001} + \frac{1}{0.004}\right) = 0.022 \text{ Np/m}$$

For Teflon, $\epsilon_r = 2.08$ and $\tan \delta = 0.0004$, so the attenuation due to conductor loss for the Teflon-filled line is

$$\alpha_{\rm c} = \frac{1.84 \times 10^{-2} \sqrt{2.08}}{2(377) \ln(0.004/0.001)} \left(\frac{1}{0.001} \div \frac{1}{0.004} \right) = 0.032 \text{ Np/m}.$$

The dielectric loss of the air-filled line is zero, but the dielectric loss of the Teflon-filled line is

$$\alpha_d = k_0 \frac{\sqrt{\epsilon_r}}{2} \tan \delta$$
$$= \frac{(104.7)\sqrt{2.08}(0.0004)}{2} = 0.030 \text{ Np/m}.$$

Finally, from (6.27), the Qs can be computed as

$$Q_{\text{air}} = \frac{\beta}{2\alpha} = \frac{104.7}{2(0.022)} = 2380,$$
$$Q_{\text{Teflon}} = \frac{\beta}{2\alpha} = \frac{104.7\sqrt{2.08}}{2(0.032 + 0.030)} = 1218.$$

Thus it is seen that the Q of the air-filled line is almost twice that of the Teflon-filled line. The Q can be further increased by using silver-plated conductors. \bigcirc

Short-Circuited $\lambda/4$ Line

A parallel type of resonance (antiresonance) can be achieved using a short-circuited transmission line of length $\lambda/4$. The input impedance of the shorted line of length ℓ is

$$Z_{in} = Z_0 \tanh(\alpha + j\beta)\ell$$

= $Z_0 \frac{\tanh \alpha \ell + j \tan \beta \ell}{1 + j \tan \beta \ell \tanh \alpha \ell}$
= $Z_0 \frac{1 - j \tanh \alpha \ell \cot \beta \ell}{\tanh \alpha \ell - j \cot \beta \ell},$ 6.28

where the last result was obtained by multiplying both numerator and denominator by $-j \cot \beta \ell$. Now assume that $\ell = \lambda/4$ at $\omega = \omega_0$, and let $\omega = \omega_0 + \Delta \omega$. Then, for a TEM line,

$$\beta \ell = \frac{\omega_0 \ell}{v_p} + \frac{\Delta \omega \ell}{v_p} = \frac{\pi}{2} + \frac{\pi \Delta \omega}{2 \omega_0},$$

and so

$$\cot eta \ell = \cot \left(rac{\pi}{2} + rac{\pi \Delta \omega}{2\omega_0}
ight) = - an rac{\pi \Delta \omega}{2\omega_0} \simeq rac{-\pi \Delta \omega}{2\omega_0}.$$

Also, as before, $\tanh \alpha \ell \simeq \alpha \ell$ for small loss. Using these results in (6.28) gives

$$Z_{\rm in} = Z_0 \frac{1 + j\alpha \ell \pi \Delta \omega / 2\omega_0}{\alpha \ell + j\pi \Delta \omega / 2\omega_0} \simeq \frac{Z_0}{\alpha \ell + j\pi \Delta \omega / 2\omega_0}, \qquad 6.29$$

since $\alpha \ell \pi \Delta \omega / 2\omega_0 \ll 1$. This result is of the same form as the impedance of a parallel *RLC* circuit, as given in (6.19):

$$Z_{in} = \frac{1}{(1/R) + 2j\Delta\omega C}.$$

Then we can identify the resistance of the equivalent circuit as

$$R = \frac{Z_0}{\alpha \ell}$$
 6.30a

and the capacitance of the equivalent circuit as

$$C = \frac{\pi}{4\omega_0 Z_0}.$$
 6.30b

The inductance of the equivalent circuit can be found as

$$L = \frac{1}{\omega_0^2 C}.$$
 6.30c

The resonator of Figure 6.4 thus has a parallel type resonance for $\ell = \lambda/4$, with an input impedance at resonance of $Z_{in} = R = Z_0/\alpha \ell$. From (6.18) and (6.30) the Q of this resonator is

$$Q = \omega_0 RC = \frac{\pi}{4\alpha\ell} = \frac{\beta}{2\alpha},$$
 6.31

since $\ell = \pi/2\beta$ at resonance.

Open-Circuited $\lambda/2$ Line

A practical resonator that is often used in microstrip circuits consists of an opencircuited length of transmission line, as shown in Figure 6.5. Such a resonator will behave as a parallel resonant circuit when the length is $\lambda/2$, or multiples of $\lambda/2$.

The input impedance of an open-circuited line of length ℓ is

$$Z_{in} = Z_0 \coth(\alpha + j\beta)\ell = Z_0 \frac{1 + j \tan \beta \ell \tanh \alpha \ell}{\tanh \alpha \ell + j \tan \beta \ell}.$$
 6.32

 $\Delta\omega\pi$

As before, assume that $\ell = \lambda/2$ at $\omega = \omega_0$, and let $\omega = \omega_0 + \Delta \omega$. Then,

$$\beta \ell = \pi + \frac{\pi \Delta \omega}{\omega_0},$$

and so

$$\tan\beta\ell=\tan\frac{-\omega}{\omega}\simeq\frac{-\omega_0}{\omega_0}.$$

 $\Delta\omega\pi$



FIGURE 6.5 An open-circuited length of lossy transmission line, and the voltage distributions for n = 1 ($\ell = \lambda/2$) and n = 2 ($\ell = \lambda$) resonators.

and $\tanh \alpha \ell \simeq \alpha \ell$. Using these results in (6.32) gives

$$Z_{\rm in} = \frac{Z_0}{\alpha \ell + j(\Delta \omega \pi / \omega_0)}.$$
6.33

Comparing with the input impedance of a parallel resonant circuit as given by (6.19) suggests that the resistance of the equivalent *RLC* circuit is,

$$R = \frac{Z_0}{\alpha \ell}, \qquad \qquad 6.34a$$

and the capacitance of the equivalent circuit is

$$C = \frac{\pi}{2\omega_0 Z_0}.$$
 6.34b

The inductance of the equivalent circuits is

$$L = \frac{1}{\omega_0^2 C}.$$
 6.34c

From
$$(6.18)$$
 and (6.34) the Q is

$$Q = \omega_0 RC = \frac{\pi}{2\alpha\ell} = \frac{\beta}{2\alpha'},$$
6.35

since $\ell = \pi/\beta$ at resonance.



EXAMPLE 6.2 A Half-Wave Microstrip Resonator

Consider a microstrip resonator constructed from a $\lambda/2$ length of 50 Ω opencircuited microstrip line. The substrate thickness is 0.159 cm, with $\epsilon_4 = 2.2$ and tan $\delta = 0.001$. The conductors are copper. Compute the length of the line for resonance at 5 GHz, and the Q of the resonator. Ignore fringing fields at the end of the line.

Solution

From (3.197), the width of a 50 Ω microstrip line on this substrate is found to be

$$W = 0.49 \, \mathrm{cm}$$

and the effective permittivity is

$$\epsilon_e = 1.87.$$

Then the resonant length can be calculated as

$$\ell = \frac{\lambda}{2} = \frac{v_p}{2f} = \frac{c}{2f\sqrt{\epsilon_e}} = \frac{3 \times 10^8}{2(5 \times 10^9)\sqrt{1.87}} = 2.19 \text{ cm}.$$

The propagation constant is

$$\beta = \frac{2\pi f}{v_p} = \frac{2\pi f \sqrt{\epsilon_e}}{c} = \frac{2\pi (5 \times 10^9) \sqrt{1.87}}{3 \times 10^8} = 143.2 \text{ rad/m}.$$

From (3.199), the attenuation due to conductor loss is

$$\alpha_{\rm c} = \frac{R_s}{Z_0 W} = \frac{1.84 \times 10^{-2}}{50(0.0049)} = 0.075 \text{ Np/m},$$

where we used R_{a} from Example 6.1. From (3.198), the attenuation due to dielectric loss is

$$\alpha_d = \frac{k_0 \epsilon_r (\epsilon_e - 1) \tan b}{2\sqrt{\epsilon_e} (\epsilon_r - 1)} = \frac{(104.7)(2.2)(0.87)(0.001)}{2\sqrt{1.87}(1.2)} = 0.0611 \text{ Np/m}.$$

Then from (6.35) the Q is

$$Q = \frac{\beta}{2\alpha} = \frac{143.2}{2(0.075 + 0.0611)} = 526.$$

6.3 ____ RECTANGULAR WAVEGUIDE CAVITIES

Resonators can also be constructed from closed sections of waveguide, which should not be surprising since waveguides are a type of transmission line. Because of radiation loss from open-ended waveguide, waveguide resonators are usually short circuited at both ends, thus forming a closed box or cavity. Electric and magnetic energy is stored within the cavity, and power can be dissipated in the metallic walls of the cavity as well as in the dielectric filling the cavity. Coupling to the resonator can be by a small aperture or a small probe or loop.

We will first derive the resonant frequencies for a general TE or TM resonant mode, and then derive an expression for the Q of the TE₁₀ mode. A complete treatment of the Q for arbitrary TE and TM modes can be made using the same procedure, but is not included here because of its length and complexity.

Resonant Frequencies

The geometry of a rectangular cavity is shown in Figure 6.6. It consists of a length d of rectangular waveguide shorted at both ends (z = 0, d). We first find the resonant frequencies of this cavity under the assumption that the cavity is lossless, then we determine the Q using the perturbation method outlined in Section 2.7. We could begin with the wave equations and use the method of separation of variables to solve for the electric and magnetic fields that satisfy the boundary conditions of the cavity, but it is easier to start with the TE and TM waveguide fields, which already satisfy the necessary boundary conditions on the side walls (x = 0, a and y = 0, b) of the cavity. Then it is only necessary to enforce the boundary conditions that $E_x = E_y = 0$ on the end walls at z = 0, d.

From Table 3.2 the transverse electric fields (E_x, E_y) of the TE_{mn} or TM_{mn} rectangular waveguide mode can be written as

$$\bar{E}_{t}(x, y, z) = \bar{e}(x, y) [A^{+} e^{-j\beta_{mn}z} + A^{-} e^{j\beta_{mn}z}], \qquad 6.36$$



FIGURE 6.6 A rectangular resonant cavity, and the electric field distributions for the TE_{101} and TE_{102} resonant modes.

where $\bar{e}(x, y)$ is the transverse variation of the mode, and A^+ , A^- are arbitrary amplitudes of the forward and backward traveling waves. The propagation constant of the m, nth TE or TM mode is

$$\beta_{mn} = \sqrt{k^2 - \left(\frac{m\pi}{a}\right)^2 - \left(\frac{n\pi}{b}\right)^2}.$$
6.37

where $k = \omega \sqrt{\mu \epsilon}$, and μ, ϵ are the permeability of permittivity of the material filling the cavity.

Applying the condition that $\bar{E}_t = 0$ at z = 0 to (6.36) implies that $A^+ = -A^-$ (as we should expect for reflection from a perfectly conducting wall). Then the condition that $\bar{E}_t = 0$ at z = d leads to the equation

$$\bar{E}_t(x, y, d) = -\tilde{e}(x, y)A^+ 2j\sin\beta_{mn}d = 0.$$

The only nontrivial $(A^+ \neq 0)$ solution thus occurs for

$$\beta_{mn}d = \ell\pi, \qquad \ell = 1.2, 3, \cdots, \qquad 6.38$$

which implies that the cavity must be an integer multiple of a half-guide wavelength long at the resonant frequency. No nontrivial solutions are possible for other lengths, or for frequencies other than the resonant frequencies. The rectangular cavity is thus a waveguide version of the short-circuited $\lambda/2$ transmission line resonator. A cutoff wavenumber for the rectangular cavity can be defined as

$$k_{mn\ell} = \sqrt{\left(\frac{m\pi}{a}\right)^2 + \left(\frac{n\pi}{b}\right)^2 + \left(\frac{\ell\pi}{d}\right)^2}.$$
6.39

Then we can refer to the $TE_{mn\ell}$ or $TM_{mn\ell}$ resonant mode of the cavity, where the indices m, n, ℓ refer to the number of variations in the standing wave pattern in the x, y, z directions, respectively. The resonant frequency of the $TE_{mn\ell}$ or $TM_{mn\ell}$ mode is then given by

$$f_{mn\ell} = \frac{ck_{mn\ell}}{2\pi\sqrt{\mu_r\epsilon_r}} = \frac{c}{2\pi\sqrt{\mu_r\epsilon_r}}\sqrt{\left(\frac{m\pi}{a}\right)^2 + \left(\frac{n\pi}{b}\right)^2 + \left(\frac{\ell\pi}{d}\right)^2}.$$
 6.40

If b < a < d, the dominant resonant mode (lowest resonant frequency) will be the TE₁₀₁ mode, corresponding to the TE₁₀ dominant waveguide mode in a shorted guide of length $\lambda_g/2$. The dominant TM mode is the TM₁₁₀ mode.

Q of the $TE_{10\ell}$ Mode

From Table 3.2, (6.36), and the fact that $A^- = -A^+$, the total fields for the TE₁₀ ℓ resonant mode can be written as

$$E_y = A^+ \sin \frac{\pi x}{a} [e^{-j\beta z} - e^{j\beta z}], \qquad 6.41a$$

$$H_{x} = \frac{-A^{+}}{Z_{\text{TE}}} \sin \frac{\pi x}{a} [e^{-j\beta z} + e^{j\beta z}], \qquad 6.41b$$

$$H_z = \frac{j\pi A^+}{k\eta a} \cos\frac{\pi x}{a} [e^{-j\beta z} - e^{j\beta z}].$$
 6.41c

Letting $E_0 = -2jA^+$ and using (6.38) allows these expressions to be reduced to

$$E_y = E_0 \sin \frac{\pi x}{a} \sin \frac{\ell \pi z}{d}, \qquad \qquad 6.42a$$

$$H_x = \frac{-jE_0}{Z_{\text{TE}}} \sin \frac{\pi x}{a} \cos \frac{\ell \pi z}{d}, \qquad 6.42b$$

$$H_z = \frac{j\pi E_0}{k\eta a} \cos \frac{\pi x}{a} \sin \frac{\ell \pi z}{d}, \qquad 6.42c$$

which clearly show that the fields form standing waves inside the cavity. We can now compute the Q of this mode by finding the stored electric and magnetic energies, and the power lost in the conducting walls and the dielectric.

The stored electric energy is, from (1.84),

$$W_e = \frac{\epsilon}{4} \int_V E_y E_y^* dv = \frac{\epsilon abd}{16} E_0^2, \qquad 6.43a$$

while the stored magnetic energy is, from (1.86),

$$W_{m} = \frac{\mu}{4} \int_{V} (H_{x}H_{x}^{*} + H_{z}H_{z}^{*})dv$$
$$= \frac{\mu abd}{16} E_{0}^{2} \left(\frac{1}{Z_{TE}^{2}} + \frac{\pi^{2}}{k^{2}\eta^{2}a^{2}}\right).$$
 6.436

Since $Z_{\text{IE}} = k\eta/\beta$, and $\beta = \beta_{10} = \sqrt{k^2 - (\pi/a)^2}$, the quantity in parentheses in (6.43b) can be reduced to

$$\left(\frac{1}{Z_{\text{TE}}^2} + \frac{\pi^2}{k^2 \eta^2 a^2}\right) = \frac{\beta^2 + (\pi/a)^2}{k^2 \eta^2} = \frac{1}{\eta^2} = \frac{\epsilon}{\mu},$$

which shows that $W_e = W_m$. Thus, the stored electric and magnetic energies are equal at resonance, analogous to the *RLC* resonant circuits of Section 6.1.

For small losses we can find the power dissipated in the cavity walls using the perturbation method of Section 2.7. Thus, the power lost in the conducting walls is given by (1.131) as

$$P_c = \frac{R_s}{2} \int_{\text{walks}} |H_t|^2 ds, \qquad 6.44$$

where $R_s \approx \sqrt{\omega \mu_o / 2\sigma}$ is the surface resistivity of the metallic walls, and H_t is the tangential magnetic field at the surface of the walls. Using (6.42b,c) in (6.44) gives,

$$P_{c} = \frac{R_{s}}{2} \left\{ 2 \int_{y=0}^{b} \int_{x=0}^{a} |H_{x}(z=0)|^{2} dx \, dy + 2 \int_{z=0}^{d} \int_{y=0}^{b} |H_{z}(x=0)|^{2} dy \, dz + 2 \int_{z=0}^{d} \int_{x=0}^{a} \left[|H_{x}(y=0)|^{2} + |H_{z}(y=0)|^{2} \right] dx \, dz \right\}$$
$$= \frac{R_{s} E_{0}^{2} \lambda^{2}}{8 \eta^{2}} \left(\frac{\ell^{2} ab}{d^{2}} + \frac{bd}{a^{2}} + \frac{\ell^{2} a}{2d} + \frac{d}{2a} \right).$$
 6.45

where use has been made of the symmetry of the cavity in doubling the contributions from the walls at x = 0, y = 0, and z = 0 to account for the contributions from the walls at x = a, y = b, and z = d, respectively. The relations that $k = 2\pi/\lambda$ and $Z_{\text{TE}} = k\eta/\beta = 2d\eta/\ell\lambda$ were also used in simplifying (6.45). Then, from (6.7), the Q of the cavity with lossy conducting walls but lossless dielectric can be found as

$$Q_{c} = \frac{2\omega_{0}W_{e}}{P_{c}}$$

$$= \frac{k^{3}ab \, d\eta}{4\pi^{2}R_{s}} \frac{1}{[(\ell^{2}ab/d^{2}) + (bd/a^{2}) + (\ell^{2}a/2d) + (d/2a)]}$$

$$= \frac{(kad)^{3}b\eta}{2\pi^{2}R_{s}} \frac{1}{(2\ell^{2}a^{3}b + 2bd^{3} + \ell^{2}a^{3}d + ad^{3})}.$$
6.46

We now compute the power lost in the dielectric. As discussed in Chapter 1, a lossy dielectric has an effective conductivity $\sigma = \omega \epsilon'' = \omega \epsilon_r \epsilon_0 \tan \delta$, where $\epsilon = \epsilon' - j\epsilon'' =$

 $\epsilon_r \epsilon_0 (1 - j \tan \delta)$, and $\tan \delta$ is the loss tangent of the material. Then the power dissipated in the dielectric is, from (1.92),

$$P_d = \frac{1}{2} \int_V \bar{J} \cdot \bar{E}^* dv = \frac{\omega \epsilon''}{2} \int_V |\bar{E}|^2 dv = \frac{abd\omega \epsilon'' |E_0|^2}{8}, \qquad 6.47$$

where \overline{E} is given by (6.42a). Then from (6.7) the Q of the cavity with a lossy dielectric filling, but with perfectly conducting walls, is

$$Q_d = \frac{2\omega W_e}{P_d} = \frac{\epsilon'}{\epsilon''} = \frac{1}{\tan \delta}.$$
6.48

The simplicity of this result is due to the fact that the integral in (6.43a) for W_e cancels with the identical integral in (6.47) for P_d . This result thus applies to Q_d for an arbitrary resonant cavity mode. When both wall losses and dielectric losses are present, the total power loss is $P_c + P_d$, so (6.7) gives the total Q as

$$Q = \left(\frac{1}{Q_c} + \frac{1}{Q_d}\right)^{-1}.$$
6.49



EXAMPLE 6.3 Design of a Rectangular Waveguide Cavity

A rectangular waveguide cavity is made from a piece of copper WR-187 Hband waveguide, with a = 4.755 cm and b = 2.215 cm. The cavity is filled with polyethylene ($\epsilon_r = 2.25$, $\tan \delta = 0.0004$). If resonance is to occur at f = 5 GHz, find the required length, d, and the resulting Q for the $\ell = 1$ and $\ell = 2$ resonant modes.

Solution The wavenumber k is

$$k = \frac{2\pi f \sqrt{\epsilon_r}}{c} = 157.08 \text{ m}^{-1}$$

From (6.40) the required length for resonance can be found as (m = 1, n = 0)

$$d = \frac{\ell \pi}{\sqrt{k^2 - (\pi/a)^2}},$$

for $\ell = 1$, $d = \frac{\pi}{\sqrt{(157.08)^2 - (\pi/0.02215)^2}} = 4.65$ cm,
for $\ell = 2$, $d = 2(4.65) = 9.30$ cm.

From Example 6.1, the surface resistivity of copper at 5 GHz is $R_s = 1.84 \times 10^{-2} \Omega$. The intrinsic impedance is

$$\eta = \frac{377}{\sqrt{\epsilon_r}} = 251.3 \ \Omega.$$

Then from (6.46) the Q due to conductor loss only is

for $\ell = 1$, $Q_c = 3380$. for $\ell = 2$, $Q_c = 3864$.

From (6.48) the Q due to dielectric loss only is, for both $\ell = 1$ and $\ell = 2$,

$$Q_d = \frac{1}{\tan \delta} = \frac{1}{0.0004} = 2500.$$

So the total Qs are, from (6.49)

for
$$\ell = 1$$
, $Q = \left(\frac{1}{3380} + \frac{1}{2500}\right)^{-1} = 1437$,
for $\ell = 2$, $Q = \left(\frac{1}{3864} + \frac{1}{2500}\right)^{-1} = 1518$.

Note that the dielectric loss has the dominant effect on the Q; higher Q could thus be obtained using an air-filled cavity. These results can be compared to those of Examples 6.1 and 6.2, which used similar types of materials at the same frequency.

6.4 CIRCULAR WAVEGUIDE CAVITIES

A cylindrical cavity resonator can be constructed from a section of circular waveguide shorted at both ends, similar to rectangular cavities. Since the dominant circular waveguide mode is the TE₁₁ mode, the dominant cylindrical cavity mode is the TE₁₁₁ mode. We will derive the resonant frequencies for the TE_{nml} and TM_{nml} circular cavity modes, and the expression for the Q of the TE_{nml} mode.

Circular cavities are often used for microwave frequency meters. The cavity is constructed with a movable top wall to allow mechanical tuning of the resonant frequency, and the cavity is loosely coupled to a waveguide with a small aperture. In operation, power will be absorbed by the cavity as it is tuned to the operating frequency of the system; this absorption can be monitored with a power meter elsewhere in the system. The tuning dial is usually directly calibrated in frequency, as in the model shown in Figure 6.7. Since frequency resolution is determined by the Q of the resonator, the TE₀₁₁ mode is often used for frequency meters because its Q is much higher than the Q of the dominant circular cavity mode. This is also the reason for a loose coupling to the cavity.

Resonant Frequencies

The geometry of a cylindrical cavity is shown in Figure 6.8. As in the case of the rectangular cavity, the solution is simplified by beginning with the circular waveguide modes, which already satisfy the necessary boundary conditions on the circular waveguide wall. From Table 3.5, the transverse electric fields (E_{ρ}, E_{ϕ}) of the TE_{nm} or TM_{nm} circular waveguide mode can be written as

$$\bar{E}_t(\rho,\phi,z) = \bar{e}(\rho,\phi) \left[A^+ e^{-j\beta_{nm}z} + A^- e^{j\beta_{nm}z} \right],$$
6.50

. .



FIGURE 6.7 Photograph of a W-band waveguide frequency meter. The knob rotates to change the length of the circular cavity resonator; the scale gives a readout of the frequency.

Photograph courtesy of Millitech Corporation, S. Deerfield, Mass.

where $\bar{e}(\rho, \phi)$ represents the transverse variation of the mode, and A^+ and A^- are arbitrary amplitudes of the forward and backward traveling waves. The propagation constant of the TE_{nm} mode is, from (3.126),

$$\beta_{nm} = \sqrt{k^2 - \left(\frac{\rho'_{nm}}{a}\right)^2}, \qquad 6.51a$$

while the propagation constant of the TM_{nm} mode is, from (3.139),

$$\beta_{nm} = \sqrt{k^2 - \left(\frac{p_{nm}}{a}\right)^2}, \qquad 6.51b$$

where $k = \omega \sqrt{\mu \epsilon}$.

Now in order to have $\tilde{E}_t = 0$ at z = 0, d, we must have $A^+ = -A^-$, and

$$A^+ \sin \beta_{nm} d = 0,$$

$$\beta_{nm} d = \ell \pi, \qquad \text{for } \ell = 0, 1, 2, 3, \dots, \qquad 6.52$$







A cylindrical resonant cavity, and the electric field distribution for resonant modes with $\ell = 1$ or $\ell = 2$.

which implies that the waveguide must be an integer number of half-guide wavelengths long. Thus, the resonant frequency of the $TE_{nm\ell}$ mode is

$$f_{nm\ell} = \frac{c}{2\pi\sqrt{\mu_r\epsilon_r}}\sqrt{\left(\frac{p'_{nm}}{a}\right)^2 + \left(\frac{\ell\pi}{d}\right)^2},\qquad 6.53a$$

and the resonant frequency of the $TM_{nm\ell}$ mode is

$$f_{nm\ell} = \frac{c}{2\pi\sqrt{\mu_r\epsilon_r}}\sqrt{\left(\frac{p_{nm}}{a}\right)^2 + \left(\frac{\ell\pi}{d}\right)^2}.$$
6.53b

Then the dominant TE mode is the TE_{111} mode, while the dominant TM mode is the TM_{110} mode. Figure 6.9 shows a *mode chart* for the lower order resonant modes of a cylindrical cavity. Such a chart is very useful for the design of cavity resonators, as it shows what modes can be excited at a given frequency, for a given cavity size.

Q of the TE_{nml} Mode

From Table 3.5. (6.50), and the fact that $A^+ = -A^-$, the fields of the TE_{nml} mode can be written as

$$H_z = H_0 J_n \left(\frac{p'_{nm}\rho}{a}\right) \cos n\phi \sin \frac{\ell \pi z}{d}, \qquad 6.54a$$



FIGURE 6.9

Adapted from data from R. E. Collin, Foundations for Microwave Engineering (New York: McGraw-Hill, 1966). Used with permission.

$$H_{\rho} = \frac{\beta a H_0}{p'_{nm}} J'_n \left(\frac{p'_{nm}\rho}{a}\right) \cos n\phi \cos \frac{\ell \pi z}{d}, \qquad 6.54b$$

$$H_{\phi} = \frac{-\beta a^2 n H_0}{(p'_{nm})^2 \rho} J_n\left(\frac{p'_{nm}\rho}{a}\right) \sin n\phi \cos \frac{\ell \pi z}{d}, \qquad 6.54c$$

$$E_{\rho} = \frac{jk\eta a^2 nH_0}{(p'_{nm})^2 \rho} J_n\left(\frac{p'_{nm}\rho}{a}\right) \sin n\phi \sin \frac{\ell \pi z}{d}, \qquad 6.54d$$

$$E_{\phi} = \frac{jk\eta aH_0}{p'_{nm}} J'_n\left(\frac{p'_{nm}\rho}{a}\right) \cos n\phi \sin \frac{\ell\pi z}{d}, \qquad 6.54\epsilon$$

$$E_z = 0, \qquad \qquad 6.54f$$

where $\eta = \sqrt{\mu/\epsilon}$, and $H_0 = -2jA^+$.

Since the time-average stored electric and magnetic energies are equal, the total stored energy is

$$W = 2W_{e} = \frac{\epsilon}{2} \int_{z=0}^{d} \int_{\phi=0}^{2\pi} \int_{\rho=0}^{a} (|E_{\rho}|^{2} + |E_{\phi}|^{2})\rho d\rho d\phi dz$$
$$= \frac{\epsilon k^{2} \eta^{2} a^{2} \pi dH_{0}^{2}}{4(p'_{nm})^{2}} \int_{\rho=0}^{a} \left[J_{n}^{\prime 2} \left(\frac{p'_{nm} \rho}{a} \right) + \left(\frac{na}{p'_{nm} \rho} \right)^{2} J_{n}^{2} \left(\frac{p'_{nm} \rho}{a} \right) \right] \rho d\phi$$
$$= \frac{\epsilon k^{2} \eta^{2} a^{4} H_{0}^{2} \pi d}{8(p'_{nm})^{2}} \left[1 - \left(\frac{n}{p'_{nm}} \right)^{2} \right] J_{n}^{2} (p'_{nm}), \qquad 6.55$$

where the integral identity of Appendix C.17 has been used. The power loss in the conducting walls is

$$P_{c} = \frac{R_{s}}{2} \int_{S} |\bar{H}_{tan}|^{2} ds$$

$$= \frac{R_{s}}{2} \left\{ \int_{z=0}^{d} \int_{\phi=0}^{2\pi} \left[|H_{\phi}(\rho = a)|^{2} + |H_{z}(\rho = a)|^{2} \right] a \, d\phi \, dz$$

$$+ 2 \int_{\phi=0}^{2\pi} \int_{\rho=0}^{a} \left[|H_{\rho}(z = 0)|^{2} + |H_{\phi}(z = 0)|^{2} \right] \rho \, d\rho \, d\phi \right\}$$

$$= \frac{R_{s}}{2} \pi H_{0}^{2} J_{n}^{2} (p_{nm}') \left\{ \frac{da}{2} \left[1 + \left(\frac{\beta an}{(p_{nm}')^{2}} \right)^{2} \right] + \left(\frac{\beta a^{2}}{p_{nm}'} \right)^{2} \left(1 - \frac{n^{2}}{(p_{nm}')^{2}} \right) \right\}. \quad 6.56$$

Then, from (6.8), the Q of the cavity with imperfectly conducting walls but lossless dielectric is

$$Q_{c} = \frac{\omega_{0}W}{P_{c}} = \frac{(ka)^{3}\eta ad}{4(p_{nm}')^{2}R_{s}} \frac{1 - \left(\frac{n}{p_{nm}'}\right)^{2}}{\left\{\frac{ad}{2}\left[1 + \left(\frac{\beta an}{(p_{nm}')^{2}}\right)^{2}\right] + \left(\frac{\beta a^{2}}{p_{nm}'}\right)^{2}\left(1 - \frac{n^{2}}{(p_{nm}')^{2}}\right)\right\}}.$$
 6.57

From (6.52) and (6.51) we see that $\beta = \ell \pi/d$ and $(ka)^2$ are constants that do not vary with frequency, for a cavity with fixed dimensions. Thus, the frequency dependence of Q_c is given by k/R_s , which varies as $1/\sqrt{f}$; this gives the variation in Q_c for a given resonant mode and cavity shape (fixed n, m, ℓ , and a/d).

Figure 6.10 shows the normalized Q due to conductor loss for various resonant modes of a cylindrical cavity. Observe that the TE₀₁₁ mode has a Q significantly higher than the lower-order TE₁₁₁, TM₀₁₀, or TM₁₁₁ modes.

To compute the Q due to dielectric loss, we must compute the power dissipated in the dielectric. Thus,

$$P_{d} = \frac{1}{2} \int_{V} \bar{J} \cdot \bar{E}^{*} dv = \frac{\omega \epsilon''}{2} \int_{V} \left[|E_{\rho}|^{2} + |E_{\phi}|^{2} \right] dv$$

$$= \frac{\omega \epsilon'' k^{2} \eta^{2} a^{2} H_{0}^{2} \pi d}{4(p'_{nm})^{2}} \int_{\rho=0}^{a} \left[\left(\frac{na}{p'_{nm} \rho} \right)^{2} J_{n}^{2} \left(\frac{p'_{nm} \rho}{a} \right) + J_{n}^{'2} \left(\frac{p'_{nm} \rho}{a} \right) \right] \rho \, d\rho$$

$$= \frac{\omega \epsilon'' k^{2} \eta^{2} a^{4} H_{0}^{2}}{8(p'_{nm})^{2}} \left[l - \left(\frac{n}{p'_{nm}} \right)^{2} \right] J_{n}^{2} \langle p'_{nm} \rangle.$$

6.58

Then (6.8) gives the Q as

$$Q_d = \frac{\omega W}{P_d} = \frac{\epsilon}{\epsilon''} = \frac{1}{\tan \delta},$$
6.59

where $\tan \delta$ is the loss tangent of the dielectric. This is the same as the result for Q_d of (6.48) for the rectangular cavity. When both conductor and dielectric losses are present, the total cavity Q can be found from (6.49).





Ø

EXAMPLE 6.4 Design of a Circular Cavity Resonator

A circular cavity resonator with d = 2a is to be designed to resonate at 5.0 GHz in the TE₀₁₁ mode. If the cavity is made from copper and is air-filled, find its dimensions and Q.

Solution

$$\lambda = \frac{300}{5000} = 0.06 \text{ m},$$
$$k = \frac{2\pi}{\lambda} = 104.7 \text{ m}^{-1}$$

From (6.53a), the resonant frequency of the TE_{011} mode is

$$f_{011} = \frac{c}{2\pi} \sqrt{\left(\frac{p_{01}'}{a}\right)^2 + \left(\frac{\pi}{d}\right)^2}.$$

Thus, since d = 2a

$$\frac{2\pi f_{011}}{\boldsymbol{c}} = k = \sqrt{\left(\frac{p_{01}'}{a}\right)^2 + \left(\frac{\pi}{2a}\right)^2}.$$

Solving for a gives

$$a = \frac{\sqrt{(p'_{01})^2 + (\pi/2)^2}}{k} = \frac{\sqrt{(3.832)^2 + (\pi/2)^2}}{104.7} = 3.96 \text{ cm}$$

Then d = 2a = 7.91 cm.

The surface resistivity of copper ($\sigma = 5.813 \times 10^7$ S/m) at 5 GHz is

$$R_s = \sqrt{rac{\omega\mu}{2\sigma}} = 0.0184 \ \Omega.$$

Then from (6.57), with n = 0, $m = \ell = 1$, and d = 2a, the Q is

$$Q_{c} = \frac{(ka)^{3}\eta ad}{4(p_{01}')^{2}R_{s}} \frac{1}{[(ad/2) + (\beta a^{2}/p_{01}')]^{2}} = \frac{ka\eta}{2R_{s}} = 42,400$$

This can be compared with the rectangular cavity of Example 6.3, which had $Q_c = 3,380$ for the TE₁₀₁ mode and $Q_c = 3,864$ for the TE₁₀₂ mode. Note from Figure 6.10 that d = 2a is close to the optimum size for maximizing the Q of the TE₀₁₁ mode.

6.5

DIELECTRIC RESONATORS

A small disc or cube of low-loss high dielectric constant material can also be used as a microwave resonator. Such dielectric resonators are similar in principle to the rectangular or cylindrical cavities previously discussed; the high dielectric constant of the resonator ensures that most of the fields are contained within the dielectric but, unlike metallic cavities, there is some field fringing or leakage from the sides and ends of the dielectric resonator. Such a resonator is generally smaller in cost, size, and weight than an equivalent metallic cavity, and can very easily be incorporated into microwave integrated circuits and coupled to planar transmission lines. Materials with dielectric constants $10 \le \epsilon_r \le 100$ are generally used, with barium tetratitanate and titanium dioxide being typical examples. Conductor losses are absent, but dielectric loss usually increases with dielectric constant; Qs of up to several thousand can be achieved, however. By using an adjustable metal plate above the resonator, the resonant frequency can be mechanically tuned.

Below we will present an approximate analysis for the resonant frequencies of the TE_{016} mode of a cylindrical dielectric resonator; this mode is the one most commonly used in practice, and is analogous to the TE_{011} mode of a circular metallic cavity.

Resonant Frequencies of TE₀₁₀ Mode

The geometry of a cylindrical dielectric resonator is shown in Figure 6.11. The basic operation of the TE₀₁ mode can be explained as follows. The dielectric resonator is considered as a short length, L, of dielectric waveguide open at both ends. The lowest order TE mode of this guide is the TE₀₁ mode, and is the dual of the TM₀₁ mode of a circular metallic waveguide. Because of the high permittivity of the resonator, propagation along the z-axis can occur inside the dielectric at the resonant frequency, but the fields will be cut off (evanescent) in the air regions around the dielectric. Thus the H_z field will look like that sketched in Figure 6.12; higher-order resonant modes will have more variations in the z direction inside the resonator. Since the resonant length, L, for the TE_{01δ} mode is less than $\lambda_g/2$, (where λ_g is the guide wavelength of the TE₀₁ dielectric waveguide mode), the symbol $\delta = 2L/\lambda_g < 1$ is used to denote the z variation of the resonant mode. Thus the equivalent circuit of the resonator looks like a length of transmission line terminated in purely reactive loads at both ends.

Our analysis will follow that of reference [2], which involves the assumption that a magnetic wall boundary condition can be imposed at $\rho = a$. This approximation is



FIGURE 6.11 Geometry of a cylindrical dielectric resonator.



FIGURE 6.12 Magnetic wall boundary condition approximation and distribution of H_z versus z for $\rho = 0$ of the first mode of the cylindrical dielectric resonator.

based on the fact that the reflection coefficient of a wave in a high dielectric constant region incident on an air-filled region approaches +1:

$$\Gamma = \frac{\eta_0 - \eta}{\eta_0 + \eta} = \frac{\sqrt{\epsilon_r} - 1}{\sqrt{\epsilon_r} + 1} \to 1, \quad \text{as } \epsilon_r \to \infty.$$

This reflection coefficient is the same as that obtained at a magnetic wall, or a perfect open circuit.

We begin by finding the fields of the TE_{01} dielectric waveguide mode with a magnetic wall boundary condition at $\rho = a$. For TE modes, $E_z = 0$, and H_z must satisfy the wave equation

$$(\nabla^2 + k^2)H_z = 0, 6.60$$

$$k = \begin{cases} \sqrt{\epsilon_{\tau}}k_0 & \text{for } |z| < L/2\\ k_0 & \text{for } |z| > L/2. \end{cases}$$
6.61

where

Since $\partial/\partial \phi = 0$, the transverse fields are given by (3.110) as follows:

$$H_{\rho} = \frac{-j\beta}{k_c^2} \frac{\partial H_z}{\partial \rho}, \qquad \qquad 6.62b$$

where $k_c^2 = k^2 - \beta^2$. Since H_z must be finite at $\rho = 0$ and zero at $\rho = a$ (the magnetic wall), we have

$$H_z = H_0 J_0(k_c \rho) e^{\pm j\beta z}, \qquad 6.63$$

where $k_c = p_{01}/a$, and $J_0(p_{01}) = 0$ ($p_{01} = 2.405$). Then from (6.62) the transverse fields are

$$E_{\phi} = \frac{j\omega\mu_0 H_0}{k_c} J'_0(k_c \rho) e^{\pm j\beta z}, \qquad 6.64a$$

$$H_{\rho} = \frac{\mp j\beta H_0}{k_c} J_0'(k_c \rho) e^{\pm j\beta z}.$$
 6.64b

Now in the dielectric region, |z| < L/2, the propagation constant is real:

$$\beta = \sqrt{\epsilon_r k_0^2 - k_c^2} = \sqrt{\epsilon_r k_0^2 - \left(\frac{p_{01}}{a}\right)^2}, \qquad 6.65a$$

and a wave impedance can be defined as

$$Z_d = \frac{E_\phi}{H_\rho} = \frac{\omega\mu_0}{\beta}.$$
 6.65b

In the air region, |z| > L/2, the propagation constant will be imaginary, so it is convenient to write

$$\alpha = \sqrt{k_c^2 - k_0^2} = \sqrt{\left(\frac{p_{01}}{a}\right)^2 - k_0^2},$$
 6.66a

and to define a wave impedance in the air region as

$$Z_a = \frac{j\omega\mu_0}{\alpha}, \qquad \qquad 6.66b$$

which is seen to be imaginary.

From symmetry, the H_z and E_{ϕ} field distributions for the lowest-order mode will be even functions about z = 0. Thus the transverse fields for the TE_{01 δ} mode can be written for |z| < L/2 as

$$E_{\phi} = A J_0'(k_c \rho) \cos \beta z, \qquad 6.67a$$

$$H_{\rho} = \frac{-jA}{Z_d} J'_0(k_c \rho) \sin \beta z, \qquad 6.67b$$

and for |z| > L/2 as

$$E_{\phi} = B J_0'(k_c \rho) e^{-\alpha |z|}, \qquad 6.68a$$

$$H_{\rho} = \frac{\pm B}{Z_a} J_0'(k_c \rho) e^{-\alpha |z|}, \qquad 6.68b$$

where A and B are unknown amplitude coefficients. In (6.68b), the \pm sign is used for z > L/2 or z < -L/2, respectively.

Matching tangential fields at z = L/2 (or z = -L/2) leads to the following two equations:

$$A\cos\frac{\beta L}{2} = Be^{-\alpha L/2}, \qquad 6.69a$$

$$\frac{-jA}{Z_d}\sin\frac{\beta L}{2} = \frac{B}{Z_a}e^{-\alpha L/2},$$
 6.69b

which can be reduced to a single transcendental equation:

$$-jZ_a\sin\frac{\beta L}{2}=Z_d\cos\frac{\beta L}{2}.$$

Using (6.65b) and (6.66b) allows this to be written as

$$\tan\frac{\beta L}{2} = \frac{\alpha}{\beta},\tag{6.70}$$

where β is given by (6.65a) and α is given by (6.66a). This equation can be solved numerically for k_0 , which determines the resonant frequency.

This solution is relatively crude, since it ignores fringing fields at the sides of the resonator, and yields accuracies only on the order of 10% (not accurate enough for most practical purposes), but it serves to illustrate the basic behavior of dielectric resonators. More accurate solutions are available in the literature [3].

The Q of the resonator can be calculated by determining the stored energy (inside and outside the dielectric cylinder), and the power dissipated in the dielectric and possibly lost to radiation. If the latter is small, the Q can be approximated as $1/\tan \delta$, as in the case of the metallic cavity resonators.

EXAMPLE 6.5 Resonant Frequency and Q of a Dielectric Resonator

Find the resonant frequency and approximate Q for the TE₀₁₆ mode of a dielectric resonator made from titania, with $\epsilon_r = 95$, and $\tan \delta = 0.001$. The resonator dimensions are a = 0.413 cm, and L = 0.8255 cm.

Solution

The transcendental equation of (6.70) must be solved for k_0 , with β and α given by (6.65a) and (6.66a). Thus,

$$\tan \frac{\beta L}{2} = \frac{\alpha}{\beta},$$

$$\alpha = \sqrt{(2.405/a)^2 - k_0^2},$$

$$\beta = \sqrt{\epsilon_r k_0^2 - (2.405/a)^2},$$

$$k_0 = \frac{2\pi f}{\beta}.$$

where

Since α and β must both be real, the possible frequency range is from f_1 to f_2 , where

$$f_1 = \frac{ck_0}{2\pi} = \frac{c(2.405)}{2\pi\sqrt{\epsilon_r}a} = 2.853 \text{ GHz.}$$
$$f_2 = \frac{ck_0}{2\pi} = \frac{c(2.405)}{2\pi a} = 27.804 \text{ GHz.}$$

Using the interval-halving method (see the Point of Interest on root-finding algorithms in Chapter 3) to find the root of the above equation gives a resonant frequency of about 3.152 GHz. This compares with a measured value of about 3.4 GHz from reference [2], indicating a 10% error. The approximate Q, due to dielectric loss, is

$$Q_d = \frac{1}{\tan \delta} = 1000.$$

6.6 FABRY-PEROT RESONATORS

In principle, the previously described resonators can be used at arbitrarily high frequencies. But examination of the expressions for Q_c due to conductor losses shows that the Q will decrease as $1/\sqrt{f}$ for a given cavity or transmission line resonator. Thus, at very high frequencies the Q of such resonators may be too small to be useful. In addition, at high frequencies the physical size of a cavity operating in a low-order mode may be too small to be practical. If a high-order mode is used, the resonances of nearby modes will be very close in frequency, and because of the finite bandwidth of these modes there may be little or no separation between them, making such resonators unusable.

A conceptual way of avoiding these difficulties is to remove the sidewalls of a cavity resonator, which has the effect of reducing conductor losses and the number of possible resonant modes. Such an *open resonator* thus consists of two parallel metal plates, as shown in Figure 6.13; this is also known as a Fabry-Perot resonator, since it is similar in principle to the optical Fabry-Perot interferometer. In order for the device to be useful, the plates must be very parallel, and large enough in extent so that no significant radiation leaves the region between the plates. These constraints can be relaxed by using spherical or parabolic-shaped reflecting mirrors to focus and confine the energy to a



FIGURE 6.13 An idealized Fabry-Perot resonator consisting of two parallel conducting plates.

stable mode pattern. Such *quasi-optical* resonators are very useful at millimeter wave and sub-millimeter wave frequencies, and are similar to those used in laser applications at infrared and visible wavelengths. Figure 6.14 shows a photograph of such a resonator. These types of resonators also find applications in the measurement of dielectric constants at millimeter wave frequencies [4]. We will first consider the operation of the idealized parallel plate resonator of Figure 6.13, and then discuss the stability of resonators with plane and spherical mirrors.

If we assume the parallel plates of Figure 6.13 to be infinite in extent, then a TEM (plane wave) standing wave field can exist between the plates with the following form:

$$E_x = E_0 \sin k_0 z, \qquad \qquad 6.71a$$

$$H_y = \frac{jE_0}{\eta_0} \cos k_0 z, \qquad 6.71b$$

where E_0 is an arbitrary amplitude constant and $\eta_0 = 377 \ \Omega$ is the intrinsic impedance of free-space. These fields already satisfy the boundary condition that $E_x = 0$ at z = 0; to satisfy the boundary conditions that $E_x = 0$ at z = d, we must have that

$$k_0 d = \ell \pi, \qquad \ell = 1, 2, 3, ...,$$

 $f_0 = \frac{ck_0}{2\pi} = \frac{c\ell}{2d}, \qquad \text{for } \ell = 1, 2, 3, ...,$ 6.72



FIGURE 6.14 Photograph of a Fabry-Perot resonator operating at 183 GHz with a mode number of 244 (nominally). One reflector is a flat plate, and the other is a movable spherical reflector.

Photograph courtesy of Millitech Corporation, S. Deerfield, Mass.

or

which determines the resonant frequencies. Note that there is only a single index, ℓ , for these modes, as opposed to three indices for rectangular or cylindrical cavities; this is a consequence of the absence of conducting sidewalls.

We can determine the Q of the resonator as follows. The stored electric energy is, for 1 m² of cross section,

$$W_{e} = \frac{\epsilon_{0}}{4} \int_{z=0}^{d} |E_{x}|^{2} dz = \frac{\epsilon_{0} |E_{0}|^{2}}{4} \int_{z=0}^{d} \sin^{2} \frac{\ell \pi z}{d} dz = \frac{\epsilon_{0} |E_{0}|^{2} d}{8}.$$
 6.73*a*

The stored magnetic energy per square meter of cross section is

$$W_m = \frac{\mu_0}{4} \int_{z=0}^d |H_y|^2 dz = \frac{\mu_0 |E_0|^2}{4\eta_0^2} \int_{z=0}^d \cos^2 \frac{\ell \pi z}{d} dz = \frac{\mu_0 |E_0|^2 d}{8\eta_0^2} = \frac{\epsilon_0 |E_0|^2 d}{8}, \quad 6.73b$$

which is seen to be equal to the stored electric energy. The power lost per square meter in both conducting plates is

$$P_c = 2\left(\frac{R_s}{2}\right)|H_g(z=0)|^2 = \frac{R_s|E_0|^2}{\eta_0^2},$$
6.74

so the Q due to conductor loss is

$$Q_c = \frac{\omega(W_e + W_m)}{P_c} = \frac{\omega\epsilon_0 d\eta_0^2}{4R_s} = \frac{\pi f_0 \epsilon_0 d\eta_0^2}{2R_s} = \frac{c\pi \ell \epsilon_0 \eta_0^2}{4R_s} = \frac{\pi \ell \eta_0}{4R_s}.$$
 6.75

which shows that the Q increases in proportion to the mode number ℓ ; ℓ is often several thousand or more for such resonators. If the region between the plates is filled with a dielectric material with tangent δ , it is easy to show that the Q due to dielectric loss is

$$Q_d = \frac{1}{\tan \delta}.$$
 6.76

Dielectric is seldom used in such resonators, however, because of its limiting effect on the Q.

Stability of Open Resonators

We now qualitatively discuss some of the properties of open resonators using curved mirrors. The general geometry is shown in Figure 6.15, which shows two spherical mirrors having radii of curvature R_1 and R_2 , and separated by a distance *d*. Depending on the focusing properties of these mirrors, the energy in the resonator may be confined to a narrow region about the axis of the mirrors (stable), or it may spread out beyond the edges of the mirrors (unstable), resulting in a high degree of loss.

Using ray optics [5], it can be shown that the open resonator geometry of Figure 6.15 will support a stable mode if the following condition is met:

$$0 \le \left(1 - \frac{d}{R_1}\right) \left(1 - \frac{d}{R_2}\right) \le 1.$$
6.77



FIGURE 6.15 Geometry of an open resonator using two spherical mirrors.

This stability criterion can be presented in graphical form, as shown in Figure 6.16. The boundaries associated with the inequality on the left-hand side of (6.77) are straight lines at $d/R_1 = 1$ and d/R - 2 = 1, while the boundaries associated with the inequality on the right-hand side of (6.77) are hyperbolas with a focus at $d/R_1 = d/R - 2 = 1$. We can now consider some special configurations.

Parallel-plane resonator. This is essentially the idealized resonator of Figure 6.13. The radii of curvature are $R_1 = R_2 = \infty$, so this configuration corresponds to the point $d/R_1 = d/R_2 = 0$ in Figure 6.16, which is seen to be right on the boundary between a stable and unstable region. Thus, any irregularities, such as a lack of parallelism in the mirrors, will result in an unstable system.

Confocal resonator. In this case, $R_1 = R_2 = d$, corresponding to a symmetrical configuration. This resonator is represented by a point between a stable region and unstable region, and so is very sensitive to irregularities.



FIGURE 6.16 Stability diagram for open resonators.

Concentric resonators. Here $R_1 = R_2 = d/2$, and the two mirrors have the same center, hence the term concentric. This configuration also lies at the edge of a stable and unstable region.

Stable resonators. Symmetrical spherical resonators can be made stable by choosing $d/R_1 = d/R_2$ around 0.6, in which case the resonator is between the confocal and parallel plane designs, or around 1.4, in which case the resonator is between the confocal and concentric designs.

6.7 EXCITATION OF RESONATORS

We now discuss how the resonators of the previous sections can be coupled to external circuitry. In general, the way in which this is done depends on the type of resonator under consideration; some typical coupling techniques are shown for various resonators in Figure 6.17. In this section we will discuss the operation of some of the more common coupling techniques, notably gap coupling and aperture coupling. First we will illustrate the concept of critical coupling, whereby a resonator can be matched to a feedline, using a lumped-element resonant circuit.

Critical Coupling

To obtain maximum power transfer between a resonator and a feedline, the resonator must be matched to the feed at the resonant frequency. The resonator is then said to be critically coupled to the feed. We will first illustrate the basic concept of critical coupling by considering the series resonant circuit shown in Figure 6.18.

From (6.9), the input impedance near resonance of the series resonant circuit of Figure 6.18 is given by

$$Z_{\rm in} = R + j2L\Delta\omega = R + j\frac{2RQ\Delta\omega}{\omega_0}, \qquad 6.78$$

and the unloaded Q is, from (6.8),

$$Q = \frac{\omega_0 L}{R}.$$
 6.79

At resonance, $\Delta \omega = 0$, so from (6.78) the input impedance is $Z_{in} = R$. In order to match the resonator to the line we must have,

$$R = Z_0. \tag{6.80}$$

Then the unloaded Q is

$$Q = \frac{\omega_0 L}{Z_0}.$$
 6.81

From (6.22), the external Q is

$$Q_e = \frac{\omega_0 L}{Z_0} = Q. \tag{6.82}$$



FIGURE 6.17 Coupling to microwave resonators. (a) A microstrip transmission line resonator gap coupled to a microstrip feedline. (b) A rectangular cavity resonator fed by a coaxial probe. (c) A circular cavity resonator aperture coupled to a rectangular waveguide. (d) A diclectric resonator coupled to a microstrip feedline. (e) A Fabry-Perot resonator fed by a waveguide horn antenna.

which shows that the external and unloaded Qs are equal under the condition of critical coupling.

It is useful to define a coefficient of coupling, g, as

$$g = \frac{Q}{Q_{g}},$$

$$- \frac{R}{Z_{0}} + \frac{L}{Z_{0}} + C$$

FIGURE 6.18 A series resonant circuit coupled to a feedline.

6.83

which can be applied to both series $(g = Z_0/R)$ and parallel $(g = R/Z_0)$ resonant circuits. Then, three cases can be distinguished.

- 1. g < 1 The resonator is said to be undercoupled to the feedline.
- **2.** g = 1 The resonator is critically coupled to the feedline.
- 3. g > 1 The resonator is said to be overcoupled to the feedline.

Figure 6.19 shows a Smith chart sketch of the impedance loci for the series resonant circuit, as given by (6.78), for various values of R corresponding to the above cases.

A Gap-Coupled Microstrip Resonator

Next we consider a $\lambda/2$ open-circuited microstrip resonator coupled to a microstrip feedline, as shown in Figure 6.17a. The gap in the microstrip line can be approximated as a series capacitor, so the equivalent circuit of this resonator and feed can be constructed



FIGURE 6.19 Smith chart illustrating coupling to a series RLC circuit.



FIGURE 6.20 Equivalent circuit of the gap-coupled microstrip resonator of Figure 6.17a.

as shown in Figure 6.20. The normalized input impedance seen by the feedline is then

$$z = \frac{Z}{Z_0} = -j \frac{\left[(1/\omega C) + Z_0 \cot \beta \ell \right]}{Z_0} = -j \left(\frac{\tan \beta \ell + b_e}{b_e \tan \beta \ell} \right), \qquad 6.84$$

where $b_c = Z_0 \omega C$ is the normalized susceptance of the coupling capacitor, C. Resonance occurs with z = 0, or when

$$\tan \beta \ell + b_c = 0. \tag{6.85}$$

The solutions to this transcendental equation are sketched in Figure 6.21. In practice, $b_c \ll 1$, so that the first resonant frequency, ω_1 , will be close to the frequency for which $\beta \ell = \pi$ (the first resonant frequency of the unloaded resonator). In this case the coupling of the feedline to the resonator has the effect of lowering its resonant frequency.

We now wish to simplify the driving point impedance of (6.84) to relate this resonator to a series *RLC* equivalent circuit. This can be accomplished by expanding $z(\omega)$ in a Taylor series about the resonant frequency, ω_1 , and assuming that b_c is small. Thus,

$$\boldsymbol{z}(\omega) = \boldsymbol{z}(\omega_1) + (\omega - \omega_1) \frac{d\boldsymbol{z}(\omega)}{d\omega} \bigg|_{\omega_1} + \cdots, \qquad 6.86$$



FIGURE 6.21 Solutions to (6.85) for the resonant frequencies of the gap-coupled microstrip resonator.

From (6.84) and (6.85), $z(\omega_1) = 0$. Then,

$$\left. \frac{dz}{d\omega} \right|_{\omega_1} = \frac{-j\sec^2\beta\ell}{b_c\tan\beta\ell} \frac{d(\beta\ell)}{d\omega} = \frac{j(1+b_c^2)}{b_c^2} \frac{\ell}{v_p} \simeq \frac{j}{b_c^2} \frac{\ell}{v_p} \simeq \frac{j\pi}{\omega_1 b_c^2},$$

since $b_c \ll 1$ and $\ell \simeq \pi v_p/\omega_1$, where v_p is the phase velocity of the transmission line (assumed TEM). Then the normalized impedance can be written as

$$z(\omega) = \frac{j\pi(\omega - \omega_1)}{\omega_1 b_c^2}.$$
 6.87

So far we have ignored losses, but for a high-Q cavity loss can be included by replacing the resonant frequency ω_1 with the complex resonant frequency given by $\omega_1(1 + j/2Q)$, which follows from (6.10). Applying this procedure to (6.87) gives the input impedance of gap-coupled lossy resonator as

$$z(\omega) = \frac{\pi}{2Qb_c^2} + j\frac{\pi(\omega - \omega_1)}{\omega_1 b_c^2}.$$
 6.88

Note that an uncoupled $\lambda/2$ open-circuited transmission line resonator looks like a parallel *RLC* circuit near resonance, but the present case of a capacitive coupled $\lambda/2$ resonator looks like a series *RLC* circuit near resonance. This is because the series coupling capacitor has the effect of inverting the driving point impedance of the resonator (see the discussion of impedance inverters in Section 8.5).

At resonance, then, the input resistance is $R = Z_0 \pi / 2Qb_c^2$. For critical coupling we must have $R = Z_0$, or

$$b_c = \sqrt{\frac{\pi}{2Q}}.$$
 6.89

The coupling coefficient of (6.83) is

$$g = \frac{R}{Z_0} = \frac{\pi}{2Qb_c^2}.$$
 6.90

If $b_c > \sqrt{\pi/2Q}$, then g < 1 and the resonator is undercoupled; if $b_c < \sqrt{\pi/2Q}$, then g > 1 and the resonator is overcoupled.



EXAMPLE 6.6 Design of a Gap-Coupled Microstrip Resonator

A resonator is made from an open-circuited 50 Ω microstrip line, and is gapcoupled to a 50 Ω feedline, as in Figure 6.17a. The resonator has a length of 2.175 cm, an effective dielectric constant of 1.9, and an attenuation of 0.01 dB/cm near its resonance. Find the value of the coupling capacitor required for critical coupling, and the resulting resonant frequency.

Solution

The first resonant frequency will occur when the resonator is about $\ell = \lambda_g/2$ in length. Thus, ignoring fringing fields, the approximate resonant frequency is

$$f_0 = \frac{v_p}{\lambda_g} = \frac{c}{2\ell\sqrt{\epsilon_e}} = \frac{3 \times 10^8}{2(0.02175)\sqrt{1.9}} = 5.00 \text{ GHz},$$

which does not include the effect of the coupling capacitor. Then from (6.35) the Q of this resonator is

$$Q = \frac{\beta}{2\alpha} = \frac{\pi}{\lambda_g \alpha} = \frac{\pi}{2\ell\alpha} = \frac{\pi (8.7 \text{ dB/Np})}{2(0.02175 \text{ m})(1 \text{ dB/m})} = 628.$$

From (6.89) the normalized coupling capacitor susceptance is

$$b_c = \sqrt{\frac{\pi}{2Q}} = \sqrt{\frac{\pi}{2(628)}} = 0.05.$$

so the coupling capacitor has a value of

$$C = \frac{b_c}{\omega Z_0} = \frac{0.05}{2\pi (5 \times 10^9)(50)} = 0.032 \text{ pF},$$

which should result in the critical coupling of the resonator to the 50 Ω feedline.

Now that C is determined, the exact resonant frequency can be found by solving the transcendental equation of (6.85). Since we know from the graphical solution of Figure 6.21 that the actual resonant frequency is slightly lower than the unloaded resonant frequency of 5.0 GHz, it is an easy matter to calculate (6.85) for several frequencies in this vicinity, which leads to a value of about 4.918 GHz. This is about 1.6% lower than the unloaded resonant frequency. Figure 6.22 shows a Smith chart plot of the input impedance of the gap-coupled resonator for coupling capacitor values that lead to under, critical, and overcoupled resonators.

An Aperture-Coupled Cavity

As a final example of resonator excitation, we will consider the aperture coupled waveguide cavity shown in Figure 6.23. As discussed in Section 4.8, a small aperture in the transverse wall acts as a shunt inductance. If we consider the first resonant mode of the cavity, which occurs when the cavity length $\ell = \lambda_g/2$, then the cavity can be considered as a transmission line resonator shorted at one end. The aperture-coupled cavity can then be modeled by the equivalent circuit shown in Figure 6.24. This circuit is basically the dual of the equivalent circuit of Figure 6.20, for the gap-coupled microstrip resonator, so we will approach the solution in the same manner.

The normalized input admittance seen by the feedline is

$$y = Z_0 Y = -j \left(\frac{Z_0}{X_L} + \cot \beta \ell \right) = -j \left(\frac{\tan \beta \ell + x_L}{x_L \tan \beta \ell} \right), \qquad 6.91$$

where $x_L = \omega L/Z_0$ is the normalized reactance of the aperture. An antiresonance occurs when the numerator of (6.91) vanishes, or when

$$\tan\beta\ell + x_L = 0, \qquad 6.92$$

which is similar in form to (6.85), for the case of the gap-coupled microstrip resonator. In practice, $x_L \ll 1$, so that the first resonant frequency, ω_1 , will be close to the resonant frequency for which $\beta \ell = \pi$, similar to the solution illustrated in Figure 6.21.



FIGURE 6.22 Smith chart plot of input impedance of the gap-coupled microstrip resonator of Example 6.6 versus frequency for various values of the coupling capacitor.

Using the same procedure as in the previous section, the input admittance of (6.91) can be expanded in a Taylor series about the resonant frequency, ω_i , assuming $x_L \ll 1$, to obtain

$$y(\omega) = y(\omega_1) + (\omega - \omega_1) \frac{dy(\omega)}{d\omega} \bigg|_{\omega_1} + \cdots \simeq \frac{j\ell}{x_{L_{\epsilon}}^2} (\omega - \omega_1) \frac{d\beta}{d\omega} \bigg|_{\omega_1}, \qquad 6.93$$

since $y(\omega_1) = 0$. For a rectangular waveguide,

$$\frac{d\beta}{d\omega} = \frac{d}{d\omega}\sqrt{k_0^2 - k_c^2} = \frac{k_0}{\beta c}$$

where c is the speed of light. Then (6.93) can be reduced to

$$y(\omega) = \frac{j\pi k_0(\omega - \omega_1)}{\beta^2 c x_L^2}.$$
6.94

In (6.94), k_0, β , and x_L should be evaluated at the resonant frequency ω_1 .



FIGURE 6.23 A rectangular waveguide aperture coupled to a rectangular cavity.

Loss can now be included by assuming a high-Q cavity and replacing ω_1 in the numerator of (6.94) with $\omega_1(1 + j/2Q)$, to obtain

$$y(\omega) \simeq \frac{\pi k_0 \omega_1}{2Q\beta^2 c x_L^2} + j \frac{\pi k_0 (\omega - \omega_1)}{\beta^2 c x_L^2}.$$
6.95

At resonance, the input resistance is $R = 2Q\beta^2 c r_L^2 Z_0 / \pi k_0 \omega_1$. To obtain critical coupling we must have $R = Z_0$, which yields the required aperture reactance as

$$X_L = Z_0 \sqrt{\frac{\pi k_0 \omega_1}{2Q\beta^2 c}}.$$
6.96

From X_L , the necessary aperture size can be found.

The next resonant mode for the aperture-coupled cavity occurs when the input impedance becomes zero, or $Y \to \infty$. From (6.91) it is seen that this occurs at a frequency such that $\tan \beta \ell = 0$, or $\beta \ell = \pi$. In this case the cavity is exactly $\lambda_g/2$ long, so a null in the transverse electric field exists at the aperture plane, and the aperture has no effect. This mode is of little practical interest, because of this loose coupling.

The excitation of a cavity resonator by an electric current probe or loop can be analyzed by the method of modal analysis, similar to that discussed in Sections 4.7 and 4.8. The procedure is complicated, however, by the fact that a complete modal expansion requires fields having irrotational (zero curl) components. The interested reader is referred to references [1] and [6].



FIGURE 6.24 Equivalent circuit of the aperture-coupled cavity.

6.8 CAVITY PERTURBATIONS

In practical applications cavity resonators are often modified by making small changes in their shape, or by the introduction of small pieces of dielectric or metallic materials. For example, the resonant frequency of a cavity can be easily tuned with a small screw (dielectric or metallic) that enters the cavity volume, or by changing the size of the cavity with a movable wall. Another application involves the determination of dielectric constant by measuring the shift in resonant frequency when a small dielectric sample is introduced into the cavity.

In some cases, the effect of such perturbations on the cavity performance can be calculated exactly, but often approximations must be made. One useful technique for doing this is the perturbational method, which assumes that the actual fields of a cavity with a small shape or material perturbation are not greatly different from those of the unperturbed cavity. Thus, this technique is similar in concept to the perturbational method introduced in Section 2.7 for treating loss in good conductors, where it was assumed that there was not a significant difference between the fields of a component with good conductors and one with perfect conductors.

In this section we will derive expressions for the approximate change in resonant frequency when a cavity is perturbed by small changes in the material filling the cavity, or by small changes in its shape.

Material Perturbations

Figure 6.25 shows a cavity perturbed by a change in the permittivity ($\Delta\epsilon$), or permeability ($\Delta\mu$), of all or part of the material filling the cavity. If \bar{E}_0 , \bar{H}_0 are the fields of the original cavity, and \bar{E} , \bar{H} are the fields of the perturbed cavity, then Maxwell's curl equations can be written for the two cases as

$$\nabla \times \bar{E}_0 = -j\omega_0 \mu \bar{H}_0, \qquad \qquad 6.97a$$

$$abla imes ar{H}_0 = j \omega_0 \epsilon ar{E}_0,$$
 6.97b

$$abla imes ar{E} = -j\omega(\mu + \Delta\mu)ar{H},$$
 6.98a

$$\nabla \times \bar{H} = j\omega(\epsilon + \Delta \epsilon)\bar{E}. \qquad 6.98b$$





where ω_0 is the resonant frequency of the original cavity and ω is the resonant frequency of the perturbed cavity.

Now multiply the conjugate of (6.97a) by \hat{H} and multiply (6.98b) by \tilde{E}_0^* to get

$$\bar{H} \cdot \nabla \times \bar{E}_0^* = j\omega_0 \mu \bar{H} \cdot \bar{H}_0^*.$$
$$\bar{E}_0^* \cdot \nabla \times \bar{H} = j\omega(\epsilon + \Delta \epsilon) \bar{E}_0^* \cdot \bar{E}.$$

Subtracting these two equations and using the vector identity (B.8) that $\nabla \cdot (\bar{A} \times \bar{B}) = \bar{B} \cdot \nabla \times \bar{A} - \bar{A} \cdot \nabla \times \bar{B}$ gives

$$\nabla \cdot (\bar{E}_0^* \times \bar{H}) = j\omega_0 \mu \bar{H} \cdot \bar{H}_0^* - j\omega(\epsilon + \Delta \epsilon) \bar{E}_0^* \cdot \bar{E}.$$
 6.99a

Similarly, we multiply the conjugate of (6.97b) by \bar{E} and multiply (6.98a) by \bar{H}_0^* to get

$$ar{E}\cdot
abla imesar{H}_0^*=-j\omega_0\epsilonar{E}_0^*\cdotar{E},\ ar{H}_0^*\cdot
abla imesar{E}=-j\omega(\mu+\Delta\mu)ar{H}_0^*\cdotar{H},$$

Subtracting these two equations and using vector identity (B.8) gives

$$\nabla \cdot (\bar{E} \times \bar{H}_0^*) = -j\omega(\mu + \Delta\mu)\bar{H}_0^* \cdot \bar{H} + j\omega_0\epsilon\bar{E}_0^* \cdot \bar{E}.$$
6.99b

Now add (6.99a) and (6.99b), integrate over the volume V_0 , and use the divergence theorem to obtain

$$\int_{V_0} \nabla \cdot (\bar{E}_0^* \times \bar{H} + \hat{E} \times \bar{H}_0^*) dv = \oint_{S_0} (\bar{E}_0^* \times \hat{H} + \bar{E} \times \bar{H}_0^*) \cdot d\bar{s} = 0$$
$$= j \int_{V_0} \{ [\omega_0 \epsilon - \omega(\epsilon + \Delta \epsilon)] \bar{E}_0^* \cdot \hat{E} + [\omega_0 \mu - \omega(\mu + \Delta \mu)] \bar{H}_0^* \cdot \bar{H} \} dv, \quad 6.100$$

where the surface integral is zero because $\hat{n} \times \vec{E} = 0$ on S_0 . Rewriting gives

$$\frac{\omega - \omega_0}{\omega} = \frac{-\int_{V_0} \left(\Delta \epsilon \bar{E} \cdot \bar{E}_0^* + \Delta \mu \bar{H} \cdot \bar{H}_0^*\right) dv}{\int_{V_0} \left(\epsilon \bar{E} \cdot \bar{E}_0^* + \mu \bar{H} \cdot \bar{H}_0^*\right) dv}.$$
6.101

This is an exact equation for the change in resonant frequency due to material perturbations, but is not in a very usable form since we generally do not know \overline{E} and \overline{H} , the exact fields in the perturbed cavity. But, if we assume that $\Delta \epsilon$ and $\Delta \mu$ are small, then we can approximate the perturbed fields \overline{E} , \overline{H} by the original fields \overline{E}_0 , \overline{H}_0 , and ω in the denominator of (6.101) by ω_0 , to give the fractional change in resonant frequency as

$$\frac{\omega - \omega_0}{\omega_0} \simeq \frac{-\int_{V_0} \left(\Delta \epsilon |\bar{E}_0|^2 + \Delta \mu |\bar{H}_0|^2\right) dv}{\int_{V_0} \left(\epsilon |\bar{E}_0|^2 + \mu |\bar{H}_0|^2\right) dv}.$$
 6.102

This result shows that any increase in ϵ or μ at any point in the cavity will decrease the resonant frequency. The reader may also observe that the terms in (6.102) can be related to the stored electric and magnetic energies in the original and perturbed cavities, so that the decrease in resonant frequency can be related to the increase in stored energy of the perturbed cavity.



EXAMPLE 6.7 Material Perturbation of a Rectangular Cavity

A rectangular cavity operating in the TE_{101} mode is perturbed by the insertion of a thin dielectric slab into the bottom of the cavity, as shown in Figure 6.26. Use the perturbational result of (6.102) to derive an expression for the change in resonant frequency.

Solution

From (6.42a-c), the fields for the unperturbed TE_{101} cavity mode can be written as

$$E_y = A \sin \frac{\pi x}{a} \sin \frac{\pi z}{d},$$
$$H_x = \frac{-jA}{Z_{\text{TE}}} \sin \frac{\pi x}{a} \cos \frac{\pi z}{d},$$
$$H_z = \frac{j\pi A}{k\eta a} \cos \frac{\pi x}{a} \sin \frac{\pi z}{d}.$$

In the numerator of (6.102), $\Delta \epsilon = (\epsilon_r - 1)\epsilon_0$ for $0 \le y \le t$, and zero elsewhere. The integral can then be evaluated as

$$\int_{V} (\Delta \epsilon |\hat{E}_{0}|^{2} + \Delta \mu |\bar{H}_{0}|^{2}) dv = (\epsilon_{r} - 1)\epsilon_{0} \int_{x=0}^{a} \int_{y=0}^{t} \int_{z=0}^{d} |E_{y}|^{2} dz \, dy \, dx$$
$$= \frac{(\epsilon_{r} - 1)\epsilon_{0} A^{2} at d}{4}.$$

The denominator of (6.102) is proportional to the total energy in the unperturbed cavity, which was evaluated in (6.43), thus,

$$\int_V (\epsilon |ar E_0|^2 + \mu |ar H_0|^2) dv = rac{abd\epsilon_0}{2} A^2.$$

Then (6.102) gives the fractional change (decrease) in resonant frequency as

$$\frac{\omega-\omega_0}{\omega_0}=\frac{-(\epsilon_{\tau}-1)t}{2b}.$$





Shape Perturbations

Changing the size of a cavity or inserting a tuning screw can be considered as a change in the shape of the cavity and, for small changes, can also be treated by the perturbation technique. Figure 6.27 shows an arbitrary cavity with a perturbation in its shape; we will derive an expression for the change in resonant frequency.

As in the case of material perturbations, let \hat{E}_0 , \bar{H}_0 , ω_0 be the fields and resonant frequency of the original cavity and let \bar{E} , \bar{H} , ω be the fields and resonant frequency of the perturbed cavity. Then Maxwell's curl equations can be written for the two cases as

$$\nabla \times \tilde{E}_0 = -j\omega_0 \mu \tilde{H}_0, \qquad \qquad 6.103a$$

$$\nabla \times \bar{H}_0 = j\omega_0 \epsilon \bar{E}_0. \qquad 6.103b$$

$$\nabla \times \bar{E} = -j\omega\mu \bar{H}, \qquad \qquad 6.104a$$

$$abla imes ar{H} = j\omega\epsilonar{E}.$$
 6.104b

Now multiply the conjugate of (6.103a) by \bar{H} and multiply (6.104b) by \bar{E}_0^* to get

$$\begin{split} \vec{H} \cdot \nabla \times \vec{E}_0^* &= j\omega_0 \mu \vec{H} \cdot \vec{H}_0^* \\ \vec{E}_0^* \cdot \nabla \times \vec{H} &= j\omega \epsilon \vec{E}_0^* \cdot \vec{E}. \end{split}$$

Subtracting these two equations and using vector identity (B.8) then gives

$$\nabla \cdot (\bar{E}_0^* \times \bar{H}) = j\omega_0 \mu \bar{H} \cdot \bar{H}_0^* - j\omega \epsilon \bar{E}_0^* \cdot \bar{E}.$$
6.105a

Similarly, we multiply the conjugate of (6.103b) by \tilde{E} and (6.104a) by \bar{H}_0^{\bullet} to get

$$\begin{split} \bar{E} \cdot \nabla \times \bar{H}_0^* &= -j\omega_0\epsilon\bar{E}\cdot\bar{E}_0^*, \\ \bar{H}_0^* \cdot \nabla \times \bar{E} &= -j\omega\mu\bar{H}_0^*\cdot\bar{H}. \end{split}$$

Subtracting and applying vector identity (B.8) gives

$$\nabla \cdot (\vec{E} \times \vec{H}_0^*) = -j\omega\mu \hat{H}_0^* \cdot \hat{H} + j\omega_0 \epsilon \bar{E} \cdot \bar{E}_0^* \qquad 6.105b$$





Now add (6.105a) and (6.105b), integrate over the volume V, and use the divergence theorem to obtain

$$\int_{V} \nabla \cdot (\bar{E} \times \bar{H}_{0}^{*} + \bar{E}_{0}^{*} \times \bar{H}) dv = \oint_{S} (\bar{E} \times \bar{H}_{0}^{*} + \bar{E}_{0}^{*} \times \bar{H}) \cdot d\bar{s}$$
$$= \oint_{S} \bar{E}_{0}^{*} \times \bar{H} \cdot d\bar{s} = -j(\omega - \omega_{0}) \int_{V} (\epsilon \bar{E} \cdot \bar{E}_{0}^{*} + \mu \bar{H} \cdot \bar{H}_{0}^{*}) dv, \qquad 6.106$$

since $\hat{n} \times \vec{E} = 0$ on S.

Since the perturbed surface $S = S_0 - \Delta S$, we can write

$$\oint_{S} \bar{E}_{0}^{*} \times \bar{H} \cdot d\bar{s} = \oint_{S_{0}} \hat{E}_{0}^{*} \times \bar{H} \cdot d\bar{s} - \oint_{\Delta S} \bar{E}_{0}^{*} \times \bar{H} \cdot d\bar{s} = -\oint_{\Delta S} \hat{E}_{0}^{*} \times \hat{H} \cdot ds,$$

because $\hat{n} \times \hat{E}_0 = 0$ on S_0 . Using this result in (6.106) gives

$$\omega - \omega_0 = \frac{-j \oint_{\Delta S} \tilde{E}_0^* \times \tilde{H} \cdot d\hat{s}}{\int_V (\epsilon \tilde{E} \cdot \tilde{E}_0^* + \mu \tilde{H} \cdot \bar{H}_0^*) dv},$$
6.107

which is an exact expression for the new resonant frequency, but not a very usable one since we generally do not initially know \bar{E}, \bar{H} , or ω . If we assume ΔS is small, and approximate \bar{E}, \bar{H} by the unperturbed values of \bar{E}_0, \bar{H}_0 , then the numerator of (6.107) can be reduced as follows:

$$\oint_{\Delta S} \bar{E}_0^* \times \tilde{H} \cdot d\bar{s} \simeq \oint_{\Delta S} \bar{E}_0^* \times \bar{H}_0 \cdot d\bar{s} = -j\omega_0 \int_{\Delta V} (\epsilon |\tilde{E}_0|^2 - \mu |\bar{H}_0|^2) dv, \qquad 6.108$$

where the last identity follows from conservation of power, as derived from the conjugate of (1.87) with σ , \bar{J}_s , and \bar{M}_s set to zero. Using this result in (6.107) gives an expression for the fractional change in resonant frequency as

$$\frac{\omega - \omega_0}{\omega_0} \simeq \frac{\int_{V_0} (\mu |\bar{H}_0|^2 - \epsilon |\bar{E}_0|^2) dv}{\int_{V_0} (\mu |\bar{H}_0|^2 + \epsilon |\bar{E}_0|^2) dv}, \qquad 6.109$$

where we have also assumed that the denominator of (6.107), which represents the total energy stored in the perturbed cavity, is approximately the same as that for the unperturbed cavity.

Equation (6,109) can be written in terms of stored energies as follows:

$$\frac{\omega - \omega_0}{\omega_0} = \frac{\Delta W_m - \Delta W_e}{W_m + W_e},$$
6.110

where ΔW_m and ΔW_e are the changes in the stored magnetic energy and electric energy, respectively, after the shape perturbation, and $W_m + W_e$ is the total stored energy in the cavity. These results show that the resonant frequency may either increase or decrease, depending on where the perturbation is located and whether it increases or decreases the cavity volume.

EXAMPLE 6.8 Shape Perturbation of a Rectangular Cavity

A thin screw of radius r_0 extends a distance ℓ through the center of the top wall of a rectangular cavity operating in the TE₁₀₁ mode, as shown in Figure 6.28.


FIGURE 6.28 A rectangular cavity perturbed by a tuning post in the center of the top wall.

If the cavity is air-filled, use (6.109) to derive an expression for the change in resonant frequency from the unperturbed cavity.

Solution

From (6.42a–c), the fields for the unperturbed TE_{101} cavity can be written as

$$E_y = A \sin \frac{\pi x}{a} \sin \frac{\pi z}{d},$$
$$H_x = \frac{-jA}{Z_{\text{TE}}} \sin \frac{\pi x}{a} \cos \frac{\pi z}{d},$$
$$H_z = \frac{j\pi A}{kna} \cos \frac{\pi x}{a} \sin \frac{\pi z}{d}.$$

Now if the screw is thin, we can assume that the fields are constant over the cross-section of the screw and can be represented by the fields at x = a/2, z = d/2;

$$E_y\left(x=rac{a}{2},y,z=rac{d}{2}
ight)=A,$$

 $H_x\left(x=rac{a}{2},y,z=rac{d}{2}
ight)=0,$
 $H_z\left(x=rac{a}{2},y,z=rac{d}{2}
ight)=0.$

Then the numerator of (6.109) can be evaluated as

$$\int_{\Delta V} (\mu |\bar{H}_0|^2 - \epsilon |\bar{E}_0|^2) dv = -\epsilon_0 \int_{\Delta V} A^2 dv = -\epsilon_0 A^2 \Delta V,$$

where $\Delta V = \pi \ell r_0^2$ is the volume of the screw. The denominator of (6.109) is, from (6.43),

$$\int_{V_0} (\mu |ar{H}_0|^2 + \epsilon |ar{E}_0|^2) dv = rac{abd\epsilon_0 A^2}{2} = rac{V_0\epsilon_0 A^2}{2},$$

where $V_0 = abd$ is the volume of the unperturbed cavity. Then (6.109) gives

$$\frac{\omega-\omega_0}{\omega_0}=\frac{-2\ell\pi r_0^2}{abd}=\frac{-2\Delta V}{V_0},$$

which indicates a lowering of the resonant frequency.

0

REFERENCES

- [1] R. E. Collin, Foundations for Microwave Engineering, Second Edition, McGraw-Hill, N.Y., 1992.
- [2] S. B. Cohn, "Microwave Bandpass Filters Containing High-Q Dielectric Resonators," IEEE Trans. Microwave Theory and Techniques, vol. MTT-16, pp. 218-227, April 1968.
- [3] M. W. Pospieszalski, "Cylindrical Dielectric Resonators and Their Applications in TEM Line Microwave Circuits," *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-27, pp. 233–238, March 1979.
- [4] J. E. Degenford and P. D. Coleman, "A Quasi-Optics Perturbation Technique for Measuring Dielectric Constants," Proc. IEEE, vol. 54, pp. 520–522, April 1966.
- [5] S. Ramo, J. R. Whinnery, and T. Van Duzer, Fields and Waves in Communication Electronics, John Wiley & Sons, N.Y., 1965.
- [6] R. E. Collin. Field Theory of Guided Waves, McGraw-Hill, N.Y., 1960.

PROBLEMS

6.1 Consider the loaded parallel resonant RLC circuit shown below. Compute the resonant frequency, unloaded Q, and loaded Q.



- 6.2 Derive an expression for the Q of a transmission line resonator consisting of a short-circuited transmission line 1λ long.
- 6.3 A transmission line resonator is fabricated from a $\lambda/4$ length of open-circuited line. Find the Q of this resonator if the complex propagation constant of the line is $\alpha + j\beta$.
- 6.4 Consider the resonator shown below, consisting of a $\lambda/2$ length of lossless transmission line shorted at both ends. At an arbitrary point z on the line, compute the impedances Z_L and Z_R seen looking to the left and to the right, and show that $Z_L = Z_R^*$. (This condition holds true for any lossless resonator and is the basis for the transverse resonance technique discussed in Section 3.9.)



6.5 A resonator is constructed from a 3.0 cm length of 100 Ω air-filled coaxial line, shorted at one end and terminated with a capacitor at the other end, as shown, (a) Determine the capacitor value to achieve the lowest-order resonance at 6.0 GHz. (b) Now assume that loss is introduced by placing a 10,000 Ω resistor in parallel with the capacitor. Calculate the Q.



6.6 A transmission line resonator is made from a length ℓ of lossless transmission line of characteristic impedance $Z_0 = 100 \ \Omega$. If the line is terminated at both ends as shown, find ℓ/λ for the first resonance, and the Q of this resonator.



- 6.7 Write the expressions for the \tilde{E} and \tilde{H} fields for a short-circuited $\lambda/2$ coaxial line resonator, and show that the time-average stored electric and magnetic energies are equal.
- **6.8** A series *RLC* resonant circuit is connected to a length of transmission line that is $\lambda/4$ long at its resonant frequency. Show that, in the vicinity of resonance, the input impedance behaves like that of a parallel *RLC* circuit.



- **6.9** An air-filled, silver-plated rectangular waveguide cavity has dimensions a = b = d = 5 cm. Find the resonant frequency and Q of the TE₁₀₁ and TE₁₀₂ modes.
- **6.10** Derive the Q for the TM₁₁₁ mode of a rectangular cavity, assuming lossy conducting walls and lossless dielectric.
- 6.11 Consider the rectangular cavity resonator shown on the next page, partially filled with dielectric. Derive a transcendental equation for the resonant frequency of the dominant mode by writing the fields in the air- and dielectric-filled regions in terms of TE_{10} waveguide modes, and enforcing boundary conditions at z = 0, d t, and d.



- **6.12** Determine the resonant frequencies of a rectangular cavity by carrying out a full separation of variables solution to the wave equation for E_z (for TM modes) and H_z (for TE modes), subject to the appropriate boundary conditions of the cavity. (Assume a solution of the form X(x)Y(y)Z(z).)
- 6.13 Find the Q for the TM_{nm0} resonant mode of a circular cavity. Consider both conductor and dielectric losses.
- **6.14** Design an air-filled circular cavity to operate in the TE₁₁₁ mode with the maximum Q at f = 7 GHz. If the cavity is silver-plated and air-filled, calculate the resulting Q.
- 6.15 An air-filled rectangular cavity resonator has its first three resonant modes at the frequencies 5.2 GHz, 6.5 GHz, and 7.2 GHz. Find the dimensions of the cavity.
- **6.16** Consider the microstrip ring resonator shown below. If the effective dielectric constant of the microstrip line is ϵ_c , find an equation for the frequency of the first resonance. Suggest some methods of coupling to this resonator.



6.17 A circular microstrip disk resonator is shown below. Solve the wave equation for TM_{nm0} modes for this structure, using the magnetic wall approximation that $H_{\phi} = 0$ at $\rho = a$. If fringing fields are neglected, show that the resonant frequency of the dominant mode is given by



- **6.18** Compute the resonant frequency of a cylindrical dielectric resonator with $\epsilon_r = 36.2$, 2a = 7.99 mm, and L = 2.14 mm.
- **6.19** Extend the analysis of Section 6.5 to derive a transcendental equation for the resonant frequency of the next resonant mode of the cylindrical dielectric resonator. $(H_z \text{ odd in } z.)$
- 6.20 Consider the rectangular dielectric resonator shown below. Assume a magnetic wall boundary condition around the edges of the cavity, and allow for evanescent fields in the $\pm z$ directions away from the dielectric, similar to the analysis of Section 6.5. Derive a transcendental equation for the resonant frequency.



- 6.21 A Fabry-Perot resonator is constructed with two large flat copper plates spaced 4 cm apart. If the operating frequency is 94 GHz, find the mode number and Q.
- **6.22** A parallel *RLC* circuit, with $R = 1000 \Omega$, L = 1.26 nH. C = 0.804 pF, is coupled with a series capacitor, C_0 , to a 50 Ω transmission line, as shown below. Determine C_0 for critical coupling to the line. What is the resonant frequency?



- **6.23** An aperture coupled rectangular waveguide cavity has a resonant frequency of 9.0 GHz and a Q of 11.000. If the waveguide dimensions are $\alpha = 2.5$ cm, b = 1.25 cm, find the normalized aperture reactance required for critical coupling.
- **6.24** At frequencies of 8.220 and 8.245 GHz, the power absorbed by a certain resonator is exactly one-half of the power absorbed by the resonator at resonance. If the reflection coefficient at resonance is 0.33, find the resonant frequency, coupling coefficient, and the unloaded and loaded Qs of the resonator.
- **6.25** A two-port transmission resonator is modeled with the equivalent circuit shown on the following Page. If ω_0 and Q are the resonant frequency and Q of the unloaded resonator, and g is the coupling coefficient to either transmission line, derive an expression for the ratio of transmitted to incident power, P_t/P_t , and sketch P_t/P_t , versus g, at resonance.



6.26 A thin slab of magnetic material is inserted next to the z = 0 wall of the rectangular cavity shown below. If the cavity is operating in the TE₁₀₁ mode, derive a perturbational expression for the change in resonant frequency caused by the magnetic material.



6.27 Derive an expression for the change in resonant frequency for the screw-tuned rectangular cavity of Example 6.8 if the screw is located at x = q/2, z = 0, where H_x is maximum and E_y is minimum.

Power Dividers and Directional Couplers

Power dividers and directional couplers are passive microwave components used for power division or power combining, as illustrated in Figure 7.1. In power division, an input signal is divided by the coupler into two (or more) signals of lesser power. The coupler may be a three-port component as shown, with or without loss, or may be a four-port component. Three-port networks take the form of T-junctions and other power dividers, while four-port networks take the form of directional couplers and hybrids. Power dividers are often of the equal-division (3 dB) type, but unequal power division ratios are also possible. Directional couplers can be designed for arbitrary power division, while hybrid junctions usually have equal power division. Hybrid junctions have either a 90° (quadrature) or a 180° (magic-T) phase shift between the outport ports.

A wide variety of waveguide couplers and power dividers were invented and characterized at the MIT Radiation Laboratory in the 1940s. These included E- and H-plane waveguide tee junctions, the Bethe hole coupler, multihole directional couplers, the Schwinger coupler, the waveguide magic-T, and various types of couplers using coaxial probes. In the mid-1950s through the 1960s, many of these couplers were reinvented to use stripline or microstrip technology. The increasing use of planar lines also led to the development of new types of couplers and dividers, such as the Wilkinson divider, the branch line hybrid, and the coupled line directional coupler.

We will first discuss some of the general properties of three- and four-port networks, and then treat the analysis and design of several of the most common types of dividers, couplers, and hybrids.

7.1 BASIC PROPERTIES OF DIVIDERS AND COUPLERS

In this section we will use the scattering matrix theory of Section 4.3 to derive some basic properties of three- and four-port networks. We will also define the terms isolation, coupling, and directivity, which are useful quantities for the characterization of couplers and hybrids.

Three-Port Networks (T-Junctions)

The simplest type of power divider is a T-junction, which is a three-port network with two inputs and one output. The scattering matrix of an arbitrary three-port network



FIGURE 7.1 Power division and combining. (a) Power division. (b) Power combining.

has nine independent elements:

$$[S] = \begin{bmatrix} S_{11} & S_{12} & S_{13} \\ S_{21} & S_{22} & S_{23} \\ S_{31} & S_{32} & S_{33} \end{bmatrix}.$$
 7.1

If the component is passive and contains no anisotropic materials, then it must be reciprocal and its [S] matrix must be symmetric $(S_{ij} = S_{ji})$. Usually, to avoid power loss, we would like to have a junction that is lossless and matched at all ports. We can easily show, however, that it is impossible to construct such a three-port lossless reciprocal network that is matched at all ports.

If all ports are matched, then $S_m = 0$, and if the network is reciprocal the scattering matrix of (7.1) reduces to

$$[S] = \begin{bmatrix} 0 & S_{12} & S_{13} \\ S_{12} & 0 & S_{23} \\ S_{13} & S_{23} & 0 \end{bmatrix}.$$
 7.2

Now if the network is also lossless, then energy conservation (4.53) requires that the scattering matrix be unitary, which leads to the following conditions [1], [2]:

$$|S_{12}|^2 + |S_{13}|^2 = 1. 7.3a$$

$$|S_{12}|^2 + |S_{23}|^2 = 1, 7.3b$$

$$|S_{13}|^2 + |S_{23}|^2 = 1, 7.3c$$

$$S_{13}^* S_{23} = 0, 7.3d$$

$$S_{23}^*S_{12} = 0.$$
 7.3e

$$S_{12}^* S_{13} = 0. 7.3f$$

Equations (7.3d-f) shows that at least two of the three parameters (S_{12} , S_{13} , S_{23}) must be zero. But this condition will always be inconsistent with one of equations (7.3a-c), implying that a three-port network cannot be lossless, reciprocal, and matched at all ports. If any one of these three conditions is relaxed, then a physically realizable device is possible.

If the three-port network is nonreciprocal, then $S_{ij} \neq S_{ji}$, and the conditions of input matching at all ports and energy conservation can be satisfied. Such a device is known as a circulator [1], and generally relies on an anisotropic material, such as ferrite, to achieve nonreciprocal behavior. Circulators will be discussed in more detail in Chapter 9, but we can demonstrate here that any matched lossless three-port network must be nonreciprocal and, thus, a circulator. The [S] matrix of a matched three-port network has the following form:

$$[S] = \begin{bmatrix} 0 & S_{12} & S_{13} \\ S_{21} & 0 & S_{23} \\ S_{31} & S_{32} & 0 \end{bmatrix}.$$
 7.4

Then if the network is lossless, [S] must be unitary, which implies the following:

$$S_{31}^* S_{32} = 0, 7.5a$$

$$S_{21}^* S_{23} = 0, 7.5b$$

$$S_{12}^*S_{13} = 0,$$
 7.5c

$$|S_{12}|^2 + |S_{13}|^2 = 1, 7.5d$$

$$|S_{21}|^2 + |S_{23}|^2 = 1, 7.5e$$

$$|S_{31}|^2 + |S_{32}|^2 = 1. 7.5f$$

These equations can be satisfied in one of two ways. Either

$$S_{12} = S_{23} = S_{31} = 0,$$
 $|S_{21}| = |S_{32}| = |S_{13}| = 1,$ 7.6a

$$S_{21} = S_{32} = S_{13} = 0.$$
 $|S_{12}| = |S_{23}| = |S_{31}| = 1.$ 7.6b

or

This result shows that $S_{ij} \neq S_{ji}$ for $i \neq j$, which implies that the device must be nonreciprocal. The [S] matrices for the two solutions of (7.6) are shown in Figure 7.2, together with the symbols for the two possible types of circulators. The only difference is in the direction of power flow between the ports. Thus, solution (7.6a) corresponds to a circulator that allows power flow only from port 1 to 2, or port 2 to 3, or port 3 to 1, while solution (7.6b) corresponds to a circulator with the opposite direction of power flow.

Alternatively, a lossless and reciprocal three-port network can be physically realized if only two of its ports are matched [1]. If ports 1 and 2 are these matched ports, then the [S] matrix can be written as

$$[S] = \begin{bmatrix} 0 & S_{12} & S_{13} \\ S_{12} & 0 & S_{23} \\ S_{13} & S_{23} & S_{33} \end{bmatrix}.$$
 7.7







FIGURE 7.3 A reciprocal, lossless three-port network matched at ports 1 and 2.

To be lossless, the following unitarity conditions must be satisfied:

$$S_{13}^* S_{23} = 0, 7.8a$$

$$S_{12}^*S_{13} + S_{23}^*S_{33} = 0, 7.8b$$

$$S_{23}^*S_{12} + S_{33}^*S_{13} = 0,$$
 7.8c

$$|S_{12}|^2 + |S_{13}|^2 = 1. 7.8d$$

$$|S_{12}|^2 + |S_{23}|^2 = 1, 7.8e$$

$$|S_{13}|^2 + |S_{23}|^2 + |S_{33}|^2 = 1.$$
 7.8°

Equations (7.8d-e) show that $|S_{13}| = |S_{23}|$, so (7.8a) leads to the result that $S_{13} = S_{23} = 0$. Then, $|S_{12}| = |S_{33}| = 1$. The scattering matrix and corresponding signal flow graph for this network are shown in Figure 7.3. where it is seen that the network actually consists of two separate components, one a matched two-port line and the other a totally mismatched one-port.

Finally, if the three-port network is allowed to be lossy, it can be reciprocal and matched at all ports: this is the case of the resistive divider, which will be discussed in Section 7.2. In addition, a lossy three-port can be made to have isolation between its output ports (for example, $S_{23} = S_{32} = 0$).

Four-Port Networks (Directional Couplers)

The [S] matrix of a reciprocal four-port network matched at all ports has the following form:

$$[S] = \begin{bmatrix} 0 & S_{12} & S_{13} & S_{14} \\ S_{12} & 0 & S_{23} & S_{24} \\ S_{13} & S_{23} & 0 & S_{34} \\ S_{14} & S_{24} & S_{34} & 0 \end{bmatrix}.$$
7.9

If the network is lossless, 10 equations result from the unitarity, or energy conservation, condition [1]. [2]. Let us consider the multiplication of row 1 and row 2, and the

multiplication of row 4 and row 3:

$$S_{13}^*S_{23} + S_{14}^*S_{24} = 0, 7.10a$$

$$S_{14}^* S_{13} + S_{24}^* S_{23} = 0. 7.10b$$

Now multiply (7.10a) by S_{24}^* and (7.10b) by S_{13}^* , and subtract to obtain

$$S_{14}^*(|S_{13}|^2 - |S_{24}|^2) = 0. 7.11$$

Similarly, the multiplication of row 1 and row 3, and the multiplication of row 4 and row 2, gives

$$S_{12}^*S_{23} + S_{14}^*S_{34} = 0, 7.12a$$

$$S_{14}^* S_{12} + S_{34}^* S_{23} = 0. 7.12b$$

Now multiply (7.12a) by S_{12} and (7.12b) by S_{34} , and subtract to obtain

$$S_{23}(|S_{12}|^2 - |S_{34}|^2) = 0. 7.13$$

One way for (7.11) and (7.13) to be satisfied is if $S_{14} = S_{23} = 0$, which results in a directional coupler. Then the self-products of the rows of the unitary [S] matrix of (7.9) yield the following equations:

$$|S_{12}|^2 + |S_{13}|^2 = 1, 7.14a$$

$$|S_{12}|^2 + |S_{24}|^2 = 1, 7.14b$$

$$|S_{13}|^2 + |S_{34}|^2 = 1, 7.14c$$

$$|S_{24}|^2 + |S_{34}|^2 = 1, 7.14d$$

which imply that $|S_{13}| = |S_{24}|$ (using 7.14a and 7.14b), and that $|S_{12}| = |S_{34}|$ (using 7.14b and 7.14d).

Further simplification can be made by choosing the phase references on three of the four ports. Thus, we choose $S_{12} = S_{34} = \alpha$. $S_{13} = \beta e^{j\theta}$, and $S_{24} = \beta e^{j\phi}$, where α and β are real, and θ and ϕ are phase constants to be determined (one of which we are still free to choose). The dot product of rows 2 and 3 gives

$$S_{12}^*S_{13} + S_{24}^*S_{34} = 0, 7.15$$

which yields a relation between the remaining phase constants as

$$\theta + \phi = \pi \pm 2n\pi. \tag{7.16}$$

If we ignore integer multiples of 2π , there are two particular choices that commonly occur in practice:

1. The Symmetrical Coupler: $\theta = \phi = \pi/2$. The phases of the terms having amplitude β are chosen equal. Then the scattering matrix has the following

form:

$$[S] = \begin{bmatrix} 0 & \alpha & j\beta & 0\\ \alpha & 0 & 0 & j\beta\\ j\beta & 0 & 0 & \alpha\\ 0 & j\beta & \alpha & 0 \end{bmatrix}.$$
 7.17

2. The Antisymmetrical Coupler: $\theta = 0, \phi = \pi$. The phases of the terms having amplitude β are chosen to be 180° apart. Then the scattering matrix has the following form:

$$[S] = \begin{bmatrix} 0 & \alpha & \beta & 0 \\ \alpha & 0 & 0 & -\beta \\ \beta & 0 & 0 & \alpha \\ 0 & -\beta & \alpha & 0 \end{bmatrix},$$
 7.18

Note that the two couplers differ only in the choice of reference planes. Also, the amplitudes α and β are not independent, as (7.14a) requires that

$$\alpha^2 + \beta^2 = 1. \tag{7.19}$$

Thus, apart from phase references, an ideal directional coupler has only one degree of freedom.

Another way for (7.11) and (7.13) to be satisfied is if $|S_{13}| = |S_{24}|$ and $|S_{12}| = |S_{34}|$. If we choose phase references, however, such that $S_{13} = S_{24} = \alpha$, and $S_{12} = S_{34} = j\beta$ (which satisfies (7.16)), then (7.10a) yields $\alpha(S_{23} + S_{14}^*) = 0$, and (7.12a) yields $\beta(S_{14}^* - S_{23}) = 0$. These two equations have two possible solutions. First, $S_{14} = S_{23} = 0$, which is the same as the above solution for the directional coupler. The other solution occurs for $\alpha = \beta = 0$, which implies that $S_{12} = S_{13} = S_{24} = S_{34} = 0$. This is the case of two decoupled two-port networks (between ports 1 and 4, and ports 2 and 3), which is of trivial interest and will not be considered further. We are thus left with the conclusion that any reciprocal, lossless, matched four-port network is a directional coupler.

The basic operation of a directional coupler can then be illustrated with the aid of Figure 7.4, which shows two commonly used symbols for a directional coupler and the port definitions. Power supplied to port 1 is coupled to port 3 (the coupled port) with the coupling factor $|S_{13}|^2 = \beta^2$, while the remainder of the input power is delivered to port 2 (the through port) with the coefficient $|S_{12}|^2 = \alpha^2 = 1 - \beta^2$. In an ideal directional coupler, no power is delivered to port 4 (the isolated port).

The following three quantities are generally used to characterize a directional coupler:

Coupling =
$$C = 10 \log \frac{P_1}{P_3} = -20 \log \beta \, dB$$
, 7.20*a*

Directivity =
$$D = 10 \log \frac{P_3}{P_4} = 20 \log \frac{\beta}{|S_{14}|} d\mathbf{B}$$
, 7.20b

Isolation =
$$I = 10 \log \frac{P_1}{P_4} = -20 \log |S_{14}| \, dB.$$
 7.20c

The coupling factor indicates the fraction of the input power that is coupled to the output port. The directivity is a measure of the coupler's ability to isolate forward and backward



FIGURE 7.4 Two commonly used symbols for directional couplers, and power flow conventions.

waves, as is the isolation. These quantities are then related as

$$I = D + C \, \mathrm{d}\mathbf{B}. \tag{7.21}$$

The ideal coupler would have infinite directivity and isolation ($S_{14} = 0$). Then both α and β could be determined from the coupling factor, C.

Hybrid couplers are special cases of directional couplers, where the coupling factor is 3 dB, which implies that $\alpha = \beta = 1/\sqrt{2}$. There are two types of hybrids. The quadrature hybrid has a 90° phase shift between ports 2 and 3 ($\theta = \phi = \pi/2$) when fed at port 1, and is an example of a symmetrical coupler. Its [S] matrix has the following form:

$$[S] = \frac{1}{\sqrt{2}} \begin{bmatrix} 0 & 1 & j & 0 \\ 1 & 0 & 0 & j \\ j & 0 & 0 & 1 \\ 0 & j & 1 & 0 \end{bmatrix}.$$
 7.22

The magic-T hybrid or rat-race hybrid has a 180° phase difference between ports 2 and 3 when fed at port 4, and is an example of an antisymmetrical coupler. Its [S] matrix has the following form:

$$[S] = \frac{1}{\sqrt{2}} \begin{bmatrix} 0 & 1 & 1 & 0 \\ 1 & 0 & 0 & -1 \\ 1 & 0 & 0 & 1 \\ 0 & -1 & 1 & 0 \end{bmatrix}.$$
 7.23



POINT OF INTEREST: Measuring Coupler Directivity

The directivity of a directional coupler is a measure of the coupler's ability to separate forward and reverse wave components, so applications of directional couplers often require high (35 dB or greater) directivity. Poor directivity will limit the accuracy of a reflectometer, and can cause variations in the coupled power level from a coupler when there is even a small mismatch on the through line. The directivity of a coupler generally cannot be measured directly because it involves a lowlevel signal that can be masked by coupled power from a reflected wave on the through arm. For example, if a coupler has C = 20 dB and D = 35 dB, with a load having RL = 30 dB, the signal level through the directivity path will be D + C = 55 dB below the input power, but the reflected power through the coupled arm will only be RL + C = 50 dB below the input power.

One way to measure coupler directivity uses a sliding matched load, as follows. First, the coupler is connected to a source and matched load, as shown in the left-hand figure below, and the coupled output power is measured. If we assume an input power P_i , this power will be $P_c = C^2 P_i$, where $C = 10^{(-C \, dB)/20}$ is the numerical voltage coupling factor of the coupler. Now reverse the position of the coupler as shown in the right-hand figure below and terminate the through line with a sliding load.



Changing the position of the sliding load introduces a variable phase shift in the signal reflected from the load and coupled to the output port. Thus the voltage at the output port can be written as

$$V_0 = V_i \left(\frac{C}{D} + C |\Gamma| e^{-j\theta}\right),$$

where V_i is the input voltage, $D = 10^{(D \text{ dB})/20} \ge 1$ is the numerical value of the directivity, $|\Gamma|$ is the reflection coefficient magnitude of the load, and θ is the path length difference between the directivity and reflected signals. Moving the sliding load changes θ , so the two signals will combine to trace out a circular locus, as shown in the following figure.



The minimum and maximum output powers are given by

$$P_{\max} = P_v \left(\frac{C}{D} - C|\Gamma|\right)^2, \qquad P_{\max} = P_i \left(\frac{C}{D} + C|\Gamma|\right)^2.$$

Now let M and m be defined in terms of these powers as follows:

$$M = \frac{P_c}{P_{\text{max}}} = \left(\frac{D}{1 + |\Gamma|D}\right)^2, \qquad m = \frac{P_{\text{max}}}{P_{\text{max}}} = \left(\frac{1 + |\Gamma|D}{1 - |\Gamma|D}\right)^2.$$

These ratios can be accurately measured directly by using a variable attenuator between the source and coupler. The directivity (numerical) can then be found as

$$D = M\left(\frac{2m}{m+1}\right).$$

This method requires that $|\Gamma| < 1/D$ or, in dB, RL > D.

Reference: M. Sucher and J. Fox. editors. Handbook of Microwave Measurements, third edition, volume II, Polytechnic Press, New York, 1963.

7.2 THE T-JUNCTION POWER DIVIDER

The T-junction power divider is a simple three-port network that can be used for power division or power combining, and can be implemented in virtually any type of transmission line medium. Figure 7.5 shows some commonly used T-junctions in waveguide and microstrip or stripline form. The junctions shown here are, in the absence of transmission line loss, lossless junctions. Thus, as discussed in the preceding section, such junctions cannot be matched simultaneously at all ports. We will treat such junctions below, followed by a discussion of the resistive divider, which can be matched at all ports but is not lossless.



FIGURE 7.5 Various T-junction power dividers. (a) E plane waveguide T. (b) H plane waveguide T. (c) Microstrip T-junction.



FIGURE 7.6 Transmission line model of a lossless T-junction.

Lossless Divider

The lossless T-junctions of Figure 7.5 can all be modelled as a junction of three transmission lines, as shown in Figure 7.6 [3]. In general, there are fringing fields and higher order modes associated with the discontinuity at such a junction, leading to stored energy that can be accounted for by a lumped susceptance, B. In order for the divider to be matched to the input line of characteristic impedance Z_0 , we must have

$$Y_{\rm in} = jB + \frac{1}{Z_1} + \frac{1}{Z_2} = \frac{1}{Z_0}.$$
 7.24

If the transmission lines are assumed to be lossless (or of low loss), then the characteristic impedances are real. If we also assume B = 0, then (7.24) reduces to

$$\frac{1}{Z_1} + \frac{1}{Z_2} = \frac{1}{Z_0}.$$
 7.25

In practice, if B is not negligible, some type of reactive tuning element can usually be added to the divider to cancel this susceptance, at least over a narrow frequency range.

The output line impedances Z_1 and Z_2 can then be selected to provide various power division ratios. Thus, for a 50 Ω input line, a 3 dB (equal split) power divider can be made by using two 100 Ω output lines. If necessary, quarter-wave transformers can be used to bring the output line impedances back to the desired levels. If the output lines are matched, then the input line will be matched, but there will be no isolation between the two output ports, and there will be a mismatch looking into the output ports.

EXAMPLE 7.1 The T-junction Power Divider

A lossless T-junction power divider has a source impedance of 50Ω . Find the output characteristic impedances so that the input power is divided in a 2:1 ratio. Compute the reflection coefficients seen looking into the output ports.

Solution

If the voltage at the junction is V_0 , as shown in Figure 7.6, the input power to the matched divider is

$$P_{\rm in} = \frac{1}{2} \frac{V_0^2}{Z_0}.$$

while the output powers are

$$P_{1} = \frac{1}{2} \frac{V_{0}^{2}}{Z_{1}} = \frac{1}{3} P_{\text{in}},$$
$$P_{2} = \frac{1}{2} \frac{V_{0}^{2}}{Z_{2}} = \frac{2}{3} P_{\text{in}}.$$

These results yield the characteristic impedances as

$$Z_1 = 3Z_0 = 150 \,\Omega,$$

 $Z_2 = \frac{3Z_0}{2} = 75 \,\Omega,$

Then the input impedance to the junction is

$$Z_{\rm in} = 75 || 150 = 50 \,\Omega,$$

so that the input is matched to the 50 Ω source.

Looking into the 150 Ω output line, we see an impedance of 50||75 = 30 Ω , while at the 75 Ω output line we see an impedance of 50||150 = 37.5 Ω . Thus, the reflection coefficients seen looking into these ports are

$$\Gamma_1 = \frac{30 - 150}{30 + 150} = -0.666,$$

$$\Gamma_2 = \frac{37.5 - 75}{37.5 + 75} = -0.333.$$

Resistive Divider

If a three-port divider contains lossy components it can be made to be matched at all ports, although the two output ports may not be isolated [3]. The circuit for such a divider is illustrated in Figure 7.7, using lumped-element resistors. An equal-split (-3 dB) divider is shown, but unequal power division ratios are also possible.

The resistive divider of Figure 7.7 can easily be analyzed using circuit theory. Assuming that all ports are terminated in the characteristic impedance Z_0 , the impedance Z, seen looking into the $Z_0/3$ resistor followed by the output line, is

$$Z = \frac{Z_0}{3} + Z_0 = \frac{4Z_0}{3}.$$
 7.26

Then the input impedance of the divider is

$$Z_{\rm in} = \frac{Z_0}{3} + \frac{2Z_0}{3} = Z_0, \qquad 7.27$$



FIGURE 7.7 An equal-split three-port resistive power divider.

which shows that the input is matched to the feed line. Since the network is symmetric from all three ports, the output ports are also matched. Thus, $S_{11} = S_{22} = S_{33} = 0$.

If the voltage at port 1 is V_1 , then by voltage division the voltage V at the center of the junction is

$$V = V_1 \frac{2Z_0/3}{Z_0/3 + 2Z_0/3} = \frac{2}{3}V_1,$$
7.28

and the output voltages are, again by voltage division

$$V_2 = V_3 = V \frac{Z_0}{Z_0 + Z_0/3} = \frac{3}{4}V = \frac{1}{2}V_1.$$
 7.29

Thus, $S_{21} = S_{31} = S_{23} = 1/2$, which is -6 dB below the input power level. The network is reciprocal, so the scattering matrix is symmetric, and can be written as

$$[S] = \frac{1}{2} \begin{bmatrix} 0 & 1 & 1 \\ 1 & 0 & 1 \\ 1 & l & 0 \end{bmatrix}.$$
 7.30

The reader may verify that this is not a unitary matrix.

The power delivered to the input of the divider is

$$P_{\rm m} = \frac{1}{2} \frac{V_1^2}{Z_0}, \qquad 7.31$$

while the output powers are

$$P_2 = P_3 = \frac{1}{2} \frac{(1/2V_1)^2}{Z_0} = \frac{1}{8} \frac{V_1^2}{Z_0} = \frac{1}{4} P_{10}, \qquad 7.32$$

which shows that half of the supplied power is dissipated in the resistors.

7.3 THE WILKINSON POWER DIVIDER

The lossless T-junction divider suffers from the problem of not being matched at all ports and, in addition, does not have any isolation between output ports. The resistive divider can be matched at all ports, but even though it is not lossless, isolation is still not achieved. From the discussion in Section 7.1, however, we know that a lossy three-port network can be made having all ports matched with isolation between the output ports. The Wilkinson power divider [4] is such a network and is the subject of the present section. It has the useful property of being lossless when the output ports are matched; that is, only reflected power is dissipated.

The Wilkinson power divider can be made to give arbitrary power division, but we will first consider the equal-split (3 dB) case. This divider is often made in microstrip or stripline form, as depicted in Figure 7.8a; the corresponding transmission line circuit is given in Figure 7.8b. We will analyze this circuit by reducing it to two simpler circuits driven by symmetric and antisymmetric sources at the output ports. This "even-odd" mode analysis technique [5] will also be useful for other networks that we will analyze in later sections.

Even-Odd Mode Analysis

For simplicity, we can normalize all impedances to the characteristic impedance Z_0 , and redraw the circuit of Figure 7.8b with voltage generators at the output ports as shown in Figure 7.9. This network has been drawn in a form that is symmetric across the midplane; the two source resistors of normalized value 2 combine in parallel to give a resistor of normalized value 1, representing the impedance of a matched source. The quarter-wave lines have a normalized characteristic impedance Z, and the shunt resistor has a normalized value of r; we shall show that, for the equal-split power divider, these values should be $Z = \sqrt{2}$ and r = 2, as given in Figure 7.8.

Now we define two separate modes of excitation for the circuit of Figure 7.9: the even mode, where $V_{g2} = V_{g3} = 2$ V, and the odd mode, where $V_{g2} = -V_{g3} = 2$ V.



FIGURE 7.8 The Wilkinson power divider. (a) An equal-split Wilkinson power divider in microstrip form. (b) Equivalent transmission line circuit.



FIGURE 7.9 The Wilkinson power divider circuit in normalized and symmetric form.

Then by superposition of these two modes, we effectively have an excitation of $V_{g2} = 4 \text{ V}$, $V_{q3} = 0$, from which we can find the S parameters of the network. We now treat these two modes separately.

Even mode. For the even-mode excitation, $V_{g2} = V_{g3} = 2$ V, and so $V_2^e = V_3^e$ and there is no current flow through the r/2 resistors or the short circuit between the inputs of the two transmission lines at port 1. Thus we can bisect the network of Figure 7.9 with open circuits at these points to obtain the network of Figure 7.10a (the grounded side of the $\lambda/4$ line is not shown). Then, looking into port 2, we see an impedance

$$Z_{\rm m}^e = \frac{Z^2}{2},$$
 7.33

since the transmission line looks like a quarter-wave transformer. Thus, if $Z = \sqrt{2}$, port 2 will be matched for even mode excitation: then $V_2^e = V$ since $Z_{in}^e = 1$. The r/2 resistor is superfluous in this case, since one end is open-circuited. Next, we find V_1^e from the transmission line equations. If we let x = 0 at port 1 and $x = -\lambda/4$ at port 2, the voltage on the transmission line section can be written as

 $V(x) = V^+ (e^{-j\beta x} + \Gamma e^{j\beta x}).$

Then,

$$V_{2}^{e} = V(-\lambda/4) = jV^{+}(1-\Gamma) = V$$

$$V_{1}^{e} = V(0) = V^{+}(1+\Gamma) = jV\frac{\Gamma+1}{\Gamma-1}.$$
7.34

The reflection coefficient Γ is that seen at port 1, looking toward the resistor of normalized value 2, so

$$\Gamma = \frac{2 - \sqrt{2}}{2 + \sqrt{2}},$$

$$V_{t}^{e} = -jV\sqrt{2}$$
7.35

and



FIGURE 7.10 Bisection of the circuit of Figure 7.9. (a) For even-mode excitation. (b) For odd-mode excitation.

Odd mode. For the odd-mode excitation, $V_{g2} = -V_{g3} = 2$ V, and so $V_2^o = -V_3^o$, and there is a voltage null along the middle of the circuit in Figure 7.9. Thus, we can bisect this circuit by grounding it at two points on its midplane to give the network of Figure 7.10b. Looking into port 2, we see an impedance of r/2, since the parallelconnected transmission line is $\lambda/4$ long and shorted at port 1, and so looks like an open circuit at port 2. Thus, port 2 will be matched for odd mode excitation if we select r = 2. Then $V_2^o = V$ and $V_1^o = 0$; for this mode of excitation all power is delivered to the r/2 resistors, with none going to port 1.

Finally, we must find the input impedance at port 1 of the Winkinson divider when ports 2 and 3 are terminated in matched loads. The resulting circuit is shown in Figure 7.11a, where it is seen that this is similar to an even mode of excitation, since $V_2 = V_3$. Thus, no current flows through the resistor of normalized value 2, so it can be removed, leaving the circuit of Figure 7.11b. We now have the parallel connection of two quarterwave transformers terminated in loads of unity (normalized). The input impedance is then

$$Z_{\rm in} = \frac{1}{2} \left(\sqrt{2}\right)^2 = 1.$$
 7.36

In summary, we can establish the following S parameters for the Wilkinson divider:

$$\begin{split} S_{11} &= 0 & (Z_{in} = 1 \text{ at port } 1) \\ S_{22} &= S_{33} = 0 & (\text{ports 2 and 3 matched for even and odd modes}) \\ S_{12} &= S_{21} = \frac{V_1^* + V_1^o}{V_2^* + V_2^o} = -j/\sqrt{2} & (\text{symmetry due to reciprocity}) \\ S_{13} &= S_{31} = -j/\sqrt{2} & (\text{symmetry of ports 2 and 3}) \\ S_{23} &= S_{32} = 0 & (\text{due to short or open at bisection}) \end{split}$$





FIGURE 7.11 Analysis of the Wilkinson divider to find S_{11} . (a) The terminated Wilkinson divider. (b) Bisection of the circuit in (a).

The preceding formula for S_{12} applies because all ports are matched when terminated with matched loads. Note that when the divider is driven at port 1 and the outputs are matched, no power is dissipated in the resistor. Thus the divider is lossless when the outputs are matched; only reflected power from ports 2 or 3 is dissipated in the resistor. Since $S_{23} = S_{32} = 0$, ports 2 and 3 are isolated.



EXAMPLE 7.2 Design and Performance of a Wilkinson Divider

Design an equal-split Wilkinson power divider for a 50 Ω system impedance at frequency f_0 , and plot the return loss (S_{11}) , insertion loss $(S_{21} = S_{31})$, and isolation $(S_{23} = S_{32})$ versus frequency from $0.5f_0$ to $1.5f_0$.

Solution

From Figure 7.8 and the above derivation, we have that the quarter-wave transmission lines in the divider should have a characteristic impedance of

$$Z=\sqrt{2}Z_0=70.7\,\Omega,$$

and the shunt resistor a value of

$$R=2Z_0=100\,\Omega.$$



FIGURE 7.12 Frequency response of an equal-split Wilkinson power divider. Port 1 is the input port: ports 2 and 3 are the output ports.

The transmission lines are $\lambda/4$ long at the frequency f_0 . Using a computeraided design program for the analysis of microwave circuits, the S parameter magnitudes were calculated and plotted in Figure 7.12.

Unequal Power Division and N-Way Wilkinson Dividers

Wilkinson-type power dividers can also be made with unequal power splits: a microstrip version is shown in Figure 7.13. If the power ratio between ports 2 and 3 is $K^2 = P_3/P_2$, then the following design equations apply:

$$Z_{03} = Z_0 \sqrt{\frac{1+K^2}{K^3}},$$
 7.37a

$$Z_{02} = K^2 Z_{03} = Z_0 \sqrt{K(1+K^2)},$$
7.37b

$$R = Z_0 \left(K + \frac{1}{K} \right).$$
 7.37c



FIGURE 7.13 A Wilkinson power divider in microstrip form having unequal power division.



FIGURE 7.14 An N-way, equal-split Wilkinson power divider.

Note that the above results reduce to the equal-split case for K = 1. Also observe that the output lines are matched to the impedances $R_2 = Z_0 K$ and $R_3 = Z_0 / K$, as opposed to the impedance Z_0 ; matching transformers can be used to transform these output impedances.

The Wilkinson divider can also be generalized to an N-way divider or combiner [4], as shown in Figure 7.14. This circuit can be matched at all ports, with isolation between all ports. A disadvantage, however, is the fact that the divider requires crossovers for the resistors for $N \ge 3$. This makes fabrication difficult in planar form. The Wilkinson divider can also be made with stepped multiple sections, for increased bandwidth. A photograph of a Wilkinson divider network is shown in Figure 7.15.

7.4 ____ WAVEGUIDE DIRECTIONAL COUPLERS

We now turn our attention to directional couplers, which are four-port devices with the characteristics discussed in Section 7.1. To review the basic operation, consider the directional coupler schematic symbols shown in Figure 7.4. Power incident at port 1 will couple to port 2 (the through port) and to port 3 (the coupled port), but not to port 4 (the isolated port). Similarly, power incident in port 2 will couple to ports 1 and 4, but not 3. Thus, ports 1 and 4 are decoupled, as are ports 2 and 3. The fraction of power coupled from port 1 to port 3 is given by C, the coupling, as defined in (7.20a), and the leakage of power from port 1 to port 4 is given by I, the isolation, as defined in (7.20c). Another quantity that can be used to characterize a coupler is the directivity. D = 1 - C (dB), which is the ratio of the power delivered to the coupling factor, as the isolated port. The ideal coupler is characterized solely by the coupling factor, as the isolation and directivity are infinite. The ideal coupler is also lossless and matched at all ports.

Directional couplers can be made in many different forms. We will first discuss waveguide couplers, followed by hybrid junctions. A hybrid junction is a special case of a directional coupler, where the coupling factor is 3 dB (equal split), and the phase relation between the output ports is either 90° (quadrature hybrid), or 180° (magic-T



FIGURE 7.15 Photograph of a four-way corporate power divider network using three microstrip Wilkinson power dividers. Note the isolation chip resistors. Courtesy of M. D. Abouzahra, MIT Lincoln Laboratory, Lexington, Mass.

or *rat-race hybrid*). Then we will discuss the implementation of directional couplers in coupled transmission line form.

Bethe Hole Coupler

The directional property of all directional couplers is produced through the use of two separate waves or wave components, which add in phase at the coupled port and are canceled at the isolated port. One of the simplest ways of doing this is to couple one waveguide to another through a single small hole in the common broad wall between the two guides. Such a coupler is known as a Bethe hole coupler, two versions of which are shown in Figure 7.16. From the small-aperture coupling theory of Section 4.8, we know that an aperture can be replaced with equivalent sources consisting of electric and magnetic dipole moments [6]. The normal electric dipole moment and the axial magnetic dipole moment radiate with even symmetry in the coupled guide, while the transverse magnetic dipole moment radiates with odd symmetry. Thus, by adjusting the relative amplitudes of these two equivalent sources, we can cancel the radiation in the direction of the isolated port, while enhancing the radiation in the direction of the coupled port. Figure 7.16 shows two ways in which these wave amplitudes can be controlled; in the coupler shown in Figure 7.16a, the two guides are parallel and the coupling is controlled by s, the aperture offset from the sidewall of the guide. For the coupler of Figure 7.16b, the wave amplitudes are controlled by the angle, θ , between the two guides.



FIGURE 7.16 Two versions of the Bethe hole directional coupler. (a) Parallel guides. (b) Skewed guides.

First consider the configuration of Figure 7.16a, with an incident TE_{10} mode into port 1. These fields can be written as

$$E_y = A \sin \frac{\pi x}{a} e^{-j\beta z}, \qquad 7.38a$$

$$H_x = \frac{-A}{Z_{10}} \sin \frac{\pi x}{a} e^{-j\beta z}, \qquad 7.38b$$

$$H_z = \frac{j\pi A}{\beta a Z_{10}} \cos \frac{\pi x}{a} e^{-j\beta z}, \qquad 7.38c$$

where $Z_{10} = k_0 \eta_0 / \beta$ is the wave impedance of the TE₁₀ mode. Then, from (4.124) and (4.125), this incident wave generates the following equivalent polarization currents at the aperture at x = s, y = b, z = 0:

$$\bar{P}_{e} = \epsilon_{0} \alpha_{e} \hat{y} A \sin \frac{\pi s}{a} \delta(x - s) \delta(y - b) \delta(z).$$
7.39a

$$\bar{P}_m = -\alpha_m A \left[\frac{-\hat{x}}{Z_{10}} \sin \frac{\pi s}{a} + \hat{z} \frac{j\pi}{\beta a Z_{10}} \cos \frac{\pi s}{a} \right] \delta(x-s) \delta(y-b) \delta(z).$$
 7.39b

Using (4.128a,b) to relate \bar{P}_e and \bar{P}_m to the currents \hat{J} and \bar{M} , and then using (4.118), (4.120), (4.122), and (4.123) gives the amplitudes of the forward and reverse traveling

waves in the bottom guide as

$$\begin{aligned} A_{10}^{+} &= \frac{-1}{P_{10}} \int_{v} \bar{E}_{10}^{-} \cdot \bar{J} \, dv + \frac{1}{P_{10}} \int_{v} \bar{H}_{10}^{-} \cdot \bar{M} \, dv \\ &= \frac{-j\omega A}{P_{10}} \left[\epsilon_{0} \alpha_{e} \sin^{2} \frac{\pi s}{a} - \frac{\mu_{0} \alpha_{m}}{Z_{10}^{2}} \left(\sin^{2} \frac{\pi s}{a} + \frac{\pi^{2}}{\beta^{2} a^{2}} \cos^{2} \frac{\pi s}{a} \right) \right], \quad 7.40a \end{aligned}$$

$$\begin{aligned} A_{10}^{-} &= \frac{-1}{P_{10}} \int_{v} \bar{E}_{10}^{+} \cdot \bar{J} \, dv + \frac{1}{P_{10}} \int_{v} \bar{H}_{10}^{+} \cdot \bar{M} \, dv \\ &= \frac{-j\omega A}{P_{10}} \left[\epsilon_{0} \alpha_{e} \sin^{2} \frac{\pi s}{a} + \frac{\mu_{0} \alpha_{m}}{Z_{10}^{2}} \left(\sin^{2} \frac{\pi s}{a} - \frac{\pi^{2}}{\beta^{2} a^{2}} \cos^{2} \frac{\pi s}{a} \right) \right]. \end{aligned}$$

where $P_{10} = ab/Z_{10}$ is the power normalization constant. Note from (7.40a,b) that the amplitude of the wave excited toward port 4 (A_{10}^+) is generally different from that excited toward port 3 (A_{10}^-) (because $H_x^+ = -H_x^-)$ so we can cancel the power delivered to port 4 by setting $A_{10}^r = 0$. If we assume that the aperture is round, then Table 4.3 gives the polarizabilities as $\alpha_e = 2r_0^3/3$ and $\alpha_m = 4r_0^3/3$, where r_0 is the radius of the aperture. Then from (7.40a) we obtain the following condition:

$$\begin{pmatrix} 2\epsilon_0 - \frac{4\mu_0}{Z_{10}^2} \end{pmatrix} \sin^2 \frac{\pi s}{a} - \frac{4\pi^2 \mu_0}{\beta^2 a^2 Z_{10}^2} \cos^2 \frac{\pi s}{a} = 0, \\ (k_0^2 - 2\beta^2) \sin^2 \frac{\pi s}{a} = \frac{2\pi^2}{a^2} \cos^2 \frac{\pi s}{a}, \\ \left(\frac{4\pi^2}{a^2} - k_0^2\right) \sin^2 \frac{\pi s}{a} = \frac{2\pi^2}{a^2}, \\ \sin \frac{\pi s}{a} = \pi \sqrt{\frac{2}{4\pi^2 - k_0^2 a^2}} = \frac{\lambda_0}{\sqrt{2(\lambda_0^2 - a^2)}}.$$
 7.41

٥r

The coupling factor is then given by

$$C = 20 \log \left| \frac{A}{A_{10}^-} \right| \, \mathrm{dB}$$
 7.42a

and the directivity by

$$D = 20 \log \left| \frac{A_{10}^-}{A_{10}^+} \right| \, \mathrm{dB}.$$
 7.42b

Thus, a Bethe hole coupler of the type shown in Figure 7.16a can be designed by first using (7.41) to find s, the position of the aperture, and then using (7.42a) to determine the aperture size, r_0 , to give the required coupling factor.

Chapter 7: Power Dividers and Directional Couplers

For the skewed geometry of Figure 7.16b, the aperture may be centered at s = a/2, and the skew angle θ adjusted for cancellation at port 4. In this case, the normal electric field does not change with θ , but the transverse magnetic field components are reduced by $\cos \theta$. We can thus account for the skew by replacing α_m in the previous derivation by $\alpha_m \cos \theta$. The wave amplitudes of (7.40a,b) then become, for s = a/2,

$$A_{10}^{+} = \frac{-j\omega A}{P_{10}} (\epsilon_0 \alpha_e - \frac{\mu_0 \alpha_m}{Z_{10}^2} \cos \theta), \qquad 7.43a$$

$$A_{10}^{-} = \frac{-j\omega A}{P_{10}} (\epsilon_0 \alpha_e + \frac{\mu_0 \alpha_m}{Z_{10}^2} \cos \theta).$$
 7.43b

Setting $A_{10}^+ = 0$ results in the following condition for the angle θ :

$$2\epsilon_0 - \frac{4\mu_0}{Z_{10}^2}\cos\theta = 0,$$

$$\cos\theta = \frac{k_0^2}{2\beta^2}.$$
7.44

or

The coupling factor then simplifies to

$$C = 20 \log \left| \frac{A}{A_{10}^{-}} \right| = -20 \log \frac{4k_0^2 r_0^3}{3ab\beta} \, \mathrm{dB}.$$
 7.45

The geometry of the skewed Bethe hole coupler is often a disadvantage in terms of fabrication and application. Also, both coupler designs operate properly only at the design frequency: deviation from this frequency will alter the coupling level and the directivity, as shown in the following example.



EXAMPLE 7.3 Bethe Hole Coupler Design and Performance

Design a Bethe hole coupler of the type shown in Figure 7.16a for X-band waveguide operating at 9 GHz, with a coupling of 20 dB. Calculate and plot the coupling and directivity from 7 to 11 GHz. Assume a round aperture.

Solution

For X-band waveguide at 9 GHz, we have the following constants:

a = 0.02286 m, b = 0.01016 m, $\lambda_0 = 0.0333 \text{ m},$ $k_0 = 188.5 \text{ m}^{-1},$ $\beta = 129.0 \text{ m}^{-1},$ $Z_{10} = 550.9 \Omega,$ $P_{10} = 4.22 \times 10^{-7} \text{ m}^2/\Omega.$ Then (7.41) can be used to find the aperture position s:

$$\sin \frac{\pi s}{a} = \frac{\lambda_0}{\sqrt{2(\lambda_0^2 - a^2)}} = 0.972,$$
$$s = \frac{a}{\pi} \sin^{-1} 0.972 = 0.424a = 9.69 \text{ mm}.$$

The coupling is 20 dB, so

$$C = 20 \text{ dB} = 20 \log \left| \frac{A}{A_{10}^{-}} \right|.$$
$$\left| \frac{A}{A_{10}^{-}} \right| = 10^{20/20} = 10,$$

Oľ

or

thus, $|A_{10}^{-}/A| = 1/10$. We now use (7.40b) to find r_0 :

$$\left|\frac{A_{10}^{-}}{A}\right| = \frac{1}{10} = \frac{\omega}{P_{10}} \left[\left(\epsilon_0 \alpha_e + \frac{\mu_0 \alpha_m}{Z_{10}^2} \right) (0.944) - \frac{\pi^2 \mu_0 \alpha_m}{\beta^2 a^2 Z_{10}^2} (0.056) \right].$$

Since $\alpha_e = 2r_0^3/3$ and $\alpha_m = 4r_0^3/3$, we obtain,

$$0.1 = 1.44 \times 10^6 r_0^3,$$

 $r_0 = 4.15 \text{ mm}$

This completes the design of the Bethe hole coupler. To evaluate the coupling and directivity versus frequency, we evaluate (7.42a) and (7.42b), using the expressions for A_{10}^- and A_{10}^+ given in (7.40a) and (7.40b). In these expressions, the aperture position and size are fixed at s = 9.69 mm and $r_0 = 4.15$ mm, and the frequency is varied. A short computer program was used to calculate the data shown in Figure 7.17. Observe that the coupling varies by less than 1 dB over the band. The directivity is very large (>60 dB) at the design



FIGURE 7.17 Coupling and directivity versus frequency for the Bethe hole coupler of Example 7.3.

frequency, but decreases to 15-20 dB at the band edges. The directivity is a more sensitive function of frequency because it depends on the cancellation of two wave components.

Design of Multihole Couplers

As seen from Example 7.3, a single-hole coupler has a relatively narrow bandwidth, at least in terms of its directivity. But if the coupler is designed with a series of coupling holes, the extra degrees of freedom can be used to increase this bandwidth. The principle of operation and design of such a multihole waveguide coupler is very similar to that of the multisection matching transformer.

First let us consider the operation of the two-hole coupler shown in Figure 7.18, Two parallel waveguides sharing a common broad wall are shown, although the same type of structure could be made in microstrip or stripline form. Two small apertures are spaced $\lambda_q/4$ apart, and couple the two guides. A wave entering at port 1 is mostly transmitted through to port 2, but some power is coupled through the two apertures. If a phase reference is taken at the first aperture, then the phase of the wave incident at the second aperture will be -90° . Each aperture will radiate a forward wave component and a backward wave component into the upper guide; in general, the forward and backward amplitudes are different. In the direction of port 3, both components are in phase, since both have traveled $\lambda_q/4$ to the second aperture. But we obtain a cancellation in the direction of port 4, since the wave coming through the second aperture travels $\lambda_a/2$ further than the wave component coming through the first aperture. Clearly, this cancellation is frequency sensitive, making the directivity a sensitive function of frequency. The coupling is less frequency dependent, since the path lengths from port 1 to port 3 are always the same. Thus, in the multihole coupler design, we synthesize the directivity response, as opposed to the coupling response, as a function of frequency.

We now consider the general case of the multihole coupler shown in Figure 7.19, where N + 1 equally spaced apertures couple two parallel waveguides. The amplitude of the incident wave in the lower left guide is A and, for small coupling, is essentially the same as the amplitude of the through wave. For instance, a 20 dB coupler has a power



FIGURE 7.18 Basic operation of a two-hole directional coupler.



FIGURE 7.19 Geometry of an N + 1 hole waveguide directional coupler.

coupling factor of $10^{-20/10} = 0.01$, so the power transmitted through waveguide A is 1 - 0.01 = 0.99 of the incident power (1% coupled to the upper guide). The voltage (or field) drop in waveguide A is $\sqrt{0.99} = 0.995$, or 0.5%. Thus, the assumption that the amplitude of the incident field is identical at each aperture is a good one. Of course, the phase will change from one aperture to the next.

As we saw in the previous section for the Bethe hole coupler, an aperture generally excites forward and backward traveling waves with different amplitudes. Thus, let

 F_n denote the coupling coefficient of the *n*th aperture in the forward direction.

 B_n denote the coupling coefficient of the *n*th aperture in the backward direction.

Then the amplitude of the forward wave can be written as

$$F = Ae^{-j\beta Nd} \sum_{n=0}^{N} F_n, \qquad 7.46$$

since all components travel the same path length. The amplitude of the backward wave is

$$B = A \sum_{n=0}^{N} B_n e^{-2j\beta nd}, \qquad 7.47$$

since the path length for the *n*th component is $2\beta nd$, where *d* is the spacing between the apertures. In (7.46) and (7.47) the phase reference is taken at the n = 0 aperture.

From the definitions in (7.20a) and (7.20b) the coupling and directivity can be computed as

$$C = -20 \log \left| \frac{F}{A} \right| = -20 \log \left| \sum_{n=0}^{N} F_n \right| dB,$$

$$D = -20 \log \left| \frac{B}{F} \right| = -20 \log \left| \frac{\sum_{n=0}^{N} B_n e^{-2j\beta nd}}{\sum_{n=0}^{N} F_n} \right|$$

$$= -C - 20 \log \left| \sum_{n=0}^{N} B_n e^{-2j\beta nd} \right| dB.$$
7.49

Chapter 7: Power Dividers and Directional Couplers

Now assume that the apertures are round holes with identical positions, s, relative to the edge of the guide, with r_n being the radius of the *n*th aperture. Then we know from Section 4.8 and the preceding section that the coupling coefficients will be proportional to the polarizabilities α_e and α_m of the aperture, and hence proportional to r_n^3 . So we can write

$$F_n = K_f r_n^3, \qquad \qquad 7.50a$$

$$B_n = K_b r_n^3, 7.50b$$

where K_f and K_b are constants for the forward and backward coupling coefficients that are the same for all apertures, but are functions of frequency. Then (7.48) and (7.49) reduce to

$$C = -20 \log |K_f| - 20 \log \sum_{n=0}^{N} r_n^3 dB,$$

$$D = -C - 20 \log |K_b| - 20 \log \left| \sum_{n=0}^{N} r_n^3 e^{-2j\beta_n d} \right|$$

$$= -C - 20 \log |K_b| - 20 \log S dB.$$
7.52

In (7.51), the second term is constant with frequency. The first term is not affected by the choice of r_n s, but is a relatively slowly varying function of frequency. Similarly, in (7.52) the first two terms are slowly varying functions of frequency, representing the directivity of a single aperture, but the last term (S) is a sensitive function of frequency due to phase cancellation in the summation. Thus we can choose the r_n s to synthesize a desired frequency response for the directivity, while the coupling should be relatively constant with frequency.

Observe that the last term in (7.52),

$$S = \left| \sum_{n=0}^{N} r_n^3 e^{-2j\beta nd} \right|, \qquad 7.53$$

is very similar in form to the expression obtained in Section 5.5 for multisection quarterwave matching transformers. As in that case, we will develop coupler designs that yield either a binomial (maximally flat) or a Chebyshev (equal ripple) response for the directivity. Another interpretation of (7.53) may be recognizable to the student familiar with basic antenna theory, as this expression is identical to the array pattern factor of an N + 1 element array with element weights r_n^3 . In that case, too, the pattern may be synthesized in terms of binomial or Chebyshev polynomials.

Binomial response. As in the case of the multisection quarter-wave matching transformers, we can obtain a binomial, or maximally flat, response for the directivity of the multihole coupler by making the coupling coefficients proportional to the binomial coefficients. Thus,

$$r_n^3 = k C_n^N, 7.54$$

where k is a constant to be determined, and C_n^N is a binomial coefficient given in (5.51). To find k, we evaluate the coupling using (7.51) to give

$$C = -20 \log |K_f| - 20 \log k - 20 \log \sum_{n=0}^{N} C_n^N \, \mathrm{dB},$$
 7,55

Since we know K_f , N, and C, we can solve for k and then find the required aperture radii from (7.54). The spacing d should be $\lambda_g/4$ at the center frequency.

Chebyshev response. First assume that N is even (an odd number of holes), and that the coupler is symmetric, so that $r_0 = r_N, r_1 = r_{N-1}$, etc. Then from (7.53) we can write S as

$$S = \left| \sum_{n=0}^{N} r_n^3 e^{-2jn\theta} \right| = 2 \sum_{n=0}^{N/2} r_n^3 \cos(N-2n)\theta,$$

where $\theta = \beta d$. To achieve a Chebyshev response we equate this to the Chebyshev polynomial of degree N:

$$S = 2\sum_{n=0}^{N/2} r_n^3 \cos(N-2n)\theta = k |T_N(\sec\theta_m\cos\theta)|, \qquad 7.56$$

where k and θ_m are constants to be determined. From (7.53) and (7.56), we see that for $\theta = 0$, $S = \sum_{n=0}^{N} r_n^3 = k |T_N(\sec \theta_m)|$. Using this result in (7.51) gives the coupling as

$$C = -20 \log |K_f| - 20 \log S \Big|_{\theta=0}$$

= -20 \log |K_f| - 20 \log k - 20 \log |T_N(\sec \theta_m)| dB. 7.57

From (7.52) the directivity is

$$D = -C - 20 \log |K_b| - 20 \log S$$

= $20 \log \frac{K_f}{K_b} + 20 \log \frac{T_N(\sec \theta_m)}{T_N(\sec \theta_m \cos \theta)} dB.$ 7.58

The term $\log K_f/K_b$ is a function of frequency, so D will not have an exact Chebyshev response. This error is usually small, however. Thus, we can assume that the smallest value of D will occur when $T_N(\sec \theta_m \cos \theta) = 1$, since $|T_N(\sec \theta_m)| \ge |T_N(\sec \theta_m \cos \theta)|$. So if D_{\min} is the specified minimum value of directivity in the passband, then θ_m can be found from the relation

$$D_{\min} = 20 \log T_N(\sec \theta_m) \, \mathrm{dB}.$$
 7.59

Alternatively, we could specify the bandwidth, which then dictates θ_m and D_{\min} . In either case, (7.57) can then be used to find k, and then (7.56) solved for the radii, r_m .

If N is odd (an even number of holes), the results for C, D_c and D_{\min} in (7.57), (7.58), and (7.59) still apply, but instead of (7.56), the following relation is used to find

the aperture radii:

$$S = 2 \sum_{n=0}^{(N-1)/2} r_n^3 \cos(N-2n)\theta = k |T_N(\sec\theta_m \cos\theta)|.$$
 7.60



EXAMPLE 7.4 Multihole Waveguide Coupler Design

Design a four-hole Chebyshev coupler in X-band waveguide using round apertures located at s = a/4. The center frequency is 9 GHz, the coupling is 20 dB, and the minimum directivity is 40 dB. Plot the directivity response from 7 to 11 GHz.

Solution

For X-band waveguide at 9 GHz, we have the following constants:

a = 0.02286 m, b = 0.01016 m, $\lambda_0 = 0.0333$ m, $k_0 = 188.5$ m⁻¹, $\beta = 129.0$ m⁻¹, $Z_{10} = 550.9 \Omega$, $P_{10} = 4.22 \times 10^{-7}$ m²/ Ω .

From (7.40a) and (7.40b), we obtain for an aperture at s = a/4:

$$|K_f| = \frac{2k_0}{3\eta_0 P_{10}} \left[\sin^2 \frac{\pi s}{a} - \frac{2\beta^2}{k_0^2} \left(\sin^2 \frac{\pi s}{a} + \frac{\pi^2}{\beta^2 a^2} \cos^2 \frac{\pi s}{a} \right) \right] = 3.953 \times 10^5,$$

$$|K_b| = \frac{2k_0}{3\eta_0 P_{10}} \left[\sin^2 \frac{\pi s}{a} + \frac{2\beta^2}{k_0^2} \left(\sin^2 \frac{\pi s}{a} - \frac{\pi^2}{\beta^2 a^2} \cos^2 \frac{\pi s}{a} \right) \right] = 3.454 \times 10^5.$$

For a four-hole coupler, N = 3, so (7.59) gives

$$40 = 20 \log T_3(\sec \theta_m) \text{ dB},$$

$$100 = T_3(\sec \theta_m) = \cosh(3 \cosh^{-1}(\sec \theta_m)),$$

$$\sec \theta_m = 3.01,$$

where (5.58b) was used. Thus $\theta_m = 70.6^\circ$ and 109.4° at the band edges. Then from (7.57) we can solve for k:

$$C = 20 = -20 \log(3.953 \times 10^5) - 20 \log k - 40 \text{ dB}.$$

20 \log k = -171.94,
k = 2.53 \times 10^{-9}.



FIGURE 7.20 Coupling and directivity versus frequency for the four-hole coupler of Example 7.4.

Finally, (7.60) and the expansion from (5.60c) for T_3 allow us to solve for the radii as follows:

$$S = 2[r_0^3 \cos 3\theta + r_1^3 \cos \theta] = k[\sec^3 \theta_m (\cos 3\theta + 3\cos \theta) - 3\sec \theta_m \cos \theta],$$
$$2r_0^3 = k \sec^3 \theta_m \quad \Rightarrow \quad r_0 = r_3 = 3.26 \text{ mm},$$
$$2r_1^3 = 3k(\sec^3 \theta_m - \sec \theta_m) \quad \Rightarrow \quad r_1 = r_2 = 4.51 \text{ mm}.$$

The resulting coupling and directivity are plotted in Figure 7.20; note the increased directivity bandwidth compared to that of the Bethe hole coupler of Example 7.3.

THE QUADRATURE (90°) HYBRID

Quadrature hybrids are 3 dB directional couplers with a 90° phase difference in the outputs of the through and coupled arms. This type of hybrid is often made in microstrip or stripline form as shown in Figure 7.21, and is also known as a branch-line hybrid. Other 3 dB couplers, such as coupled line couplers or Lange couplers, can also be used as quadrature couplers; these components will be discussed in later sections. Here we will analyze the operation of the quadrature hybrid using an even-odd mode decomposition technique similar to that used for the Wilkinson power divider.

With reference to Figure 7.21 the basic operation of the branch-line coupler is as follows. With all ports matched, power entering port 1 is evenly divided between ports 2 and 3, with a 90° phase shift between these outputs. No power is coupled to port 4 (the isolated port). Thus, the [S] matrix will have the following form:

$$[S] = \frac{-1}{\sqrt{2}} \begin{bmatrix} 0 & j & 1 & 0 \\ j & 0 & 0 & 1 \\ 1 & 0 & 0 & j \\ 0 & 1 & j & 0 \end{bmatrix}.$$
 7.61





Observe that the branch-line hybrid has a high degree of symmetry, as any port can be used as the input port. The output ports will always be on the opposite side of the junction from the input port, and the isolated port will be the remaining port on the same side as the input port. This symmetry is reflected in the scattering matrix, as each row can be obtained as a transposition of the first row.

Even-Odd Mode Analysis

We first draw the schematic circuit of the branch-line coupler in normalized form, as in Figure 7.22, where it is understood that each line represents a transmission line with indicated characteristic impedance normalized to Z_0 . The common ground return for each transmission line is not shown. We assume that a wave of unit amplitude $A_1 = 1$ is incident at port 1.

Now the circuit of Figure 7.22 can be decomposed into the superposition of an even-mode excitation and an odd-mode excitation [5], as shown in Figure 7.23. Note that adding the two sets of excitations produces the original excitation of Figure 7.22, and since the circuit is linear, the actual response (the scattered waves) can be obtained from the sum of the responses to the even and odd excitations.

Because of the symmetry or antisymmetry of the excitation. the four-port network can be decomposed into a set of two decoupled two-port networks, as shown in Figure 7.23.



FIGURE 7.22 Circuit of the branch-line hybrid coupler in normalized form.


FIGURE 7.23 Decomposition of the branch-line coupler into even- and odd-mode excitations. (a) Even mode (e). (b) Odd mode (o).

Since the amplitudes of the incident waves for these two-ports are $\pm 1/2$, the amplitudes of the emerging wave at each port of the branch-line hybrid can be expressed as

$$\dot{B}_1 = \frac{1}{2}\Gamma_e + \frac{1}{2}\Gamma_o, \qquad 7.62a$$

$$B_2 = \frac{1}{2}T_e + \frac{1}{2}T_o, \qquad 7.62b$$

$$B_3 = \frac{1}{2}T_e - \frac{1}{2}T_o, 7.62c$$

$$B_4 = \frac{1}{2}\Gamma_e - \frac{1}{2}\Gamma_o.$$
 7.62*d*

where $\Gamma_{e,o}$ and $T_{e,o}$ are the even- and odd-mode reflection and transmission coefficients for the two-port networks of Figure 7.23. First consider the calculation of Γ_c and T_c , for the even-mode two-port circuit. This can best be done by multiplying the *ABCD* matrices of each cascade component in that circuit, to give

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{e} = \underbrace{\begin{bmatrix} 1 & 0 \\ j & 1 \end{bmatrix}}_{V=j} \underbrace{\begin{bmatrix} 0 & j/\sqrt{2} \\ j\sqrt{2} & 0 \end{bmatrix}}_{V=j} \underbrace{\begin{bmatrix} 1 & 0 \\ j & 1 \end{bmatrix}}_{Y=j} = \frac{1}{\sqrt{2}} \begin{bmatrix} -1 & j \\ j & -1 \end{bmatrix},$$
7.63
Shuat
$$X = j$$

$$X = j$$

$$Y = j$$

where the individual matrices can be found from Table 4.1, and the admittance of the shunt open-circuited $\lambda/8$ stubs is $Y = j \tan \beta \ell = j$. Then Table 4.2 can be used to convert from ABCD parameters (defined here with $Z_{\alpha} = 1$) to S parameters, which are equivalent to the reflection and transmission coefficients. Thus,

$$\Gamma_e = \frac{A+B-C-D}{A+B+C+D} \approx \frac{(-1+j-j+1)/\sqrt{2}}{(-1+j+j-1)/\sqrt{2}} = 0, \qquad 7.64a$$

$$T_e = \frac{2}{A+B+C+D} = \frac{2}{(-1+j+j-1)/\sqrt{2}} = \frac{-1}{\sqrt{2}}(1+j).$$
 7.64

Similarly, for the odd mode we obtain

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{o} = \frac{1}{\sqrt{2}} \begin{bmatrix} I & j \\ j & I \end{bmatrix}.$$
 7.65

which gives the reflection and transmission coefficients as

$$\Gamma_o = 0, \qquad \qquad 7.66a$$

$$T_o = \frac{1}{\sqrt{2}}(1-j).$$
 7.66b

11

Then using (7.64) and (7.66) in (7.62) gives the following results:

$$B_1 = 0$$
 (port 1 is matched), 7.67a

$$B_2 = -\frac{j}{\sqrt{2}}$$
 (half-power, -90° phase shift from port 1 to 2), 7.676

$$B_3 = -\frac{1}{\sqrt{2}}$$
 (half-power, -180° phase shift from port 1 to 3), 7.67c

$$B_4 = 0 \qquad (\text{no power to port 4}). \qquad 7.67d$$

These results agree with the first row and column of the [S] matrix given in (7.61); the remaining elements can easily be found by transposition.

In practice, due to the quarter-wave length requirement, the bandwidth of a branchline hybrid is limited to 10–20%. But as with multisection matching transformers and multihole directional couplers, the bandwidth of a branch-line hybrid can be increased to a decade or more by using multiple sections in cascade. Figure 7.24 shows a photograph of a quadrature hybrid. In addition, the basic design can be modified for unequal power division and/or different characteristic impedances at the output ports. Another practical point to be aware of 1s the fact that discontinuity effects at the junctions of the branch-line coupler may require that the shunt arms be lengthened by $10^{\circ}-20^{\circ}$.

EXAMPLE 7.5 Design and Performance of a Quadrature Hybrid

Design a 50 Ω branch-line quadrature hybrid junction, and plot the S parameter magnitudes from $0.5f_0$ to $1.5f_0$, where f_0 is the design frequency.



FIGURE 7.24 Photograph of a microstrip quadrature hybrid prototype. Courtesy of M. D. Abouzahra, MIT Lincoln Laboratory, Lexington, Mass.

Solution

After the preceding analysis, the design of a quadrature hybrid is trivial. The lines are $\lambda/4$ at the design frequency f_0 , and the branch-line impedances are

$$\frac{Z_0}{\sqrt{2}} = \frac{50}{\sqrt{2}} = 35.4\,\Omega.$$

The calculated frequency response is plotted in Figure 7.25. Note that we obtain perfect 3 dB power division in ports 2 and 3, and perfect isolation and return loss at ports 4 and 1, respectively, at the design frequency f_0 . All of these quantities, however, degrade quickly as the frequency departs from f_0 .

7.6 COUPLED LINE DIRECTIONAL COUPLERS

When two unshielded transmission lines are close together, power can be coupled between the lines due to the interaction of the electromagnetic fields of each line. Such lines are referred to as coupled transmission lines, and usually consist of three conductors in close proximity, although more conductors can be used. Figure 7.26 shows several examples of coupled transmission lines. Coupled transmission lines are usually assumed to operate in the TEM mode, which is rigorously valid for stripline structures and approximately valid for microstrip structures. In general, a three-wire line, like those of Figure 7.26, can support two distinct propagating modes. This feature can be used to implement directional couplers, hybrids, and filters.



FIGURE 7.25 S parameter magnitudes versus frequency for the branch-line coupler of Example 7.5.

We will first discuss the theory of coupled lines and present some design data for coupled stripline and coupled microstrip. Then we will analyze the operation of a singlesection directional coupler, and extend these results to multisection coupler design.

Coupled Line Theory

The coupled lines of Figure 7.26, or any other three-wire line, can be represented by the structure shown in Figure 7.27. If we assume a TEM type of propagation, then the electrical characteristics of the coupled lines can be completely determined from the effective capacitances between the lines and the velocity of propagation on the line. As depicted in Figure 7.27, C_{12} represents the capacitance the two strip conductors in the absence of the ground conductor, while C_{11} and C_{22} represent the capacitance



FIGURE 7.26 Various coupled transmission line geometries. (a) Coupled stripline (planar, or edge-coupled). (b) Coupled stripline (stacked, or broadside-coupled). (c) Coupled microstrip.



FIGURE 7.27 A three-wire coupled transmission line and its equivalent capacitance network.

between the two strip conductors in the absence ℓ_{-} he ground conductor, while C_{11} and C_{22} represent the capacitance between one strip inductor and ground, in the absence of the other strip conductor. If the strip conductors are identical in size and location relative to the ground conductor, then $C_{11} = C_{22}$. Note that the designation of "ground" for the third conductor has no special relevance beyond the fact that it is convenient, since in many applications this conductor is the ground plane of a stripline or microstrip circuit.

Now consider two special types of excitations for the coupled line: the even mode, where the currents in the strip conductors are equal in amplitude and in the same direction, and the odd mode, where the currents in the strip conductors are equal in amplitude but in opposite directions. The electric field lines for these two cases are sketched in Figure 7.28.

For the even mode, the electric field has even symmetry about the center line, and no current flows between the two strip conductors. This leads to the equivalent circuit



FIGURE 7.28 Even- and odd-mode excitations for a coupled line, and the resulting equivalent capacitance networks. (a) Even-mode excitation. (b) Odd-mode excitation.

shown, where C_{12} is effectively open-circuited. Then the resulting capacitance of either line to ground for the even mode is

$$C_e = C_{11} = C_{22}, 7.68$$

assuming that the two strip conductors are identical in size and location. Then the characteristic impedance for the even mode is

$$Z_{0e} = \sqrt{\frac{L}{C_e}} = \frac{\sqrt{LC_e}}{C_e} = \frac{1}{vC_e},$$
7.69

where v is the velocity of propagation on the line.

For the odd mode, the electric field lines have an odd symmetry about the center line, and a voltage null exists between the two strip conductors. We can imagine this as a ground plane through the middle of C_{12} , which leads to the equivalent circuit as shown. In this case, the effective capacitance between either strip conductor and ground is

$$C_o = C_{11} + 2C_{12} = C_{22} + 2C_{12}, 7.70$$

and the characteristic impedance for the odd mode is

$$Z_{0o} = \frac{1}{vC_o}.$$
 7.71

In words, $Z_{0r}(Z_{0n})$ is the characteristic impedance of one of the strip conductors relative to ground when the coupled line is operated in the even (odd) mode. An arbitrary excitation of a coupled line can always be treated as a superposition of appropriate amplitudes of even and odd modes.

If the coupled line is purely TEM, such as coaxial, parallel plate, or stripline, analytical techniques such as conformal mapping [7] can be used to evaluate the capacitances per unit length of line, and the even- and odd-mode characteristic impedances can then be determined. For quasi-TEM lines, such as microstrip, these results can be obtained numerically or by approximate quasi-static techniques [8]. In either case, such calculations are generally too involved for our consideration, so we will present only two examples of design data for coupled lines.

For a symmetric coupled stripline of the type shown in Figure 7.26a, the design graph in Figure 7.29 can be used to determine the necessary strip widths and spacing for a given set of characteristic impedances, Z_{0e} and Z_{0o} , and the dielectric constant. This graph should cover ranges of parameters for most practical applications, and can be used for any dielectric constant, since stripline supports a purely TEM mode.

For microstrip, the results do not scale with dielectric constant, so design graphs must be made for specific values of dielectric constant. Figure 7.30 shows such a design graph for coupled microstrip lines on a substrate with $\epsilon_r = 10$. Another difficulty with microstrip coupled lines is the fact that the phase velocity is usually different for the two modes of propagation, since the two modes operate with different field configurations in the vicinity of the air-dielectric interface. This can have a degrading effect on coupler directivity.



FIGURE 7.29 Normalized even- and odd-mode characteristic impedance design data for edgecoupled striplines.

Ø

EXAMPLE 7.6 Impedance of a Simple Coupled Line

For the coupled stripline geometry of Figure 7.26b, let W >> S and W >> b, so that fringing fields can be ignored, and determine the even- and odd-mode characteristic impedances.

Solution

We first find the equivalent network capacitances, C_{11} and C_{12} (since the line is symmetric, $C_{22} = C_{11}$). C_{11} is the capacitance of one of the strip conductors to the ground planes, in the absence of the other strip conductor.

The capacitance of a parallel plate capacitor with plate area, A, and plate separation, d, is

$$C = \frac{\epsilon A}{d}$$

with ϵ being the permittivity of the material between the plates. This formula ignores fringing fields.

Now C_{11} is formed by the parallel combination of two capacitors from one strip to the two ground planes. Thus, the capacitance per unit length is

$$\tilde{C}_{11} = \frac{\epsilon_r \epsilon_0 W}{(b-S)/2} + \frac{\epsilon_r \epsilon_0 W}{(b+S)/2} = \frac{4b\epsilon_r \epsilon_0 W}{b^2 - S^2} \quad \text{Fd/m}.$$



FIGURE 7.30 Even- and odd-mode characteristic impedance design data for coupled microstrip lines.

The capacitance between the strips is, per unit length,

$$\bar{C}_{12} = rac{\epsilon_r \epsilon_0 W}{S}$$
 Fd/m

Then from (7.68) and (7.70), the even- and odd-mode capacitances are

$$\begin{split} \bar{C}_{e} &= \bar{C}_{11} = \frac{4b\epsilon_{r}\epsilon_{0}W}{b^{2} - S^{2}} \quad \mathrm{Fd/m}, \\ \bar{C}_{o} &= \bar{C}_{11} + 2\bar{C}_{12} = 2\epsilon_{r}\epsilon_{0}W\left(\frac{2b}{b^{2} - S^{2}} + \frac{1}{S}\right) \quad \mathrm{Fd/m} \end{split}$$

The phase velocity on the line is

$$v = 1/\sqrt{\epsilon_r \epsilon_0 \mu_0},$$

so the characteristic impedances are

$$Z_{0e} = \frac{1}{v\bar{C}_e} = Z_0 \frac{b^2 - S^2}{4b\bar{W}\sqrt{\epsilon_r}},$$

$$Z_{0e} = \frac{1}{v\bar{C}_e} = Z_0 \frac{1}{2\bar{W}\sqrt{\epsilon_r}[2b/(b^2 - S^2) + 1/S]}.$$

Design of Coupled Line Couplers

With the preceding definitions of the even- and odd-mode characteristic impedances, we can apply an even-odd mode analysis to a length of coupled line to arrive at the design equations for a single-section coupled line coupler. Such a line is shown in Figure 7.31. This four-port network is terminated in the impedance Z_0 at three of its ports, and driven with a voltage generator of 2 V and internal impedance Z_0 at port 1. We will show that a coupler can be designed with arbitrary coupling such that the input (port 1) is matched, while port 4 is isolated. Port 2 is the through port, and port 3 is the coupled port. In Figure 7.31, a ground conductor is understood to be common to both strip conductors.

For this problem we will apply the even-odd mode analysis technique in conjunction with the input impedances of the line, as opposed to the reflection and transmission coefficients of the line. So by superposition, the excitation at port 1 in Figure 7.31 can be treated as the sum of the even- and odd-mode excitations shown in Figure 7.32. From symmetry, we can see that $I_1^e = I_3^e$, $I_4^e = I_2^e$, $V_1^e = V_3^e$, and $V_4^e = V_2^e$ for the even modes, while $I_1^a = -I_3^o$, $I_4^o = -I_2^o$, $V_1^b = -V_3^o$, and $V_4^e = -V_2^o$ for the odd mode.

The input impedance at port 1 of the coupler of Figure 7.31 can thus be expressed as

$$Z_{\rm in} = \frac{V_{\rm i}}{I_{\rm 1}} = \frac{V_{\rm i}^e + V_{\rm i}^o}{I_{\rm i}^e + I_{\rm 1}^o}.$$
 7.72



FIGURE 7.31 A single-section coupled line coupler. (a) Geometry and port designations. (b) The schematic circuit.

.



FIGURE 7.32 Decomposition of the coupled line coupler circuit of Figure 7.31 into even- and odd-mode excitations. (a) Even mode. (b) Odd mode.

Now if we let Z_{in}^e be the input impedance at port 1 for the even mode, and Z_{in}^o be the input impedance for the odd mode, then we have

$$Z_{in}^e = Z_{0e} \frac{Z_0 + jZ_{0e} \tan \theta}{Z_{9e} + jZ_0 \tan \theta}, \qquad 7.73a$$

$$Z_{in}^{o} = Z_{0o} \frac{Z_0 + j Z_{0o} \tan \theta}{Z_{0o} + j Z_0 \tan \theta}.$$
 7.73b

since, for each mode, the line looks like a transmission line of characteristic impedance Z_{0e} or Z_{0o} , terminated in a load impedance, Z_0 . Then by voltage division

$$V_{\rm l}^{o} = V \frac{Z_{\rm in}^{o}}{Z_{\rm in}^{o} + Z_{\rm 0}},$$
7.74a

$$V_1^e = V \frac{Z_{in}^e}{Z_{in}^e + Z_0},$$
 7.74b

$$I_1^o = \frac{V}{Z_{in}^o + Z_0},$$
 7.75*a*

$$I_1^e = \frac{V}{Z_{i_0}^e + Z_0}.$$
 7.75b

Using these results in (7.72) yields

$$Z_{\rm in} = \frac{Z_{\rm in}^o(Z_{\rm in}^e + Z_0) + Z_{\rm in}^e(Z_{\rm in}^o + Z_0)}{Z_{\rm in}^e + Z_{\rm in}^o + 2Z_0} = Z_0 + \frac{2(Z_{\rm in}^o Z_{\rm in}^e - Z_0^2)}{Z_{\rm in}^e + Z_{\rm in}^o + 2Z_0}.$$
 7.76

Now if we let

$$Z_0 = \sqrt{Z_{0e} Z_{0o}}, \qquad 7.77$$

then (7.73a,b) reduce to

$$Z_{in}^{e} = Z_{0e} \frac{\sqrt{Z_{0e}} + j\sqrt{Z_{0e}}\tan\theta}{\sqrt{Z_{0e}} + j\sqrt{Z_{0e}}\tan\theta},$$
$$Z_{in}^{o} = Z_{0o} \frac{\sqrt{Z_{0e}} + j\sqrt{Z_{0o}}\tan\theta}{\sqrt{Z_{0o}} + j\sqrt{Z_{0e}}\tan\theta},$$

so that $Z_{in}^{\epsilon}Z_{in}^{o}=Z_{0e}Z_{0o}=Z_{0}^{2}$, and (7.76) reduces to

$$Z_{\rm in} = Z_0. \tag{7.78}$$

Thus, as long as (7.77) is satisfied, port 1 (and, by symmetry, all other ports) will be matched.

Now if (7.77) is satisfied, so that $Z_{in} = Z_0$, we have that $V_1 = V$, by voltage division. The voltage at port 3 is

$$V_{3} = V_{3}^{e} + V_{3}^{o} = V_{1}^{e} - V_{1}^{o} = V \left[\frac{Z_{in}^{e}}{Z_{in}^{e} + Z_{0}} - \frac{Z_{in}^{o}}{Z_{in}^{o} + Z_{0}} \right].$$
 7.79

where (7.74) has been used. From (7.73) and (7.77), we can show that

$$\frac{Z_{in}^{e}}{Z_{in}^{e} + Z_{0}} = \frac{Z_{0} + jZ_{0e}\tan\theta}{2Z_{0} + j(Z_{0e} + Z_{0o})\tan\theta},$$
$$\frac{Z_{in}^{o}}{Z_{in}^{o} + Z_{0}} = \frac{Z_{0} + jZ_{0o}\tan\theta}{2Z_{0} + j(Z_{0e} + Z_{0o})\tan\theta}.$$

so that (7.79) reduces to

$$V_3 = V \frac{j(Z_{0e} - Z_{0o}) \tan \theta}{2Z_0 + j(Z_{0e} + Z_{0o}) \tan \theta}.$$
 7.80

Now define C as

$$C = \frac{Z_{0e} - Z_{0o}}{Z_{0e} + Z_{0o}},$$
 7.81

which we will soon see is actually the midband voltage coupling coefficient, V_3/V . Then,

$$\sqrt{1 - C^{2}} = \frac{2Z_{0}}{Z_{0e} + Z_{0o}},$$

$$V_{3} = V \frac{jC \tan \theta}{\sqrt{1 - C^{2}} + j \tan \theta}.$$
7.82

so that



FIGURE 7.33 Coupled and through port voltages (squared) versus frequency for the coupled line coupler of Figure 7.31.

Similarly, we can show that

$$V_4 = V_4^e + V_4^o = V_2^e - V_2^o = 0, 7.83$$

and

$$V_2 = V_2^e + V_2^o = V \frac{\sqrt{1 - C^2}}{\sqrt{1 - C^2} \cos \theta} + j \sin \theta}.$$
 7.84

Equations (7.82) and (7.84) can be used to plot the coupled and through port voltages versus frequency, as shown in Figure 7.33. At very low frequencies ($\theta << \pi/2$), virtually all power is transmitted through port 2, with none being coupled to port 3. For $\theta = \pi/2$, the coupling to port 3 is at its first maximum; this is where the coupler is generally operated, for small size and minimum line loss. Otherwise, the response is periodic, with maxima in V_3 for $\theta = \pi/2, 3\pi/2, \ldots$

For $\theta = \pi/2$, the coupler is $\lambda/4$ long, and (7.82) and (7.84) reduce to

$$\frac{V_3}{V} = C, 7.85$$

$$\frac{V_2}{V} = -j\sqrt{1-C^2},$$
 7.86

which shows that C < 1 is the voltage coupling factor at the design frequency, $\theta = \pi/2$. Note that these results satisfy power conservation, since $P_{\rm in} = (1/2)|V|^2/Z_0$, while the output powers are $P_2 = (1/2)|V_2|^2/Z_0 = (1/2)(1 - C^2)|V|^2/Z_0$, $P_3 = (1/2)|C|^2|V|^2/Z_0$, $P_4 = 0$, so that $P_{\rm in} = P_2 + P_3 + P_4$. Also observe that there is a 90° phase shift between the two output port voltages; thus this coupler can be used as a quadrature hybrid. And, as long as (7.77) is satisfied, the coupler will be matched at the input and have perfect isolation, at any frequency.

Finally, if the characteristic impedance, Z_0 , and the voltage coupling coefficient, C, are specified, then the following design equations for the required even- and odd-mode

characteristic impedances can be easily derived from (7.77) and (7.81):

$$Z_{0e} = Z_0 \sqrt{\frac{1+C}{1-C}},$$
 7.87*a*

$$Z_{0o} = Z_0 \sqrt{\frac{1-C}{1+C}}.$$
 7.87b

In the above analysis, it was assumed that the even and odd modes of the coupled line structure have the same velocities of propagation, so that the line has the same electrical length for both modes. For a coupled microstrip, or other non-TEM, line this condition will generally not be satisfied, and the coupler will have poor directivity. The fact that coupled microstrip lines have unequal even- and odd-mode phase velocities can be intuitively explained by considering the field line plots of Figure 7.28, which show that the even mode has less fringing field in the air region than the odd mode. Thus its effective dielectric constant should be higher, indicating a smaller phase velocity for the even mode. Techniques for compensating coupled microstrip lines to achieve equal even- and odd-mode phase velocities include the use of dielectric overlays and anistropic substrates.

This type of coupler is best suited for weak couplings, as tight coupling requires lines that are too close together to be practical, or a combination of even- and odd-mode characteristic impedances that is nonrealizable.



EXAMPLE 7.7 Single-Section Coupler Design and Performance

Design a 20 dB single-section coupled line coupler in stripline with a 0.158 cm ground plane spacing, dielectric constant of 2.56, a characteristic impedance of 50 Ω , and a center frequency of 3 GHz. Plot the coupling and directivity from 1 to 5 GHz.

Solution

The voltage coupling factor is

$$C = 10^{-20/20} = 0.1.$$

From (7.87), the even- and odd-mode characteristic impedances are

$$Z_{0e} = 50\sqrt{\frac{1.1}{0.9}} = 55.28 \,\Omega,$$

 $Z_{0e} = 50\sqrt{\frac{0.9}{1.1}} = 45.23 \,\Omega.$

To use Figure 7.29, we have that

$$\sqrt{\epsilon_{ au}} Z_{0e} = 88.4,$$

 $\sqrt{\epsilon_{ au}} Z_{0o} = 72.4,$

and so, W/b = 0.72, S/b = 0.34. This gives a conductor width of

$$W = 0.72b = 0.114$$
 cm,



FIGURE 7.34 Coupling versus frequency for the single-section coupler of Example 7.7.

and a conductor separation of

$$S = 0.34b = 0.054$$
 cm.

Note that these lines are quite close together, which may make fabrication difficult.

The coupling and directivity are plotted in Figure 7.34.

Design of Multisection Coupled Line Couplers

As Figure 7.33 shows, the coupling of a single-section coupled line coupler is limited in bandwidth due to the $\lambda/4$ length requirement. As in the case of matching transformers and waveguide couplers, bandwidth can be increased by using multiple sections. In fact, there is a very close relation between multisection coupled line couplers and multisection quarter-wave transformers [9].

Because the phase characteristics are usually better, multisection coupled line couplers are generally made with an odd number of sections, as shown in Figure 7.35. Thus, we will assume that N is odd. We will also assume that the coupling is weak $(C \le 10 \text{ dB})$, and that each section is $\lambda/4 \log (\theta = \pi/2)$ at the center frequency.

Now for a single coupled line section, with $C \ll 1$, (7.82) and (7.84) simplify to

$$\frac{V_3}{V_1} = \frac{jC\tan\theta}{\sqrt{1-C^2} + j\tan\theta} \simeq \frac{jC\tan\theta}{1+j\tan\theta} = jC\sin\theta e^{-j\theta}, \qquad 7.88a$$

$$\frac{V_2}{V_1} = \frac{\sqrt{1-C^2}}{\sqrt{1-C^2}\cos\theta + j\sin\theta} \simeq e^{-j\theta}.$$
7.88b

Then for $\theta = \pi/2$, we have that $V_3/V_1 = C$ and $V_2/V_1 = -j$. This approximation is equivalent to assuming that no power is lost on the through path from one section to the next, and is similar to the multisection waveguide coupler analysis. It is a good assumption for small C, even though power conservation is violated.



FIGURE 7.35 An N-section coupled line coupler.

Using these results, the total voltage at the coupled port (port 3) of the cascaded coupler in Figure 7.35 can be expressed as

$$V_{3} = (jC_{1}\sin\theta e^{-j\theta})V_{1} + (jC_{2}\sin\theta e^{-j\theta})V_{1}e^{-2j\theta} + \dots + (jC_{N}\sin\theta e^{-j\theta})V_{1}e^{-2j(N-1)\theta},$$

$$7.89$$

where C_n is the voltage coupling coefficient of the *n*th section. If we assume that the coupler is symmetric, so that $C_1 = C_N$, $C_2 = C_{N-1}$, etc., (7.89) can be simplified to

$$V_{3} = jV_{1} \sin \theta e^{-j\theta} [C_{1}(1 + e^{-2j(N-1)\theta}) + C_{2}(e^{-2j\theta} + e^{-2j(N-2)\theta}) + \dots + C_{M} e^{-j(N-1)\theta}]$$

$$= 2jV_{1} \sin \theta e^{-jN\theta} \bigg[C_{1} \cos(N-1)\theta + C_{2} \cos(N-3)\theta + \dots + \frac{1}{2}C_{M} \bigg],$$

$$7.90$$

where M = (N + 1)/2.

At the center frequency, we define the voltage coupling factor C_0 :

$$C_0 = \left| \frac{V_3}{V_1} \right|_{\theta = \pi/2}.$$
7.91

Equation (7.90) is in the form of a Fourier series for the coupling, as a function of frequency. Thus, we can synthesize a desired coupling response by choosing the coupling coefficients, C_n . Note that in this case, we synthesize the coupling response, while in the case of the multihole waveguide coupler we synthesized the directivity response. This is because the path for the uncoupled arm of the multisection coupled line coupler is in the forward direction, and so is less dependent on frequency than the coupled arm path, which is in the reverse direction; this is the opposite situation from the multihole waveguide coupler.

Multisection couplers of this form can achieve decade bandwidths, but coupling levels must be low. Because of the longer electrical length, it is more critical to have

Chapter 7: Power Dividers and Directional Couplers

equal even- and odd-mode phase velocities than it is for the single-section coupler. This usually means that stripline is the preferred medium for such couplers. Mismatched phase velocities will degrade the coupler directivity, as will junction discontinuities, load mismatches, and fabrication tolerances. A photograph of a coupled line coupler is shown in Figure 7.36.



EXAMPLE 7.8 Multisection Coupler Design and Performance

Design a three-section 20 dB coupler with a binomial (maximally flat) response, a system impedance of 50 Ω , and a center frequency of 3 GHz. Plot the coupling and directivity from 1 to 5 GHz.

Solution

For a maximally flat response for a three-section (N = 3) coupler, we require that

$$\left. \frac{d^n}{d\theta^n} C(\theta) \right|_{\theta=\pi/2} = 0.$$
 for $n = 1, 2$

From (7.90).

$$C = \left| \frac{V_3}{V_1} \right| = 2\sin\theta \left[C_1\cos 2\theta + \frac{1}{2}C_2 \right]$$
$$= C_1(\sin 3\theta - \sin\theta) + C_2\sin\theta$$
$$= C_1\sin 3\theta + (C_2 - C_1)\sin\theta,$$



FIGURE 7.36 Photograph of a single-section microstrip coupled line coupler. Courtesy of M. D. Abouzahra, MIT Lincoln Laboratory, Lexington, Mass.

so
$$\frac{dC}{d\theta} = [3C_1 \cos 3\theta + (C_2 - C_1) \cos \theta] \Big|_{\pi/2} = 0,$$

 $\frac{d^2C}{d\theta^2} = [-9C_1 \sin 3\theta - (C_2 - C_1) \sin \theta] \Big|_{\pi/2} = 10C_1 - C_2 = 0.$

Now at midband, $\theta = \pi/2$ and $C_0 = 20$ dB. Thus, $C = 10^{-20/20} = 0.1 = C_2 - 2C_1$. Solving these two equations for C_1 and C_2 gives

$$C_1 = C_3 = 0.0125,$$

 $C_2 = 0.125.$

Then from (7.87) the even- and odd-mode characteristic impedances for each section are

$$Z_{0e}^{1} = Z_{0e}^{3} = 50\sqrt{\frac{1.0125}{0.9875}} = 50.63\,\Omega,$$

$$Z_{0e}^{1} = Z_{0e}^{3} = 50\sqrt{\frac{0.9875}{1.0125}} = 49.38\,\Omega,$$

$$Z_{0e}^{2} = 50\sqrt{\frac{1.125}{0.875}} = 56.69\,\Omega,$$

$$Z_{0e}^{2} = 50\sqrt{\frac{0.875}{1.125}} = 44.10\,\Omega.$$

The coupling and directivity for this coupler are plotted in Figure 7.37. \bigcirc



FIGURE 7.37 Coupling versus frequency for the three-section binomial coupler of Example 7.8.

7.7 THE LANGE COUPLER

Generally, the coupling in a coupled line coupler is too loose to achieve coupling factors of 3 dB or 6 dB. One way to increase the coupling between edge-coupled lines is to use several lines parallel to each other, so that the fringing fields at both edges of a line contribute to the coupling. Probably the most practical implementation of this idea is the Lange coupler [10], shown in Figure 7.38a. Here, four coupled lines are used with interconnections to provide tight coupling. This coupler can easily achieve 3 dB coupling ratios, with an octave or more bandwidth. The design tends to compensate for unequal even- and odd-mode phase velocities, which also improves the bandwidth. There is a 90° phase difference between the output lines (ports 2 and 3), so the Lange





The Lange coupler. (a) Layout in microstrip form. (b) The unfolded Lange coupler.

coupler is a type of quadrature coupler. The main disadvantage of the Lange coupler is probably practical, as the lines are very narrow, close together, and it is difficult to fabricate the necessary bonding wires across the lines. This type of coupled line geometry is also referred to as interdigitated; such structures can also be used for filter circuits.

The unfolded Lange coupler [11], shown in Figure 7.38b, operates essentially the same as the original Lange coupler, but is easier to model with an equivalent circuit. Such an equivalent circuit consists of a four-wire coupled line structure, as shown in Figure 7.39a. All the lines have the same width and spacing. If we make the reasonable assumption that each line couples only to its nearest neighbor, and ignore more distant couplings, then we effectively have a two-wire coupled line circuit, as shown in Figure 7.39b. Then, if we can derive the even- and odd-mode characteristic impedances, Z_{e4} and Z_{o4} , of the four-wire circuit of Figure 7.39a in terms of Z_{0e} and Z_{0o} , the even- and odd-mode characteristic impedances of any adjacent pair of lines, we can apply the coupled line coupler results of Section 7.6 to analyze the Lange coupler.

Figure 7.40a shows the effective capacitances between the conductors of the fourwire coupled line of Figure 7.39a. Unlike the two-line case of Section 7.6, the capacitances of the four lines to ground are different depending on whether the line is on the outside (1 and 4), or on the inside (2 and 3). An approximate relation between these capacitances is given as [12]

$$C_{\rm in} = C_{\rm ex} - \frac{C_{\rm ex}C_m}{C_{\rm ex} + C_m}.$$
7.92



FIGURE 7.39

Equivalent circuits for the unfolded Lange coupler. (a) Four-wire coupled line model. (b) Approximate two-wire coupled line model.



FIGURE 7.40 Effective capacitance networks for the unfolded Lange coupler equivalent circuits of Figure 7.39. (a) Effective capacitance for the four-wire model. (b) Effective capacitance for the two-wire model.

For an even-mode excitation, all four conductors in Figure 7.40a are at the same potential, so C_m has no effect and the total capacitance of any line to ground is

$$C_{e4} = C_{ex} + C_{in}.$$
 7.93a

For an odd-mode excitation, electric walls effectively exist through the middle of each C_m , so the capacitance of any line to ground is

$$C_{\rm o4} = C_{\rm ex} + C_{\rm in} + 6C_m. 7.93b$$

The even- and odd-mode characteristic impedances are then

$$Z_{e4}=\frac{1}{vC_{e4}}, \qquad 7.94a$$

$$Z_{o4} = \frac{1}{vC_{o4}},$$
 7.94b

where v is the velocity of propagation on the line.

Now consider any isolated pair of adjacent conductors in the four-line model; the effective capacitances are as shown in Figure 7.40b. The even- and odd-mode capacitances are

$$C_e = C_{\rm ex}, \qquad 7.95a$$

$$C_o = C_{\rm ex} + 2C_m. \tag{7.95b}$$

Solving (7.95) for C_{ex} and C_m , and substituting into (7.93) with the aid of (7.92) gives the even-odd mode capacitances of the four-wire line in terms of a two-wire coupled line:

$$C_{e4} = \frac{C_e(3C_e + C_o)}{C_e + C_o},$$
7.96a

$$C_{o4} = \frac{C_o(3C_o + C_e)}{C_e + C_o}.$$
 7.96b

Since characteristic impedances are related to capacitance as $Z_0 = 1/vC$, we can rewrite (7.96) to give the even/odd mode characteristic impedances of the Lange coupler in terms

of the characteristic impedances of a two-conductor line which is identical to any pair of adjacent lines in the coupler:

$$Z_{e4} = \frac{Z_{0o} + Z_{0e}}{3Z_{0o} + Z_{0e}} Z_{0e}, \qquad 7.97a$$

$$Z_{o4} = \frac{Z_{0o} + Z_{0e}}{3Z_{0e} + Z_{0o}} Z_{0o}, \qquad 7.97b$$

where Z_{0e} , Z_{0o} are the even- and odd-mode characteristic impedances of the two-conductor pair.

Now we can apply the results of Section 7.6 to the coupler of Figure 7.39b. From (7.77) the characteristic impedance is

$$Z_0 = \sqrt{Z_{e4} Z_{o4}} = \sqrt{\frac{Z_{0c} Z_{0o} (Z_{0o} + Z_{0e})^2}{(3Z_{0o} + Z_{0e})(3Z_{0e} + Z_{0o})^2}},$$
7.98

while the voltage coupling coefficient is, from (7.81),

$$C = \frac{Z_{e4} - Z_{o4}}{Z_{e4} + Z_{o4}} = \frac{3(Z_{0e}^2 - Z_{0o}^2)}{3(Z_{0e}^2 + Z_{0o}^2) + 2Z_{0e}Z_{0o}},$$
7.99

where (7.97) was used. For design purposes, it is useful to invert these results to give the necessary even- and odd-mode impedances for a desired characteristic impedance and coupling coefficient:

$$Z_{0e} = \frac{4C - 3 + \sqrt{9 - 8C^2}}{2C\sqrt{(1 - C)}/(1 + C)} Z_{0}, \qquad 7.100a$$

$$Z_{0o} = \frac{4C + 3 - \sqrt{9 - 8C^2}}{2C\sqrt{(1+C)}/(1-C)} Z_0.$$
 7.100b

These results are approximate because of the simplifications involved with the application of two-line characteristic impedances to the four-line circuit, and because of the assumption of equal even- and odd-mode phase velocities. In practice, however, these results generally give sufficient acuracy. If necessary, a more complete analysis can be made to directly determine Z_{e4} and Z_{o4} for the four-line circuit, as in reference [13].

7.8

- THE 180° HYBRID

The 180° hybrid junction is a four-port network with a 180° phase shift between the two output ports. It can also be operated so that the outputs are in phase. With reference to the 180° hybrid symbol shown in Figure 7.41, a signal applied to port 1 will be evenly split into two in-phase components at ports 2 and 3, and port 4 will be isolated. If the input is applied to port 4, it will be equally split into two components with a 180° phase difference at ports 2 and 3, and port 1 will be isolated. When operated as a combiner, with input signals applied at ports 2 and 3, the sum of the inputs will be formed at port 1, while the difference will be formed at port 4. Hence, ports 1 and 4 are teferred to as the sum and difference ports, respectively. The scattering matrix for the



FIGURE 7.41 Symbol for a 180° hybrid junction.

ideal 3 dB 180° hybrid thus has the following form:

$$[S] = \frac{-j}{\sqrt{2}} \begin{bmatrix} 0 & 1 & 1 & 0 \\ 1 & 0 & 0 & -1 \\ 1 & 0 & 0 & 1 \\ 0 & -1 & 1 & 0 \end{bmatrix}.$$
 7.10

The reader may verify that this matrix is unitary and symmetric.

The 180° hybrid can be fabricated in several forms. The ring hybrid, or rat-race, shown in Figures 7.42 and 7.43a, can easily be constructed in planar (microstrip or stripline) form, although waveguide versions are also possible. Another type of planar 180° hybrid uses tapered matching lines and coupled lines, as shown in Figure 7.43b. Yet another type of hybrid is the hybrid waveguide junction, or magic-T, shown in Figure 7.43c. We will first analyze the ring hybrid, using an even-odd mode analysis similar to that used for the branch-line hybrid, and use a similar technique for the analysis of the tapered line hyrid. Then we will qualitatively discuss the operation of the waveguide magic-T.



FIGURE 7.42 Photograph of a microstrip ring hybrid. Courtesy of M. D. Abouzahra, MFT Lincoln Laboratory, Lexington, Mass.



FIGURE 7.43 Hybrid junctions. (a) A ring hybrid, or *rat-race*, in microstrip or stripline form. (b) A tapered coupled line hybrid. (c) A waveguide hybrid junction, or *magic-T*.

Even-Odd Mode Analysis of the Ring Hybrid

First consider a unit amplitude wave incident at port I (the sum port) of the ring hybrid of Figure 7.43a. At the ring junction this wave will divide into two components, which both arrive in phase at ports 2 and 3, and 180° out of phase at port 4. Using the even-odd mode analysis technique [5], we can decompose this case into a superposition of the two simpler circuits and excitations shown in Figure 7.44. Then the amplitudes of the scattered waves from the ring hybrid will be

$$B_i = \frac{1}{2}\Gamma_e + \frac{1}{2}\Gamma_o, \qquad \qquad 7.102a$$

$$B_2 = \frac{1}{2}T_e + \frac{1}{2}T_a, \qquad 7.102b$$





FIGURE 7.44

Even- and odd-mode decomposition of the ring hybrid when port 1 is excited with a unit amplitude incident wave. (a) Even mode. (b) Odd mode.

$$B_3 = \frac{1}{2}\Gamma_e - \frac{1}{2}\Gamma_o, \qquad 7.102c$$

$$B_4 = \frac{1}{2}T_e - \frac{1}{2}T_o.$$
 7.102d

We can evaluate the required reflection and transmission coefficients defined in Figure 7.44 using the ABCD matrix for the even- and odd-mode two-port circuits in Figure 7.44. The results are

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{e} = \begin{bmatrix} 1 & j\sqrt{2} \\ j\sqrt{2} & -1 \end{bmatrix},$$
 7.103*a*

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{o} = \begin{bmatrix} -1 & j\sqrt{2} \\ j\sqrt{2} & 1 \end{bmatrix}.$$
 7.103b

Then with the aid of Table 4.2 we have

$$\Gamma_e = \frac{-j}{\sqrt{2}}, \qquad 7.104a$$

$$T_e = \frac{-j}{\sqrt{2}}, \qquad \qquad 7.104b$$

7.8 The 180° Hybrid

$$\Gamma_{o} = \frac{j}{\sqrt{2}}, \qquad \qquad 7.104c$$

$$T_o = \frac{-j}{\sqrt{2}}.$$
 7.104d

Using these results in (7.102) gives

•

$$B_1 = 0,$$
 7.105*a*

$$B_2 = \frac{-j}{\sqrt{2}}, \qquad \qquad 7.105b$$

$$B_3 = \frac{-j}{\sqrt{2}},$$
 7.105*c*

$$B_4 = 0,$$
 7.105d

which shows that the input port is matched, port 4 is isolated, and the input power is evenly divided in phase between ports 2 and 3. These results form the first row and column of the scattering matrix in (7.101).

Now consider a unit amplitude wave incident at port 4 (the difference port) of the ring hybrid of Figure 7.43a. The two wave components on the ring will arrive in phase at ports 2 and 3, with a net phase difference of 180° between these ports. The two wave components will be 180° out of phase at port 1. This case can be decomposed into a superposition of the two simpler circuits and excitations shown in Figure 7.45. Then the amplitudes of the scattered waves will be

$$B_2 = \frac{1}{2}\Gamma_e - \frac{1}{2}\Gamma_o, \qquad 7.106b$$

$$B_4 = \frac{1}{2}\Gamma_e + \frac{1}{2}\Gamma_o.$$
 7.106*d*

The ABCD matrices for the even- and odd-mode circuits of Figure 7.45 are

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{e} = \begin{bmatrix} -1 & j\sqrt{2} \\ j\sqrt{2} & 1 \end{bmatrix}.$$
 7.107*a*

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{o} = \begin{bmatrix} 1 & j\sqrt{2} \\ j\sqrt{2} & 1 \end{bmatrix}.$$
 7.107b

Then from Table 4.2, the necessary reflection and transmission coefficients are

$$\Gamma_e = \frac{j}{\sqrt{2}}, \qquad 7.108a$$

$$T_e = \frac{-j}{\sqrt{2}}, \qquad \qquad 7.108b$$



FIGURE 7.45 Even- and odd-mode decomposition of the ring hybrid when port 4 is excited with a unit amplitude incident wave. (a) Even mode. (b) Odd mode.

$$\Gamma_o = \frac{-j}{\sqrt{2}}, \qquad \qquad 7.108c$$

$$T_o = \frac{-j}{\sqrt{2}}.$$
 7.108*d*

Using these results in (7.106) gives

$$B_1 = 0,$$
 7.109a

$$B_2 = \frac{j}{\sqrt{2}}, \qquad 7.109b$$

$$B_3 = \frac{-j}{\sqrt{2}}.$$
 7.109*c*

$$B_4 = 0,$$
 7.109d

which shows that the input port is matched, port 1 is isolated, and the input power is evenly divided into ports 2 and 3 with a 180° phase difference. These results form the fourth row and column of the scattering matrix of (7.101). The remaining elements in this matrix can be found from symmetry considerations.



FIGURE 7.46 S parameter magnitudes versus frequency for the ring hybrid of Example 7.9,

The bandwidth of the ring hybrid is limited by the frequency dependence of the ring lengths, but is generally on the order of 20–30%. Increased bandwidth can be obtained by using additional sections, or a symmetric ring circuit as suggested in reference [14].

Ø

EXAMPLE 7.9 Design and Performance of a Ring Hybrid

Design a 180° ring hybrid for a 50 Ω system impedance, and plot the magnitude of the S parameters (S_{1j}) from 0.5 f_0 to 1.5 f_0 , where f_0 is the design frequency.

Solution

With reference to Figure 7.43a, the characteristic impedance of the ring transmission line is

$$\sqrt{2}Z_0=70.7\,\Omega,$$

while the feedline impedances are 50 Ω . The S parameter magnitudes are plotted versus frequency in Figure 7.46.

Even-Odd Mode Analysis of the Tapered Coupled Line Hybrid

The tapered coupled line 180° hybrid [15], shown in Figure 7.43b, can provide any power division ratio with a bandwidth of a decade or more. This hybrid is also referred to as an asymmetric tapered coupled line coupler.

The schematic circuit of this coupler is shown in Figure 7.47; the ports have been numbered to correspond functionally to the ports of the 180° hybrids in Figures 7.41 and 7.43. The coupler consists of two coupled lines with tapering characteristic impedances over the length 0 < z < L. At z = 0 the lines are very weakly coupled so that $Z_{0e}(0) = Z_{0o}(0) = Z_0$, while at z = L the coupling is such that $Z_{0e}(L) = Z_0/k$ and $Z_{0o}(L) = kZ_0$, where $0 \le k \le 1$ is a coupling factor which we will relate to the voltage coupling factor. The even mode of the coupled line thus matches a load impedance of Z_0/k (at z = L) to Z_0 , while the odd mode matches a load of kZ_0 to



FIGURE 7.47 (a) Schematic diagram of the tapered coupled line hybrid. (b) The variation of characteristic impedances.

 Z_0 ; note that $Z_{0e}(z)Z_{0o}(z) = Z_0^2$ for all z. The Klopfenstein taper is generally used for these tapered matching lines. For L < z < 2L, the lines are uncoupled, and both have a characteristic impedance Z_0 : these lines are required for phase compensation of the coupled line section. The length of each section, $\theta = \beta L$, must be the same, and should be electrically long to provide a good impedance match over the desired bandwidth.

First consider an incident voltage wave of amplitude V applied to port 4, the difference input. This excitation can be reduced to the superposition of an even-mode excitation and an odd-mode excitation, as shown in Figure 7.48a.b. At the junctions of the coupled and uncoupled lines (z = L), the reflection coefficients seen by the even or odd modes of the tapered lines are

$$\Gamma_e' = \frac{Z_0 - Z_0/k}{Z_0 + Z_0/k} = \frac{k-1}{k+1},$$
7.110a

$$\Gamma'_o = \frac{Z_0 - kZ_0}{Z_0 + kZ_0} = \frac{1 - k}{1 + k}.$$
7.110b

Then at z = 0 these coefficients are transformed to

$$\Gamma_e = \frac{k-1}{k+1} e^{-2j\theta}, \qquad 7.111a$$

$$\Gamma_o = \frac{1-k}{1+k} e^{-2j\theta}.$$
7.111b

Then by superposition the scattering parameters of ports 2 and 4 are as follows:

$$S_{44} = \frac{1}{2}(\Gamma_e + \Gamma_o) = 0, \qquad 7.112a$$

$$S_{24} = \frac{1}{2}(\Gamma_e - \Gamma_o) = \frac{k-1}{k+1}e^{-2j\theta}.$$
 7.112b

By symmetry, we also have that $S_{22} = 0$ and $S_{42} = S_{24}$.

To evaluate the transmission coefficients into ports 1 and 3, we will use the ABCD parameters for the equivalent circuits shown in Figure 7.49, where the tapered matching sections have been assumed to be ideal, and replaced with transformers. The ABCD matrix of the transmission line-transformer-transmission line cascade can be found by multiplying the three individual ABCD matrices for these components, but it is easier to use the fact that the transmission line sections affect only the phase of the transmission coefficients. The ABCD matrix of the transmission line sections affect only the phase of the transmission coefficients. The ABCD matrix of the transmission line sections affect only the phase of the transmission coefficients.

$$\begin{bmatrix} \sqrt{k} & \mathbf{0} \\ \mathbf{0} & 1/\sqrt{k} \end{bmatrix},$$

for the even mode, and

$$\begin{bmatrix} 1/\sqrt{k} & 0 \\ 0 & \sqrt{k} \end{bmatrix}$$
.



FIGURE 7.48 Excitation of the tapered coupled line hybrid. (a) Even-mode excitation. (b) Oddmode excitation.



FIGURE 7.49 Equivalent circuits for the tapered coupled line hybrid, for transmission from port 4 to port 3. (a) Even-mode case. (b) Odd-mode case,

for the odd mode. Then the even- and odd-mode transmission coefficients are

$$T_e = T_o = \frac{2\sqrt{k}}{k+1}e^{-2j\theta},$$
 7.113

since $T = 2/(A + B/Z_0 + CZ_0 + D) = 2\sqrt{k}/(k + 1)$ for both modes; the $e^{-2j\theta}$ factor accounts for the phase delay of the two transmission line sections. We can then evaluate the following S parameters:

$$S_{34} = \frac{1}{2}(T_e + T_o) = \frac{2\sqrt{k}}{k+1}e^{-2j\theta}, \qquad 7.114a$$

$$S_{14} = \frac{1}{2}(T_e - T_o) = 0.$$
 7.114b

The voltage coupling factor from port 4 to port 3 is then

$$|\beta = |S_{34}| = \frac{2\sqrt{k}}{k+1}, \qquad 0 < \beta < 1,$$
 7.115*a*

while the voltage coupling factor from port 4 to port 2 is

$$\alpha = |S_{24}| = -\frac{k-1}{k+1}, \quad 0 < \alpha < 1.$$
 7.115b

Power conservation is verified by the fact that

$$|S_{24}|^2 + |S_{34}|^2 = \alpha^2 + \beta^2 = 1.$$

If we now apply even- and odd-mode excitations at ports 1 and 3, so that superposition yields an incident voltage wave at port 1, we can derive the remaining scattering parameters. With a phase reference at the input ports, the even- and odd-mode reflection coefficients at port 1 will be

$$\Gamma_e = \frac{1-k}{1+k} e^{-2j\theta}, \qquad 7.116a$$

$$\Gamma_o = \frac{k-1}{k+1} e^{-2j\theta}.$$
7.116b

Then we can calculate the following S parameters:

$$S_{11} = \frac{1}{2}(\Gamma_e + \Gamma_o) = 0, \qquad 7.117a$$

$$S_{31} = \frac{1}{2}(\Gamma_e - \Gamma_o) = \frac{1 - k}{1 + k}e^{-2j\theta} = \alpha e^{-2j\theta}.$$
 7.117b

From symmetry, we also have that $S_{33} = 0$, $S_{13} = S_{31}$, and that $S_{14} = S_{32}$, $S_{12} = S_{34}$. The tapered coupled line 180° hybrid thus has the following scattering matrix:

$$[S] = \begin{bmatrix} 0 & \beta & \alpha & 0 \\ \beta & 0 & 0 & -\alpha \\ \alpha & 0 & 0 & \beta \\ 0 & -\alpha & \beta & 0 \end{bmatrix} e^{-2j\theta}.$$
 7.118

Waveguide Magic-T

The waveguide magic-T hybrid junction in Figure 7.43c has terminal properties similar to those of the ring hybrid, and a scattering matrix similar in form to (7.101). A rigorous analysis of this junction is too complicated to present here, but we can explain its operation in a qualitative sense by considering the field lines for excitations at the sum and difference ports.

First consider a TE_{10} mode incident at port 1. The resulting E_y field lines are illustrated in Figure 7.50a, where it is seen that there is an odd symmetry about guide 4. Since the field lines of a TE_{10} mode in guide 4 would have even symmetry, there is no coupling between ports 1 and 4. There is identical coupling to ports 2 and 3, however, resulting in an in-phase, equal-split power division.

For a TE₁₀ mode incident at port 4, the field lines are as shown in Figure 7.50b. Again ports 1 and 4 are decoupled, due to symmetry (or reciprocity). Ports 2 and 3 are excited equally by the incident wave, but with a 180° phase difference.

In practice, tuning posts or irises are often used for matching; such components must be placed symmetrically to maintain proper operation of the hybrid.

7.9

OTHER COUPLERS

While we have discussed the general properties of couplers, and have analyzed and derived design data for several of the most frequently used couplers, there are many other types of couplers that we have not treated in detail. In this section, we will briefly describe some of these.



FIGURE 7.50 Electric field lines for a waveguide hybrid junction. (a) Incident wave at port 1, (b) Incident wave at port 4.

Moreno crossed-guide coupler. This is a waveguide directional coupler, consisting of two waveguides at right angles, with coupling provided by two apertures in the common broad wall of the guides. See Figure 7.51. By proper design [16], the two wave components excited by these apertures can be made to cancel in the back direction. The apertures usually consist of crossed slots, in order to couple tightly to the fields of both guides.



FIGURE 7.51 The Moreno crossed-guide coupler.

Schwinger reversed-phase coupler. This waveguide coupler is designed so that the path lengths for the two coupling apertures are the same for the uncoupled port, so that the directivity is essentially independent of frequency. Cancellation in the isolated port is accomplished by placing the slots on opposite sides of the centerline of the waveguide walls, as shown in Figure 7.52, which couple to magnetic dipoles with a 180° phase difference. Then, the $\lambda g/4$ slot spacing leads to in-phase combining at the coupled (backward) port, but this coupling is very frequency sensitive. This is the opposite situation from that of the multihole waveguide coupler discussed in Section 7.4.

Riblet short-slot coupler. Figure 7.53 shows a Riblet short-slot coupler, consisting of two waveguides with a common sidewall. Coupling takes place in the region where part of the common wall has been removed. In this region, both the TE_{10} (even) and the TE_{20} (odd) mode are excited, and by proper design can be made to cause cancellation at the isolated port and addition at the coupled port. The width of the interaction region must generally be reduced to prevent propagation of the undesired TE_{30} mode. This coupler can usually be made smaller than other waveguide couplers.

Symmetric tapered coupled line coupler. We saw that a continuously tapered transmission line matching transformer was the logical extension of the multisection matching transformer. Similarly, the multisection coupled line coupler can be extended to a continuous taper, yielding a coupled line coupler with good bandwidth characteristics. Such a coupler is shown in Figure 7.54. Generally, both the conductor width and separation can be adjusted to provide a synthesized coupling or directivity response. One way to do this involves the computer optimization of a stepped-section approximation to the continuous taper [17]. This coupler provides a 90° phase shift between the outputs.



FIGURE 7.52 The Schwinger reversed-phase coupler.



Couplers with apertures in planar lines. Many of the above-mentioned waveguide couplers can also be fabricated with planar lines such as microstrip, stripline, dielectric image lines, or various combinations of these. Some possibilities are illustrated in Figure 7.55. In principle, the design of such couplers can be carried out using the smallhole coupling theory and analysis techniques used in this chapter. The evaluation of the fields of planar lines, however, is usually much more complicated than for rectangular waveguides.

414

POINT OF INTEREST: The Reflectometer

A reflectometer is a circuit that uses a directional coupler to isolate and sample the incident and reflected powers from a mismatched load. It forms the heart of a scalar or vector network analyzer, as it can be used to measure the reflection coefficient of a one-port network and, in a more general configuration, the S parameters of a two-port network. It can also be used as an SWR meter, or as a power monitor in systems applications.

The basic reflectometer circuit shown below can be used to measure the reflection coefficient magnitude of an unknown load. If we assume a reasonably matched coupler with loose coupling (C << 1), so that $\sqrt{1-C^2} \simeq 1$, then the circuit can be represented by the signal flow graph



FIGURE 7.55 Various aperture coupled planar line couplers. (a) Microstrip-to-microstrip coupler. (b) Microstrip-to-waveguide coupler. (c) Microstrip-to-dielectric image line coupler.

shown below. In operation, the directional coupler provides a sample, V_i , of the incident wave, and a sample, V_r , of the reflected wave. A ratio meter with an appropriately calibrated scale can then measure these voltages and provide a reading in terms of reflection coefficient magnitude, or SWR.



Realistic directional couplers, however, have finite directivity, which means that both the incident and reflected powers will contribute to both V_i and V_r , leading to an error. If we assume a unit incident wave from the source, inspection of the signal flow graph leads to the following expressions for V_i and V_r :

$$V_{i} = C + \frac{C}{D} \Gamma \epsilon^{j\theta},$$
$$V_{\tau} = \frac{C}{D} + C \Gamma e^{j\phi},$$

where Γ is the reflection coefficient of the load. $D = 10^{(D \text{ dB}/20)}$ is the numerical directivity of the coupler, and θ, ϕ are unknown phase delay differences through the circuit. Then the maximum and minimum values of the magnitude of V_r/V_1 can be written as

$$\left|\frac{V_r}{V_r}\right|_{\text{max}} = \frac{|\Gamma| \pm \frac{1}{D}}{1 \mp \frac{|\Gamma|}{D}},$$

For a coupler with infinite directivity this reduces to the desired result of $[\Gamma]$. Otherwise a measurement uncertainty of approximately $\pm (1 + |\Gamma|)/D$ is introduced. Good accuracy thus requires a coupler with high directivity, preferably greater than 40 dB.

REFERENCES

- [1] A. E. Bailey, Ed., Microwave Measurement, Peter Peregrinus, London, 1985.
- [2] R. E. Collin. Foundations for Microwave Engineering, Second Edition, McGraw-Hill, N. Y., 1992.
- [3] F. E. Gardiol, Introduction to Microwaves, Artech House, Dedham, Mass., 1984.
- [4] E. Wilkinson, "An N-Way Hybrid Power Divider," IRE Trans. on Microwave Theory and Techniques, vol. MTT-8, pp. 116–118, January 1960.
- [5] J. Reed and G. J. Wheeler, "A Method of Analysis of Symmetrical Four-Port Networks," IRE Trans. on Microwave Theory and Techniques, vol. MTT-4, pp. 246–252, October 1956.
- [6] C. G. Montgomery, R. H. Dicke. and E. M. Purcell. Principles of Microwave Circuits, MIT Radiation Laboratory Series, vol. 8, McGraw-Hill, N. Y., 1948.
- [7] H. Howe, Stripline Circuit Design, Artech House, Dedham. Mass., 1974.
- [8] K. C. Gupta, R. Garg, and I. J. Bahl, *Microstrip Lines and Slot Lines*. Artech House, Dedham, Mass., 1979.
- [9] L. Young, "The Analytical Equivalence of the TEM-Mode Directional Couplers and Transmission-Line Stepped Impedance Filters," Proc. IEEE, vol. 110, pp. 275–281, February 1963.
- [10] J. Lange, "Interdigitated Stripline Quadrature Hybrid," IEEE Trans. Microwave Theory and Techniques, vol. MTT-17, pp. 1150–1151, December 1969.
- [11] R. Waugh and D. LaCombe, "Unfolding the Lange Coupler," IEEE Trans. Microwave Theory and Techniques. vol. MTT-20, pp. 777–779, November 1972.
- [12] W. P. Ou, "Design Equations for an Interdigitated Directional Coupler," *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-23, pp. 253–255, February 1973.
- [13] D. Paolino, "Design More Accurate Interdigitated Couplers," *Microwaves*, vol. 15, pp. 34–38, May 1976.
- [14] J. Hughes and K. Wilson, "High Power Multiple IMPATT Amplifiers." Proc. European Microwave Conference. pp. 118-122, 1974.
- [15] R. H. DuHamel and M. E. Armstrong, "The Tapered-Line Magic-T." Abstracts of 15th Annual Symposium of the USAF Antenna Research and Development Program, Monticello, Ill., October 12-14, 1965.
- [16] T. N. Anderson, "Directional Coupler Design Nomograms," *Microwave Journal*, vol. 2, pp. 34–38, May 1959.
- [17] D. W. Kammler, "The Design of Discrete N-Section and Continuously Tapered Symmetrical Microwave TEM Directional Couplers." *IEEE Trans. on Microwave Theory and Techniques*, vol. MTT-17, pp. 577-590, August 1969.
PROBLEMS

7.1 Consider the T-junction of three lines with characteristic impedances Z_1 , Z_2 , and Z_3 , as shown on the next page. Demonstrate that it is impossible for all three lines to be matched, when looking toward the junction.



7.2 A directional coupler has the scattering matrix given below. Find the directivity, coupling, isolation, and return loss at the input port when the other ports are terminated in matched loads.

$$[S] = \begin{bmatrix} 0.05 \angle 30 & 0.96 \angle 0 & 0.1 \angle 90 & 0.05 \angle 90 \\ 0.96 \angle 0 & 0.05 \angle 30 & 0.05 \angle 90 & 0.1 \angle 90 \\ 0.1 \angle 90 & 0.05 \angle 90 & 0.04 \angle 30 & 0.96 \angle 0 \\ 0.05 \angle 90 & 0.1 \angle 90 & 0.96 \angle 0 & 0.05 \angle 30 \end{bmatrix}$$

7.3 Two identical 90° couplers with C = 8.34 dB are connected as shown below. Find the resulting phase and amplitudes at ports 2' and 3', relative to port 1.



- 7.4 A 4 W power source is connected to the input of a directional coupler with C = 20 dB, D = 35 dB, and an insertion loss of 0.5 dB. Find the output powers (in dBm) at the through, coupled, and isolated ports. Assume all ports to be matched.
- 7.5 Design a lossless T-junction divider with a 30 Ω source impedance to give a 3:1 power split. Design quarter-wave matching transformers to convert the impedances of the output lines to 30 Ω . Determine the magnitude of the S parameters for this circuit, using a 30 Ω characteristic impedance.
- 7.6 Consider the T and π resistive attenuator circuits shown on the next page. If the input and output are matched to Z_0 , and the ratio of output voltage to input voltage is α , derive the design equations for R_1 and R_2 for each circuit. If $Z_0 = 50 \Omega$, compute R_1 and R_2 for 3 dB, 10 dB, and 20 dB attenuators of each type.



7.7 Design a three-port resistive divider for an equal power split and a 100 Ω system impedance. If port 3 is matched, calculate the change in output power at port 3 (in dB) when port 2 is connected either to a matched load or to a load having a mismatch of $\Gamma = 0.3$.



7.8 Consider the general resistive divider shown below. For an arbitrary power division ratio, $\alpha = P_2/P_3$, derive expressions for the resistors R_1 , R_2 , and R_3 , and the output characteristic impedances Z_{o21}, Z_{o3} so that all ports are matched, assuming the source impedance is Z_0 .



- 7.9 Design a Wilkinson power divider with a power division ratio of $P_3/P_2 = 1/3$, and a source impedance of 50 Ω .
- 7.10 Derive the design equations in (7.37a,b,c) for the unequal-split Wilkinson divider.
- 7.11 For the Bethe hole coupler of the type shown in Figure 7.16a, derive a design for s so that port 3 is the isolated port.
- 7.12 Design a Bethe hole coupler of the type shown in Figure 7.16a for Ku-band waveguide operating at 11 GHz. The required coupling is 20 dB.
- 7.13 Design a Bethe hole coupler of the type shown in Figure 7.16b for Ku-band waveguide operating at 17 GHz. The required coupling is 30 dB.

- 7.14 Design a five-hole directional coupler in Ku-band waveguide with a binomial directivity response. The center frequency is 17.5 GHz, and the required coupling is 20 dB. Use round apertures centered across the broad wall of the waveguides.
- 7.15 Repeat Problem 7.14 for a design with a Chebyshev response, having a minimum directivity of 30 dB.
- 7.16 Develop the necessary equations required to design a two-hole directional coupler using two waveguides with apertures in a common sidewall, as shown below.



7.17 Consider the general branch-line coupler shown below, having shunt arm characteristic impedances Z_a , and series arm characteristic impedances Z_b . Using an even-odd mode analysis, derive design equations for a quadrature hybrid coupler with an arbitrary power division ratio of $\alpha = P_2/P_3$, and with the input port (port 1) matched. Assume all arms are $\lambda/4$ long. Is port 4 isolated, in general?



- 7.18 An edge-coupled stripline with a ground plane spacing of 0.32 cm and a dielectric constant of 2.2 is required to have even- and odd-mode characteristic impedances of $Z_{0e} = 70 \Omega$ and $Z_{0o} = 40 \Omega$. Find the necessary strip widths and spacing.
- 7.19 A coupled microstrip line on a substrate with $\epsilon_r = 10$ and d = 0.16 cm has strip widths of 0.16 cm and a strip spacing of 0.064 cm. Find the even- and odd-mode characteristic impedances.
- 7.20 Repeat the derivation in Section 7.6 for the design equations of a single-section coupled line coupler using reflection and transmission coefficients, instead of voltages and currents.
- 7.21 Design a single-section coupled line coupler with a coupling of 19.1 dB. a system impedance of 60 Ω , and a center frequency of 8 GHz. If the coupler is to be made in stripline (edge-coupled), with $\epsilon_r = 2.2$ and b = 0.32 cm, find the necessary strip widths and separation.
- 7.22 Repeat Problem 7.21 for a coupling factor of 5 dB. Is this a practical design?
- 7.23 Derive equations (7.83) and (7.84).

- 7.24 Design a 25 dB three-section coupled line coupler with a maximally flat coupling response. Assume $Z_0 = 50 \Omega$, and find $Z_{0e_1} Z_{0o}$ for each section. Use CAD to plot the coupling factor (in dB) versus frequency.
- **7.25** Repeat Problem 7.24 for a coupler with an equal-ripple coupling response, where the ripple in the coupling is 1 dB over the passband.
- **7.26** For the Lange coupler, derive the design equations (7.100) for Z_{0e} and Z_{0o} from (7.98) and (7.99).
- 7.27 Design a 3 dB Lange coupler for operation at 5 GHz. If the coupler is to be fabricated in microstrip on an alumina substrate with $\epsilon_r = 10$ and d = 1.0 mm, compute Z_{0e} and Z_{0o} for the two adjacent lines, and find the necessary spacing and widths of the lines.
- 7.28 Consider the four-port hybrid transformer shown below. Determine the scattering matrix for this device, and show that it is similar in form to the scattering matrix for the 180° hybrid. Let the port characteristic impedances be $Z_{01} = Z_{04} = Z_0$; $Z_{02} = Z_{03} = 2Z_0$. (This type of transformer is often used in telephone circuits.)



- 7.29 An input signal V_1 is applied to the sum port of a 180° hybrid, and another signal V_4 is applied to the difference port. What are the output signals?
- 7.30 Calculate the even- and odd-mode characteristic impedances for a tapered coupled line 180° hybrid coupler with a 3 dB coupling ratio and a 50 Ω characteristic impedance.
- 7.31 Find the S parameters for the four-port Bagley polygon power divider shown below.



7.32 For the symmetric hybrid shown below, calculate the output voltages if port 1 is fed with an incident wave of $1\angle 0$ V. Assume the outputs are matched.



Microwave Filters

A microwave filter is a two-port network used to control the frequency response at a certain point in a microwave system by providing transmission at frequencies within the passband of the filter and attenuation in the stopband of the filter. Typical frequency responses include low-pass, high-pass, bandpass, and band-reject characteristics. Applications can be found in virtually any type of microwave communication, radar, or test and measurement system.

Microwave filter theory and practice began in the years preceding World War II, by pioneers such as Mason, Sykes, Darlington, Fano, Lawson, and Richards. The image parameter method of filter design was developed in the late 1930s and was useful for low-frequency filters in radio and telephony. In the early 1950s a group at Stanford Resarch Institute, consisting of G. Matthaei, L. Young, E. Jones, S. Cohn, and others, became very active in filter and coupler development. A voluminous handbook on filters and couplers resulted from this work and remains a valuable reference [1]. Today, most microwave filter design is done with sophisticated computer-aided design (CAD) packages based on the insertion loss method. Because of continuing advancements in network synthesis with distributed elements, the use of low-temperature superconductors, and the incorporation of active devices in filter circuits, microwave filter design remains an active research area.

We begin our discussion of filter theory and design with the frequency characteristics of periodic structures, which consist of a transmission line or waveguide periodically loaded with reactive elements. These structures are of interest in themselves, because of the application to slow-wave components and traveling-wave amplifier design, and **also** because they exhibit basic passband-stopband responses that lead to the image parameter method of filter design.

Filters designed using the *image parameter method* consist of a cascade of simpler two-port filter sections to provide the desired cutoff frequencies and attenuation characteristics, but do not allow the specification of a frequency response over the complete operating range. Thus, although the procedure is relatively simple, the design of filters by the image parameter method often must be iterated many times to achieve the desired results.

A more modern procedure, called the *insertion loss method*, uses network synthesis techniques to design filters with a completely specified frequency response. The design is simplified by beginning with low-pass filter prototypes that are normalized in terms

of impedance and frequency. Transformations are then applied to convert the prototype designs to the desired frequency range and impedance level.

Both the image parameter and insertion loss method of filter design provide lumpedelement circuits. For microwave applications such designs usually must be modified to use distributed elements consisting of transmission line sections. The Richard's transformation and the Kuroda identities provide this step. We will also discuss transmission line filters using stepped impedances and coupled lines; filters using coupled resonators will also be briefly described.

The subject of microwave filters is quite extensive, due to the importance of these components in practical systems and the wide variety of possible implementations. We give here a treatment of only the basic principles and some of the more common filter designs, and refer the reader to references such as [1], [2], [3], and [4] for further discussion.

8.1 PERIODIC STRUCTURES

An infinite transmission line or waveguide periodically loaded with reactive elements is referred to as a periodic structure. As shown in Figure 8.1, periodic structures can take various forms, depending on the transmission line media being used. Often the loading elements are formed as discontinuities in the line, but in any case they can be modeled as lumped reactances across a transmission line as shown in Figure 8.2. Periodic structures support slow-wave propogation (slower than the phase velocity of the unloaded line), and have passband and stopband characteristics similar to those of filters; they find application in traveling-wave tubes, masers, phase shifters, and antennas.



FIGURE 8.1 Examples of periodic structures. (a) Periodic stubs on a microstrip line. (b) Periodic diaphragms in a waveguide.

Analysis of Infinite Periodic Structures

We begin by studying the propagation characteristics of the infinite loaded line shown in Figure 8.2. Each unit cell of this line consists of a length d of transmission line with a shunt susceptance across the midpoint of the line; the susceptance b is normalized to the characteristic impedance, Z_0 . If we consider the infinite line as being composed of a cascade of identical two-port networks, we can relate the voltages and currents on either side of the *n*th unit cell using the *ABCD* matrix:

$$\begin{bmatrix} V_n \\ I_n \end{bmatrix} = \begin{bmatrix} A & B \\ C & D \end{bmatrix} \begin{bmatrix} V_{n+1} \\ I_{n+1} \end{bmatrix},$$
8.1

where A, B, C, and D are the matrix parameters for a cascade of a transmission line section of length d/2, a shunt susceptance b, and another transmission line section of length d/2. From Table 4.1 we then have, in normalized form.

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix} = \begin{bmatrix} \cos\frac{\theta}{2} & j\sin\frac{\theta}{2} \\ j\sin\frac{\theta}{2} & \cos\frac{\theta}{2} \end{bmatrix} \begin{bmatrix} 1 & 0 \\ jb & 1 \end{bmatrix} \begin{bmatrix} \cos\frac{\theta}{2} & j\sin\frac{\theta}{2} \\ j\sin\frac{\theta}{2} & \cos\frac{\theta}{2} \end{bmatrix}$$
$$= \begin{bmatrix} \left(\cos\theta - \frac{b}{2}\sin\theta\right) & j\left(\sin\theta + \frac{b}{2}\cos\theta - \frac{b}{2}\right) \\ j\left(\sin\theta + \frac{b}{2}\cos\theta + \frac{b}{2}\right) & \left(\cos\theta - \frac{b}{2}\sin\theta\right) \end{bmatrix}.$$
 8.2

where $\theta = kd$, and k is the propagation constant of the unloaded line. The reader can verify that AD - BC = 1, as required for reciprocal networks.

Now for any wave propagating in the +z direction, we must have

$$V(z) = V(0)e^{-\gamma z}, \qquad 8.3a$$

$$I(z) = I(0)e^{-\gamma z}, \qquad 8.3b$$

for a phase reference at z = 0. Since the structure is infinitely long, the voltage and current at the *n*th terminals can differ from the voltage and current at the n+1 terminals



FIGURE 8.2 Equivalent circuit of a periodically loaded transmission line. The unloaded line has characteristic impedance Z_0 and propagation constant k.

only by the propagation factor, $e^{-\gamma d}$. Thus,

$$V_{n+1} = V_n e^{-\gamma d}, \qquad 8.4a$$

$$I_{n+1} = I_n e^{-\gamma d}.$$
 8.4b

Using this result in (8.1) gives the following:

or
$$\begin{bmatrix} V_n \\ I_n \end{bmatrix} = \begin{bmatrix} A & B \\ C & D \end{bmatrix} \begin{bmatrix} V_{n+1} \\ I_{n+1} \end{bmatrix} = \begin{bmatrix} V_{n+1}e^{\gamma d} \\ I_{n+1}e^{\gamma d} \end{bmatrix},$$
$$\begin{bmatrix} A - e^{\gamma d} & B \\ C & D - e^{\gamma d} \end{bmatrix} \begin{bmatrix} V_{n+1} \\ I_{n+1} \end{bmatrix} = 0.$$

For a nontrivial solution, the determinant of the above matrix must vanish:

$$AD + e^{2\gamma d} - (A + D)e^{\gamma d} - BC = 0,$$
 8.6

or, since AD - BC = 1,

$$1 + e^{2\gamma d} - (A + D)e^{\gamma d} = 0,$$

$$e^{-\gamma d} + e^{\gamma d} = A + D,$$

$$\cosh \gamma d = \frac{A + D}{2} = \cos \theta - \frac{b}{2} \sin \theta,$$
8.7

where (8.2) was used for the values of A and D. Now if $\gamma = \alpha + j\beta$, we have that

$$\cosh \gamma d = \cosh \alpha d \cos \beta d + j \sinh \alpha d \sin \beta d = \cos \theta - \frac{b}{2} \sin \theta.$$

Since the right-hand side of (8.8) is purely real, we must have either $\alpha = 0$ or $\beta = 0$.

Case 1: $\alpha = 0, \beta \neq 0$. This case corresponds to a nonattenuating, propagating wave on the periodic structure, and defines the passband of the structure. Then (8.8) reduces to

$$\cos\beta d = \cos\theta - \frac{b}{2}\sin\theta, \qquad 8.9a$$

which can be solved for β if the magnitude of the right-hand side is less than or equal to unity. Note that there are an infinite number of values of β that can satisfy (8.9a).

Case 2: $\alpha \neq 0, \beta = 0, \pi$. In this case the wave does not propagate, but is attenuated along the line; this defines the stopband of the structure. Because the line is lossless, power is not dissipated, but is reflected back to the input of the line. The magnitude of (8.8) reduces to

$$\cosh \alpha d = |\cos \theta - \frac{b}{2} \sin \theta| \ge 1,$$
 8.9b

which has only one solution $(\alpha > 0)$ for positively traveling waves: $\alpha < 0$ applies for negatively traveling waves. If $\cos \theta - (b/2) \sin \theta \le -1$. (8.9b) is obtained from (8.8) by letting $\beta = \pi$; then all the lumped loads on the line are $\lambda/2$ apart, yielding an input impedance the same as if $\beta = 0$.

Chapter 8: Microwave Filters

Thus, depending on the frequency and normalized susceptance values, the periodically loaded line will exhibit either passbands or stopbands, and so can be considered as a type of filter. It is important to note that the voltage and current waves defined in (8.3) and (8.4) are meaningful only when measured at the terminals of the unit cells, and do not apply to voltages and currents that may exist at points within a unit cell. These waves are sometimes referred to as *Bloch waves* because of their similarity to the elastic waves that propagate through periodic crystal lattices.

Besides the propagation constant of the waves on the periodically loaded line, we will also be interested in the characteristic impedance for these waves. We can define a characteristic impedance at the unit cell terminals as

$$Z_B = Z_0 \frac{V_{n+1}}{I_{n+1}},$$
 8.10

since V_{n+1} and I_{n+1} in the above derivation were normalized quantities. This impedance is also referred to as the Bloch impedance. From (8.5) we have that

$$(A - e^{\gamma d})V_{n+1} + BI_{n+1} = 0.$$

so (8.10) yields

$$Z_B = rac{-BZ_0}{A - e^{\gamma d}}$$

From (8.6) we can solve for $e^{\gamma d}$ in terms of A and D as follows:

$$e^{\gamma d} = \frac{(A+D) \pm \sqrt{(A+D)^2 - 4}}{2}.$$

Then the Bloch impedance has two solutions given by

$$Z_B^{\pm} = \frac{-2BZ_0}{2A - A - D \mp \sqrt{(A+D)^2 - 4}}.$$
 8.14

For symmetrical unit cells (as assumed in Figure (8.2)) we will always have A = D. In this case (8.11) reduces to

$$Z_B^{\pm} = \frac{\pm B Z_0}{\sqrt{A^2 - 1}}.$$
 8.12

The \pm solutions correspond to the characteristic impedance for positively and negatively traveling waves, respectively. For symmetrical networks these impedances are the same except for the sign; the characteristic impedance for a negatively traveling wave turns out to be negative because we have defined I_n in Figure 8.2 as always being in the positive direction.

From (8.2) we see that B is always purely imaginary. If $\alpha = 0, \beta \neq 0$ (passband), then (8.7) shows that $\cosh \gamma d = A \leq 1$ (for symmetrical networks) and (8.12) shows that Z_B will be real. If $\alpha \neq 0, \beta = 0$ (stopband), then (8.7) shows that $\cosh \gamma d = A \geq 1$, and (8.12) shows that Z_B is imaginary. This situation is similar to that for the wave impedance of a waveguide, which is real for propagating modes and imaginary for cutoff, or evanescent, modes.

Terminated Periodic Structures

Next consider a truncated periodic structure, terminated in a load impedance Z_L , as shown in Figure 8.3. At the terminals of an arbitrary unit cell, the incident and reflected voltages and currents can be written as (assuming operation in the passband)

$$V_n = V_0^+ e^{-j\beta nd} + V_0^- e^{j\beta nd}, \qquad 8.13a$$

$$I_{n} = I_{0}^{+} e^{-j\beta nd} + I_{0}^{-} e^{j\beta nd} = \frac{V_{0}^{+}}{Z_{B}^{+}} e^{-j\beta nd} + \frac{V_{0}^{-}}{Z_{B}^{-}} e^{j\beta nd}, \qquad 8.13b$$

where we have replaced γz in (8.3) with $j\beta nd$, since we are interested only in terminal quantities.

Now define the following incident and reflected voltages at the nth unit cell:

$$V_n^+ = V_0^+ e^{-j\beta nd}, 8.14a$$

$$V_n^- = V_0^- e^{j\beta nd}.$$
 8.14b

Then (8.13) can be written as

$$V_n = V_n^+ + V_n^-, 8.15a$$

$$I_n = \frac{V_n^+}{Z_B^+} + \frac{V_n^-}{Z_B^-}.$$
 8.156

At the load, where n = N, we have

$$V_N = V_N^+ + V_N^- = Z_L I_N = Z_L \left(\frac{V_N^+}{Z_B^+} + \frac{V_N^-}{Z_B^-} \right), \qquad 8.16$$

so the reflection coefficient at the load can be found as

$$\Gamma = \frac{V_N^-}{V_N^+} = -\frac{Z_L/Z_B^+ - 1}{Z_L/Z_B^- - 1}.$$
8.17

If the unit cell network is symmetrical (A = D), then $Z_B^+ = -Z_B^- = Z_B$, which reduces (8.17) to the familiar result that

$$\Gamma = \frac{Z_L - Z_B}{Z_L + Z_B}.$$
8.18





So to avoid reflections on the terminated periodic structure, we must have $Z_L = Z_B$, which is real for a lossless structure operating in a passband. If necessary, a quarter-wave transformer can be used between the periodically loaded line and the load.

k-*β* Diagrams and Wave Velocities

When studying the passband and stopband characteristics of a periodic structure, it is useful to plot the propagation constant, β , versus the propagation constant of the unloaded line, k (or ω). Such a graph is called a k- β diagram, or Brillouin diagram (after L. Brillouin, a physicist who studied wave propagation in periodic crystal structures).

The k- β diagram can be plotted from (8.9a), which is the dispersion relation for a general periodic structure. In fact, a k- β diagram can be used to study the dispersion characteristics of many types of microwave components and transmission lines. For instance, consider the dispersion relation for a waveguide mode:

$$eta=\sqrt{k^2-k_a^2}, \ k=\sqrt{eta^2+k_c^2}, \ 8.19$$

where k_c is the cutoff wavenumber of the mode, k is the free-space wavenumber, and β is the propagation constant of the mode. Relation (8.19) is plotted in the k- β diagram of Figure 8.4. For values of $k < k_c$, there is no real solution for β , so the mode is nonpropagating. For $k > k_c$, the mode propagates, and k approaches β for large values of β (TEM propagation).

The k- β diagram is also useful for interpreting the various wave velocities associated with a dispersive structure. The phase velocity is

$$\boldsymbol{v_p} = \frac{\omega}{\beta} = c\frac{k}{\beta}, \qquad \qquad 8.20$$



FIGURE 8.4 k- β diagram for a waveguide mode.

OT

which is seen to be equal to c (speed of light) times the slope of the line from the origin to the operating point on the k- β diagram. The group velocity is

$$v_{g} = \frac{d\omega}{d\beta} = c \frac{dk}{d\beta},$$
 8.21

which is the slope of the k- β curve at the operating point. Thus, referring to Figure 8.4, we see that the phase velocity for a propagating waveguide mode is infinite at cutoff and approaches c (from above) as k increases. The group velocity, however, is zero at cutoff and approaches c (from below) as k increases. We finish our discussion of periodic structures with a practical example of a capacitively loaded line.



EXAMPLE 8.1 Analysis of a Periodic Structure

Consider a periodic capacitively loaded line, as shown in Figure 8.5 (such a line may be implemented as in Figure 8.1 with short capacitive stubs). If $Z_0 = 50 \ \Omega, d = 1.0 \ \text{cm}$, and $C_0 = 2.666 \ \text{pF}$, sketch the k- β diagram and compute the propagation constant, phase velocity, and Bloch impedance at $f = 3.0 \ \text{GHz}$. Assume $k = k_0$.

Solution

We can rewrite the dispersion relation of (8.9a) as

ł

$$\cos\beta d = \cos k_0 d - \left(\frac{C_0 Z_0 c}{2d}\right) k_0 d \sin k_0 d$$

Then

$$\frac{C_0 Z_0 c}{2d} = \frac{(2.666 \times 10^{-12})(50)(3 \times 10^8)}{2(0.01)} = 2.0,$$

so we have

$$\coseta d = \cos k_0 d - 2k_0 d \sin k_0 d$$

The most straightforward way to proceed at this point is to numerically evaluate the right-hand side of the above equation for a set of values of k_0d starting at zero. When the magnitude of the right-hand side is unity or less, we have a passband and can solve for βd . Otherwise we have a stopband. Calculation shows that the first passband exists for $0 \le k_0d \le 0.96$. The second passband does not begin until the sin k_0d term changes sign at $k_0d = \pi$. As k_0d increases, an







FIGURE 8.6 k- β diagram for Example 8.1.

infinite number of passbands are possible, but they become narrower. Figure 8.6 shows the $k-\beta$ diagram for the first two passbands.

At 3.0 GHz, we have

$$k_0 d = \frac{2\pi (3 \times 10^9)}{3 \times 10^8} (0.01) = 0.6283 = 36^\circ,$$

so $\beta d = 1.5$ and the propagation constant is $\beta = 150$ rad/m. The phase velocity is

$$v_p = \frac{k_0 c}{\beta} = \frac{0.6283}{1.5}c = 0.42c,$$

which is much less than the speed of light, indicating that this is a slow-wave structure. To evaluate the Bloch impedance, we use (8.2) and (8.12):

$$\frac{b}{2} = \frac{\omega C_0 Z_0}{2} = 1.256,$$
$$\theta = k_0 d = 36^\circ,$$
$$A = \cos \theta - \frac{b}{2} \sin \theta = 0.0707,$$
$$B = j \left(\sin \theta + \frac{b}{2} \cos \theta - \frac{b}{2}\right) = j0.3479.$$

Then,

$$Z_B = \frac{BZ_0}{\sqrt{A^2 - 1}} = \frac{(j0.3479)(50)}{j\sqrt{1 - (0.0707)^2}} = 17.4 \ \Omega.$$

Ο

8.2_____ FILTER DESIGN BY THE IMAGE PARAMETER METHOD

The image parameter method of filter design involves the specification of passband and stopband characteristics for a cascade of two-port networks, and so is similar in concept to the periodic structures that were studied in Section 8.1. The method is relatively simple but has the disadvantage that an arbitrary frequency response cannot be incorporated into the design. This is in contrast to the insertion loss method, which is the subject of the following section. Nevertheless, the image parameter method is useful for simple filters and provides a link between infinite periodic structures and practical filter design. The image parameter method also finds application in solid-state traveling-wave amplifier design.

Image Impedances and Transfer Functions for Two-Port Networks

We begin with definitions of the image impedances and voltage transfer function for an arbitrary reciprocal two-port network; these results are required for the analysis and design of lilters by the image parameter method.

Consider the arbitrary two-port network shown in Figure 8.7, where the network is specified by its ABCD parameters. Note that the reference direction for the current at port 2 has been chosen according to the convention for ABCD parameters. The image impedances, Z_{i1} and Z_{i2} , are defined for this network as follows:

 Z_{i1} = input impedance at port 1 when port 2 is terminated with Z_{i2} .

 Z_{i2} = input impedance at port 2 when port 1 is terminated with Z_{i1} .

Thus both ports are matched when terminated in their image impedances. We will now derive expressions for the image impedances in terms of the ABCD parameters of a network.

The port voltages and currents are related as

$$V_1 = AV_2 + BI_2,$$
 8.22a

$$I_1 = CV_2 + DI_2. 8.22b$$



FIGURE 8.7 A two-port network terminated in its image impedances.

The input impedance at port 1, with port 2 terminated in Z_{i2} , is

$$Z_{\text{inl}} = \frac{V_1}{I_1} = \frac{AV_2 + BI_2}{CV_2 + DI_2} = \frac{AZ_{i2} + B}{CZ_{i2} + D},$$
8.23

since $V_2 = Z_{i2}I_2$.

Now solve (8.22) for V_2 , I_2 by inverting the ABCD matrix. Since AD - BC = 1 for a reciprocal network, we obtain

$$V_2 = DV_1 - BI_1, \qquad 8.24a$$

$$I_2 = -CV_1 + AI_1. 8.24b$$

Then the input impedance at port 2, with port 1 terminated in Z_{i1} , can be found as

$$Z_{in2} = \frac{-V_2}{I_2} = -\frac{DV_1 - BI_1}{-CV_1 + AI_1} = \frac{DZ_{i1} + B}{CZ_{i1} + A},$$
8.25

since $V_1 = -Z_{i1}I_1$ (circuit of Figure 8.7).

We desire that $Z_{in1} = Z_{i1}$ and $Z_{in2} = Z_{i2}$, so (8.23) and (8.25) give two equations for the image impedances:

$$Z_{i1}(CZ_{i2}+D) = AZ_{i2}+B, \qquad 8.26a$$

$$Z_{i1}D - B = Z_{i2}(A - CZ_{i1}).$$
 8.26b

Solving for Z_{i1} and Z_{i2} gives

$$Z_{i1} = \sqrt{\frac{AB}{CD}}, \qquad 8.27a$$

$$Z_{i2} = \sqrt{\frac{BD}{AC}}, \qquad 8.27b$$

with $Z_{i2} = DZ_{i1}/A$. If the network is symmetric, then A = D and $Z_{i1} = Z_{i2}$ as expected.

Now consider the voltage transfer function for a two-port network terminated in its image impedances. With reference to Figure 8.8 and (8.24a), the output voltage at port 2 can be expressed as

$$V_2 = DV_1 - BI_1 = \left(D - \frac{B}{Z_{i1}}\right)V_1$$
 8.28

(since we now have $V_1 = I_1 Z_{\mu}$) so the voltage ratio is

$$\frac{V_2}{V_1} = D - \frac{B}{Z_{i1}} = D - B\sqrt{\frac{CD}{AB}} = \sqrt{\frac{D}{A}}(\sqrt{AD} - \sqrt{BC}).$$
 8.29a

Similarly, the current ratio is

$$\frac{I_2}{I_1} = -C\frac{V_1}{I_1} + A = -CZ_{i1} + A = \sqrt{\frac{A}{D}}(\sqrt{AD} - \sqrt{BC}). \qquad 8.29b$$



FIGURE 8.8 A two-port network terminated in its image impedances and driven with a voltage generator.

The factor $\sqrt{D/A}$ occurs in reciprocal positions in (8.29a) and (8.29b), and so can be interpreted as a transformer turns ratio. Apart from this factor, we can define a propagation factor for the network as

$$e^{-\gamma} = \sqrt{AD} - \sqrt{BC}, \qquad 8.30$$

with $\gamma = \alpha + j\beta$ as usual. Since $e^{\gamma} = 1/(\sqrt{AD} - \sqrt{BC}) = (AD - BC)/(\sqrt{AD} - \sqrt{BC}) = \sqrt{AD} + \sqrt{BC}$, and $\cosh \gamma = (e^{\gamma} + e^{-\gamma})/2$, we also have that

$$\cosh \gamma = \sqrt{AD}, \qquad 8.31$$

Two important types of two-port networks are the T and π circuits, which can be made in symmetric form. Table 8.1 lists the image impedances and propagation factors, along with other useful parameters, for these two networks.

Constant-k Filter Sections

Now we are ready to develop low-pass and high-pass filter sections. First consider the T network shown in Figure 8.9; intuitively, we can see that this is a low-pass filter network because the series inductors and shunt capacitor tend to block high-frequency signals while passing low-frequency signals. Comparing with the results given in Table 8.1, we have $Z_1 = j\omega L$ and $Z_2 = 1/j\omega C$, so the image impedance is

$$Z_{iT} = \sqrt{\frac{L}{C}} \sqrt{1 - \frac{\omega^2 LC}{4}}.$$
8.32

If we define a cutoff frequency, ω_c , as

$$\omega_c = \frac{2}{\sqrt{LC}},$$
8.33

and a nominal characteristic impedance, R_0 , as

$$R_0 = \sqrt{\frac{L}{C}} = k, \qquad 8.34$$



TABLE 8.1 Image Parameters for T and π Networks

where k is a constant, then (8.32) can be rewritten as

$$Z_{iT} = R_0 \sqrt{1 - \frac{\omega^2}{\omega_c^2}}.$$
8.35

Then $Z_{iT} = R_0$ for $\omega = 0$.

The propagation factor, also from Table 8.1, is

$$e^{\gamma} = 1 - \frac{2\omega^2}{\omega_c^2} + \frac{2\omega}{\omega_c}\sqrt{\frac{\omega^2}{\omega_c^2} - 1}.$$
 8.36



Now consider two frequency regions:

1. For $\omega < \omega_c$: This is the passband of the filter section. Equation (8.35) shows that Z_{iT} is real, and (8.36) shows that γ is imaginary, since $\omega^2/\omega_c^2 - 1$ is negative and $|e^{\gamma}| = 1$:

$$|e^{\gamma}|^{2} = \left(1 - \frac{2\omega^{2}}{\omega_{c}^{2}}\right)^{2} + \frac{4\omega^{2}}{\omega_{c}^{2}}\left(1 - \frac{\omega^{2}}{\omega_{c}^{2}}\right) = 1.$$

2. For $\omega > \omega_c$: This is the stopband of the filter section. Equation (8.35) shows that Z_{iT} is imaginary, and (8.36) shows that γ is real, since e^{γ} is real and $-1 < e^{\gamma} < 0$ (as seen from the limits as $\omega \to \omega_c$ and $\omega \to \infty$). The attenuation rate for $\omega >> \omega_c$ is 40 dB/decade.

Typical phase and attenuation constants are sketched in Figure 8.10. Observe that the attenuation, α , is zero or relatively small near the cutoff frequency, although $\alpha \to \infty$ as $\omega \to \infty$. This type of filter is known as a constant-k low-pass prototype. There are only two parameters to choose (L and C), which are determined by ω_c , the cutoff frequency, and R_0 , the image impedance at zero frequency.

The above results are valid only when the filter section is terminated in its image impedance at both ports. This is a major weakness of the design, because the image impedance is a function of frequency which is not likely to match a given source or load impedance. This disadvantage, as well as the fact that the attenuation is rather small near cutoff, can be remedied with the modified m-derived sections to be discussed shortly.

For the low-pass π network of Figure 8.9, we have that $Z_1 = j\omega L$ and $Z_2 = 1/j\omega C$, so the propagation factor is the same as that for the low-pass T network. The cutoff frequency, ω_c , and nominal characteristic impedance, R_0 , are the same as the corresponding quantities for the T network as given in (8.33) and (8.34). At $\omega = 0$ we have that $Z_{iT} = Z_{i\pi} = R_0$, where $Z_{i\pi}$ is the image impedance of the low-pass π -network, but Z_{iT} and $Z_{i\pi}$ are generally not equal at other frequencies.





Typical passband and stopband characteristics of the low-pass constant-k sections of Figure 8.9.



FIGURE 8.11 High-pass constant-k filter sections in T and π form. (a) T-section (b) π -section.

High-pass constant-k sections are shown in Figure 8.11; we see that the positions of the inductors and capacitors are reversed from those in the low-pass prototype. The design equations are easily shown to be

$$R_{0} = \sqrt{\frac{\mathcal{L}}{C}}, \qquad 8.37$$

$$\omega_c = \frac{1}{2\sqrt{LC}}.$$
 8.38

m-Derived Filter Sections

We have seen that the constant-k filter section suffers from the disadvantages of a relatively slow attenuation rate past cutoff, and a nonconstant image impedance. The *m*-derived filter section is a modification of the constant-k section designed to overcome these problems. As shown in Figure 8.12a,b the impedances Z_1 and Z_2 in a constant-k T-section are replaced with Z'_1 and Z'_2 , and we let

$$Z_1' = m Z_1.$$
 8.39

Then we choose Z'_{2} to obtain the same value of Z_{iT} as for the constant-k section. Thus, from Table 8.1,

$$Z_{iT} = \sqrt{Z_1 Z_2 + \frac{Z_1^2}{4}} = \sqrt{Z_1' Z_2' + \frac{Z_1'^2}{4}} = \sqrt{m Z_1 Z_2' + \frac{m^2 Z_1^2}{4}}.$$
 8.40

Solving for Z'_2 gives

$$Z_2' = \frac{Z_2}{m} + \frac{Z_1}{4m} - \frac{mZ_1}{4} = \frac{Z_2}{m} + \frac{(1-m^2)}{4m} Z_1.$$
 8.41

Because the impedances Z_1 and Z_2 represent reactive elements, Z'_2 represents two elements in series, as indicated in Figure 8.12c. Note that m = 1 reduces to the original constant-k section.

For a low-pass filter, we have $Z_1 = j\omega L$ and $Z_2 = 1/j\omega C$. Then (8.39) and (8.41) give the *m*-derived components as

$$Z_1' = j\omega Lm, \qquad 8.42a$$



FIGURE 8.12 Development of an m-derived filter section from a constant-k section. (a) Constantk section. (b) General m-derived section. (c) Final m-derived section.

$$Z'_{2} = \frac{1}{j\omega Cm} + \frac{(1-m^{2})}{4m} j\omega L,$$
8.42b

which results in the circuit of Figure 8.13. Now consider the propagation factor for the m-derived section. From Table 8.1,

$$e^{\gamma} = 1 + \frac{Z_1'}{2Z_2'} + \sqrt{\frac{Z_1'}{Z_2'}} \left(1 + \frac{Z_1'}{4Z_2'}\right).$$
8.43

For the low-pass *m*-derived filter,

$$\frac{Z'_1}{Z'_2} = \frac{j\omega Lm}{(1/j\omega Cm) + j\omega L(1-m^2)/4m} = \frac{-(2\omega m/\omega_c)^2}{1 - (1-m^2)(\omega/\omega_c)^2},$$





where $\omega_c = 2/\sqrt{LC}$ as before. Then,

$$l + rac{Z_1'}{4Z_2'} = rac{1 - (\omega/\omega_c)^2}{1 - (1 - m^2)(\omega/\omega_c)^2}$$

If we restrict 0 < m < 1, then these results show that e^{γ} is real and $|e^{\gamma}| > 1$ for $\omega > \omega_{e^{\gamma}}$. Thus the stopband begins at $\omega = \omega_c$, as for the constant-k section. However, when $\omega = \omega_{\infty}$, where

$$\omega_{\infty} = \frac{\omega_{e}}{\sqrt{1 - m^2}},$$
8.44

the denominators vanish and e^{γ} becomes infinite, implying infinite attenuation. Physically, this pole in the attenuation characteristic is caused by the resonance of the series LC resonator in the shunt arm of the T; this is easily verified by showing that the resonant frequency of this LC resonator is ω_{∞} . Note that (8.44) indicates that $\omega_{\infty} > \omega_{c}$, so infinite attenuation occurs after the cutoff frequency, ω_{c} , as illustrated in Figure 8.14. The position of the pole at ω_{∞} can be controlled with the value of m.

We now have a very sharp cutoff response, but one problem with the *m*-derived section is that its attenuation decreases for $\omega > \omega_{\infty}$. Since it is often desirable to have infinite attenuation as $\omega \to \infty$, the *m*-derived section can be cascaded with a constant-k section to give the composite attenuation response shown in Figure 8.14.

The *m*-derived *T*-section was designed so that its image impedance was identical to that of the constant-k section (independent of *m*), so we still have the problem of a nonconstant image impedance. But the image impedance of the π -equivalent will depend on *m*, and this extra degree of freedom can be used to design an optimum matching section.

The easiest way to obtain the corresponding π -section is to consider it as a piece of an infinite cascade of *m*-derived *T*-sections, as shown in Figure 8.15a,b. Then the







FIGURE 8.15 Development of an *m*-derived π -section. (a) Infinite cascade of *m*-derived *T*-sections. (b) A de-embedded π -equivalent.

image impedance of this network is, using the results of Table 8.1 and (8.35),

$$\mathbf{Z}_{int} = \frac{Z_1' Z_2'}{Z_{if}} = \frac{Z_1 Z_2 + Z_1^2 (1 - m^2)/4}{R_0 \sqrt{1 - (\omega/\omega_c)^2}}.$$
8.45

Now $Z_1Z_2 = L/C = R_0^2$ and $Z_1^2 = -\omega^2 L^2 = -4R_0^2(\omega/\omega_c)^2$, so (8.45) reduces to

$$Z_{i\pi} = \frac{1 - (1 - m^2)(\omega/\omega_c)^2}{\sqrt{1 - (\omega/\omega_c)^2}} R_0.$$
 8.46

Since this impedance is a function of m, we can choose m to minimize the variation of $Z_{i\pi}$ over the passband of the filter. Figure 8.16 shows this variation with frequency for several values of m; a value of m = 0.6 generally gives the best results.

This type of *m*-derived section can then be used at the input and output of the filter to provide a nearly constant impedance match to and from R_0 . But the image impedance of the constant-*k* and *m*-derived *T*-sections, Z_{iT} , does not match $Z_{i\pi}$; this problem can be surmounted by bisecting the π -sections, as shown in Figure 8.17. The image impedances of this circuit are $Z_{i1} = Z_{iT}$ and $Z_{i2} = Z_{i\pi}$, which can be shown by finding its ABCD parameters:

$$A = 1 + \frac{Z_1'}{4Z_2'}, \qquad 8.47a$$

$$B=\frac{Z_1'}{2},\qquad \qquad 8.47b$$



FIGURE 8.16 Variation of $Z_{i\pi}$ in the passband of a low-pass *m*-derived section for various values of *m*.





$$C = \frac{1}{2Z_2'}, \qquad \qquad 8.47c$$

$$D=1, 8.47d$$

and then using (8.27) for Z_{i1} and Z_{i2} :

$$Z_{i1} = \sqrt{Z_1' Z_2' + \frac{Z_1'^2}{4}} = Z_{iT}.$$
 8.48a

$$Z_{22} = \sqrt{\frac{Z_1' Z_2'}{1 + Z_1' / 4 Z_2'}} = \frac{Z_1' Z_2'}{Z_{iT}} = Z_{in}, \qquad 8.48b$$

where (8.40) has been used for Z_{iT} .

Composite Filters

By combining in cascade the constant-k, m-derived sharp cutoff, and the m-derived matching sections we can realize a filter with the desired attenuation and matching properties. This type of design is called a composite filter, and is shown in Figure 8.18. The



FIGURE 8.18 The final four-stage composite filter.

sharp-cutoff section, with m < 0.6, places an attenuation pole near the cutoff frequency to provide a sharp attenuation response; the constant-k section provides high attenuation further into the stopband. The bisected- π sections at the ends of the filter match the nominal source and load impedance, R_0 , to the internal image impedances, Z_{iT} , of the constant-k and m-derived sections. Table 8.2 summarizes the design equations for lowand high-pass composite filters; notice that once the cutoff frequency and impedance are specified, there is only one degree of freedom (the value of m for the sharp-cutoff section) left to control the filter response. The following example illustrates the design procedure.

Ø

EXAMPLE 8.2 Low-Pass Composite Filter Design

Design a low-pass composite filter with a cutoff frequency of 2 MHz and impedance of 75 Ω . Place the infinite attenuation pole at 2.05 MHz, and plot the frequency response from 0 to 4 MHz.

Solution

All the component values can be found from Table 8.2. For the constant-k section:

$$L = \frac{2R_0}{\omega_c} = 11.94 \ \mu \text{H}, \qquad C = \frac{2}{R_0 \omega_c} = 2.122 \ \text{nF}.$$

For the *m*-derived sharp-cutoff section:

$$m = \sqrt{1 - \left(\frac{f_c}{f_{\infty}}\right)^2} = 0.2195,$$
$$\frac{mL}{2} = 1.310 \ \mu\text{H},$$
$$mC = 465.8 \ \text{pF},$$
$$\frac{1 - m^2}{4m}L = 12.94 \ \mu\text{H}.$$



TABLE 8.2 Summary of Composite Filter Design

For the m = 0.6 matching sections:

$$\frac{mL}{2} \approx 3.582 \ \mu\text{H},$$
$$\frac{mC}{2} = 636.5 \ \text{pF},$$
$$\frac{1-m^2}{2m}L = 6.368 \ \mu\text{H}.$$



FIGURE 8.19 Low-pass composite filter for Example 8.2.



FIGURE 8.20 Prequency response for the low-pass filter of Example 8.2.

The completed filter circuit is shown in Figure 8.19; the series pairs of inductors between the sections can be combined. Figure 8.20 shows the resulting frequency response for $|S_{12}|$. Note the sharp dip at f = 2.05 MHz due to the m = 0.2195 section, and the pole at 2.56 MHz, which is due to the m = 0.6matching sections.

8.3 FILTER DESIGN BY THE INSERTION LOSS METHOD

The perfect filter would have zero insertion loss in the passband, infinite attenuation in the stopband, and a linear phase response (to avoid signal distortion) in the passband. Of course, such filters do not exist in practice, so compromises must be made; herein lies the art of filter design.

The image parameter method of the previous section may yield a usable filter response, but if not there is no clear-cut way to improve the design. The insertion loss method, however, allows a high degree of control over the passband and stopband amplitude and phase characteristics, with a systematic way to synthesize a desired response. The necessary design trade-offs can be evaluated to best meet the application requirements. If, for example, a minimum insertion loss is most important, a binomial response could be used; a Chebyshev response would satisfy a requirement for the sharpest cutoff. If it is possible to sacrifice the attenuation rate, a better phase response can be obtained by using a linear phase filter design. And in all cases, the insertion loss method allows filter performance to be improved in a straightforward manner, at the expense of a higher order filter. For the filter prototypes to be discussed below, the order of the filter is equal to the number of reactive elements.

Characterization by Power Loss Ratio

In the insertion loss method a filter response is defined by its insertion loss, or *power* loss ratio, P_{LR} :

$$P_{\rm LR} = \frac{\text{Power available from source}}{\text{Power delivered to load}} = \frac{P_{\rm inc}}{P_{\rm load}} = \frac{1}{1 - |\Gamma(\omega)|^2}.$$
 8.49

Observe that this quantity is the reciprocal of $|S_{12}|^2$ if both load and source are matched. The insertion loss (IL) in dB is

$$\mathbf{IL} = 10 \log P_{\mathrm{LR}}.$$

From Section 4.1 we know that $|\Gamma(\omega)|^2$ is an even function of ω ; therefore it can be expressed as a polynomial in ω^2 . Thus we can write

$$|\Gamma(\omega)|^2 = \frac{M(\omega^2)}{M(\omega^2) + N(\omega^2)},$$
8.51

where M and N are real polynomials in ω^2 . Substituting this form in (8.49) gives the following:

$$P_{\rm LR} = 1 + \frac{M(\omega^2)}{N(\omega^2)}.$$
 8.52

Thus, for a filter to be physically realizable its power loss ratio must be of the form in (8.52). Notice that specifying the power loss ratio simultaneously constrains the reflection coefficient, $\Gamma(\omega)$. We now discuss some practical filter responses.

Maximally flat. This characteristic is also called the binomial or Butterworth response, and is optimum in the sense that it provides the flattest possible passband response for a given filter complexity, or order. For a low-pass filter, it is specified by

$$P_{\rm LR} = 1 + k^2 \left(\frac{\omega}{\omega_c}\right)^{2N},$$
8:53

where N is the order of the filter, and ω_c is the cutoff frequency. The passband extends from $\omega = 0$ to $\omega = \omega_c$; at the band edge the power loss ratio is $1 + k^2$. If we choose

this as the -3 dB point, as is common, we have k = 1, which we will assume from now on. For $\omega > \omega_c$, the attenuation increases monotonically with frequency, as shown in Figure 8.21. For $\omega >> \omega_c$, $P_{LR} \simeq k^2 (\omega/\omega_c)^{2N}$, which shows that the insertion loss increases at the rate of 20N dB/decade. Like the binomial response for multisection quarter-wave matching transformers, the first (2N - 1) derivatives of (8.53) are zero at $\omega = 0$.

Equal ripple. If a Chebyshev polynomial is used to specify the insertion loss of an *N*-order low-pass filter as

$$P_{\rm LR} = 1 + k^2 T_N^2 \left(\frac{\omega}{\omega_c}\right), \qquad 8.54$$

then a sharper cutoff will result, although the passband response will have ripples of amplitude $1+k^2$, as shown in Figure 8.21, since $T_N(x)$ oscillates between ± 1 for $|x| \leq 1$. Thus, k^2 determines the passband ripple level. For large x, $T_N(x) \simeq 1/2(2x)^N$, so for $\omega >> \omega_c$ the insertion loss becomes

$$P_{
m LR}\simeq rac{k^2}{4}\left(rac{2\omega}{\omega_c}
ight)^{2N},$$

which also increases at the rate of 20N dB/decade. But the insertion loss for the Chebyshev case is $(2^{2N})/4$ greater than the binomial response, at any given frequency where $\omega >> \omega_c$.

Elliptic function. The maximally flat and equal-ripple responses both have monotonically increasing attenuation in the stopband. In many applications it is adequate to specify a minimum stopband attenuation, in which case a better cutoff rate can be obtained. Such filters are called elliptic function filters [3], and have equal-ripple responses in the passband as well as the stopband, as shown in Figure 8.22. The maximum attenuation in the passband, A_{max} , can be specified, as well as the minimum attenuation



FIGURE 8.21 Maximally flat and equal-ripple low-pass filter responses (N = 3).



FIGURE 8.22 Elliptic function low-pass filter response.

in the stopband, A_{min} . Elliptic function filters are difficult to synthesize, so we will not consider them further: the interested reader is referred to reference [3].

Linear phase. The above filters specify the amplitude response, but in some applications (such as multiplexing filters for communication systems) it is important to have a linear phase response in the passband to avoid signal distortion. It turns out that a sharp-cutoff response is generally incompatible with a good phase response, so the phase response of the filter must be deliberately synthesized, usually resulting in an inferior amplitude cutoff characteristic. A linear phase characteristic can be achieved with the following phase response:

$$\phi(\omega) = A\omega \left[1 + p \left(\frac{\omega}{\omega_c} \right)^{2N} \right], \qquad 8.55$$

where $\phi(\omega)$ is the phase of the voltage transfer function of the filter, and p is a constant. A related quantity is the group delay, defined as

$$\tau_d = \frac{d\phi}{d\omega} = A \left[1 + p(2N+1) \left(\frac{\omega}{\omega_c} \right)^{2N} \right], \qquad 8.56$$

which shows that the group delay for a linear phase filter is a maximally flat function.

More general filter specifications can be obtained, but the above cases are the most common. We will next discuss the design of low-pass filter prototypes which are normalized in terms of impedance and frequency; this normalization simplifies the design of filters for arbitrary frequency, impedance, and type (low-pass, high-pass, bandpass, or bandstop). The low-pass prototypes are then scaled to the desired frequency and impedance; and the lumped-element components replaced with distributed circuit elements for implementation at microwave frequencies. This design process is illustrated in Figure 8.23.



FIGURE 8.23 The process of filter design by the insertion loss method.

Maximally Flat Low-Pass Filter Prototype

Consider the two-element low-pass filter prototype shown in Figure 8.24; we will derive the normalized element values, L and C, for a maximally flat response. We assume a source impedance of 1 Ω , and a cutoff frequency $\omega_c = 1$. From (8.53), the desired power loss ratio will be, for N = 2,

$$P_{\mathbf{l},\mathbf{R}} = 1 + \omega^4. \tag{8.57}$$

The input impedance of this filter is

$$Z_{
m in}=j\omega L+rac{R(1-j\omega RC)}{1+\omega^2 R^2 C^2}.$$
 8.58

Since

$$\Gamma \equiv rac{Z_{
m in}-1}{Z_{
m in}+1},$$

the power loss ratio can be written as

$$P_{\rm LR} = \frac{l}{1 - |\Gamma|^2} = \frac{l}{1 - [(Z_{\rm in} - 1)/(Z_{\rm in}^* - 1)][(Z_{\rm in}^* - 1)/(Z_{\rm in}^* + 1)]} = \frac{|Z_{\rm in} + 1|^2}{2(Z_{\rm in} + Z_{\rm in}^*)}.$$
8.59

Now,

$$Z_{
m in} + Z_{
m in}^{*} = rac{2R}{1+\omega^2 R^2 C^2},$$

and

$$|Z_{\mathrm{in}}+1|^2 = \left(rac{R}{1+\omega^2 R^2 C^2}+1
ight)^2 + \left(\omega L - rac{\omega C R^2}{1+\omega^2 R^2 C^2}
ight)^2,$$



FIGURE 8.24 Low-pass filter prototype, N = 2.

so (8.59) becomes

$$P_{LR} = \frac{1 + \omega^2 R^2 C^2}{4R} \left[\left(\frac{R}{1 + \omega^2 R^2 C^2} + 1 \right)^2 + \left(\omega L - \frac{\omega C R^2}{1 + \omega^2 R^2 C^2} \right)^2 \right]$$

= $\frac{1}{4R} (R^2 + 2R + 1 + R^2 \omega^2 C^2 + \omega^2 L^2 + \omega^4 L^2 C^2 R^2 - 2\omega^2 L C R^2)$
= $1 + \frac{1}{4R} \left[(1 - R)^2 + (R^2 C^2 + L^2 - 2L C R^2) \omega^2 + L^2 C^2 R^2 \omega^4 \right].$ 8.60

Notice that this expression is a polynomial in ω^2 . Comparing to the desired response of (8.57) shows that R = 1, since $P_{LR} = 1$ for $\omega = 0$. In addition, the coefficient of ω^2 must vanish, so

$$C^2 + L^2 - 2LC = (C - L)^2 = 0,$$

or L = C. Then for the coefficient of ω^4 to be unity we must have

$$\frac{1}{4}L^2C^2 = \frac{1}{4}L^4 = 1,$$
$$L = C = \sqrt{2}.$$

OF

In principle, this procedure can be extended to find the element values for filters with an arbitrary number of elements, N, but clearly this is not practical for large N. For a normalized low-pass design where the source impedance is 1 Ω and the cutoff frequency is $\omega_c = 1$, however, the element values for the ladder-type circuits of Figure 8.25 can





N	g_1	g_2	<i>9</i> 3	<u>9</u> 4	g_5	g_b	g_7	$g_{\mathtt{B}}$	g_9	g 10	g_{11}
1	2,0000	1.0000									
2	1.4142	1.4142	1.0000								
3	1.0000	2.0000	1.0000	1.0000							
4	0.7654	1.8478	1.8478	0.7654	1.0000						
5	0.6180	1.6180	2.0000	1.6180	0.6180	1.0000					
6	0.5176	1.4142	1.9318	1.9318	1.4142	0.5176	0000.1				
7	0.4450	1.2470	1.8019	2.0000	1.8019	1.2470	0.4450	1.0000			
8	0.3902	1.1111	1.6629	1.9615	1.9615	1.6629	1.1111	0.3902	1.0000		
9	0.3473	1.0000	1.5321	1,8794	2.0000	1.8794	1.5321	1.0000	0.3473	1.0000	
0	0.3129	0.9080	1.4142	1.7820	1.9754	l.9754	1.7820	1.4142	0.9080	0.3129	1.0000

TABLE 8.3 Element Values for Maximally Flat Low-Pass Filter Prototypes ($g_0 = 1$, $\omega_c = 1$, N = 1 to 10)

Source: Reprinted from G. L. Matthaei, L. Young, and E. M. T. Jones, *Microwave Filters, Impedance-Matching Networks, and Coupling Structures* (Dedham, Mass.: Artech House, 1980) with permission.

be tabulated [1]. Table 8.3 gives such element values for maximally flat low-pass filter prototypes for N = 1 to 10. (Notice that the values for N = 2 agree with the above analytical solution.) This data is used with either of the ladder circuits of Figure 8.25 in the following way. The element values are numbered from g_0 at the generator impedance to g_{N+1} at the load impedance, for a filter having N reactive elements. The elements alternate between series and shunt connections, and g_k has the following definition:

$$g_0 = \begin{cases} \text{generator resistance (network of Figure 8.25a)} \\ \text{generator conductance (network of Figure 8.25b)} \\ g_k \\ g_{k=1 \text{ to } N} \end{cases} = \begin{cases} \text{inductance for series inductors} \\ \text{capacitance for shunt capacitors} \\ g_{N+1} \end{cases} = \begin{cases} \text{load resistance if } g_N \text{ is a shunt capacitor} \\ \text{load conductance if } g_N \text{ is a series inductor} \end{cases}$$

Then the circuits of Figure 8.25 can be considered as the dual of each other, and both will give the same response.

Finally, as a matter of practical design procedure, it will be necessary to determine the size, or order, of the filter. This is usually dictated by a specification on the insertion loss at some frequency in the stopband of the filter. Figure 8.26 shows the attenuation characteristics for various N, versus normalized frequency. If a filter with N > 10 is required, a good result can usually be obtained by cascading two designs of lower order.



EXAMPLE 8.3 Low-Pass Filter Design

A maximally flat low-pass filter is to be designed with a cutoff frequency of 8 GHz and a minimum attenuation of 20 dB at 11 GHz. How many filter elements are required?



FIGURE 8.26 Attenuation versus normalized frequency for maximally flat filter prototypes. Adapted from G. L. Matthaei, L. Young, and E. M. T. Jones, *Microwave Filters, Impedance-Matching Networks, and Coupling Structures* (Dedham, Mass.: Artech House, 1980) with permission.

Solution We have $\omega/2\pi = 11$ GHz and $\omega_c/2\pi = 8$ GHz, so

$$\left|\frac{\omega}{\omega_c}\right| - 1 = \frac{11}{8} - 1 = 0.375.$$

Then from Figure 8.26 we see that an attenuation of 20 dB at this frequency requires that $N \ge 8$. Further design details will be discussed in Section 8.4. \bigcirc

Equal-Ripple Low-Pass Filter Prototype

For an equal-ripple low-pass filter with a cutoff frequency $\omega_c = 1$, the power loss ratio from (8.54) is

$$P_{\rm LR} = 1 + k^2 T_N^2(\omega), \qquad 8.61$$

where $1 + k^2$ is the ripple level in the passband. Since the Chebyshev polynomials have

the property that

$$T_N(0) = \begin{cases} 0 & \text{for } N \text{ odd,} \\ 1 & \text{for } N \text{ even,} \end{cases}$$

equation (8.61) shows that the fifter will have a unity power loss ratio at $\omega = 0$ for N odd, but a power loss ratio of $1 + k^2$ at $\omega = 0$ for N even. Thus, there are two cases to consider, depending on N.

For the two-element filter of Figure 8.24, the power loss ratio is given in terms of the component values in (8.60). From (5.56b), we see that $T_2(x) = 2x^2 - 1$, so equating (8.61) to (8.60) gives

$$1 + k^{2}(4\omega^{4} - 4\omega^{2} + 1) = 1 + \frac{1}{4R}[(1 - R)^{2} + (R^{2}C^{2} + L^{2} - 2LCR^{2})\omega^{2} + L^{2}C^{2}R^{2}\omega^{4}],$$
8.62

which can be solved for R, L, and C if the ripple level (as determined by k^2) is known. Thus, at $\omega = 0$ we have that

$$k^{2} = \frac{(1-R)^{2}}{4R},$$

$$R = 1 + 2k^{2} \pm 2k\sqrt{1+k^{2}} \quad \text{(for } N \text{ even).}$$
8.63

or

Equating coefficients of ω^2 and ω^4 yields the additional relations,

$$4k^{2} = \frac{1}{4R}L^{2}C^{2}R^{2},$$

$$-4k^{2} = \frac{1}{4R}(R^{2}C^{2} + L^{2} - 2LCR^{2}),$$

which can be used to find L and C. Note that (8.63) gives a value for R that is not unity, so there will be an impedance mismatch if the load actually has a unity (normalized) impedance; this can be corrected with a quarter-wave transformer, or by using an additional filter element to make N odd. For odd N, it can be shown that R = 1. (This is because there is a unity power loss ratio at $\omega = 0$ for N odd.)

Tables exist for designing equal-ripple low-pass filters with a normalized source impedance and cutoff frequency ($\omega_c' = 1$) [1], and can be applied to either of the ladder circuits of Figure 8.25. This design data depends on the specified passband ripple level; Table 8.4 lists element values for normalized low-pass filter prototypes having 0.5 dB or 3.0 dB ripple, for N = 1 to 10. Notice that the load impedance $g_{N+1} \neq 1$ for even N. If the stopband attenuation is specified, the curves in Figures 8.27a,b can be used to determine the necessary value of N for these ripple values.

Linear Phase Low-Pass Filter Prototypes

Filters having a maximally flat time delay, or a linear phase response, can be designed in the same way, but things are somewhat more complicated because the phase of the voltage transfer function is not as simply expressed as is its amplitude. Design

					0.5 -	dB Ripple	!			_	
N	g_1	g_2	g_3	g_4	Y 5	g_6	<i>g</i> 7	g_8	<i>9</i> 9	9 10	g_{11}
1	0.6986	1.0000			_						
2	1.4029	0.7071	1.9841								
3	1.5963	1.0967	1.5963	1.0000							
4	1.6703	1.1926	2.3661	0.8419	1.9841						
5	l.7058	1.2296	2.5408	1.2296	1.7058	1.0000					
6	1.7254	1.2479	2.6064	1.3137	2.4758	0.8696	1.9841				
7	1.7372	1.2583	2.6381	1.3444	2.6381	1.2583	1.7372	1.000			
8	1.7451	1.2647	2.6564	1.3590	2.6964	1.3389	2.5093	0.8796	1.9841		
9	1.7504	1.2690	2.6678	1.3673	2.7239	1. 36 73	2.6678	1.2690	1.7504	1.0000	
10	1.7543	1.2721	2.6754	1.3725	2,7392	1.3806	2.7231	1.3485	2.5239	0.8842	1.9841
					3,0 0	dB Ripple	:		·		
N	<u></u> g1	g_2	<i>g</i> 3	g_4	<u>g</u> 5	g_6	g 7	g_8	<i>9</i> 9	$m{g}_{10}$	g_{11}
1	1,9953	1.0000									
2	3.1013	0.5339	5.8095								
3	3.3487	0.7117	3,3487	1.0000							
4	3.4389	0.7483	4.3471	0.5920	5.8095						
5	3,4817	0.7618	4.5381	0.7618	3.4817	1.0000					
6	3,5045	0.7685	4.6061	0.7929	4,4641	0.6033	5.8095				
7	3.5182	0.7723	4.6386	0.8039	4.6386	0.7723	3.5182	1.0000			
8	3.5277	0.7745	4.6575	0.8089	4.6990	0.8018	4.4990	0.6073	5.8095		
9	3.5340	0.7760	4.6692	0.8118	4.7272	0.8118	4.6692	0.7760	3.5340	1.0000	
10	3.5384	0.7771	4.6768	0.8136	4.7425	0.8164	4.72 60	0.8 051	4.5142	0.6091	5.8095

TABLE 8.4 Element Values for Equal-Ripple Low-Pass Filter Prototypes ($g_0 = 1$, $\omega_c = 1$, N = 1 to 10, 0.5 dB and 3.0 dB ripple)

Source: Reprinted from G. L. Matthaci, L. Young, and E. M. T. Jones, Microwave Filters, Impedance-Matching Networks, and Coupling Structures (Dedham, Mass.: Artech House, 1980) with permission.

values have been derived for such filters [1], however, again for the ladder circuits of Figure 8.25, and are given in Table 8.5 for a normalized source impedance and cutoff frequency ($\omega'_c = 1$). The resulting group delay in the passband will be $\tau_d = 1/\omega'_c = 1$.

8.4 ____ FILTER TRANSFORMATIONS

The low-pass filter prototypes of the previous section were normalized designs having a source impedance of $R_s = 1 \ \Omega$ and a cutoff frequency of $\omega_c = 1$. Here we show how these designs can be scaled in terms of impedance and frequency, and converted to give high-pass, bandpass, or bandstop characteristics. Several examples will be presented to illustrate the design procedure.


FIGURE 8.27

Attenuation versus normalized frequency for equal-ripple filter prototypes. (a) 0.5 dB ripple level. (b) 3.0 dB ripple level.

Adapted from G. L. Matthaet, L. Young, and E. M. T. Iones, *Microwave Filters, Impedance-Matching Networks, and Coupling Structures* (Dedham, Mass.: Artech House, 1980) with permission.

		· -, -		-							
N	<i>g</i> ı	<i>g</i> 2	<i>y</i> 3	∯4	g_5	<u>g</u> 6	<i>g</i> 7	<i>9</i> 8	<i>9</i> 9	g 10	911
1	2.0000	1.0000									
2	1.5774	0.4226	i.000 0								
3	1.2550	0.5528	0.1922	1.0000							
4	1.0598	0.5116	0.3181	0.1104	1.0000						
5	0.9303	0.4577	0.3312	0.2090	0.0718	1.0000					
6	0.8377	0.4116	0.3158	0.2364	0.1480	0.0505	1.0000				
7	0.7677	0.3744	0.2944	0.2378	0.1778	0.1104	0.0375	1.0000			
8	0.7125	0.3446	0.2735	0.2297	0.1867	0.1387	0.0855	0.0289	1.0000		
9	0.6678	0.3203	0.2547	0.2184	0.1859	0.1506	0.1111	0.0682	0.0230	1.0000	
10	0.6305	0.3002	0.2384	0.2066	0.1808	0.1539	0.1240	0.0911	0.0557	0.0187	0000.1
											-

TABLE 8.5 Element Values for Maximally Flat Time Delay Low-Pass Filter Prototypes ($g_0 = 1$, $w_c = 1$, N = 1 to 10)

Source: Reprinted from G. L. Matthaei, L. Young, and E. M. T. Jones, Microwave Filters, Impedance-Matching Networks, and Coupling Structures (Dedham, Mass.: Artech House, 1980) with permission.

Impedance and Frequency Scaling

Impedance scaling. In the prototype design, the source and load resistances are unity (except for equal-ripple filters with even N, which have nonunity load resistance). A source resistance of R_0 can be obtained by multiplying the impedances of the prototype design by R_0 . Then, if we let primes denote impedance scaled quantities, we have the new filter component values given by

$$L' = R_0 L, \qquad 8.64a$$

$$C' = \frac{C}{R_0}, \qquad 8.64b$$

$$R'_s = R_0, \qquad \qquad 8.64c$$

$$R_L' = R_0 R_L, \qquad 8.64d$$

where L, C, and R_L are the component values for the original prototype.

Frequency scaling for low-pass filters. To change the cutoff frequency of a lowpass prototype from unity to ω_c requires that we scale the frequency dependence of the filter by the factor $1/\omega_c$, which is accomplished by replacing ω by ω/ω_c :

$$\omega \leftarrow \frac{\omega}{\omega_c}$$
. 8.65

Then the new power loss ratio will be

$$P_{\rm LR}^{\prime}(\omega) = P_{\rm LR}\left(\frac{\omega}{\omega_c}\right),$$

where ω_c is the new cutoff frequency: cutoff occurs when $\omega/\omega_c = 1$, or $\omega = \omega_c$. This transformation can be viewed as a stretching, or expansion, of the original passband, as illustrated in Figure 8.28a.b.

The new element values are determined by applying the substitution of (8.65) to the series reactances, $j\omega L_k$, and shunt susceptances, $j\omega C_k$, of the prototype filter. Thus,

$$egin{aligned} jX_k &= jrac{\omega}{\omega_c}L_k = j\omega L_k',\ jB_k &= jrac{\omega}{\omega_c}C_k = j\omega C_k', \end{aligned}$$

which shows that the new element values are given by

$$L'_k = \frac{L_k}{\omega_c},$$
 8.66a

$$C'_k = \frac{C_k}{\omega_c}.$$
 8.66b

When both impedance and frequency scaling are required, the results of (8.64) can be combined with (8.66) to give

$$L'_{k} = \frac{R_0 L_k}{\omega_c}, \qquad \qquad 8.67a$$

$$C_k' = \frac{C_k}{R_0 \omega_c}.$$
8.67b

Low-pass to high-pass transformation. The frequency substitution where,

$$\omega \leftarrow -\frac{\omega_c}{\omega},$$
 8.68

can be used to convert a low-pass response to a high-pass response, as shown in Figure 8.28c. This substitution maps $\omega = 0$ to $\omega = \pm \infty$, and vice versa; cutoff occurs when



FIGURE 8.28 Frequency scaling for low-pass filters and transformation to a high-pass response. (a) Low pass filter prototype response for $\omega_c = 1$. (b) Frequency scaling for low-pass response. (c) Transformation to high-pass response.

Chapter 8: Microwave Filters

 $\omega = \pm \omega_c$. The negative sign is needed to convert inductors (and capacitors) to realizable capacitors (and inductors). Applying (8.68) to the series reactances, $j\omega L_k$, and the shunt susceptances, $j\omega C_k$, of the prototype filter gives

$$egin{aligned} jX_k &= -jrac{\omega_c}{\omega}L_k = rac{1}{j\omega C_k'}, \ jB_k &= -jrac{\omega_c}{\omega}C_k = rac{1}{j\omega L_k'}, \end{aligned}$$

which shows that series inductors L_k must be replaced with capacitors C'_k , and shunt capacitors C_k must be replaced with inductors L'_k . The new component values are given by

$$C_k' = \frac{1}{\omega_c L_k},$$

$$L'_k = \frac{1}{\omega_c C_k}.$$
 8.696

Impedance scaling can be included by using (8.64) to give

$$C'_{k} = \frac{1}{R_{0}\omega_{c}L_{k}}.$$
8.70a

$$L'_k = \frac{R_0}{\omega_c C_k}.$$
8.70b



EXAMPLE 8.4 Low-Pass Filter Design Comparison

Design a maximally flat low-pass filter with a cutoff frequency of 2 GHz, impedance of 50 Ω , and at least 15 dB insertion loss at 3 GHz. Compute and plot the amplitude response and group delay for f = 0 to 4 GHz, and compare with an equal-rapple (3.0 dB ripple) and linear phase filter having the same order.

Solution

First find the required order of the maximally flat filter to satisfy the insertion loss specification at 3 GHz. We have that $|\omega/\omega_c| - 1 = 0.5$; from Figure 8.26 we see that N = 5 will be sufficient. Then Table 8.3 gives the prototype element values as

 $g_1 = 0.618,$ $g_2 = 1.618,$ $g_3 = 2.000,$ $g_4 = 1.618,$ $g_5 = 0.618.$

_-- _ _



FIGURE 8.29 Low-pass maximally flat filter circuit for Example 8.4.

Then (8.67) can be used to obtain the scaled element values:

 $C'_1 = 0.984 \text{ pF},$ $L'_2 = 6.438 \text{ nH},$ $C'_3 = 3.183 \text{ pF},$ $L'_4 = 6.438 \text{ nH}.$ $C'_5 = 0.984 \text{ pF}.$

The final filter circuit is shown in Figure 8.29; the ladder circuit of Figure 8.25a was used, but that of Figure 8.25b could have been used just as well.

The component values for the equal-ripple filter and the linear phase filter, for N = 5, can be determined from Tables 8.4 and 8.5. The amplitude and group delay results for these three filters are shown in Figure 8.30. These results clearly show the trade-offs involved with the three types of filters. The equal-ripple response has the sharpest cutoff, but the worst group delay characteristics. The maximally flat response has a flatter attenuation characteristic in the passband, but a slightly lower cutoff rate. The linear phase filter has the worst cutoff rate, but a very good group delay characteristic.

Bandpass and Bandstop Transformation

Low-pass prototype filter designs can also be transformed to have the bandpass or bandstop responses illustrated in Figure 8.31. If ω_1 and ω_2 denote the edges of the passband, then a bandpass response can be obtained using the following frequency substitution:

$$\omega \longleftarrow \frac{\omega_0}{\omega_2 - \omega_1} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega} \right) = \frac{1}{\Delta} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega} \right), \qquad 8.71$$

where

$$\Delta = \frac{\omega_2 - \omega_1}{\omega_0}$$
 8.72

is the fractional bandwidth of the passband. The center frequency, ω_0 , could be chosen as the arithmetic mean of ω_1 and ω_2 , but the equations are simpler if it is chosen as the



FIGURE 8.30 Frequency response of the filter design of Example 8.4. (a) Amplitude response. (b) Group delay response.

geometric mean:

$$\omega_0 = \sqrt{\omega_1 \omega_2}.$$
 8.73

Then the transformation of (8.71) maps the bandpass characteristics of Figure 8.31b to the low-pass response of Figure 8.31a as follows:

When
$$\omega = \omega_0$$
, $\frac{1}{\Delta} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega} \right) = 0;$





When
$$\omega = \omega_1$$
, $\frac{1}{\Delta} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega} \right) = \frac{1}{\Delta} \left(\frac{\omega_1^2 - \omega_0^2}{\omega_0 \omega_1} \right) = -1;$
When $\omega = \omega_2$, $\frac{1}{\Delta} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega} \right) = \frac{1}{\Delta} \left(\frac{\omega_2^2 - \omega_0^2}{\omega_0 \omega_2} \right) = 1.$

The new filter elements are determined by using (8.71) in the expressions for the series reactance and shunt susceptances. Thus,

$$jX_{k} = \frac{j}{\Delta} \left(\frac{\omega}{\omega_{0}} - \frac{\omega_{0}}{\omega} \right) L_{k} = j\frac{\omega L_{k}}{\Delta\omega_{0}} - j\frac{\omega_{0}L_{k}}{\Delta\omega} = j\omega L_{k}' - j\frac{1}{\omega C_{k}'}$$

which shows that a series inductor, L_k , is transformed to a series LC circuit with element values,

$$L'_{k} = \frac{L_{k}}{\Delta\omega_{0}}, \qquad \qquad 8.74a$$

$$C'_{k} = \frac{\Delta}{\omega_{0}L_{k}}.$$
8.74b

Similarly,

$$jB_{k} = \frac{j}{\Delta} \left(\frac{\omega}{\omega_{0}} - \frac{\omega_{0}}{\omega} \right) C_{k} = j \frac{\omega C_{k}}{\Delta \omega_{0}} - j \frac{\omega_{0} C_{k}}{\Delta \omega} = j \omega C_{k}' - j \frac{1}{\omega L_{k}'}.$$

which shows that a shunt capacitor, C_k , is transformed to a shunt LC circuit with element values,

$$L'_{k} = \frac{\Delta}{\omega_0 C_k}, \qquad 8.74c$$

$$C'_{k} = \frac{C_{k}}{\Delta\omega_{0}}.$$
8.74d

The low-pass filter elements are thus converted to series resonant circuits (low impedance at resonance) in the series arms, and to parallel resonant circuits (high impedance at resonance) in the shunt arms. Notice that both series and parallel resonator elements have a resonant frequency of ω_0 . The inverse transformation can be used to obtain a bandstop response. Thus,

$$\omega \longleftarrow \Delta \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega}\right)^{-1},$$
 8.75

where Δ and ω_0 have the same definitions as in (8.72) and (8.73). Then series inductors of the low-pass prototype are converted to parallel *LC* circuits having element values given by

$$C'_k = \frac{1}{\omega_0 \Delta L_k}.$$
 8.766

The shunt capacitor of the low-pass prototype is converted to series LC circuits having element values given by

$$L'_{k} = \frac{1}{\omega_0 \Delta C_{k}}, \qquad 8.76c$$

$$C'_{k} = \frac{\Delta C_{k}}{\omega_{0}}.$$
8.76d

The element transformations from a low-pass prototype to a highpass, bandpass, or bandstop filter are summarized in Table 8.6. These results do not include impedance scaling, which can be made using (8.64).

EXAMPLE 8.5 Bandpass Filter Design

Design a bandpass filter having a 0.5 dB equal-ripple response, with N = 3. The center frequency is 1 GHz, the bandwidth is 10%, and the impedance is 50 Ω .

Solution

From Table 8.4 the element values for the low-pass prototype circuit of Figure 8.25b are given as

$$g_1 = 1.5963 = L_1,$$

$$g_2 = 1.0967 = C_2,$$

$$g_3 = 1.5963 = L_3,$$

$$g_4 = 1.000 = R_L.$$

Then (8.64) and (8.74) give the impedance-scaled and frequency-transformed element values for the circuit of Figure 8.32 as

$$L'_{1} = \frac{L_{1}Z_{o}}{\omega_{o}\Delta} = 127.0 \text{ nH},$$
$$C'_{1} = \frac{\Delta}{\omega_{o}L_{1}Z_{o}} = 0.199 \text{ pF},$$



TABLE 8.6 Summary of Prototype Filter Transformations

$$L'_{2} = \frac{\Delta Z_{o}}{\omega_{o}C_{2}} = 0.726 \text{ nH},$$

$$C'_{2} = \frac{C_{2}}{\omega_{o}\Delta Z_{o}} = 34.91 \text{ pF}.$$

$$L'_{3} = \frac{L_{3}Z_{o}}{\omega_{o}\Delta} = 127.0 \text{ nH},$$

$$C'_{3} = \frac{\Delta}{\omega_{o}L_{3}Z_{o}} = 0.199 \text{ pF}.$$

The resulting amplitude response is shown in Figure 8.33.





FIGURE 8.32 Bandpass filter circuit for Example 8.5.



FIGURE 8.33 Amplitude response for the bandpass filter of Example 8.5.

8.5 FILTER IMPLEMENTATION

The lumped-element filter design discussed in the previous sections generally works well at low frequencies, but two problems arise at microwave frequencies. First, lumped elements such as inductors and capacitors are generally available only for a limited range of values and are difficult to implement at microwave frequencies, but must be approximated with distributed components. In addition, at microwave frequencies the distances between filter components is not negligible. Richard's transformation is used to convert lumped elements to transmission line sections, while Kuroda's identities can be used to separate filter elements by using transmission line sections. Because such additional transmission line sections do not affect the filter response, this type of design is called *redundant* filter synthesis. It is possible to design microwave filters that take advantage of these sections to improve the filter response [4]; such *nonredundant* synthesis does not have a lumped-element counterpart.

Richard's Transformation

The transformation,

$$\Omega = \tan \beta \ell = \tan \left(\frac{\omega \ell}{v_p}\right), \qquad 8.77$$

maps the ω plane to the Ω plane, which repeats with a period of $\omega \ell / v_p = 2\pi$. This transformation was introduced by P. Richard [6] to synthesize an LC network using

open- and short-circuited transmission lines. Thus, if we replace the frequency variable ω with Ω , the reactance of an inductor can be written as

$$jX_L = j\,\Omega L = jL\,\tan\beta\ell, \qquad 8.78a$$

and the susceptance of a capacitor can be written as

$$jB_C = j\Omega C = jC \tan \beta \ell.$$
 8.78b

These results indicate that an inductor can be replaced with a short-circuited stub of length $\beta \ell$ and characteristic impedance L, while a capacitor can be replaced with an opencircuited stub of length $\beta \ell$ and characteristic impedance 1/C. A unity filter impedance is assumed.

Cutoff occurs at unity frequency for a low-pass filter prototype; to obtain the same cutoff frequency for the Richard's-transformed filter, (8.77) shows that

$$\Omega = 1 = \tan \beta \ell,$$

which gives a stub length of $\ell = \lambda/8$, where λ is the wavelength of the line at the cutoff frequency, ω_c . At the frequency $\omega_0 = 2\omega_c$, the lines will be $\lambda/4$ long, and an attenuation pole will occur. At frequencies away from ω_c , the impedances of the stubs will no longer match the original lumped-element impedances, and the filter response will differ from the desired prototype response. Also, the response will be periodic in frequency, repeating every $4\omega_c$.

In principle, then, the inductors and capacitors of a lumped-element filter design can be replaced with short-circuited and open-circuited stubs, as illustrated in Figure 8.34. Since the lengths of all the stubs are the same ($\lambda/8$ at ω_c), these lines are called *commensurate lines*.



FIGURE 8.34

Richard's transformation. (a) For an inductor to a short-circuited stub. (b) For a capacitor to an open-circuited stub.

Kuroda's Identities

The four Kuroda identities use redundant transmission line sections to achieve a more practical microwave filter implementation by performing any of the following operations:

- Physically separate transmission line stubs
- Transform series stubs into shunt stubs, or vice versa
- · Change impractical characteristic impedances into more realizable ones

The additional transmission line sections are called *unit elements* and are $\lambda/8$ long at ω_c ; the unit elements are thus commensurate with the stubs used to implement the inductors and capacitors of the prototype design.

The four identities are illustrated in Table 8.7, where each box represents a unit element, or transmission line, of the indicated characteristic impedance and length ($\lambda/8$ at ω_c). The inductors and capacitors represent short-circuit and open-circuit stubs, respectively. We will prove the equivalence of the first case, and then show how to use these identities in Example 8.6.





8.5 Filter Implementation

The two circuits of identity (a) in Table 8.7 can be redrawn as shown in Figure 8.35; we will show that these two networks are equivalent by showing that their ABCD matrices are identical. From Table 4.1, the ABCD matrix of a length ℓ of transmission line with characteristic impedance Z_1 is

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix} = \begin{bmatrix} \cos \beta \ell & j Z_1 \sin \beta \ell \\ \frac{j}{Z_1} \sin \beta \ell & \cos \beta \ell \end{bmatrix} = \frac{1}{\sqrt{1+\Omega^2}} \begin{bmatrix} 1 & j \Omega Z_1 \\ \frac{j \Omega}{Z_1} & 1 \end{bmatrix}, \quad 8.79$$

where $\Omega = \tan \beta \ell$. Now the open-circuited short stub in the first circuit in Figure 8.35 has an impedance of $-jZ_2 \cot \beta \ell = -jZ_2/\Omega$, so the *ABCD* matrix of the entire circuit is

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{L} = \begin{bmatrix} 1 & 0 \\ \frac{j\Omega}{Z_{2}} & 1 \end{bmatrix} \begin{bmatrix} 1 & j\Omega Z_{1} \\ \frac{j\Omega}{Z_{1}} & 1 \end{bmatrix} \frac{1}{\sqrt{1+\Omega^{2}}}$$
$$= \frac{1}{\sqrt{1+\Omega^{2}}} \begin{bmatrix} 1 & j\Omega Z_{1} \\ j\Omega \left(\frac{1}{Z_{1}} + \frac{1}{Z_{2}}\right) & 1 - \Omega^{2} \frac{Z_{1}}{Z_{2}} \end{bmatrix}.$$
 8.80a

The short-circuited series stub in the second circuit in Figure 8.35 has an impedance of $j(Z_1/n^2) \tan \beta \ell = j(\Omega Z_1/n^2)$, so the ABCD matrix of the entire circuit is

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{R} = \begin{bmatrix} 1 & j\frac{\Omega Z_{2}}{n^{2}} \\ \frac{j\Omega n^{2}}{Z_{2}} & 1 \end{bmatrix} \begin{bmatrix} 1 & \frac{j\Omega Z_{1}}{n^{2}} \\ 0 & 1 \end{bmatrix} \frac{1}{\sqrt{1+\Omega^{2}}}$$
$$= \frac{1}{\sqrt{1+\Omega^{2}}} \begin{bmatrix} 1 & \frac{j\Omega}{n^{2}}(Z_{1}+Z_{2}) \\ \frac{j\Omega n^{2}}{Z_{2}} & 1-\Omega^{2}\frac{Z_{1}}{Z_{2}} \end{bmatrix}.$$
8.80b

The results in (8.80a) and (8.80b) are identical if we choose $n^2 = 1 + Z_2/Z_1$. The other identities in Table 8.7 can be proved in the same way.



FIGURE 8.35 Equivalent circuits illustrating Kuroda identity (a) in Table 8.7.



EXAMPLE 8.6 Low-Pass Filter Design Using Stubs

Design a low-pass filter for fabrication using microstrip lines. The specifications are: cutoff frequency of 4 GHz, third order, impedance of 50 Ω , and a 3 dB equal-ripple characteristic.

Solution

From Table 8.4, the normalized low-pass prototype element values are

$$g_1 = 3.3487 = L_1,$$

 $g_2 = 0.7117 = C_2.$





(b)



FIGURE 8.36

Filter design procedure for Example 8.6. (a) Lumped-element low-pass filter prototype. (b) Using Richard's transformations to convert inductors and capacitors to series and shunt stubs. (c) Adding unit elements at ends of filter.

 $g_3 = 3.3487 = L_3,$ $g_4 = 1.0000 = R_L,$

with the lumped-element circuit shown in Figure 8.36a.

The pext step is to use Richard's transformations to convert series inductors to series stubs, and shunt capacitors to shunt stubs, as shown in Figure 8.36b. According to (8.78), the characteristic impedance of a series stub (inductor) is L, and the characteristic impedance of a shunt stub (capacitor) is 1/C. For commensurate line synthesis, all stubs are $\lambda/8$ long at $\omega = \omega_c$. (It is usually most convenient to work with normalized quantities until the last step in the design.)

The series stubs of Figure 8.36b would be very difficult to implement in microstrip form, so we will use one of the Kuroda identities to convert these to shunt stubs. First, we must add unit elements at either end of the filter, as shown in Figure 8.36c. These redundant elements do not affect filter performance since



FIGURE 8.36 Continued. (d) Applying the second Kuroda identity. (e) After impedance and frequency scaling. (f) Microstrip fabrication of final filter.

they are matched to the source and load ($Z_0 = 1$). Then we can apply Kuroda identity (b) from Table 8.7 to both ends of the filter. In both cases we have that

$$n^2 = 1 + \frac{Z_2}{Z_1} = 1 + \frac{1}{3.3487} = 1.299.$$

The result is shown in Figure 8.36d.

Finally, we impedance and frequency scale the circuit, which simply involves multiplying the normalized characteristic impedances by 50 Ω and choosing the line and stub lengths to be $\lambda/8$ at 4 GHz. The final circuit is shown in Figure 8.36c, with a microstrip layout in Figure 8.36f.

The calculated amplitude response of this design is plotted in Figure 8.37, along with the response of the lumpcd-element version. Note that the passband characteristics are very similar up to 4 GHz, but the distributed-element filter has a sharper cutoff. Also notice that the distributed-element filter has a response which repeats every 16 GHz, as a result of the periodic nature of Richard's transformation.

Impedance and Admittance Inverters

As we have seen, it is often desirable to use only series, or only shunt, elements when implementing a filter with a particular type of transmission line. The Kuroda identities can be used for conversions of this form, but another possibility is to use impedance (K)or admittance (J) inverters [1], [4], [7]. Such inverters are especially useful for bandpass or bandstop filters with narrow (< 10%) bandwidths.



FIGURE 8.37 Amplitude responses of lumped-element and distributed-element low-pass filter of Example 8.6.

The conceptual operation of impedance and admittance inverters is illustrated in Figure 8.38; since these inverters essentially form the inverse of the load impedance or admittance, they can be used to transform series-connected elements to shunt-connected elements, or vice versa. This procedure will be illustrated in later sections for bandpass and bandstop filters.

In its simplest form, a J or K inverter can be constructed using a quarter-wave transformer of the appropriate characteristic impedance, as shown in Figure 8.38b. This implementation also allows the *ABCD* matrix of the inverter to be easily found from the *ABCD* parameters for a length of transmission line given in Table 4.1. Many other types of circuits can also be used as J or K inverters, with one such alternative being shown in Figure 8.38c. Inverters of this form turn out to be useful for modeling the coupled resonator filters of Section 8.8. The lengths, $\theta/2$, of the transmission line sections are generally required to be negative for this type of inverter, but this poses no problem if these lines can be absorbed into connecting transmission lines on either side.



FIGURE 8.38

Impedance and admittance inverters. (a) Operation of impedance and admittance inverters. (b) Implementation as quarter-wave transformers. (c) An alternative implementation.

470

8.6____ STEPPED-IMPEDANCE LOW-PASS FILTERS

A relatively easy way to implement low-pass filters in microstrip or stripline is to use alternating sections of very high and very low characteristic impedance lines. Such filters are usually referred to as *stepped-impedance*, or hi-Z, low-Z filters, and are popular because they are easier to design and take up less space than a similar low-pass filter using stubs. Because of the approximations involved, however, their electrical performance is not as good, so the use of such filters is usually limited to applications where a sharp cutoff is not required (for instance, in rejecting out-of-band mixer products).

Approximate Equivalent Circuits for Short Transmission Line Sections

We begin by finding the approximate equivalent circuits for a short length of transmission line having either a very large or very small characteristic impedance. The ABCD parameters of a length, ℓ , of line having characteristic impedance Z_0 are given in Table 4.1; the conversion in Table 4.2 can then be used to find the Z-parameters as

$$Z_{11} = Z_{22} = \frac{A}{C} = -jZ_0 \cos\beta\ell,$$
 8.81a

$$Z_{12} = Z_{21} = rac{1}{C} = -jZ_0 \csc eta \ell.$$
 8.81b

The series elements of the T-equivalent circuit are

$$Z_{11} - Z_{12} = -jZ_0 \left[\frac{\cos \beta \ell - 1}{\sin \beta \ell} \right] = jZ_0 \tan \left(\frac{\beta \ell}{2} \right), \qquad 8.82$$

while the shunt element of the T-equivalent is Z_{12} . So if $\beta \ell < \pi/2$, the series elements have a positive reactance (inductors), while the shunt element has a negative reactance (capacitor). We thus have the equivalent circuit shown in Figure 8.39a, where

$$\frac{X}{2} = Z_0 \tan\left(\frac{\beta\ell}{2}\right), \qquad 8.83a$$

$$B = \frac{1}{Z_0} \sin \beta \ell. \qquad 8.83b$$

Now assume a short length of line (say $\beta \ell < \pi/4$) and a large characteristic impedance. Then (8.83) approximately reduces to

$$X\simeq Z_0eta\ell,$$
 8.84a

$$B\simeq 0,$$
 8.84b

which implies the equivalent circuit of Figure 8.39b (a series inductor). For a short length of line and a small characteristic impedance, (8.83) approximately reduces to

$$X\simeq 0,$$
 8.85a

$$B \simeq Y_0 \beta \ell$$
, $8.85b$



FIGURE 8.39 Approximate equivalent circuits for short sections of transmission lines. (a) Tequivalent circuit for a transmission line section having $\beta \ell << \pi/2$. (b) Equivalent circuit for small $\beta \ell$ and large Z_0 . (c) Equivalent circuit for small $\beta \ell$ and small Z_0 .

which implies the equivalent circuit of Figure 8.39c (a shunt capacitor). So the series inductors of a low-pass prototype can be replaced with high-impedance line sections $(Z_0 = Z_h)$, and the shunt capacitors can be replaced with low-impedance line sections $(Z_0 = Z_\ell)$. The ratio Z_h/Z_ℓ should be as high as possible, so the actual values of Z_h and Z_ℓ are usually set to the highest and lowest characteristic impedance that can be practically fabricated. The lengths of the lines can then be determined from (8.84) and (8.85); to get the best response near cutoff, these lengths should be evaluated at $\omega = \omega_c$. Combining the results of (8.84) and (8.85) with the scaling equations of (8.67) allows the electrical lengths of the inductor sections to be calculated as

$$\beta \ell = \frac{LR_0}{Z_h}$$
 (inductor), 8.86a

and the electrical length of the capacitor sections as

$$\beta \ell = \frac{CZ_{\ell}}{R_0} \qquad \text{(capacitor)}, \qquad 8.86b$$

where R_0 is the filter impedance and L and C are the normalized element values (the g_k s) of the low-pass prototype.



EXAMPLE 8.7 Stepped-Impedance Filter Design

Design a stepped-impedance low-pass filter having a maximally flat response and a cutoff frequency of 2.5 GHz. It is necessary to have more than 20 dB insertion loss at 4.0 GHz. The filter impedance is 50Ω ; the highest practical line impedance is 150Ω , and the lowest is 10Ω . Solution

To use Figure 8.26, we calculate

$$\frac{\omega}{\omega_c} - 1 = \frac{4.0}{2.5} - 1 = 0.6,$$

then the figure indicates N = 6 should give the necessary attenuation at 4.0 GHz. Table 8.3 gives the low-pass prototype values as

 $g_1 = 0.517 = C_1,$ $g_2 = 1.414 = L_2,$ $g_3 = 1.932 = C_3,$ $g_4 = 1.932 = L_4,$ $g_5 = 1.414 = C_5,$ $g_6 = 0.517 = L_6.$

The low-pass prototype circuit is shown in Figure 8.40a.

Next, we use (8.86) to find the electrical lengths of the hi-Z, low-Z transmission line sections to replace the series inductors and shunt capacitors:

$$\beta \ell_1 = g_1 \frac{Z_\ell}{R_0} = 5.9^\circ,$$



FIGURE 8.40 Filter design for Example 8.7. (a) Low-pass filter prototype circuit. (b) Steppedimpedance implementation. (c) Microstrip layout of final filter.

$$\beta \ell_2 = g_2 \frac{R_0}{Z_h} = 27.0^\circ,$$

$$\beta \ell_3 = g_3 \frac{Z_\ell}{R_0} = 22.1^\circ,$$

$$\beta \ell_4 = g_4 \frac{R_0}{Z_h} = 36.9^\circ,$$

$$\beta \ell_5 = g_5 \frac{Z_\ell}{R_0} = 16.2^\circ,$$

$$\beta \ell_6 = g_6 \frac{R_0}{Z_h} = 9.9^\circ.$$

The final filter circuit is shown in Figure 8.40b, where $Z_{\ell} = 10 \Omega$ and $Z_h = 150 \Omega$. Note that $\beta \ell < \pi/4$ in all cases. A layout in microstrip is shown in Figure 8.40c.

Figure 8.41 shows the calculated amplitude response, compared with the response of the corresponding lumped-element filter. The passband characteristics are very similar, but the lumped-element circuit gives more attenuation at higher frequencies. This is because the stepped-impedance filter elements depart significantly from the lumped-element values at the higher frequencies. The stepped-impedance filter may have other passbands at higher frequencies, but the response will not be perfectly periodic because the lines are not commensurate.



FIGURE 8.41 Amplitude response of the stepped-impedance low-pass filter of Example 8.7, compared with the corresponding lumped-element design.

8.7 COUPLED LINE FILTERS

The parallel coupled transmission lines discussed in Section 7.6 (for directional couplers) can also be used to construct many types of filters. Fabrication of multisection bandpass or bandstop coupled line filters is particularly easy in microstrip or stripline form, for bandwidths less than about 20%. Wider bandwidth filters generally require very tightly coupled lines, which are difficult to fabricate. We will first study the filter characteristics of a single quarter-wave coupled line section, and then show how these sections can be used to design a bandpass filter [7]. Other filter designs using coupled lines can be found in reference [1].

Filter Properties of a Coupled Line Section

A parallel coupled line section is shown in Figure 8.42a, with port voltage and current definitions. We will derive the open-circuit impedance matrix for this four-port network by considering the superposition of even- and odd-mode excitations [8], which are shown in Figure 8.42b. Thus, the current sources i_1 and i_3 drive the line in the even mode, while i_2 and i_4 drive the line in the odd mode. By superposition, we see that the total port currents, I_i , can be expressed in terms of the even- and odd-mode currents as

$$I_1 = i_1 + i_2, 8.87a$$

$$I_2 = i_1 - i_2$$
, 8.87b

$$I_3 = i_3 - i_4,$$
 8.87c

$$I_4 = i_3 + i_4.$$
 8.87d

First consider the line as being driven in the even mode by the i_1 current sources. If the other ports are open-circuited, the impedance seen at port 1 or 2 is

$$Z_{in}^e = -jZ_{0e} \cot \beta \ell. \qquad 8.88$$

The voltage on either conductor can be expressed as

$$v_{a}^{1}(z) = v_{b}^{1}(z) = V_{e}^{+} [e^{-j\beta(z-\ell)} + e^{j\beta(z-\ell)}]$$

= $2V_{e}^{+} \cos\beta(\ell-z),$ 8.89

so the voltage at port 1 or 2 is

$$v_a^1(0) = v_b^1(0) = 2V_e^+ \cos\beta\ell = i_1 Z_{in}^e$$

This result and (8.88) can be used to rewrite (8.89) in terms of i_i as

$$v_a^1(z) = v_b^1(z) = -jZ_{0e} \frac{\cos\beta(\ell-z)}{\sin\beta\ell} i_l.$$
8.90

Similarly, the voltages due to current sources i_3 driving the line in the even mode are

$$v_a^3(z) = v_b^3(z) = -jZ_{0e} \frac{\cos\beta z}{\sin\beta \ell} i_3.$$
 8.91



FIGURE 8.42 Definitions pertaining to a coupled line filter section. (a) A parallel coupled line section with port voltage and current definitions. (b) A parallel coupled line section with even- and odd-mode current sources. (c) A two-port coupled line section having a bandpass response.

Now consider the line as being driven in the odd mode by current i_2 . If the other ports are open-circuited, the impedance seen at port 1 or 2 is

$$Z_{\rm in}^o = -j Z_{\theta o} \cot \beta \ell.$$
 8.92

The voltage on either conductor can be expressed as

$$v_a^2(z) = -v_b^2(z) = V_0^+ [e^{-j\beta(z-\ell)} + e^{j\beta(z-\ell)}] = 2V_0^+ \cos\beta(\ell-z).$$
 8.93

Then the voltage at port 1 or port 2 is

$$v_a^2(0) = -v_b^2(0) = 2V_0^+ \cos\beta\ell = i_2 Z_{\rm in}^o$$

This result and (8.92) can be used to rewrite (8.93) in terms of i_2 as

$$\boldsymbol{v}_a^2(\boldsymbol{z}) = -\boldsymbol{v}_b^2(\boldsymbol{z}) \approx -jZ_{0\nu}\frac{\cos\beta(\ell-\boldsymbol{z})}{\sin\beta\ell}\boldsymbol{i}_2.$$
 8.94

Similarly, the voltages due to current i_4 driving the line in the odd mode are

$$v_a^4(z) = -v_b^4(z) = -jZ_{0o} \frac{\cos\beta z}{\sin\beta\ell} i_4.$$
 8.95

Now the total voltage at port 1 is

$$V_{1} = v_{a}^{1}(0) + v_{a}^{2}(0) + v_{a}^{3}(0) + v_{a}^{4}(0)$$

= $-j(Z_{0e}i_{1} + Z_{0o}i_{2})\cot\theta - j(Z_{0e}i_{3} + Z_{0o}i_{4})\csc\theta$, 8.96

where the results of (8.90), (8.91), (8.94), and (8.95) were used, and $\theta = \beta \ell$. Next, we solve (8.87) for the i_j in terms of the Is:

$$i_1 = \frac{1}{2}(I_1 + I_2),$$
 8.97 α

$$i_2 = \frac{1}{2}(I_1 - I_2),$$
 8.97b

$$i_3 = \frac{1}{2}(I_3 + I_4),$$
 8.97c

$$i_4 = \frac{1}{2}(I_4 - I_3),$$
 8.97d

and use these results in (8.96):

$$V_{1} = \frac{-j}{2} (Z_{0e}I_{1} + Z_{0e}I_{2} + Z_{0o}I_{1} - Z_{0o}I_{2}) \cot \theta$$

$$\frac{-j}{2} (Z_{0e}I_{3} + Z_{0e}I_{4} + Z_{0o}I_{4} - Z_{0o}I_{3}) \csc \theta.$$
 8.98

This result yields the top row of the open-circuit impedance matrix [Z] that describes the coupled line section. From symmetry, all other matrix elements can be found once the first row is known. The matrix elements are then

$$Z_{11} = Z_{22} = Z_{33} = Z_{44} = \frac{-j}{2} (Z_{0e} + Z_{0a}) \cot \theta, \qquad 8.99a$$

$$Z_{12} = Z_{21} = Z_{34} = Z_{43} = \frac{-j}{2} (Z_{0c} - Z_{0o}) \cot \theta, \qquad 8.99b$$

$$Z_{13} = Z_{31} = Z_{24} = Z_{42} = \frac{-j}{2}(Z_{0e} - Z_{0o})\csc\theta, \qquad 8.99c$$

$$Z_{14} = Z_{41} = Z_{23} = Z_{32} = \frac{-j}{2} (Z_{0e} + Z_{0o}) \csc \theta. \qquad 8.99d$$

A two-port network can be formed from the coupled line section by terminating two of the four ports in either open or short circuits; there are ten possible combinations. as illustrated in Table 8.8. As indicated in this table, the various circuits have different frequency responses, including low-pass, bandpass, all pass, and all stop. For bandpass filters, we are most interested in the case shown in Figure 8.42c, as open circuits are easier to fabricate than are short circuits. In this case, $I_2 = I_4 = 0$, so the four-port impedance matrix equations reduce to

$$V_1 = Z_{11}I_1 + Z_{13}I_3, \qquad 8.100a$$

$$V_3 = Z_{31}I_1 + Z_{33}I_3, \qquad 8.100b$$

where Z_{ij} is given in (8.99).

We can analyze the filter characteristics of this circuit by calculating the image impedance (which is the same at ports 1 and 3), and the propagation constant. From Table 8.1, the image impedance in terms of the Z-parameters is

$$Z_{i} = \sqrt{Z_{11}^{2} - \frac{Z_{11}Z_{13}^{2}}{Z_{33}}}$$
$$= \frac{1}{2}\sqrt{(Z_{0e} - Z_{0o})^{2} \csc^{2}\theta - (Z_{0e} + Z_{0o})^{2} \cot^{2}\theta}.$$
 8.101

When the coupled line section is $\lambda/4$ long ($\theta = \pi/2$), the image impedance reduces to

$$Z_i = \frac{1}{2}(Z_{0e} - Z_{0o}), \qquad 8.102$$

which is real and positive, since $Z_{0c} > Z_{0c}$. But when $\theta \to 0$ or $\pi, Z_i \to \pm j\infty$, indicating a stopband. The real part of the image impedance is sketched in Figure 8.43, where the cutoff frequencies can be found from (8.101) as

$$\cos\theta_1 = -\cos\theta_2 = \frac{Z_{0e} - Z_{0o}}{Z_{0e} + Z_{0o}}$$

The propagation constant can also be calculated from the results of Table 8.1 as

$$\cos\beta = \sqrt{\frac{Z_{11}Z_{33}}{Z_{13}^2}} = \frac{Z_{11}}{Z_{13}} = \frac{Z_{0e} + Z_{0e}}{Z_{0e} - Z_{0e}}\cos\theta, \qquad 8.103$$

which shows β is real for $\theta_1 < \theta < \theta_2 = \pi - \theta_1$, where $\cos \theta_1 = (Z_{0c} - Z_{0o})/(Z_{0e} + Z_{0o})$.

Design of Coupled Line Bandpass Filters

Narrowband bandpass filters can be made with cascaded coupled line sections of the form shown in Figure 8.42c. To derive the design equations for filters of this type, we first show that a single coupled line section can be approximately modeled by the equivalent circuit shown in Figure 8.44. We will do this by calculating the image impedance and propagation constant of the equivalent circuit and showing that they are approximately equal to those of the coupled line section for $\theta = \pi/2$, which will correspond to the center frequency of the bandpass response.

Image Impedance Circuit Response $Re(Z_{l1})$ $Z_{li} = \frac{2Z_{0e}Z_{0e}\cos\theta}{\sqrt{(Z_{0e} + Z_{0e})^2\cos^2\theta - (Z_{0e} - Z_{0e})^2}}$ $Z_{i2} = \frac{Z_{0e}Z_{0o}}{Z_{i1}}$ $\circ \frac{1}{Z_{i2}}$ $\overline{Z_{1}}$ $\frac{\pi}{2}$ 0 \overline{u} 3372 Low pass $Re(Z_{i1})$ $rac{1}{z_{i1}}$ $Z_{\rm cl} = \frac{2Z_{\rm Qe}Z_{\rm Qo}\sin\theta}{\sqrt{(Z_{\rm Qe} - Z_{\rm Qo})^2 - (Z_{\rm Qe} + Z_{\rm Qo})^2\cos^2\theta}}$ 0 $\frac{\pi}{2}$ $\overline{Z_{t1}}$ π 3π Bandpass 2 $Re(Z_{\rm H})$ $Z_{i1} = \frac{\sqrt{(Z_{0e} - Z_{0e})^2 - (Z_{0e} + Z_{0e})^2 \cos^2 \theta}}{2\sin \theta}$ $\frac{1}{Z_{11}}$ 0 $\frac{\pi}{2}$ π 3# $\overline{Z_{i1}}$ 2 Bandpass $Re(Z_{i1})$ $Z_{11} = \frac{\sqrt{Z_{0e} Z_{0o}} \sqrt{(Z_{0e} - Z_{0o})^2 - (Z_{0e} + Z_{0o})^2 \cos^2 \theta}}{(Z_{0e} + Z_{0o}) \sin \theta}$ Z_{Ω} $Z_{i2} = \frac{Z_{0e} Z_{0e}}{Z_{i1}}$ 34 0 $\frac{\pi}{2}$ π $\overline{Z_{,1}}$ Bandpass $Z_{\rm d} = \frac{Z_{\rm 0e} + Z_{\rm 0o}}{2}$ Ali pass $\overline{Z_{i1}}$ Z_{i1} $Z_{t\bar{t}} = \frac{2Z_{0c}Z_{0o}}{Z_{0c} + Z_{0o}}$ All pass Z.1 $Z_{i1} = \sqrt{Z_{0e} Z_{0e}}$ All pass $-0 = \frac{1}{Z_{i1}} = -j \frac{2Z_{0e}Z_{0e}}{Z_{0e} + Z_{0e}} \cot \theta$ All stop $Z_{i2} = \frac{Z_{0e}Z_{0e}}{Z_{i1}}$ $Z_{i1} = j \sqrt{Z_{0e} Z_{0e}} \tan \theta$ All stop $Z_{i1} = -j \sqrt{Z_{0}} Z_{0a} \cot \theta$ All stop $\overrightarrow{Z_{i1}}^{\circ}$

TABLE 8.8 Ten Canonical Coupled Line Circuits

.



FIGURE 8.43 The real part of the image impedance of the bandpass network of Figure 8.42c.

The ABCD parameters of the equivalent circuit can be computed using the ABCDmatrices for transmission lines from Table 4.1:

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix} = \begin{bmatrix} \cos\theta & jZ_0 \sin\theta \\ \frac{j\sin\theta}{Z_0} & \cos\theta \end{bmatrix} \begin{bmatrix} 0 & -j/J \\ -jJ & 0 \end{bmatrix} \begin{bmatrix} \cos\theta & jZ_0 \sin\theta \\ \frac{j\sin\theta}{Z_0} & \cos\theta \end{bmatrix}$$
$$= \begin{bmatrix} \left(JZ_0 + \frac{1}{JZ_0}\right) \sin\theta\cos\theta & j\left(JZ_0^2 \sin^2\theta - \frac{\cos^2\theta}{J}\right) \\ j\left(\frac{1}{JZ_0^2} \sin^2\theta - J\cos^2\theta\right) & \left(JZ_0 + \frac{1}{JZ_0}\right) \sin\theta\cos\theta \end{bmatrix}.$$
8.104

The ABCD parameters of the admittance inverter were obtained by considering it as a quarter-wave length of transmission of characteristic impedance, 1/J. From (8.27) the image impedance of the equivalent circuit is

$$Z_{i} = \sqrt{\frac{B}{C}} = \sqrt{\frac{JZ_{0}^{2}\sin^{2}\theta - (1/J)\cos^{2}\theta}{(1/JZ_{0}^{2})\sin^{2}\theta - J\cos^{2}\theta}},$$
8.105

which reduces to the following value at the center frequency, $\theta = \pi/2$:

$$Z_i = J Z_0^2$$
. 8.106



Equivalent circuit of the coupled line section of Figure 8.42c. **FIGURE 8,44**

From (8.31) the propagation constant is

$$\cos \beta = A = \left(JZ_0 + \frac{1}{JZ_0}\right) \sin \theta \cos \theta. \qquad 8.107$$

Equating the image impedances in (8.102) and (8.106), and the propagation constants of (8.103) and (8.107) yields the following equations:

$$\begin{aligned} \frac{1}{2}(Z_{0e} - Z_{0o}) &= JZ_0^2, \\ \frac{Z_{0e} + Z_{0o}}{Z_{0e} - Z_{0o}} &= JZ_0 + \frac{1}{JZ_0}, \end{aligned}$$

where we have assumed $\sin \theta \simeq 1$ for θ near $\pi/2$. These equations can be solved for the even- and odd-mode line impedances to give

$$Z_{0e} = Z_0 [1 + JZ_0 + (JZ_0)^2], \qquad 8.108a$$

$$Z_{0o} = Z_0 [1 - JZ_0 + (JZ_0)^2].$$
 8.1086

Now consider a bandpass filter composed of a cascade of N + 1 coupled line sections, as shown in Figure 8.45a. The sections are numbered from left to right, with the load on the right, but the filter can be reversed without affecting the response. Since each coupled line section has an equivalent circuit of the form shown in Figure 8.44, the equivalent circuit of the cascade is as shown in Figure 8.45b. Between any two consecutive inverters we have a transmission line section that is effectively 2θ in length. This line is approximately $\lambda/2$ long in the vicinity of the bandpass region of the filter, and has an approximate equivalent circuit that consists of a shunt parallel LC resonator, as in Figure 8.45c.

The first step in establishing this equivalence is to find the parameters for the T-equivalent and ideal transformer circuit of Figure 8.45c (an exact equivalent). The ABCD matrix for this circuit can be calculated using the results in Table 4.1 for a T-circuit and an ideal transformer:

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix} = \begin{bmatrix} \frac{Z_{11}}{Z_{12}} & \frac{Z_{12}^2 - Z_{12}^2}{Z_{12}} \\ \frac{1}{Z_{12}} & \frac{Z_{11}}{Z_{12}} \end{bmatrix} \begin{bmatrix} -1 & 0 \\ 0 & -1 \end{bmatrix} = \begin{bmatrix} \frac{-Z_{11}}{Z_{12}} & \frac{Z_{12}^2 - Z_{11}^2}{Z_{12}} \\ \frac{-1}{Z_{12}} & \frac{-Z_{11}}{Z_{12}} \end{bmatrix}.$$
 8.109

Equating this result to the ABCD parameters for a transmission line of length 2θ and characteristic impedance Z_0 gives the parameters of the equivalent circuit as

$$Z_{12} = \frac{-1}{C} = \frac{jZ_0}{\sin 2\theta},$$
 8.110a

$$Z_{11} = Z_{22} = -Z_{12}A = -jZ_0 \cot 2\theta. \qquad 8.110b$$

Then the series arm impedance is

$$Z_{11} - Z_{12} = -jZ_0 \frac{\cos 2\theta + 1}{\sin 2\theta} = -jZ_0 \cot \theta.$$
 8.111







(c)







FIGURE 8.45

Development of an equivalent circuit for derivation of design equations for a coupled line bandpass filter. (a) Layout of an N+1 section coupled line bandpass filter. (b) Using equivalent circuit of Figure 8.44 for each coupled line section. (c) Equivalent circuit for transmission lines of length 2θ . (d) Equivalent circuit of the admittance inverters. (e) Using results of (c) and (d) for the N = 2 case. (f) Lumped-element circuit for a bandpass filter for N = 2.

Chapter 8: Microwave Filters

The 1:-1 transformer provides a 180° phase shift, which cannot be obtained with the T-network alone; since this does not affect the amplitude response of the filter, it can be discarded. For $\theta \sim \pi/2$ the series arm impedances of (8.111) are near zero, and can also be ignored. The shunt impedance Z_{12} , however, looks like the impedance of a parallel resonant circuit for $\theta \sim \pi/2$. If we let $\omega = \omega_0 + \Delta \omega$, where $\theta = \pi/2$ at the center frequency ω_0 , then we have $2\theta = \beta \ell = \omega \ell/v_p = (\omega_0 + \Delta \omega)\pi/\omega_0 = \pi(1 + \Delta \omega/\omega_0)$, so (8.110a) can be written for small $\Delta \omega$ as

$$Z_{12} = \frac{jZ_0}{\sin \pi (1 + \Delta \omega / \omega_0)} \simeq \frac{-jZ_0 \omega_0}{\pi (\omega - \omega_0)}.$$
 8.112

From Section 6.1 the impedance near resonance of a parallel LC circuit is

$$Z = \frac{-jL\omega_0^2}{2(\omega - \omega_0)},$$
 8.113

with $\omega_0^2 = 1/LC$. Equating this to (8.112) gives the equivalent inductor and capacitor values as

$$L = \frac{2Z_0}{\pi\omega_0}, \qquad \qquad 8.114a$$

$$C = \frac{1}{\omega_0^2 L} = \frac{\pi}{2Z_0\omega_0}.$$
 8.114b

.

The end sections of the circuit of Figure 8.45b require a different treatment. The lines of length θ on either end of the filter are matched to Z_0 , and so can be ignored. The end inverters, J_1 and J_{N+1} , can each be represented as a transformer followed by a $\lambda/4$ section of line, as shown in Figure 8.45d. The ABCD matrix of a transformer with a turns ratio N in cascade with a quarter-wave line is

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix} = \begin{bmatrix} \frac{1}{N} & 0 \\ 0 & N \end{bmatrix} \begin{bmatrix} 0 & -jZ_0 \\ \frac{-j}{Z_0} & 0 \end{bmatrix} = \begin{bmatrix} 0 & \frac{-jZ_0}{N} \\ \frac{-jN}{Z_0} & 0 \end{bmatrix}.$$
 8.115

Comparing this to the ABCD matrix of an admittance inverter (part of (8.104)) shows that the necessary turns ratio is $N = JZ_0$. The $\lambda/4$ line merely produces a phase shift, and so can be ignored.

Using these results for the interior and end sections allows the circuit of Figure 8.45b to be transformed into the circuit of Figure 8.45e, which is specialized to the N = 2 case. We see that each pair of coupled line sections leads to an equivalent shunt LC resonator, and an admittance inverter occurs between each pair of LC resonators. Next, we show that the admittance inverters have the effect of transforming a shunt LC resonator into a series LC resonator, leading to the final equivalent circuit of Figure 8.45f (shown for N = 2). This will then allow the admittance inverter constants, J_n , to be determined from the element values of a low-pass prototype. We will demonstrate this for the N = 2 case.

With reference to Figure 8.45e, the admittance just to the right of the J_2 inverter is

$$j\omega C_2 + \frac{1}{j\omega L_2} + Z_0 J_3^2 = j\sqrt{\frac{C_2}{L_2}} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega}\right) + Z_0 J_3^2,$$

since the transformer scales the load admittance by the square of the turns ratio. Then the admittance seen at the input of the filter is

$$Y = \frac{1}{J_1^2 Z_0^2} \left\{ j \omega C_1 + \frac{1}{j \omega L_1} + \frac{J_2^2}{j \sqrt{C_2/L_2} \left[(\omega/\omega_0) - (\omega_0/\omega) \right] + Z_0 J_3^2} \right\}$$
$$= \frac{1}{J_1^2 Z_0^2} \left\{ j \sqrt{\frac{C_1}{L_1}} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega} \right) + \frac{J_2^2}{j \sqrt{C_2/L_2} \left[(\omega/\omega_0) - (\omega_0/\omega) \right] + Z_0 J_3^2} \right\}.$$
 8.116

These results also use the fact, from (8.114), that $L_n C_n = 1/\omega_0^2$ for all LC resonators. Now the admittance seen looking into the circuit of Figure 8.45f is

$$Y = j\omega C_1' + \frac{1}{j\omega L_1'} + \frac{1}{j\omega L_2' + (1/j\omega C_2') + Z_0}$$

= $j\sqrt{\frac{C_1'}{L_1'}} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega}\right) + \frac{1}{j\sqrt{L_2'/C_2'}\left[(\omega/\omega_0) - (\omega_0/\omega)\right] + Z_0'}$ 8.117

which is identical in form to (8.116). Thus, the two circuits will be equivalent if the following conditions are met:

$$\frac{1}{J_1^2 Z_0^2} \sqrt{\frac{C_1}{L_1}} = \sqrt{\frac{C_1'}{L_1'}},$$
 8.118*a*

$$\frac{J_1^2 Z_0^2}{J_2^2} \sqrt{\frac{C_2}{L_2}} = \sqrt{\frac{L_2'}{C_2'}},$$
 8.118b

$$\frac{J_1^2 Z_0^3 J_3^2}{J_2^2} = Z_0. 8.118c$$

We know L_n and C_n from (8.114); L'_n and C'_n are determined from the element values of a lumped-element low-pass prototype which has been impedance scaled and frequency transformed to a bandpass filter. Using the results in Table 8.6 and the impedance scaling formulas of (8.64) allows the L'_n and C'_n values to be written as

$$L_1' = \frac{\Delta Z_0}{\omega_0 g_1}, \qquad \qquad 8.119a$$

$$C_1'=rac{g_1}{\Delta\omega_0 Z_0},$$
 8.119b

$$L_2' = \frac{g_2 Z_0}{\Delta \omega_0},$$
 8.119c

$$C_2' = \frac{\Delta}{\omega_0 g_2 Z_0}, \qquad \qquad 8.119d$$

where $\Delta = (\omega_2 - \omega_1)/\omega_0$ is the fractional bandwidth of the filter. Then (8.118) can be solved for the inverter constants with the following results (for N = 2):

$$J_{1}Z_{0} = \left(\frac{C_{1}L_{1}'}{L_{1}C_{1}'}\right)^{1/4} = \sqrt{\frac{\pi\Delta}{2g_{1}}}, \qquad 8.120a$$

$$J_2 Z_0 = J_1 Z_0^2 \left(\frac{C_2 C_2'}{L_2 L_2'}\right)^{1/4} = \frac{\pi \Delta}{2\sqrt{g_1 g_2}}, \qquad 8.120b$$

$$J_3 Z_0 = \frac{J_2}{J_1} = \sqrt{\frac{\pi \overline{\Delta}}{2g_2}}.$$
 8.120c

After the J_n s are found, Z_{0e} and Z_{0o} for each coupled line section can be calculated from (8.108).

The above results were derived for the special case of N = 2 (three coupled line sections), but more general results can be derived for any number of sections, and for the case where $Z_L \neq Z_0$ (or $g_{N+1} \neq 1$, as in the case of an equal-ripple response with N even). Thus, the design equations for a bandpass filter with N + 1 coupled line sections are

$$Z_0 J_1 = \sqrt{\frac{\pi \Delta}{2g_1}}, \qquad \qquad 8.121a$$

$$Z_{\emptyset}J_n = \frac{\pi\Delta}{2\sqrt{g_{n-1}g_n}},$$
 for $n = 2, 3, ..., N,$ 8.121b

$$Z_0 J_{N+1} = \sqrt{\frac{\pi \Delta}{2g_N g_{N+1}}}.$$
 8.121c

The even and odd mode characteristic impedances for each section are then found from (8.108).

Ø

EXAMPLE 8.8 Coupled Line Bandpass Filter Design

Design a coupled line bandpass filter with N = 3 and a 0.5 dB equal-ripple response. The center frequency is 2.0 GHz, the bandwidth is 10%, and $Z_0 = 50 \Omega$. What is the attenuation at 1.8 GHz?

Solution

The fractional bandwidth is $\Delta = 0.1$. We can use Figure 8.27a to obtain the attenuation at 1.8 GHz, but first we must use (8.71) to convert this frequency to the normalized low-pass form ($\omega_c = 1$):

$$\omega \leftarrow \frac{1}{\Delta} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega} \right) = \frac{1}{0.1} \left(\frac{1.8}{2.0} - \frac{2.0}{1.8} \right) = -2.11.$$

Then the value on the horizontal scale of Figure 8.27a is

$$\left|\frac{\omega}{\omega_c}\right| - 1 = |-2.11| - 1 = 1.11,$$

which indicates an attenuation of about 20 dB for N = 3.

The low-pass prototype values, g_n , are given in Table 8.4; then (8.121) can be used to calculate the admittance inverter constants, J_n . Finally, the even- and odd-mode characteristic impedances can be found from (8.108). These results are summarized in the following table:

n	g_n	$Z_0 J_n$	$Z_{0e}(\Omega)$	$Z_{0a}(\Omega)$
I	1.5963	0.3137	70.61	39.24
2	1.0967	0.1187	56.64	44.77
3	1.5963	0.1187	56.64	44.77
4	1.0000	0.3137	70.61	39.24

Note that the filter sections are symmetric about the midpoint. The calculated response of this filter is shown in Figure 8.46; passbands also occur at 6 GHz, 10 GHz, etc.

Many other types of filters can be constructed using coupled line sections; most of these are of the bandpass or bandstop variety. One particularly compact design is the interdigitated filter, which can be obtained from a coupled line filter by folding the lines at their midpoints; see [1] and [3] for details.



FIGURE 8.46 Amplitude response of the coupled line bandpass filter of Example 8.8.

8.8 ____ FILTERS USING COUPLED RESONATORS

We have seen that bandpass and bandstop filters require elements that behave as series or parallel resonant circuits; the coupled line bandpass filters of the previous section were of this type. Here we will consider several other types of microwave filters that use transmission line or cavity resonators.

Bandstop and Bandpass Filters Using Quarter-Wave Resonators

From Chapter 6 we know that quarter-wave open-circuited or short-circuited transmission line stubs look like series or parallel resonant circuits, respectively. Thus we can use such stubs in shunt along a transmission line to implement bandpass or bandstop filters, as shown in Figure 8.47. Quarter-wavelength sections of line between the stubs act as admittance inverters to effectively convert alternate shunt resonators to series resonators. The stubs and the transmission line sections are $\lambda/4$ long at the center frequency, ω_0 .

For narrow bandwidths the response of such a filter using N stubs is essentially the same as that of a coupled line filter using N + 1 sections. The internal impedance of the stub filter is Z_0 , while in the case of the coupled line filter end sections are required to transform the impedance level. This makes the stub filter more compact and easier to design. A disadvantage, however, is that a filter using stub resonators often requires characteristic impedances that are difficult to realize in practice,



FIGURE 8.47

Bandstop and bandpass filters using shunt transmission line resonators ($\theta = \pi/2$ at the center frequency). (a) Bandstop filter. (b) Bandpass filter.

We first consider a bandstop filter using N open-circuited stubs, as shown in Figure 8.47a. The design equations for the required stub characteristic impedances, Z_{0n} , will be derived in terms of the element values of a low-pass prototype through the use of an equivalent circuit. The analysis of the bandpass version, using short-circuited stubs, follows the same procedure so the design equations for this case are presented without detailed derivation.

As indicated in Figure 8.48a, an open-circuited stub can be approximated as a series LC resonator when its length is near 90°. The input impedance of an open-circuited transmission line of characteristic impedance Z_{0n} is

 $Z = -jZ_{0n}\cot\theta,$



 C_{n}



(a)







Equivalent circuit for the bandstop filter of Figure 8.47a. (a) Equivalent circuit of open-circuited stub for θ near $\pi/2$. (b) Equivalent filter circuit using resonators and admittance inverters. (c) Equivalent lumped-element bandstop filter.

where $\theta = \pi/2$ for $\omega = \omega_0$. If we let $\omega = \omega_0 + \Delta \omega$, where $\Delta \omega \ll \omega_0$, then $\theta = \pi/2(1 + \Delta \omega/\omega_0)$, and this impedance can be approximated as

$$Z = j Z_{0n} \tan \frac{\pi \Delta \omega}{2\omega_0} \simeq \frac{j Z_{0n} \pi (\omega - \omega_0)}{2\omega_0}, \qquad 8.122$$

for frequencies in the vicinity of the center frequency, ω_0 . The impedance of a series LC circuit is

$$Z = j\omega L_n + \frac{1}{j\omega C_n} = j\sqrt{\frac{L_n}{C_n}} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega}\right) \simeq 2j\sqrt{\frac{L_n}{C_n}} \frac{\omega - \omega_0}{\omega_0} \simeq 2jL_n(\omega - \omega_0),$$
8.123

where $L_n C_n = 1/\omega_0^2$. Equating (8.122) and (8.123) gives the characteristic impedance of the stub in terms of the resonator parameters:

$$Z_{0n} = rac{4\omega_0 L_n}{\pi}.$$
 8.124

Then, if we consider the quarter-wave sections of line between the stubs as ideal admittance inverters, the bandstop filter of Figure 8.47a can be represented by the equivalent circuit of Figure 8.48b. Next, the circuit elements of this equivalent circuit can be related to those of the lumped-element bandstop filter prototype of Figure 8.48c.

With reference to Figure 8.48b, the admittance, Y, seen looking toward the L_2C_2 resonator is

$$Y = \frac{1}{j\omega L_2 + (1/j\omega C_2)} + \frac{1}{Z_0^2} \left[\frac{1}{j\omega L_1 + (1/j\omega C_1)} + \frac{1}{Z_0} \right]^{-1}$$

= $\frac{1}{j\sqrt{L_2/C_2} \left[(\omega/\omega_0) - (\omega_0/\omega) \right]}$
+ $\frac{1}{Z_0^2} \left\{ \frac{1}{j\sqrt{L_1/C_1} \left[(\omega/\omega_0) - (\omega_0/\omega) \right] + (1/Z_0)} \right\}^{-1}$. 8.125

The admittance at the corresponding point in the circuit of Figure 8.48c is

$$\dot{Y} = \frac{1}{j\omega L_{2}' + (1/j\omega C_{2}')} + \left[\frac{1}{j\omega C_{1}' + (1/j\omega L_{1}')} + Z_{0}\right]^{-1}$$
$$= \frac{1}{j\sqrt{L_{2}'/C_{2}'}\left[(\omega/\omega_{0}) - (\omega_{0}/\omega)\right]}$$
$$+ \left\{\frac{1}{j\sqrt{C_{1}'/L_{1}'}\left[(\omega/\omega_{0}) - (\omega_{0}/\omega)\right]} + Z_{0}\right\}^{-1}.$$
 8.126
These two results will be equivalent if the following conditions are satisfied:

$$\frac{1}{Z_0^2} \sqrt{\frac{L_1}{C_1}} = \sqrt{\frac{C_1'}{L_1'}},$$
8.127*a*

$$\sqrt{\frac{L_2}{C_2}} = \sqrt{\frac{L_2'}{C_2'}}.$$
 8.127b

Since $L_n C_n = L'_n C'_n = 1/\omega_0^2$, these results can be solved for L_n :

$$L_1 = \frac{Z_0^2}{\omega_0^2 L_1'},$$
 8.128*a*

$$L_2 = L_2'.$$
 8.128b

Then using (8.124) and the impedance-scaled bandstop filter elements from Table 8.6 gives the stub characteristic impedances as

$$Z_{01}=rac{4Z_0^1}{\pi\omega_0 L_1'}=rac{4Z_0}{\pi g_1\Delta}, \hspace{1.5cm} 8.129a$$

$$Z_{02} = rac{4\omega_0 L_2'}{\pi} = rac{4Z_0}{\pi g_2 \Delta}.$$
 8.129b

where $\Delta = (\omega_2 - \omega_1)/\omega_0$ is the fractional bandwidth of the filter. It is easy to show that the general result for the characteristic impedances of a bandstop filter is

$$Z_{0n} = \frac{4Z_0}{\pi g_n \Delta}.$$
8.130

For a bandpass filter using short-circuited stub resonators the corresponding result is,

$$Z_{0n} = \frac{\pi Z_0 \Delta}{4g_n}.$$
 8.131

These results only apply to filters having input and output impedances of Z_0 , and so cannot be used for equal-ripple designs with N even.

EXAMPLE 8.9 Bandstop Filter Design

Design a bandstop filter using three quarter-wave open-circuit stubs. The center frequency is 2.0 GHz, the bandwidth is 15%, and the impedance is 50 Ω . Use an equal-ripple response, with a 0.5 dB ripple level.

Solution

The fractional bandwidth is $\Delta = 0.15$. Table 8.4 gives the low-pass prototype values, g_n , for N = 3. Then the characteristic impedances of the stubs can be

490

found from (8.130). The results are listed in the following table:

n,	Ĝn	Z_{0n}
1	1.5963	265.9 Ω
2	J.0967	387.0Ω
3	1.5963	265.9 Ω

The filter circuit is shown in Figure 8.47a, with all stubs and transmission line sections $\lambda/4$ long at 2.0 GHz. The calculated attenuation for this filter is shown in Figure 8.49; the ripple in the passbands is somewhat greater than 0.5 dB, as a result of the approximations involved in the development of the design equations.

The performance of quarter-wave resonator filters can be improved by allowing the characteristic impedances of the interconnecting lines to be variable; then an exact correspondence with coupled line bandpass or bandstop filters can be demonstrated. Design details for this case can be found in reference [1].

Bandpass Filters Using Capacitively Coupled Resonators

Another type of bandpass filter that can be conveniently fabricated in microstrip or stripline form is the capacitive-gap coupled resonator filter shown in Figure 8.50. An Nth order filter of this form will use N resonant sections of transmission line with N + 1 capacitive gaps between them. These gaps can be approximated as series capacitors; design data relating the capacitance to the gap size and transmission line parameters



FIGURE 8.49 Amplitude response of the bandstop filter of Example 8.9.

is given in graphical form in reference [1]. The filter can then be modeled as shown in Figure 8.50b. The resonators are approximately $\lambda/2$ long at the center frequency, ω_0 .

Next, we redraw the equivalent circuit of Figure 8.50b with negative-length transmission line sections on either side of the series capacitors. The lines of length ϕ will be $\lambda/2$ long at ω_0 , so the electrical length, θ_i , of the *i*th section in Figures 8.50a,b is

$$\theta_i = \pi + \frac{1}{2}\phi_i + \frac{1}{2}\phi_{i+1}$$
, for $i = 1, 2, ..., N$, 8.132

with $\phi_i < 0$. The reason for doing this is that the combination of series capacitor and negative-length transmission lines forms the equivalent circuit of an admittance inverter, as seen from Figure 8.38c. In order for this equivalence to be valid, the following



FIGURE 8.50 Development of the equivalence of a capacitive-gap coupled resonator bandpass filter to the coupled line bandpass filter of Figure 8.45 (a) The capacitive-gap coupled resonator bandpass filter. (b) Transmission line model. (c) Transmission line model with negative-length sections forming admittance inverters ($\phi_i/2 <$ 0). (d) Equivalent circuit using inverters and $\lambda/2$ resonators ($\phi = \pi$ at ω_0). This circuit is now identical in form with the coupled line bandpass filter equivalent circuit in Figure 8.45b.

relationship must hold between the electrical length of the lines and the capacitive susceptance:

$$\phi_i = -\tan^{-1}(2Z_0B_i).$$
8.133

Then the resulting inverter constant can be related to the capacitive susceptance as

$$B_i = \frac{J_i}{1 - (Z_0 J_i)^2}.$$
8.134

(These results are given in Figure 8.38, and their derivation is requested in Problem 8.15.)

The capacitive-gap coupled filter can then be modeled as shown in Figure 8.50d. Now consider the equivalent circuit shown in Figure 8.45b for a coupled line bandpass filter. Since these two circuits are identical (as $\phi = 2\theta = \pi$ at the center frequency), we can use the results from the coupled line filter analysis to complete the present problem. Thus, we can use (8.121) to find the admittance inverter constants, J_i , from the low-pass prototype values (g_i) and the fractional bandwidth, Δ . As in the case of the coupled line filter, there will be N + 1 inverter constants for an Nth order filter. Then (8.134) can be used to find the susceptance, B_i , for the *i*th coupling gap. Finally, the electrical length of the resonator sections can be found from (8.132) and (8.133):

$$\theta_i = \pi - \frac{1}{2} [\tan^{-1}(2Z_0B_i) + \tan^{-1}(2Z_0B_{i+1})].$$
 8.135



EXAMPLE 8.10 Coupled Resonator Bandpass Filter Design

Design a bandpass filter using capacitive coupled resonators, with a 0.5 dB equal-ripple passband characteristic. The center frequency is 2.0 GHz, the bandwidth is 10%, and the impedance is 50 Ω . At least 20 dB of attenuation is required at 2.2 GHz.

Solution

We first determine the order of the filter to satisfy the attenuation specification at 2.2 GHz. Using (8.71) to convert to normalized frequency gives

$$\omega \leftarrow \frac{1}{\Delta} \left(\frac{\omega}{\omega_0} - \frac{\omega_0}{\omega} \right) = \frac{1}{0.1} \left(\frac{2.2}{2.0} - \frac{2.0}{2.2} \right) = 1.91.$$

Then,

$$\left|\frac{\omega}{\omega_c}\right| - 1 = 1.91 - 1.0 = 0.91.$$

From Figure 8.27a, we see that N = 3 should satisfy the attenuation specification at 2.2 GHz. The low-pass prototype values are given in Table 8.4, from which the inverter constants can be calculated using (8.121). Then the coupling susceptances can be found from (8.134), and the coupling capacitor values as

$$C_n = rac{B_n}{\omega_0}$$

.

n	g_n	$Z_0 J_n$	B_n	C.	<i>θ</i> "	
1	1,5963	0.3137	6.96×10^{-3}	0.554 pF	155.8°	
2	1.0967	0.1187	2.41×10^{-3}	0.192 pF	166.5°	
3	1.5963	0.1187	2.41×10^{-3}	0.192 pF	155.8°	
4	1.0000	0.3137	6.96×10^{-3}	0.554 pF		

Finally, the resonator lengths can be calculated from (8.135). The following table summarizes these results.

The calculated amplitude response is plotted in Figure 8.51. The specifications of this filter are the same as the coupled line bandpass filter of Example 8.8, and comparison of the results in Figures 8.51 and 8.46 shows that the responses are identical near the passband region.

Direct-Coupled Waveguide Cavity Filters

Another type of bandpass filter that is commonly made in waveguide form is the direct-coupled cavity filter shown in Figure 8.52a. Inductive irises spaced along the waveguide form resonators that are approximately $\lambda_g/2$ in length. The equivalent circuit is shown in Figure 8.52b. Observe that this circuit is the dual of the capacitive-gap coupled resonator filter circuit of Figure 8.50b; therefore, the design and operation of these two filters will be very similar. One difference is that the waveguide filter must be designed in terms of $(\beta/k_0)\omega = \beta c$, instead of ω , because the electrical lengths



FIGURE 8.51

Amplitude response for the capacitive-gap coupled resonator bandpass filter of Example 8.10.



FIGURE 8.52 (a) A direct-coupled waveguide cavity filter. (b) Its equivalent circuit.

of the transmission line sections are proportional to β , the propagation constant of the waveguide. In addition, the reactance of the inductive diaphragms is proportional to β , rather than ω . Design equations and other details for waveguide cavity filters can be found in the literature [1].

REFERENCES

- [1] G. L. Matthaei, L. Young, and E. M. T. Jones, *Microwave Filters, Impedance-Matching Networks, and Coupling Structures*, Artech House, Dedham, Mass., 1980.
- [2] R. E. Collin, Foundations for Microwave Engineering, Second Edition, McGraw-Hill, N. Y., 1992.
- [3] J. A. G. Malherbe, Microwave Transmission Line Filters, Artech House, Dedham. Mass., 1979.
- [4] W. A. Davis, Microwave Semiconductor Circuit Design, Van Nostrand Reinhold, N. Y., 1984.
- [5] R. F. Harrington, Time-Harmonic Electromagnetic Fields, McGraw-Hill, N. Y., 1961.
- [6] P. I. Richard, "Resistor-Transmission Line Circuits," Proc. of the IRE, vol. 36, pp. 217-220, February 1948.
- [7] S. B. Cohn, "Parallel-Coupled Transmission-Line-Resonator Filters," *IRE Trans. Microwave Theory* and Techniques, vol. MTT-6, pp. 223–231. April 1958.
- [8] E. M. T. Jones and J. T. Bolljahn, "Coupled-Strip-Transmission Line Filters and Directional Couplers," IRE Trans. Microwave Theory and Techniques, vol. MTT-4, pp. 78-81, April 1956.

PROBLEMS

8.1 Consider the finite periodic structure shown on the next page, consisting of eight 80 Ω resistors spaced at intervals of $\lambda/2$ along a transmission line with $Z_0 = 50 \Omega$. Find the voltage V(z) along the line, and plot |V(z)| versus z.



8.2 Sketch the $k-\beta$ diagram for the infinite periodic structure shown below. Assume $Z_0 = 100 \Omega$, d = 1.0 cm, $k = k_0$, and $L_0 = 3.0 \text{ nH}$.



8.3 Verify the expression for the image impedance of a π -network given in Table 8.1.

8.4 Compute the image impedances and propagation factor for the network shown below.



- **8.5** Design a composite low-pass filter by the image parameter method with the following specifications: $R_0 = 50 \Omega$, $f_c = 50$ MHz, and $f_{\infty} = 52$ MHz. Use CAD to plot the insertion loss versus frequency.
- **8.6** Design a composite high-pass filter by the image parameter method with the following specifications: $R_0 = 75 \Omega$, $f_c = 50$ MHz, and $f_{\infty} = 48$ MHz. Use CAD to plot the insertion loss versus frequency.
- 8.7 Solve the design equations in Section 8.3 for the elements of an N = 2 equal-ripple filter if the ripple specification is 1.0 dB.
- **8.8** Design a low-pass maximally flat filter having a passband of 0 to 3 GHz, and an attenuation of 20 dB at 5 GHz. The characteristic impedance is 75Ω . Use CAD to plot the insertion loss versus frequency.
- **8.9** Design a five-section high-pass filter with a 3 dB equal-ripple response, a cutoff frequency of 1 GHz, and an impedance of 50 Ω . What is the resulting attenuation at 0.6 GHz? Use CAD to plot the insertion loss versus frequency.
- **8.10** Design a four-section bandpass filter having a maximally flat group delay response. The bandwidth should be 5% with a center frequency of 2 GHz. The impedance is 50 Ω . Use CAD to plot the insertion loss versus frequency.
- **8.11** Design a three-section bandstop filter with a 0.5 dB equal-ripple response, a bandwidth of 10% centered at 3 GHz, and an impedance of 75 Ω . What is the resulting attenuation at 3.1 GHz? Use CAD to plot the insertion loss versus frequency.

- 8.12 Verify the second Kuroda identity in Table 8.7 by calculating the ABCD matrices for both circuits.
- **8.13** Design a low-pass third-order maximally flat filter using only series stubs. The cutoff frequency is 6 GHz and the impedance is 50Ω . Use CAD to plot the insertion loss versus frequency.
- 8.14 Design a low-pass fourth-order maximally flat filter using only shunt stubs. The cutoff frequency is 8 GHz and the impedance is 50Ω . Use CAD to plot the insertion loss versus frequency.
- **8.15** Verify the operation of the admittance inverter of Figure 8.38c by calculating its ABCD matrix and comparing it to the ABCD matrix of the admittance inverter made from a quarter-wave line.
- **8.16** Show that the π equivalent circuit for a short length of transmission line leads to equivalent circuits identical to those in Figure 8.39b and c. for large and small characteristic impedance, respectively.
- 8.17 Design a stepped-impedance low-pass filter having a cutoff frequency of 4.0 GHz and a fifth-order, 0.5 dB equal-ripple response. Assume $R_0 = 100 \Omega$, $Z_\ell = 15 \Omega$, and $Z_h = 200 \Omega$. Use CAD to plot the insertion loss versus frequency.
- 8.18 Design a stepped-impedance low-pass filter with $f_c = 2.0$ GHz and $R_0 = 50 \Omega$, using the exact transmission line equivalent circuit of Figure 8.39a. Assume a maximally flat N = 5 response, and solve for the necessary line lengths and impedances if $Z_{\ell} = 10 \Omega$ and $Z_h = 150 \Omega$. Use CAD to plot the insertion loss versus frequency.
- **8.19** Design a four-section coupled line bandpass filter with a maximally flat response. The passband is 3.00 to 3.50 GHz, and the impedance is 50Ω . What is the attenuation at 2.9 GHz? Use CAD to plot the insertion loss versus frequency.
- **8.20** Design a maximally flat bandstop filter using four open-circuited quarter-wave stub resonators. The center frequency is 3 GHz, the bandwidth is 15%, and the impedance is 40 Ω . Use CAD to plot the insertion loss versus frequency.
- **8.21** Design a bandpass filter using three quarter-wave short-circuited stub resonators. The filter should have an equal-ripple response with a 0.5 dB ripple. a passband from 3.00 to 3.50 GHz, and an impedance of 50Ω . Use CAD to plot the insertion loss versus frequency.
- 8.22 Derive the design equation of (8.131) for bandpass filters using quarter-wave shorted stub resonators.
- 8.23 Design a bandpass filter using capacitive-gap coupled resonators. The response should be maximally flat, with a center frequency of 4 GHz, a bandwidth of 12%, and at least 12 dB attenuation at 3.6 GHz. The characteristic impedance is 50 Ω . Find the electrical line lengths and the coupling capacitor values. Use CAD to plot the insertion loss versus frequency.
- **8.24** Derive the design equations for an *N*-section direct-coupled waveguide cavity filter. (*Hint*: Use the impedance inverter of Figure 8.38c to put the equivalent circuit in the form of the coupled line filter equivalent in Figure 8.45b.)

Theory and Design of Ferrimagnetic Components

The components and networks discussed up to this point have all been reciprocal. That is, the response between any two ports, i and j, of a component did not depend on the direction of signal flow (thus, $S_{ij} = S_{ji}$). This will always be the case when the component consists of passive and isotropic material, but if anisotropic (different properties in different directions) materials are used, nonreciprocal behavior can be obtained. This allows the implementation of a wide variety of devices having directional properties.

In Chapter 1 we discussed materials with electric anisotropy (tensor permittivity), and magnetic anisotropy (tensor permeability). The most practical anisotropic materials for microwave applications are ferromagnetic compounds such as YIG (yttrium iron garnet), and ferrites composed of iron oxides and various other elements such as aluminum, cobalt, manganese, and nickel. In contrast to ferromagnetic materials (e.g., iron, steel), ferrimagnetic compounds have high resistivity and a significant amount of anisotropy at microwave frequencies. As we will see, the magnetic anisotropy of a ferrimagnetic material is actually induced by applying a DC magnetic bias field. This field aligns the magnetic dipoles in the ferrite material to produce a net (nonzero) magnetic dipole moment, and causes the magnetic dipoles to precess at a frequency controlled by the strength of the bias field. A microwave signal circularly polarized in the same direction as this precession will interact strongly with the dipole moments, while an oppositely polarized field will interact less strongly. Since, for a given direction of rotation, the sense of polarization changes with the direction of propagation, a microwave signal will propagate through a ferrite differently in different directions. This effect can be utilized to fabricate directional devices such as isolators, circulators, and gyrators. Another useful characteristic of ferrimagnetic materials is that the interaction with an applied microwave signal can be controlled by adjusting the strength of the bias field. This effect leads to a variety of control devices such as phase shifters, switches, and tunable resonators and filters.

We will begin by considering the microscopic behavior of a ferrimagnetic material and its interaction with a microwave signal to derive the permeability tensor. This macroscopic description of the material can then be used with Maxwell's equations to analyze wave propagation in an infinite ferrite medium, and in a ferrite-loaded waveguide. These canonical problems will illustrate the nonreciprocal propagation properties of ferrimagnetic materials, including Faraday rotation and birefringence effects, and will be used in later sections when discussing the operation and design of waveguide phase shifters and isolators.

9.1 BASIC PROPERTIES OF FERRIMAGNETIC MATERIALS

In this section we will show how the permeability tensor for a ferrimagnetic material can be deduced from a relatively simple microscopic view of the atom. We will also discuss how loss affects the permeability tensor, and the demagnetization field inside a finite-sized piece of ferrite.

The Permeability Tensor

The magnetic properties of a material are due to the existence of magnetic dipole moments, which arise primarily from electron spin. From quantum mechanical considerations [1], the magnetic dipole moment of an electron due to its spin is given by

$$m = \frac{q\hbar}{2m_{\odot}} = 9.27 \times 10^{-24} \text{ A-m}^2,$$
 9.1

where \hbar is Planck's constant divided by 2π , q is the electron charge, and m_e is the mass of the electron. An electron in orbit around a nucleus gives rise to an effective current loop, and thus an additional magnetic moment, but this effect is generally insignificant compared to the magnetic moment due to spin. The Landé g factor is a measure of the relative contributions of the orbital moment and the spin moment to the total magnetic moment: g = 1 when the moment is due only to orbital motion, and g = 2 when the moment is due only to spin. For most microwave ferrite materials, g is in the range of 1.98 to 2.01, so g = 2 is a good approximation.

In most solids, electron spins occur in pairs with opposite signs so the overall magnetic moment is negligible. In a magnetic material, however, a large fraction of the electron spins are unpaired (more left-hand spins than right-hand spins, or vice versa), but are generally oriented in random directions so that the net magnetic moment is still small. An external magnetic field, however, can cause the dipole moments to align in the same direction to produce a large overall magnetic moment. The existence of exchange forces can keep adjacent electron spins aligned after the external field is removed; the material is then said to be permanently magnetized.

An electron has a spin angular momentum given in terms of Planck's constant as [1], [2]

$$s = \frac{\hbar}{2}.$$
 9.2

The vector direction of this momentum is opposite the direction of the spin magnetic dipole moment, as indicated in Figure 9.1. The ratio of the spin magnetic moment to the spin angular momentum is a constant called the gyromagnetic ratio:

$$\gamma = \frac{m}{s} = \frac{q}{m_e} = 1.759 \times 10^{11} \text{ C/Kg.}$$
 9.3

where (9.1) and (9.2) have been used. Then we can write the following vector relation between the magnetic moment and the angular momentum:

$$\bar{m} = -\gamma \bar{s},$$

n 4

where the negative sign is due to the fact that these vectors are oppositely directed.



FIGURE 9.1 Spin magnetic dipole moment and angular momentum vectors for a spinning electron.

When a magnetic bias field $\bar{H}_0 = \hat{z}H_0$ is present, a torque will be exerted on the magnetic dipole:

$$\bar{T} = \bar{m} \times \bar{B}_0 = \mu_0 \bar{m} \times \bar{H}_0 = -\mu_0 \gamma \bar{s} \times \bar{H}_0.$$
9.5

Since torque is equal to the time rate of change of angular momentum, we have

$$\frac{d\bar{s}}{dt} = \frac{-1}{\gamma} \frac{d\bar{m}}{dt} = \bar{T} = \mu_0 \bar{m} \times \bar{H}_0,$$
$$\frac{d\bar{m}}{dt} = -\mu_0 \gamma \bar{m} \times \bar{H}_0.$$
9.6

οг

This is the equation of motion for the magnetic dipole moment, \bar{m} . We will solve this equation to show that the magnetic dipole precesses around the H_0 -field vector, like a spinning top precesses around a vertical axis.

Writing (9.6) in terms of its three vector components gives

$$\frac{dm_x}{dt} = -\mu_0 \gamma \dot{m}_y H_0, \qquad 9.7a$$

$$\frac{dm_y}{dt} = \mu_0 \gamma m_x H_0, \qquad 9.7b$$

$$\frac{dm_z}{dt} = 0. \qquad \qquad \textbf{9.7c}$$

Now use (9.7a,b) to obtain two equations for m_x and m_y :

500

$$\frac{d^2 m_y}{dt^2} + \omega_0^2 m_y = 0, 9.8b$$

where

$$\omega_0 = \mu_0 \gamma H_0. \tag{9.9}$$

is called the Larmor, or precession, frequency. One solution to (9.8) that is compatible with (9.7a,b) is given by

$$m_{\star} = A \cos \omega_0 t, \qquad \qquad 9.10a$$

$$m_y = A \sin \omega_0 t. \qquad 9.10b$$

Equation (9.7c) shows that m_2 is a constant, and (9.1) shows that the magnitude of \tilde{m} is also a constant, so we have the relation that

$$|\tilde{m}|^2 = \left(\frac{q\hbar}{2m_e}\right)^2 = m_x^2 + m_y^2 + m_z^2 = A^2 + m_z^2. \qquad 9.11$$

Thus the precession angle, θ , between \overline{m} and \overline{H}_0 (the z-axis) is given by

$$\sin\theta = \frac{\sqrt{m_x^2 + m_y^2}}{|\bar{m}|} = \frac{A}{|\bar{m}|}.$$
 9.12

The projection of \bar{m} on the xy plane is given by (9.10), which shows that \bar{m} traces a circular path in this plane. The position of this projection at time t is given by $\phi = \omega_0 t$, so the angular rate of rotation is $d\phi/dt = \omega_0$, the precession frequency. In the absence of any damping forces, the actual precession angle will be determined by the initial position of the magnetic dipole, and the dipole will precess about \bar{H}_0 at this angle indefinitely (free precession). In reality, however, the existence of damping forces will cause the magnetic dipole moment to spiral in from its initial angle until \bar{m} is aligned with \bar{H}_0 ($\theta = 0$).

Now assume that there are N unbalanced electron spins (magnetic dipoles) per unit volume, so that the total magnetization is

$$ar{M}=Nar{m},$$
 9.13

and the equation of motion in (9.6) becomes,

$$\frac{d\bar{M}}{dt} = -\mu_0 \gamma \bar{M} \times \bar{H}, \qquad \qquad 9.14$$

where \tilde{H} is the internal applied field. (Note: In Chapter 1 we used \tilde{P}_m for magnetization and \tilde{M} for magnetic currents; here we use \tilde{M} for magnetization, as this is common practice in ferrimagnetics work. Since we will not be using magnetic currents in this chapter, there should be no confusion.) As the strength of the bias field H_0 is increased, more magnetic dipole moments will align with H_0 until all are aligned, and \tilde{M} reaches an upper limit. See Figure 9.2. The material is then said to be magnetically saturated, and M_s is denoted as the saturation magnetization. M_s is thus a physical property of the ferrite material, and typically ranges from $4\pi M_s = 300$ to 5000 gauss. (Appendix H lists the saturation magnetization, ferrite materials can be very lossy at microwave frequencies, and the RF interaction is reduced. Thus ferrites are usually operated in the saturated state, and this assumption is made for the remainder of this chapter.



FIGURE 9.2 Magnetic moment of a ferrimagnetic material versus bias field, H_0 .

The saturation magnetization of a material is a strong function of temperature, decreasing as temperature increases, as illustrated in Figure 9.3. This effect can be understood by noting that the vibrational energy of an atom increases with temperature, making it more difficult to align all the magnetic dipoles. At a high enough temperature the thermal energy is greater than the energy supplied by the internal magnetic field, and a zero net magnetization results. This temperature is called the Curie temperature, T_C .

We now consider the interaction of a small AC (microwave) magnetic field with a magnetically saturated ferrite material. Such a field will cause a forced precession of the dipole moments around the $\bar{H}_0(\bar{z})$ axis at the frequency of the applied AC field, much like the operation of an AC synchronous motor. The small-signal approximation will apply to all the ferrite components of interest to us, but there are applications where high-power signals can be used to obtain useful nonlinear effects.

If \overline{H} is the applied AC field, the total magnetic field is

$$\bar{H}_t = H_0 \hat{z} + \bar{H}, \qquad 9.15$$



FIGURE 9.3 Magnetic moment of a ferrimagnetic material versus temperature, T.

where we assume that $|\bar{H}| << H_0$. This field produces a total magnetization in the ferrite material given by

$$\bar{M}_t = M_s \hat{z} + \bar{M}, \qquad 9.16$$

where M_s is the (DC) saturation magnetization and \overline{M} is the additional (AC) magnetization (in the xy plane) caused by \overline{H} . Substituting (9.16) and (9.15) into (9.14) gives the following component equations of motion:

$$\frac{dM_x}{dt} = -\mu_0 \gamma M_y (H_0 + H_z) + \mu_0 \gamma (M_s + M_z) H_y, \qquad 9.17a$$

$$\frac{dM_y}{dt} = \mu_0 \gamma M_x (H_0 + H_z) - \mu_0 \gamma (M_s + M_z) H_x, \qquad 9.17b$$

$$\frac{dM_z}{dt} = -\mu_0 \gamma M_x H_y + \mu_0 \gamma M_y H_x. \qquad 9.17c$$

since $dM_s/dt = 0$. Since $|\bar{H}| \ll H_0$, we have $|\bar{M}||\bar{H}| \ll |\bar{M}|H_0$ and $|\bar{M}||\bar{H}| \ll M_s|\bar{H}|$, so we can ignore MH products. Then (9.17) reduces to

$$\frac{dM_x}{dt} = -\omega_0 M_y + \omega_m H_y, \qquad 9.18a$$

$$\frac{dM_y}{dt} = \omega_0 M_x - \omega_m H_x, \qquad 9.18b$$

$$\frac{dM_z}{dt} = 0, \qquad 9.18c$$

where $\omega_0 = \mu_0 \gamma H_0$ and $\omega_m = \mu_0 \gamma M_s$. Solving (9.18a,b) for M_x and M_y gives the following equations:

$$\frac{d^2 M_x}{dt^2} + \omega_0^2 M_x = \omega_m \frac{dH_y}{dt} + \omega_0 \omega_m H_x, \qquad 9.19a$$

$$\frac{d^2 M_y}{dt^2} + \omega_0^2 M_y = -\omega_m \frac{dH_x}{dt} + \omega_0 \omega_m H_y. \qquad 9.19b$$

These are the equations of motion for the forced precession of the magnetic dipoles, assuming small-signal conditions. It is now an easy step to arrive at the permeability tensor for ferrites; after doing this, we will try to gain some physical insight into the magnetic interaction process by considering circularly polarized AC fields.

If the AC \hat{H} field has an $e^{j\omega t}$ time-harmonic dependence, the AC steady-state form of (9.19) reduces to the following phasor equations:

$$(\omega_0^2 - \omega^2)M_x = \omega_0 \omega_m H_x + j \omega \omega_m H_y, \qquad 9.206$$

$$(\omega_0^2 - \omega^2)M_y = -j\omega\omega_m H_x + \omega_0\omega_m H_y, \qquad 9.209$$

which shows the linear relationship between \bar{H} and \bar{M} . As in (1.24), (9.20) can be written with a tensor susceptibility, $[\chi]$, to relate \bar{H} and \bar{M} :

$$\bar{M} = [\chi]\bar{H} = \begin{bmatrix} \chi_{xx} & \chi_{xy} & 0 \\ \chi_{yx} & \chi_{yy} & 0 \\ 0 & 0 & 0 \end{bmatrix} \bar{H}, \qquad 9.21$$

where the elements of $[\chi]$ are given by

$$\chi_{xx} = \chi_{yy} = \frac{\omega_0 \omega_m}{\omega_0^2 - \omega^2}, \qquad \qquad 9.22a$$

$$\chi_{xy} = -\chi_{yx} = rac{j\omega\omega_m}{\omega_0^2 - \omega^2}.$$
 9.22b

The \hat{z} component of \hat{H} does not affect the magnetic moment of the material, under the above assumptions.

To relate \overline{B} and \overline{H} , we have from (1.23) that

$$\bar{B} = \mu_0(\bar{M} + \bar{H}) = [\mu]\bar{H},$$
 9.23

where the tensor permeability $[\mu]$ is given by

$$[\mu] = \mu_0([U] + [\chi]) = \begin{bmatrix} \mu & j\kappa & 0\\ -j\kappa & \mu & 0\\ 0 & 0 & \mu_0 \end{bmatrix} \quad (\hat{z} \text{ bias}). \qquad 9.24$$

The elements of the permeability tensor are then

$$\mu = \mu_0 (1 + \chi_{xx}) = \mu_0 (1 + \chi_{yy}) = \mu_0 \left(1 + \frac{\omega_0 \omega_m}{\omega_0^2 - \omega^2} \right), \qquad 9.25a$$

$$\kappa = -j\mu_0\chi_{xy} = j\mu_0\chi_{yx} = \mu_0\frac{\omega\omega_m}{\omega_0^2 - \omega^2}.$$
9.25b

A material having a permeability tensor of this form is called gyrotropic; note that an \hat{x} (or \hat{y}) component of \bar{H} gives rise to both \hat{x} and \hat{y} components of \bar{B} , with a 90^c phase shift between them.

If the direction of bias is reversed, both H_0 and M_s will change signs, so ω_0 and ω_m will change signs. Equation (9.25) then shows that μ will be unchanged, but κ will change sign. If the bias field is suddenly removed ($H_0 = 0$), the ferrite will generally remain magnetized ($0 < |M| < M_s$); only by demagnetizing the ferrite (with a decreasing AC bias field, for example) can M = 0 be obtained. Since the results of (9.22) and (9.25) assume a saturated ferrite sample, both M_s and H_0 should be set to zero for the unbiased, demagnetized case. Then $\omega_0 = \omega_m = 0$, and (9.25) show that $\mu = \mu_0$ and $\kappa = 0$, as expected for a nonmagnetic material.

The tensor results of (9.24) assume bias in the \hat{z} direction. If the ferrite is biased in a different direction the permeability tensor will be transformed according to the change in coordinates. Thus, if $\bar{H}_0 = \hat{x}H_0$, the permeability tensor will be

$$[\mu] = \begin{bmatrix} \mu_0 & 0 & 0 \\ 0 & \mu & j\kappa \\ 0 & -j\kappa & \mu \end{bmatrix} \quad (\hat{x} \text{ bias}), \qquad 9.26$$

while if $\ddot{H}_0 = \hat{y}H_0$ the permeability tensor will be

$$[\mu] = \begin{bmatrix} \mu & 0 & -j\kappa \\ 0 & \mu_0 & 0 \\ j\kappa & 0 & \mu \end{bmatrix} \quad (\hat{y} \text{ bias}). \qquad 9.27$$

A comment must be made about units. By tradition most practical work in magnetics is done with CGS units, with magnetization measured in gauss (1 gauss = 10^{-4} weber/m²), and field strength measured in oersteds ($4\pi \times 10^{-3}$ oersted = 1 A/m). Thus, $\mu_0 =$ 1 gauss/oersted in CGS units, implying that B and H have the same numerical values in a nonmagnetic material. Saturation magnetization is usually expressed as $4\pi M_s$ gauss: the corresponding MKS value is then $\mu_0 M_s$ weber/m² = 10^{-4} ($4\pi M_s$ gauss). In CGS units, the Larmor frequency can be expressed as $f_0 = \omega_0/2\pi = \mu_0\gamma H_0/2\pi = (2.8 \text{ MH/oersted})$ (H_0 oersted), and $f_m = \omega_m/2\pi = \mu_0\gamma M_s/2\pi = (2.8 \text{ MHz/oersted}) \cdot (4\pi M_s$ gauss). In practice, these units are convenient and easy to use.

Circularly Polarized Fields

To get a better physical understanding of the interaction of an AC signal with a saturated ferrimagnetic material we will consider circularly polarized fields. As discussed in Section 1.5, a right-hand circularly polarized field can be expressed in phasor form as

$$\bar{H}^+ = H^+(\hat{x} - j\hat{y}).$$
 9.28a

and in time-domain form as

$$\bar{\mathcal{H}}^+ = \operatorname{Re}\{\bar{H}^+ e^{j\omega t}\} = H^+(\hat{x}\cos\omega t + \hat{y}\sin\omega t), \qquad 9.233$$

where we have assumed the amplitude H^+ as real. This latter form shows that $\bar{\mathcal{H}}^+$ is a vector which rotates with time, such that at time t it is oriented at the angle ωt from the x-axis: thus its angular velocity is ω . (Also note that $|\tilde{\mathcal{H}}^+| = H^+ \neq |\tilde{H}^+|$.) Applying the RHCP field of (9.28a) to (9.20) gives the magnetization components as

$$egin{aligned} M_x^+ &= rac{\omega_m}{\omega_0 - \omega} H^+, \ M_y^+ &= rac{-j \omega_m}{\omega_0 - \omega} H^+, \end{aligned}$$

so the magnetization vector resulting from \tilde{H}^+ can be written as

$$\bar{M}^{+} = M_{x}^{+}\hat{x} + M_{y}^{+}\hat{y} = \frac{\omega_{m}}{\omega_{0} - \omega}H^{+}(\hat{x} - j\hat{y}), \qquad 9.29$$

which shows that the magnetization is also RHCP, and so rotates with angular velocity ω in synchronism with the driving field, \bar{H}^+ . Since \bar{M}^+ and \bar{H}^+ are vectors in the same direction, we can write $\bar{B}^+ = \mu_0(\bar{M}^+ + \bar{H}^+) = \mu^+\bar{H}^+$, where μ^+ is the effective permeability for an RHCP wave given by

$$\mu^{+} = \mu_0 \left(1 + \frac{\omega_m}{\omega_0 - \omega} \right). \tag{9.30}$$

The angle, θ_M , between $\tilde{\mathcal{M}}^+$ and the z-axis is given by

$$\tan \theta_M = \frac{|\bar{\mathcal{M}}^+|}{M_s} = \frac{\omega_m H^+}{(\omega_0 - \omega)M_s} = \frac{\omega_0 H^+}{(\omega_0 - \omega)H_0}, \qquad 9.31$$

while the angle, θ_H , between \hat{H}^+ and the z-axis is given by

$$\tan \theta_H \approx \frac{|\bar{\mathcal{H}}^+|}{H_0} = \frac{H^+}{H_0}.$$
 9.32

For frequencies such that $\omega < 2\omega_0$, (9.31) and (9.32) show that $\theta_M > \theta_H$, as illustrated in Figure 9.4a. In this case the magnetic dipole is precessing in the same direction as it would freely precess in the absence of \bar{H}^+ .

Now consider a left-hand circularly polarized field, expressed in phasor form as

$$\bar{H}^- = H^-(\hat{x} + j\hat{y}),$$
 9.33a

and in time-domain form as

$$\hat{\mathcal{H}}^{-} = \operatorname{Re}\{\bar{H}^{-}e^{j\omega t}\} = H^{-}(\hat{x}\cos\omega t - \hat{y}\sin\omega t). \qquad 9.33b$$

Equation (9.33b) shows that $\tilde{\mathcal{H}}^-$ is a vector rotating in the $-\omega$ (left-hand) direction. Applying the LHCP field of (9.33a) to (9.20) gives the magnetization components as

$$M_{x}^{-} = rac{\omega_{m}}{\omega_{0} + \omega} H^{-},$$

 $M_{y}^{-} = rac{j\omega_{m}}{\omega_{0} + \omega} H^{-},$

so the vector magnetization can be written as

$$\bar{M}^{-} = M_{x}^{-}\hat{x} + M_{y}^{-}\hat{y} = \frac{\omega_{m}}{\omega_{0} + \omega}H^{-}(\hat{x} + j\hat{y}), \qquad 9.34$$

which shows that the magnetization is LHCP, rotating in synchronism with \hat{H}^- . Writing $\bar{B}^- = \mu_0(\bar{M}^- + \bar{H}^-) = \mu^- \bar{H}^-$, gives the effective permeability for an LHCP wave as

$$\mu^{-} = \mu_0 \left(1 + \frac{\omega_m}{\omega_0 + \omega} \right). \tag{9.35}$$



FIGURE 9.4 Forced precession of a magnetic dipole with circularly polarized fields. (a) RHCP, $\theta_M > \theta_H$. (b) LHCP, $\theta_M < \theta_H$.

The angle, θ_M , between \overline{M}^- and the z-axis is given by

$$\tan \theta_M = \frac{|\overline{M}^-|}{M_s} = \frac{\omega_m H^-}{(\omega_0 + \omega)M_s} = \frac{\omega_0 H^-}{(\omega_0 + \omega)H_0},$$
9.36

which is seen to be less than θ_H of (9.32), as shown in Figure 9.4b. In this case the magnetic dipole is precessing in the opposite direction to its free precession.

Thus we see that the interaction of a circularly polarized wave with a biased ferrite depends on the sense of the polarization (RHCP or LHCP). This is because the bias field sets up a preferential precession direction coinciding with the direction of forced precession for an RHCP wave but opposite to that of an LHCP wave. As we will see in Section 9.2, this effect leads to nonreciprocal propagation characteristics.

Effect of Loss

Equations (9.22) and (9.25) show that the elements of the susceptibility or permeability tensors become infinite when the frequency, ω , equals the Larmor frequency, ω_0 . This effect is known as gyromagnetic resonance, and occurs when the forced precession frequency is equal to the free precession frequency. In the absence of loss the response may be unbounded, in the same way that the response of an *LC* resonant circuit will be unbounded when driven with an AC signal having a frequency equal to the resonant frequency of the *LC* circuit. All real ferrite materials, however, have various magnetic loss mechanisms that damp out such singularities.

As with other resonant systems, loss can be accounted for by making the resonant frequency complex:

$$\omega_0 \leftarrow \omega_0 + j\alpha\omega,$$
 9.37

where α is a damping factor. Substituting (9.37) into (9.22) makes the susceptibilities complex:

$$\chi_{xx} = \chi'_{xx} - j\chi''_{xx} \qquad 9.386$$

$$\chi_{inv} = \chi_{inv}'' + j\chi_{inv}' \qquad 9.38b$$

where the real and imaginary parts are given by

$$\chi'_{xx} = \frac{\omega_0 \omega_m (\omega_0^2 - \omega^2) + \omega_0 \omega_m \omega^2 \alpha^2}{[\omega_0^2 - \omega^2 (1 + \alpha^2)]^2 + 4\omega_0^2 \omega^2 \alpha^2},$$
9.39a

$$\chi_{xx}'' = \frac{\alpha \omega \omega_m [\omega_0^2 + \omega^2 (1 + \alpha^2)]}{[\omega_0^2 - \omega^2 (1 + \alpha^2)]^2 + 4\omega_0^2 \omega^2 \alpha^2},$$
9.39b

$$\chi'_{xy} = \frac{\omega \omega_m [\omega_0^2 - \omega^2 (1 + \alpha^2)]}{[\omega_0^2 - \omega^2 (1 + \alpha^2)]^2 + 4\omega_0^2 \omega^2 \alpha^2},$$
9.396

$$\chi_{xy}'' = \frac{2\omega_0 \omega_m \omega^2 \alpha}{[\omega_0^2 - \omega^2 (1 + \alpha^2)]^2 + 4\omega_0^2 \omega^2 \alpha^2}.$$
 9.39d



FIGURE 9.5 Complex susceptibilities for a typical ferrite. (a) Real and imaginary parts of χ_{xx} . (b) Real and imaginary parts of χ_{xy} .

Equation (9.37) can also be applied to (9.25) to give a complex $\mu = \mu' - j\mu''$, and $\kappa = \kappa' - j\kappa''$; this is why (9.38b) appears to define χ'_{xy} and χ''_{xy} backward, as $\chi_{xy} = j\kappa/\mu_0$. For most ferrite materials the loss is small, so $\alpha << 1$, and the $(1 + \alpha^2)$ terms in (9.39) can be approximated as unity. The real and imaginary parts of the susceptibilities of (9.39) are sketched in Figure 9.5 for a typical ferrite.

The damping factor, α , is related to the *linewidth*, ΔH , of the susceptibility curve near resonance. Consider the plot of $\chi_{se}^{\prime\prime}$ versus bias field, H_0 , shown in Figure 9.6.





For a fixed frequency, ω , resonance occurs when $H_0 = H_r$, such that $\omega_0 = \mu_0 \gamma H_r$. The linewidth, ΔH , is defined as the width of the curve of χ''_{xx} versus H_0 where χ''_{xx} has decreased to half its peak value. If we assume $(1 + \alpha^2) \simeq 1$, (9.39b) shows that the maximum value of χ''_{xx} is $\omega_m/2\alpha\omega$, and occurs when $\omega = \omega_0$. Now let ω_{02} be the Larmor frequency for which $H_0 = H_2$, where χ''_{xx} has decreased to half its maximum value. Then we can solve (9.39b) for α in terms of ω_{02} :

$$\frac{\alpha\omega\omega_m(\omega_{02}^2+\omega^2)}{(\omega_{02}^2-\omega^2)^2+4\omega_{02}^2\omega^2\alpha^2}=\frac{\omega_m}{4\alpha\omega},$$
$$4\alpha^2\omega^4=(\omega_{02}^2-\omega^2)^2.$$
$$\omega_{02}=\omega\sqrt{1+2\alpha}\simeq\omega(1+\alpha).$$

Then $\Delta\omega_0 = 2(\omega_{02} - \omega_0) \simeq 2[\omega(1 + \alpha) - \omega] = 2\alpha\omega$, and using (9.9) gives the linewidth as

$$\Delta H = \frac{\Delta \omega_0}{\mu_0 \gamma} = \frac{2\alpha \omega}{\mu_0 \gamma}.$$
 9.40

Typical linewidths range from less than 100 Oe (for yttrium iron garnet) to 100-500 Oe (for ferrites): single-crystal YIG can have a linewidth as low as 0.3 Oe. Also note that this loss is separate from the dielectric loss that a ferrimagnetic material may have.

Demagnetization Factors

The DC bias field, H_0 , internal to a ferrite sample is generally different from the externally applied field, H_0 , because of the boundary conditions at the surface of the ferrite. To illustrate this effect, consider a thin ferrite plate, as shown in Figure 9.7. When the applied field is normal to the plate, continuity of B_n at the surface of the plate gives

$$B_n = \mu_0 H_a = \mu_0 (M_s + H_0),$$

so the internal magnetic bias field is

$$H_0 = H_a - M_s.$$

This shows that the internal field is less than the applied field by an amount equal to the saturation magnetization. When the applied field is parallel to the ferrite plate, continuity





of H_t at the surfaces of the plate gives

$$H_t = H_a = H_0.$$

In this case the internal field is not reduced. In general, the internal field (AC or DC), \bar{H} , is affected by the shape of the ferrite sample and its orientation with respect to the external field, \bar{H}_e , and can be expressed as

$$\bar{H} = \bar{H}_r - N\bar{M}, \qquad 9.41$$

where $N = N_x$, N_y , or N_z is called the *demagnetization factor* for that direction of the external field. Different shapes have different demagnetization factors, which depend on the direction of the applied field. Table 9.1 lists the demagnetization factors for a few simple shapes. The demagnetization factors are defined such that $N_x + N_y + N_z = 1$.

The demagnetization factors can also be used to relate the internal and external RF fields near the boundary of a ferrite sample. For a z-biased ferrite with transverse RF fields, (9.41) reduces to

$$H_x = H_{xe} - N_x M_x, \qquad 9.42a$$

$$H_y = H_{ye} - N_y M_y, \qquad 9.42b$$

$$H_z = H_n - N_z M_s, \qquad \qquad 9.42c$$

where H_{xe} . H_{ye} are the RF fields external to the ferrite, and H_a is the externally applied

	Shape	N _x	Ny	Nz
Thin disk or plate		0	0	l I
Thin rod		12	1	0
Sphore		<u>1</u> <u>3</u>		<u>1</u> 3

TABLE 9.1 Demagnetization Factors for Some Simple Shapes

bias field. Equation (9.21) relates the internal transverse RF fields and magnetization as

$$M_x = \chi_{xx}H_x + \chi_{xy}H_y,$$
$$M_y = \chi_{yx}H_x + \chi_{yy}H_y.$$

Using (9.42a,b) to eliminate H_x and H_y gives

$$M_x = \chi_{xx}H_{xe} + \chi_{xy}H_{ye} - \chi_{xx}N_xM_x - \chi_{xy}N_yM_y,$$

$$M_y = \chi_{yx}H_{xe} + \chi_{yy}H_{ye} - \chi_{yx}N_xM_x - \chi_{yy}N_yM_y.$$

These equations can be solved for M_x , M_y to give

$$M_x = \frac{\chi_{xx}(1 + \chi_{yy}N_y) - \chi_{xy}\chi_{yx}N_y}{D}H_{xe} + \frac{\chi_{xy}}{D}H_{ye}, \qquad 9.43a$$

$$M_{y} = \frac{\chi_{yx}}{D} H_{xe} + \frac{\chi_{yy}(1 + \chi_{xx}N_{x}) - \chi_{yx}\chi_{xy}N_{x}}{D} H_{ye}, \qquad 9.43b.$$

where

$$D = (1 + \chi_{xx} N_x)(1 + \chi_{yy} N_y) - \chi_{yx} \chi_{xy} N_x N_y.$$
 9.44

This result is of the form $\tilde{M} = [\chi_{\varepsilon}]\hat{H}$, where the coefficients of $H_{x\varepsilon}$ and $H_{y\varepsilon}$ in (9.43) can be defined as "external" susceptibilities since they relate magnetization to the external RF fields.

For an infinite ferrite medium gyromagnetic resonance occurs when the denominator of the susceptibilities of (9.22) vanishes, at the frequency $\omega_r = \omega = \omega_0$. But for a finite-sized ferrite sample the gyromagnetic resonance frequency is altered by the demagnetization factors, and given by the condition that D = 0 in (9.43). Using the expressions in (9.22) for the susceptibilities in (9.44), and setting the result equal to zero gives

$$\left(1+\frac{\omega_0\omega_m N_x}{\omega_0^2-\omega^2}\right)\left(1+\frac{\omega_0\omega_m N_y}{\omega_0^2-\omega^2}\right)-\frac{\omega^2\omega_m^2}{(\omega_0^2-\omega^2)^2}N_xN_y=0.$$

After some algebraic manipulations this result can be reduced to give the resonance frequency, ω_r , as

$$\omega_r = \omega = \sqrt{(\omega_0 + \omega_m N_x)(\omega_0 + \omega_m N_y)}$$
9.45

Since $\omega_0 = \mu_0 \gamma H_0 = \mu_0 \gamma (H_a - N_z M_s)$, and $\omega_m = \mu_0 \gamma M_s$, (9.45) can be rewritten in terms of the applied bias field strength and saturation magnetization as

$$\omega_r = \mu_0 \gamma \sqrt{[H_a + (N_x - N_z)M_s][H_a + (N_y - N_z)M_s]}.$$
9.46

This result is known as Kittel's equation [4].



POINT OF INTEREST: Permanent Magnets

Since ferrite components such as isolators, gyrators, and circulators generally use permanent magnets to supply the required DC bias field, it may be useful to mention some of the important characteristics of permanent magnets.

A permanent magnet is made by placing the magnetic material in a strong magnetic field, and then removing the field, to leave the material magnetized in a remanent state. Unless the magnet shape forms a closed path (like a toroid), the demagnetization factors at the magnet ends will cause a slightly negative H field to be induced in the magnet. Thus the "operating point" of a permanent magnet will be in the second quadrant of the B-H hysteresis curve for the magnet material. This portion of the curve is called the demagnetization curve. A typical example is shown below.



The residual magnetization, for H = 0, is called the remanence, B_r , of the material. This quantity characterizes the strength of the magnet, so generally a magnet material is chosen to have a large remanence. Another important parameter is the coercivity, H_c , which is the value of the negative H field required to reduce the magnetization to zero. A good permanent magnet should have a high coercivity to reduce the effects of vibration, temperature changes, and external fields, which can lead to a loss of magnetization. An overall figure of merit for a permanent magnet is sometimes given as the maximum value of the BH product, $(BH)_{max}$, on the demagnetization curve. This quantity is essentially the maximum magnetic energy density that can be stored by the magnet, and can be useful in electromechanical applications. The following table lists the remanence, coercivity, and $(BH)_{max}$ for some of the most common permanent magnet materials.

Material	Composition	B, (Oe)	Н _с (G)	(<i>BH</i>) _{max} (G-Oe)×10 ⁶
ALNICO 5	Al, Ni, Co, Cu	12,000	720	5.0
ALNICO 8	Al, Ni, Co, Cu, T _i	7.100	2,000	5.5
ALNICO 9	Al, Ni, Co, Cu, T ₁	10,400	1,600	8.S
Remailoy	Mo, Co, Fe	10,500	250	1.1
Platinum Cobalt	Pt. Co	6,450	4,300	9.5
Ceramic	$BaO_6Fe_2O_3$	3,950	2,400	3 .5
Cobalt Samarium	Co. Sm	8,400	7,000	L6.0

<u>9.2</u>

PLANE WAVE PROPAGATION IN A FERRITE MEDIUM

The previous section gives an explanation of the microscopic phenomena that occur inside a biased ferrite material to produce a tensor permeability of the form given in (9.24) (or in (9.26) or (9.27), depending on the bias direction). Once we have this macroscopic description of the ferrite material, we can solve Maxwell's equations for wave propagation in various geometries involving ferrite materials. We begin with plane wave propagation in an infinite ferrite medium, for propagation either in the direction of bias, or propagation transverse to the bias field. These problems will illustrate the important effects of Faraday rotation and birefringence. 512

Propagation in Direction of Bias (Faraday Rotation)

Consider an infinite ferrite-filled region with a DC bias field given by $\bar{H}_0 = \hat{z}H_0$, and a tensor permittivity $[\mu]$ given by (9.24). Maxwell's equations can be written as

$$\nabla \times E \approx -j\omega[\mu]H, \qquad 9.47a$$

$$abla imes ar{H} = j\omega\epsilonar{E},$$
 9.47b

$$\nabla \cdot \bar{D} \approx 0, \qquad 9.47_{\rm C}$$

$$abla \cdot \vec{B} \approx 0.$$
 9.47 d

Now assume plane wave propagation in the z direction, with $\partial/\partial x = \partial/\partial y = 0$. Then the electric and magnetic fields will have the following form:

$$\bar{E} = \bar{E}_0 e^{-j\beta z}, \qquad \qquad 9.48a$$

$$H = \bar{H}_0 e^{-j\beta z}, \qquad \qquad 9.48h$$

The two curl equations of (9.47a,b) then reduce to the following, after using (9.24);

$$jeta E_y = -j\omega(\mu H_x + j\kappa H_y),$$
 9.49a

$$-j\beta E_x = -j\omega(-j\kappa H_x + \mu H_y), \qquad 9.49b$$

$$0 = -j\omega\mu_0 H_z, \qquad 9.49c$$

$$j\beta H_y = j\omega\epsilon E_x, \qquad 9.49d$$

$$-jeta H_x = j\omega\epsilon E_y,$$
 9.49e

$$0 = j\omega\epsilon E_z. \qquad 9.49f$$

Equations (9.49c) and (9.49f) show that $E_z = H_z = 0$, as expected for TEM planes. Then we also have $\nabla \cdot \overline{D} = \nabla \cdot \overline{B} = 0$, since $\partial/\partial x = \partial/\partial y = 0$. Equations (9.49d,e) give relations between the transverse field components as

$$Y = \frac{H_y}{E_x} \approx \frac{-H_x}{E_y} = \frac{\omega\epsilon}{\beta},$$
9.50

where Y is the wave admittance. Using (9.50) in (9.49a) and (9.49b) to eliminate H_x and H_y gives the following results:

$$j\omega^2 \epsilon \kappa E_x + (\beta^2 - \omega^2 \mu \epsilon) E_y = 0, \qquad 9.51a$$

$$(\beta^2 - \omega^2 \mu \epsilon) E_x - j \omega^2 \epsilon \kappa E_y = 0.$$
9.516

For a nontrivial solution for E_x and E_y the determinant of this set of equations must vanish:

$$\omega^{4} \epsilon^{2} \kappa^{2} - (\beta^{2} - \omega^{2} \mu \epsilon)^{2} = 0,$$

$$\beta_{\pm} = \omega \sqrt{\epsilon(\mu \pm \kappa)}.$$

9.52

0f

So there are two possible propagation constants, β_+ and β_- .

First consider the fields associated with β_+ , which can be found by substituting β_+ into (9.51a), or (9.51b):

$$j\omega^2\epsilon\kappa E_x+\omega^2\epsilon\kappa E_y=0,$$

or

$$E_y = -jE_x.$$

Then the electric field of (9.48a) must have the following form:

$$\bar{E}_{+} = E_0(\hat{x} - j\hat{y})e^{-j\beta_{+}z}, \qquad 9.53a$$

which is seen to be a right-hand circularly polarized plane wave. Using (9.50) gives the associated magnetic field as

$$\bar{H}_{+} = E_0 Y_{+} (j\hat{x} + \hat{y}) e^{-j\beta + z}, \qquad 9.53b$$

where Y_+ is the wave admittance for this wave:

$$Y_{+} = \frac{\omega\epsilon}{\beta_{+}} = \sqrt{\frac{\epsilon}{\mu + \kappa}}.$$
 9.53c

Similarly, the fields associated with β_{-} are left-hand circularly polarized:

$$\bar{E}_{-} = E_0(\hat{x} + j\hat{y})e^{-j\beta_{-}z}, \qquad 9.54a$$

$$\dot{H}_{-} = E_0 Y_{-} (-j\hat{x} + \hat{y}) e^{-j\beta_{-}\hat{x}}, \qquad 9.54b$$

where Y_{-} is the wave admittance for this wave:

$$Y_{-} = \frac{\omega\epsilon}{\beta_{-}} = \sqrt{\frac{\epsilon}{\mu - \kappa}}.$$
 9.54c

Thus we see that RHCP and LHCP plane waves are the source-free modes of the \hat{z} -biased ferrite medium, and these waves propagate through the ferrite medium with different propagation constants. As discussed in the previous section, the physical explanation for this effect is that the magnetic bias field creates a preferred direction for magnetic dipole precession, and one sense of circular polarization causes precession in this preferred direction while the other sense of polarization causes precession in the opposite direction. Also note that for an RHCP wave, the ferrite material can be represented with an effective permeability of $\mu + \kappa$, while for an LHCP wave the effective permeability is $\mu - \kappa$. In mathematical terms, we can state that $(\mu + \kappa)$ and $(\mu - \kappa)$, or β_+ and β_- , are the *eigenvalues* of the system of equations in (9.51), and that \vec{E}_+ and \vec{E}_- are the associated *eigenvectors*. When Iosses are present, the attenuation constants for RHCP and LHCP waves will also be different.

Now consider a linearly polarized electric field at z = 0, represented as the sum of an RHCP and an LHCP wave:

$$\bar{E}|_{z=0} = \hat{x}E_0 = \frac{E_0}{2}(\hat{x} - j\hat{y}) + \frac{E_0}{2}(\hat{x} + j\hat{y}). \qquad 9.55$$

The RHCP component will propagate in the z direction as $e^{-j\beta_+z}$, and the LHCP component will propagate as $e^{-j\beta_-3}$, so the total field of (9.55) will propagate as

$$\begin{split} \bar{E} &= \frac{E_0}{2} (\hat{x} - j\hat{y}) e^{-j\beta_+ z} + \frac{E_0}{2} (\hat{x} + j\hat{y}) e^{-j\beta_- z} \\ &= \frac{E_0}{2} \hat{x} (e^{-j\beta_+ z} + e^{-j\beta_- z}) - j \frac{E_0}{2} \hat{y} (e^{-j\beta_+ z} - e^{-j\beta_- z}) \\ &= E_0 \left[\hat{x} \cos\left(\frac{\beta_+ - \beta_-}{2}\right) z - \hat{y} \sin\left(\frac{\beta_+ - \beta_-}{2}\right) z \right] e^{-j(\beta_+ + \beta_-)z/2}. \quad 9.56 \end{split}$$

This is still a linearly polarized wave, but one whose polarization rotates as the wave propagates along the z-axis. At a given point along the z-axis the polarization direction measured from the x-axis is given by

$$\phi = \tan^{-1} \frac{E_y}{E_x} = \tan^{-1} \left[-\tan\left(\frac{\beta_+ - \beta_-}{2}\right) z \right] = -\left(\frac{\beta_+ - \beta_-}{2}\right) z. \qquad 9.57$$

This effect is called *Faraday rotation*. after Michael Faraday, who first observed this phenomenon during his study of the propagation of light through liquids that had magnetic properties. Note that for a fixed position on the z-axis, the polarization angle is fixed, unlike the case for a circularly polarized wave, where the polarization would rotate with time.

For $\omega < \omega_0$, μ and κ are positive and $\mu > \kappa$. Then $\beta_+ > \beta_-$, and (9.57) shows that ϕ becomes more negative as z increases, meaning that the polarization (direction of \overline{E}) rotates counterclockwise as we look in the +z direction. Reversing the bias direction (sign of H_0 and M_s) changes the sign of κ , which changes the direction of rotation to clockwise. Similarly, for +z bias, a wave traveling in the -z direction will rotate its polarization clockwise as we look in the direction of propagation (-z); if we were looking in the +z direction, however, the direction of rotation would be counterclockwise (same as a wave propagating in the +z direction). Thus, a wave that travels from z = 0to z = L and back again to z = 0 undergoes a total polarization rotation of 2ϕ , where ϕ is given in (9.57) with z = L. So, unlike the situation of a screw being driven into a block of wood and then backed out, the polarization does not "unwind" when the direction of propagation is reversed. Faraday rotation is thus seen to be a nonreciprocal effect.



EXAMPLE 9.1 Plane Wave Propagation in a Ferrite Medium

Consider an infinite ferrite medium with $4\pi M_s = 1800$ gauss, $\Delta H = 75$ oersted, $\epsilon_r = 14$, and $\tan \delta = 0.001$. If the bias field strength is $H_0 = 3570$ oersted, calculate and plot the phase and attenuation constants for RHCP and LHCP plane waves versus frequency, for f = 0 to 20 GHz.

Solution

The Larmor precession frequency is

$$f_0 = \frac{\omega_0}{2\pi} = (2.8 \text{ MHz/oersted})(3570 \text{ oersted}) = 10.0 \text{ GHz},$$

and

$$f_m = \frac{\omega_m}{2\pi} = (2.8 \text{ MHz/oersted}) (1800 \text{ gauss}) = 5.04 \text{ GHz}.$$

At each frequency we can compute the complex propagation constant as

$$\gamma_{\pm} = \alpha_{\pm} + j\beta_{\pm} = j\omega\sqrt{\epsilon(\mu\pm\kappa)},$$

where $\epsilon \approx \epsilon_0 \epsilon_r (1 - j \tan \delta)$ is the complex permittivity, and μ, κ are given by (9.25). The following substitution for ω_0 is used to account for ferrimagnetic loss:

$$\omega_0 \longleftarrow \omega_0 + j \frac{\mu_0 \gamma \Delta H}{2},$$

or $f_0 \longleftarrow f_0 + j \frac{(2.8 \text{ MHz/Oe})(75 \text{ Oe})}{2} = (10. + j0.105) \text{ GHz}.$

which is derived from (9.37) and (9.40). The quantities $(\mu \pm \kappa)$ can be simplified to the following, by using (9.25):

$$\mu + \kappa = \mu_0 \left(1 + \frac{\omega_m}{\omega_0 - \omega} \right),$$
$$\mu - \kappa = \mu_0 \left(1 + \frac{\omega_m}{\omega_0 + \omega} \right).$$

The phase and attenuation constants are plotted in Figure 9.8, normalized to the free-space wavenumber, k_0 .

Observe that β_+ and α_+ (for an RHCP wave) show a resonance near f = $f_0 = 10$ GHz: β_- and α_- (for an LHCP wave) do not, however, because the singularities in μ and κ cancel in the $(\mu - \kappa)$ term contained in γ_{-} . Also note from Figure 9.8 that a stopband (β_+ near zero, large α_+) exists for RHCP waves for frequencies between f_0 and $f_0 + f_m$ (between ω_0 and $\omega_0 + \omega_m$). For frequencies in this range, the above expression for $(\mu + \kappa)$ shows that this quantity is negative, and $\beta_{\pm} = 0$ (in the absence of loss), so an RHCP wave incident on such a ferrite medium would be totally reflected. О

Propagation Transverse to Bias (Birefringence)

24

Now consider the case where an infinite ferrite region is biased in the \hat{x} direction, transverse to the direction of propagation; the permeability tensor is given in (9.26). For plane wave fields of the form in (9.48), Maxwell's curl equations reduce to

$$j\beta E_y = -j\omega\mu_0 H_x, \qquad \qquad 9.58a$$





$$0 = -j\omega(-j\kappa H_y + \mu H_z), \qquad 9.58c$$

$$j_{\beta} \mathcal{H}_{y} = j \omega \epsilon E_{x}, \qquad 9.58d$$

$$-j\beta H_x = j\omega\epsilon E_y, \qquad 9.58e$$

$$0 = j\omega\epsilon E_z. \qquad 9.58f$$

Then $E_z = 0$, and $\nabla \cdot \hat{D} = 0$ since $\partial/\partial x = \partial/\partial y = 0$. Equations (9.58d,e) give an admittance relation between the transverse field components:

$$Y = \frac{H_y}{E_x} = \frac{-H_x}{E_y} = \frac{\omega\epsilon}{\beta},$$
 9.59

Using (9.59) in (9.58a,b) to eliminate H_x and H_y , and using (9.58c) in (9.58b) to eliminate H_z give the following results:

$$\beta^2 E_y = \omega^2 \mu_0 \epsilon E_y, \qquad 9.60a$$

$$\mu(\beta^2 - \omega^2 \mu \epsilon) E_x = -\omega^2 \epsilon \kappa^2 E_x. \qquad 9.60b$$

One solution to (9.60) occurs for

$$\beta_o = \omega \sqrt{\mu_0 \epsilon}, \qquad \qquad 9.61$$

with $E_x = 0$. Then the complete fields are

$$\bar{E}_o = \hat{y} E_0 e^{-j \hat{B}_o z}, \qquad 9.62a$$

$$\bar{H}_{o} = -\hat{x}E_{0}Y_{o}e^{-j\beta_{o}z}, \qquad 9.62b$$

since (9.59) shows that $H_y = 0$ when $E_x = 0$, and (9.58c) shows that $H_z = 0$ when $H_y = 0$. The admittance is

$$Y_o = \frac{\omega\epsilon}{\beta_o} = \sqrt{\frac{\epsilon}{\mu_0}}.$$
 9.63

This wave is called the *ordinary wave*, because it is unaffected by the magnetization of the ferrite. This happens whenever the magnetic field components transverse to the bias direction are zero $(H_y = H_z = 0)$. The wave propagates in either the +z or -z direction with the same propagation constant, which is independent of H_0 .

Another solution to (9.60) occurs for

$$eta_e = \omega \sqrt{\mu_e \epsilon},$$
 9.64

with $E_y = 0$, where μ_e is an effective permeability given by

$$\mu_{\rm e} = \frac{\mu^2 - \kappa^2}{\mu}.$$
 9.65

This wave is called the *extraordinary wave*, and is affected by the ferrite magnetization. Note that the effective permeability may be negative for certain values of ω , ω_0 . The electric field is

$$ar{E}_e = \hat{x} E_0 e^{-j\beta_e z}$$
. 9.660

Since $E_y = 0$, (9.58e) shows that $H_x = 0$. H_y can be found from (9.58d), and H_z from (9.58c), giving the complete magnetic field as

$$\tilde{H}_{\epsilon} = E_0 Y_e \left(\hat{y} + \hat{z} \frac{j\kappa}{\mu} \right) e^{-j\beta_e z}, \qquad 9.66b$$

where

$$Y_e = \frac{\omega\epsilon}{\beta_e} = \sqrt{\frac{\epsilon}{\mu_e}}.$$
 9.67

These fields constitute a linearly polarized wave, but note that the magnetic field has a component in the direction of propagation. Except for the existence of H_z , the extraordinary wave has electric and magnetic fields that are perpendicular to the corresponding fields of the ordinary wave. Thus, a wave polarized in the y direction will have a propagation constant β_o (ordinary wave), but a wave polarized in the x direction will have a propagation constant β_e (extraordinary wave). This effect, where the propagation constant depends on the polarization direction, is called *birefringence* [2]. Birefringence often occurs in optics work, where the index of refraction can have different values depending on the polarization. The double image seen through a calcite crystal is an example of this effect.

From (9.65) we can see that μ_{ϵ} , the effective permeability for the extraordinary wave, can be negative if $\kappa^2 > \mu^2$. This condition depends on the values of ω , ω_0 , and ω_m , or f. H_0 , and M_s , but for a fixed frequency and saturation magnetization there will always be some range of bias field for which $\mu_{\epsilon} < 0$ (ignoring loss). When this occurs β_e will become imaginary, as seen from (9.64), which implies that the wave will be cutoff, or evanescent. An \hat{x} polarized plane wave incident at the interface of such a ferrite region would be totally reflected. The effective permeability is plotted versus bias field strength in Figure 9.9, for several values of frequency and saturation magnetization.



FIGURE 9.9 Effective permeability, μ_{e} , versus bias field, H_0 , for various saturation magnetizations and frequencies.

9.3 PROPAGATION IN A FERRITE-LOADED RECTANGULAR WAVEGUIDE

In the previous section we introduced the effects of a ferrite material on electromagnetic waves by considering the propagation of plane waves in an infinite ferrite medium. In practice, however, most ferrite components use waveguide or other types of transmission lines loaded with ferrite material. Most of these geometries are very difficult to analyze. Nevertheless, it is worth the effort to treat some of the easier cases, involving ferrite-loaded rectangular waveguides. in order to quantitatively demonstrate the operation and design of several types of practical ferrite components.

TEmo Modes of Waveguide with a Single Ferrite Slab

We first consider the geometry shown in Figure 9.10, where a rectangular waveguide is loaded with a vertical slab of ferrite material, biased in the \hat{y} direction. This geometry and its analysis will be used in later sections to treat the operation and design of resonance isolators, field-displacement isolators, and remanent (nonreciprocal) phase shifters.

In the ferrite slab, Maxwell's equations can be written as

$$\nabla \times \bar{E} = -j\omega[\mu]\bar{H}, \qquad 9.68a$$

$$abla imes ar{H} = j\omega\epsilonar{E},$$
 9.68b

where $[\mu]$ is the permeability tensor for \hat{y} bias, as given in (9.27). Then if we let $\bar{E}(x, y, z) = [\bar{e}(x, y) + \hat{z}e_z(x, y)]e^{-j\beta z}$ and $\bar{H}(x, y, z) = [\bar{h}(x, y) + \hat{z}h_z(x, y)]e^{-j\beta z}$, (9.68) reduces to

$$\frac{\partial e_z}{\partial y} + j\beta e_y = -j\omega(\mu h_x + j\kappa h_z), \qquad 9.69a$$

$$-jeta e_x - rac{\partial e_x}{\partial x} = -j\omega\mu_0 h_y,$$
 9.69b





9.10 Geometry of a rectangular waveguide loaded with a transversely biased ferrite slab.

$$\frac{\partial e_y}{\partial x} - \frac{\partial e_x}{\partial y} = -j\omega(-j\kappa h_x + \mu h_z), \qquad 9.69c$$

$$\frac{\partial h_z}{\partial y} + j\beta h_y = j\omega\epsilon e_x, \qquad 9.69d$$

$$-j\beta h_x - \frac{\partial h_z}{\partial x} = j\omega\epsilon e_y, \qquad \qquad 9.69e$$

$$\frac{\partial h_y}{\partial x} - \frac{\partial h_x}{\partial y} = j\omega\epsilon e_x. \qquad 9.69f$$

For TE_{mo} modes, we know that $E_z = 0$ and $\partial/\partial y = 0$. Then (9.69b) and (9.69d) imply that $e_x = h_y = 0$ (since $\beta^2 \neq \omega^2 \mu_0 \epsilon$ for a waveguide mode) and so (9.69) reduces to three equations:

$$j\beta e_y = -j\omega(\mu h_x + j\kappa h_z), \qquad 9.70a$$

$$\frac{\partial e_y}{\partial x} = -j\omega(-j\kappa h_x + \mu h_z), \qquad 9.70b$$

$$j\omega\epsilon e_y = -j\beta h_x - rac{\partial h_z}{\partial x}.$$
 9.70c

We can solve (9.70a,b) for h_x and h_z as follows. Multiply (9.70a) by μ and (9.70b) by $-j\kappa$, then add to obtain

$$h_x = \frac{1}{\omega \mu \mu_e} \left(-\mu \beta e_y + \kappa \frac{\partial e_y}{\partial x} \right).$$
9.71a

Now multiply (9.70a) by $j\kappa$ and (9.71a) by μ , then add to obtain

$$h_z = \frac{i}{\omega \mu \mu_e} \left(-\kappa \beta e_y + \mu \frac{\partial e_y}{\partial x} \right).$$
 9.71b

where $\mu_e = (\mu^2 - \kappa^2)/\mu$. Substituting (9.71) into (9.70c) gives a wave equation for e_y :

$$j\omega\epsilon e_{y} = \frac{-j\beta}{\omega\mu\mu_{e}} \left(-\mu\beta e_{y} + \kappa\frac{\partial e_{y}}{\partial x}\right) - \frac{j}{\omega\mu\mu_{e}} \left(-\kappa\beta\frac{\partial e_{y}}{\partial x} + \mu\frac{\partial^{2}e_{y}}{\partial x^{2}}\right).$$
$$\left(\frac{\partial^{2}}{\partial x^{2}} + k_{f}^{2}\right)e_{y} = 0,$$
9.72

ог

where k_f is defined as a cutoff wavenumber for the ferrite:

$$k_f^2 = \omega^2 \mu_e \epsilon - \beta^2. \qquad \qquad 9.73$$

We can obtain the corresponding results for the air regions by letting $\mu = \mu_0, \kappa = 0$, and $\epsilon_r = 1$, to obtain

$$\left(\frac{\partial^2}{\partial x^2} + k_a^2\right)e_y = 0, \qquad 9.74$$

where k_a is the cutoff wavenumber for the air regions:

$$k_a^2 = k_0^2 - \beta^2 9.75$$

The magnetic field in the air region is given by

$$h_{x} = \frac{-\beta}{\omega\mu_{0}}e_{y} = \frac{-1}{Z_{w}}e_{y}, \qquad 9.76a$$
$$h_{z} = \frac{j}{\omega\mu_{0}}\frac{\partial e_{y}}{\partial x}. \qquad 9.76b$$

The solutions for e_y in the air-ferrite-air regions of the waveguide are then

$$e_y = \begin{cases} A \sin k_a x, & \text{for } \mathbf{0} < \mathbf{i} < \mathbf{c}, \\ B \sin k_f (x - c) + C \sin k_f (c + t - x), & \text{for } c < x < c + t, \\ D \sin k_a (a - x), & \text{for } c + t < x < \mathbf{a}, \end{cases}$$

which have been constructed to facilitate the enforcement of boundary conditions at x = 0, c, c + t, and a [3]. We will also need h_z , which can be found from (9.77a), (9.71b), and (9.76b):

$$h_{z} = \begin{cases} (jk_{a}A/\omega\mu_{o})\cos k_{a}x, & \text{for } 0 < x < c, \\ (j/\omega\mu\mu_{e})\{-\kappa\beta[B\sin k_{f}(x-c)+C\sin k_{f}(c+t-x)] \\ +\mu k_{f}[B\cos k_{f}(x-c)-C\cos k_{f}(c+t-x)]\}, & \text{for } c < x < c+t, \\ (-jk_{a}D/\omega\mu_{o})\cos k_{a}(a-x), & \text{for } c+t < x < a, \\ 9 77b \end{cases}$$

Matching e_y and h_z at x = c and x = c+t = a-d gives four equations for the constants A, B, C, D:

$$A\sin k_a c = C\sin k_f t, \qquad 9.78a$$

$$B\sin k_f t = D\sin k_a d, \qquad 9.78b$$

$$A\frac{k_a}{\mu_o}\cos k_a c = B\frac{k_f}{\mu_e} - C\frac{1}{\mu\mu_e}(\kappa\beta\sin k_f t + \mu k_f\cos k_f t), \qquad 9.78c$$

$$B\frac{1}{\mu\mu_e}(-\kappa\beta\sin k_f t + \mu k_f\cos k_f t) - C\frac{k_f}{\mu_e} = -D\frac{k_a}{\mu_o}\cos k_a d. \qquad 9.78d$$

Solving (9.78a) and (9.78b) for C and D, substituting into (9.78c) and (9.78d), and then eliminating A or B gives the following transcendental equation for the propagation constant, β :

$$\left(\frac{k_f}{\mu_e}\right)^2 + \left(\frac{\kappa\beta}{\mu\mu_e}\right)^2 - k_a \cot k_a c \left(\frac{k_f}{\mu_o\mu_e} \cot k_f t - \frac{\kappa\beta}{\mu_o\mu\mu_e}\right) - \left(\frac{k_a}{\mu_o}\right)^2 \\ \times \cot k_a c \cot k_a d - k_a \cot k_a d \left(\frac{k_f}{\mu_o\mu_e} \cot k_f t + \frac{\kappa\beta}{\mu_o\mu\mu_e}\right) = 0.$$
9.79

After using (9.73) and (9.75) to express the cutoff wavenumbers k_f and k_a in terms of β , (9.79) can be solved numerically. The fact that (9.79) contains terms that are odd in $\kappa\beta$ indicates that the resulting wave propagation will be nonreciprocal, since changing the direction of the bias field (which is equivalent to changing the direction of propagation) changes the sign of κ , which leads to a different solution for β . We will identify these two solutions as β_+ and β_- , for positive bias and propagation in the +2 direction (positive κ), or in the -z direction (negative κ), respectively. The effects of magnetic loss can easily be included by allowing ω_0 to be complex, as in (9.37). In later sections we will also need to evaluate the electric field in the guide, as given in (9.77a). If we choose the arbitrary amplitude constant as A, then B, C, and D can be found in terms of A by using (9.78a), (9.78b), and (9.78c). Note from (9.75) that if $\beta > k_a$, then k_a will be imaginary. In this case, the sin $k_a x$ function of (9.77a) becomes $j \sinh |k_a|x$, indicating an almost exponential variation in the field distribution.

A useful approximate result can be obtained for the differential phase shift, $\beta_{+} - \beta_{-}$, by expanding β in (9.79) in a Taylor series about t = 0. This can be accomplished with implicit differentiation after using (9.73) and (9.75) to express k_f and k_a in terms of β [4]. The result is

$$\beta_{+} - \beta_{-} = \frac{-2k_{c}t\kappa}{a\mu}\sin 2k_{c}c = -2k_{c}\frac{\kappa}{\mu}\frac{\Delta S}{S}\sin 2k_{c}c, \qquad 9.80$$

where $k_c = \pi/a$ is the cutoff frequency of the empty guide, and $\Delta S/S = t/a$ is the *filling factor*, or ratio of slab cross-sectional area to waveguide cross-sectional area. Thus, this formula can be applied to other geometries such as waveguides loaded with small ferrite strips or rods, although the appropriate demagnetization factors may be required for some ferrite shapes. The result in (9.80) is accurate, however, only for very small ferrite cross sections, typically for $\Delta S/S < 0.01$.

This same technique can be used to obtain an approximate expression for the forward and reverse attenuation constants, in terms of the imaginary parts of the susceptibilities defined in (9.39):

$$\alpha_{\pm} = \frac{\Delta S}{S\beta_o} (\beta_o^2 \chi_{xx}'' \sin^2 k_c x + k_c^2 \chi_{zz}'' \cos^2 k_c x \mp \chi_{xy}'' k_c \beta_o \sin 2k_c x), \qquad 9.81$$

where $\beta_{0} = \sqrt{k_{0}^{2} - k_{c}^{2}}$ is the propagation constant of the empty guide. This result will be useful in the design of resonance isolators. Both (9.80) and (9.81) can also be derived using a perturbation method with the empty waveguide fields [4], and so are usually referred to as the perturbation theory results.

TE_{mo} Modes of Waveguide with Two Symmetrical Ferrite Slabs

A related geometry is the rectangular waveguide loaded with two symmetrically placed ferrite slabs, as shown in Figure 9.11. With equal but opposite \hat{y} -directed bias fields on the ferrite slabs, this configuration provides a useful model for the nonreciprocal remanent phase shifter, which will be discussed in Section 9.5. Its analysis is very similar to that of the single-slab geometry.

Since the h_y and h_z fields (including the bias fields) are antisymmetric about the midplane of the waveguide at x = a/2, a magnetic wall can be placed at this point. Then we only need to consider the region for 0 < x < a/2. The electric field in this region can be written as

$$e_y = \begin{cases} A \sin k_a x, & 0 < x < c, \\ B \sin k_f (x - c) + C \sin k_f (c + t - x), & c < x < c + t, \\ D \cos k_a (a/2 - x), & c + t < x < a/2, \end{cases}$$
9.82a



FIGURE 9.11 Geometry of a rectangular waveguide loaded with two symmetrical ferrite states

which is similar in form to (9.77a), except that the expression for c + t < x < a/2 was constructed to have a maximum at x = a/2 (since h_z must be zero at x = a/2). The cutoff wavenumbers k_f and k_a are defined in (9.73) and (9.75).

Using (9.71) and (9.76) gives the h_z field as,

$$h_{z} = \begin{cases} (jk_{a}A/\omega\mu_{a})\cos k_{a}x, & 0 < x < c, \\ (j/\omega\mu\mu_{e})\{-\kappa\beta[B\sin k_{f}(x-c) + C\sin k_{f}(c+t-x)] \\ +\mu k_{f}[B\cos k_{f}(x-c) - C\cos k_{f}(c+t-x)]\}, & c < x < c+t, \\ (jk_{a}D/\omega\mu_{o})\sin k_{a}(a/2-x), & c+t < x < a/2, \\ & 0 \text{ and } \end{cases}$$

Matching e_y and h_z at x = c and x = c + t = a/2 - d gives four equations for the constants A, B, C, D:

$$A\sin k_a c \approx C\sin k_f t, \qquad 9.83a$$

$$B\sin k_f t = D\cos k_a d, \qquad 9.83b$$

Т

$$A\frac{k_a}{\mu_o}\cos k_a c = B\frac{k_f}{\mu_e} - C\frac{1}{\mu\mu_e}(\kappa\beta\sin k_f t + \mu k_f\cos k_f t), \qquad 9.83e$$

$$\frac{B}{\mu\mu_e}(-\kappa\beta\sin k_f t + \mu k_f\cos k_f t) - C\frac{k_f}{\mu_e} = D\frac{k_a}{\mu_o}\sin k_a d. \qquad 9.83d$$

Reducing these results gives a transcendental equation for the propagation constant, β :

$$\left(\frac{k_f}{\mu_e}\right)^2 + \left(\frac{\kappa\beta}{\mu\mu_e}\right)^2 - k_a \cot k_a c \left(\frac{k_f}{\mu_o\mu_e} \cot k_f t - \frac{\kappa\beta}{\mu_o\mu\mu_e}\right) + \left(\frac{k_a}{\mu_o}\right)^2$$

$$\times \cot k_a c \tan k_a d + k_a \tan k_a d \left(\frac{k_f}{\mu_o \mu_e} \cot k_f t + \frac{\kappa \beta}{\mu_o \mu \mu_e}\right) = 0 \qquad 9.84$$

This equation can be solved numerically for β . As in (9.79) for the single-slab case, κ and β appear in (9.84) only as $\kappa\beta$, κ^2 , or β^2 , which implies nonreciprocal propagation, since changing the sign of κ (or bias fields) necessitates a change in sign for β (propagation direction) for the same root. At first glance it may seem that, for

the same waveguide and slab dimensions and parameters, two slabs would give twice the phase shift of one slab, but this is generally untrue because the fields are highly concentrated in the ferrite regions.

9.4 FERRITE ISOLATORS

One of the most useful microwave ferrite components is the *isolator*, which is a two-port device having unidirectional transmission characteristics. The S matrix for an ideal isolator has the form

$$[S] = \begin{bmatrix} 0 & 0 \\ 1 & 0 \end{bmatrix}, \qquad 9.85$$

indicating that both ports are matched, but transmission occurs only in the direction from port 1 to port 2. Since [S] is not unitary, the isolator must be lossy. And, of course, [S] is not symmetric, since an isolator is a nonreciprocal component.

A common application uses an isolator between a high-power source and a load to prevent possible reflections from damaging the source. An isolator can be used in place of a matching or tuning network, but it should be realized that any power reflected from the load will be absorbed by the isolator, as opposed to being reflected back to the load, which is the case when a matching network is used.

Although there are several types of ferrite isolators, we will concentrate on the resonance isolator and the field displacement isolator. These devices are of practical importance, and can be analyzed and designed using the results for the ferrite slab-loaded waveguide of the previous section.

Resonance Isolators

We have seen that a circularly polarized plane wave rotating in the same direction as the precessing magnetic dipoles of a ferrite medium will have a strong interaction with the material, while a circularly polarized wave rotating in the opposite direction will have a weaker interaction. Such a result was illustrated in Example 9.1, where the attenuation of a circularly polarized wave was very large near the gyromagnetic resonance of the ferrite, while the attenuation of a wave propagating in the opposite direction was very small. This effect can be used to construct an isolator; such isolators must operate near gyromagnetic resonance and so are called *resonance isolators*. Resonance isolators usually consist of a ferrite slab or strip mounted at a certain point in a waveguide. We will discuss the two isolator geometries shown in Figure 9.12.

Ideally, the RF fields inside the ferrite material should be circularly polarized. In an empty rectangular waveguide the magnetic fields of the TE_{10} mode can be written as

$$H_x = \frac{j\beta_o}{k_c} A \sin k_c x e^{-j\beta_o z},$$
$$H_z = A \cos k_c x e^{-j\beta_o z},$$

where $k_c = \pi/a$ is the cutoff wavenumber and $\beta_o = \sqrt{k_o^2 - k_c^2}$ is the propagation constant of the empty guide. Since a circularly polarized wave must satisfy the condition

that $H_x/H_z = \pm j$, the location, x, of the CP point of the empty guide is given by

$$\tan k_c x = \pm \frac{k_c}{\beta_o}.$$
 9.86

Ferrite loading, however, may perturb the fields so that (9.86) may not give the actual optimum position, or it may prevent the internal fields from being circularly polarized for any position.

First consider the full-height *E*-plane slab geometry of Figure 9.12a; we can analyze this case using the exact results from the previous section. Alternatively, we could use the perturbation result of (9.81), but this would require the use of a demagnetization factor for h_a , and would be less accurate than the exact results. Thus, for a given set of parameters, (9.79) can be solved numerically for the complex propagation constants of the forward and reverse waves of the ferrite-loaded guide. It is necessary to include the effect of magnetic loss, which can be done by using (9.37) for the complex resonant frequency, ω_0 , in the expressions for μ and κ . The imaginary part of ω_0 can be related to the linewidth, ΔH , of the ferrite through (9.40). Usually the waveguide width, *a*, frequency, ω , and ferrite parameters $4\pi M_s$, and ϵ_r will be fixed, and the bias field and slab position and thickness will be determined to give the optimum design.

Ideally, the forward attenuation constant (α_+) would be zero, with a nonzero attenuation constant (α_-) in the reverse direction. But for the *E*-plane ferrite slab there is no position x = c where the fields are perfectly CP in the ferrite (this is because the demagnetization factor $N_x \simeq 1$ [4]). Hence the forward and reverse waves both contain an RHCP component and an LHCP component, so ideal attenuation characteristics cannot be obtained. The optimum design, then, generally minimizes the forward attenuation, which determines the slab position. Alternatively, it may be desired to maximize the ratio of the reverse to forward attenuations. Since the maximum reverse attenuation generally does not occur at the same slab position as the minimum forward attenuation, such a design will involve a trade-off of the forward loss.




For a long, thin slab, the demagnetization factors are approximately those of a thin disk: $N_x \simeq 1$, $N_y = N_z = 0$. It can then be shown via the Kittel equation of (9.45) that the gyromagnetic resonance frequency of the slab is given by

$$\omega = \sqrt{\omega_0(\omega_0 + \omega_m)}.$$
 9.87

which determines H_0 , given the operating frequency and saturation magnetization. This is an approximate result: the transcendental equation of (9.79) accounts for demagnetization exactly, so the actual internal bias field, H_0 , can be found by numerically solving (9.79) for the attenuation constants for values of H_0 near the approximate value given by (9.87),

Once the slab position, c, and bias field, H_o , have been found the slab length, L, can be chosen to give the desired total reverse attenuation (or isolation) as $(\alpha_{-})L$. The slab thickness can also be used to adjust this value. Typical numerical results are given in Example 9.2.

One advantage of this geometry is that the full-height slab is easy to bias with an external C-shaped permanent magnet, with no demagnetization factor. But it suffers from several disadvantages:

- Zero forward attenuation cannot be obtained because the internal magnetic field is not truly circularly polarized.
- The bandwidth of the isolator is relatively parrow, dictated essentially by the linewidth, ΔH , of the ferrite.
- The geometry is not well suited for high-power applications because of poor heat transfer from the middle of the slab, and an increase in temperature will cause a change in $M_{s_{\gamma}}$ which will degrade performance.

The first two problems noted above can be remedied to a significant degree by adding a dielectric loading slab; see reference [5] for details.



EXAMPLE 9.2 Ferrite Resonance Isolator Design

Design an *E*-plane resonance isolator in X-band waveguide to operate at 10 GHz with a minimum forward insertion loss and 30 dB reverse attenuation. Use a 0.5 mm thick ferrite slab with $4\pi M_s = 1700 \text{ G}, \Delta H = 200 \text{ O}\epsilon$, and $\epsilon_r = 13$. Determine the bandwidth for which the reverse attenuation is at least 27 dB.

Solution

The complex roots of (9.79) were found numerically using an interval-halving routine followed by a Newton-Raphson iteration. The approximate bias field, H_0 , given by (9.87) is 2820 Oe, but numerical results indicate the actual field to be closer to 2840 Oe for resonance at 10 GHz. Figure 9.13a shows the calculated forward (α_+) and reverse (α_-) attenuation constants at 10 GHz versus slab position, and it can be seen that the minimum forward attenuation occurs for c/a = 0.125; the reverse attenuation at this point is $\alpha_- = 12.4$ dB/cm. Figure 9.13b shows the attenuation constants versus frequency for this slab position. For a total reverse attenuation of 20 dB, the length of the slab must be

$$L = \frac{30 \text{ dB}}{12.4 \text{ dB/cm}} = 2.4 \text{ cm}.$$



 $4\pi M_{g} = 1700$ G $H_{0} = 2840$ Oe $\Delta H = 200$ Oe a = 2.286 cm t = 0.05 cm $\epsilon_{g} = 13$

FIGURE 9.13 Forward and reverse attenuation constants for the resonance isolator of Example 9.2. (a) Versus slab position. (b) Versus frequency.

For the total reverse attenuation to be at least 27 dB, we must have

$$\alpha_{-} > \frac{27 \text{ dB}}{2.4 \text{ cm}} = 11.3 \text{ dB/cm}$$

So the bandwidth according to the above definition is, from the data of Figure 9.13b, less than 2%. This figure could be improved by using a ferrite with a larger linewidth, at the expense of a longer or thicker slab and a higher forward attenuation. \bigcirc

Next we consider a resonance isolator using the *H*-plane slab geometry of Figure 9.12b. If the slab is much thinner than it is wide, the demagnetization factors will approximately be $N_x = N_z = 0$, $N_y = 1$. This means that a stronger applied bias field will be required to produce the internal field, H_0 , in the y direction. But the RF magnetic field components, h_x and h_z , will not be affected by the air-ferrite boundary since $N_x = N_z = 0$, and perfect circular polarized fields will exist in the ferrite when it is positioned at the CP point of the empty guide, as given by (9.86). Another advantage of this geometry is that it has better thermal properties than the *E*-plane version, since the ferrite slab has a large surface area in contact with a waveguide wall for heat dissipation.

Unlike the full-height E-plane slab case, the H-plane geometry of Figure 9.12b cannot be analyzed exactly. But if the slab occupies only a very small fraction of the total guide cross section ($\Delta S/S \ll 1$, where ΔS and S are the cross sectional areas

of the slab and waveguide, respectively), the perturbational result for α_+ in (9.81) can be used with reasonable results. This expression is given in terms of the susceptibilities $\chi_{xx} = \chi'_{xx} - j\chi''_{xx}$, $\chi_{zz} = \chi'_{zz} - j\chi''_{zz}$, and $\chi_{xy} = \chi''_{xy} + j\chi'_{xy}$, as defined for a \hat{y} -biased ferrite in a manner similar to (9.22). For ferrite shapes other than a thin *H*-plane slab, these susceptibilities would have to be modified with the appropriate demagnetization factors, as in (9.43) [4].

As seen from the susceptibility expressions of (9.22), gyromagnetic resonance for this geometry will occur when $\omega = \omega_0$, which determines the internal bias field, H_0 . The center of the slab is positioned at the circular polarization point of the empty guide, as given by (9.86). This should result in a near-zero forward attenuation constant. The total reverse attenuation, or isolation, can be controlled with either the length, L, of the ferrite slab or its cross section ΔS , since (9.81) shows α_{\pm} is proportional to $\Delta S/S$. If $\Delta S/S$ is too large, however, the purity of circular polarization over the slab cross section will be degraded, and forward loss will increase. One practical alternative is to use a second identical ferrite slab on the top wall of the guide, to double $\Delta S/S$ without significantly degrading polarization purity.

The Field Displacement Isolator

Another type of isolator uses the fact that the electric field distributions of the forward and reverse waves in a ferrite slab-loaded waveguide can be quite different. As illustrated in Figure 9.14, the electric field for the forward wave can be made to vanish at the side of the ferrite slab at x = c + t, while the electric field of the reverse wave can be quite large at this same point. Then if a thin resistive sheet is placed in this position, the forward wave will be essentially unaffected while the reverse wave will be attenuated. Such an isolator is called a *field displacement isolator*, high values of isolation with a relatively compact device can be obtained with bandwidths on the order





of 10%. Another advantage of the field displacement isolator over the resonance isolator is that a much smaller bias field is required, since it operates well below resonance,

The main problem in designing a field displacement isolator is to determine the design parameters that produce field distributions like those shown in Figure 9.14. The general form of the electric field is given in (9.77a), from the analysis of the ferrite slab-loaded waveguide. This shows that for the electric field of the forward wave to have a sinusoidal dependence for c + t < x < a, and to vanish at x = c + t, the cutoff wavenumber k_a^+ must be real and satisfy the condition that

$$k_a^+ = \frac{\pi}{d}.$$
 9.88

where d = a - c - t. In addition, the electric field of the reverse wave should have a hyperbolic dependence for c + t < x < a, which implies that k_a^- must be imaginary. Since from (9.75), $k_a^2 = k_0^2 - \beta^2$, the above conditions imply that $\beta^+ < k_0$ and $\beta^- > k_0$, where $k_0 = \omega \sqrt{\mu_0 \epsilon_0}$. These conditions on β_{∞} depend critically on the slab position, which must be determined by numerically solving (9.79) for the propagation constants. The slab thickness also affects this result, but less critically; a typical value is t = a/10.

It also turns out that in order to satisfy (9.88), to force $E_y = 0$ at x = c + t, $\mu_e = (\mu^2 - \kappa^2)/\mu$ must be negative. This requirement can be intuitively understood by thinking of the waveguide mode for c + t < x < a as a superposition of two obliquely traveling plane waves. The magnetic field components H_x and H_z of these waves are both perpendicular to the bias field, a situation which is similar to the extraordinary plane waves discussed in Section 9.2, where it was seen that propagation would not occur for $\mu_e < 0$. Applying this cutoff condition to the ferrite-loaded waveguide will allow a null in E_y for the forward wave to be formed at x = c + t.

The condition that μ_e be negative depends on the frequency, saturation magnetization, and bias field. Figure 9.9 shows the dependence of μ_e versus bias field for several frequencies and saturation magnetization. This type of data can be used to select the saturation magnetization and bias field to give $\mu_e < 0$ at the design frequency. Observe that higher frequencies will require a ferrite with higher saturation magnetization, and a higher bias field, but $\mu_e < 0$ always occurs before the resonance in μ_e at $\sqrt{\omega_0(\omega_0 + \omega_m)}$. Further design details will be given in the following example.



EXAMPLE 9.3 Field Displacement Isolator Design

Design a field displacement isolator in X-band waveguide to operate at 11 GHz. The ferrite has $4\pi M_s = 3000$ G, and $\epsilon_r = 13$. Ferrite loss can be ignored.

Solution

We first determine the internal bias field, H_0 , such that $\mu_e < 0$. This can be found from Figure 9.9, which shows μ_e/μ_o versus H_0 for $4\pi M_s = 3000$ G at 11 GHz. We see that $H_0 = 1200$ Oe should be sufficient. Also note from this figure that a ferrite with a smaller saturation magnetization would require a much larger bias field,

Next we determine the slab position, c/a, by numerically solving (9.79) for the propagation constants, β_{\pm} , as a function of c/a. The slab thickness was set to t = 0.25 cm, which is approximately a/10. Figure 9.15a shows the



FIGURE 9.15 Propagation constants and electric field distribution for the field displacement isolator of Example 9.3. (a) Forward and reverse propagation constants versus slab position. (b) Electric field amplitudes for the forward and reverse waves.

resulting propagation constants, as well as the locus of points where β_+ and c/a satisfy the condition of (9.88). The intersection of β_+ with this locus will insure that $E_y = 0$ at x = c + t for the forward wave; this intersection occurs for a slab position of c/a = 0.028. The resulting propagation constants are $\beta_+ = 0.724k_0 < k_0$ and $\beta_- = 1.607k_0 > k_0$.

The electric fields are plotted in Figure 9.15b. Note that the forward wave has a null at the face of the ferrite slab, while the reverse wave has a peak

530 Chapter 9: Theory and Design of Ferrimagnetic Components

(the relative amplitudes of these fields are arbitrary). Then a resistive sheet can be placed at this point to attenuate the reverse wave. The actual isolation will depend on the resistivity of this sheet; a value of 75 Ω per square is typical.

9.5 ____ FERRITE PHASE SHIFTERS

Another important application of ferrite materials is in *phase shifters*, which are two-port components that provide variable phase shift by changing the bias field of the ferrite. (Microwave diodes and FETs can also be used to implement phase shifters; see Section 10.3.) Phase shifters find application in test and measurements systems, but the most significant use is in phased array antennas where the antenna beam can be steered in space by electronically controlled phase shifters. Because of this demand, many different types of phase shifters have been developed, both reciprocal (same phase shift in either direction) and nonreciprocal [2], [6]. One of the most useful designs is the latching (or *remanent*) nonreciprocal phase shifter using a ferrite toroid in a rectangular waveguide; we can analyze this geometry with a reasonable degree of approximation using the double ferrite slab geometry discussed in Section 9.3. Then we will qualitatively discuss the operation of a few other types of phase shifters.

Nonreciprocal Latching Phase Shifter

The geometry of a latching phase shifter is shown in Figure 9.16; it consists of a toroidal ferrite core symmetrically located in the waveguide with a bias wire passing through its center. When the ferrite is magnetized, the magnetization of the sidewalls of the toroid will be oppositely directed and perpendicular to the plane of circular polarization of the RF fields. Since the sense of circular polarization is also opposite on opposite sides of the waveguide, a strong interaction between the RF fields and the ferrite can be obtained. Of course, the presence of the ferrite perturbs the waveguide fields (the fields tend to concentrate in the ferrite), so the circular polarization point does not occur at tan $k_c x = k_c/\beta_0$, as it does for an empty guide.

In principle, such a geometry can be used to provide a continuously variable (analog) phase shift by varying the bias current. But a more useful technique employs the magnetic





hysteresis of the ferrite to provide a phase shift that can be switched between two values (digital). A typical hysteresis curve is shown in Figure 9.17, showing the variation in magnetization, M, with bias field, H_0 . When the fertite is initially demagnetized and the bias field is off, both M and H_0 are zero. As the bias field is increased, the magnetization increases along the dashed line path until the ferrite is magnetically saturated, and $M = M_{s}$. If the bias field is now reduced to zero, the magnetization will decrease to a remanent condition (like a permanent magnet), where $M = M_r$. A bias field in the opposite direction will saturate the ferrite with $M = -M_s$, whereupon the removal of the bias field will leave the ferrite in a remanent state with $M = -M_{\tau}$. Thus we can "latch" the ferrite magnetization in one of two states, where $M = \pm M_r$. giving a digital phase shift. The amount of differential phase shift between these two states is controlled by the length of the ferrite toroid. In practice, several sections having individual bias lines and decreasing lengths are used in series to give binary differential phase shifts of 180° , 90° , 45° , etc. to as fine a resolution as desired (or can be afforded). An important advantage of the latching mode of operation is that the bias current does not have to be continuously applied, but only pulsed with one polarity or the other to change the polarity of the remanent magnetization; switching speeds can be on the order of a few microseconds. The bias wire can be oriented perpendicular to the electric field in the guide, with a negligible perturbing effect. The top and bottom walls of the ferrite toroid have very little magnetic interaction with the RF fields because the magnetization is not perpendicular to the plane of circular polarization, and the top and bottom magnetizations are oppositely directed. So these walls provide mainly a dielectric loading effect, and the essential operating features of the remanent phase shifters can be obtained by considering the simpler dual ferrite slab geometry of Section 9.3.

For a given operating frequency and waveguide size, the design of a remanent dual slab phase shifter mainly involves the determination of the slab thickness, t, the spacing between the slabs, s = 2d = a - 2c - 2t (see Figure 9.11), and the length of the slabs for the desired phase shift. This requires the propagation constants, β_{\pm} , for the dual slab geometry, which can be numerically evaluated from the transcendental equation of (9.84). This equation requires values for μ and κ , which can be determined from (9.25)



FIGURE 9.17 A hysteresis curve for a ferrite toroid.

for the remanent state by setting $H_0 = 0$ ($\omega_0 = 0$) and $M_s = M_r$ ($\omega_m = \mu_0 \gamma M_s$

$$\mu = \mu_0, \qquad 9.89^{\circ}_{\circ}, \qquad 9$$

The differential phase shift, $\beta_+ - \beta_-$, is linearly proportional to κ , for κ/μ_0 up to about 0.5. Then, since κ is proportional to M_r , as seen by (9.89b), it follows that a shorter ferrite can be used to provide a given phase shift if a ferrite with a higher remanent magnetization is selected. The insertion loss of the phase shifter decreases with length, but is a function of the ferrite linewidth, ΔH . A figure of merit commonly used to characterize phase shifters is the ratio of phase shift to insertion loss, measured in degrees/dB.



EXAMPLE 9.4 Remanent Phase Shifter Design

Design a two-slab remanent phase shifter at 10 GHz using X-band waveguide with ferrite having $4\pi M_r = 1786$ G and $\epsilon_r = 13$. Assume that the ferrite slabs are spaced 1 mm apart. Determine the slab thicknesses for maximum differential phase shift, and the lengths of the slabs for 180° and 90° phase shifter sections.

Solution From (9.89) we have that

$$\frac{\mu}{\mu_0} = 1,$$

$$\frac{\kappa}{\mu_0} = \pm \frac{\omega_m}{\omega} = \pm \frac{(2.8 \text{ MHz/Oe})(1786 \text{ G})}{10.000 \text{ MHz}} = \pm 0.5$$

Using a numerical root-finding technique, such as interval halving, we can solve (9.84) for the propagation constants β_+ and β_- by using positive and negative values of κ . Figure 9.18 shows the resulting differential phase shift. $(\beta_+ - \beta_-)/k_0$, versus slab thickness, *t*. for several slab spacings. Observe that the phase shift increases as the spacing, *s*, between the slabs decreases, and as the slab thickness increases, for t/a up to about 0.12.

From the curve in Figure 9.18 for s = 1 mm, we see that the optimum slab thickness for maximum phase shift is t/a = 0.12, or t = 2.74 mm, since a = 2.286 cm for X-band guide. The corresponding normalized differential phase shift is 0.40, so

$$\beta_{+} - \beta_{-} = 0.4k_0 = 0.4\left(\frac{2.09 \text{ rad}}{\text{cm}}\right)$$

= 0.836 rad/cm = 48°/cm



FIGURE 9.18 Differential phase shift for the two-slab remanent phase shifter of Examnle 9.4.

The ferrite length required for the 180° phase shift section is then

$$L = \frac{180^{\circ}}{48^{\circ}/\text{cm}} = 3.75 \,\text{cm}.$$

while the length required for a 90° section is

$$L = \frac{90^{\circ}}{48^{\circ}/\text{cm}} = 1.88 \text{ cm.}$$

Other Types of Ferrite Phase Shifters

Many other types of ferrite phase shifters have been developed, with various combinations of rectangular or circular waveguide, transverse or longitudinal biasing, latching or continuous phase variation, and reciprocal or nonreciprocal operation. Phase shifters using printed transmission lines have also been proposed. Even though PIN diode and FET circuits offer a less bulky and more integratable alternative to ferrite components, ferrite phase shifters often have advantages in terms of cost, power handling capacity, and power requirements. But there is still a great need for a low-cost, compact phase shifter.

Several waveguide phase shifter designs are derived from the nonreciprocal Faraday rotation phase shifter shown in Figure 9.19. In operation, a rectangular waveguide TE_{10} mode entering at the left is converted to a TE_{11} circular waveguide mode with a short transition section. Then a quarter-wave dielectric plate, oriented 45^c from the electric field vector, converts the wave to an RHCP wave by providing a 90° phase difference between the field components that are parallel and perpendicular to the plate. In the



FIGURE 9.19 Nonreciprocal Faraday rotation phase shifter.

ferrite-loaded region the phase delay is $\beta_+ z$, which can be controlled with the bias field strength. The second quarter-wave plate converts the wave back to a linearly polarized field. The operation is similar for a wave entering at the right, except now the phase delay is $\beta_- z$; the phase shift is thus nonreciprocal. The ferrite rod is biased longitudinally, in the direction of propagation, with a solenoid coil. This type of phase shifter can be made reciprocal by using nonreciprocal quarter-wave plates to convert a linearly polarized wave to the same sense of circular polarization for either propagation direction.

The Reggia-Spencer phase shifter, shown in Figure 9.20, is a popular reciprocal phase shifter. In either rectangular or circular waveguide form, a longitudinally biased ferrite rod is centered in the guide. When the diameter of the rod is greater than a certain critical size, the fields become tightly bound to the ferrite and are circularly polarized.



FIGURE 9.20 Reggia-Spencer reciprocal phase shifter.



FIGURE 9.21 Symbol for a gyrator, which has a differential phase shift of 180°.



FIGURE 9.22 An isolator constructed with a gyrator and two quadrature hybrids. The forward wave (\rightarrow) is passed, while the reverse wave (\leftarrow) is absorbed in the matched load of the first hybrid.

A large reciprocal phase shift can be obtained over relatively short lengths, although the phase shift is rather frequency sensitive.

The Gyrator

An important canonical nonreciprocal component is the gyrator, which is a two-port device having a 180° differential phase shift. The schematic symbol for a gyrator is shown in Figure 9.21, and the scattering matrix for an ideal gyrator is

$$[S] = \begin{bmatrix} 0 & 1 \\ -1 & 0 \end{bmatrix}, \qquad 9.90$$

which shows that it is lossless, matched, and nonreciprocal. Using the gyrator as a basic nonreciprocal building block in combination with reciprocal dividers and couplers can lead to useful equivalent circuits for nonreciprocal components such as isolators and circulators. Figure 9.22, for example, shows an equivalent circuit for an isolator using a gyrator and two quadrature hybrids.

The gyrator can be implemented as a phase shifter with a 180° differential phase shift; bias can be provided with a permanent magnet, making the gyrator a passive device.

9.6 FERRITE CIRCULATORS

As we discussed in Section 7.1, a *circulator* is a three-port device that can be lossless and matched at all ports; by using the unitary properties of the scattering matrix we were able to show how such a device must be nonreciprocal. The scattering matrix for an ideal circulator thus has the following form:

$$[S] = \begin{bmatrix} 0 & 0 & 1 \\ 1 & 0 & 0 \\ 0 & 1 & 0 \end{bmatrix},$$
 9.91

which shows that power flow can occur from ports 1 to 2, 2 to 3, and 3 to 1, but not in the reverse direction. By transposing the port indices, the opposite circularity can be obtained. In practice, this result can be produced by changing the polarity of the ferrite bias field. Most circulators use permanent magnets for the bias field, but if an electromagnet is used the circulator can operate in a latching (remanent) mode as a single-pole double-throw (SPDT) switch. A circulator can also be used as an isolator by terminating one of the ports with a matched load. A junction circulator is shown in Figure 9.23,

We will first discuss the properties of an imperfectly matched circulator in terms of its scattering matrix. Then we will analyze the operation of the stripline junction, circulator. The operation of waveguide circulators is similar in principle.



FIGURE 9.23 Photograph of a disassembled ferrite junction circulator, showing the stripline conductor, the ferrite disks, and the bias magnet. The middle port of the circulator is terminated with a matched load, so this circulator is actually configured as an isolator. Note the change in the width of the stripline conductors, due to the different dielectric constants of the ferrite and the surrounding plastic material.

Properties of a Mismatched Circulator

If we assume that a circulator has circular symmetry around its three ports and is lossless, but not perfectly matched, its scattering matrix can be written as

$$[S] = \begin{bmatrix} \Gamma & \beta & \alpha \\ \alpha & \Gamma & \beta \\ \beta & \alpha & \Gamma \end{bmatrix}.$$
 9.92

Since the circulator is assumed lossless, [S] must be unitary, which implies the following two conditions:

$$|\Gamma|^2 + |\beta|^2 + |\alpha|^2 = 1, \qquad 9.93a$$

$$\Gamma \beta^* + \alpha \Gamma^* + \beta \alpha^* = 0.$$
 9.93b

If the circulator were matched ($\Gamma = 0$), then (9.93) shows that either $\alpha = 0$ and $|\beta| = 1$, or $\beta = 0$ and $|\alpha| = 1$; this describes the ideal circulator with its two possible circularity states. Observe that this condition depends only on a lossless and matched device.

Now assume small imperfections, such that $|\Gamma| << 1$. To be specific, consider the circularity state where power flows primarily in the 1-2-3 direction, so that $|\alpha|$ is close to unity and $|\beta|$ is small. Then $\exists \Gamma \sim 0$, and (9.93b) shows that $\alpha \Gamma^* + \beta \alpha^* \simeq 0$, so $|\Gamma| \simeq |\beta|$. Then (9.93a) shows that $|\alpha|^2 \simeq 1 - 2|\beta|^2 \simeq 1 - 2|\Gamma|^2$, or $|\alpha| \simeq 1 - |\Gamma|^2$. Then the scattering matrix of (9.92) can be written as

$$[S] = \begin{bmatrix} \Gamma & \Gamma & 1 - \Gamma^{2} \\ 1 - \Gamma^{2} & \Gamma & \Gamma \\ \Gamma & 1 - \Gamma^{2} & \Gamma \end{bmatrix}.$$
 9.94

ignoring phase factors. This result shows that circulator isolation, $\beta \simeq \Gamma$, and transmission, $\alpha \simeq 1 - \Gamma^2$, both deteriorate as the input ports become mismatched.

Junction Circulator

The stripline junction circulator geometry is shown in Figure 9.24, and in the photograph of Figure 9.23. Two ferrite disks fill the spaces between the center metallic disk and the ground planes of the stripline. Three stripline conductors are attached to the periphery of the center disk at 120° intervals, forming the three ports of the circulator. The DC bias field is applied normal to the ground planes.

In operation, the ferrite disks form a resonant cavity; in the absence of a bias field, this cavity has a single lowest-order resonant mode with a $\cos \phi$ (or $\sin \phi$) dependence. When the ferrite is biased this mode breaks into two resonant modes with slightly different resonant frequencies. The operating frequency of the circulator can then be chosen so that the superposition of these two modes add at the output port and cancel at the isolated port.

We can analyze the junction circulator by treating it as a thin cavity resonator with electric walls on the top and bottom, and an approximate magnetic wall on the side. Then $E_{\rho} = E_{\phi} \simeq 0$, and $\partial/\partial z = 0$, so we have TM modes. Since E_z on either side of



FIGURE 9.24 A stripline junction circulator. (a) Pictorial view. (b) Geometry.

the center conducting disk is antisymmetric, we need only consider the solution for one of the ferrite disks [7].

We begin by transforming (9.23), $\tilde{B} = [\mu]\tilde{H}$, from rectangular to cylindrical coordinates:

$$B_{\rho} = B_{x} \cos \phi + B_{y} \sin \phi$$

$$= (\mu H_{x} + j\kappa H_{y}) \cos \phi + (-j\kappa H_{x} + \mu H_{y}) \sin \phi$$

$$= \mu H_{\rho} + j\kappa H_{\phi},$$

$$B_{\phi} = -B_{x} \sin \phi + B_{y} \cos \phi$$

$$= -(\mu H_{x} + j\kappa H_{y}) \sin \phi + (-j\kappa H_{x} + \mu H_{y}) \cos \phi$$

$$= -j\kappa H_{\rho} + \mu H_{\phi}.$$

9.95b

So we have that

$$\begin{bmatrix} B_{\rho} \\ B_{\phi} \\ B_{z} \end{bmatrix} = [\mu] \begin{bmatrix} H_{\rho} \\ H_{\phi} \\ H_{z} \end{bmatrix},$$
 9.96

where $\{\mu\}$ is the same matrix as for rectangular coordinates, as given in (9.24).

In cylindrical coordinates, with $\partial/\partial z = 0$, Maxwell's curl equations reduce to the following:

$$\frac{1}{\rho}\frac{\partial E_z}{\partial \phi} = -j\omega(\mu H_\rho + j\kappa H_\phi), \qquad \qquad 9.97a$$

$$-\frac{\partial E_z}{\partial \rho} = -j\omega(-j\kappa H_\rho + \mu H_\phi), \qquad 9.97b$$

$$\frac{1}{\rho} \left[\frac{\partial(\rho H_{\phi})}{\partial \rho} - \frac{\partial H_{\rho}}{\partial \phi} \right] = j\omega \epsilon E_z, \qquad 9.97c$$

Solving (9.97a,b) for H_{ρ} and H_{ϕ} in terms of E_z gives

$$H_{\rho} = \frac{jY}{k\mu} \left(\frac{\mu}{\rho} \frac{\partial E_z}{\partial \phi} + j\kappa \frac{\partial E_z}{\partial \rho} \right), \qquad 9.98a$$

$$H_{\phi} = \frac{-jY}{k\mu} \left(\frac{-j\kappa}{\rho} \frac{\partial E_z}{\partial \phi} + \mu \frac{\partial E_z}{\partial \rho} \right), \qquad 9.98b$$

where $k^2 = \omega^2 \epsilon (\mu^2 - \kappa^2) / \mu = \omega^2 \epsilon \mu_e$ is an effective wavenumber, and $Y = \sqrt{\epsilon/\mu_e}$ is an effective admittance. Using (9.98) to eliminate H_ρ and H_ϕ in (9.97c) gives a wave equation for E_z :

$$\frac{\partial^2 E_z}{\partial \rho^2} + \frac{1}{\rho} \frac{\partial E_z}{\partial \rho} + \frac{1}{\rho^2} \frac{\partial^2 E_z}{\partial \phi^2} + k^2 E_z = 0.$$
9.99

This equation is identical in form to the equation for E_z for the TM mode of a circular waveguide, so the general solution can be written as

$$E_{zn} = \left[A_{+n} e^{jn\phi} + A_{-n} e^{-jn\phi} \right] J_n(k\rho), \qquad 9.100a$$

where we have excluded the solution with $Y_n(k\rho)$ because E_z must be finite at $\rho = 0$. We will also need $H_{\rho n}$, which can be found using (9.98b):

$$H_{\phi n} = -jY \left\{ A_{+n} e^{jn\phi} \left[J'_n(k\rho) + \frac{n\kappa}{k\rho\mu} J_n(k\rho) \right] + A_{-n} e^{-jn\phi} \left[J'_n(k\rho) - \frac{n\kappa}{k\rho\mu} J_n(k\rho) \right] \right\}.$$
9.100b

The resonant modes can now be found by enforcing the boundary condition that $H_{\phi} = 0$ at $\rho = a$.

If the ferrite is not magnetized, then $H_0 \neq M_s = 0$ and $\omega_0 = \omega_m = 0$ so that $\kappa \neq 0$ and $\mu = \mu_e = \mu_o$, and resonance occurs when

$$J_n'(ka)=0,$$

or $ka = x_0 = p'_{11} = 1.841$. Define this frequency as ω_0 (not to be confused with $\omega_0 = \gamma \mu_0 H_0$):

$$\omega_0 = \frac{x_0}{a\sqrt{\epsilon\mu_e}} = \frac{1.841}{a\sqrt{\epsilon\mu_o}}.$$
 9.101

When the ferrite is magnetized there are two possible resonant modes for each value of n, as associated with either a $e^{jn\phi}$ variation or $e^{-jn\phi}$ variation. The resonance condition for the two n = 1 modes is

$$\frac{\kappa}{\mu x} J_1(x) \pm J_1'(x) = 0, \qquad 9.102$$

where x = ka. This result shows the nonreciprocal property of the circulator, since changing the sign of κ (the polarity of the bias field) in (9.102) leads to the other root and propagation in the opposite direction in φ .

If we let x_+ and x_- be the two roots of (9.102), then the resonant frequencies for these two n = 1 modes can be expressed as

$$\omega_{\pm} = \frac{x_{\pm}}{a\sqrt{\epsilon\mu_c}} \qquad \qquad 9.103$$

We can develop an approximate result for ω_{\pm} if we assume that κ/μ is small, so that ω_{\pm} will be close to ω_0 of (9.101). Using a Taylor series about x_0 for the two terms in (9.102) gives the following results, since $J'_1(x_0) = 0$:

$$J_{1}(x) \simeq J_{1}(x_{0}) + (x - x_{0})J_{1}'(x_{0}) = J_{1}(x_{0}),$$

$$J_{1}'(x) \simeq J_{1}'(x_{0}) + (x - x_{0})J_{1}''(x_{0})$$

$$= -(x - x_{0})\left(1 - \frac{1}{x_{0}^{2}}\right)J_{1}(x_{0}).$$

Then (9.102) becomes

$$\frac{\kappa}{\mu x_0} \neq (x_{\pm} - x_0) \left(1 - \frac{1}{x_0^2} \right) = 0,$$

$$x_{\pm} = x_0 \left(1 \pm 0.418 \frac{\kappa}{\mu} \right),$$

9.104

٥r

since $x_0 = 1.841$. This result gives the resonant frequencies as

$$\omega_{\pm} = \omega_0 \left(1 \pm 0.418 \frac{\kappa}{\mu} \right). \qquad 9.105$$

Note that ω_{\pm} approaches ω_0 as $\kappa \to 0$, and that

$$\omega_{+} \leq \omega_{0} \leq \omega_{+}$$

Now we can use these two modes to design a circulator. The amplitudes of these modes provide two degrees of freedom that can be used to provide coupling from the input to the output port, and to provide cancellation at the isolated port. It will turn out that ω_0 will be the operating frequency, between the resonances of the ω_{\pm} modes. Thus,

 $H_{\phi} \neq 0$ over the periphery of the fertite disks, since $\omega \neq \omega_{\pm}$. If we select port 1 as the input, port 2 as the output, and port 3 as the isolated port, as in Figure 9.24, we can assume the following E_z field at the ports at $\rho = \alpha$:

$$E_{z}(\rho = a, \phi) = \begin{cases} E_{0}, & \text{for } \phi = 0 \quad (\text{Port 1}), \\ -\tilde{E}_{0}, & \text{for } \phi = 120^{\circ} \text{ (Port 2)}, \\ 0, & \text{for } \phi = 240^{\circ} \text{ (Port 3)}. \end{cases}$$

If the feedlines are narrow, the E_z field will be relatively constant across their width. The corresponding H_{ϕ} field should be

$$H_{\varphi}(\rho = a, \phi) = \begin{cases} H_0, & \text{for } -\psi < \phi < \psi, \\ H_0, & \text{for } 120^\circ - \psi < \phi < 120^\circ + \psi, \\ 0, & \text{elsewhere.} \end{cases}$$

Equating (9.106a) to E_z of (9.100a) gives the mode amplitude constants as

$$A_{\pm 1} \approx \frac{E_0(1 \pm j/\sqrt{3})}{2J_1(ka)}.$$
 9.107*a*

$$A_{-1} \neq \frac{E_0(1-j/\sqrt{3})}{2J_1(ka)}.$$
 9.107b

Then (9.100a,b) can be reduced to give the electric and magnetic fields as

$$E_{z1} = \frac{E_0 J_1(k\rho)}{2J_1(ka)} \left[\left(1 + \frac{j}{\sqrt{3}} \right) e^{j\phi} + \left(1 - \frac{j}{\sqrt{3}} \right) e^{-j\phi} \right] \\ = \frac{E_0 J_1(k\rho)}{J_1(ka)} \left(\cos \phi - \frac{\sin \phi}{\sqrt{3}} \right), \qquad 9.108a$$

$$H_{\phi 1} = \frac{-jY E_0}{2J_1(ka)} \left\{ \left(1 + \frac{j}{\sqrt{3}} \right) \left[J_1'(k\rho) + \frac{\kappa}{k\rho\mu} J_1(k\rho) \right] e^{j\phi} \\ + \left(1 - \frac{j}{\sqrt{3}} \right) \left[J_1'(k\rho) - \frac{\kappa}{k\rho\mu} J_1(k\rho) \right] e^{-j\phi} \right\}. \qquad 9.108b$$

To approximately equate $H_{\phi 1}$ to H_{ϕ} in (9.106b) requires that H_{ϕ} be expanded in a Fourier series:

$$H_{\phi}(\rho = a, \phi) = \sum_{n = -\infty}^{\infty} C_n e^{jn\phi} = \frac{2H_0\psi}{\pi} + \frac{H_0}{\pi} \sum_{n=1}^{\infty} \left[(1 + e^{-j2\pi n/3})e^{jn\phi} + (1 + e^{j2\pi n/3})e^{-jn\phi} \right] \times \frac{\sin n\psi}{n}.$$
9.109

The n = 1 term of this result is

$$H_{\phi 1}(\rho=a,\phi)=\frac{-j\sqrt{3}H_0\sin\psi}{2\pi}\left[\left(1+\frac{j}{\sqrt{3}}\right)e^{j\phi}-\left(1-\frac{j}{\sqrt{3}}\right)e^{-j\phi}\right],$$

which can now be equated to (9.108b) for $\rho = a$. Equivalence can be obtained if two conditions are met:

71.19 5

$$J_{1}(\kappa a) = 0,$$
$$\frac{Y E_{0} \kappa}{k a \mu} = \frac{\sqrt{3} H_{0} \sin \psi}{\pi}$$

The first condition is identical to the condition for resonance in the absence of bias, which implies that the operating frequency is ω_0 , as given by (9.101). For a given operating frequency, (9.101) can then be used to find the disk radius, *a*. The second condition can be related to the wave impedance at port 1 or 2:

$$Z_w = \frac{E_0}{H_0} = \frac{\sqrt{3}ka\mu\sin\psi}{\pi Y\kappa} \simeq \frac{\mu\sin\psi}{\kappa Y},$$
9.110

since $\sqrt{3}ka/\pi = \sqrt{3}(1.841)/\pi \simeq 1.0$. Thus, Z_w can be controlled for impedance matching by adjusting κ/μ via the bias field.

We can compute the power flows at the three ports as follows:

$$P_{ia} = P_i = -\hat{\rho} \cdot \bar{E} \times \hat{H}^* = E_z H_\phi \bigg|_{\phi=0} = \frac{E_0 H_0 \sin \psi}{\pi} = \frac{E_0^2 \kappa Y}{\pi \mu}, \qquad 9.111a$$

$$P_{\text{out}} = P_2 = \hat{\rho} \cdot \tilde{E} \times \bar{H}^* = -E_z H_{\phi} \bigg|_{\phi = 120^{\circ}} = \frac{E_0 H_0 \sin \psi}{\pi} = \frac{E_0^2 \kappa Y}{\pi \mu}, \qquad 9.111b$$

$$P_{\rm iso} = P_3 = \hat{\rho} \cdot \bar{E} \times \bar{H}^* = -E_z H_\phi \bigg|_{\phi = 240^\circ} = 0.$$
 9.111c

This shows that power flow occurs from port 1 to 2, but not from 1 to 3. By the azimuthal symmetry of the circulator, this also implies that power can be coupled from port 2 to 3, or from port 3 to 1, but not in the reverse directions.

The electric field of (9.108a) is sketched in Figure 9.25 along the periphery of the circulator, showing that the amplitudes and phases of the $e^{\pm j\phi}$ modes are such that their superposition gives a null at the isolated port, with equal voltages at the input and output ports. This result ignores the loading effect of the input and output lines, which will distort the field from that shown in Figure 9.25. This design is narrowband, but bandwidth can be improved using dielectric loading; the analysis then requires consideration of higher order modes.

and



FIGURE 9.25 Magnitude of the electric field around the periphery of the junction circulator.

REFERENCES

- [1] R. F. Soohoo, Microwave Magnetics, Harper and Row, N. Y., 1985.
- [2] A. J. Baden Fuller, Ferrites at Microwave Frequencies, Peter Peregrinus, London, 1987.
- [3] R. E. Collin, Field Theory of Guided Waves, McGraw-Hill, N. Y., 1960,
- [4] B. Lax and K. J. Button, Microwave Ferrites and Ferrimagnetics, McGraw-Hill, N. Y., 1962.
- [5] F. E. Gardiol and A. S. Vander Vorst, "Computer Analysis of E-plane Resonance Isolators," *IEEE Trans, Microwave Theory and Techniques*, vol. MTT-19, pp. 315–322, March 1971.
- [6] G. P. Rodrigue, "A Generation of Microwave Ferrite Devices." Proc. IEEE, vol. 76, pp. 121–137, February 1988.
- [7] C. E. Fay and R. L. Comstock, "Operation of the Ferrite Junction Circulator," IEEE Trans. Microwave Theory and Techniques, vol. MTT-13, pp. 15-27, January 1965.

PROBLEMS

- **9.1** A certain ferrite material has a saturation magnetization of $4\pi M_s = 1780$ G. Ignoring loss, calculate the elements of the permeability tensor at f = 10 GHz for two cases: (1) no bias field and ferrite demagnetized ($M_s = H_0 = 0$) and (2) a z-directed bias field of 1000 oersted.
- 9.2 Consider the following field transformations from rectangular to circular polarized components:

$$\begin{array}{ll} B^+ = (B_x + jB_y)/2, & H^+ = (H_x + jH_y)/2, \\ B^- = (B_x - jB_y)/2, & H^- = (H_x - jH_y)/2, \\ B_z = B_z, & H_z = H_z. \end{array}$$

For a z-biased ferrite medium, show that the relation between \tilde{B} and \tilde{H} can be expressed in terms of a diagonal tensor permeability as follows:

$$\begin{bmatrix} B^+\\ B^-\\ B_{\pi} \end{bmatrix} = \begin{bmatrix} (\mu+\kappa) & 0 & 0\\ 0 & (\mu-\kappa) & 0\\ 0 & 0 & \mu_o \end{bmatrix} \begin{bmatrix} H^+\\ H^-\\ H_{\pi} \end{bmatrix}$$

- **9.3** A YIG sphere with $4\pi M_s \approx 1780$ G lies in a uniform magnetic field having a strength of 1200 Oe. What is the magnetic field strength inside the YIG sphere?
- **9.4** A thin rod is biased along its axis with an external applied field of $H_a = 1000$ Oe. If $4\pi M_s \approx 600$ G, calculate the gyromagnetic resonance frequency for the rod.
- 9.5 An infinite lossless ferrite medium with a saturation magnetization of $4\pi M_s \approx 1200$ G and a dielectric constant of 10 is biased to a field strength of 500 oersted. At 8 GHz, calculate the differential phase shift per meter between an RHCP and an LHCP plane wave propagating in the direction of bias. If a linearly polarized wave is propagating in this material, what is the distance it must travel in order that its polarization is rotated 90°?
- **9.6** An infinite lossless ferrite medium with a saturation magnetization of $4\pi M_s \approx 1200$ G and a dielectric constant of 10 is biased in the \hat{x} direction with a field strength of 2000 oersted. At 4 GHz, two plane waves propagate in the +z direction, one linearly polarized in x and the other linearly polarized in y. What is the distance these two waves must travel so that the differential phase shift between them is 270°?
- 9.7 Consider a circularly polarized plane wave normally incident on an infinite ferrite medium, as shown in the following figure. Calculate the reflection and transmission coefficients for an RHCP (Γ^+, T^-) and an LHCP (Γ^-, T^-) incident wave. HINT: The transmitted wave will be polarized in the same sense as the incident wave, but the reflected wave will be oppositely polarized.



- 9.8 An infinite lossless ferrite material with $4\pi M_s = 1200 \text{ G}$ is biased in the \hat{x} direction with $\hat{H}_0 = H_0 \hat{x}$. Determine the range of H_0 , in oersteds, where an extraordinary wave (polarized in \hat{x} , propagating in \hat{z}) will be cutoff. The frequency is 4 GHz.
- 9.9 Find the forward and reverse propagation constants for a waveguide half-filled with a transversely biased ferrite. (The geometry of Figure 9.10 with c = 0 and t = a/2.) Assume a = 1.0 cm, f = 10 GHz, $4\pi M_s = 1700$ G, and $\epsilon_r = 13$. Plot versus $H_0 = 0$ to 1500 Oe. Ignore loss and the fact that the ferrite may not be saturated for small H_0 .
- **9.10** Find the forward and reverse propagation constants for a waveguide filled with two pieces of oppositely biased ferrite. (The geometry of Figure 9.11 with c = 0 and t = a/2.) Assume a = 1.0 cm, f = 10 GHz, $4\pi M_s = 1700$ G, and $\epsilon_r = 13$. Plot versus $H_0 = 0$ to 1500 Oe. Ignore loss and the fact that the ferrite may not be saturated for small H_0 .
- 9.11 Consider a wide, thin ferrite slab in a rectangular X-band waveguide, as shown in Figure 9.12b. If f = 10 GHz, $4\pi M_s = 1700 \text{ G}$, c = a/4, and $\Delta S = 2 \text{ mm}^2$, use the perturbation formula of (9.80) to plot the differential phase shift, $(\beta_+ \beta_-)/k_0$, versus the bias field for $H_0 = 0$ to 1200 Oe. Ignore loss.

- 9.12 An *E*-plane resonance isolator with the geometry of Figure 9.12a is to be designed to operate at 8 GHz, with a ferrite having a saturation magnetization of $4\pi M_s = 1500$ G. What is the approximate bias field, H_0 , required for resonance? What is the required bias field if the *H*-plane geometry of Figure 9.12b is used?
- 9.13 Design a resonance isolator using the *H*-plane ferrite slab geometry of Figure 9.12b in an X-band waveguide. The isolator should have minimum forward insertion loss, and a reverse attenuation of 30 dB at 10 GHz. Use a ferrite slab having $\Delta S/S = 0.01$, $4\pi M_s = 1700$ G, and $\Delta H = 200$ Oe.
- 9.14 Calculate and plot the two normalized positions, r/a, where the magnetic fields of the TE₁₀ mode of an empty rectangular waveguide are circularly polarized, for $k_0 = k_c$ to $2k_c$.
- 9.15 The latching ferrite phase shifter shown in the figure below uses the birefringence effect. In state 1, the ferrite is magnetized so that $H_0 = 0$ and $\overline{M} = M_r \hat{x}$. In state 2, the ferrite is magnetized so that $H_0 = 0$ and $\overline{M} = M_r \hat{y}$. If f = 6 GHz, $\epsilon_r = 10$, $4\pi M_r = 1500$ G, and L = 2.78 cm, calculate the differential phase shift between the two states. Assume the incident plane wave is \hat{x} polarized for both states, and ignore reflections.



- **9.16** Rework Example 9.4 with a slab spacing of s = 2 mm, and a remanent magnetization of 1000 G. (Assume all other parameters as unchanged, and that the differential phase shift is linearly proportional to κ .)
- 9.17 Consider a latching phase shifter constructed with a wide, thin *H*-plane ferrite slab in an X-band waveguide, as shown in Figure 9.12b. If f = 9 GHz, $4\pi M_r = 1200$ G, c = a/4, and $\Delta S = 2$ mm², use the perturbation formula of (9.80) to calculate the required length for a differential phase shift of 22.5°.
- **9.18** Design a gyrator using the twin *H*-plane ferrite slab geometry shown below. The frequency is 9.0 GHz, and the saturation magnetization is $4\pi M_s = 1700$ G. The cross-sectional area of each slab is 3.0 mm², and the guide is X-band waveguide. The permanent magnet has a field strength of $H_a = 4000$ Oe. Determine the internal field in the ferrite, H_0 , and use the perturbation formula of (9.80) to determine the optimum location of the slabs and the length. L. to give the necessary 180° differential phase shift.



- 9.19 Draw an equivalent circuit for a circulator using a gyrator and two couplers.
- **9.20** A certain lossless circulator has a return loss of 10 dB. What is the isolation? What is the isolation if the return loss is 20 dB?

Active Microwave Circuits

The components and circuits that we have discussed so far have been linear and passive, but any useful microwave system will require some nonlinear and active components. Such devices include diodes, transistors, and tubes, which can be used for detection, mixing, amplification, frequency multiplication, switching, and as sources. Active circuit design is a broad and rapidly evolving field, so we can only present some of the basic concepts and principles here, and refer the reader to the references for more detail. We will also avoid any discussion of the physics of diodes, transistors, or tubes, since for our purposes it will be adequate to characterize these devices in terms of their terminal properties.

The earliest detector diode was probably the "cat-whisker" crystal detector used in early radio work. The advent of tubes as detectors and amplifiers eliminated this component in most radio systems, but the crystal diode was later used by Southworth in his 1930s experiments with waveguides, since the tube detectors of that era could not operate at such high frequencies. Southworth's co-workers at Bell Labs, including A. P. King and R. S. Ohl, greatly improved the crystal detector with better materials and a rugged cartridge package. Frequency conversion and heterodyning were also first developed for radio applications, in the 1920s. These same techniques were later applied to microwave radar receiver design at the MIT Radiation Laboratory during World War II using crystal diodes as mixers [1], but it was not until the 1960s that microwave solid-state devices saw significant development. PIN diodes were invented, and used as microwave switches and phase shifters. The basic theory of the field effect transistor (FET) was developed by Shockley in 1952, and the first FETs were fabricated on silicon. The first gallium arsenide Schottky barrier FET was made by C. A. Mead at Cal Tech in 1965, and microwave gallium arsenide FETs were developed in the late 1960s [2].

The logical trend for microwave circuits has since been to integrate transmission lines, active devices, and other components on a single semiconductor substrate to form a monolithic microwave integrated circuit (MMIC). The first single-function MMICs were developed in the late 1960s, but more sophisticated circuits such as multistage FET amplifiers, 3- or 4-bit phase shifters, complete transmit/receive radar modules, and other circuits were being fabricated as MMICs [2]. The present trend is toward MMICs with higher performance, lower cost, and greater complexity.

The electrical performance of a microwave system can be affected by many factors, but the effect of noise is probably one of the most fundamental. Thus we begin with a 548

discussion of the sources of noise, and the characterization of components in terms of noise temperature and noise figure. Next we discuss the small-signal characteristics of detector diodes, and their application to the frequency conversion functions of rectification, detection, and mixing. Then we show how PIN diodes can be used for a variety of control circuits, including switches and phase shifters. Finally, we give brief overviews of microwave integrated circuits, and microwave solid-state and tube sources. We will discuss transistor amplifier and oscillator circuit design in Chapter 11.

10.1 NOISE IN MICROWAVE SYSTEMS

Noise power is a result of random processes such as the flow of charges or holes in an electron tube or solid-state device, propagation through the ionosphere or other ionized gas, or, most basic of all, the thermal vibrations in any component at a temperature above absolute zero. Noise can be passed into a microwave system from external sources, or generated within the system itself. In either case the noise level of a system sets the lower limit on the strength of a signal that can be detected in the presence of the noise. Thus, it is generally desired to minimize the residual noise level of a radar or communications receiver, to achieve the best performance. In some cases, such as radiometers or radio astronomy systems, the desired signal is actually the noise power received by an antenna, and it is necessary to distinguish between the received noise power and the undesired noise generated by the receiver system itself.

Dynamic Range and Sources of Noise

In previous chapters we have implicitly assumed that all components were *linear*, meaning that the output is directly proportional to the input, and *deterministic*, meaning that the output is predictable from the input. In reality no component can perform in this way over an unlimited range of input/output signal levels. In practice, however, there is a range of signal levels over which such assumptions are valid; this range is called the dynamic range of the component.

As an example, consider a realistic microwave transistor amplifier having a gain of 10 dB, as shown in Figure 10.1. If the amplifier were ideal, the output power would be related to the input power as

$$P_{\rm out} = 10 P_{\rm in}$$

and this relation would hold true for any value of P_{in} . Thus if $P_{in} = 0$, we would have $P_{out} = 0$, and if $P_{in} = 10^6$ W, we would have $P_{out} = 10^7$ W. Obviously neither of these conditions will be true in practice. Because of noise generated by the amplifier itself, as well as external noise that might be fed into the amplifier. a certain nonzero noise power will be delivered by the amplifier even when the input power is zero. For very high input powers, the amplifier will be destroyed. Thus, the actual relation between the input and output power will be as shown in Figure 10.1. At very low input power levels, the output will be dominated by the noise of the amplifier. This level is often called the noise floor of the component or system; typical values may range from -60 dBm to -100 dBm over the bandwidth of the system, with lower values being obtainable with



FIGURE 10.1 Illustrating the dynamic range of a realistic amplifier.

cooled components. Above the noise floor, the amplifier has a range of input powers for which $P_{out} = 10P_{out}$ is closely approximated. This is the usable dynamic range of the component. At the upper end of the dynamic range, the output begins to saturate, meaning that the output power no longer increases linearly as the input power increases. A quantitative measure of the onset of saturation is given by the 1 dB *compression point*, which is defined as the input power for which the output is 1 dB below that of the ideal amplifier (the corresponding output power level can also be used to specify this point). If the input power is excessive, the amplifier can be destroyed.

Noise is usually generated by the random motions of charges or charge carriers in devices and materials. Such motions can be caused by any of several mechanisms, leading to various sources of noise:

- Thermal noise is the most basic type of noise, being caused by thermal vibration of bound charges. Also known as Johnson or Nyquist noise.
- Shot noise is due to random fluctuations of charge carriers in an electron tube or solid-state device.
- Flicker noise occurs in solid-state components and vacuum tubes. Flicker noise power varies inversely with frequency, and so is often called 1/f-noise.
- *Plasma noise* is caused by random motion of charges in an ionized gas, such as a plasma, the ionosphere, or sparking electrical contacts.

• Quantum noise results from the quantized nature of charge carriers and photons; often insignificant relative to other noise sources.

It is sometimes necessary for measurement purposes to have a calibrated noise source. Passive noise generators consist of a resistor held at a constant temperature, either in a temperature-controlled oven or a cryogenic flask. Active noise sources can be made using gas-discharge tubes or avalanche diodes; such sources generally give much higher noise power than passive sources.

Noise Power and Equivalent Noise Temperature

Consider a resistor at a temperature of T degrees kelvin (K), as depicted in Figure 10.2. The electrons in this resistor are in random motion, with a kinetic energy that is proportional to the temperature, T. These random motions produce small, random voltage fluctuations at the resistor terminals, as illustrated in Figure 10.2. This voltage has a zero average value, but a nonzero rms value given by Planck's black body radiation law,

$$v_n = \sqrt{\frac{4hfBR}{e^{hf/kT} - 1}},$$
 10.1

where

 $h = 6.546 \times 10^{-34}$ J-sec is Planck's constant.

 $k = 1.380 \times 10^{-23}$ J/°K is Boltzmann's constant.

T is the temperature in degrees kelvin (K).

B is the bandwidth of the system in Hz.

f is the center frequency of the bandwidth in Hz.

R is the resistance, in Ω .

This result comes from quantum mechanical considerations, and is valid for any frequency, f. At microwave frequencies the above result can be simplified by making use



FIGURE 10.2 A random voltage generated by a noisy resistor.

of the fact that $hf \ll kT$. (As a worst-case example, let f = 100 GHz and T = 100 K. Then $hf = 6.5 \times 10^{-23} \ll kT = 1.4 \times 10^{-21}$.) Using the first two terms of a Taylor series expansion for the exponential gives

$$e^{hf/kt}-1\simeq \frac{hf}{kT}.$$

so that (10.1) reduces to

$$v_n = \sqrt{4kTBR}.$$
 10.2

This is the Rayleigh-Jeans approximation, and is the form most commonly used in microwave work [3]. For very high frequencies or very low temperatures, however, this approximation may be invalid, in which case (10.1) should be used. Note that this noise power is independent of frequency; such a noise source has a power spectral density that is constant with frequency, and is referred to as a white noise source. The noise power is directly proportional to the bandwidth, which in practice is usually limited by the passband of the microwave system. Since independent white noise sources can be treated as Gaussian distributed random variables, the noise powers (variances) are additive.

The noisy resistor of Figure 10.2 can be replaced with a Thévenin equivalent circuit consisting of a noiseless resistor and a generator with a voltage given by (10.2), as shown in Figure 10.3. Connecting a load resistor R results in maximum power transfer from the noisy resistor, with the result that power delivered to the load in a bandwidth B, is

$$P_n = \left(\frac{v_n}{2R}\right)^2 R = \frac{v_n^2}{4R} = kTB,$$

since v_n is an rms voltage. This important result gives the maximum available noise power from the noisy resistor at temperature T. Observe the following trends:

- As $B \rightarrow 0, P_n \rightarrow 0$. This means that systems with smaller bandwidths collect less noise power.
- As $T \rightarrow 0, P_n \rightarrow 0$. This means that cooler devices and components generate less noise power.
- As B→∞, P_n→∞. This is the so-called ultraviolet catastrophe, which does not occur in reality because (10.2–10.3) are not valid as f (or B) →∞; (10.1) must be used in this case.



FIGURE 10.3 Equivalent circuit of a noisy resistor delivering maximum power to a load resistor through an ideal bandpass filter.

Chapter 10: Active Microwave Circuits

If an arbitrary source of noise (thermal or nonthermal) is "white." so that the noise power is not a strong function of frequency, it can be modeled as an equivalent thermal noise source, and characterized with an equivalent noise temperature. Thus, consider the arbitrary white noise source of Figure 10.4, which has a driving-point impedance of Rand delivers a noise power P_{θ} to a load resistor R. This noise source can be replaced by a noisy resistor of value R, at temperature T_{e} , where T_{e} is an equivalent temperature selected so that the same noise power is delivered to the load. That is,

$$T_{e} = \frac{P_{s}}{kB}.$$
 10.4

Components and systems can then be characterized by saying that they have an equivalent noise temperature of T_e : this implies some fixed bandwidth, B, which is generally the bandwidth of the component or system.

For example, consider a noisy amplifier with a bandwidth B and gain G. Let the amplifier be matched to noiseless source and load resistors, as shown in Figure 10.5. If the source resistor is at a (hypothetical) temperature of $T_s = 0 K$, then the input power to the amplifier will be $P_i = 0$, and the output noise power P_o will be due only to the noise generated by the amplifier itself. We can obtain the same load noise power by driving an ideal noiseless amplifier with a resistor at a temperature,

$$T_e = \frac{P_o}{GkB},$$
 10.5

so that the output power in both cases is $P_o = GkT_eB$. Then T_e is the equivalent noise temperature of the amplifier.

Active noise sources use a diode or tube to provide a calibrated noise power output, and are useful for test and measurement applications. Active noise generators can be characterized by an equivalent noise temperature, but a more common measure of noise power for such components is the *excess noise ratio* (ENR), defined as

ENR(dB) =
$$10 \log \frac{P_N - P_o}{P_o} = 10 \log \frac{T_N - T_0}{T_0}$$
, 10.6



FIGURE 10.4 The equivalent noise temperature, T_e , of an arbitrary white noise source.



FIGURE 10.5 Defining the equivalent noise temperature of a noisy amplifier. (a) Noisy amplifier. (b) Noiseless amplifier.

where P_N and T_N are the noise power and equivalent temperature of the generator, and P_o and T_0 are the noise power and temperature associated with a room-temperature passive source (a matched load). Solid-state noise generators typically have ENRs ranging from 20 to 40 dB.

Measurement of Noise Temperature by the Y-Factor Method

In principle, the equivalent noise temperature of a component can be determined by measuring the output power when a matched load at 0 K is connected at the input of the component. In practice, of course, the 0 K source temperature cannot be achieved, so a different method must be used. If two loads at significantly different temperatures are available, then the Y-factor method can be applied.

This technique is illustrated in Figure 10.6, where the amplifier (or other component) under test is connected to one of two matched loads at different temperatures, and the output power is measured for each case. Let T_1 be the temperature of the hot load, and T_2 the temperature of the cold load ($T_1 > T_2$), and let P_1 and P_2 be the respective powers



FIGURE 10.6 The Y-factor method for measuring the equivalent noise temperature of an amplifier.

measured at the amplifier output. The output power consists of noise power generated by the amplifier as well as noise power from the source resistor. Thus we have

$$P_1 = GkT_1B + GkT_cB, 10.7a$$

$$P_2 = GkT_2B + GkT_eB, 10.7b$$

which are two equations for the two unknowns, T_e and GB (the gain-bandwidth product of the amplifier). Define the Y-factor as

$$Y = \frac{P_1}{P_2} = \frac{T_1 + T_e}{T_2 + T_e} > 1,$$
 10.8

which is determined via the power measurements. Then (10.7) can be solved for the equivalent noise temperature.

$$T_e = \frac{T_1 - YT_2}{Y - 1},$$
 10.9

in terms of the load temperatures and the Y-factor.

Observe that to obtain accurate results from this method, the two source temperatures must not be too close together. If they are, P_1 will be close to P_2 , Y will be close to unity, and the evaluation of (10.9) will involve the subtractions of numbers close to each other, resulting in a loss of accuracy. In practice, one noise source is usually a load resistor at room temperature (T_0) , while the other noise source is either "hotter" or "colder," depending on whether T_{ϵ} is greater or lesser than T_0 . An active noise generator can be used as a "hotter" source, while a "colder" source can be obtained by immersinga load resistor in liquid nitrogen (T = 77 K), or liquid helium (T = 4 K).



EXAMPLE 10.1 Noise Temperature Measurement

An X-band amplifier has a gain of 20 dB and a 1 GHz bandwidth. Its equivalent noise temperature is to be measured via the Y-factor method. The following data is obtained:

for
$$T_1 = 290$$
 K, $P_1 = -62.0$ dBm.
for $T_2 = 77$ K. $P_2 = -64.7$ dBm.

Determine the equivalent noise temperature of the amplifier. If the amplifier is used with a source having an equivalent noise temperature of $T_s = 450$ K, what is the output noise power in dBm?

Solution

From (10.8), the Y-factor in dB is

 $Y = (P_1 - P_2) d\mathbf{B} = (-62.0) - (-64.7) = 2.7 d\mathbf{B},$

which is a numeric value of Y = 1.86. Then using (10.9) gives the equivalent noise temperature as

$$T_e = \frac{T_1 - YT_2}{Y - 1} = \frac{290 - (1.86)(77)}{1.86 - 1} = 170 \text{ K}.$$

If a source with an equivalent noise temperature of $T_s = 450$ K drives the amplifier, the noise power into the amplifier will be kT_sB . The total noise power out of the amplifier will be

$$P_o = GkT_sB + GkT_eB = 100(1.38 \times 10^{-23})(10^9)(450 + 170)$$
$$= 8.56 \times 10^{-10} \text{ W} = -60.7 \text{ dBm}.$$

Noise Figure

We have seen that a noisy microwave component can be characterized by an equivalent noise temperature. An alternative characterization is the *noise figure* of the component, which is a measure of the degradation in the signal-to-noise ratio between the input and output of the component. The signal-to-noise ratio is the ratio of desired signal power to undesired noise power, and so is dependent on the signal power. When noise and a desired signal are applied to the input of a noiseless network, both noise and signal will be attenuated or amplified by the same factor, so that the signal-to-noise ratio will be unchanged. But if the network is noisy, the output noise power will be increased more than the output signal power, so that the output signal-to-noise ratio will be reduced. The noise figure, F, is a measure of this reduction in signal-to-noise ratio, and is defined as

$$F = \frac{S_i/N_i}{S_o/N_o} \ge 1, \qquad 10.10$$

where S_i , N_i are the input signal and noise powers, and S_o , N_o are the output signal and noise powers. By definition, the input noise power is assumed to be the noise power resulting from a matched resistor at $T_0 = 290$ K; that is, $N_i = kT_0B$.

Consider Figure 10.7, which shows noise power N_i and signal power S_i being fed into a noisy two-port network. The network is characterized by a gain G, a bandwidth B, and an equivalent noise temperature, T_e . The input noise power is $N_i = kT_0B$, and the output noise power is a sum of the amplified input noise and the internally generated noise: $N_o = kGB(T_0 + T_e)$. The output signal power is $S_o = GS_i$. Using these results in (10.10) gives the noise figure as

$$F = \frac{S_i}{kT_0B} \frac{kGB(T_0 + T_e)}{GS_i} = 1 + \frac{T_e}{T_0} \ge 1.$$
 10.11



FIGURE 10.7 Determining the noise figure of a noisy network.

In dB, $F = 10 \log(1 + T_e/T_0) dB \ge 0$. If the network were noiseless, T_e would be zero, giving F = 1, or 0 dB. Solving (10.11) for T_e gives

$$T_e = (F - 1)T_0. 10.12$$

It is important to keep in mind two things concerning the definition of noise figure: noise figure is defined for a matched input source, and for a noise source that consists of a resistor at temperature $T_0 = 290$ K. Noise figure and equivalent noise temperatures are interchangeable characterizations of the noise properties of a component.

An important special case occurs in practice when the two-port network is a passive, lossy component, such as an attenuator or lossy transmission line, held at a temperature, T. Consider such a network with a matched source resistor, which is also at temperature T, as shown in Figure 10.8. The gain, G, of a lossy network is less than unity; the loss factor, L, can be defined as L = 1/G > 1. Because the entire system is in thermal equilibrium at the temperature T, and has a driving point impedance of R, the output noise power must be $P_o = kTB$. But we can also think of this power as coming from the source resistor (through the lossy line), and from the noise generated by the line itself. Thus we also have that

$$P_o = kTB = GkTB + GN_{\text{added}}, \qquad 10.13$$

where N_{added} is the noise generated by the line, as if it appeared at the input terminals of the line. Solving (10.13) for this power gives

$$N_{\text{addod}} = \frac{1-G}{G}kTB = (L-1)kTB.$$
 10.14

Then (10.4) shows that the lossy line has an equivalent noise temperature (as referred to the input) given by

$$T_e = \frac{1 - G}{G}T = (L - 1)T.$$
 10.15

Then from (10.11) the noise figure is

$$F = 1 + (L-1)\frac{T}{T_0}.$$
 10.16

If the line is at temperature T_0 , then F = L. For instance, a 6 dB attenuator at room temperature has a noise figure of F = 6 dB.



FIGURE 10.8 Determining the noise figure of a lossy line or attenuator with loss L and temperature T.



EXAMPLE 10.2 Dynamic Range

A 10-12 GHz amplifier has a gain of 20 dB, a noise figure of 3.5 dB, and an output power of 10 dBm at its 1 dB compression point. What is the dynamic range of this amplifier?

Solution

The upper end of the dynamic range is set by the 1 dB compression point, which corresponds to an output power of 10 dBm. The lower end is set by the output noise power, N_{0} , due to the amplifier itself. From (10.12), the equivalent noise temperature at the input of the amplifier is

$$T_c = (F-1)T_0 = (10^{3.5/10} - 1)(290) = 359K.$$

So the output noise power is, from (10.5)

$$N_{e} = GkT_{e}B.$$

In dBm.

$$N_o = 20 + 10 \log \frac{(1.38 \times 10^{-23})(359)(2 \times 10^9) W}{10^{-3} W}$$

= -60.0 dBm.

So the dynamic range is 10 dBm -(-60.0 dBm) = 70.0 dB. Note that noise figure cannot be used directly in this problem because the input noise power is not kT_0B .

Noise Figure of a Cascaded System

In a typical microwave system the input signal travels through a cascade of many different components, each of which may degrade the signal-to-noise ratio to some degree. If we know the noise figure (or noise temperature) of the individual stages, we can determine the noise figure (or noise temperature) of the cascade connection of stages. We will see that the noise performance of the first stage is usually the most critical, an interesting result that is very important in practice.

Consider the cascade of two components, having gains G_1, G_2 , noise figures F_1, F_2 , and noise temperature T_{e1}, T_{e2} , as shown in Figure 10.9. We wish to find the overall noise figure and noise temperature of the cascade, as if it were a single component. The overall gain of the cascade is G_1G_2 .

Using noise temperatures, the noise power at the output of the first stage is

$$N_1 = G_1 k T_0 B + G_1 k T_{e1} B. ag{0.17}$$

since $N_i = kT_0B$ for noise figure calculations. The noise power at the output of the second stage is

$$N_{a} = G_{2}N_{1} + G_{2}kT_{e2}B$$

= $G_{1}G_{2}kB(T_{0} + T_{e1} + \frac{1}{G_{1}}T_{e2}).$ 10.18



FIGURE 10.9 Noise figure and equivalent noise temperature of a cascaded system. (a) Two cascaded networks. (b) Equivalent network.

For the equivalent system we have

$$N_o = G_1 G_2 k B (T_{\rm cts} + T_0), 10.19$$

so comparison with (10.18) gives the noise temperature of the cascade system as

$$T_{\rm cas} = T_{e1} + \frac{1}{G_1} T_{e2}.$$
 10.20

Using (10.12) to convert the temperatures in (10.20) to noise figures yields the noise figure of the cascade system as

$$F_{\rm cas} = F_1 + \frac{1}{G_1}(F_2 - 1).$$
 10.21

Equations (10.20) and (10.21) show that the noise characteristics of a cascaded system are dominated by the characteristics of the first stage, since the effect of the second stage is reduced by the gain of the first. Thus, for the best overall system noise performance, the first stage should have a low noise figure and at least moderate gain. Expense and effort should be devoted primarily to the first stage, as opposed to later stages, since later stages have a diminished impact on the overall noise performance.

Equations (10.20) and (10.21) can be generalized to an arbitrary number of stages, as follows:

$$T_{\rm cas} = T_{\rm e1} + \frac{T_{e2}}{G_1} + \frac{T_{e3}}{G_1 G_2} + \cdots,$$
 10.22

$$F_{\rm cas} = F_1 + \frac{F_2 - 1}{G_1} + \frac{F_3 - 1}{G_1 G_2} + \cdots$$
 10.23



EXAMPLE 10.3 Noise Figure of a Cascade System

An antenna is connected to a low-noise amplifier with a piece of coaxial transmission line. The amplifier has a gain of 15 dB, a bandwidth of 100 MHz, and a noise temperature of 150 K. The coaxial line has an attenuation of 2 dB. Find the noise figure of the transmission line-amplifier cascade. What would be the noise figure if the amplifier were placed at the antenna, eliminating the transmission line? Assume all components are at an ambient temperature of T = 300 K.

Solution

The loss factor of the coaxial line is $L = 10^{2/10} = 1.58$, so from (10.16) the noise figure of the line is

$$F_{\ell} = 1 + (L - 1)\frac{T}{T_0} = 1 + (1.58 - 1)\frac{300}{290} = 1.60 = 2.04 \text{ dB}.$$

From (10.11), the noise figure of the amplifier is

$$F_a = 1 + \frac{T_e}{T_0} = 1 + \frac{150}{290} = 1.52 = 1.81 \,\mathrm{dB}.$$

Then (10.21) gives the noise figure of the cascade as

$$F_{\text{cas}} = F_{\ell} + \frac{1}{G_{\ell}}(F_{q} - 1) = 1.60 + 1.58(1.52 - 1) = 2.42 = 3.84 \text{ dB}.$$

since $1/G_{\ell} = L = 1.58$ for the coaxial line. Without the transmission line, the noise figure would be that of the amplifier itself, or 1.81 dB. So we see that the effect of the lossy feedline reduces the noise figure of the system by about 2 dB—a substantial amount. Sometimes such a line cannot be avoided in the front end of a receiver. Its effect, however, will be deleterious, because not only does the line itself add noise but, since its gain is less than unity, it increases the effect of the noise of the next stage.

10.2 DETECTORS AND MIXERS

Detectors and mixers use a nonlinear device to achieve *frequency conversion* of an input signal [1]. Microwave diodes are most commonly used as the nonlinear element, but transistors can also be used. Figure 10.10 illustrates the three basic frequency conversion functions of rectification, detection, and mixing. We will first discuss the nonlinear voltage-current characteristics of a diode, and then use a small-signal analysis to describe the operation of various circuits that perform these functions.

Diode Rectifiers and Detectors

A diode is basically a nonlinear resistor, with a DC V-I characteristic that can be expressed as

$$I(V) = I_s(e^{\alpha V} - 1), 10.24$$

where $\alpha = q/nkT$, and q is the charge of an electron, k is Boltzmann's constant, T is temperature, n is the idealty factor, and I_s is the saturation current [4], [5]. Typically,









FIGURE 10.10 Basic operations of rectification, detection, and mixing. (a) Diode rectifier. (b) Diode detector. (c) Mixer.

 I_s is between 10^{-6} and 10^{-15} A. and $\alpha = q/nkT$ is approximately 1/(25 mV) for T = 290 K. The idealty factor, *n*, depends on the structure of the diode itself, and can vary from 1.2 for Schottky barrier diodes to about 2.0 for point-contact silicon diodes. Figure 10.11 shows a typical diode V-I characteristic. Now let the diode voltage be

$$V = V_0 + v_1$$
 10.25

where V_0 is a DC bias voltage and v is a small AC signal voltage. Then (10.24) can be expanded in a Taylor series about V_0 as follows:

$$I(V) = I_0 + v \frac{dI}{dV} \bigg|_{V_0} + \frac{1}{2} v^2 \frac{d^2 I}{dV^2} \bigg|_{V_0} + \cdots, \qquad 10.26$$


FIGURE 10.11 V-I characteristics of a diode.

where $I_0 = I(V_0)$ is the DC bias current. The first derivative can be evaluated as

$$\left. \frac{dI}{dV} \right|_{V_0} = \alpha I_s e^{\alpha V_0} = \alpha (I_0 + I_s) = G_d = \frac{1}{R_j}, \qquad 10.27$$

which defines R_j , the junction resistance of the diode, and $G_d = 1/R_j$, which is called the dynamic conductance of the diode. The second derivative is

$$\frac{d^2I}{dV^2}\Big|_{V_0} = \frac{dG_d}{dV}\Big|_{V_0} = \alpha^2 I_s e^{\alpha V_0} = \alpha^2 (I_0 + I_s) = \alpha G_d = G'_d.$$
 10.28

Then (10.26) can be rewritten as the sum of the DC bias current, I_0 , and an AC current, i;

$$I(V) = I_0 + i = I_0 + vG_d + \frac{v^2}{2}G'_d + \cdots$$
 10.29

The three-term approximation for the diode current in (10.29) is called the *small-signal approximation*, and will be adequate for most of our purposes.

The small-signal approximation is based on the DC voltage-current relationship of (10.24), and shows that the equivalent circuit of a diode will involve a nonlinear resistance. In practice, however, the AC characteristics of a diode also involve reactive effects due to the structure and packaging of the diode. A typical equivalent circuit for a diode is shown in Figure 10.12. The leads and contacts of the diode package lead to a series inductance, L_p , and shunt capacitance, C_p . The series resistor, R_s , accounts for



FIGURE 10.12 Equivalent AC circuit model for a diode.

contact and current-spreading resistance. C_j and R_j are the junction capacitance and resistance, and are bias-dependent.



EXAMPLE 10.4 Diode Package Effects

A diode in an axial-lead package has the following equivalent circuit parameters: $C_p = 0.10 \text{ pF}$, $L_p = 2.0 \text{ nH}$, $C_j = 0.15 \text{ pF}$, $R_s = 10 \Omega$, and $I_s = 0.1 \mu \text{A}$. Calculate and plot the impedance of this diode from 4 to 14 GHz, for a bias current $I_0 = 0$ and $I_0 = 60 \mu \text{A}$. Ignore the change in C_j with bias, and assume $\alpha = 1/(25 \text{ mV})$.

Solution

From (10.27) the junction resistance for the two bias states is

for
$$I_0 = 0$$
, $R_j = \frac{1}{\alpha(I_0 + I_s)} = \frac{25 \text{ mV}}{0.1 \,\mu\text{A}} = 2.5 \times 10^5 \,\Omega$,
for $I_0 = 60 \,\mu\text{A}$, $R_j = \frac{1}{\alpha(I_0 + I_s)} = \frac{25 \text{ mV}}{(60 + 0.1) \,\mu\text{A}} = 417 \,\Omega$.

Then the input impedance can be calculated from the equivalent circuit of Figure 10.12; the result is plotted versus frequency on a 50 Ω Smith chart in Figure 10.13.





In a rectifier application, a diode is used to convert a fraction of an RF input signal to DC power. Rectification is a very common function, and is used for power monitors, automatic gain control circuits, and signal strength indicators. If the diode voltage consists of a DC bias voltage and a small-signal RF voltage,

$$V = V_0 + v_0 \cos \omega_0 t, 10.30$$

then (10.29) shows that the diode current will be

$$I = I_0 + v_0 G_d \cos \omega_0 t + \frac{v_0^2}{2} G'_d \cos^2 \omega_0 t$$

= $I_0 + \frac{v_0^2}{4} G'_d + v_0 G_d \cos \omega_0 t + \frac{v_0^2}{4} G'_d \cos 2\omega_0 t.$ 10.31

 I_0 is the bias current and $v_0^2 G'_d/4$ is the DC rectified current. The output also contains AC signals of frequency ω_0 , and $2\omega_0$ (and higher-order harmonics), which are usually filtered out with a simple low-pass filter. A current sensitivity, β_i , can be defined as a measure of the change in DC output current for a given input RF power. From (10.29) the RF input power is $v_0^2 G_d/2$ (using only the first term), while (10.31) shows the change in DC current is $v_0^2 G'_d/4$. The current sensitivity is then

$$\beta_i = \frac{\Delta I_{\rm dc}}{P_{\rm in}} = \frac{G'_d}{2G_d} \,\,\mathrm{A/W}.$$
 10.32

An open-circuit voltage sensitivity, β_n , can be defined in terms of the voltage drop across the junction resistance when the diode is open-circuited. Thus,

$$\beta_v = \beta_i R_r \tag{10.33}$$

Typical values for the voltage sensitivity of a diode range from 400 to 1500 mV/mW.

In a detector application the nonlinearity of a diode is used to demodulate an amplitude modulated RF carrier. For this case, the diode voltage can be expressed as

$$v(t) = v_0(1 + m \cos \omega_m t) \cos \omega_0 t. \qquad 10.34$$

where ω_m is the modulation frequency, ω_0 is the RF carrier frequency ($\omega_0 >> \omega_m$), and m is defined as the modulation index ($0 \le m \le 1$). Using (10.34) in (10.29) gives the diode current:

$$i(t) = v_0 G_d (1 + m \cos \omega_m t) \cos \omega_0 t + \frac{v_0^2}{2} G'_d (1 + m \cos \omega_m t)^2 \cos^2 \omega_0 t$$

$$= v_0 G_d \left[\cos \omega_0 t + \frac{m}{2} \cos(\omega_0 + \omega_m) t + \frac{m}{2} \cos(\omega_0 - \omega_m) t \right]$$

$$+ \frac{v_0^2}{4} G'_d \left[1 + \frac{m^2}{2} + 2m \cos \omega_m t + \frac{m^2}{2} \cos 2\omega_m t + \cos 2\omega_0 t + m \cos(2\omega_0 + \omega_m) t + m \cos(2\omega_0 - \omega_m) t + \frac{m^2}{2} \cos 2\omega_0 t + \frac{m^2}{4} \cos 2(\omega_0 + \omega_m) t + \frac{m^2}{4} \cos 2(\omega_0 - \omega_m) t \right].$$
10.35



FIGURE 10.14 Output spectrum of a detected AM modulated signal.

The frequency spectrum of this output is shown in Figure 10.14. The output current terms which are linear in the diode voltage (terms multiplying v_0G_d) have frequencies of ω_0 and $\omega_0 \pm \omega_m$, while the terms that are proportional to the square of the diode voltage (terms multiplying $v_0^2G'_d/2$) include the frequencies and relative amplitudes listed in Table 10.1.

The desired demodulated output of frequency ω_m is easily separated from the undesired components with a low-pass filter. Observe that the amplitude of this current is $mv_0^2G'_d/2$, which is proportional to the power of the input signal. This square-law behavior is the usual operating condition for detector diodes, but can be obtained only over a restricted range of input powers. If the input power is too large, small-signal conditions will not apply, and the output will become saturated and approach a linear, and then a constant, *i* versus *P* characteristic. At very low signal levels the input signal will be lost in the noise floor of the device. Figure 10,15 shows the typical v_{out} versus P_{in} characteristic, where the output voltage can be considered as the voltage drop across a resistor in series with the diode. Square-law operation is particularly important for applications where power levels are inferred from detector voltage, as in SWR indicators and signal level indicators. Detectors may be DC biased to an operating point that provides the best sensitivity.

TABLE 10.1	Frequencies and	Relative	Amplitudes	of the	Square-L	.aw	Output	of a
Detected AM	Signal							

Frequency	Relative Amplitude	
0	$\frac{1}{1+m^2/2}$	
\$4 7 476	Źm.	
200	$m^2/2$	
$2\omega_0$	$1 + m^2/2$	
$2\omega_0\pm\omega_m$	m	
$2(\omega_0\pm\omega_m)$	$m^{2}/4$	



FIGURE 10.15 Square-law region for a typical diode detector.

Single-Ended Mixer

A mixer uses the nonlinearity of a diode to generate an output spectrum consisting of the sum and difference frequencies of two input signals. In a receiver application, a low-level RF signal and an RF local oscillator (LO) signal are mixed together to produce an intermediate frequency (IF). $f_{\rm IF} = f_{\rm RF} - f_{\rm LO}$, and a much higher frequency, $f_{\rm RF} + f_{\rm LO}$, which is filtered out. See Figure 10.16a. The IF signal usually has a frequency between 10 and 100 MHz, and can be amplified with a low-noise amplifier. This is called a heterodyne receiver, and is useful because it has much better sensitivity and noise characteristics (using an IF amplifier minimizes 1/f noise) than the direct detection scheme discussed in the previous section. A heterodyne system also has the advantage of being able to tune over a band by simply changing the LO frequency, without the need for a high-gain, wideband RF amplifier.

As shown in Figure 10.16b, a mixer can also be used in a transmitter to offset the frequency of an RF signal by an amount equal to $f_{\rm IF}$. This is a convenient technique, as it allows the use of identical local oscillators in the transmitter and receiver; a single oscillator may serve this purpose in a radar or transceiver system.

There are several types of mixer circuits, but the simplest is the single-ended mixer, single-ended mixers often are used as part of more sophisticated mixers. A typical single-ended mixer circuit is shown in Figure 10.17, where an RF signal,

$$v_{\rm RF}(t) = v_{\rm r} \, \cos \, \omega_{\rm r} t, \qquad 10.36$$

is combined with an LO signal,

$$v_{\rm LO}(t) = v_0 \cos \omega_0 t. \qquad 10.37$$

and fed into a diode. The combiner may be a simple T-junction combiner, or a directional coupler. An RF matching circuit may precede the diode, and the diode may be biased through chokes that allow DC to pass while blocking RF. From (10.29), the diode current



FIGURE 10.16 Frequency conversion in a receiver and transmitter. (a) Down-conversion in a heterodyne receiver. (b) Up-conversion in a transmitter.

will consist of a constant DC bias term, and RF and LO signals of frequencies ω_r and ω_0 , due to the term which is linear in v. The v^2 term will give rise to the following output current:

$$i = \frac{G'_d}{2} (v_r \cos \omega_r t + v_0 \cos \omega_0 t)^2$$

= $\frac{G'_d}{2} (v_r^2 \cos^2 \omega_r t + 2v_r v_0 \cos \omega_r t \cos \omega_0 t + v_0^2 \cos^2 \omega_0 t)$



FIGURE 10.17 Single-ended mixer circuit.

$$= \frac{G'_{d}}{4} [v_{r}^{2} + v_{0}^{2} + v_{r}^{2} \cos 2\omega_{r}t + v_{0}^{2} \cos 2\omega_{0}t + 2v_{r}v_{0}\cos(\omega_{r} - \omega_{0})t + 2v_{r}v_{0}\cos(\omega_{r} + \omega_{0})t].$$
10.38

The DC terms can be ignored, and the $2\omega_r$ and $2\omega_0$ terms will be filtered out. The most important terms are those of frequency $\omega_r \pm \omega_0$.

For a receiver or down-converter, the $\omega_r - \omega_0$ term will become the IF signal. Note that, for a given local oscillator frequency, there will be two RF frequencies that will mix down to the same IF frequency. If the RF frequency is $\omega_r = \omega_0 + \omega_i$, then the output frequencies of the mixer will be $\omega_r \pm \omega_0 = 2\omega_0 + \omega_i$, and ω_i ; if the RF frequency is $\omega_r = \omega_0 - \omega_i$, the mixer output frequencies will be $\omega_r \pm \omega_0 = 2\omega_0 - \omega_i$, and $-\omega_i$. This latter output is called the *image response* of the mixer, and is indistinguishable from the direct response. It can be eliminated by RF filtering at the input of the mixer, but this is difficult because the desired RF frequency ($\omega_0 + \omega_i$) is relatively close to the spurious image frequency at ($\omega_0 - \omega_i$), since generally $\omega_i << \omega_0$. Another way to eliminate the image response is by using an image rejection mixer,

In an up-converter, or modulation, application the two inputs will usually be a local oscillator and an IF oscillator, as in Figure 10.16b. The IF signal would be modulated with the desired information signal. Then the output will be $\omega_0 \pm \omega_i$, where ω_i is the IF frequency. The frequency $\omega_0 \pm \omega_i$ is called the *upper sideband* (USB), while $\omega_0 - \omega_i$ is called the *lower sideband* (LSB). Double sideband (DSB) modulation retains both sidebands, while single sideband (SSB) modulation removes one of the sidebands by filtering or by using an image rejection mixer (also called a single sideband modulator).

Mixer design involves impedance matching the three ports, which is complicated by the fact that several frequencies and their harmonics are involved. Undesired harmonic power can be dissipated in resistive terminations, or blocked with reactive terminations. Resistive loads increase the loss of the mixer, and reactive loads are usually very frequency sensitive. An important figure of merit for a mixer is the *conversion loss*, defined as

$$L_C = 10 \log \frac{\text{available RF input power}}{\text{IF output power}} \text{ dB.}$$
 10.39

Practical mixers usually have a conversion loss between 4 and 7 dB. One factor that strongly affects the conversion loss of a mixer is the local oscillator signal (or *pump*) power level: minimum conversion loss usually occurs for LO powers between 0 and 10 dBm. This power level is large enough to violate the small-signal approximation of (10.29), so results using such a model may not be very accurate. Precise design requires numerical solution of the nonlinear equation that describes the diode characteristics [4].

Because a mixer is often the first or second component in a receiver system, its noise characteristics can be of critical importance. When specifying the noise figure of a mixer (or a receiver that uses a mixer), a distinction must be made as to whether the input is a single sideband signal or a double sideband signal. This is because the mixer will produce an IF output for two RF frequencies ($\omega_0 \pm \omega_i$), and therefore collect noise power at both frequencies. When used with a DSB input, the mixer will have desired signals at both RF frequencies, while an SSB input provides the desired signal only at one of these frequencies. Thus the DSB noise figure will be 3 dB lower than the SSB noise figure.

Besides conversion loss, there are several other characteristics that describe mixer performance. Impedance matching at the RF and LO inputs is important for good signal sensitivity and noise figure. In many applications it is desirable to have good isolation between the RF and LO ports so that, for example, LO power will not be radiated out the receive antenna. Other factors include the cancellation of AM noise from the LO, and supression of higher-order harmonics. The single-ended mixer performs reasonably well in terms of all these characteristics, but the mixer designs discussed below can be used to obtain substantially better performance for some specific characteristics.

Balanced Mixer

A balanced mixer combines two or more identical single-ended mixers with a 3 dB hybrid junction (90° or 180°) to give either better input SWR or better RF/LO isolation. The balanced mixer can also give cancellation of AM noise from the local oscillator. Figure 10.18 shows a photograph of a microstrip circuit that contains several balanced mixers.

The circuit for a balanced mixer is shown in Figure 10.19; it consists of two singleended mixers with matched characteristics, driven with a 3 dB coupler. Although not shown, the single-ended mixers will require matching and bias networks. We first consider the case where a small random noise voltage, $v_n(t)$, is superimposed on the local oscillator signal. Then the RF and LO voltages at the input of the hybrid can be expressed as

$$v_{\rm RF}(t) = v_r \cos \omega_r t, \qquad 10.40$$

$$v_{\text{LO}}(t) = [v_0 + v_n(t)] \cos \omega_0 t,$$
 10.41

where $v_r \ll v_0$, and $v_n(t) \ll v_0$. If we have a 90° hybrid, the voltages across the two diodes are

$$v_{1}(t) = v_{\tau} \cos(\omega_{\tau}t - 90^{\circ}) + (v_{0} + v_{n}) \cos(\omega_{0}t - 180^{\circ})$$

= $v_{\tau} \sin\omega_{\tau}t - (v_{0} + v_{n}) \cos\omega_{0}t$, 10.42a
 $v_{2}(t) = v_{\tau} \cos(\omega_{\tau}t - 180^{\circ}) + (v_{0} + v_{n}) \cos(\omega_{0}t - 90^{\circ})$
= $-v_{\tau} \cos\omega_{\tau}t + (v_{0} + v_{n}) \sin\omega_{0}t$. 10.42b

The quadratic term of the diode V-I characteristic will give rise to the desired mixer products, so we will consider only this term and assume identical diodes so that diode currents can be represented as

$$i_1 = k v_1^2,$$
 10.43a

$$i_2 = -kv_2^2,$$
 10.43b

where the negative sign in (10.43b) accounts for the reversed polarity of the diodes. Using (10.42) in (10.43) gives the diode currents as

$$egin{aligned} &i_1 = k [v_r^2 \sin^2 \omega_r t + (v_0 + v_n)^2 \, \cos^2 \omega_0 t - 2 v_r (v_0 + v_n) \sin \omega_r t \, \cos \, \omega_0 t], \ &i_2 = -k [v_r^2 \, \cos^2 \omega_r t + (v_0 + v_n)^2 \sin^2 \omega_0 t - 2 v_r (v_0 + v_n) \, \cos \, \omega_r t \sin \omega_0 t]. \end{aligned}$$



FIGURE 10.18

Photograph of a 35 GHz microstrip monopulse ratar receiver circuit. Three balanced mixers using ring hybrids are shown, along with three stepped-impedance low-pass filters, and six quadrature hybrids. Eight feedlines are aperture coupled to microstrip antennas on the reverse side. The circuit also contains a Gunn diode source for the local oscillator.

Courtesy of Millitech Corporation, S. Deerfield, Mass,





After low-pass filtering, the remaining terms will be DC, noise, and IF frequency terms:

$$i_1 = k \left[\frac{1}{2} v_r^2 + \frac{1}{2} (v_0 + v_n)^2 - v_r (v_0 + v_n) \sin \omega_i t \right]. \qquad 10.44a$$

$$i_2 = -k \left[\frac{1}{2} v_r^2 + \frac{1}{2} (v_0 + v_n)^2 + v_r (v_0 + v_n) \sin \omega_i t \right], \qquad 10.44b$$

where $\omega_i = \omega_r - \omega_0$ is the IF frequency. Combining these currents gives the IF output as

$$i_{\mathrm{IP}} = i_1 + i_2 = -2kv_r(v_0 + v_n)\sin\omega_i t \simeq -2kv_rv_0\sin\omega_i t, \qquad 10.45$$

since $v_n \ll v_0$. This result shows that the first-order terms in the noise voltage are canceled by the mixer, while the desired IF signals combine in phase. Practical mixers can give from 15 to 30 dB of AM noise rejection.

Now consider reflection of the input RF and LO signals from the diodes. If we have a balanced mixer with a 90° hybrid, the input RF signal will give rise to the following reflected waves (phasors) from the diodes:

$$V_{\Gamma 1} = \Gamma V_1 = \frac{\Gamma V_r}{\sqrt{2}}, \qquad 10.46a$$

$$V_{\Gamma 2} = \Gamma V_2 = -j \frac{\Gamma V_r}{\sqrt{2}}, \qquad 10.46b$$

where Γ is the reflection coefficient of each diode, and V_r is the phasor RF input voltage. These two reflections will then arrive and combine back at the RF and LO input ports with the following amplitudes:

$$V_{\Gamma}^{RF} = \frac{V_{\Gamma 1}}{\sqrt{2}} - j \frac{V_{\Gamma 2}}{\sqrt{2}} = \frac{1}{2} \Gamma V_r - \frac{1}{2} \Gamma V_r = 0, \qquad 10.47a$$

$$V_{\Gamma}^{LO} \approx \frac{V_{\Gamma 2}}{\sqrt{2}} - j \frac{V_{\Gamma 1}}{\sqrt{2}} = -\frac{1}{2} j \Gamma V_{r} - \frac{1}{2} j \Gamma V_{r} = -j \Gamma V_{r}.$$
 10.476

Thus the RF input is matched, but the reflected wave appears at the LO port. Similarly, when the LO port is driven, the reflected wave will appear at the RF port. So the RF and LO inputs of a mixer using a 90° hybrid will have good SWR characteristics, but the isolation between the RF and LO ports will be poor.

Alternatively, if a 180° hybrid is used with the RF applied to the sum port and the LO applied to the difference port, the RF waves reflected from the diodes will be

$$V_{\Gamma 1} = V_{\Gamma 2} = \frac{\Gamma V_r}{\sqrt{2}}.$$
 10.48

Then the reflections back at the sum and difference ports will be

$$V_{\Gamma}^{\Sigma} = \frac{V_{\Gamma 1}}{\sqrt{2}} + \frac{V_{\Gamma 2}}{\sqrt{2}} = \Gamma V_{T}$$
 (0.49a)

$$V_{\Gamma}^{\Delta} = \frac{V_{\Gamma 1}}{\sqrt{2}} - \frac{V_{\Gamma 2}}{\sqrt{2}} = 0.$$
 10.49b

The LO waves reflected from the diodes will be

$$V_{\Gamma_1} = -V_{\Gamma^2} = \frac{\Gamma V_r}{\sqrt{2}},$$
 10.50

and the reflections back at the sum and difference ports will be

$$V_{\Gamma}^{\Sigma} = \frac{V_{\Gamma 1}}{\sqrt{2}} + \frac{V_{\Gamma 2}}{\sqrt{2}} = 0, \qquad 10.51a$$

$$V_{\Gamma}^{\Delta} = \frac{V_{\Gamma 1}}{\sqrt{2}} - \frac{V_{\Gamma 2}}{\sqrt{2}} = \Gamma V_r. \qquad 10.51b$$

In both cases, the mismatch appears at the corresponding input port, while the RF and LO ports are isolated.

Other Types of Mixers

Like the balanced mixers described above, there are several other mixer circuits that can be used to enhance or reduce various modulation products and harmonics [4]. Some of these are briefly described below.

Antiparallel diode mixer. A circuit that is often used for subharmonically pumped mixers for millimeter wave applications uses a back-to-back pair of diodes, as shown in Figure 10.20. In operation, the local oscillator frequency is one-half of the usual LO frequency ($\omega_{\tau} - \omega_i$), and the diode nonlinearity generates a second harmonic of the LO frequency to mix with ω_{τ} and produce the desired output frequency. Actually, most mixers can be used in this manner, but the antiparallel diode pair creates a symmetrical V-I characteristic that suppresses the fundamental mixing product of the RF and LO signals and leads to a better conversion loss. This configuration also suppresses AM noise from the local oscillator.

Double-balanced mixer. The single-ended mixer has an output consisting of all harmonic combinations of the RF and LO signals. The balanced mixer using a 180° hybrid suppresses all even harmonics of the LO. The double-balanced mixer, shown in Figure 10.21, can suppress even harmonics of both the LO and RF signals. This



FIGURE 10.20 Subharmonically pumped mixer using an antiparallel diode pair.





leads to a very low conversion loss. It uses two 180° hybrids, so it has good RF/LO isolation, but poor input SWR. It uses four diodes in a ring configuration, although a "star" arrangement can also be used.

Image rejection mixer. We have already noted that two distinct RF signals, $\omega_r = \omega_0 \pm \omega_i$, can produce the same IF frequency, ω_i , when mixed with a local oscillator of frequency ω_0 . These two signals can be thought of as the upper and lower sidebands of ω_0 modulated by ω_i , and are usually referred to as the real (desired) and image (undesired) mixer responses. The real response can be arbitrarily selected as either the USB or the LSB. The image rejection mixer of Figure 10.22 can be used to isolate these two responses into separate LSB ($\omega_r = \omega_0 - \omega_i$) and USB ($\omega_r = \omega_0 + \omega_i$) signals. When used as an up-converter or modulator, this mixer can produce a single-sideband output signal.

The operation of the image rejection mixer can be explained as follows. Let the input RF signal consist of both upper and lower sidebands:

$$v_{\tau} = v_U \cos(\omega_0 + \omega_i)t + v_L \cos(\omega_0 - \omega_i)t. \qquad 10.52$$



FIGURE 10.22 Image rejection mixer circuit.

Then the input to the two mixers is

$$v_r^{\mathcal{A}} = \frac{v_{\mathcal{U}}}{\sqrt{2}} \cos\left(\omega_0 + \omega_i\right)t + \frac{v_{\mathcal{L}}}{\sqrt{2}} \cos\left(\omega_0 - \omega_i\right)t, \qquad 10.53a$$

$$v_r^B = \frac{v_U}{\sqrt{2}} \cos\left[(\omega_0 + \omega_i)t - 90^\circ\right] + \frac{v_L}{\sqrt{2}} \cos\left[(\omega_0 - \omega_i)t - 90^\circ\right]. \quad 10.53b$$

After mixing with an LO signal of $\cos \omega_0 t$, the IF outputs of the mixers are

$$v_i^A = \frac{kv_U}{2\sqrt{2}}\cos\omega_i t + \frac{kv_L}{2\sqrt{2}}\cos\omega_i t, \qquad 10.54a$$

$$v_i^{\mathcal{B}} = \frac{k v_U}{2\sqrt{2}} \cos(\omega_i t - 90^\circ) + \frac{k v_L}{2\sqrt{2}} \cos(\omega_i t + 90^\circ).$$
 10.54b

Combining these two signals in the 90° hybrid at the IF output gives the top output signal as

$$v_{1} = \frac{k}{4} [v_{U} \cos \omega_{i} t + v_{L} \cos \omega_{i} t + v_{U} \cos(\omega_{i} t + 180^{\circ}) + v_{L} \cos \omega_{i} t]$$
$$= \frac{k v_{L}}{2} \cos \omega_{i} t, \qquad 10.55a$$

which is the LSB component. The bottom IF output is

$$v_{2} = \frac{k}{4} [v_{U} \cos(\omega_{i}t - 90^{\circ}) + v_{L} \cos(\omega_{i}t + 90^{\circ}) + v_{U} \cos(\omega_{i}t - 90^{\circ}) + v_{L} \cos(\omega_{i}t - 90^{\circ})]$$
$$= \frac{kv_{U}}{2} \sin \omega_{i}t, \qquad 10.55b$$

which is the USB component. Image rejection or isolation ratios of 20 dB or more are typical. Table 10.2 summarizes some of the basic characteristics of the various mixers we have discussed.

Mixer Type	Number of Diodes	RF SWR	RF/LO Isolation	L_C	Third-Order Intercept
Single ended	1	Poor	Fair	Good	13 dBm
Balanced (90°)	2	Good	Poor	Good	13 dBm
Balanced (180°)	2	Fair	Excellent	Good	13 dBm
Double balanced	4	Poor	Excellent	Excellent	18 dBm
Image rejection	8	Good	Good	Good	15 dBm

TABLE 10.2 Basic Characteristics of Some Mixers

Intermodulation Products

Frequency conversion in a detector or mixer is made possible through the use of a nonlinear device, such as a diode. We have seen, however, that this nonlinearity also gives rise to a number of undesired harmonics and mixer products. These spurious signals increase the conversion loss of a mixer, and can also lead to signal distortion. A similar effect can occur in amplifiers, since the active devices used for amplification (transistors, diodes, or tubes) are nonlinear. In general, a system using a nonlinear device has a voltage transfer function that can be written as a Taylor series:

$$v_{\text{out}} = a_0 + a_1 v_{\text{in}} + a_2 v_{\text{in}}^2 + a_3 v_{\text{in}}^3 + \cdots$$
 10.56

For a detector or mixer, the a_0 term corresponds to the DC bias voltage, while the desired detected or mixed output is part of the v_{in}^2 term. For an amplifier, the linear v_{in} term provides the desired response. The operation of a subharmonically pumped mixer depends on the v_{in}^3 term. Thus, depending on the application, one of these terms provides the desired output, while the remaining terms produce undesired spurious signals.

If the input to the system consists of a single frequency (or tone), say $v_{in} = \cos \omega_1 t$, then the output voltage given by (10.56) will consist of all harmonics, $m\omega_1$, of the input signal. These harmonics are classified by their order, which is equal to m. Thus, for an amplifier, the first-order harmonic (fundamental) is the desired response, and the presence of higher-order harmonics is called harmonic distortion. If an amplifier had a bandwidth of an octave or more, the second-order distortion product of a low-frequency signal could be in the passband of the amplifier. In a mixer application, single-tone distortion products are generally eliminated by filtering.

More serious problems arise when the input to the system consists of two relatively closely spaced frequencies (two-tone), say $v_{in} = \cos \omega_1 t + \cos \omega_2 t$. Then the output spectrum will consist of all harmonics of the form $m\omega_1 + n\omega_2$, where m and n may be positive or negative integers; the order of a given product is then defined as |m| + |n|. The v_{in}^2 term of (10.56) will produce harmonics at the frequencies $2\omega_{11}/2\omega_2$. $\omega_1 - \omega_2$, and $\omega_1 + \omega_2$, which are all second-order products. These frequencies are generally far away from the fundamentals ω_1 and ω_2 , and so can easily be filtered. Such filtering may be impossible for a broadband amplifier, or receiver system, however. The $\omega_1 - \omega_2$ product is usually the desired result for a mixer. The v_{in}^3 term will lead to third-order products such as $3\omega_1$, $3\omega_2$, $2\omega_1 + \omega_2$, and $2\omega_2 + \omega_1$, which can be filtered, and to the products $2\omega_1 - \omega_2$ and $2\omega_2 - \omega_1$, which generally cannot be filtered, even in a narrow-band system. Such products that arise from mixing two input signals are called intermodulation distortion; the third-order two-tone intermodulation products $2\omega_1 - \omega_2$ and $2\omega_2 - \omega_1$ are especially important because they may set the dynamic range or bandwidth of the system. Higherorder terms in (10.56) may also contribute such harmonics, but usually the dominant contributions come from the lowest-order terms.

A measure of the second- or third-order intermodulation distortion is given by the intercept points, which are points on the graph of output power versus input power for the nonlinear component or system under consideration. Such an intercept diagram is shown in Figure 10.23. A plot of output signal ($\omega_1 \pm \omega_2$ for mixer, or ω_1 and ω_2 for an amplifier) power versus input power has a slope of unity for small signal levels. As the input power increases, saturation sets in, causing clipping of the output waveform



FIGURE 10.23 Intercept diagram for a nonlinear component.

and signal distortion. This distortion manifests itself by diverting part of the input power to various harmonics. Equation (10.56) shows that the power in a second-order product varies as v_{in}^2 , so the curve of output power for this product will have a slope of two. If the linear part of the small signal gain curve is extended, it will intercept the second-order product power curve at the second-order intercept point. This point can be specified by either the input or the output power at the intersection, and is a measure of the amount of second-order intermodulation distortion. The component would actually be operated well below this point. A similar intercept point is defined for third-order intermodulation distortion. Because this product is due to the v_{in}^3 term, its curve has a slope of three. Besides the fact that this type of distortion is more difficult to filter than other distortions, its intercept point usually occurs at a lower power level than the second-order intercept point. If two components having individual intercept output powers of I_1 and I_2 , and gains G_1 and G_2 , are connected in cascade, the total intercept output power, I_T , can be shown to be given as

$$I_T = \left(\frac{1}{G_2 I_1} + \frac{1}{I_2}\right)^{-1}.$$
 10.57



POINT OF INTEREST: The Spectrum Analyzer

A spectrum analyzer gives a frequency-domain representation of an input signal, displaying the average power density versus frequency. Thus, its function is dual to that of the oscilloscope, which displays a time-domain representation of an input signal. A spectrum analyzer is basically a sensitive receiver that tunes over a specified frequency band and gives a video output that is proportional to the signal power in a narrow bandwidth. Spectrum analyzers are invaluable for measuring modulation products, harmonic and intermodulation distortion, noise and interference effects.

The diagram below shows a simplified block diagram of a spectrum analyzer. A microwave spectrum analyzer can typically cover any frequency band from several hundred megahertz to tens of gigahertz. The frequency resolution is set by the IF bandwidth, and is adjustable from about 100 Hz to 1 MHz. A sweep generator is used to repetitively scan the receiver over the desired frequency band by adjusting the local oscillator frequency, and to provide horizontal deflection of the display. An important part of the modern spectrum analyzer is the YIG-tuned bandpass filter at the input to the mixer. This filter is tuned along with the local oscillator, and acts as a preselector to reduce spurious intermodulation products. An IF amplifier with a logarithmic response is generally used to accommodate a wide dynamic range. Of course, like many modern test instruments, state-of-the-art spectrum analyzers often contain microprocessors to control the system and the measurement process. This improves performance and makes the analyzer more versatile, but can be a disadvantage in that the computer tends to remove the user from the physical reality of the measurement.



10.3 PIN DIODE CONTROL CIRCUITS

Switches are used extensively in microwave systems, for directing signal or power flow between other components. Switches can also be used to construct other types of control circuits, such as phase shifters and attenuators. Mechanical switches can be made in waveguide or coaxial form, and can handle high powers, but are bulky and slow. PIN diodes, however, can be used to construct an electronic switching element easily integrated with planar circuitry and capable of high-speed operation. (Switching speeds of 10 nanoseconds or less are typical.) FETs can also be used as switching elements.

The PIN diode has V-I characteristics that make it a good RF switching element. When reverse biased, a small series junction capacitance leads to a relatively high diode impedance, while a forward bias current removes the junction capacitance and leaves the diode in a low impedance state [5]. Equivalent circuits for these two states are shown in Figure 10.24. Typical values for the parameters are: $C_j = 1$ pF, or less; $L_i = 0.5$ nH, or less; $R_r = 5 \Omega$, or less; $R_f = 1 \Omega$, or less. The equivalent circuits do not include



FIGURE 10.24 Equivalent circuits for the ON and OFF states of a PIN diode. (a) Reverse bias (OFF) state. (b) Forward bias (ON) state.

parasitic effects due to packaging, which may be important. The forward bias current is typically 10-30 mA, and the reverse bias voltage is typically 40-60 V. The bias signal must be applied to the diode with RF chokes and DC blocks to isolate it from the RF signal.

Single-Pole Switches

A PIN diode can be used in either a series or a shunt configuration to form a single-pole. single-throw RF switch. These circuits are shown in Figure 10.25, with bias networks. In the series configuration of Figure 10.25a, the switch is on when the diode is forward biased, while in the shunt configuration the switch is on when the diode is reversed biased. In both cases, input power is reflected when the switch is in the OFF state. The DC blocks should have a very low impedance at the RF operating frequency, while the RF choke inductors should have a very high RF impedance. In some designs, high impedance quarter-wavelength lines can be used in place of the chokes, to provide RF blocking.

Ideally, a switch would have zero insertion loss in the ON state, and infinite attenuation in the OFF state. Realistic switching elements, of course, result in some insertion





loss for the ON state, and finite attenuation for the OFF state. Knowing the diode parameters for the equivalent circuits of Figure 10.24 allows the insertion loss for the ON and OFF states to be calculated for the series and shunt switches. With reference to the simplified switch circuits of Figure 10.26, we can define the insertion loss in terms of the actual load voltage, V_L , and V_0 , which is the load voltage which would appear if the switch (Z_d) were absent:

$$IL = -20 \log \left| \frac{\tilde{V}_L}{V_0} \right|.$$
 10.58

Simple circuit analysis applied to the two cases of Figure 10.26 gives the following results:

$$IL = -20 \log \left| \frac{2Z_0}{2Z_0 + Z_d} \right| \qquad \text{(series switch)}, \qquad 10.59a$$

$$IL = -20\log \left| \frac{2Z_d}{2Z_d + Z_0} \right| \qquad \text{(shunt switch).} \qquad 10.59b$$

In both cases, Z_d is the diode impedance for either the reverse or forward bias state. Thus,

$$Z_{d} = \begin{cases} Z_{r} = R_{\tau} + j(\omega L_{i} - 1/\omega C_{j}) & \text{for reverse bias} \\ Z_{f} = R_{f} + j\omega L_{i} & \text{for forward bias.} \end{cases}$$
10.60

The ON state or OFF state insertion loss of a switch can usually be improved by adding an external reactance in series or in parallel with the diode, to compensate for the reactance of the diode. This technique usually reduces the bandwidth, however.

Several single-throw switches can be combined to form a variety of multiple-pole and/or multiple-throw configurations [6]. Figure 10.27 shows series and shunt circuits for a single-pole, double-throw switch; such a switch requires at least two switching





Simplified equivalent circuits for the series and shunt single-pole PIN diode switches. (a) Series switch, (b) Shunt switch.





elements. In operation, one diode is biased in the low impedance state, with the other diode biased in the high impedance state. The input signal is switched from one output to the other by reversing the diode states. The quarter-wave lines of the shunt circuit limit the bandwidth of this configuration.

EXAMPLE 10.5 Single-pole PIN Diode Switch

A single-pole switch is to be constructed using a PIN diode with the following parameters: $C_j = 0.1$ pF, $R_r = 1 \Omega$, $R_f = 5.0 \Omega$, $L_i = 0.4$ nH. If the operating frequency is 5 GHz, and $Z_0 = 50 \Omega$, what circuit (series or shunt) should be used to obtain the greatest ratio of off-to-on attenuation?

Solution

We first compute the diode impedance for the reverse and forward bias states, using (10.60):

$$Z_d = \begin{cases} Z_r = R_r + j(\omega L_i - 1/\omega C_j) &= 1.0 - j305.7 \ \Omega \\ Z_f = R_f + j\omega L_i &= 0.5 + j12.6 \ \Omega. \end{cases}$$

Then using (10.59) gives the insertion losses for the ON and OFF states of the series and shunt switches as follows:

For the series circuit,

$$IL_{\text{off}} = -20 \log \left| \frac{2Z_0}{2Z_0 + Z_f} \right| = 0.11 \text{ dB},$$
$$IL_{\text{off}} = -20 \log \left| \frac{2Z_0}{2Z_0 + Z_r} \right| = 10.16 \text{ dB}.$$

For the shunt circuit.

$$IL_{\rm on} = -20 \log \left| \frac{2Z_r}{2Z_r + Z_0} \right| = 0.03 \, \mathrm{dB},$$
$$IL_{\rm off} = -20 \log \left| \frac{2Z_f}{2Z_f + Z_0} \right| = 7.07 \, \mathrm{dB}.$$

So the series configuration has the greatest difference in attenuation between the ON and OFF states, but the shunt circuit has the lowest ON insertion loss.

PIN Diode Phase Shifters

Several types of microwave phase shifters can be constructed with PIN diode switching elements. Compared with ferrite phase shifters, diode phase shifters have the advantages of small size, integrability with planar circuitry, and high speed. The power requirements for diode phase shifters, however, are generally greater than those for a latching ferrite phase shifter, because diodes require continuous bias current while the latching ferrite device requires only a pulsed current to change its state. There are basically three types of PIN diode phase shifters: *switched line*, *loaded line*, and *reflection*.

The switched-line phase shifter is the most straightforward type, using two singlepole double-throw switches to route the signal flow between one of two transmission lines of different length. See Figure 10.28. The differential phase shift between the two





paths is given by

$$\Delta \phi = \beta (\ell_2 - \ell_1), \qquad 10.61$$

where β is the propagation constant of the line. If the transmission lines are TEM (or quasi-TEM, like microstrip), this phase shift is a linear function of frequency, which implies a true time delay between the input and output ports. This is a useful feature in broadband systems, because distortion is minimized. This type of phase shifter is also inherently reciprocal, and can be used for both receive and transmit functions. The insertion loss of the switched line phase shifter is equal to the loss of the SPDT switches plus line losses.

Like many other types of phase shifters, the switched-line phase shifter is usually designed for binary phase shifts of $\Delta \phi = 180^{\circ}$, 90° , 45° , etc. One potential problem with this type of phase shifter is that resonances can occur in the OFF line, if its length is near a multiple of $\lambda/2$. The resonant frequency will be slightly shifted due to the series junction capacitances of the reversed biased diodes, so the lengths ℓ_1 and ℓ_2 should be chosen with this effect taken into account.

A design that is useful for small amounts of phase shift (generally 45° , or less) is the loaded-line phase shifter. The basic principle of this type of phase shifter can be illustrated with the circuit of Figure 10.29a, which shows a transmission line loaded with a shunt susceptance, *jB*. The reflection and transmission coefficients can be written as

$$\Gamma = \frac{1 - (1 + jb)}{1 + (1 + jb)} = \frac{-jb}{2 + jb},$$
10.62a

$$T = 1 + \Gamma = \frac{2}{2 + jb}.$$
 10.62b

where $b = BZ_0$ is the normalized susceptance. Thus the phase shift in the transmitted



FIGURE 10.29

Loaded-line phase shifters. (a) Basic circuit. (b) Practical loaded-line phase shifter and its equivalent circuit.

wave introduced by the load is

$$\Delta \phi = \tan^{-1} \frac{b}{2}.$$
 10.63

which can be made positive or negative, depending on the sign of b. A disadvantage is the insertion loss that is inherently present, due to the reflection from the shunt load. And increasing b to obtain a larger $\Delta \phi$ entails a greater insertion loss, as seen from (10.52a).

The reflections from the shunt susceptance can be reduced by using the circuit of Figure 10.29b, where two shunt loads are separated by a $\lambda/4$ length of line. Then the partial reflection from the second load will be 180° out of phase with the partial reflection from the first load, leading to a cancellation. We can analyze this circuit by calculating its ABCD matrix and comparing it to the ABCD matrix of an equivalent line having a length θ_e and characteristic impedance Z_e . Thus, for the loaded line,

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix} = \begin{bmatrix} 1 & 0 \\ jB & 1 \end{bmatrix} \begin{bmatrix} 0 & jZ_0 \\ j/Z_0 & 0 \end{bmatrix} \begin{bmatrix} 1 & 0 \\ jB & 1 \end{bmatrix}$$
$$= \begin{bmatrix} -BZ_0 & jZ_0 \\ j(1/Z_0 - B^2Z_0) & -BZ_0 \end{bmatrix}, \qquad 10.64a$$

while the equivalent transmission line has an ABCD matrix given by

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix} = \begin{bmatrix} \cos \theta_e & j Z_e \sin \theta_e \\ j \sin \theta_e / Z_e & \cos \theta_e \end{bmatrix}.$$
 10.64b

So we have that

$$\cos\theta_e = -BZ_0 = -b, \qquad 10.65a$$

$$Z_e = Z_0 \cos \theta_e = \frac{Z_0}{\sqrt{1 - b^2}}.$$
 10.65b

For small b, θ_e will be close to $\pi/2$, and these results reduce to

$$heta_e \simeq rac{\pi}{2} + b,$$
 10.66a

$$Z_e \simeq Z_0 \left(1 + \frac{b}{2}\right). \tag{10.66b}$$

The susceptance, B, can be implemented with a lumped inductor or capacitor, or with a stub, and switched between two states with an SPST diode switch.

The third type of PIN diode phase shifter is the reflection phase shifter, which uses an SPST switch to control the path length of a reflected signal. Usually a quadrature hybrid is used to provide a two-port circuit, although other types of hybrids, or even a circulator, could be used for this purpose.

Figure 10.30 shows a reflection-type phase shifter using a quadrature hybrid. In operation, an input signal divides equally among the two right-hand ports of the hybrid. The diodes are both biased in the same state (forward or reverse biased), so the waves reflected from the two terminations will add in phase at the indicated output port. Turning the diodes on or off changes the total path length for both reflected waves by $\Delta \phi$,



FIGURE 10.30 A reflection phase shifter using a quadrature hybrid.

producing a phase shift of $\Delta \phi$ at the output. Ideally, the diodes would look like short circuits in their on state, and open circuits in their off state, so that the reflection coefficients at the right side of the hybrid can be written as $\Gamma = e^{j\phi}$ for the diodes in their ON state, and $\Gamma = e^{j(\phi + \Delta \phi)}$ for the diodes in their OFF state. There are infinite numbers of choices of line lengths that give the desired $\Delta \phi$ (that is, the value of ϕ is a degree of freedom), but it can be shown that bandwidth is optimized if the reflection coefficients for the two states are phase conjugates. Thus, if $\Delta \phi = 180^{\circ}$, the best bandwidth will be obtained if $\Gamma = \pm j$, or $\phi = 0$, 2π , etc.

A good input match for the reflection-type phase shifter requires that the diodes be well-matched. The insertion loss is limited by the loss of the hybrid, as well as the forward and reverse resistances of the diodes. Impedance transformation sections can be used to improve performance in this regard.

10.4 MICROWAVE INTEGRATED CIRCUITS

The trend of any maturing electrical technology is toward smaller size, lighter weight, lower cost, and increased complexity. Microwave technology has been moving in this direction for the last 10–20 years, with the development of microwave integrated circuits [2]. This technology serves to replace bulky and expensive waveguide and coaxial components with small and inexpensive planar components, and is analogous to the digital integrated circuitry that has led to the rapid increase in sophistication of computer systems. Microwave integrated circuitry (MIC) can incorporate transmission lines, discrete resistors, capacitors, and inductors, and active devices such as diodes and transistors. MIC technology has advanced to the point where complete microwave subsystems, such as receiver front ends and radar transmit/receive modules, can be integrated on a chip that is only a few square millimeters in size.

There are two distinct types of microwave integrated circuits. *Hybrid* MICs have one layer of metallization for conductors and transmission lines, with discrete components (resistors, capacitors, transistors, diodes, etc.) bonded to the substrate. In a thin-film hybrid MIC, some of the simpler components are deposited on the substrate. Hybrid MICs were first developed in the 1960s, and still provide a very flexible and cost-effective means for circuit implementation. *Monolithic* microwave integrated circuits (MMICs) are a more recent development, where the active and passive circuit elements are grown on the substrate. The substrate is a semiconductor material, and several layers

of metal, dielectric, and resistive films are used. Below we will briefly describe these two types of MICs, in terms of the materials and fabrication processes that are required and the relative merits of each type of circuitry.

Hybrid Microwave Integrated Circuits

Material selection is an important consideration for any type of MIC: characteristics such as electrical conductivity, dielectric constant, loss tangent, thermal transfer, mechanical strength, and manufacturing compatability must be evaluated. Generally the substrate material is of primary importance. For hybrid MICs, alumina, quartz, and Tefton fiber are commonly used for substrates. Alumina is a rigid ceramic-like material with a dielectric constant of about 9–10. A high dielectric constant is often desirable for lower frequency circuits because it results in a smaller circuit size. At higher frequencies, however, the substrate thickness must be decreased to prevent radiation loss and other spurious effects; then the transmission lines (typically microstrip, slotline, or coplanar waveguide) can become too narrow to be practical. Quartz has a lower dielectric constant (~ 4) which, with its rigidity, makes it useful for higher frequency (> 20 GHz) circuits. Tefton and similar types of soft plastic substrates have dielectric constants ranging from 2 to 10, and can provide a large substrate area at a low cost, as long as rigidity and good thermal transfer are not required. Transmission line conductors for hybrid MICs are typically copper or gold.

After the circuit has been designed, the next step is to lay out the components and their interconnections, and make a mask for the metallization layer. This step is usually performed with a CAD system for integrated circuit layout, such as MICAD or CALMA[®]. The mask itself may be made on rubylith (a soft mylar film), usually at a magnified scale $(2\times, 5\times, 10\times, \text{etc.})$ for a high accuracy. Then an actual-size mask is made on a thin sheet of glass or quartz. The metallized substrate is coated with photoresist, covered with the mask, and exposed to a light source. The substrate can then be etched to remove the unwanted areas of metal. Plated-through, or via, holes can be made by evaporating a layer of metal inside a hole that has been drilled in the substrate. Finally, the discrete components are soldered or wire-bonded to the conductors. This is generally the most labor-intensive part of hybrid MIC fabrication, and therefore the most expensive part of the process.

Then the MIC can be tested. Often provision is made for variations in component values and other circuit tolerances by providing tuning or trim stubs that can be manually trimmed for each circuit. This increases circuit yield, but also increases cost since trimming involves labor at a highly skilled level. The layout of a typical hybrid MIC circuit is shown in Figure 10.31. A photograph of a hybrid MIC module is shown in Figure 10.32.

Monolithic Microwave Integrated Circuits

Progress in GaAs material processing and device development since the late 1970s has led to the feasibility of the monolithic microwave integrated circuit, where all pas-

[®]Registered trademark of General Electric Company.



FIGURE 10.31 Layout of a hybrid microwave integrated circuit.



FIGURE 10.32

Photograph of one of the 25,344 hybrid integrated T/R modules used in Raytheon's Ground Based Radar system. This X-band module contains phase shifters, amplifiers, switches, couplers, a ferrite circulator, and associated control and bias circuitry.

Courtesy of Raytheon Company, Lexington, Mass.

sive and active components required for a given circuit can be grown or implanted in the substrate. Potentially, the MMIC can be made at low cost because the manual tabor required for fabricating hybrid MICs is eliminated. In addition, a single wafer can contain a large number of circuits, all of which can be processed and fabricated simultaneously.

The substrate of an MMIC must be a semiconductor material to accommodate the fabrication of active devices; the type of devices and the frequency range dictate the type of substrate material. Thus, silicon bipolar transistors can be used up to several gigahertz, silicon-on-sapphire (SOS) MESFETs can be used up to several gigahertz, and submicron gate-length GaAs FETs have been used up to 60 GHz. The GaAs FET is a very versatile circuit element, finding applications in low-noise amplifiers, high-gain amplifiers, broadband amplifiers, mixers, oscillators, phase shifters, and switches. Thus, GaAs is probably the most common substrate for MMICs, but silicon, silicon-on-sapphire, and indium phosphide (InP) are sometimes used.

Transmission lines and other conductors are usually made with gold metallization. To improve adhesion of the gold to the substrate, a thin layer of chromium or titanium is generally deposited first. These metals are relatively lossy, so the gold layer must be made at least several skin depths thick to reduce attenuation. Capacitors and overlaying lines require insulating dielectric films, such as SiO, SiO₂, Si₂N₄, and Ta₂O₅. These materials have a high dielectric constant and low loss, and are compatible with integrated circuit processing. Resistors require the deposition of lossy films; NiCr, Ta, Ti, and doped GaAs are commonly used.

Designing an MMIC requires extensive use of CAD software, for circuit design and optimization as well as mask generation. Careful consideration must be given to the circuit design to allow for component variations and tolerances, and the fact that circuit trimming after fabrication will be difficult, or impossible (and defeats the goal of low-cost production). Thus, effects such as transmission line discontinuities, bias networks, spurious coupling, and package resonances must be taken into account.

After the circuit design has been finalized, the masks can be generated. One or more masks are generally required for each processing step. Processing begins by forming an active layer in the semiconductor substrate for the necessary active devices; this can be done by ion implantation or by epitaxial techniques. Then active areas are isolated by etching or additional implantation, leaving mesas for the active devices. Next, ohmic contacts are made to the active device areas by alloying a gold or gold/germanium layer onto the substrate. FET gates are then formed with a titanium/platinum/gold compound deposited between the source and drain areas. At this time the active device processing has been essentially completed, and intermediate tests can be made to evaluate the wafer. If it meets specifications, the next step is to deposit the first layer of metallization for contacts, transmission lines, inductors, and other conducting areas. Then resistors are formed by depositing resistive films, and the dielectric films required for capacitors and overlays are deposited. A second layer of metallization completes the formation of capacitors and any remaining interconnections. The final processing steps involve the bottom, or backside, of the substrate. First it is lapped to the required thickness, then via holes are formed by etching and plating. Via holes provide ground connections to the circuitry on the top side of the substrate, and provide a heat dissipation path from



FIGURE 10.33 Layout of a monolithic microwave integrated circuit.

the active devices to the ground plane. After the processing has been completed, the individual circuits can be cut from the wafer, and tested. Figure 10.33 shows the structure of a typical MMIC, and Figure 10.34 shows a photograph of an X-band monolithic integrated power amplifier.

Monolithic microwave integrated circuits are not without some disadvantages, when compared with hybrid MICs or other type of circuitry. First, MMICs tend to waste large areas of relatively expensive semiconductor substrate for components such as transmission lines and hybrids. Also, the processing steps and required tolerances for an MMIC are very critical, resulting in low yields. These factors tend to make MMICs expensive, especially when made in small quantities (less than several hundred). MMICs generally require a more thorough design procedure to include effects such as component tolerances and discontinuities, and debugging, tuning, or trimming after fabrication is difficult. Because their small size limits heat dissipation, MMICs cannot be used for circuits requiring more than moderate power levels. And high-Q resonators and filters are difficult to implement in MMIC form because of the inherent resistive losses in MMIC materials.

Besides the obvious features of small size and weight, MMICs have some unique advantages over other types of circuits. Since it is very easy to fabricate additional FETs in an MMIC design, circuit flexibility and performance can often be enhanced with little additional cost. Also, monolithically integrated devices have much less parasitic reactance than discrete packaged devices, so MMIC circuits can often be made with broader bandwidth than hybrid circuits. And MMICs generally give very reproducible results. especially for circuits from the same wafer.



FIGURE 10.34 Photograph of a monolithic integrated X-band power amplifier. This circuit uses eight heterojunction bipolar transistors with power dividers/combiners at the input and output to produce 5 watts.

Courtesy of M. Adlerstein and R. Wohlert, Raytheon Company, Lexington, Mass.

10.5 OVERVIEW OF MICROWAVE SOURCES

A source of microwave power is obviously essential for any microwave system. Communication and radar systems generally use a relatively high-power source for the transmitter, and one or more low-power sources for local oscillator and down conversion functions in the receiver. Radar transmitters are often operated in a pulsed mode, and peak powers that are much greater than the continuous power rating of the source can then be attained. Electronic warfare systems use sources in much the same way as a radar system, with the additional requirement for tunability over a wide bandwidth. Radiometer and radio astronomy receiver systems require low-power sources for local oscillators (although it can be argued that the primary source of microwave power for such systems is the radiation emitted from the hot body under observation). Test and measurement systems usually require a low-power microwave source, often tunable over a wide bandwidth. And the microwave oven, that most common of all microwave systems, requires a single-frequency high-power source.

At present, these requirements are met with a variety of solid-state and microwave tube sources. Generally the division is between solid-state sources for low power and low frequencies, and tubes for high power and/or high frequencies. Figure 10.35 illustrates the power versus frequency performance for these two types of sources. Solid-state sources have the advantages of small size, ruggedness, low cost, and compatability with microwave integrated circuits, and so are usually preferred whenever they can meet the



FIGURE 10.35 Power versus frequency performance of solid-state sources and microwave. tubes.

necessary power and frequency requirements. But very high power applications are dominated by microwave tubes, and even though the power and frequency performance of solid-state sources is steadily improving, it appears that the need for microwave tubes will not be eliminated any time soon.

Presenting the thorough discussion of the operation and design of microwave solidstate and tube sources that these components deserve is well beyond the scope of this book. Instead, we will briefly describe and summarize the performance of several of the most common types of solid-state and microwave tube sources.

Solid-State Sources

Solid-state microwave sources can be categorized as two-terminal devices (diodes), or three-terminal devices (transistor oscillators). The most common diode sources are the Gunn diode and the IMPATT diode, both of which directly convert a DC bias to RF power in the frequency range of about 2 to 100 GHz. The Gunn diode is a transferred-electron device that uses a bulk semiconductor (usually GaAs or InP), as opposed to a pn junction [5]. This effect leads to a negative-resistance characteristic that can be employed with an external resonator to produce a stable oscillator. DC to RF efficiencies are generally less than 10%. Figure 10.36 shows the power (continuous and pulsed) versus frequency performance for a variety of commercially available Gunn sources. Gunn diodes can also be used as negative-resistance reflection-type amplifiers. Figure 10.37 shows a photograph of two commercially available Gunn diode sources.

The IMPATT (IMPact ionization Avalanche Transit Time) diode uses a reversebiased pn junction to generate microwave power [5]. The material is usually silicon or gallium arsenide, and the diode is operated with a relatively high voltage (70-100 V)to achieve a reverse-biased avalanche breakdown current. When coupled with a high-Qresonator and biased at an appropriate operating point, a negative-resistance effect can



FIGURE 10.36Power versus frequency performance of Gunn diodes. • pulsed; • continuous.MA: Microwave Associates; BTL: Bell Telephone Labs; CA: Cayuga.

be achieved at the RF operating frequency, and oscillation will occur. Oscillator circuit design for IMPATT diodes essentially follows the procedure outlined in Section 11.5 for negative-resistance oscillators. IMPATT sources are generally more noisy than sources using Gunn diodes, but are capable of higher powers and higher DC to RF conversion efficiencies. IMPATTs also have better temperature stability than Gunn diodes. Figure 10.38 shows the power versus frequency performance for typical commercial IMPATT sources. IMPATT diodes can also be used as negative-resistance amplifiers.





Transistor oscillator design will be discussed in Section 11.5. Such sources generally have lower frequency and power capabilities when compared to Gunn or IMPATT sources, but do offer several advantages over diodes. First, oscillators using GaAs FETs are readily compatible with MIC or MMIC circuitry, allowing easy integration with FET amplifiers and mixers, while diode devices are less compatible. Also, a transistor oscillator circuit is much more flexible than a diode source. This is because the negativeresistance oscillation mechanism of a diode is determined and limited by the physical characteristics of the device itself, while the operating characteristics of a transistor source can be adjusted to a greater degree by the oscillator circuitry. Thus, transistor oscillators allow more control of the frequency of oscillation, temperature stability, and output noise than do diode sources. Transistor oscillator circuits also lend themselves well to frequency tuning, phase or injection locking, and to various modulation requirements. Transistor sources are relatively efficient, but presently not capable of very high power outputs.

An increasingly popular fixed-frequency transistor oscillator design, especially useful for local oscillator applications, is the transistor dielectric resonator oscillator (TDRO). This uses a small, high-Q dielectric resonator as the frequency-determining load for an FET oscillator circuit. A very good temperature stability can be achieved, and the resonator is compatible with MIC design, leading to a compact and low-cost unit.

Tunable sources are necessary in many types of electronic warfare systems, frequencyhopping radar and communications systems, and test systems. Transistor oscillators can be made tunable by using an adjustable element in the resonant load, such as a varactor diode or a magnetically-biased YIG sphere [6]. When reverse-biased, the junction capacitance of a varactor diode can be controlled with the DC bias voltage. Thus, a voltage-controlled oscillator (VCO) can be made by using a reverse-biased varactor



FIGURE 10.38 Power versus frequency performance of IMPATT diodes.

diode in the tank circuit of a transistor oscillator. In a YIG-tuned oscillator (YTO), a single-crystal YIG sphere is used to control the inductance of a coil in the tank circuit of the oscillator. Since YIG is a ferrimagnetic material, its effective permeability can be controlled with an external DC magnetic bias field, thus controlling the oscillator frequency. YIG oscillators can be made to tune over a decade or more of bandwidth, while varactor-tuned oscillators are limited to a tuning range of about an octave. YIG-tuned oscillators, however, cannot be tuned as fast as varactor oscillators.

In many applications the RF power requirement exceeds the power capacity of a single solid-state source. But because of the many advantages offered by solid-state sources, a lot of effort has been directed toward increasing output power through the use of various power combining techniques. Thus, the outputs of two or more sources are combined in phase, effectively multiplying the output power of a single source by the number of individual sources being used. In principle, an unlimited amount of RF power can be generated in this manner; in practice, however, factors such as high-order modes and combiner losses limit the multiplication to about 10-20 dB.

Power combining can be done by combining powers at the device level or at the circuit level. In some applications, power can be combined spatially by using an array of

antennas, where each element is fed with a separate source. At the device level, several diode or transistor junctions are essentially connected in parallel over an electrically small region, and used as a single device. This technique is thus limited to a relatively few device junctions. At the circuit level, the power output from N devices can be combined with an N-way combiner. The combining circuit may be an N-way Wilkinson-type network, or a similar type of planar combining network. Resonant cavities can also be used for this purpose. These various techniques all have their own advantages and disadvantages in terms of efficiency, bandwidth, isolation between sources, and circuit complexity.

Microwave Tubes

The first truly practical microwave source was the *magnetron tube*, developed in England in the 1930s, and later providing the impetus for the development of microwave radar during World War II. Since then, a large variety of microwave tubes have been designed for the generation and amplification of microwave power. In recent years, solid-state devices have been progressively filling the roles that were once reserved for microwave tubes, generally with a multitude of advantages. But tubes are still essential for the generation of very high powers (10 kW to 10 MW), and for the higher millimeter wave frequencies (100 GHz and higher). Here we will provide a brief overview of some of the most common microwave tubes, and their basic characteristics. Several of these tubes are not actually sources by themselves, but are high-power amplifiers. Such tubes are used in conjunction with lower power sources (often solid-state sources) in transmitter systems.

There is a wide variety of tube geometries, as well as a wide variety of principles on which tube operation is based, but all tubes have several common features. First, all tubes involve the interaction of an electron beam with an electromagnetic field, inside a glass or metal vacuum envelope. Thus, a way must be provided for RF energy to be coupled outside the envelope; this is usually accomplished with transparent windows or coaxial coupling probes or loops. Next, a hot cathode is used to generate a stream of electrons by thermionic emission. Cathodes are usually fabricated from a barium oxidecoated metal surface, or an impregnated tungsten surface. The electron stream is then focused into a narrow beam by a focusing anode with a high voltage bias. Alternatively, a solenoidal electromagnet can be used to focus the electron beam. For pulsed operation, a beam modulating electrode is used between the cathode and anode. A positive bias voltage will attract electrons from the cathode, and turn the beam on, while a negative bias will turn the beam off. After the electron beam leaves the region of the tube where the desired interaction with the RF field takes place, a collector element is used to provide a complete current path back to the cathode power supply. The assembly of the cathode, focusing anode, and modulating electrode is called the *electron gun*. Because of the requirement for a high vacuum, and the need to dissipate large amounts of heat, microwave tubes are generally very large and bulky. In addition, tubes often require large, heavy biasing magnets, and high voltage power supplies. Factors to consider when choosing a particular type of tube include power output, frequency, bandwidth, tuning range, and noise.

Microwave tubes can be grouped into two categories, depending on the type of electron beam-field interaction. In *linear-beam*, or "O," type tubes the electron beam traverses the length of the tube, and is parallel to the electric field. In the *crossed-field*, or "m." type tube the focusing field is perpendicular to the accelerating electric field. Microwave tubes can also be classified as either oscillators or amplifiers.

The klystron is a linear-beam tube that is widely used as both an amplifier and an oscillator. In a klystron amplifier, the electron beam passes through two or more resonant cavities. The first cavity accepts an RF input and modulates the electron beam by bunching it into high-density and low-density regions. The bunched beam then travels to the next cavity, which accentuates the bunching effect. At the final cavity the RF power is extracted, at a highly amplified level. Two cavities can produce up to about 20 dB of gain, while using four cavities (about the practical limit) can give 80-90 dB gain. Klystrons are capable of peak powers in the megawatt range, with RF output/DC input power conversion efficiencies of 30-50%.

The reflex klystron is a single-cavity klystron tube which operates as an oscillator by using a reflector electrode after the cavity to provide positive feedback via the electron beam. It can be tuned by mechanically adjusting the cavity size. The major disadvantage of klystrons is their narrow bandwidth, which is a result of the high-Q cavities required for electron bunching. Klystrons have very low AM and FM noise tevels.

The narrow bandwidth of the klystron amplifier is overcome in the *traveling wave tube* (TWT). The TWT is a linear-beam amplifier that uses an electron gun and a focusing magnet to accelerate a beam of electrons through an interaction region. Usually the interaction region consists of a slow-wave helix structure, with an RF input at the electron gun end, and an RF output at the collector end. The helical structure slows down the propagating RF wave so that it travels at the same velocity as the wave and beam travel along the interaction region, and amplification is effected. Then the amplified signal is coupled from the end of the helix. The TWT has the highest bandwidth of any amplifier tube, ranging from 30 to 120%; this makes it very useful for electronic warfare systems, which require high power over broad bandwidths. It has a power rating of several hundred watts (typically), but this can be increased to several kilowatts by using an interaction region consisting of a set of coupled cavities: the bandwidth will be reduced, however. The efficiency of the TWT is relatively small, typically ranging from 20 to 40%.

A variation of the TWT is the *backward wave oscillator* (BWO). The difference between a TWT and the BWO is that in the BWO, the RF wave travels along the helix from the collector toward the electron gun. Thus the signal for amplification is provided by the bunched electron beam itself, and oscillation occurs. A very useful feature of the BWO is that its output frequency can be tuned by varying the DC voltage between the cathode and the helix; tuning ranges of an octave or more can be achieved. The power output of the BWO, however, is relatively low (typically less than 1 W), so these tubes are generally being replaced with solid-state sources.

Another type of linear-beam oscillator tube is the *extended interaction oscillator* (EIO). The EIO is very similar to a klystron, and uses an interaction region consisting of several cavities coupled together, with positive feedback to support oscillation. It has a narrow tuning bandwidth, and a moderate efficiency, but it can supply high powers at frequencies up to several hundred GHz. Only the gyratron can deliver more power.

Crossed-field tubes include the *magnetron*, the *crossed-field amplifier*, and the gyratron. As previously mentioned, the magnetron was the first high-power microwave source. It consists of a cylindrical cathode surrounded by a cylindrical anode with several cavity resonators along the inside of its periphery. A magnetic bias field is applied parallel to the cathode-anode axis. In operation, a cloud of electrons is formed which rotates around the cathode in the interaction region. As with linear-beam devices, electron bunching occurs, and energy is transferred from the electron beam to the RF wave. RF power can be coupled out of the tube with a probe, loop, or aperture window.

Magnetrons are capable of very high power outputs—on the order of several kilowatts. And the magnetron has an efficiency of 80% or more. A significant disadvantage, however, is that they are very noisy, and cannot maintain frequency or phase coherency when operated in a pulsed mode. These factors are important for high-performance pulsed radars, where processing techniques operate on a sequence of returned pulses. (Modern radars of this type today generally use a stable low-noise solid-state source, followed by a TWT for power amplification.) The application of magnetrons is now primarily for microwave cooking.

The crossed-field amplifier (CFA) has a geometry similar to a TWT, but employs a crossed-field interaction that is similar to that of the magnetron. The RF input is applied to a slow-wave structure in the interaction region of the CFA, but the electron beam is deflected by a negatively biased electrode, called the sole, to force the beam perpendicular to the slow-wave structure. In addition, a magnetic bias field is applied perpendicular to this electric field, and perpendicular to the electron beam direction. The magnetic field exerts a force on the electron beam that counteracts the field from the sole. In the absence of an RF input, the electric and magnetic fields are adjusted so that their effects on the electron beam cancel, leaving the beam to travel parallel to the slow-wave structure. Applying an RF field causes velocity modulation of the beam, and bunching occurs. The beam is also periodically deflected toward the slow-wave circuit, producing an amplified signal. Crossed-field amplifiers have very good efficiencies—up to 80%, but the gain is limited to 10-15 dB. Also, the CFA has a noisier output than either a klystron amplifier or TWT. Its bandwidth can be up to 40%.

Another crossed-field tube is the gyratron, which can be used as an amplifier or an oscillator. This tube consists of an electron gun with input and output cavities along the axis of the electron beam, similar to a klystron amplifier. But the gyratron also has a solenoidal bias magnet that provides an axial magnetic field. This field forces the electrons to travel in tight spirals down the length of the tube. The electron velocity is high enough so that relativistic effects are important. Bunching occurs, and energy from the transverse component of the electron velocity is coupled to the RF field.

A significant feature of the gyratron is that the frequency of operation is determined by the bias field strength and the electron velocity, as opposed to the dimensions of the tube itself. This makes the gyratron especially useful for millimeter wave frequencies; it offers the highest output power (10-100 kW) of any tube in this frequency range. It also has a high efficiency for tubes in the millimeter wave range. The gyratron is a relatively new type of tube, but it is rapidly replacing tubes such as reflex ktystrons and EIOs as sources of millimeter wave power.

Figures 10.39 and 10.40 summarize the power versus frequency performance of microwave tube oscillators and amplifiers,









a.

:
REFERENCES

- M. E. Hines, "The Virtues of Nonlinearity-Detection. Frequency Conversion, Parametric Amplification and Harmonic Generation." *IEFE Trans. Microwave Theory and Techniques*, vol. MTT-32, pp. 1097–1104. September 1984.
- [2] D. N. McQuiddy, Jr., J. W. Wassel, J. B. Lagrange, and W. R. Wisseman, "Monolithic Microwave Integrated Circuits: An Historical Perspective." *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-32, pp. 997–1008, September 1984.
- [3] F. T. Ulaby, R. K. Moore, and A. K. Fung, Microwave Remote Sensing: Active and Passive, Volume I, Microwave Remote Sensing, Fundamentals and Radiometry. Addison-Wesley, Reading, Mass, 1981.
- [4] S. A. Maas, Microwave Mixers, Artech House, Dedham, Mass, 1986.
- [5] S. Y. Yngvesson, Microwave Semiconductor Devices, Kluwer Academic Publishers, 1991.
- [6] I. Bahl and P. Bhartia, Microwave Solid-State Circuit Design, Wiley Interscience, N. Y. 1988.

PROBLEMS

- 10.1 The Y-factor method is to be used to measure the equivalent noise temperature of a component. A hot load of $T_1 = 300$ K and a cold load of $T_2 = 77$ K will be used. If the noise temperature of the amplifier is $T_c = 250$ K, what will be the ratio of power meter readings at the output of the component for the two loads?
- 10.2 Assume that measurement error introduces an uncertainty of ΔY into the measurement of Y in a Y-factor measurement. Derive an expression for the normalized error, $\Delta T_e/T_e$, of the equivalent noise temperature in terms of $\Delta Y/Y$ and the temperatures T_1 . T_2 , and T_e . Minimize this result with respect to T_e to obtain an expression for T_e in terms of T_1 and T_2 that will result in minimum error.
- 10.3 It is necessary to connect an antenna to a low-noise receiver with a transmission line. The frequency is 10 GHz, and the distance is 2 meters. The choices are: copper X-band waveguide, RG-8/U coaxial cable, or copper circular waveguide with an inner diameter of 2.0 cm. Which type of line should be used for the best noise figure? Disregard impedance mismatch.
- 10.4 A certain transmission line has a noise figure F = 1 dB at a temperature $T_0 = 290$ K. Calculate and plot the noise figure of this line as its physical temperature ranges from T = 0 K to 1000 K.
- 10.5 An amplifier with a bandwidth of 1 GHz has a gain of 15 dB and a noise temperature of 250 K. If the 1 dB compression point occurs for an input power level of -10 dBm, what is the dynamic range of the amplifier?
- 10.6 An amplifier with a gain of 12 dB, a bandwidth of 150 MHz, and a noise figure of 4.0 dB feeds a receiver with a noise temperature of 900 K. Find the noise figure of the overall system.
- 10.7 Consider the microwave system shown on the next page, where the bandwidth is 1 GHz centered at 20 GHz, and the physical temperature of the system is T = 300 K. What is the equivalent noise temperature of the source? What is the noise figure of the amplifier, in dB? What is the noise figure of the cascaded transmission line and amplifier, in dB? When the noisy source is connected to the system, what is the total noise power output of the amplifier, in dBm?



10.8 Consider the wireless local area network (WLAN) receiver front-end shown below, where the bandwidth of the bandpass filter is 100 MHz centered at 2.4 GHz. If the system is at room temperature, find the noise figure of the overall system. What is the resulting signal-to-noise ratio at the output, if the input signal power level is -90 dBm? Can the components be rearranged to give a better noise figure?



- 10.9 A two-way power divider has one output port terminated in a matched load, as shown below. Find the equivalent noise temperature of the resulting two-port network if the divider is
 - (a) an equal-split two-way resistive divider,
 - (b) a two-way Wilkinson divider, and
 - (c) a 3 dB quadrature hybrid. Assume the divider in each case is matched, and at room temperature.



- 10.10 A diode has the following parameters: $C_{j} = 0.1$ pF, $R_{s} = 15 \Omega$. $I_{s} = 0.1 \mu$ A, and $L_{p} = C_{p} = 0$. Compute the open-circuit voltage sensitivity at 10 GHz for $I_{0} = 0$. 20, and 50 μ A. Assume $\alpha = 1/25$ mV, and neglect the effect of bias current on the junction capacitance.
- 10.11 An input signal composed of two closely spaced frequencies, ω_1 and ω_2 , is applied to a mixer with an LO frequency of ω_0 . (See the input spectrum shown below.) Calculate and sketch the resulting output spectrum due to the v^2 term of the mixer response.



10.12 A phase detector produces an output signal proportional to the phase difference between two RF input signals. Let these input signals be expressed as

$$v_1 = v_0 \cos \omega t,$$

$$v_2 = v_0 \cos(\omega t + \theta).$$

If these two signals are applied to a single-balanced mixer using a 90° hybrid, show that the IF output signal, after low-pass filtering, is given by

$$i = k v_0^2 \sin \theta$$
,

where k is a constant. If the mixer uses a 180° hybrid, show that the corresponding output signal is given by

$$i = k v_0^2 \cos \theta.$$

- 10.13 A two-tone input with a 6 dB difference in the two signal levels is applied to a nonlinear component. What is the relative power ratio of the resulting two third-order intermodulation products $2\omega_1 - \omega_2$ and $2\omega_2 - \omega_1$, if ω_1 and ω_2 are close together.
- **10.14** Derive (10.57) for third-order intercept points, assuming the individual distortion products add in phase at the output of the cascade.
- 10.15 A single-pole, single-throw switch uses a PIN diode in a shunt configuration. The frequency is 4 GHz, $Z_0 = 50 \Omega$, and the diode parameters are $C_2 = 0.5 \text{ pF}$, $R_r = 0.5 \Omega$, $R_f = 0.3 \Omega$, $L_s = 0.3 \text{ nH}$. Find the electrical length of an open-circuited shunt stub, placed across the diode, to minimize the insertion loss for the ON state of the switch. Calculate the resulting insertion losses for the ON and OFF states.
- 10.16 A single-pole, single-throw switch is constructed using two identical PIN diodes in the arrangement shown below. In the ON state, the series diode is forward biased and the shunt diode is reversed biased; and vice versa for the OFF state. If f = 6 GHz, $Z_0 = 50 \Omega$, $C_f = 0.1$ pF, $R_r = 0.5 \Omega$, $R_f = 0.3 \Omega$, and $L_s = 0.4$ nH, determine the insertion losses for the ON and OFF states.



10.17 Consider the loaded-line phase shifter shown below. If $Z_0 = 50 \Omega$, find the necessary stub lengths for a differential phase shift of 45°, and calculate the resulting insertion loss for both states of the phase shifter. Assume all lines are lossless, and that the diodes can be approximated as ideal shorts or opens.



Design of Microwave Amplifiers and Oscillators

Amplification is one of the most basic and prevalent microwave circuit functions. Early microwave amplifiers used klystron or traveling-wave tubes, or two-terminal solidstate devices such as tunnel or varactor diodes. But due to the dramatic improvements in transistor processing technology that have occurred since the 1970s, most microwave amplifiers today use three-terminal solid-state devices such as gallium arsenide field effect transistors (FETs), silicon bipolar transistors, heterojunction bipolar transistors (HBTs), and high electron mobility transistors (HEMTs) [1]–[3]. Microwave transistor amplifiers are rugged, low-cost, reliable, and can be easily integrated in both hybrid and monolithic microwave integrated circuits. They can presently be used at frequencies up to 100 GHz in a wide variety of applications requiring low-noise figure, broad bandwidth, and medium power capacity. As discussed in Section 10.5, microwave tubes are still required for very high power and/or very high frequency applications, but continuing improvement in the performance of microwave transistors is steadily reducing the need for microwave tubes.

The development of microwave sources paralleled that of microwave amplifiers, with early sources using tubes (magnetrons and klystrons), followed by the application of twoterminal negative-resistance semiconductor devices such as Gunn and IMPATT diodes. The availability of microwave transistors has led to the development of solid-state sources capable of operating over a wide range of frequencies with medium power levels. Such sources are universally found in all modern microwave radar, communication, and instrumentation systems.

We begin this chapter with a brief overview of microwave FET and bipolar transistors, their small-signal equivalent circuits, and some biasing considerations. Since our emphasis will be on circuit design using transistors, as opposed to the physics of the device itself, we will treat transistors primarily in terms of their terminal characteristics. using either S parameters or an equivalent circuit model. Next we develop some general results for the gain and stability of a two-port network in terms of its S parameters, and apply this theory to the design of single-stage transistor amplifier design in Section 11.3. Section 11.4 discusses bandwidth considerations for transistor amplifiers, and details the theory and design of balanced amplifiers and distributed amplifiers. Finally, we discuss microwave oscillators using one-port negative resistance loads, transistor oscillators, and dielectric resonator oscillators.

11.1 CHARACTERISTICS OF MICROWAVE TRANSISTORS

Microwave transistors are used as amplifiers, oscillators, switches, phase shifters, mixers, and active filters. Most of these applications use either silicon bipolar transistors or GaAs field effect transistors (FETs). Silicon bipolar device technology is very mature and inexpensive compared to GaAs transistor technology. Bipolar transistors are capable of higher gain and power capacity at lower frequencies, but GaAs FETs generally have better noise figures and can operate at much higher frequencies. Present silicon bipolar transistors such as the heterojunction bipolar transistor (HBT) allow operation well into the millimeter wave region. GaAs FETs have been used at frequencies in excess of 100 GHz. Table 11.1 compares the gain and noise figure versus frequency for present-day microwave transistors [1].

In this section we will give a brief discussion of the basic construction of microwave GaAs FETs and silicon bipolar transistors, along with small-signal equivalent circuit models for these devices, and DC biasing considerations. The design of amplifiers and oscillators relies primarily on the terminal characteristics of the transistor, and these can be expressed either in terms of the two-port S parameters of the device, or in terms of the component values of an equivalent circuit. We will use the S parameter method for most of our design work, as this is a procedure that is both accurate and convenient, although it does have the drawback of requiring knowledge of the transistor S parameters (usually through measurement) over the frequency band of interest. This is usually not a serious problem unless a very wide frequency range is being considered, since the S parameters of microwave transistors typically change fairly slowly with frequency. In contrast, the use of a good transistor equivalent circuit model involves only a few circuit parameters which are generally stable over a wide frequency range. An equivalent circuit model can also provide a closer linkage between the operation of the device and its physical parameters.

Microwave Field Effect Transistors (FETs)

Microwave field effect transistors can be used at frequencies well into the millimeter wave range with high gain and low noise figure, making them the device of choice for hybrid and monolithic integrated circuits at frequencies above 5 to 10 GHz [1].

Frequency (GHz)	GaAs FET		GaAs HEMT		Silicon Bipolar		GaAs HBT	
	Gain	Fran	Gain	Fmin	Gain	F _{min}	Gain	Fmu
4	20	0.5			15	2.5		
8	16	0.7	_	_	9	4.5		<u></u>
12	12	1.0	22	0.5	6	8.0	20	4,0
18	8	1.2	16	0.9	— -		16	-
36			12	1.7	_		10	
60			8	2.6		— —	7	

TABLE 11.1 Comparison of Gain and Noise Figure of Microwave Transistors (Gain and Noise Figure in dB)



FIGURE 11.1 (a) Cross section of a GaAs FET; (b) top view, showing drain, gate, and source contacts.

Figure 11.1 shows the construction of a typical GaAs FET. The desirable gain and noise features of the GaAs FET are a result of the higher electron mobility of GaAs compared to silicon, and the absence of shot noise. In operation, electrons are drawn from the source to the drain by the positive V_{ds} supply voltage. An input signal voltage on the gate then modulates these majority electron carriers, producing voltage amplification. The maximum frequency of operation is limited by the gate length; presently manufactured FETs have gate lengths on the order of 0.3 to 0.6 μ m, with corresponding upper frequency limits of 100 to 50 GHz.

A small-signal equivalent circuit for β microwave GaAs FET is shown in Figure 11.2, for a common-source configuration. The components and typical values for this circuit model are listed below:

 R_i (series gate resistance) = 7 Ω R_{ds} (drain-to-source resistance) = 400 Ω C_{gs} (gate-to-source capacitance) = 0.3 pF C_{ds} (drain-to-source capacitance) = 0.12 pF C_{gd} (gate-to-drain capacitance) = 0.01 pF g_m (transconductance) = 40 mS

This model does not include package parasities, which typically introduce small series resistances and inductances at the three terminals due to ohmic contacts and bonding leads. The dependent current generator $g_m V_c$ depends on the voltage across the gate-tosource capacitor C_{gs} , leading to a value of $|S_{21}| > 1$ under normal operating conditions



FIGURE 11.2 Small-signal equivalent circuit for a GaAs FET in the common-source configuration.

(where port 1 is at the gate, and port 2 is at the drain). The reverse signal path, given by S_{12} , is due solely to the capacitance C_{gd} . As can be seen from the above data, this is typically a very small capacitor which can often be ignored in practice. In this case, $S_{12} = 0$, and the device is said to be *unilateral*.

The equivalent circuit model of Figure 11.2 can be used to determine the upper frequency of operation for the transistor. The short-circuit current gain, G_i^{sc} , is defined as the ratio of drain to gate current when the output is short-circuited. For the unilateral case, where C_{ad} is assumed to be zero, this can be derived as

$$G_i^{sc} = \left| \frac{I_d}{I_g} \right| = \left| \frac{g_m V_c}{I_g} \right| = \frac{g_m}{\omega C_{gs}}.$$
 11.1

The upper frequency limit, f_T , is the frequency where the short-circuit current gain is unity, thus we have that

$$f_T = \frac{g_m}{2\pi C_{gs}}.$$

For proper operation, the transistor must be DC biased at an appropriate operating point. This depends on the application (low-noise, high-gain, high-power), the class of the amplifier (class A, class AB, class B), and the type of transistor (bipolar, FET, HBT, HEMT). Figure 11.3a shows a typical family of DC I_{ds} versus V_{ds} curves for a GaAs FET. For low-noise design, the drain current is generally chosen to be about 15% of I_{dss} (the saturated drain-to-source current). High-power circuits generally use higher values of drain current. DC bias voltage must be applied to the gate and drain, without disturbing the RF signal paths. This can be done as shown in Figure 11.3b, which shows the biasing and decoupling circuitry for a dual polarity supply. The RF chokes provide a very low DC resistance for biasing, and a very high impedance at RF frequencies to prevent the microwave signal from being shorted by the bias supply. Similarly,



FIGURE 11.3 (a) DC characteristics of a GaAs FET, (b) biasing and decoupling circuit for a GaAs FET.

the input and output decoupling capacitors block DC from the input and output lines, while allowing passage of microwave signals. There are many other types of bias circuits that provide compensation for temperature and device variations, and that can work with single-polarity power supplies.

Microwave Bipolar Transistors

Microwave bipolar transistors are usually of the *npn* type, and are often preferred over GaAs FETs at frequencies below 2 to 4 GHz because of higher gain and lower cost. Bipolar transistors are subject to shot noise as well as thermal noise effects, so their noise figure is not as good as that of FETs. Figure 11.4 shows the construction of a typical silicon bipolar transistor. In contrast to the FET, the bipolar transistor is current driven, with the base current modulating the collector current. The upper frequency limit of the bipolar transistor is controlled primarily by the base length, which is on the order of $0.1 \ \mu m$.

A small-signal equivalent circuit model for a microwave bipolar transistor is shown in Figure 11.5, for a common emitter configuration. The components of this circuit, along with typical values, are listed below:

- R_b (base resistance) = 7 Ω
- R_{π} (equivalent π resistance) = 110 Ω
- C_{π} (equivalent π capacitance) = 18 pF
- $C_{\rm c}$ (collector capacitance) = 18 pF

 g_m (transconductance) = 900 mS







FIGURE 11.5 Simplified hybrid- π equivalent circuit for a microwave bipolar transistor in the common emitter configuration.

Observe that the transconductance is much higher than that of the GaAs FET, leading to higher power gain at lower frequencies. The larger capacitances in the bipolar transistor model serve to reduce the gain at higher frequencies. The model in Figure 11.5 is popular because of its similarity to the FET equivalent circuit, but more sophisticated equivalent circuits may be advantageous for use over wide frequency ranges [1]. In addition, this model does not include parasitic resistances and inductances due to the base and emitter leads.

The equivalent circuit of Figure 11.5 can be used to estimate the upper frequency limit, f_T , where the short-circuit current gain is unity. The result is similar to that found above for the FET:

$$f_T = \frac{g_m}{2\pi C_\pi}.$$
 11.3

Figure 11.6a shows typical DC operating characteristics for a bipolar transistor. As with the FET, the biasing point for a bipolar transistor depends on the application and type of transistor, with low collector currents generally giving the best noise figure, and higher collector currents giving the best power gain. Figure 11.6b shows a typical bias and decoupling circuit for a bipolar transistor that requires only a single polarity supply.





(a) DC characteristics of a silicon bipolar transistor; (b) biasing and decoupling circuit for a bipolar transistor.

11.2 GAIN AND STABILITY

In this section we develop several expressions for the gain and stability of a general two-port amplifier circuit in terms of the S parameters of the transistor. These results will be used in the following sections for amplifier and oscillator design.

Two-Port Power Gains

Consider an arbitrary two-port network [S] connected to source and load impedances Z_S and Z_L , respectively, as shown in Figure 11.7. We will derive expressions for three types of power gain in terms of the S parameters of the two-port network and the reflection coefficients, Γ_S and Γ_L , of the source and load.

- Power Gain $= G = P_L/P_{in}$ is the ratio of power dissipated in the load Z_L to the power delivered to the input of the two-port network. This gain is independent of Z_S , although some active circuits are strongly dependent on Z_S .
- Available Gain $= G_A = P_{ava}/P_{avs}$ is the ratio of the power available from the two-port network to the power available from the source. This assumes conjugate matching of both the source and the load, and depends on Z_S but not Z_L .
- Transducer Power Gain = $G_T = P_L/P_{av_s}$ is the ratio of the power available from the two-port network to the power available from the source. This depends on both Z_S and Z_L .

These definitions differ primarily in the way the source and load are matched to the two-port device; if the input and output are both conjugately matched to the two-port, then the gain is maximized and $G = G_A = G_T$.

With reference to Figure 11.7, the reflection coefficient seen looking toward the load is

$$\Gamma_L \approx \frac{Z_L - Z_0}{Z_L + Z_0}, \qquad 11.4a$$

while the reflection coefficient seen looking toward the source is

$$\Gamma_S = \frac{Z_S - Z_0}{Z_S + Z_0},\tag{11.4b}$$

where Z_0 is the characteristic impedance reference for the S parameters of the two-port network.



FIGURE 11.7 A two-port network with general source and load impedances.

In general, the input impedance of the terminated two-port network will be mismatched with a reflection coefficient given by Γ_{in} , which can be determined using a signal flow graph (see Example 4.7), or by the following analysis. From the definition of the S parameters that $V_2^+ = \Gamma_L V_2^-$, we have

$$V_1^- = S_{11}V_1^+ + S_{12}V_2^+ = S_{11}V_1^+ + S_{12}\Gamma_L V_2^-, \qquad 11.5a$$

$$V_2^- = S_{21}V_1^+ + S_{22}V_2^+ = S_{21}V_1^+ + S_{22}\Gamma_LV_2^-.$$
 11.5b

Eliminating V_2^- from (11.5a) and solving for V_1^-/V_1^+ gives

$$\Gamma_{\rm in} = \frac{V_1^{-1}}{V_1^{+}} = S_{11} + \frac{S_{12}S_{21}\Gamma_L}{1 - S_{22}\Gamma_L} = \frac{Z_{\rm in} - Z_0}{Z_{\rm in} + Z_0},$$
11.6a

where Z_{in} is the impedance seen looking into port 1 of the terminated network. Similarly, the reflection coefficient seen looking into port 2 of the network when port 1 is terminated by Z_S is

$$\Gamma_{\text{out}} = \frac{V_2^-}{V_2^+} = S_{22} + \frac{S_{12}S_{21}\Gamma_S}{1 - S_{11}\Gamma_S}.$$
 11.6b

By voltage division,

$$V_{1} = V_{S} \frac{Z_{\text{in}}}{Z_{S} + Z_{\text{in}}} = V_{1}^{+} + V_{1}^{-} = V_{i}^{+} (1 + \Gamma_{\text{in}}).$$

Using

$$Z_{\rm in} = Z_0 \frac{1 + \Gamma_{\rm in}}{1 - \Gamma_{\rm in}},$$

from (11.6a) and solving for V_1^+ in terms of V_S gives

$$V_{1}^{+} = \frac{V_{S}}{2} \frac{(1 - \Gamma_{S})}{(1 - \Gamma_{S}\Gamma_{in})}.$$
 11.7

If peak values are assumed for all voltages, the average power delivered to the network is

$$P_{\rm in} = \frac{1}{2Z_0} \left| V_{\rm f}^+ \right|^2 \left(1 - \left| \Gamma_{\rm in} \right|^2 \right) = \frac{\left| V_S \right|^2}{8Z_0} \frac{\left| 1 - \Gamma_S \right|^2}{\left| 1 - \Gamma_S \Gamma_{\rm in} \right|^2} \left(1 - \left| \Gamma_{\rm in} \right|^2 \right)$$
 11.8

where (11.7) was used. The power delivered to the load is

$$P_{L} = \frac{\left|V_{2}^{-}\right|^{2}}{2Z_{0}} \left(1 - \left|\Gamma_{L}\right|^{2}\right).$$
 11.9

Solving for V_2^- from (11.5b), substituting into (11.9), and using (11.7) gives

$$P_{L} = \frac{|V_{1}^{+}|^{2}}{2Z_{0}} \frac{|S_{21}|^{2} \left(1 - |\Gamma_{L}|^{2}\right)}{\left|1 - S_{22}\Gamma_{L}\right|^{2}} \approx \frac{|V_{S}|^{2}}{8Z_{0}} \frac{|S_{21}|^{2} \left(1 - |\Gamma_{L}|^{2}\right) \left|1 - \Gamma_{S}\right|^{2}}{\left|1 - \Gamma_{S}\Gamma_{in}\right|^{2}}$$
(1.10)

The power gain can then be expressed as

$$G \approx \frac{P_L}{P_{\rm in}} = \frac{|S_{21}|^2 \left(1 - |\Gamma_L|^2\right)}{\left(1 - |\Gamma_{\rm in}|^2\right) \left|1 - S_{22}\Gamma_L\right|^2}$$
 1132

The power available from the source, P_{avs} , is the maximum power that can be delivered to the network. This occurs when the input impedance of the terminated network is conjugately matched to the source impedance, as discussed in Section 2.6. Thus, from (11.8),

$$P_{avs} = P_{in} \Big|_{\Gamma_{i0} = \Gamma_{S}^{*}} = \frac{|V_{S}|^{2}}{8Z_{0}} \frac{|1 - \Gamma_{S}|^{2}}{\left(1 - |\Gamma_{S}|^{2}\right)}$$
 11.12

Similarly, the power available from the network, P_{ava} , is the maximum power that can be delivered to the load. Thus, from (11.10),

$$P_{\text{avn}} = P_L \bigg|_{\Gamma_I = \Gamma_{\text{out}}^*} = \frac{|V_S|^2}{8Z_0} \frac{|S_{21}|^2 \left(1 - |\Gamma_{\text{out}}|^2\right) |1 - \Gamma_S|^2}{|1 - S_{22}\Gamma_{\text{out}}^*|^2 |1 - \Gamma_S\Gamma_{\text{in}}|^2} \bigg|_{\Gamma_L = \Gamma_{\text{out}}^*}$$
 11.13

In (11.13), Γ_{in} must be evaluated for $\Gamma_L = \Gamma_{out}^*$. From (11.6a), it can be shown that

$$\left|1 - \Gamma_{S}\Gamma_{in}\right|^{2} \bigg|_{\Gamma_{L} = \Gamma_{out}^{*}} = \frac{\left|1 - S_{11}\Gamma_{S}\right|^{2} \left(1 - \left|\Gamma_{out}\right|^{2}\right)^{2}}{\left|1 - S_{22}\Gamma_{out}^{*}\right|^{2}},$$

which reduces (11,13) to

$$P_{\text{avn}} = \frac{|V_S|^2}{8Z_0} \frac{|S_{21}|^2 |1 - \Gamma_S|^2}{|1 - S_{11}\Gamma_S|^2 (1 - |\Gamma_{\text{out}}|^2)}$$
 11.14

Observe that P_{avs} and P_{avn} have been expressed in terms of the source voltage, V_{s_1} which is independent of the input or load impedances. There would be confusion if these quantities were expressed in terms of V_1^+ , since V_1^+ is different for each of the calculations of P_L , P_{avs} , and P_{avn} .

Using (11.14) and (11.12), the available power gain is then

$$G_{A} = \frac{P_{\text{avn}}}{P_{\text{avs}}} = \frac{|S_{21}|^{2} \left(1 - |\Gamma_{S}|^{2}\right)}{|I - S_{11}\Gamma_{S}|^{2} \left(1 - |\Gamma_{\text{out}}|^{2}\right)}$$
 11.15

From (11.10) and (11.12), the transducer power gain is

$$G_{T} = \frac{P_{L}}{P_{\text{avs}}} = \frac{|S_{21}|^{2} \left(1 - |\Gamma_{S}|^{2}\right) \left(1 - |\Gamma_{L}|^{2}\right)}{|1 - \Gamma_{S}\Gamma_{\text{in}}|^{2} |1 - S_{22}\Gamma_{L}|^{2}}$$
11.16

A special case of the transducer power gain occurs when both the input and output are matched for zero reflection (in contrast to conjugate matching). Then $\Gamma_L = \Gamma_S = 0$, and

(11.16) reduces to

$$G_T = |S_{21}|^2 \,. \tag{11.17}$$

Another special case is the unilateral transducer power gain, G_{TU} , where $S_{12} = 0$ (or is negligibly small). This nonreciprocal characteristic is common to many practical amplifier circuits. From (11.6a), $\Gamma_{in} = S_{11}$ when $S_{12} = 0$, so (11.16) gives the unilateral transducer gain as

$$G_{TU} = \frac{|S_{21}|^2 \left(1 - |\Gamma_S|^2\right) \left(1 - |\Gamma_L|^2\right)}{\left|1 - S_{11}\Gamma_S\right|^2 \left|1 - S_{22}\Gamma_L\right|^2}$$
 11.18



EXAMPLE 11.1 Comparison of Power Gain Definitions

A microwave transistor has the following S parameters at 10 GHz, with a 50 Ω reference impedance:

$$S_{11} = 0.45 \angle 150^{\circ}$$

$$S_{12} = 0.01 \angle -10^{\circ}$$

$$S_{21} = 2.05 \angle 10^{\circ}$$

$$S_{22} = 0.40 \angle -150^{\circ}$$

The source impedance is $Z_S = 20 \ \Omega$ and the load impedance is $Z_L = 30 \ \Omega$. Compute the power gain, the available gain, and the transducer power gain.

Solution

From (11.4a,b) the reflection coefficients at the source and load are

$$\Gamma_S = \frac{Z_S - Z_0}{Z_S + Z_0} = \frac{20 - 50}{20 + 50} = -0.429,$$

$$\Gamma_L = \frac{Z_L - Z_0}{Z_L + Z_0} = \frac{30 - 50}{30 + 50} = -0.250.$$

From (11.6a,b) the reflection coefficients seen looking at the input and output of the terminated network are

$$\begin{split} \Gamma_{\rm in} &= S_{11} + \frac{S_{12}S_{21}\Gamma_L}{1 - S_{22}\Gamma_L} = 0.45\angle \underline{150}^\circ + \frac{(0.01\angle -\underline{10}^\circ)(2.05\angle \underline{10}^\circ)(-0.250)}{1 - (0.40\angle -\underline{150}^\circ)(-0.250)} \\ &= 0.455\angle \underline{150}^\circ, \\ \Gamma_{\rm out} &= S_{22} + \frac{S_{12}S_{21}\Gamma_S}{1 - S_{11}\Gamma_S} = 0.40\angle -\underline{150}^\circ + \frac{(0.01\angle -\underline{10}^\circ)(2.05\angle \underline{10}^\circ)(-0.429)}{1 - (0.45\angle \underline{150}^\circ)(-0.429)} \\ &= 0.408\angle -\underline{151}^\circ. \end{split}$$

Then from (11.1) the power gain is

$$G = \frac{|S_{21}|^2 \left(1 - |\Gamma_L|^2\right)}{\left(1 - |\Gamma_{10}|^2\right) |1 - S_{22}\Gamma_L|^2} = \frac{(2.05)^2 \left[1 - (0.250)^2\right]}{\left|1 - (0.40 \angle -150^\circ\right) (-0.250)\right|^2 \left[1 - (0.455)^2\right]}$$

= 5.94.

From (11.15) the available power gain is

$$G_{A} = \frac{|S_{21}|^{2} \left(1 - |\Gamma_{S}|^{2}\right)}{\left|1 - S_{11}\Gamma_{S}\right|^{2} \left(1 - |\Gamma_{out}|^{2}\right)} = \frac{(2.05)^{2} \left[1 - (0.429)^{2}\right]}{\left|1 - (0.45 \cancel{-150}^{\circ}) (-0.429)\right|^{2} \left[1 - (0.408)^{2}\right]}$$

= 5.85.

From (11.16) the transducer power gain is

$$G_{T} = \frac{|S_{21}|^{2} \left(1 - |\Gamma_{S}|^{2}\right) \left(1 - |\Gamma_{L}|^{2}\right)}{|1 - \Gamma_{S}\Gamma_{i0}|^{2} |1 - S_{22}\Gamma_{L}|^{2}}$$

=
$$\frac{(2.05)^{2} \left[1 - (0.429)^{2}\right] \left[1 - (0.250)^{2}\right]}{|1 - (0.40/-150^{\circ})(-0.250)|^{2} |1 - (-0.429)(0.455/150^{\circ})|^{2}} = 5.49.$$

A single-stage microwave transistor amplifier can be modeled by the circuit of Figure 11.8, where a matching network is used on both sides of the transistor to transform the input and output impedance Z_0 to the source and load impedances Z_S and Z_L . The most useful gain definition for amplifier design is the transducer power gain of (11.16), which accounts for both source and load mismatch. Thus, from (11.16), we can define separate effective gain factors for the input (source) matching network, the transistor itself, and the output (load) matching network as follows:

$$G_S = \frac{1 - \left|\Gamma_S\right|^2}{\left|1 - \Gamma_{in}\Gamma_S\right|^2},$$
11.19a

$$G_0 = |S_{21}|^2$$
, 11.19b

$$G_L = \frac{1 - |\Gamma_L|^2}{\left|1 - S_{22}\Gamma_L\right|^2}.$$
 11.19c





Then the overall transducer gain is $G_T = G_S G_0 G_L$. The effective gains from G_S and G_L are due to the impedance matching of the transistor to the impedance Z_0 .

If the transistor is unilateral, so that $S_{12} = 0$ or is small enough to be ignored, then (11.6) reduces to $\Gamma_{in} = S_{11}, \Gamma_{out} = S_{22}$, and the unilateral transducer gain reduces to $G_{TU} = G_S G_0 G_L$, where

$$G_{S} = \frac{1 - |\Gamma_{S}|^{2}}{\left|1 - S_{11}\Gamma_{S}\right|^{2}},$$
11.20*a*

$$G_0 = |S_{21}|^2 \,, \qquad 11.206$$

$$G_L = \frac{1 - |\Gamma_L|^2}{\left|1 - S_{22}\Gamma_L\right|^2}.$$
 11.20c

The above results have been derived using the S parameters of the transistor, but it is possible to obtain alternative expressions for gain in terms of the equivalent circuit parameters of the transistor. As an example, consider the evaluation of the unilateral transducer gain for a conjugately matched GaAs FET using the equivalent circuit of Figure 11.2 (with $C_{gd} = 0$). To conjugately match the transistor we choose source and load impedances as shown in Figure 11.9. Setting the series source inductive reactance $X = 1/(\omega C_{gs})$ will make $Z_{in} = Z_S^*$, and setting the short load inductive susceptance $B = -\omega C_{ds}$ will make $Z_{out} = Z_L^*$; this effectively eliminates the reactive elements from the FET equivalent circuit. Then by voltage division $V_c = V_S/(2j\omega R_i C_{gs})$, and the gain can be easily evaluated as

$$G_{TU} = \frac{P_L}{P_{\text{avs}}} = \frac{\frac{1}{8} \left| g_m V_c \right|^2 R_{ds}}{\frac{1}{8} \left| V_S \right|^2 / R_i} = \frac{g_m^2 R_{ds}}{4\omega^2 R_i C_{gs}^2} = \frac{R_{ds}}{4R_i} \left(\frac{f_T}{f} \right)^2.$$
 11.21

where the last step has been written in terms of the cutoff frequency, f_T , from (11.2). This shows the interesting result that the gain of a conjugately matched FET amplifier drops off as $1/f^2$, or 6 dB per octave. A photograph of a laboratory prototype GaAs FET amplifier is shown in Figure 11.10.



FIGURE 11.9 Unilateral FET equivalent circuit and source and load terminations for the calculation of unilateral transducer power gain.





Stability

Next we must consider the stability of the amplifier. In the circuit of Figure 11.8, oscillation is possible if either the input or output port impedance has a negative real part; this would then imply that $|\Gamma_{in}| > 1$ or $|\Gamma_{out}| > 1$. Because Γ_{in} and Γ_{out} depend on the source and load matching networks, the stability of the amplifier depends on Γ_s and Γ_L as presented by the matching networks. Thus, we define two types of stability:

- **1.** Unconditional stability: The network is unconditionally stable if $|\Gamma_{in}| < 1$ and $|\Gamma_{out}| < 1$ for all passive source and load impedances (i.e., $|\Gamma_S| < 1$ and $|\Gamma_L| < 1$).
- 2. Conditional stability: The network is conditionally stable if $|\Gamma_{in}| < 1$ and $|\Gamma_{out}| < 1$ only for a certain range of passive source and load impedances. This case is also referred to as potentially unstable.

Note that the stability condition of a network is frequency dependent, so that it is possible for an amplifier to be stable at its design frequency but unstable at other frequencies.

Applying the above requirements for unconditional stability to (11.6) gives the following conditions that must be satisfied by Γ_S and Γ_L if the amplifier is to be unconditionally stable:

$$|\Gamma_{in}| = \left| S_{11} + \frac{S_{12}S_{21}\Gamma_L}{1 - S_{22}\Gamma_L} \right| \ll 1_i$$
 11.22a

$$|\Gamma_{\text{out}}| = \left| S_{22} + \frac{S_{12}S_{21}\Gamma_S}{1 - S_{11}\Gamma_S} \right| < 1.$$
 11.22b

If the device is unilateral $(S_{12} = 0)$, these conditions reduce to the simple results that $|S_{11}| < 1$ and $|S_{22}| < 1$ are sufficient for unconditional stability. Otherwise, the inequalities of (11.22) define a range of values for Γ_S and Γ_L where the amplifier will be stable. Finding this range for Γ_S and Γ_L can be facilitated by using the Smith chart, and plotting the input and output stability circles. The stability circles are defined as the loci in the Γ_L (or Γ_S) plane for which $|\Gamma_{in}| = 1$ (or $|\Gamma_{out}| = 1$). The stability circles then define the boundaries between stable and potentially unstable regions of Γ_S and Γ_L . Γ_S and Γ_L must lie on the Smith chart ($|\Gamma_S| < 1$, $|\Gamma_L| < 1$ for passive matching networks).

We can derive the equation for the output stability circle as follows (the procedure for the input stability circle is the same—just interchange S_{11} and S_{22}). First use (11.22a) to express the condition that $|\Gamma_{1n}| = 1$ as

$$\left| S_{11} + \frac{S_{12}S_{21}\Gamma_L}{1 - S_{22}\Gamma_L} \right| = 1,$$

$$|S_{11}(1 - S_{22}\Gamma_L) + S_{12}S_{21}\Gamma_L| = |1 - S_{22}\Gamma_L|.$$
11.23

OΓ

Now define Δ (the determinant of the scattering matrix) as

$$\Delta = S_{11}S_{22} - S_{12}S_{21}. \tag{11.24}$$

Then we can write the above result as

$$|S_{11} - \Delta \Gamma_L| = |1 - S_{22} \Gamma_L|.$$
 11.25

Now square both sides and simplify to obtain,

$$\begin{split} |S_{11}|^{2} + |\Delta|^{2} |\Gamma_{L}|^{2} - (\Delta \Gamma_{L} S_{11}^{*} + \Delta^{*} \Gamma_{L}^{*} S_{11}) &= 1 + |S_{22}|^{2} |\Gamma_{L}|^{2} - (S_{22}^{*} \Gamma_{L}^{*} + S_{22} \Gamma_{L}) \\ (|S_{22}|^{2} - |\Delta|^{2}) \Gamma_{L} \Gamma_{L}^{*} - (S_{22} - \Delta S_{11}^{*}) \Gamma_{L} - (S_{22}^{*} - \Delta^{*} S_{11}) \Gamma_{L}^{*} &= |S_{11}|^{2} - 1 \\ \Gamma_{L} \Gamma_{L}^{*} - \frac{(S_{22} - \Delta S_{11}^{*}) \Gamma_{L} + (S_{22}^{*} - \Delta^{*} S_{11}) \Gamma_{L}^{*}}{|S_{22}|^{2} - |\Delta|^{2}} &= \frac{|S_{11}|^{2} - 1}{|S_{22}|^{2} - |\Delta|^{2}}. \end{split}$$
11.26

Now complete the square by adding $|S_{22} - \Delta S_{11}^*|^2/(|S_{22}|^2 - |\Delta|^2)^2$ to both sides:

or
$$\left| \Gamma_{L} - \frac{(S_{22} - \Delta S_{11}^{*})^{*}}{|S_{22}|^{2} - |\Delta|^{2}} \right|^{2} = \frac{|S_{11}|^{2} - 1}{|S_{22}|^{2} - |\Delta|^{2}} + \frac{|S_{22} - \Delta S_{11}^{*}|^{2}}{(|S_{22}|^{2} - |\Delta|^{2})^{2}},$$
$$\left| \Gamma_{L} - \frac{(S_{22} - \Delta S_{11}^{*})^{*}}{|S_{22}|^{2} - |\Delta|^{2}} \right| = \left| \frac{S_{12}S_{21}}{|S_{22}|^{2} - |\Delta|^{2}} \right|.$$
11.27

In the complex Γ plane, an equation of the form $|\Gamma - C| = R$ represents a circle with center at C (a complex number) and a radius R (a real number). Thus, (11.27) defines the output stability circle with a center C_L and radius R_L , where

$$C_L = \frac{(S_{22} - \Delta S_{11}^*)^*}{|S_{22}|^2 - |\Delta|^2} \qquad \text{(center)}, \qquad 11.28a$$

$$R_L = \left| \frac{S_{12} S_{21}}{|S_{22}|^2 - |\Delta|^2} \right| \qquad \text{(radius)}. \qquad 11.28b$$

Similar results can be obtained for the input stability circle by interchanging S_{11} and S_{22} :

$$C_S = \frac{(S_{11} - \Delta S_{22}^*)^*}{|S_{11}|^2 - |\Delta|^2}$$
 (center), 11.29a

$$R_{S} = \left| \frac{S_{12} S_{21}}{|S_{11}|^{2} - |\Delta|^{2}} \right|$$
 (radius). 11.29b

Given the S parameters of the device, we can plot the input and output stability circles to define where $|\Gamma_{in}| = 1$ and $|\Gamma_{out}| = 1$. On one side of the input stability circle we will have $|\Gamma_{out}| < 1$, while on the other side we will have $|\Gamma_{out}| > 1$. Similarly, we will have $|\Gamma_{in}| < 1$ on one side of the output stability circle, and $|\Gamma_{in}| > 1$ on the other side. So we now need to determine which areas on the Smith chart represent the stable region, for which $|\Gamma_{in}| < 1$ and $|\Gamma_{out}| < 1$.

Consider the output stability circles plotted in the Γ_L plane for $|S_{11}| < 1$ and $|S_{11}| > 1$, as shown in Figure 11.11. If we set $Z_L = Z_0$, then $\Gamma_L = 0$ and (11.22a) shows that $|\Gamma_{in}| = |S_{11}|$. Now if $|S_{11}| < 1$, then $|\Gamma_{in}| < 1$, so $\Gamma_L = 0$ must be in a stable region. This means that the center of the Smith chart ($\Gamma_L = 0$) is in the stable region, so all of the Smith chart ($|\Gamma_L| < 1$) that is exterior to the stability circle defines the stable range for Γ_L . This region is shaded in Figure 11.11a. Alternatively, if we set $Z_L = Z_0$ but have $|S_{11}| > 1$, then $|\Gamma_{in}| > 1$ for $\Gamma_L = 0$ and the center of the Smith chart nust be in an unstable region. In this case the stable region is the inside region of the stability circle that intersects the Smith chart, as illustrated in Figure 11.11b. Similar results apply to the input stability circle.

If the device is unconditionally stable, the stability circles must be completely outside (or totally enclose) the Smith chart. We can state this result mathematically as

$$||C_L| - R_L| > 1,$$
 for $|S_{11}| < 1,$ 11.30a

$$||C_S| - R_S| > 1,$$
 for $|S_{22}| < 1.$ 11.30b

If $|S_{11}| > 1$ or $|S_{22}| > 1$, the amplifier cannot be unconditionally stable because we can always have a source or load impedance of Z_0 leading to $\Gamma_S = 0$ or $\Gamma_L = 0$, thus causing $|\Gamma_{in}| > 1$ or $|\Gamma_{out}| > 1$.

Alternatively, it can be shown that the amplifier will be unconditionally stable if the following necessary and sufficient conditions are met:

$$K = \frac{1 - |S_{11}|^2 - |S_{22}|^2 + |\Delta|^2}{2|S_{12}S_{21}|} > 1,$$
11.31

and

$$|\Delta| < 1.$$
 11.32

We can prove this result by starting with the inequalities of (11.22). If the amplifier is to be unconditionally stable, the first condition that must be met is, from (11.22a),

$$|\Gamma_{\rm in}| = \left|S_{11} + \frac{S_{12}S_{21}\Gamma_L}{1 - S_{22}\Gamma_L}\right| < 1,$$



FIGURE 11.11 Output stability circles for a conditionally stable device. (a) $|S_{11}| < 1$. (b) $|S_{11}| > 1$.

for all $|\Gamma_L| < 1$. This inequality can be rewritten as

$$\left| \frac{1}{S_{22}} \left(S_{22} S_{11} + \frac{S_{12} S_{21} S_{22} \Gamma_L}{1 - S_{22} \Gamma_L} \right) \right| < 1,$$

$$\left| \frac{1}{S_{22}} \left(\Delta + \frac{S_{12} S_{21}}{1 - S_{22} \Gamma_L} \right) \right| < 1.$$
11.33

or

Now let $\Gamma_L = |\Gamma_L| e^{j\theta}$; allowable values of Γ_L for a passive load and matching network must lie within the unit circle defined by $|\Gamma_L| = 1$. The factor $1/(1 - S_{22}\Gamma_L)$ maps this circle into a new circle with a center at $1/(1 - |S_{22}|^2)$ and a radius of $|S_{22}|/(1 - |S_{22}|^2)$. This can be seen by setting $|\Gamma_L| = 1$ and writing this factor as

$$\frac{1}{1-S_{23}e^{j\theta}}=\frac{1}{1-|S_{22}|e^{j\phi}},$$

and noting that the maximum and minimum excursions of this factor occur for $\phi = 0$ and π , respectively, and are given as $1/(1-|S_{22}|)$ and $1/(1+|S_{22}|)$, respectively. The center of the circle is then the average of these values, $1/2 \left[1/(1-|S_{22}|) + 1/(1+|S_{22}|) \right] = 1/(1-|S_{22}|^2)$ while the radius is given by half the difference: $1/2 \left[1/(1-|S_{22}|) - 1/(1+|S_{22}|) \right] = |S_{22}|/(1-|S_{22}|^2)$. Thus we have that $1/(1-S_{22}e^{j\theta}) = 1/(1-|S_{22}|^2) + (|S_{22}|e^{j\psi})/(1-|S_{22}|^2)$.

Using this result allows us to rewrite (11.33) as

$$\frac{1}{|S_{22}|} \left| \left(\Delta + \frac{S_{12}S_{21}}{1 - |S_{22}|^2} \right) + \frac{S_{12}S_{21}|S_{22}|e^{j\psi}}{1 - |S_{22}|^2} \right| < 1,$$
 11.34

which must be true for all values of the angle, ψ . The left-hand side is maximum for that value of ψ that makes the phase angles of the two terms identical, so (11.34) can

be simplified to

$$\frac{1}{|S_{22}|} \left| \Delta + \frac{S_{12}S_{21}}{1 - |S_{22}|^2} \right| + \frac{|S_{12}S_{21}|}{1 - |S_{22}|^2} < 1,$$

$$0 \le \frac{1}{|S_{22}|} \left| \Delta + \frac{S_{12}S_{21}}{1 - |S_{22}|^2} \right| < 1 - \frac{|S_{12}S_{21}|}{1 - |S_{22}|^2}.$$
 11.35

or

Squaring both sides of this inequality and simplifying gives

$$|S_{12}S_{21}|^{2} + |\Delta|^{2}(1 - |S_{22}|^{2}) + \Delta S_{12}^{*}S_{21}^{*} + \Delta^{*}S_{12}S_{21} < |S_{22}|^{2}(1 - 2|S_{12}S_{21}| - |S_{22}|^{2}).$$

Using the result that $\Delta S_{12}^{*}S_{21}^{*} + \Delta^{*}S_{12}S_{21} = |S_{11}|^{2}|S_{22}|^{2} - |S_{12}S_{21}|^{2} - |\Delta|^{2}$ then gives
$$2|S_{12}S_{21}| < 1 - |S_{11}|^{2} - |S_{22}|^{2} + |\Delta|^{2}, \qquad 11.36$$

which is identical to the condition stated in (11.31). Since K remains unchanged after an interchange of S_{11} and S_{22} , we can conclude that (11.31) applies to $|\Gamma_{out}| < 1$, as well as $|\Gamma_{in}| < 1$.

Next, the right-hand side of (11.35) indicates that,

$$0 < 1 - |S_{22}|^2 - |S_{12}S_{21}|,$$

and the corresponding result obtained from (11.22b) is

$$0 < 1 - |S_{11}|^2 - |S_{12}S_{21}|.$$

Adding these two inequalities gives

$$2|S_{12}S_{21}| < 2 - |S_{11}|^2 - |S_{22}|^2.$$

From the triangle inequality we know that

$$|\Delta| = |S_{11}S_{22} - S_{12}S_{21}| \le |S_{11}S_{22}| + |S_{12}S_{21}|,$$

so the above result can be reduced to

$$\begin{aligned} 2(|\Delta| - |S_{11}S_{22}|) &< 2 - |S_{11}|^2 - |S_{22}|^2, \\ |\Delta| &< 1 - \frac{1}{2}(|S_{11}|^2 - 2|S_{11}S_{22}| + |S_{22}|^2), \\ |\Delta| &< 1 - \frac{1}{2}(|S_{11}| - |S_{22}|)^2 < 1, \end{aligned}$$

$$\begin{aligned} 11.37 \end{aligned}$$

which is identical to (11.32). Thus, we have shown that a two-port network will be unconditionally stable if and only if K > 1 and $|\Delta| < 1$.

If the device is only conditionally stable, operating points for Γ_S and Γ_L must be chosen in the stable region, and it is good practice to check the stability at several frequencies near the design frequency. If it is possible to accept a design with less than maximum gain, the transistor can usually be made to be unconditionally stable by using resistive loading [2].

While the $K-\Delta$ test of (11.31)–(11.32) is a mathematically rigorous condition for unconditional stability, it cannot be used to compare the relative stability of two or

more devices since it involves constraints on two parameters. Recently, however, a new criterion has been derived that combines the K- Δ parameters into a test involving only a single parameter, μ . The derivation of this criteria is straightforward, but lengthy, so we refer the reader to the literature [6], and present only the result for unconditional stability:

$$\mu = \frac{1 - |S_{11}|^2}{|S_{22} - S_{11}^* \Delta| + |S_{21}S_{12}|} > 1.$$
 11.38

Thus, if $\mu > 1$, the device is unconditionally stable. In addition, it can be said that larger values of μ imply greater stability.



EXAMPLE 11.2 Transistor Stability

The S parameters for the HP HFET-102 GaAs FET at 2 GHz with a bias voltage $V_{gs} = 0$ are given as follows ($Z_0 = 50 \Omega$):

$$S_{11} = 0.894 / -60.6^{\circ},$$

$$S_{21} = 3.122 / 123.6^{\circ},$$

$$S_{12} = 0.020 / 62.4^{\circ},$$

$$S_{22} = 0.781 / -27.6^{\circ}.$$

Determine the stability of this transistor by calculating K and $|\Delta|$, and plot the stability circles.

Solution From (11.31) and (11.24) we compute K and $|\Delta|$ as

$$\Delta = S_{11}S_{22} - S_{12}S_{21} = 0.696 \underline{/-83}^{\circ},$$
$$K = \frac{1 + |\Delta|^2 - |S_{11}|^2 - |S_{22}|^2}{2|S_{12}S_{21}|} = 0.607$$

We have $|\Delta| = 0.696 < 1$, but K < 1, so the device is potentially unstable. The centers and radii of the stability circles are given by (11.28) and (11.29):

$$C_{L} = \frac{(S_{22} - \Delta S_{11}^{*})^{*}}{|S_{22}|^{2} - |\Delta|^{2}} = 1.361 \angle 47^{\circ},$$

$$R_{L} = \frac{|S_{12}S_{21}|}{|S_{22}|^{2} - |\Delta|^{2}} = 0.50,$$

$$C_{S} = \frac{(S_{11} - \Delta S_{22}^{*})^{*}}{|S_{11}|^{2} - |\Delta|^{2}} \approx 1.132 \angle 68^{\circ},$$

$$R_{S} = \frac{|S_{12}S_{21}|}{|S_{11}|^{2} - |\Delta|^{2}} = 0.199.$$



FIGURE 11.12 Stability circles for Example 11.2.

This data can be used to plot the input and output stability circles, as shown in Figure 11.12. Since $|S_{11}| < 1$ and $|S_{22}| < 1$, the central part of the Smith chart represents the stable operating region for Γ_S and Γ_L . The unstable regions are darkened. Equation (11.38) gives $\mu = 0.86$ for this device.

11.3 SINGLE-STAGE TRANSISTOR AMPLIFIER DESIGN

Design for Maximum Gain (Conjugate Matching)

After the stability of the transistor has been determined, and the stable regions for Γ_S and Γ_L have been located on the Smith chart, the input and output matching sections.

can be designed. Since G_0 of (11.19b) is fixed for a given transistor, the overall gain of the amplifier will be controlled by the gains, G_S and G_L , of the matching sections. Maximum gain will be realized when these sections provide a conjugate match between the amplifier source or load impedance and the transistor. Because most transistors appear as a significant impedance mismatch (large $|S_{11}|$ and $|S_{22}|$), the resulting frequency response will be narrowband. In the next section we will discuss how to design for less than maximum gain, with a corresponding improvement in bandwidth. Broadband amplifier design will be discussed in Section 11.4.

With reference to Figure 11.8 and our discussion in Section 2.6 on conjugate impedance matching, we know that maximum power transfer from the input matching network to the transistor will occur when

$$\Gamma_{\rm in} = \Gamma_S^*, \qquad \qquad 11.39a$$

and the maximum power transfer from the transistor to the output matching network will occur when

$$\Gamma_{\rm out} = \Gamma_L^*. \tag{11.39b}$$

Then, assuming lossless matching sections, these conditions will maximize the overall transducer gain. From (11.16), this maximum gain will be given by

$$G_{T_{\rm max}} = \frac{1}{1 - |\Gamma_S|^2} |S_{21}|^2 \frac{1 - |\Gamma_L|^2}{|1 - S_{22}\Gamma_L|^2}.$$
 11.40

In the general case with a bilateral transistor, Γ_{in} is affected by Γ_{out} , and vice versa, so that the input and output sections must be matched simultaneously. Using (11.39) in (11.6) gives the necessary equations:

$$\Gamma_S^* = S_{11} + \frac{S_{12}S_{21}\Gamma_L}{1 - S_{22}\Gamma_L},$$
11.41a

$$\Gamma_{L}^{\bullet} = S_{22} + \frac{S_{12}S_{21}\Gamma_{S}}{1 - S_{11}\Gamma_{S}}.$$
 11.41b

We can solve for Γ_S by first rewriting these equations as follows:

$$\begin{split} \Gamma_{S} &= S_{11}^{*} + \frac{S_{12}^{*}S_{21}^{*}}{1/\Gamma_{L}^{*} - S_{22}^{*}}, \\ \Gamma_{L}^{*} &= \frac{S_{22} - \Delta\Gamma_{S}}{1 - S_{11}\Gamma_{S}}, \end{split}$$

where $\Delta = S_{11}S_{22} - S_{12}S_{21}$. Substituting this expression for Γ_L^* into the expression for Γ_S and expanding gives

$$\Gamma_{S}(1 - |S_{22}|^{2}) + \Gamma_{S}^{2}(\Delta S_{22}^{*} - S_{11}) = \Gamma_{S}(\Delta S_{11}^{*}S_{22}^{*} - |S_{11}|^{2} - \Delta S_{12}^{*}S_{21}^{*}) + S_{11}^{*}(1 - |S_{22}|^{2}) + S_{12}^{*}S_{21}^{*}S_{22}.$$

Using the result that $\Delta(S_{11}^*S_{22}^*-S_{12}^*S_{21}^*) = |\Delta|^2$ allows this to be rewritten as a quadratic equation for Γ_S :

$$(S_{11} - \Delta S_{22}^*)\Gamma_S^2 + (|\Delta|^2 - |S_{11}|^2 + |S_{22}|^2 - 1)\Gamma_S + (S_{11}^* - \Delta^* S_{22}) = 0.$$
 11.42

The solution is

$$\Gamma_S = \frac{B_1 \pm \sqrt{B_1^2 - 4|C_1|^2}}{2C_1}.$$
 11.43a

Similarly, the solution for Γ_L can be written as

$$\Gamma_L = \frac{B_2 \pm \sqrt{B_2^2 - 4|C_2|^2}}{2C_2}.$$
 11.43b

The variables B_1 , C_1 , B_2 . C_2 are defined as

$$B_1 = 1 + |S_{11}|^2 - |S_{22}|^2 - |\Delta|^2, \qquad 11.44a$$

$$B_2 = 1 + |S_{22}|^2 - |S_{11}|^2 - |\Delta|^2, \qquad 11.44b$$

$$C_1 = S_{11} - \Delta S_{22}^*, \qquad 11.44c$$

$$C_2 = S_{22} - \Delta S_{11}^*. \qquad 11.44d$$

The results are much simpler for the unilateral case. When $S_{12} = 0$, (11.41) shows that $\Gamma_S = S_{11}^*$ and $\Gamma_L = S_{22}^*$, and then maximum transducer gain of (11.40) reduces to

$$G_{TU_{\text{max}}} = \frac{1}{1 - |S_{11}|^2} |S_{21}|^2 \frac{1}{1 - |S_{22}|^2}.$$
 11.45

EXAMPLE 11.3 Conjugately Matched Amplifier Design

Design an amplifier for maximum gain at 4.0 GHz using single-stub matching sections. Calculate and plot the input return loss and the gain from 3 to 5 GHz. The GaAs FET has the following S parameters ($Z_0 = 50 \Omega$):

f (GHz)	S_{11}	S_{21}	S_{12}	S_{22}
3.0	0.80∠ <u>-89</u> °	2.86 <u>/99</u> °	0.03 <u>/56</u> °	0.76∠ <u>-41</u> °
4.0	0.72∠ <u>−116</u> °	2.60 <u>/76</u> °	0.03 <u>∕57</u> °	0.73∠ <u>-54</u> °
5.0	0.66∠ <u>142</u> °	2.39 <u>/54</u> °	0.03∠ <u>62</u> °	0.72∠ <u>–68</u> °

Solution

We first check the stability of the transistor by calculating Δ and K at 4.0 GHz:

$$\Delta = S_{11}S_{22} - S_{12}S_{21} = 0.488 \angle -162^{\circ},$$
$$K = \frac{1 - |S_{11}|^2 - |S_{22}|^2 + |\Delta|^2}{2|S_{12}S_{21}|} = 1.195.$$



Since $|\Delta| < 1$ and K > 1, the transistor is unconditionally stable at 4.0 GHz. There is no need to plot the stability circles.

For maximum gain, we should design the matching sections for a conjugate match to the transistor. Thus, $\Gamma_S \approx \Gamma_{\rm en}^*$ and $\Gamma_L \approx \Gamma_{\rm out}^*$, and Γ_S , Γ_L can be determined from (11.43):

$$\Gamma_{S} = \frac{B_{1} \pm \sqrt{B_{1}^{2} - 4|C_{1}|^{2}}}{2C_{1}} = 0.872 \angle 123^{\circ}$$

$$\Gamma_{L} = \frac{B_{2} \pm \sqrt{B_{2}^{2} - 4|C_{2}|^{2}}}{2C_{2}} = 0.876 \angle 61^{\circ}.$$

Then the effective gain factors of (11.19) can be calculated as

$$G_S = \frac{1}{1 - |\Gamma_S|^2} = 4.17 = 6.20 \text{ dB},$$

$$G_0 = |S_{21}|^2 = 6.76 = 8.30 \text{ dB},$$

$$G_L = \frac{1 - |\Gamma_L|^2}{|1 - S_{22}\Gamma_L|^2} = 1.67 = 2.22 \text{ dB}.$$

So the overall transducer gain will be

$$G_{T_{max}} = 6.20 + 8.30 + 2.22 = 16.7 \text{ dB}.$$

The matching networks can easily be determined using the Smith chart. For the input matching section, we first plot Γ_S , as shown in Figure 11.13a. The impedance, Z_S , represented by this reflection coefficient is the impedance seen looking into the matching section toward the source impedance, Z_0 . Thus, the matching section must transform Z_0 to the impedance Z_S . There are several ways of doing this, but we will use an open-circuited shunt stub followed by a length of line. Thus we convert to the normalized admittance y_s , and work backward (toward the load on the Smith chart) to find that a line of length 0.120 λ will bring us to the 1 + jb circle. Then we see that the required stub admittance is +j3.5, for an open-circuited stub length of 0.206 λ . A similar procedure gives a line length of 0.206 λ and a stub length of 0.206 λ for the output matching circuit.

The final amplifier circuit is shown in Figure 11.13b. This circuit only shows the RF components: the amplifier will also require some bias circuitry. The return loss and gain were calculated using a CAD package, interpolating the necessary S parameters from the table given above. The results are plotted in Figure 11.13c, and show the expected gain of 16.7 dB at 4.0 GHz, with a very good return loss. The bandwidth where the gain drops by 1 dB is about 2.5%.



FIGURE 11.13 Circuit design and frequency response for the transistor amplifier of Example 11.3. (a) Smith chart for the design of the input matching network.

Constant Gain Circles and Design for Specified Gain (Unilateral Device)

In many cases it is preferable to design for less than the maximum obtainable gain, to improve bandwidth or to obtain a specific value of amplifier gain. This can be done by designing the input and output matching sections to have less than maximum gains; in other words, mismatches are purposely introduced to reduce the overall gain. The design procedure is facilitated by plotting *constant gain circles* on the Smith chart, to represent loci of Γ_S and Γ_L that give fixed values of gain (G_S and G_L). To simplify our discussion, we will only treat the case of a unilateral device; the more general case of a bilateral device must sometimes be considered in practice and is discussed in detail in references [1], [2], and [3].



FIGURE 11.13 Continued. (b) RF circuit. (c) Frequency response.

In many practical cases $|S_{12}|$ is small enough to be ignored, and the device can then be assumed to be unilateral. This greatly simplifies the design procedure. The error in the transducer gain caused by approximating $|S_{12}|$ as zero is given by the ratio G_T/G_{TU} . It can be shown that this ratio is bounded by

$$\frac{1}{(1+U)^2} < \frac{G_T}{G_{TU}} < \frac{1}{(1-U)^2},$$
11.46

where U is defined as the unilateral figure of merit,

$$U = \frac{|S_{12}||S_{21}||S_{11}||S_{22}|}{(1 - |S_{11}|^2)(1 - |S_{22}|^2)}.$$
 11.47

Usually an error of a few tenths of a dB or less justifies the unilateral assumption.

The expression for G_S and G_L for the unilateral case are given by (11.20a) and (11.20c):

$$G_{S} = \frac{1 - |\Gamma_{S}|^{2}}{|1 - S_{11}\Gamma_{S}|^{2}},$$
$$G_{L} = \frac{1 - |\Gamma_{L}|^{2}}{|1 - S_{22}\Gamma_{L}|^{2}}.$$

These gains are maximized when $\Gamma_S = S_{11}^*$ and $\Gamma_L = S_{22}^*$, resulting in the maximum values given by

$$G_{S_{\max}} = \frac{1}{1 - |S_{11}|^2}, \qquad 11.48a$$

$$G_{L_{\max}} = \frac{1}{1 - |S_{22}|^2}.$$
 11.48b

Now define normalized gain factors g_S and g_L as

$$g_{S} = \frac{G_{S}}{G_{S_{\text{max}}}} = \frac{1 - |\Gamma_{S}|^{2}}{|1 - S_{11}\Gamma_{S}|^{2}} (1 - |S_{11}|^{2}), \qquad 11.49a$$

$$g_L = \frac{G_L}{G_{L_{max}}} = \frac{1 - |\Gamma_L|^2}{|1 - S_{22}\Gamma_L|^2} (1 - |S_{22}|^2).$$
 11.49b

Then we have that $0 \le g_S \le 1$, and $0 \le g_L \le 1$.

For fixed values of g_S and g_L , (11.49) represents circles in the Γ_S or Γ_L plane. To show this, consider (11.49a), which can be expanded to give

$$g_{S}|1 - S_{II}\Gamma_{S}|^{2} = (1 - |\Gamma_{S}|^{2})(1 - |S_{II}|^{2}),$$

$$(g_{S}|S_{II}|^{2} + 1 - |S_{II}|^{2})|\Gamma_{S}|^{2} - g_{S}(S_{II}\Gamma_{S} + S_{II}^{*}\Gamma_{S}^{*}) = 1 - |S_{II}|^{2} - g_{S},$$

$$\Gamma_{S}\Gamma_{S}^{*} - \frac{g_{S}(S_{II}\Gamma_{S} + S_{II}^{*}\Gamma_{S}^{*})}{1 - (1 - g_{S})|S_{II}|^{2}} = \frac{1 - |S_{II}|^{2} - g_{S}}{1 - (1 - g_{S})|S_{II}|^{2}}.$$
11.50

Now add $(g_S^2 |S_{11}|^2)/[1 - (1 - g_S)|S_{11}|^2]^2$ to both sides to complete the square:

$$\left|\Gamma_{S} - \frac{g_{S}S_{H}^{*}}{1 - (1 - g_{S})|S_{H}|^{2}}\right|^{2} = \frac{(1 - |S_{H}|^{2} - g_{S})[1 - (1 - g_{S})|S_{H}|^{2}] + g_{S}^{2}|S_{H}|^{2}}{[1 - (1 - g_{S})|S_{H}|^{2}]^{2}}.$$

Simplifying gives,

$$\left|\Gamma_{S} - \frac{g_{S}S_{11}^{*}}{1 - (1 - g_{S})|S_{11}|^{2}}\right| = \frac{\sqrt{1 - g_{S}}(1 - |S_{11}|^{2})}{1 - (1 - g_{S})|S_{11}|^{2}},$$
11.51

which is the equation of a circle with its center and radius given by

$$C_S = \frac{g_S S_{11}^*}{1 - (1 - g_S) |S_{11}|^2},$$
11.52*a*

$$R_S = \frac{\sqrt{1 - g_S}(1 - |S_{11}|^2)}{1 - (1 - g_S)|S_{11}|^2}.$$
 (1.52b)

The results for the constant gain circles of the output section can be shown to be,

$$C_L = \frac{g_L S_{22}^*}{1 - (1 - g_L) |S_{22}|^2},$$
11.53*a*

$$R_L = \frac{\sqrt{1 - g_L}(1 - |S_{22}|^2)}{1 - (1 - g_L)|S_{22}|^2}.$$
 11.53b

The centers of each family of circles lie along straight lines give by the angle of S_{11}^* or S_{22}^* . Note that when g_S (or g_L) = 1 (maximum gain), the radius R_S (or R_L) = 0, and the center reduces to S_{11}^* (or S_{22}^*), as expected. Also, it can be shown that the 0 dB gain circles ($G_S = 1$ or $G_L = 1$) will always pass through the center of the Smith chart. These results can be used to plot a family of circles of constant gain for the input and output sections. Then Γ_S and Γ_L can be chosen along these circles to provide the desired gains. The choices for Γ_S and Γ_L are not unique, but it makes sense to choose points close to the center of the Smith chart to minimize the mismatch and thus maximize the bandwidth. Alternatively, as we will see in the next section, the input network mismatch can be chosen to provide a low-noise design.



EXAMPLE 11.4 Amplifier Design for Specified Gain

Design an amplifier to have a gain of 11 dB at 4.0 GHz. Plot constant gain circles for $G_S = 2$ dB and 3 dB, and $G_L = 0$ dB and 1 dB. Calculate and plot the input return loss and overall amplifier gain from 3 to 5 GHz. The FET has the following S parameters ($Z_0 = 50 \Omega$):

f (GHz)	S_{11}	S_{21}	S_{12}	S_{22}
3	0.80 <u>∠−90</u> °	2.8∠100°	0	0.66 <u>2-50</u> ª
4	0.75∠ <u>−120</u> °	2.5 <u>∠80</u> °	0	0.60∠ <u>−70</u> °
5	0.71∠ <u>−140</u> ¢	2.3∠ <u>60</u> °	0	0.58 <u>∠−85</u> °

Solution

Since $S_{12} = 0$ and $|S_{11}| < 1$ and $|S_{22}| < 1$, the transistor is unilateral and unconditionally stable. From (11.48) we calculate the maximum matching section gains as

$$G_{S_{\text{max}}} = \frac{1}{1 - |S_{11}|^2} = 2.29 = 3.6 \text{ dB},$$

 $G_{L_{\text{max}}} = \frac{1}{1 - |S_{22}|^2} = 1.56 = 1.9 \text{ dB}.$

The gain of the mismatched transistor is

$$G_o = |S_{21}|^2 = 6.25 = 8.0 \text{ dB},$$

so the maximum unilateral transducer gain is

$$G_{TU_{\text{max}}} = 3.6 + 1.9 + 8.0 = 13.5 \text{ dB}.$$

Thus we have 2.5 dB more gain than is required by the specifications.

We use (11.49), (11.52), and (11.53) to calculate the following data for the constant gain circles:

$G_S = 3 \mathrm{d}\mathbf{B}$	$g_S=0.875$	$C_S = 0.706 \angle 120^{\circ}$	$R_{S} = 0.166$
$G_S = 2 \mathrm{dB}$	$g_S = 0.691$	$C_S=0.627 \angle 120^\circ$	$R_S = 0.294$
$G_L = 1 \mathrm{dB}$	$g_L = 0.806$	$C_L = 0.520 \angle \underline{70}^{\circ}$	$R_L = 0.303$
$G_L = 0 \mathrm{dB}$	$g_L = 0.640$	$C_L = 0.440 \angle \underline{70}^\circ$	$R_{S} = 0.440$

The constant gain circles are shown in Figure 11.14a. We choose $G_S = 2 \text{ dB}$



(a)

FIGURE 11.14 Circuit design and frequency response for the transistor amplifier of Example 11.4. (a) Constant gain circles.

and $G_L = 1$ dB, for an overall amplifier gain of 11 dB. Then we select Γ_S and Γ_L along these circles as shown, to minimize the distance from the center of the chart (this places Γ_S and Γ_L along the radial lines at 120° and 70°, respectively). Thus, $\Gamma_S = 0.33/120^\circ$ and $\Gamma_L = 0.22/70^\circ$, and the matching networks can be designed using shunt stubs as in Example 11.3.

The final amplifier circuit is shown in Figure 11.14b. The response was calculated using CAD software, with interpolation of the given S parameter data. The results are shown in Figure 11.14c, where it is seen the desired gain of 11 dB is achieved at 4.0 GHz. The bandwidth over which the gain varies by ± 1 dB or less is about 25%, which is considerably better than the bandwidth of the maximum gain design Example 11.3. The return loss, however, is not very good, being only about 5 dB at the design frequency. This is due to the deliberate mismatch introduced into the matching sections to achieve the specified gain.



FIGURE 11.14 Continued. (b) RF circuit. (c) Transducer gain and return loss.

Low-Noise Amplifier Design

Besides stability and gain. another important design consideration for a microwave amplifier is its noise figure. In receiver applications especially, it is often required to have a preamplifier with as low a noise figure as possible since, as we saw in Section 10.1, the first stage of a receiver front end has the dominant effect on the noise performance of the overall system. Generally it is not possible to obtain both minimum noise figure and maximum gain for an amplifier, so some sort of compromise must be made. This can be done by using constant gain circles and *circles of constant noise figure* to select a usable trade-off between noise figure and gain. Here we will derive the equations for constant noise figure circles, and show how they are used in transistor amplifier design.

As derived in references [4] and [5], the noise figure of a two-port amplifier can be expressed as

$$F = F_{\rm nun} + \frac{R_N}{G_S} |Y_S - Y_{\rm opt}|^2.$$
 11.54

where the following definitions apply:

 $Y_S = G_S + jB_S$ = source admittance presented to transistor.

 $Y_{opt} = optimum$ source admittance that results in minimum noise figure.

 $F_{\rm min}$ = minimum noise figure of transistor, attained when $Y_S = Y_{\rm opt}$.

 R_N = equivalent noise resistance of transistor.

 $G_S =$ real part of source admittance.

Instead of the admittance Y_S and Y_{opc} , we can use the reflection coefficients Γ_S and Γ_{opt} , where

$$Y_{S} = \frac{1}{Z_{0}} \frac{1 - \Gamma_{S}}{1 + \Gamma_{S}},$$
11.55a

$$Y_{\text{opt}} = \frac{1}{Z_0} \frac{1 - \Gamma_{\text{opt}}}{1 + \Gamma_{\text{opt}}}.$$
 11.55b

 Γ_S is the source reflection coefficient defined in Figure 11.8. The quantities F_{\min} . Γ_{opt} , and R_N are characteristics of the particular transistor being used, and are called the *noise* parameters of the device; they may be given by the manufacturer, or measured.

Using (11.55), the quantity $|Y_S - Y_{opt}|^2$ can be expressed in terms of Γ_S and Γ_{opt} :

$$|Y_{S} - Y_{opt}|^{2} = \frac{4}{Z_{0}^{2}} \frac{|\Gamma_{S} - \Gamma_{opt}|^{2}}{|1 + \Gamma_{S}|^{2}|1 + \Gamma_{opt}|^{2}}.$$
 (1.56)

Also,

$$G_{S} = \operatorname{Re}\{Y_{S}\} = \frac{1}{2Z_{0}} \left(\frac{1 - \Gamma_{S}}{1 + \Gamma_{S}} + \frac{1 - \Gamma_{S}^{*}}{1 + \Gamma_{S}^{*}} \right) = \frac{1}{Z_{0}} \frac{1 - |\Gamma_{S}|^{2}}{|1 + \Gamma_{S}|^{2}}.$$
 (1).57

Using these results in (11.54) gives the noise figure as

$$F = F_{\min} + \frac{4R_N}{Z_0} \frac{|\Gamma_S - \Gamma_{opt}|^2}{(1 - |\Gamma_S|^2)|1 + \Gamma_{opt}|^2}.$$
 11.58

For a fixed noise figure, F, we can show that this result defines a circle in the Γ_S plane. First define the *noise figure parameter*, N, as

$$N = \frac{|\Gamma_{S} - \Gamma_{opt}|^{2}}{1 - |\Gamma_{S}|^{2}} = \frac{F - F_{min}}{4R_{N}/Z_{0}}|1 + \Gamma_{opt}|^{2}, \qquad 11.59$$

which is a constant, for a given noise figure and set of noise parameters. Then rewrite (11.59) as

$$(\Gamma_{S} - \Gamma_{opt})(\Gamma_{S}^{*} - \Gamma_{opt}^{*}) = N(1 - |\Gamma_{S}|^{2}),$$

$$\Gamma_{S}\Gamma_{S}^{*} - (\Gamma_{S}\Gamma_{opt}^{*} + \Gamma_{S}^{*}\Gamma_{opt}) + \Gamma_{opt}\Gamma_{opt}^{*} = N - N|\Gamma_{S}|^{2},$$

$$\Gamma_{S}\Gamma_{S}^{*} - \frac{(\Gamma_{S}\Gamma_{opt}^{*} + \Gamma_{S}^{*}\Gamma_{opt})}{N+1} = \frac{N - |\Gamma_{opt}|^{2}}{N+1}.$$

Now add $|\Gamma_{opt}|^2/(N+1)^2$ to both sides to complete the square to obtain

$$\left|\Gamma_{S} - \frac{\Gamma_{\text{opt}}}{N+1}\right| = \frac{\sqrt{N(N+1-|\Gamma_{\text{opt}}|^{2})}}{(N+1)}.$$
 11.60

This result defines circles of constant noise figure with centers at

$$C_F = rac{\Gamma_{opt}}{N+1},$$
 11.61a

and radii of

$$R_F = \frac{\sqrt{N(N+1-|\Gamma_{\rm opt}|^2)}}{N+1}.$$
 (1.6)b



EXAMPLE 11.5 Low-Noise Amplifier Design

A GaAs FET is biased for minimum noise figure, and has the following S parameters and noise parameters at 4 GHz ($Z_0 = 50 \Omega$): $S_{11} = 0.6 \angle -60^\circ$, $S_{21} = 1.9 \angle 81^\circ$, $S_{12} = 0.05 \angle 26^\circ$, $S_{22} = 0.5 \angle -60^\circ$; $F_{\min} = 1.6 \text{ dB}$, $\Gamma_{\text{opt}} = 0.62 \angle 100^\circ$, $R_N = 20 \Omega$. For design purposes, assume the device is unilateral, and calculate the maximum error in G_T resulting from this assumption. Then design an amplifier having a 2.0 dB noise figure with the maximum gain that is compatible with this noise figure.

Solution

We first compute the unilateral figure of merit from (11.47):

$$U = \frac{|S_{12}S_{21}S_{11}S_{22}|}{(1 - |S_{11}|^2)(1 - |S_{22}|^2)} = 0.059.$$

Then from (11.46) the ratio G_T/G_{TU} is bounded as

$$\frac{1}{(1+U)^2} < \frac{G_T}{G_{TU}} < \frac{1}{(1-U)^2},$$
$$0.891 < \frac{G_T}{G_{TU}} < 1.130.$$

0ľ

In dB,

$$-0.50 < G_T - G_{TU} < 0.53$$
 dB.

where G_T and G_{TU} are now in dB. Thus, we should expect less than about ± 0.5 dB error in gain.

Next, we use (11.59) and (11.61) to compute the center and radius of the 2 dB noise figure circle:

$$N = \frac{F - F_{\min}}{4R_N/Z_0} |1 + \Gamma_{opt}|^2 = \frac{1.58 - 1.445}{4(20/50)} |1 + 0.62 \angle 100^{\circ}|^2$$

= 0.0986.
$$C_F = \frac{\Gamma_{opt}}{N+1} = 0.56 \angle 100^{\circ}$$
$$R_F = \frac{\sqrt{N(N+1 - |\Gamma_{opt}|^2)}}{N+1} = 0.24.$$

This noise figure circle is plotted in Figure 11.15a. Minimum noise figure $(F_{\min} = 1.6 \text{ dB})$ occurs for $\Gamma_S \approx \Gamma_{opt} = 0.62 \angle 100^{\circ}$.

Next we calculate data for several input section constant gain circles. From (11.52),

$G_{S}(dB)$	ĝs	C_S	R_{S}
1.0	0.805	0.52 <u>/60</u> °	0.300
1.5	0.904	0.56∠ <u>60</u> °	0.205
1.7	0.946	0.58 <u>∠60</u> °	0.150

These circles are also plotted in Figure 11.15a. We see that the $G_S = 1.7$ dB gain circle just intersects the F = 2 dB noise figure circle, and that any higher gain will result in a worse noise figure. From the Smith chart the optimum solution is then $\Gamma_S = 0.53 / 75^\circ$, yielding $G_S = 1.7$ dB and F = 2.0 dB.

solution is then $\Gamma_S = 0.53 \angle 75^\circ$, yielding $G_S = 1.7$ dB and F = 2.0 dB. For the output section we choose $\Gamma_L = S_{22}^* = 0.5 \angle 60^\circ$ for a maximum G_L of

$$G_L = \frac{1}{1 - |S_{22}|^2} = 1.33 = 1.25 \text{ dB}$$

The transistor gain is

$$G_0 = |S_{21}|^2 = 3.61 = 5.58 \text{ dB},$$



(a)



FIGURE 11.15

Circuit design for the transistor amplifier of Example 11.5. (a) Constant gain and noise figure circles. (b) RF circuit.

÷

so the overall transducer gain will be

$$G_{TU} = G_S + G_0 + G_L = 8.53 \text{ dB},$$

A complete AC circuit for the amplifier, using open-circuited shunt stubs in the matching sections, is shown in Figure 11.15b. A computer analysis of the circuit gave a gain of 8.36 dB.

11.4 BROADBAND TRANSISTOR AMPLIFIER DESIGN

The ideal microwave amplifier would have constant gain and good input matching over the desired frequency bandwidth. As the examples of the last section have shown, conjugate matching will give maximum gain only over a relatively narrow bandwidth, while designing for less than maximum gain will improve the gain bandwidth, but the input and output ports of the amplifier will be poorly matched. These problems are primarily a result of the fact that microwave transistors typically are not well matched to 50Ω , and large impedance mismatches are governed by the Bode-Fano gain-bandwidth criterion discussed in Section 5.9. Another factor is that $|S_{21}|$ decreases with frequency at the rate of 6 dB/octave. For these reasons, special consideration must be given to the problem of designing broadband amplifiers. Some of the common approaches to this problem are listed below; note in each case that an improvement in bandwidth is achieved only at the expense of gain, complexity, or similar factors.

- Compensated matching networks: Input and output matching sections can be designed to compensate for the gain rolloff in $|S_{21}|$, but generally at the expense of the input and output match.
- Resistive matching networks: Good input and output matching can be obtained by using resistive matching networks, with a corresponding loss in gain and increase in noise figure.
- Negative feedback: Negative feedback can be used to flatten the gain response
 of the transistor, improve the input and output match, and improve the stability
 of the device. Amplifier bandwidths in excess of a decade are possible with this
 method, at the expense of gain and noise figure.
- Balanced amplifiers: Two amplifiers having 90° couplers at their input and output can provide good matching over an octave bandwidth, or more. The gain is equal to that of a single amplifier, however, and the design requires two transistors and twice the DC power.
- Distributed amplifiers: Several transistors are cascaded together along a transmission line, giving good gain, matching, and noise figure over a wide bandwidth. The circuit is large, and does not give as much gain as a cascade amplifier with the same number of stages.

Below we discuss in detail the operation of balanced and distributed amplifiers.

Balanced Amplifiers

As we saw in Example 11.4, a fairly flat gain response can be obtained if the amplifier is designed for less than maximum gain, but the input and output matching will


FIGURE 11.16 A balanced amplifier using 90° hybrid couplers.

be poor. The balanced amplifier circuit solves this problem by using two 90° couplers to cancel input and output reflections from two identical amplifiers. The basic circuit of a balanced amplifier is shown in Figure 11.16. The first 90° hybrid coupler divides the input signal into two equal-amplitude components with a 90° phase difference, which drive the two amplifiers. The second coupler recombines the amplifier outputs. Because of the phasing properties of the hybrid coupler, reflections from the amplifier inputs cancel at the input to the hybrid, resulting in an improved impedance match; a similar effect occurs at the output of the balanced amplifier. The gain bandwidth is not improved over that of the single amplifier sections. This type of circuit is more complex than a single-stage amplifier since it requires two hybrid couplers and two separate amplifier sections, but it has a number of interesting advantages:

- The individual amplifier stages can be optimized for gain flatness or noise figure, without concern for input and output matching.
- Reflections are absorbed in the coupler terminations, improving input/output matching as well as the stability of the individual amplifiers.
- The circuit provides a graceful degradation of a −6 dB loss in gain if a single amplifier section fails.
- Bandwidth can be an octave or more, primarily limited by the bandwidth of the coupler.

In practice, balanced MMIC amplifiers often use Lange couplers, which are broadband and very compact, but quadrature hybrids and Wilkinson power dividers (with an extra 90° line on one arm) can also be used.

If we assume ideal hybrid couplers, then with reference to Figure 11.16 the voltages incident at the amplifiers can be written as

$$V_{A1}^{+} = \frac{1}{\sqrt{2}} V_{1}^{+}, \qquad 11.62a$$

$$V_{B1}^{+} = \frac{-j}{\sqrt{2}} V_{1}^{+}, \qquad 11.62b$$

where V_1^+ is the incident input voltage. Then the output voltage can be found as

$$V_2^- = \frac{-j}{\sqrt{2}}V_{A2}^+ + \frac{1}{\sqrt{2}}V_{B2}^+ = \frac{-j}{\sqrt{2}}G_A V_{A1}^+ + \frac{1}{\sqrt{2}}G_B V_{B1}^+ = \frac{-j}{2}V_1^+ (G_A + G_B)$$
 11.63

where (11.62) was used. Then we can write S_{21} as

$$S_{21} = \frac{V_2^+}{V_1^+} = \frac{-j}{2} \left(G_A + G_B \right), \qquad 11.64$$

which shows that the overall gain of the balanced amplifier is the average of the individual amplifier voltage gains.

The total reflected voltage at the input can be written as

$$V_{1}^{-} = \frac{1}{\sqrt{2}}V_{A1}^{-} + \frac{-j}{\sqrt{2}}V_{B1}^{-} = \frac{1}{\sqrt{2}}\Gamma_{A}V_{A1}^{+} + \frac{-j}{\sqrt{2}}\Gamma_{B}V_{B1}^{+} = \frac{1}{2}V_{1}^{+}(\Gamma_{A} - \Gamma_{B}).$$
 11.65

Then we can write S_{11} as

$$S_{11} = \frac{V_1^-}{V_1^+} = \frac{1}{2} (\Gamma_A - \Gamma_B).$$
 11.66

If the amplifiers are identical, then $G_A = G_B$ and $\Gamma_A = \Gamma_B$, and (11.66) shows that $S_{11} = 0$, and (11.64) shows that the gain of the balanced amplifier will be the same as the gain of an individual amplifier. If one amplifier fails, the overall gain will drop by 6 dB, with the remaining power lost in the coupler terminations. It can also be shown that the noise figure of the balanced amplifier is $F = (F_A + F_B)/2$, where F_A and F_B are the noise figures of the individual amplifiers.



EXAMPLE 11.6 Performance and Optimization of a Balanced Amplifier

Use the amplifier of Example 11.4 in a balanced amplifier configuration operating from 3 to 5 GHz. Use quadrature hybrids, and plot the gain and return loss over this frequency range. Using microwave CAD software, optimize the amplifier matching networks to give 10 dB gain over this band.

Solution

The amplifier of Example 11.4 was designed for a gain 11 dB at 4 GHz. As seen from Figure 11.14c, the gain varies by a few dB from 3 to 5 GHz, and the return loss is no better than 5 dB. We can design a quadrature hybrid, according to the discussion in Section 7.5, to have a center frequency of 4 GHz. Then the balanced amplifier configuration of Figure 11.16 can be modeled using a microwave CAD package, with the results shown in Figure 11.17. Note the dramatic improvement in return loss over the band, as compared with the result for the original amplifier in Figure 11.14c. The input matching is best at 4 GHz since this was the design frequency of the coupler; a coupler with better bandwidth will give improved results at the band edges. Also observe that the gain at 4 GHz is still 11 dB, and that it drops by a few dB at the band edges.

Most modern microwave CAD software packages have an optimization feature with which a small set of design variables can be adjusted to optimize a particular performance variable. In the present example, we will reduce the gain specification to 10 dB, and allow the CAD software to adjust the four



FIGURE 11.17 Gain and return loss, before and after optimization, for the balanced amplifier of Example 11.6.

transmission line stub and line lengths in the amplifier circuit of Figure 11.14b to give the best fit to this gain over the frequency range 3 to 5 GHz. Both amplifiers in the balanced circuit remain identical, so we should still see the improved input matching.

The results of this optimization are shown in Figure 11.17, where it can be seen that the gain response is much flatter over the operating band. The input match is still very good in the vicinity of the center frequency, with a slightly worse result at the low-frequency end. The optimized stub and line lengths for the amplifier matching networks are listed below:

Matching Network Parameter	Before Optimization	After Optimization
Input section stub length	0.100λ	0.109)
Input section line length	0.179λ	0.113λ
Output section line length	0.045λ	0.134λ
Output section stub length	0.432λ	0.461λ

These represent fairly small deviations from the lengths in the original matching networks.

Distributed Amplifiers

The concept of the distributed amplifier dates back to the 1940s [1], when it was used in the design of broadband vacuum tube amplifiers. With recent advances in microwave integrated circuit and device processing technology, the distributed amplifier has found new applications in broadband microwave amplifiers. Bandwidths in excess of a decade are possible, with good input and output matching. Distributed amplifiers are not capable



FIGURE 11.18 Configuration of an N-stage distributed amplifier.

of very high gains or very low noise figure, however, and generally are larger in size than an amplifier having comparable gain over a narrower bandwidth.

The basic configuration of a microwave distributed amplifier is shown in Figure 11.18. A cascade of N identical FETs have their gates connected to a transmission line having a characteristic impedance Z_g , with a spacing of ℓ_g , while the drains are connected to a transmission line of characteristic impedance Z_d , with a spacing ℓ_d . The operation of the distributed amplifier is very similar to that of the multihole waveguide coupler discussed in Section 7.4. The input signal propagates down the gate line, with each FET tapping off some of the input power. The amplified output signals from the FETs form a traveling wave on the drain line. The propagation constants and lengths of the gate and drain lines are chosen for constructive phasing of the output signals, and the termination impedances on the lines serve to absorb waves traveling in the reverse directions. The gate and drain capacitances of the FET effectively become part of the gate and drain transmission lines, while the gate and drain resistances introduce loss on these lines. This type of circuit is also known as a traveling wave amplifier.

Here we will analyze the distributed amplifier in terms of the loaded gate and drain transmission lines [7], although it is also possible to apply the concept of image parameters [8], or to simply use CAD software. An analytical treatment has the advantage of illustrating the underlying principles of operation of the amplifier, while the numerical CAD approach is recommended for better accuracy and optimization capabilities.

The first step in the analysis of the distributed amplifier is to employ the unilateral ($C_{gd} = 0$) version of the FET equivalent circuit of Figure 11.2 to decompose the circuit of Figure 11.18 into separate loaded transmission lines for the gate and drain terminals. These are shown in Figures 11.19 and 11.20. The gate and drain transmission lines are typically microstrip: the ground conductors are not shown in Figure 11.18, but they are in Figures 11.19 and 11.20. The gate and drain lines are isolated except for the coupling through the dependent current sources, where $I_{dn} = g_m V_{cn}$, and are matched at both ends. Figures 11.19b and 11.20b show the equivalent circuits for a single unit cell from the gate and drain lines, respectively. L_g and C_g are the inductance and capacitance per unit length of the gate transmission line, while $R_n \ell_g$ and C_{gs}/ℓ_g represent the equivalent per unit length loading due to the FET input resistance R_i and gate-to-source capacitance C_{gs} . Similar definitions apply to the quantities L_d , C_d , $R_{ds}\ell_d$, and C_{ds}/ℓ_d for the drain line. Thus we have taken the lumped loading



FIGURE 11.19 (a) Transmission line circuit for the gate line of the distributed amplifier; (b) equivalent circuit of a single unit cell of the gate line.

of each FET and distributed its circuit parameters over the transmission lines of each unit cell. This approximation is generally valid when the electrical lengths of the unit cells are small.

We can now use basic transmission line theory to find the effective characteristic impedance and propagation constants of the gate and drain lines. For the gate line, the





FIGURE 11.20 (a) Transmission line circuit for the drain line of the distributed amplifier; (b) equivalent circuit of a single unit cell of the drain line.

series impedance and shunt admittance per unit length can be written as

$$Z = j\omega L_g, \qquad \qquad 11.67a$$

$$Y = j\omega C_g + \frac{j\omega C_{gs}/\ell_g}{1 + j\omega R_i C_{gs}}.$$
 11.67b

If we assume that loss can be neglected for the calculation of characteristic impedance, as discussed in Section 2.7, then we have

$$Z_g = \sqrt{\frac{Z}{Y}} = \sqrt{\frac{L_g}{C_g + C_{gs}/\ell_g}}.$$
 11.68

For the calculation of the propagation constant we retain the resistive term, since this will lead to attenuation:

$$\gamma_g = lpha_g + jeta_g = \sqrt{ZY} = \sqrt{j\omega L_g \left[j\omega C_g + rac{j\omega C_{gs}/\ell_g}{1 + j\omega R_i C_{gs}}
ight]}.$$

If we assume small loss, such that $\omega R_i C_{gs} \ll 1$, then the above result can be simplified as follows:

$$\gamma_{g} = \alpha_{g} + j\beta_{g} \cong \sqrt{-\omega^{2}L_{g} \left[C_{g} + C_{gs} \left(1 - j\omega R_{i}C_{gs}\right)/\ell_{g}\right]}$$
$$\cong \frac{\omega^{2}R_{i}C_{gs}^{2}Z_{g}}{2\ell_{g}} + j\omega\sqrt{L_{g} \left(C_{g} + C_{gs}/\ell_{g}\right)}$$
11.69

For the drain line, the series impedance and shunt admittance per unit length are

$$Z = j\omega L_d, \qquad \qquad 11900$$

$$Y = \frac{1}{R_{ds}\ell_d} + j\omega \left(C_d + C_{ds}/\ell_d \right).$$
 11.70b

The characteristic impedance of the drain line can be written as

$$Z_d = \sqrt{\frac{Z}{Y}} = \sqrt{\frac{L_d}{C_d + C_{ds}/\ell_d}},$$
 11.71

and the propagation constant can be simplified using the small loss approximation as

$$\gamma_{d} = \alpha_{d} + j\beta_{d} = \sqrt{ZY} = \sqrt{j\omega L_{d}} \left[\frac{1}{R_{ds}\ell_{d}} + j\omega \left(C_{d} + C_{ds}/\ell_{d} \right) \right]$$

$$\approx \frac{Z_{d}}{2R_{ds}\ell_{d}} + j\omega \sqrt{L_{d} \left(C_{d} + C_{ds}/\ell_{d} \right)}.$$
11.72

For an incident input voltage, V_i , the voltage on the gate-to-source capacitance of the *n*th FET can be written as

$$V_{cn} = V_i e^{-(n-1)\gamma_g \ell_g} \left(\frac{1}{1 + j\omega R_i C_{gs}} \right).$$
 11.73

for a phase reference at the first FET. The factor in parentheses in (11.73) accounts for voltage division between R_i and C_{gs} ; for typical FET parameters $\omega R_i C_{gs} \ll 1$, so this factor can be approximated as unity over the bandwidth of the amplifier. The output current on the drain line can be found by recognizing that each current generator contributes waves of the form $-\frac{1}{2}I_{dn}e^{\pm\gamma_d z}$ in each direction. Since $I_{dn} = g_m V_{cn}$, the total output current at the Nth terminal of the drain line is

$$I_{o} = -\frac{1}{2} \sum_{n=1}^{n} I_{dn} e^{-(N-n)\gamma_{d}\ell_{d}} = -\frac{g_{m}V_{i}}{2} e^{-N\gamma_{d}\ell_{d}} e^{\gamma_{g}\ell_{g}} \sum_{n=1}^{N} e^{-n(\gamma_{g}\ell_{g} - \gamma_{d}\ell_{d})}, \qquad 11.74$$

The terms in the summation will add in phase only when $\beta_g \ell_g = \beta_d \ell_d$, so that the phase delays on the gate and drain lines are synchronized. There is also a backward traveling wave component on the drain line, but the individual contributions to this wave will not be in phase; the residual will be absorbed in the termination Z_d . Using the summation formula that

$$\sum_{n=1}^{N} x^n = \frac{x^{N+1} - x}{x - 1}$$

allows (11.74) to be simplified as follows:

$$I_{o} = -\frac{g_{m}V_{i}}{2} \frac{e^{\gamma_{d}\ell_{d}} \left[e^{-N\gamma_{g}\ell_{g}} - e^{-N\gamma_{d}\ell_{d}}\right]}{e^{-(\gamma_{g}\ell_{g} - \gamma_{d}\ell_{d})} - 1} = -\frac{g_{m}V_{i}}{2} \frac{e^{-N\gamma_{g}\ell_{g}} - e^{-N\gamma_{d}\ell_{d}}}{e^{-\gamma_{g}\ell_{g}} - e^{-\gamma_{d}\ell_{d}}}.$$
 11.75

For matched input and output ports, the amplifier gain can then be calculated as

$$G = \frac{P_{\text{out}}}{P_{\text{in}}} = \frac{\frac{1}{2} |I_o|^2 Z_d}{\frac{1}{2} |V_i|^2 / Z_g} = \frac{g_m^2 Z_d Z_g}{4} \left| \frac{e^{-N\gamma_g \ell_g} - e^{-N\gamma_d \ell_d}}{e^{-\gamma_g \ell_g} - e^{-\gamma_d \ell_d}} \right|^2.$$
 11.76

Applying the synchronization condition that $\beta_g \ell_g = \beta_d \ell_d$ allows this result to be further simplified to

$$G = \frac{g_m^2 Z_d Z_g}{4} \frac{\left(e^{-N\alpha_g \ell_g} - e^{-N\alpha_d \ell_d}\right)^2}{\left(e^{-\alpha_g \ell_g} - e^{-\alpha_d \ell_d}\right)^2}.$$
 11.77

If the losses are small, the denominator in (11.77) can be approximated as $(\alpha_g \ell_g - \alpha_d \ell_d)$.

Several interesting aspects of the distributed amplifier can be deduced from the gain expression of (11.77). For the ideal case of a lossless amplifier, the gain reduces to

$$G \approx \frac{g_m^2 Z_d Z_g N^2}{4},$$

showing that gain increases as N^2 . This is in contrast to the gain of a cascade of N amplifier stages, which increases as $(G_0)^N$. When loss is present, (11.77) shows that the gain of a distributed amplifier approaches zero as $N \to \infty$. This surprising behavior is explained by the fact that the input voltage on the gate line decays exponentially, so the FETs at the end of the amplifier receive no input signal; similarly, the amplified signals from the FETs near the beginning of the amplifier are attenuated along the drain line. The multiplicative increase in gain with N is not enough to compensate for an exponential decay for large N. This implies that, for a given set of FET parameters, there will be an optimum value of N that maximizes the gain of a distributed amplifier. This can be found by differentiating (11.77) with respect to N, and setting the result to zero to obtain

$$N_{\rm opt} = \frac{\ln\left(\alpha_g \ell_g / \alpha_d \ell_d\right)}{\alpha_g \ell_g - \alpha_d \ell_d}.$$
11.78

This result depends on frequency, the parameters of the FETs, and the line lengths through the attenuation constants given in (11.69) and (11.72).

EXAMPLE 11.7 Distributed Amplifier Performance

Use (11.77) to calculate the gain of a distributed amplifier from 1 to 18 GHz, for N = 2, 4, 8, and 16 stages. Assume $Z_d = Z_g = Z_0 = 50 \Omega$, and the following FET parameters: $R_i = 10 \Omega$, $R_{ds} = 300 \Omega$, $C_{gs} = 0.27$ pF, and $g_m = 35$ mS. Find the optimum value of N that will give maximum gain at 16 GHz.

Solution

We use (11.69) and (11.72) to evaluate the attenuation constants α_g and α_d , and then compute the gain versus frequency and N using (11.77). Note that the products $\alpha_g \ell_g$ and $\alpha_d \ell_d$ are independent of ℓ_g and ℓ_d :

$$lpha_g\ell_g=rac{\omega^2R_iC_{gs}^2Z_0}{2} lpha_d\ell_d=rac{Z_0}{2R_{ds}}$$

The results are shown in Figure 11.21. Observe that the gain drops off with frequency faster for larger N, and that at high frequencies the gain for N = 16 is less than the gain for smaller N. The optimum size for maximum gain at 16 GHz can be calculated using (11.78). At 16 GHz we have $\alpha_g \ell_g = 0.184$ and $\alpha_d \ell_d = 0.083$. The optimum size is then

$$N_{\rm opt} = \frac{\ln \left(\alpha_g \ell_g / \alpha_d \ell_d \right)}{\alpha_g \ell_g - \alpha_d \ell_d} = \frac{\ln \left(0.184 / 0.083 \right)}{0.184 - 0.083} = 7.9,$$

or eight stages. Finally, note that $\omega R_i C_{gs} = 0.31$ at 18 GHz, justifying the approximation of unity for the voltage divider factor of (11.73).



FIGURE 11.21 Gain versus frequency for the distributed amplifier of Example 11.7.

11.5 __ OSCILLATOR DESIGN

A microwave oscillator converts DC power to RF power, and so is one of the most basic and essential components in a microwave system. A solid-state oscillator uses an active device, such as a diode or transistor, in conjunction with a passive circuit to produce a sinusoidal steady-state RF signal. At startup, however, oscillation is triggered by transients or noise, after which a properly designed oscillator will reach a stable oscillation state. This process requires that the active device be nonlinear. In addition, since the device is producing RF power, it must have a negative resistance. Because of this active and nonlinear element, the complete analysis of oscillator operation is very difficult.

We will first discuss the operation and design of one-port negative resistance oscillators. Such circuits represent oscillators that use IMPATT or Gunn diodes. Then we treat transistor oscillators, where either an FET or bipolar device is operated with a passive termination that produces a negative resistance at its input port. That is, the transistor is operated in an unstable region, in contrast to an amplifier application where a stable operating point is required. Finally, we discuss the design of dielectric resonator oscillators.

One-Port Negative Resistance Oscillators

Here we present some of the basic principles of operation for one-port negative oscillators; much of this material will also apply to two-port (transistor) oscillators.



FIGURE 11.22 Circuit for a one-port negative-resistance oscillator.

Figure 11.22 shows the canonical RF circuit for a one-port negative-resistance oscillator, where $Z_{in} = R_{in} + jX_{in}$ is the input impedance of the active device (e.g., a biased diode). In general, this impedance is current (or voltage) dependent, as well as frequency dependent, which we can indicate by writing $Z_{in}(I, j\omega) = R_{in}(I, j\omega) + jX_{in}(I, j\omega)$. The device is terminated with a passive load impedance, $Z_L = R_L + jX_L$. Applying Kirchhoff's voltage law gives

$$(Z_L + Z_{\rm in})I = 0. 11.79$$

If oscillation is occurring, such that the RF current I is nonzero, then the following conditions must be satisfied:

$$R_L + R_{\rm in} = 0,$$
 11.80a

$$X_L + X_{\rm in} = 0. 11.80b$$

Since the load is passive, $R_L > 0$ and (11.80a) indicates that $R_{in} < 0$. Thus, while a positive resistance implies energy dissipation, a negative resistance implies an energy source. The condition of (11.80b) controls the frequency of oscillation. The condition in (11.79), that $Z_L = -Z_{in}$ for steady-state oscillation, implies that the reflection coefficients Γ_L and Γ_{in} are related as

$$\Gamma_L = \frac{Z_L - Z_0}{Z_L + Z_0} = \frac{-Z_{\rm in} - Z_0}{-Z_{\rm in} + Z_0} = \frac{Z_{\rm in} + Z_0}{Z_{\rm in} - Z_0} = \frac{1}{\Gamma_{\rm in}}.$$
11.83

The process of oscillation depends on the nonlinear behavior of Z_{in} , as follows. Initially, it is necessary for the overall circuit to be unstable at a certain frequency, that is, $R_{in}(I, j\omega) + R_L < 0$. Then any transient excitation or noise will cause an oscillation to build up at the frequency, ω . As I increases, $R_{in}(I, j\omega)$ must become less negative until the current I_0 is reached such that $R_{in}(I_0, j\omega_0) + R_L = 0$, and $X_{in}(I_0, j\omega_0) + X_L(j\omega_0) = 0$. Then the oscillator is running in a stable state. The final frequency, ω_0 , generally differs from the startup frequency because X_{in} is current dependent, so that $X_{in}(I, j\omega) \neq X_{in}(I_0, j\omega_0)$.

Thus we see that the conditions of (11.80) are not enough to guarantee a stable state of oscillation. In particular, stability requires that any perturbation in current or

frequency will be damped out, allowing the oscillator to return to its original state. This condition can be quantified by considering the effect of a small change, δI , in the current and a small change, δs , in the complex frequency $s = \alpha + j\omega$. If we let $Z_T(I,s) = Z_{in}(I,s) + Z_L(s)$, then we can write a Taylor series for $Z_T(I,s)$ about the operating point I_0, ω_0 as

$$Z_T(I,s) = Z_T(I_0,s_0) + \frac{\partial Z_T}{\partial s} \bigg|_{s_0,I_0} \delta s + \frac{\partial Z_T}{\partial I} \bigg|_{s_0,I_0} \delta I = 0, \qquad 11.82$$

since $Z_T(I, s)$ must still equal zero if oscillation is occurring. In (11.82), $s_0 = j\omega_0$ is the complex frequency at the original operating point. Now use the fact that $Z_T(I_0, s_0) = 0$, and that $\frac{\partial Z_T}{\partial s} = -j \frac{\partial Z_T}{\partial \omega}$, to solve (11.82) for $\delta s = \delta \alpha + j \delta \omega$:

$$\delta s = \delta \alpha + j \delta \omega = \frac{-\partial Z_T / \partial I}{\partial Z_T / \partial s} \bigg|_{s_0, I_0} \delta I = \frac{-j (\partial Z_T / \partial I) (\partial Z_T^* / \partial \omega)}{\left| \partial Z_T / \partial \omega \right|^2} \delta I.$$
 11.83

Now if the transient caused by δI and $\delta \omega$ is to decay, we must have $\delta \alpha < 0$ when $\delta I > 0$. Equation (11.83) then implies that

$$I_{m}\left\{\frac{\partial Z_{T}}{\partial I}\frac{\partial Z_{T}^{*}}{\partial \omega}\right\} < 0,$$

$$\frac{\partial R_{T}}{\partial I}\frac{\partial X_{T}}{\partial \omega} - \frac{\partial X_{T}}{\partial I}\frac{\partial R_{T}}{\partial \omega} > 0.$$
 11.84

OT

For a passive load, $\partial R_L / \partial I = \partial X_L / \partial I = \partial R_L / \partial \omega = 0$, so (11.84) reduces to

$$\frac{\partial R_{\rm in}}{\partial I} \frac{\partial}{\partial \omega} (X_L + X_{\rm in}) - \frac{\partial X_{\rm in}}{\partial I} \frac{\partial R_{\rm in}}{\partial \omega} > 0.$$
 11.85

As discussed above, we usually have that $\partial R_{in}/\partial l > 0$ [3]. So (11.85) can be satisfied if $\partial (X_L + X_{in})/\partial \omega \gg 0$, which implies that a high-Q circuit will result in maximum oscillator stability. Cavity and dielectric resonators are often used for this purpose.

Effective oscillator design requires the consideration of several other issues, such as the selection of an operating point for stable operation and maximum power output, frequency-pulling, large-signal effects, and noise characteristics. But we must leave these topics to more advanced texts [1].





FIGURE 11.23 Load matching circuit for the one-port oscillator of Example 11.8.

Solution

From either the Smith chart (see Problem 11.20), or by direct calculation, we find the input impedance as

$$Z_{\rm in} = -44 + j123 \ \Omega.$$

Then, by (11.80), the load impedance must be

$$Z_L = 44 - j123 \ \Omega.$$

A shunt stub and series section of line can be used to convert 50 Ω to Z_L , as shown in the circuit of Figure 11.23.

Transistor Oscillators

In a transistor oscillator, a negative-resistance one-port is effectively created by terminating a potentially unstable transistor with an impedance designed to drive the device in an unstable region. The circuit model is shown in Figure 11.24; the actual power



FIGURE 11.24 Circuit for a two-port transistor oscillator.

output port can be on either side of the transistor. In the case of an amplifier, we preferred a device with a high degree of stability—ideally, an unconditionally stable device. For an oscillator, we require a device with a high degree of instability. Typically, common source or common gate FET configurations are used (common emitter or common base for bipolar devices), often with positive feedback to enhance the instability of the device. After the transistor configuration is selected, the output stability circle can be drawn in the Γ_T plane, and Γ_T selected to produce a large value of negative resistance at the input to the transistor. Then the load impedance Z_L can be chosen to match Z_{in} . Because such a design uses the small-signal S parameters, and because R_{in} will become less negative as the oscillator power builds up, it is necessary to choose R_L so that $R_L + R_{in} < 0$. Otherwise, oscillation will cease when the increasing power increases R_{in} to the point where $R_L + R_{in} > 0$. In practice, a value of,

$$R_L = \frac{-R_{\rm in}}{3}, \qquad \qquad 11.86a$$

is typically used. The reactive part of Z_L is chosen to resonate the circuit,

$$X_L = -X_{\rm m}.$$
 11.86b

When oscillation occurs between the load network and the transistor, oscillation will simultaneously occur at the output port, which we can show as follows. For steady-state oscillation at the input port, we must have $\Gamma_L\Gamma_{in} = 1$, as derived in (11.81). Then from (11.6a) (after replacing Γ_L with Γ_T), we have

$$\frac{1}{\Gamma_L} = \Gamma_{\rm m} = S_{11} + \frac{S_{12}S_{21}\Gamma_T}{1 - S_{22}\Gamma_T} = \frac{S_{11} - \Delta\Gamma_T}{1 - S_{22}\Gamma_T},$$
11.87

where $\Delta = S_{11}S_{22} - S_{12}S_{21}$. Solving for Γ_T gives.

$$\Gamma_T = \frac{1 - S_{11} \Gamma_L}{S_{22} - \Delta \Gamma_L}.$$
 11.88

Then from (11.6b) we have that

$$\Gamma_{\text{out}} = S_{22} + \frac{S_{12}S_{21}\Gamma_L}{1 - S_{11}\Gamma_L} = \frac{S_{22} - \Delta\Gamma_L}{1 - S_{11}\Gamma_L},$$
11.89

which shows that $\Gamma_T \Gamma_{out} = 1$, and hence $Z_T = -Z_{out}$. Thus, the condition for oscillation of the terminating network is satisfied. Note that the appropriate S parameters to use in the above development are generally the large-signal parameters of the transistor.



EXAMPLE 11.9 Transistor Oscillator Design

Design a transistor oscillator at 4 GHz using a GaAs FET in a common gate configuration, with a 5 nH inductor in series with the gate to increase the instability. Choose a terminating network to match to a 50 Ω load, and an appropriate tuning network. The S parameters of the transistor in a common source configuration are $(Z_0 = 50 \Omega)$: $S_{11} = 0.72 / -116^\circ$, $S_{21} = 2.60 / 76^\circ$, $S_{12} = 0.03 / 57^\circ$, $S_{22} = 0.73 / -54^\circ$.

Solution

The first step is to convert the common source S parameters to the S parameters that apply to the transistor in a common gate configuration with a series inductor. (See Figure 11.25a.) This is most easily done using a microwave CAD package. The new S parameters are

$$S'_{11} = 2.18 \angle -35^{\circ},$$

$$S'_{21} = 2.75 \angle 96^{\circ},$$

$$S'_{12} = 1.26 \angle 18^{\circ},$$

$$S'_{22} = 0.52 \angle 155^{\circ}.$$

Note that $|S'_{11}|$ is significantly greater than $|S_{11}|$, which suggests that the configuration of Figure 11.25a is more unstable than the common source configuration. Calculating the output stability circle (Γ_T plane) parameters from (11.28) gives

$$C_T = \frac{(S'_{22} - \Delta' S'^{4}_{11})^*}{|S'_{22}|^2 - |\Delta'|^2} = 1.08 \underline{\beta}3^\circ,$$
$$R_T = \left|\frac{S'_{12}S'_{21}}{|S'_{22}|^2 - |\Delta'|^2}\right| = 0.665.$$

Since $|S'_{11}| = 2.18 > 1$, the stable region is inside this circle, as shown in the Smith chart in Figure 11.25b.

There is a great amount of freedom in our choice for Γ_T , but one objective is to make $|\Gamma_{in}|$ large. Thus we try several values of Γ_T located on the opposite side of the chart from the stability circle, and select $\Gamma_T = 0.59 \angle -104^\circ$. Then we can design a single-stub matching network to convert a 50 Ω load to $Z_T =$ $20 - j35 \Omega$, as shown in Figure 11.25a.

For the given value of Γ_T , we calculate Γ_{in} as,

$$\Gamma_{\rm in} = S_{11}' + \frac{S_{12}'S_{21}'\Gamma_T}{1 - S_{22}'\Gamma_T} = 3.96\angle -2.4^\circ,$$

or $Z_{in} = -84 - j1.9 \Omega$. Then, from (11.86), we find Z_L as

$$Z_L = rac{-R_{
m in}}{3} - jX_{
m in} = 28 + j1.9 \; \Omega.$$

Using $R_{\rm in}/3$ should ensure enough instability for the startup of oscillation. The easiest way to implement the impedance Z_L is to use a 90 Ω load with a short length of line, as shown in the figure. It is likely that the steady-state oscillation frequency will differ from 4 GHz because of the nonlinearity of the transistor parameters.



FIGURE 11.25 Circuit design for the transistor oscillator of Example 11.9. (a) Oscillator circuit, (b) Smith chart for determining Γ_T .

Dielectric Resonator Oscillators

As we saw from the result of (11.85), oscillator stability is enhanced with the use of a high-Q tuning network. The Q of a resonant network using lumped elements or microstrip lines and stubs is typically limited to a few hundred (see Examples 6.1 and 6.2), and while waveguide cavity resonators can have Qs of 10^4 or more, they are not wellsuited for integration in miniature microwave integrated circuitry. Another disadvantage of metal cavities is the significant frequency drift caused by dimensional expansion due to a variation in temperature. The dielectric cavity resonator discussed in Section 6.5 overcomes most of these disadvantages, as it can have an unloaded Q as high as several thousand, is compact and easily integrated with planar circuitry, and can be made from ceramic materials that have excellent temperature stability. For these reasons, transistor dielectric resonator oscillators (DROs) are becoming increasingly common over the entire microwave and millimeter wave frequency range.

A dielectric resonator is usually coupled to an oscillator circuit by positioning it in close proximity to a microstrip line, as shown in Figure (1.26a. The resonator operates in the TE_{01b} mode, and couples to the fringing magnetic field of the microstrip line. The strength of coupling is determined by the spacing, d, between the resonator and microstrip line. Because coupling is via the magnetic field, the resonator appears as a series load on the microstrip line, as shown in the equivalent circuit of Figure 11.26b. The resonator is modeled as a parallel *RLC* circuit, and the coupling to the feedline is modeled by the turns ratio, N, of the transformer. Using the result of (6.19) for the impedance of a parallel *RLC* resonator, we can express the equivalent series impedance, Z, seen by the microstrip line as

$$Z = \frac{N^2 R}{1 + j2Q\Delta\omega/\omega_0},$$
 11.90

where $Q = R/\omega_0 L$ is the unloaded resonator Q, $\omega_0 = 1/\sqrt{LC}$ is the resonant frequency, and $\Delta \omega = \omega - \omega_0$. The coupling factor, defined in (6.83), between the resonator and the





feedline is the ratio of the unloaded to external Q, and can be found as

$$g = \frac{\hat{Q}}{Q_e} = \frac{R/\omega_0 L}{R_L/N^2 \omega_0 L} = \frac{N^2 R}{2Z_0},$$
 11.91

where $R_L = 2Z_0$ is the load resistance for a feedline with source and termination resistances Z_0 . In some cases the feedline is terminated with an open-circuit $\lambda/4$ from the resonator to maximize the magnetic field at that point; in this case $R_L = Z_0$ and the coupling factor is twice the value given in (11.91).

The reflection coefficient seen on the terminated microstrip line looking toward the resonator can be written as

$$\Gamma = \frac{(Z_0 + N^2 R) - Z_0}{(Z_0 + N^2 R) + Z_0} = \frac{N^2 R}{2Z_0 + N^2 R} = \frac{g}{1 + g}.$$
 11.92

This allows the coupling coefficient to be found from $g = \Gamma/(1 - \Gamma)$ after the simple procedure of measuring Γ at resonance; the resonant frequency and Q can also be found by measurement. Alternatively, these quantities can be calculated using approximate analytical solutions [9]. Note that this procedure leaves a degree of freedom between Nand R, since only the product N^2R is uniquely determined.

There are many oscillator configurations using common source (emitter), common gate (base), or common drain (collector) connections of either FET or bipolar transistors, in addition to the optional use of series or shunt elements to increase the instability of the device [1]-[3]. A dielectric resonator can be incorporated into the circuit to provide frequency stability using either the parallel feedback arrangement of Figure 11.27a, or the series feedback technique shown in Figure 11.27b. The parallel configuration uses a resonator coupled to two microstrip lines, functioning as a high-Q bandpass filter that couples a portion of the transistor output back to its input. The amount of coupling is controlled by the spacing between the resonator and the lines, and the phase is controlled



FIGURE 11.27 (a) Dielectric resonator oscillator using parallel feedback: (b) dielectric resonator oscillator using series feedback.

by the length of the lines. The series feedback configuration is simpler, using only a single microstrip feedline, but typically does not have a tuning range as wide as that obtained with parallel feedback. Design of an oscillator using parallel feedback is most conveniently done using a microwave CAD package, but a dielectric resonator oscillator using series feedback can be designed using the same procedure that was discussed in the previous section on two-port oscillators.



EXAMPLE 11.10 Dielectric Resonator Oscillator Design

A wireless local area network application requires a local oscillator operating at 2.4 GHz. Design a dielectric resonator oscillator using the series feedback circuit of Figure 11.27b with a bipolar transistor having the following S parameters ($Z_0 = 50 \Omega$): $S_{11} = 1.8 / 130^\circ$, $S_{12} = 0.4 / 45^\circ$, $S_{21} = 3.8 / 36^\circ$, $S_{22} = 0.7 / -63^\circ$. Determine the required coupling coefficient for the dielectric resonator, and a microstrip matching network for the termination network. The termination network should include the output load impedance. Plot the magnitude of Γ_{out} versus $\Delta f / f_0$, for small variations in frequency about the design value, assuming an unloaded resonator Q of 1000.

Solution

The DRO circuit is shown in Figure 11.28a. The dielectric resonator is placed $\lambda/4$ from the open end of the microstrip line; the line length ℓ_r can be adjusted to match the phase of the required value of Γ_L . In contrast to the oscillator of the previous example, the output load impedance for this circuit is part of the terminating network.

The stability circles for the load and termination sides of the transistor can be plotted if desired, but are not necessary to the design, since we will begin by choosing Γ_L to provide a large value of $|\Gamma_{out}|$. From (11.89) we have

$$\Gamma_{\rm out} = S_{22} + \frac{S_{12}S_{21}\Gamma_L}{1 - S_{11}\Gamma_L},$$



FIGURE 11.28 (a) Circuit for the dielectric resonator of Example 11.10.

which indicates that we can maximize Γ_{out} by making $1 - S_{11}\Gamma_L$ close to zero. Thus we choose $\Gamma_L = 0.6 - 130^\circ$, which gives $\Gamma_{out} = 10.7 - 132^\circ$. This corresponds to an impedance

$$Z_{\text{out}} = Z_0 \frac{1 + \Gamma_{\text{out}}}{1 - \Gamma_{\text{out}}} = 50 \frac{1 + 10.7 \cancel{132}^{\circ}}{1 - 10.7 \cancel{132}^{\circ}} = -43.7 + j6.1 \ \Omega.$$

Applying the analogous startup condition of (11.86) for the termination side gives the required termination impedance as

$$Z_T = rac{-R_{
m out}}{3} - jX_{
m out} = 5.5 - j6.1 \ \Omega.$$

The termination matching network can now be designed using a Smith chart. The shortest transmission line length for matching Z_T to the load impedance Z_0 is $\ell_t = 0.481\lambda$, and the required open-circuit stub length is $\ell_s = 0.307\lambda$.

Next we match Γ_L to the resonator network. From (11.90) we know that the equivalent impedance of the resonator seen by the microstrip line is real at the resonant frequency, so the phase angle of the reflection coefficient at this point, Γ'_L , must be either zero or 180°. For an undercoupled parallel *RLC* resonator, $R < Z_0$, so the proper phase will be 180°, which can be achieved by transformation through the line length ℓ_r . The magnitude of the reflection coefficient is unchanged, so we have the relation

$$\Gamma_L' = \Gamma_L e^{2j\beta\ell_r} = \left(0.6 \angle 130^\circ\right) e^{2j\beta\ell_r} = 0.6 \angle 180^\circ.$$

which gives $\ell_r = 0.431\lambda$. The equivalent impedance of the resonator at resonance is then

$$Z'_L = Z_0 \frac{1 + \Gamma'_L}{1 - \Gamma'_L} = 12.5 \,\Omega.$$

The coupling coefficient can be found using (11.91), with a factor of two to account for the $\lambda/4$ stub termination, as

$$g = \frac{N^2 R}{Z_0} = \frac{12.5}{50} = 0.25.$$

The variation of $|\Gamma_{out}|$ with frequency will give an indication of the frequency stability of the oscillator. We can calculate Γ_{out} from (11.89), after first using (11.90) to compute Z'_L , Γ'_L , and then transforming down the line of length ℓ_r to obtain Γ_L . The electrical line length can be approximated as constant for the small changes in frequency associated with this calculation. A short computer program or a microwave CAD package can be used to generate data for $-0.01 < \Delta f/f_0 < 0.01$, which is shown in the graph of Figure 11.28b. Observe that $|\Gamma_{out}|$ decreases rapidly with a change in frequency as small as a few hundredths of a percent, demonstrating the sharp selectivity that can be obtained with a dielectric resonator.



FIGURE 11.28 Continued. (b) $|\Gamma_{out}|$ vs. frequency in Example 11.10.

REFERENCES

- G. D. Vendelin, A. M. Pavio, and U. L. Rohde, Microwave Circuit Design Using Linear and Nonlinear Techniques, Wiley, N. Y., 1990.
- [2] G. Gonzalez, Microwave Transistor Amplifiers: Analysis and Design, Prentice-Hall, N. J., 1984.
- [3] I. Bahl and P. Bhartia, Microwave Solid-State Circuit Design, Wiley Interscience, N. Y., 1988.
- [4] T. T. Ha, Solid-State Microwave Amplifier Design, Wiley, N. Y., 1981.
- [5] C. Gentile. Microwave Amplifiers and Oscillators, McGraw-Hill, N. Y., 1987.
- [6] M. L. Edwards and J. H. Sinksy. "A New Criteria for Linear 2-Port Stability Using a Single Geometrically Derived Parameter." *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-40, pp. 2803–2811, December 1992.
- [7] Y. Ayasli, R. L. Mozzi, J. L. Vorhous, L. D. Reynolds, and R. A. Pucel, "A Monolithic GaAs 1-13 GHz Traveling-Wave Amplifier," *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-30, pp. 976–981, July 1982.
- [8] J. B. Beyer, S. N. Prasad, R. C. Becker, J. E. Nordman, and G. K. Hohenwarter, "MESFET Distributed Amplifier Design Guidelines," *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-32, pp. 268–275, March 1984.
- [9] Y. Komatsu and Y. Murakami, "Coupling Coefficient Between Microstrip Line and Dielectric Resonator," IEEE Trans. Microwave Theory and Techniques, vol. MTT-31. pp. 34-40, January 1983.

PROBLEMS

11.1 Use the equivalent circuit of Figure 11.5 to derive the expression for the short-circuit current gain of a bipolar transistor. Assume a unilateral device, where $C_e = 0$.

- 11.2 Derive expressions for the y parameters of an FET using the unilateral equivalent circuit model. Evaluate these parameters at 5 GHz for the following FET characteristics: $R_i = 7 \Omega$, $R_{ds} = 400 \Omega$, $C_{gs} = 0.3 \text{ pF}$, $C_{ds} = 0.12 \text{ pF}$, $C_{gd} = 0$, $g_m = 30 \text{ mS}$. Convert the y parameters to S parameters for a 50 Ω system impedance, and find the unilateral transducer gain assuming conjugately matched source and load impedances. Compare with the value computed using (11.21).
- 11.3 Consider the microwave network shown below, consisting of a 50 Ω source, a 50 Ω , 3 dB matched attenuator, and a 50 Ω load. Compute the available power gain, the transducer power gain, and the actual power gain. How do these gains change if the load is changed to 25 Ω ? How do these gains change if the source impedance is changed to 25 Ω ?



- **11.4** A microwave transistor has the following S parameters: $S_{11} = 0.34 \angle -170^\circ$, $S_{21} = 4.3 \angle 80^\circ$, $S_{12} = 0.06 \angle 70^\circ$, and $S_{22} = 0.45 \angle -25^\circ$. Determine the stability, and plot the stability circles if the device is potentially unstable.
- **11.5** Repeat Problem 11.4 for the following transistor S parameters: $S_{11} = 0.8 \angle -90^\circ$, $S_{21} = 5.1 \angle 80^\circ$, $S_{12} = 0.3 \angle 70^\circ$, and $S_{22} = 0.62 \angle -40^\circ$.
- 11.6 Use the μ -parameter test to determine which of the following devices are unconditionally stable, and of those, which has the greatest stability:

Device	S_{11}	$S_{1\bar{z}}$	S ₂₁	S ₂₂
A	$\begin{array}{r} 0.34 \underline{/-170}^{\circ} \\ 0.75 \underline{/-60}^{\circ} \\ 0.65 \underline{/-140}^{\circ} \end{array}$	0.06 <u>/70</u> *	4.3 <u>/80</u> °	0.45∠ <u>-25</u> °
19		0.2 <u>/70</u> °	5.0 <u>/90</u> °	0.51 <u>∕60</u> °
C		0.04 <u>/60</u> °	2.4 <u>/50</u> °	0.70 <u>∕-65</u> °

- 11.7 Show that for a unilateral device, where $S_{12} = 0$, the μ -parameter test of (11.38) implies that $|S_{11}| < 1$ and $|S_{22}| < 1$ for unconditional stability.
- 11.8 Design an amplifier for maximum gain at 5.0 GHz with a GaAs FET that has the following S parameters ($Z_0 = 50 \Omega$): $S_{11} = 0.65 \angle -140^\circ$, $S_{21} = 2.4 \angle 50^\circ$, $S_{12} = 0.04 \angle 60^\circ$, $S_{22} = 0.70 \angle -65^\circ$. Design matching sections using open-circuited shunt stubs.
- 11.9 Design an amplifier with maximum G_{TU} using a transistor with the following S parameters ($Z_0 = 50 \Omega$) at 6.0 GHz: $S_{11} = 0.61 \pm -170^\circ$, $S_{21} = 2.24 \pm 32^\circ$. $S_{12} = 0.5_{22} = 0.72 \pm -83^\circ$. Design L-section matching sections using lumped elements.
- 11.10 Design an amplifier to have a gain of 10 dB at 6.0 GHz, using a transistor with the following S parameters ($Z_0 = 50 \ \Omega$): $S_{11} = 0.61 \ \angle -170^\circ$, $S_{21} = 2.24 \ \angle 32^\circ$, $S_{12} = 0$, $S_{22} = 0.72 \ \angle -83^\circ$. Plot (and use) constant gain circles for $G_S = 1$ dB and $G_L = 2$ dB. Use matching sections with open-circuited shunt stubs.
- 11.11 Compute the unilateral figure of merit for the transistor of Problem 11.4. What is the maximum error in the transducer gain if an amplifier is designed assuming the device is unilateral?
- 11.12 Show that the 0 dB gain circle for G_S ($G_S = 1$), defined by (11.52), will pass through the center of the Smith chart.

- 11.13 A GaAs FET has the following scattering and noise parameters at 8 GHz ($Z_0 = 50 \Omega$): $S_{11} = 0.7 \angle -110^{\circ}$, $S_{12} = 0.02 \angle 60^{\circ}$, $S_{21} = 3.5 \angle 60^{\circ}$, $S_{22} = 0.8 \angle -70^{\circ}$, $F_{min} = 2.5 \text{ dB}$, $\Gamma_{opt} = 0.70 \angle 120^{\circ}$, $R_N = 15 \Omega$. Design an amplifier with minimum noise figure, and maximum possible gain. Use open-circuited shunt stubs in the matching sections.
- 11.14 A GaAs FET has the following scattering and noise parameters at 6 GHz ($Z_0 = 50 \Omega$): $S_{11} = 0.6 \angle -60^{\circ}$, $S_{21} = 2.0 \angle 81^{\circ}$, $S_{12} = 0$, $S_{22} = 0.7 \angle -60^{\circ}$, $F_{\min} = 2.0$ dB, $\Gamma_{opt} = 0.62 \angle 100^{\circ}$, $R_N = 20 \Omega$. Design an amplifier to have a gain of 6 dB, and the minimum noise figure possible with this gain. Use open-circuited shunt stubs in the matching sections.
- **11.15** Repeat Problem 11.14, but design the amplifier for a noise figure of 2.5 dB, and the maximum possible gain that can be achieved with this noise figure.
- 11.16 Repeat the analysis of the balanced amplifier of Example 11.6 using a 3 dB coupled line hybrid coupler. Use CAD software to optimize the input and output matching networks of the amplifiers to obtain a flat 10 dB gain response from 3 to 5 GHz, and compare the results with those obtained using the quadrature hybrid.
- 11.17 If the individual amplifier stages in a balanced amplifier have mismatches of Γ_A and Γ_B at their output ports, show that the output mismatch of the balanced amplifier is $S_{22} = (\Gamma_A \Gamma_B)/2$.
- 11.18 Derive the result for the optimum size of a distributed amplifier given in (11.78).
- **11.19** Consider a distributed amplifier using FETs having the following parameters: $R_i = 5 \Omega$, $R_{ds} = 200 \Omega$, $C_{gs} = 0.35$ pF, and $g_m = 40$ mS. Calculate and plot the gain from 2 to 20 GHz, for N = 4 and N = 16 sections. Find the optimum value of N that will give maximum gain at 18 GHz.
- 11.20 Prove that the standard Smith chart can be used for negative resistances by plotting $1/\Gamma^*$ (instead of Γ). Then the resistance circle values are read as negative, while the reactance circles are unchanged.
- 11.21 Design a transistor oscillator at 6 GHz using an FET in a common source configuration driving a 50 Ω load on the drain side. The S parameters are $(Z_0 = 50 \ \Omega)$: $S_{11} = 0.9 \angle -150^\circ$, $S_{21} = 2.6 \angle 50^\circ$, $S_{12} = 0.2 \angle -15^\circ$, $S_{22} = 0.5 \angle -105^\circ$. Calculate and plot the output stability circle, and choose Γ_T for $|\Gamma_{10}| >> 1$. Design load and terminating networks.
- **11.22** For either the one-port negative resistance oscillator of Figure 11.23, or the two-port transistor oscillator of Figure 11.24, show that $\Gamma_L \Gamma_{10} = 1$ for steady-state oscillation.
- 11.23 Repeat the oscillator design of Example 11.10 by replacing the dielectric resonator and microstrip feedline with a single-stub tuner to match Γ_L to a 50 Ω load. Find the Q of the tuner and 50 Ω load, then compute and plot $|\Gamma_{out}|$ versus $\Delta f/f_0$. Compare with the result in Figure 11.28b for the dielectric resonator case.
- 11.24 Repeat the dielectric oscillator design of Example 11.10 using a GaAs FET having the following S parameters: $S_{11} = 1.2 \angle 150^{\circ}$, $S_{12} = 0.2 \angle 120^{\circ}$, $S_{21} = 3.7 \angle -72^{\circ}$, $S_{22} = 1.3 \angle -67^{\circ}$.

Introduction to Microwave Systems

A microwave system consists of passive and active microwave components arranged to perform a useful function. Probably the two most important examples are microwave radar systems and microwave communication systems, but there are many others. In this chapter we will describe the basic operation of several types of microwave systems to give a general overview of the application of microwave technology, and to show how many of the subjects of earlier chapters fit into the overall scheme of complete microwave systems.

An important component in any radar or communication system is the antenna, so we will discuss some of the basic properties of antennas in Section 12.1. Then communication, radar, and radiometry systems are treated as important applications of microwave technology. Propagation effects, biological effects, and other miscellaneous applications are also briefly discussed.

All of the above topics are of sufficient depth that many books have been written for each. Thus our purpose here is not to give a complete and thorough treatment of these subjects, but instead to introduce these topics as a way of placing the other material in this book in the larger systems context. The interested reader is referred to the references at the end of the chapter for more complete treatments.

12.1 SYSTEM ASPECTS OF ANTENNAS

In this section we will discuss some of the basic characteristics of antennas that will be needed for our study of radar and communications systems. In essence, an antenna is a component that converts a wave propagating on a transmission line to a plane wave propagating in free-space (transmission), or vice versa (reception). There is a wide variety of antenna types and geometries, but all antennas can be described with a set of parameters and terms, as discussed below.

Definition of Important Antenna Parameters

Radiation pattern. The power radiated (or received) by an antenna is a function of angular position and radial distance from the antenna. At electrically large distances (many wavelengths), r, the power density drops off as $1/r^2$ in any direction. The variation of power density with angular position is determined by the type and design of

the antenna, and can be graphically represented as a radiation pattern plot. Such a plot may take the form of a three-dimensional graph of power versus elevation and azimuth angles, but more commonly the radiation pattern is represented by principal plane pattern plots, such as the E-plane or H-plane where one angle is held fixed while the other is varied. The transmit and receive patterns of an antenna are identical if the antenna contains no nonreciprocal materials or components.

Far-field. This is the region away from an antenna where the radiated wave essentially takes the form of a plane wave. A commonly used criterion is $2D^2/\lambda$, where D is the maximum linear dimension of the antenna, and λ is the operating wavelength. Radiation patterns are generally assumed to be in the far-field of the antenna.

Directivity. Many antennas are used to transmit or receive power in a fixed direction, and so it may be desired to maximize the radiation pattern, or antenna response, in this direction. A quantitative measure of this response is the directive gain of the antenna, for a given direction. The maximum value of directive gain is the directivity of the antenna. The directivity depends solely on the shape of the radiation pattern.

Efficiency. Like other microwave components, an antenna may dissipate power due to conductor loss or dielectric loss, and so an antenna efficiency can be defined as the ratio of total power radiated by the antenna to the input power of the antenna.

Gain. Antenna gain is the product of efficiency and directivity, and accounts for the fact that loss reduces the power density radiated in a given direction.

Impedance. An antenna presents a driving-point impedance to the source or load to which it is connected, and so impedance mismatch with a feed line can occur. This mismatch degrades antenna performance, and is dependent on the external circuitry which is connected to the antenna.

Bandwidth. The usable frequency bandwidth of an antenna may be limited by impedance mismatch or pattern deterioration. Matching networks can sometimes be used to increase the impedance bandwidth of an antenna.

Polarization. The polarization of an antenna refers to the polarization of the electric field vector of the radiated wave. Typical polarizations include linear (vertical or horizontal), and circular (RHCP or LHCP). Some antennas are designed to operate with two polarizations.

Size/complexity. A basic characteristic of an antenna is that, for efficient operation, its size must be $\lambda/2$ or greater, and that higher gain can only be achieved with larger size. In addition, features such as pattern shaping, polarization control, or large bandwidth are generally obtained only with an increase in the complexity of the antenna.

Basic Types of Antennas

Because of the diversity implied in the above characteristics, and the fact that the operation of an antenna is very dependent on its geometry, a wide variety of antenna types and geometries have been developed. We summarize some of these below, and in Figure 12.1.



- FIGURE 12.1 Photograph of various millimeter wave antennas. Clockwise from top: a highgain 38 GHz reflector antenna with radome, a prime-focus parabolic antenna, a corrugated conical horn antenna, a 38 GHz planar microstrip array, a pyramidal horn antenna with a Gunn diode module, and a multibeam reflector antenna. Courtesy of H. Syrigos, Alpha Industries, Inc., Woburn. Mass.
 - Wire antennas are probably the simplest types of antennas, and most of the earliest antennas (e.g., as used by Hertz and Marconi) were of this type. Wire dipoles (and monopoles mounted on a ground plane) are most commonly used at lower frequencies (HF to UHF), and have relatively low gains. Wire antennas are easy to fabricate, easy to feed, and are lightweight.
 - Aperture antennas are often just flared sections of waveguide (a horn), or even open-ended waveguides. Slots in a waveguide or a ground plane are additional examples. Aperture antennas are most commonly used at microwave frequencies and have moderate gains.

- Printed antennas are a relatively new type of antenna consisting of printed conductors on a microstrip or similar type of substrate, and are thus compatible with planar microwave circuit technology. Printed antennas usually operate at microwave frequencies and have low gains.
- Reflector antennas achieve high gain by focusing the radiation from a small feed antenna with an electrically large reflector. Because of the large electrical size, reflectors usually operate at microwave frequencies. Reflectors are relatively easy to fabricate and are rugged, but can be large and unwieldly.

All of the above antennas are single-element antennas; another class of antenna is the array. An array antenna consists of a number of antenna elements (usually all of the same type) arranged in a uniform grid. By controlling the amplitude and phase of the excitation of each element, the radiation pattern of the array can be controlled in terms of sidelobe levels or main beam pointing position. A phased array antenna uses electronically controlled phase shifters to rapidly scan the main beam; such antennas are particularly useful for tracking radar applications.

Antenna Pattern Characteristics

Many of the important characteristics of an antenna are given solely in terms of its radiation pattern. The radiation pattern is a plot of the radiation intensity. $F(\theta, \phi)$, of the antenna, defined as

$$F(\theta,\phi) = r^2 |\bar{E}(\theta,\phi) \times \bar{H}^*(\theta,\phi)| = r^2 S(\theta,\phi), \qquad 12.1$$

where \overline{E} and \overline{H} are the electric and magnetic fields radiated to the far-field zone of the antenna, and S is the magnitude of the Poynting vector. Since the \overline{E} and \overline{H} fields will decay as 1/r, the r^2 factor in (12.1) effectively removes this range dependence.

Principal plane radiation patterns are made by taking orthogonal cuts through the main beam of the antenna. A typical example, shown in Figure 12.2, identifies the most important pattern features. Most antennas have a radiation pattern with a well-formed main beam; the direction of the peak of the main beam is referred to as the main beam direction, or scan angle. The sharpness of the main beam is quantified by the 3 dB beamwidth; this beamwidth may be different in the two principal planes. Next to the main beam are the sidelobes; the height of the largest sidelobe relative to the main beam is called the peak sidelobe level (or just sidelobe level), and is usually given in dB. The pattern plot is usually in dB. normalized to the peak of the main beam. A rectangular plot is shown in Figure 12.2, but polar plots are also commonly used.

An antenna having a radiation pattern with relatively narrow beamwidths in both principal planes is sometimes called a *pencil-beam antenna*. Similarly, an antenna having a narrow beamwidth in one plane and a much broader beamwidth in the orthogonal plane is referred to as a *fan-beam antenna*.

Like the beamwidth, the directivity is a measure of the sharpness of the main beam, or the extent to which the pattern is focussed in a given direction. It is defined as,

$$D = \frac{4\pi F_{\text{max}}}{P_{\text{rsd}}} = \frac{4\pi F_{\text{max}}}{\int_{\phi=0}^{2\pi} \int_{\theta=0}^{\pi} F(\theta, \phi) \sin \theta \, d\theta \, d\phi},$$
 12.2



FIGURE 12.2 A typical antenna radiation pattern.

where F_{max} is the maximum value of radiation intensity, F, and P_{rad} is the total power radiated by the antenna. An isotropic antenna is a hypothetical antenna having $F(\theta, \phi) =$ constant; from (12.2) the directivity of such an antenna would be unity, or 0 dB. No real antenna has an isotropic pattern, however, so directivity will always be greater than unity. Narrower beamwidth implies a larger directivity.

Since beamwidth and directivity are measures of the same thing, we might expect a relation between them. In fact, the relationship depends on the shape of the main beam, as well as the rest of the radiation pattern, but an approximate result that is useful for many cases of practical interest is given by

$$D \simeq \frac{32,400}{\theta_1 \theta_2},$$
 12.3

where θ_1 , θ_2 are the beamwidths, in degrees, of the main beam in the two principal planes.

If a transmitting antenna radiates a total power of P_{rad} with an isotropic pattern (equal power density in all directions), then the power density (Poynting vector magnitude) at a distance r from the antenna is given by P_{rad} divided by the surface area of the sphere through which all power passes: $S = P_{rad}/4\pi r^2$ W/m². If the antenna has a directivity D, then the power density in the direction of the main beam (the maximum power density) will be

$$S = \frac{P_{\rm rat}D}{4\pi\tau^2} \,\,\mathrm{W/m^2},$$
 12.4

since. by the definition of (12.2), the directivity can be considered as the ratio of the peak radiation intensity to the average radiation intensity. The result in (12.4) can also be derived by combining (12.1) and (12.2).

For receiving antennas, it is useful to define the effective area of the antenna as

$$A_e = \frac{P_\ell}{S}.$$
 12.5

where P_i is the power delivered to the load of the antenna (assumed lossless), and S is the incident power density. Since $P_f = A_c S$, the effective area can be thought of as the "capture area" that an antenna presents to an incident plane wave. It can be shown [1] that the following exact relation exists between the directivity and effective area of an antenna:

$$D = \frac{4\pi A_e}{\lambda^2}.$$
 12.6

This shows that the directivity is proportional to the effective area of an antenna. The effective area of an antenna may be greater or less than its physical cross-sectional area, but for many large aperture antennas it is useful to define an aperture efficiency, η_a , that relates the effective area to the physical area, A:

$$A_e = \eta_a A.$$
 12.7

Then, using (12.6), the directivity of such an antenna can be written in terms of its physical area as,

$$D = \frac{4\pi\eta_{\alpha}A}{\lambda^2}.$$
 12.8

Typical aperture efficiencies for large antennas range from 0.60 to 1.0, depending on the type of antenna, excitation, and other factors.



660

EXAMPLE 12.1 Characteristics of a Short Dipole Antenna

A short wire dipole of length ℓ , where $\ell \ll \lambda$, is oriented along the z-axis and has far-zone fields given by the following expressions:

$$E_{ heta} = rac{jk_0\eta_0I_0\ell}{4\pi r}\sin heta e^{-jk_0r},$$
 $H_{\phi} = rac{jk_0I_0\ell}{4\pi r}\sin heta e^{-jk_0r},$

where I_0 is the current on the dipole (constant along the length of the dipole). Use these results to compute the radiation intensity, F, the directivity, and the effective area of the dipole.

Solution

First observe that $E_{\theta} = \eta_0 H_{\phi}$, which is characteristic of plane wave propagation. From (12.1) the radiation intensity is given as

$$F(\theta,\phi) = r^2 |\bar{E} \times \bar{H}^*| = \eta_{\Theta} \left(\frac{k_0 I_0 \ell}{4\pi}\right)^2 \sin^2 \theta.$$

The dipole thus has a beam maximum at $\theta = 90^{\circ}$, and the pattern is independent of ϕ . From (12.2) we compute the directivity:

$$P_{\rm rad} = C \int_{\phi=0}^{2\pi} \int_{\theta=0}^{\pi} \sin^3 \theta \, d\theta d\, \phi = \frac{8\pi C}{3}.$$

where C is a constant that will be canceled in the next step. Since $F_{\text{max}} = C$, we have

$$D = \frac{4\pi F_{\max}}{P_{\text{rad}}} = 1.5.$$

Using (12.6) gives the effective area of the dipole as

$$A_e = \frac{\lambda^2 D}{4\pi} = \frac{3\lambda^2}{8\pi}.$$

Table 12.1 lists the directivities for some common types of antennas.

Antenna Efficiency, Gain, and Temperature

The radiation efficiency of an antenna is defined as

$$\eta = rac{P_{
m rad}}{P_{
m iq}}, \quad 0 \leq \eta \leq 1$$
 , 12.9

where P_{rad} is the total power radiated by the antenna and P_{in} is the input power to the antenna. Thus, the radiation efficiency is a measure of how much power is lost in the antenna, to dielectric and conductor losses. These losses reduce the radiated power in any given direction, so in terms of the input power the power density in the direction of the main beam is

$$S = \frac{\eta P_{\rm in} D}{4\pi r^2} \, \mathrm{W/m^2},$$
 12.10

as derived from (12.4) and (12.9). Most antennas have relatively high radiation efficiencies, typically ranging from $\eta = 0.6$ to 0.95.

TABLE 12.1	Directivities of Common Antennas	

Antenna	Directivity
Short dipole	1.5
$\lambda/2$ dipole	1.6
$\lambda/4$ monopole	3.2
Rectangular aperture $(a \times b)$ (uniform excitation)	$4\pi ab/\lambda^2$
Rectangular aperture $(a \times b)$ (TE ₁₀ excitation)	$32ab/\pi\lambda^2$

Another factor which can reduce the effective radiated power is reflection loss due to impedance mismatch between the feed line and the antenna; the radiated power is reduced by a factor of $(1 - |\Gamma|^2)$, where Γ is the reflection coefficient between the line and the antenna. This loss factor is identical in effect to the radiation efficiency, but is not included in the radiation efficiency (or in the gain of the antenna) because it is dependent on the external circuitry connected to the antenna. while the radiation efficiency is a characteristic of the antenna itself.

While the directivity depends only on the shape of the radiation pattern, and not the efficiency, the gain includes the effect of loss. Thus

$$G = \eta D, \qquad 12.11$$

which shows that antenna gain never exceeds the directivity.

If the entire pattern of a lossless antenna $(\eta = 1)$ sees a background with a temperature T_b , the equivalent noise temperature of the antenna will be T_b . But if the antenna is so lossy that $\eta \simeq 0$, the (matched) antenna will look like a resistor at temperature T_p , where T_p is the physical temperature of the antenna, and so will have an equivalent noise temperature T_p . For other values of efficiency, the antenna will have an equivalent noise temperature given as the weighted average of these two results:

$$T_{\epsilon} = \eta T_b + (1 - \eta) T_p.$$
12.12

Thus the antenna of a receiver system picks up noise power from the background at which it is aimed, as well as the thermal noise resulting from the internal losses of the antenna. Noise power can be received from the main beam of the antenna, as well as all the sidelobes, albeit at a reduced level. The temperature T_b is the equivalent temperature of the background, and is referred to as the brightness temperature. If this temperature varies over the pattern of the antenna, an integration is required [1], [2] to determine the net temperature seen by the antenna. For example, the overhead sky has a brightness temperature of about 5 K, while near the horizon the brightness temperature can increase to 100-150 K.

12.2 MICROWAVE COMMUNICATION SYSTEMS

Microwave communication links are an important practical application of microwave technology, and are used to carry voice, data, or television signals over distances ranging from intercity links to deep-space spacecraft. In this section we will give a brief introduction to microwave communications systems, concentrating on microwave radio systems.

Types of Communication Systems

Microwave communication systems can be grouped into two types: guided-wave systems, where the signal is transmitted over a low-loss cable or waveguide, and radio links, where the signal propagates through space. In both cases the information-carrying signal will have a much lower bandwidth than the microwave carrier frequency. Some typical channel bandwidths are given in Table 12.2.

Using modulation and multiplexing techniques, a microwave link can carry a large number of individual channels; this is one of the principal advantages of a microwave

Туре	Bandwidth
Voice	4 kHz
Television	6 MHz
Digital voice (PCM)	64 kHz
Digital data	50 kHz-1,5 MHz

TABLE 12.2 Typical Communication Channel Bandwidths

communication link. For example, a nominal 4 GHz microwave carrier with a relatively modest 10% bandwidth can carry 100,000 voice channels, or 66 television channels.

Guided-wave communication channels may use coaxial line or waveguide, but the high attenuation of coax generally limits its application to frequencies below 1 GHz. At microwave and millimeter wave frequencies, less loss is obtained with circular waveguides. Ever since the 1930s, when waveguide was first being developed, the TE_{01} circular waveguide mode has created interest for communications systems because its attenuation decreases with frequency (see Section 3.4). The problem is that the TE_{01} mode is not the dominant mode of circular waveguide, so power can be coupled to undesired propagating modes; this spurious power effectively increases the attenuation of the guide. Consequently, very few microwave links using waveguide have been developed beyond the prototype stage, although recently there has been interest in millimeter wave waveguide links [3]. Instead, fiber optic cables (which are essentially waveguides at optical frequencies), are being beavily used for long-distance telephone traffic, as they offer low-loss, very high bandwidths, and a high degree of ruggedness.

Because of loss, the power level on any transmission line or waveguide (including optical fiber) decreases exponentially with distance. as $e^{-2\alpha z}$, where α is the attenuation constant of the line, and z is the distance from the source. As we saw in the previous section, however, the power radiated from an antenna falls off as $1/R^2$, where R is the distance from the antenna. Thus, in the absence of other effects (e.g., propagation loss, polarization mismatch, obstructions), a radio link will intrinsically have less path loss than a guided-wave link, for large distances. A radio link is also advantageous in that right-of-ways on land are not required, unlike guided-wave systems. Of course, radio links are mandatory for satellite or spacecraft applications.

Microwave radio propagation is essentially *line-of-sight*, meaning that microwave signals travel in straight lines, and do not follow the contour of the earth or reflect off the ionosphere, as do lower frequency signals. So for long-distance links on the ground, repeater stations are required at frequent intervals, to receive and retransmit the signals (usually in both directions). A communications satellite is a repeater also, connecting widely separated earth stations. Figure 12.3 shows a photograph of an antenna for the IRIDIUM satellite system.

The Friis Power Transmission Formula

A basic microwave radio link is shown in Figure 12.4, where P_t is the transmitter power, G_t and G_r are the transmit and receive antenna gains, and R is the distance between the antennas. It is desired to find P_r , the power received at the receiver.



FIGURE 12.3 Photograph of one of the three L-band antenna arrays for a Motorola IRIDIUM communications satellite. The IRIDIUM system consists of 66 satellites in low earth orbit to provide global personal satellite TDMA communications services, including voice, fax, and paging.

Courtesy of Raytheon Company.

If we assume that the main beams of both antennas are aligned with each other, then the radiated power density at the receive antenna can be found from (12.10) and (12.11) as

$$S = \frac{P_I G_I}{4\pi R^2} \text{ W/m}^2.$$
 12.13

Then, from (12.5), the power collected by the receive antenna is

$$P_r = SA_e$$
,





where A_c is the effective area of the receive antenna. We can express A_{ε} in terms of the gain of the receive antenna using (12.6) and (12.11):

$$A_{\rm e} = \frac{\lambda^2 G_r}{4\pi}.$$
 12.14

which then includes the effect of losses in the receive antenna. Combining these results gives the received power as

$$P_r = P_t \frac{G_t G_r \lambda^2}{(4\pi R)^2},$$
12.15

which is known as the *Friis power transmission equation* [1]. This result does not include the effects of antenna impedance mismatch, polarization mismatch, or losses due to propagation effects.

Note that the Friis equation shows that the received power is proportional to the gain of either antenna, and decays as $1/R^2$. Figure 12.5 illustrates the difference in path loss between free-space propagation and guided wave propagation by plotting the attenuation versus path length for three types of transmission lines and free-space. The three transmission lines include a coaxial cable, a circular waveguide (TE₀₁ mode), and an optical fiber, each with an attenuation constant that is typical for such a line. The curves are normalized to 0 dB at R = 1 km, to eliminate the dependences on frequency and antenna gain. Observe that the path loss for the transmission lines becomes very severe for large distances, but that free-space path loss behaves quite differently. Of course, signals on transmission lines would not be susceptible to propagation effects, weather conditions, and external interference.



FIGURE 12.5 Attenuation for various transmission schemes.

EXAMPLE 12.2 Satellite Communications Link

An earth station with a transmitter power of 120 W, a frequency of 6 GHz, and an antenna gain of 42 dB transmits to a satellite repeater. The receiver antenna on the satellite has a gain of 31 dB, and the satellite is in a synchronous orbit 35,900 km above the earth. What is the received power, in dBm?

Solution

The numerical antenna gains are

$$G_t = 10^{42/10} = 15849,$$

 $G_n = 10^{31/10} = 1259.$

The wavelength is $\lambda = 0.05$ m. Then from (12.15) the received power is

$$P_r = P_t \frac{G_t G_r \lambda^2}{(4\pi R)^2} = \frac{120(15849)(1259)(0.05)^2}{(4\pi)^2 (35.9 \times 10^6)^2} = 2.94 \times 10^{-11} \text{W}$$

= -75.3 dBm.

Microwave Transmitters and Receivers

A basic amplitude-modulated microwave radio system is shown in Figure 12.6. The input baseband signal (voice, data, video) is assumed to be at a frequency f_m ; a low-pass filter serves to remove frequencies beyond the passband of the channel. Next, a



(a)



FIGURE 12.6 block diagrams of an AM microwave transmitter and receiver. (a) Microwave radio transmitter. (b) Microwave radio receiver.

local oscillator signal is mixed with the baseband signal to produce the modulated carrier. The mixer essentially performs a product function (see Section 10.2), and yields a double sideband signal. That is, the output of the mixer contains a lower sideband, $f_{\rm LO} - f_m$, as well as an upper sideband, $f_{\rm LO} + f_m$. This process is called *up-conversion*. The local oscillator signal, $f_{\rm LO}$, is a microwave signal, and so is much higher in frequency than f_m . The power amplifier amplifies the signal, which is then radiated by the antenna.

At the receiver, the received signal is first amplified by a low-noise amplifier, although this stage may be eliminated in some systems. The mixer then *down-converts* the microwave signal to an intermediate frequency (IF), using a local oscillator which is offset from the transmitter local oscillator frequency by an amount equal to the IF frequency. The low-noise amplifier-mixer stage is the most critical in terms of overall noise figure, so it is important to use low-noise components and to keep losses to a minimum.

After the mixer, the IF signal is filtered to eliminate undesired harmonics, and amplified by the IF amplifier. The IF amplifier/filter has high gain with a narrow bandwidth (about $2f_m$), and leads to less noise power than if a high-gain RF amplifier were used alone. Using an IF amplifier also minimizes the effect of 1/f noise. In addition, the receiver can easily be tuned by changing the LO frequency. This system is known as a superheterodyne receiver. The output of the IF amplifier goes to the detector, from which the baseband signal, f_m , is recovered.

There are many variations on the above design, with the most common being the use of different modulation schemes. Single sideband (SSB) modulation generates a signal with only one sideband (either $f_{\rm LO} - f_m$ or $f_{\rm LO} + f_m$), which uses one-half the bandwidth of the double-sideband case. A single sideband signal can be generated using a single-sideband mixer, or by filtering out one of the sidebands from a double sideband signal. Better S/N ratios can be obtained with frequency modulation (FM), where the frequency of the RF carrier is varied according to the modulation voltage variation.

Noise Characterization of a Microwave Receiver

Let us now use the results of Section 10.1 to analyze the noise characteristics of a complete antenna-transmission line-receiver front end, as shown in Figure 12.7. In this





system the total noise power at the output of the receiver, N_o , will be due to contributions from the antenna pattern, the loss in the antenna, the loss in the transmission line, and from the receiver components. This noise power will determine the minimum detectable signal level for the receiver and, for a given transmitter power, the maximum range of the communication link.

The receiver components in Figure 12.7 consist of an RF amplifier with gain G_{RF} and noise temperature T_{RF} , a mixer with an RF-to-IF conversion loss factor L_M and noise temperature T_M , and an IF amplifier with gain G_{IF} and noise temperature T_{IF} . The noise effects of later stages can be ignored, since the overall noise figure is dominated by the characteristics of the first few stages. The component noise temperatures can be related to noise figures as $T = (F-1)T_0$. From (10.22) the equivalent noise temperature of the receiver can be found as

$$T_{\text{REC}} = T_{\text{RF}} + \frac{T_M}{G_{\text{RF}}} + \frac{T_{\text{IF}}L_M}{G_{\text{RF}}} .$$
 12.16

The transmission line connecting the antenna to the receiver has a loss L_T , and is at a physical temperature T_p . So from (10.15) its equivalent noise temperature is

$$T_{\rm TL} = (L_T - 1)T_p.$$
 12.17

Again using (10.22), the noise temperature of the transmission line (TL) and receiver cascade is

$$T_{\text{TL}+\text{REC}} = T_{\text{TL}} + L_T T_{\text{REC}}$$
$$= (L_T - 1)T_p + L_T T_{\text{REC}}.$$
 12.18

This noise temperature is defined at the antenna terminals (the input to the transmission line).

As discussed in Section 12.1, the entire antenna pattern can collect noise power. If the antenna has a reasonably high gain with relatively low sidelobes, we can assume that all noise power comes via the main beam, so that the noise temperature of the antenna is given by (12.12):

$$T_A = \eta T_b + (1 - \eta) T_p.$$
 12.19

where η is the efficiency of the antenna. T_p is its physical temperature, and T_b is the equivalent brightness temperature of the background seen by the main beam. (One must be careful with this approximation, as it is quite possible for the noise power collected by the sidelobes to exceed the noise power collected by the main beam, if the sidelobes are aimed at a hot background.) The noise power at the antenna terminals, which is also the noise power delivered to the transmission line, is

$$N_i = kBT_A = kB[\eta T_b + (1 - \eta)T_p],$$
 12.20

where B is the system bandwidth. If S_i is the received power at the antenna terminals, then the input signal-to-noise ratio at the antenna terminals is S_i/N_i .

The output signal power is

$$S_o = \frac{S_i G_{\rm RF} G_{\rm IF}}{L_T L_M} = S_i G_{\rm SYS}, \qquad 12.21$$
where G_{SYS} has been defined as a system power gain. The output noise power is

$$N_{o} = [N_{i} + kBT_{\text{TL}+\text{REC}}]G_{\text{SYS}}$$

= $kB(T_{A} + T_{\text{TL}+\text{REC}})G_{\text{SYS}}$
= $kB[\eta T_{b} + (1 - \eta)T_{p} + (L_{T} - 1)T_{p} + L_{T}T_{\text{REC}}]G_{\text{SYS}}$
= $kBT_{\text{SYS}}G_{\text{SYS}}$, [2.22]

where T_{SYS} has been defined as the overall system noise temperature. The output signal-to-noise ratio is

$$\frac{S_o}{N_o} = \frac{S_i}{kBT_{\rm SYS}} = \frac{S_i}{kB[\eta T_b + (1-\eta)T_p + (L_T - 1)T_p + L_T T_{\rm REC}]}.$$
 12.23

It may be possible to improve this signal-to-noise ratio by various signal processing techniques. Note that it may appear to be convenient to use an overall system noise figure to calculate the degradation in signal-to-noise ratio from input to output for the above system, but one must be very careful with such an approach because noise figure is defined for $N_i = kT_0B$, which is not the case here. It is often less confusing to work directly with noise temperatures and powers, as we did above.

EXAMPLE 12.3 Signal-to-Noise Ratio of a Microwave Receiver

A microwave receiver like that of Figure 12.7 has the following parameters:

$$f = 4.0 \text{ GHz}, \quad G_{\text{RF}} = 20 \text{ dB},$$

$$B = 1 \text{ MHz}, \quad F_{\text{RF}} = 3.0 \text{ dB},$$

$$G_A = 26 \text{ dB}, \quad L_M = 6.0 \text{ dB},$$

$$\eta = 0.90, \quad F_M = 7.0 \text{ dB},$$

$$T_p = 300 \text{ K}, \quad G_{1\text{F}} = 30 \text{ dB},$$

$$T_b = 200 \text{ K}, \quad F_{1\text{F}} = 1.1 \text{ dB}.$$

$$L_T = 1.5 \text{ dB},$$

If the received power at the antenna terminals is $S_i = -80$ dBm, calculate the input and output signal-to-noise ratios.

Solution

We first convert the above dB quantities to numerical values, and noise figures to noise temperatures:

$$G_{\rm RF} = 10^{20/10} = 100,$$

$$G_{\rm IF} = 10^{30/10} = 1000,$$

$$L_T = 10^{1.5/10} = 1.41,$$

$$L_M = 10^{6/10} = 4.0,$$

$$T_M = (F_M - 1)T_0 = (10^{7/10} - 1)(290) = 1163 \text{ K},$$

$$T_{\rm RF} = (F_{\rm RF} - 1)T_0 = (10^{3/10} - 1)(290) = 289 \text{ K},$$

$$T_{\rm IF} = (F_{\rm IF} - 1)T_0 = (10^{1.1/10} - 1)(290) = 84 \text{ K}.$$

Then from (12.16), (12.17), and (12.19) the noise temperatures of the receiver, transmission line, and antenna are

$$T_{\text{REC}} = T_{\text{RF}} + \frac{T_M}{G_{\text{RF}}} + \frac{T_{\text{IF}}L_M}{G_{\text{RF}}} = 289 + \frac{1163}{100} + \frac{84(4.0)}{100} = 304 \text{ K},$$

$$T_{\text{TL}} = (L_T - 1)T_p = (1.41 - 1)300 = 123 \text{ K},$$

$$T_A = \eta T_b + (1 - \eta)T_P = 0.9(200) + (1 - 0.9)(300) = 210 \text{ K}.$$

Then the input noise power, from (12.20), is

$$N_i = kBT_A = 1.38 \times 10^{-23} (10^6)(210) = 2.9 \times 10^{-15} \text{W} = -115 \text{ dBm}.$$

So the input signal-to-noise ratio is,

$$\frac{S_i}{N_i} = -80 + 115 = 35 \text{ dB}.$$

From (12.22) the total system noise temperature is

$$T_{\text{SYS}} = T_A + T_{\text{TL}} + L_T T_{\text{REC}} = 210 + 123 + (1.41)(304) = 762 \text{ K}.$$

This result clearly shows the noise contributions of the various components. The output signal-to-noise ratio is found from (12.23) as

$$\frac{S_o}{N_o} = \frac{S_i}{kBT_{SYS}},$$

$$kBT_{SYS} = 1.38 \times 10^{-23} (10^6) (762) = 1.05 \times 10^{-14} \text{ W} = -110 \text{ dBm},$$

$$\frac{S_o}{N_o} = -80 + 110 = 30 \text{ dB}.$$

\$O

Frequency-Multiplexed Systems

To take advantage of the wide absolute bandwidth of a microwave channel, a large number of narrow bandwidth channels can be multiplexed together. This is essentially a process of up-converting groups of voice, data, or video channels to consecutive frequency bands in the RF spectrum of the transmitter, and then down-converting at the receiver. The procedure is known as frequency-division multiplexing (FDM), and is in contrast to time-division multiplexing (TDM). where small "time-slices" of a large number of input signals are transmitted sequentially.

A simplified FDM radio is shown in Figure 12.8. At the transmitter, several narrowband baseband channels are first multiplexed together to form a group. For example, a standard grouping for voice channels is to multiplex twelve 4 kHz (nominal) bandwidth channels into a 60–108 kHz frequency band. This can be done using single-sideband mixers and a set of precision local oscillators. Several of these groups can then be combined into one RF channel, and several RF channels may be accommodated on the same RF carrier. Two RF channels are shown in Figure 12.8. To avoid wasted bandwidth, all mixing should be single-sideband.

The resulting multiplexed signal is then amplified and transmitted. At the receiver, RF channel dropping filters are used to separate the RF channels. These filters are usually three- or four-port components, where the desired channel is filtered out through



FIGURE 12.8 Frequency division multiplexed microwave radio. (a) FDM transmitter. (b) FDM receiver.

one port, and the remaining spectrum is passed through to the next channel dropping filter. Each RF channel is then down-converted. IF filtered and amplified, and demultiplexed into the original baseband channels. Practical FDM systems have capacities of 10^5 or more telephone channels. A frequency-hopping system is similar to a FDM system, but

uses a psuedo-random sequence of transmission frequencies to provide immunity from interference and jamming.

12.3 RADAR SYSTEMS

Radar, or Radio Detection And Ranging, is probably the most prevalent application of microwave technology. In its basic operation, a transmitter sends out a signal which is partly reflected by a distant target, and then detected by a sensitive receiver. If a narrow beam antenna is used, the target's direction can be accurately given by the position of the antenna. The distance to the target is determined by the time required for the signal⁴ to travel to the target and back, and the tadial velocity of the target is related to the Doppler shift of the return signal. Below are listed some of the typical applications of radar systems.

Civilian Applications

- Airport surveillance
- Marine navigation
- Weather radar
- Altimetry
- Aircraft landing
- Burglar alarms
- Speed measurement (police radar)
- Mapping

Military Applications

- Air and marine navigation
- Detection and tracking of aircraft, missiles, spacecraft
- Missile guidance
- Fire control for missiles and artillery
- Weapon fuses
- Reconnaissance

Scientific Applications

- Astronomy
- Mapping and imaging
- Precision distance measurement
- Remote sensing of natural resources

Early radar work in the United States and Britain began in the 1930s using very high frequency (VHF) sources. A major breakthrough occurred in the early 1940s with the British invention of the magnetron as a reliable source of high-power microwaves. Higher frequencies allowed the use of reasonably sized antennas with high gain, allowing mechanical tracking of targets with good angular resolution. Radar was quickly developed in Great Britain and the United States, and played an important role in World War II. Figure 12.9 shows a photograph of the phased array radar for the PATRIOT missile system. We will now derive the radar equation, which governs the basic operation of most radars, and then describe some of the more common types of radar systems.

The Radar Equation

Two basic radar systems are illustrated in Figure 12.10; in the *monostatic system* the same antenna is used for both transmit and receive, while the *bistatic system* uses two separate antennas for these functions. Most radars are of the monostatic type, but in some applications (such as missile fire control) the target is illuminated by a separate transmit antenna. Separate antennas are also sometimes used to achieve the necessary isolation between transmitter and receiver.

Here we will consider the monostatic case, but the bistatic case is very similar. If the transmitter radiates a power P_t through an antenna of gain G, the power density incident on the target is, from (12.4) and (12.11),

$$S_t = \frac{P_t G}{4\pi R^2},$$
 12.24

where R is the distance to the target. It is assumed that the target is in the main beam direction of the antenna. The target will scatter the incident power in various directions;



FIGURE 12.9 Photograph of the PATRIOT phased array radar. This is a C-band multifunction radar that provides tactical air defense, including target search and tracking, and missile fire control. The phased array antenna uses 5000 ferrite phase shifters to electronically scan the antenna beam.

Photo provided by Raytheon Company.



FIGURE 12.10 Basic monostatic and bistatic radar systems. (a) Monostatic radar system. (b) Bistatic radar system.

the ratio of the scattered power in a given direction to the incident power density is defined as the *radar cross section*, σ , of the target. Mathematically,

$$\sigma = \frac{P_s}{S_t},$$
 12.25

where P_s is the total power scattered by the target. The radar cross section thus has the dimensions of area, and is a property of the target itself. It depends on the incident and reflection angles, as well as the polarization of the incident wave.

Since the target acts as a finite-sized source, the power density of the reradiated field must decay as $1/4\pi R^2$ away from the target. Thus the power density of the scattered field back at the receive antenna must be

$$S_r = \frac{P_t G \sigma}{(4\pi R^2)^2}.$$
 12.26

Then using (12.14) for the effective area of the antenna gives the received power as

$$P_r = \frac{P_t G^2 \lambda^2 \sigma}{(4\pi)^3 R^4}.$$
 12.27

This is the radar equation. Note that the received power varies as $1/R^4$, which implies that a sensitive low-noise receiver is needed to detect targets at long ranges.

Because of noise received by the antenna and generated in the receiver, there will be some minimum detectable power that can be discriminated by the receiver. If this power is P_{min} , then (12.27) can be rewritten to give the maximum range as

$$R_{\max} = \left[\frac{P_t G^2 \sigma \lambda^2}{(4\pi)^3 P_{\min}}\right]^{1/4}.$$
 12.28

Signal processing can effectively reduce the minimum detectable signal, and so increase the usable range. One very common processing technique used with pulse radars is pulse integration, where a sequence of N received pulses are integrated over time. The effect is to reduce the noise level, which has a zero mean, relative to the returned pulse level, resulting in an improvement factor of approximately N [4].

Of course, the above results seldom describe the performance of an actual radar system. Factors such as propagation effects, the statistical nature of the detection process, and external interference often serve to reduce the usable range of a radar system.

EXAMPLE 12.4 Application of the Radar Range Equation

A pulsed radar operating at 10 GHz has an antenna with a gain of 28 dB, and a transmitter power of 2 kW (pulse power). If it is desired to detect a target with a cross section of 12 m², and the minimum detectable signal is $P_{\min} = -90$ dBm, what is the maximum range of the radar?

Solution The required numerical values are

$$G = 10^{23/10} = 631,$$

 $P_{\min} = 10^{-90/10} \text{ mW} = 10^{-12} \text{ W}.$
 $\lambda = 0.03 \text{ m}.$

Then the radar range equation of (12.28) gives the maximum range as

$$R_{\text{max}} = \left[\frac{(2 \times 10^3)(631)^2(12)(.03)^2}{(4\pi)^3(10^{-12})}\right]^{1/4}$$

= 8114 m.

Pulse Radar

A pulse radar determines target range by measuring the round-trip time of a pulsed microwave signal. Figure 12.1) shows a typical pulse radar system block diagram. The transmitter portion consists of a single-sideband mixer used to frequency offset a microwave oscillator of frequency f_0 by an amount equal to the IF frequency. After power amplification, pulses of this signal are transmitted by the antenna. The transmit/receive switch is controlled by the pulse generator to give a transmit pulse width τ , with a pulse repetition frequency (PRF) of $f_r = 1/T_r$. The transmit pulse thus consists of a short burst of a microwave signal at the frequency $f_0 + f_{\rm IF}$. Typical pulse durations range from 100 ms to 50 ns: shorter pulses give better range resolution, but longer pulses result in a better signal-to-noise ratio after receiver processing. Typical pulse repetition frequencies range from 100 Hz to 100 kHz; higher PRFs give more returned pulses per unit time,



FIGURE 12.11 A pulse radar system and timing diagram.

which improves performance, but lower PRFs avoid range ambiguities that can occur when $R > cT_r/2$.

In the receive mode, the returned signal is amplified and mixed with the local oscillator of frequency f_0 to produce the desired IF signal. The local oscillator is used for both up-conversion in the transmitter as well as down-conversion in the receiver; this simplifies the system and avoids the problem of frequency drift, which would be a consideration if separate oscillators were used. The IF signal is amplified, detected, and fed to a video amplifier/display. Search radars often use a continuously rotating antenna for 360° azimuthal coverage; in this case the display shows a polar plot of target range versus angle. Many modern radars use a computer for the processing of the detected signal and display of target information.

The transmit/receive (T/R) switch in the pulse radar actually performs two functions: forming the transmit pulse train, and switching the antenna between the transmitter and receiver. This latter function is also known as *duplexing*. In principle, the duplexing function could be achieved with a circulator, but an important requirement is that a high degree of isolation (about 80–100 dB) be provided between the transmitter and receiver, to avoid transmitter leakage into the receiver which would drown the target return (or possibly damage the receiver). As circulators typically achieve only 20–30 dB of isolation, some type of switch, with high isolation, is required. If necessary, further isolation can be obtained by using additional switches along the path of the transmitter circuit.

Doppler Radar

If the target has a velocity component along the line-of-sight of the radar, the returned signal will be shifted in frequency relative to the transmitted frequency, due to the doppler effect. If the transmitted frequency is f_o , and the radial target velocity is v_i , then the shift in frequency, or the doppler frequency, will be

$$f_d = \frac{2vf_o}{c},$$
 12.29

where c is the velocity of light. The received frequency is then $f_o \pm f_d$, where the plus sign corresponds to an approaching target and the minus sign corresponds to a receding target.

Figure 12.12 shows a basic doppler radar system. Observe that it is much simpler than a pulse radar, since a continuous wave signal is used, and the transmit oscillator can also be used as a local oscillator for the receive mixer, because the received signal is frequency offset by the doppler frequency. The filter following the mixer should have a passband corresponding to the expected minimum and maximum target velocities. It is important that the filter have high attenuation at zero frequency, to eliminate the effect of clutter return and transmitter leakage at the frequency f_0 , as these signals would down-convert to zero frequency. Then a high degree of isolation is not necessary between transmitter and receiver, and a circulator can be used. This type of filter response also helps to reduce the effect of 1/f noise.

The above radar cannot distinguish between approaching and receding targets, as the sign of f_d is lost in the detection process. Such information can be recovered, however, by using a mixer that produces separately the upper and lower sideband products.

Since the return of a pulse radar from a moving target will contain a doppler shift, it is possible to determine both the range and velocity (and position, if a narrow beam antenna is used) of a target with a single radar. Such a radar is known as a *pulse-doppler radar*,





and offers several advantages over pulse or doppler radars. One problem with a pulse radar is that it is impossible to distinguish between a true target and clutter returns from the ground, trees, buildings, etc. Such clutter returns may be picked up from the antenna sidelobes. But if the target is moving (e.g., as in an airport surveillance radar application), the doppler shift can be used to separate its return from clutter, which is stationary.

Radar Cross Section

A radar target is characterized by its radar cross section, as defined in (12.25), which gives the ratio of scattered power to incident power density. The cross section of a target depends on the frequency and polarization of the incident wave, and on the incident and reflected angles relative to the target. Thus we can define a monostatic cross section (incident and reflected angles identical), and a bistatic cross section (incident and reflected angles identical).

For simple shapes the radar cross section can be calculated as an electromagnetic boundary value problem; more complex targets require numerical techniques, or measurement to find the cross section. The radar cross section of a conducting sphere can be calculated exactly; the monostatic result is shown in Figure 12.13, normalized to πa^2 , the physical cross-sectional area of the sphere. Note that the cross section increases very quickly with size for electrically small spheres ($a << \lambda$). This region is called the *Rayleigh region*, and it can be shown that σ varies as $(a/\lambda)^4$ in this region. (This strong dependence on frequency explains why the sky is blue, as the blue component of sunlight scatters more strongly from atmospheric particles than do the lower frequency red components.)

For electrically large spheres, where $a >> \lambda$, the radar cross section of the sphere is equal to its physical cross section, πa^2 . This is the optical region, where geometrical optics are valid. Many other shapes, such as flat plates at normal incidence, also have cross sections that approach the physical area for electrically large sizes.



FIGURE 12.13 Monostatic radar cross section of a conducting sphere.

TABLE 12.3	Typical	Radar	Cross	Sections
-------------------	---------	-------	-------	----------

Between the Rayleigh region and the optical region is the *resonance region*, where the electrical size of the sphere is on the order of a wavelength. Here the cross section is oscillating with frequency, due to phase addition and cancellation of various scattered field components. Of particular note is the fact that the cross section may reach quite high values in this region.

Complex targets such as aircraft or ships generally have cross sections that vary rapidly with frequency and aspect angle. In military applications it is often desirable to minimize the radar cross section of vehicles, to reduce detectability. This can be accomplished by using radar absorbing materials (lossy dielectrics) in the construction of the vehicle. Table 12.3 lists the approximate radar cross sections of a variety of different targets.

12.4 RADIOMETRY

A radar system obtains information about a target by transmitting a signal and receiving the echo from the target, and thus can be described as an active remote sensing system. Radiometry, however, is a passive technique which develops information about a target solely from the microwave portion of the blackbody radiation (noise) that it either emits directly or reflects from surrounding bodies. A *radiometer* is a sensitive receiver specially designed to measure this noise power.

Theory and Applications of Radiometry

As discussed in Section 10.1, a body in thermodynamic equilibrium at a temperature T radiates energy according to Planck's radiation law. In the microwave region this result reduces to P = kTB, where k is Boltzmann's constant, B is the system bandwidth, and P is the radiated power. This result strictly applies only to a blackbody, which is defined as an idealized material which absorbs all incident energy, and reflects none; a

blackbody also radiates energy at the same rate as it absorbs energy, thus maintaining thermal equilibrium. A nonideal body will partially reflect incident energy, and so does not radiate as much power as would a blackbody at the same temperature. A measure of the power radiated by a body relative to that radiated by an ideal blackbody at the same temperature is the emissivity, e, defined as

$$e = \frac{P}{kTB},$$
 12.30

where P is the power radiated by the nonideal body, and kTB is the power that would be emitted by a perfect blackbody. Thus, $0 \le e \le 1$, and e = 1 for a perfect blackbody.

As we saw in Section 10.1, noise power can also be quantified in terms of equivalent temperature. Thus for radiometric purposes we can define a brightness temperature, T_B , as

$$T_B = eT. 12.31$$

where T is the physical temperature of the body. This shows that, radiometrically, a body always looks cooler than its actual temperature, since $0 \le e \le 1$.

Now consider Figure 12.14, which shows the antenna of a radiometer receiving noise powers from various sources. The antenna is pointed at a region of the earth which has an apparent brightness temperature T_B . The atmosphere emits radiation in all directions: the component radiated directly toward the antenna is T_{AD} , while the power reflected from the earth to the antenna is T_{AR} . There may also be noise powers that enter the sidelobes of the antennas, from the sun or other sources. Thus, we can see that the total brightness temperature seen by the radiometer is a function of the scene under observation, as well as the observation angle, frequency, polarization, attenuation of the atmosphere, and the antenna pattern. The objective of radiometry is to infer information about the scene from the measured brightness temperature, and an analysis of the radiometric mechanisms that relate brightness temperature to physical conditions of the scene. For example, the power reflected from a uniform layer of snow over soil





can be treated as plane wave reflection from a multilayer dielectric region, leading to the development of an algorithm that gives the thickness of the snow in terms of measured brightness temperature at various frequencies.

Microwave radiometry is a relatively new area of technology, and one which is strongly interdisciplinary, drawing on results from fields such as electrical engineering, oceanography, geophysics, and atmospheric and space sciences, to name a few. Below, some of the more typical applications of microwave radiometry are listed.

Environmental Applications

- Measurement of soil moisture
- Flood mapping
- Snow cover/lee cover mapping
- Ocean surface windspeed
- Atmospheric temperature profile
- Atmospheric humidity profile

Military Applications

- Target detection
- Target recognition
- Surveillance
- Mapping

Astronomy Applications

- Planetary mapping
- Solar emission mapping
- Mapping of galactic objects
- · Measurement of cosmological background radiation

Figure 12.15 shows a photograph of a radiometer used to measure the water vapor profile of the atmosphere.

Total Power Radiometer

The aspect of radiometry that is of most interest to the microwave engineer is the design of the radiometer itself. The basic problem is to build a receiver that can distinguish between the desired radiometric noise and the inherent noise of the receiver, even though the radiometric power is usually less than the receiver noise power. Although it is not a very practical instrument, we will first consider the total power radiometer, because it represents a simple and direct approach to the problem and serves to illustrate the difficulties involved in radiometer design.

The block diagram of a typical total power radiometer is shown in Figure 12.16. The front end of the receiver is a standard superheterodyne circuit consisting of an RF amplifier, a mixer/local oscillator, and an IF stage. The IF filter determines the system bandwidth, B. The detector is generally a square-law device, so that its output voltage is proportional to the input power. The integrator is essentially a low-pass filter with a cutoff frequency of $1/\tau$, and serves to smooth out short-term variations in the noise





Courtesy of the Microwave Remote Sensing Laboratory. University of Massachusetts at Amberst.

power. For simplicity, we assume that the antenna is lossless, although in practice antenna loss will affect the apparent temperature of the antenna, as given in (12.12).

If the antenna is pointed at a background scene with a brightness temperature T_B , the antenna power will be $P_A = kT_B B$; this is the desired signal. The receiver contributes noise which can be characterized as a power $P_R = kT_R B$ at the receiver input, where T_R is the overall noise temperature of the receiver. Thus the output voltage of the radiometer is

$$V_a = G(T_B + T_R)kB, 12.32$$

where G is the overall gain constant of the radiometer. Conceptually, the system is



FIGURE 12.16 Total power radiometer block diagram.

calibrated by replacing the antenna input with two calibrated noise sources, from which the system constants GkB and GT_RkB can be determined. (This is similar to the Yfactor method for measuring noise temperature,) Then the desired brightness temperature, T_B , can be measured with the system.

Two types of errors occur with this radiometer. First is an error, ΔT_N , in the measured brightness temperature due to noise fluctuations. Since noise is a random process, the measured noise power may vary from one integration period to the next. The integrator (or low-pass filter) acts to smooth out ripples in V_o with frequency components greater than $1/\tau$. It can be shown that the remaining error is [2]

$$\Delta T_N = \frac{T_B + T_R}{\sqrt{B\tau}}.$$
12.33

This result shows that if a longer measurement time, τ , can be tolerated, the error due to noise fluctuation can be reduced to a negligible value.

A more serious error is due to random variations in the system gain, G. Such variations generally occur in the RF amplifier, mixer, or IF amplifier, over a period of one second or longer. So if the system is calibrated with a certain value of G, which changes by the time a measurement is made, an error will occur, as given in reference [2] as

$$\Delta T_{\mathcal{G}} = (T_B + T_R) \frac{\Delta G}{G}, \qquad 12.34$$

where ΔG is the rms change in the system gain, G.

It will be useful to consider some typical numbers at this time. For example, a 10 GHz total power radiometer may have a bandwidth of 100 MHz, a receiver temperature of $T_R = 500$ K, an integrator time constant of $\tau = 0.01$ s, and a system gain variation $\Delta G/G = 0.01$. If the antenna temperature is $T_B = 300$ K, (12.33) gives the error due to noise fluctuations as, $\Delta T_N = 0.8$ K, while (12.34) gives the error due to gain variations as $\Delta T_G = 8$ K. These results, which are based on reasonably realistic data, show that

gain variation is the most detrimental factor affecting the accuracy of the total power radiometer.

The Dicke Radiometer

We have seen that the dominant factor affecting the accuracy of the total power radiometer is the variation of gain of the overall system. Since such gain variations have a relatively long time constant (> 1 second), it is conceptually possible to eliminate this error by repeatedly calibrating the radiometer at rapid rate. This is the principle behind the operation of the Dicke null-balancing radiometer.

A system diagram is shown in Figure 12.17. The superheterodyne receiver is identical to the total power radiometer, but the input is periodically switched between the antenna and a variable power noise source; this switch is called the Dicke switch. The output of the square-law detector drives a synchronous demodulator, which consists of a switch and a difference circuit. The demodulator switch operates in synchronism with the Dicke switch, so that the output of the subtractor is proportional to the difference between the noise powers from the antenna, T_B , and the reference noise source, T_{REF} . The output of the subtractor is then used as an error signal to a feedback control circuit, which controls the power level of the reference noise source so that V_o approaches zero. In this balanced state, $T_B = T_{REF}$, and T_B can be determined from the control voltage, V_c . The square-wave sampling frequency, f_s , is chosen to be much faster than the drift time of the system gain, so that this effect is virtually eliminated. Typical sampling frequencies range from 10 to 1000 Hz.



FIGURE 12.17 Balanced Dicke radiometer block diagram.

A typical radiometer would measure the brightness temperature T_B over a range of about 50-300 K; this then implies that the reference noise source would have to cover this same range, which is difficult to do in practice. Thus, there are several variations on the above design, differing essentially in the way that the reference noise power is controlled or added to the system. One possible method is to use a constant T_{REF} which is somewhat hotter than the maximum T_B to be measured. The amount of reference noise power delivered to the system is then controlled by varying the pulse width of the sampling waveform. Another approach is to use a constant reference noise power, and vary the gain of the IF stage during the reference sample time to achieve a null output. Other possibilities, including alternatives to the Dicke radiometer, are discussed in the literature [2].

12.5 MICROWAVE PROPAGATION

In free-space electromagnetic waves propagate in straight lines without attenuation or other adverse effects. Free-space, however, is an idealization that is only approximated when microwave energy propagates through the atmosphere or in the presence of the earth. In practice the performance of a communication, radar, or radiometry system may be seriously affected by propagation effects such as reflection, refraction, attenuation, or diffraction. Below we discuss some specific propagation phenomenon that can influence the operation of microwave systems. It is important to realize that propagation effects generally cannot be quantified in any exact or rigorous sense, but can only be described in terms of their statistics.

Atmospheric Effects

The relative permittivity of the atmosphere is close to unity, but is actually a function of air pressure, temperature, and humidity. An empirical result which is useful at microwave frequencies is given by [5]

$$\epsilon_{\tau} = \left[1 + 10^{-6} \left(\frac{79P}{T} - \frac{11V}{T} + \frac{3.8 \times 10^5 V}{T^2}\right)\right]^2.$$
 12.35

where P is the barometric pressure in millibars, T is the temperature in kelvin, and V is the water vapor pressure in millibars. This result shows that permittivity generally decreases (approaches unity) as altitude increases, since pressure and humidity decrease with height faster than does temperature. This change in permittivity with altitude causes radio waves to bend toward the earth, as depicted in Figure 12.18. Such refraction of radio waves can sometimes be useful, since it may extend the range of radar and communication systems beyond the limit imposed by the presence of the earth's horizon.

If an antenna is at a height, h, above the earth, simple geometry gives the line-ofsight distance to the horizon as

$$d = \sqrt{2Rh}, \qquad 12.36$$



FIGURE 12.18 Refraction of radio waves by the atmosphere.

where R is the radius of the earth. From Figure 12.18 we see that the effect of refraction on range can be accounted for by using an effective earth radius kR, where k > 1. A value commonly used [4] is k = 4/3, but this is only an average value which changes with weather conditions. In a radar system, refraction effects can lead to errors when determining the elevation of a target close to the horizon.

Weather conditions can sometimes produce a temperature inversion, where the temperature increases with altitude. Equation (12.35) then shows that the atmospheric permittivity will decrease much faster than normal, with increasing altitude. This condition can sometimes lead to *ducting* (also called trapping, or anomalous propagation), where a radio wave can propagate long distances parallel to the earth's surface, via the duct created by the layer of air along the temperature inversion. The situation is very similar to propagation in a dielectric waveguide. Such ducts can range in height from 50–500 feet, and may be near the earth's surface, or higher in altitude.

Another atmospheric effect is attenuation, caused primarily by the absorption of microwave energy by water vapor and molecular oxygen. Maximum absorption occurs when the frequency coincides with one of the molecular resonances of water or oxygen, thus atmospheric attenuation has distinct peaks at these frequencies. Figure 12.19 shows the atmospheric attenuation vs. frequency. At frequencies below 10 GHz the atmosphere has very little effect on the strength of a signal. At 22.2 and 183.3 GHz, resonance peaks occur due to water vapor resonances, while resonances of molecular oxygen cause peaks at 60 and 120 GHz. Thus there are "windows" in the millimeter wave band near 35, 94, and 135 GHz where radar and communication systems can operate with minimum loss. Precipitation such as rain, snow, or fog will increase the attenuation, especially at higher frequencies. The effect of atmospheric attenuation can be included in system design when using the Friis transmission equation or the radar equation.

In some instances the system frequency may be chosen at a point of maximum atmospheric attenuation. Remote sensing of the atmosphere (temperature, water vapor, rain rate) is often done with radiometers operating near 20 or 55 GHz, to maximize the sensing of atmospheric conditions (see Figure 12.15). Another interesting example is spacecraft-to-spacecraft communication at 60 GHz. This millimeter wave frequency has the advantages of a large bandwidth and small antennas with high gains and, since the



FIGURE 12.19 Average atmospheric attenuation versus frequency (horizontal polarization).

atmosphere is very lossy at this frequency, the possibilities of interference, jamming, and eavesdropping from earth are greatly reduced.

Ground Effects

The most obvious effect of the presence of the ground on microwave propagation is reflection from the earth's surface (land or sea). As shown in Figure 12.20, a radar target (or receiver antenna) may be illuminated by both a direct wave from the transmitter and a wave reflected from the ground. The reflected wave is generally smaller in amplitude than the direct wave, because of the larger distance it travels, the fact that it usually radiates from the sidelobe region of the transmit antenna, and because the ground is not a perfect reflector. Nevertheless, the received signal at the target or receiver will be the vector sum of the two wave components and, depending on the relative phases of the two



FIGURE 12.20 Direct and reflected waves over the earth's surface.

waves, may be greater or less than the direct wave alone. Because the distances involved are usually very large in terms of the electrical wavelength, even a small variation in the permittivity of the atmosphere can cause *fading* (long term fluctuations) or *scintillation* (short term fluctuations) in the signal strength. These effects can also be caused by reflections from inhomogeneities in the atmosphere.

In communication systems such fading can sometimes be reduced by making use of the fact that the fading of two communication channels having different frequencies, polarizations, or physical locations is essentially independent. Thus a communication link can reduce fading by combining the outputs of two (or more) such channels; this is called a diversity system.

Another ground effect is *diffraction*, whereby a radio wave scatters energy in the vicinity of the line-of-sight boundary at the horizon, thus giving a range slightly beyond the horizon. This effect is usually very small at microwave frequencies. Of course, when obstacles such as hills, mountains, or buildings are in the path of propagation, diffraction effects can be stronger.

In a radar system, unwanted reflections often occur from terrain, vegetation, trees, buildings, and the surface of the sea. Such clutter echoes generally degrade or mask the return of a true target, or show up as a false target, in the context of a surveillance or tracking radar. In mapping or remote sensing applications such clutter returns may actually constitute the desired signal.

Plasma Effects

A *plasma* is a gas consisting of ionized particles. The ionosphere consists of spherical layers of atmosphere with particles which have been ionized by solar radiation, and thus forms a plasma region. A very dense plasma is formed on a spacecraft as it reenters the atmosphere, due to the high temperatures produced by friction. Plasmas are also produced by lightning, meteor showers, and nuclear explosions.

A plasma is characterized by the number of ions per unit volume; depending on this density and the frequency, a wave might be reflected, absorbed, or transmitted by the plasma medium. An effective permittivity can be defined for a uniform plasma region as

$$\epsilon_{e} = \epsilon_{o} \left(1 - \frac{\omega_{p}^{2}}{\omega^{2}} \right), \qquad 12.37$$

where

$$\omega_p = \sqrt{\frac{Nq^2}{m\epsilon_0}}$$
12.38

is the plasma frequency. In (12.38), q is the charge of the electron, m is the mass of the electron, and N is the number of ionized particles per unit volume. By studying the solution of Maxwell's equations for plane wave propagation in such a medium, it can be shown that wave propagation through a plasma is only possible for $\omega > \omega_{\mu}$. Lower frequency waves will be totally reflected. If a magnetic field is present, the plasma becomes anisotropic, and the analysis is more complicated. The earth's magnetic field may be strong enough to produce such an anisotropy in some cases.

The ionosphere consists of several different layers with varying ion densities; in order of increasing ion density, these layers are referred to as D, E, F_1 , and F_2 . The

characteristics of these layers depends on seasonal weather and solar cycles, but the average plasma frequency is about 8 MHz. Thus, signals at frequencies less than 8 MHz (e.g., short-wave radio) can reflect off the ionosphere to travel distances well beyond the **horizon**. Higher frequency signals, however, will pass through the ionosphere.

A similar effect occurs with a spacecraft entering the atmosphere. The high velocity of the spacecraft causes a very dense plasma to form around the vehicle. The electron density is high enough so that, from (12.38), the plasma frequency is very high, thus inhibiting communication with the spacecraft until its velocity has decreased. Besides this blackout effect, the plasma layer may also cause a large impedance mismatch between the antenna and its feed line.

12.6 OTHER APPLICATIONS AND TOPICS

Microwave Heating

To the average consumer, the term "microwave" connotes a microwave oven, which is used in many households for heating food; industrial and medical applications also exist for microwave heating. As shown in Figure 12.21, a microwave oven is a relatively simple system consisting of a high-power source, a waveguide feed, and the oven cavity. The source is generally a magnetron tube operating at 2.45 GHz, although 915 MHz is sometimes used when greater penetration is desired. Power output is usually between 500 and 1500 W. The oven cavity has metallic walls, and is electrically large. To reduce the effect of uneven heating caused by standing waves in the oven, a "mode stirrer," which is just a metallic fan blade, is used to perturb the field distribution inside the oven. The food is also rotated with a motorized platter.

In a conventional oven a gas or charcoal fire, or an electric heating element, generates heat outside of the material to be heated. The outside of the material then gets heated by convection, and the inside of the material by conduction. In microwave heating, by contrast, the inside of the material gets heated first. The process through which this occurs primarily involves the conduction losses in materials with large loss tangents [5], [6]. An interesting fact is that the loss tangents of many foods decrease with increasing





temperature, so that microwave heating is to some extent self-regulating. The result is that microwave cooking generally gives faster and more uniform heating of food, as compared with conventional cooking. The efficiency of a microwave oven, when defined as the ratio of power converted to heat (in the food) to the power supplied to the oven, is generally less than 50%; this is usually greater than the cooking efficiency of a conventional oven, however.

The most critical issue in the design of a microwave oven is safety. Since a very high power source is used, leakage levels must be very small to avoid exposing the user to harmful radiation. Thus the magnetron, feed waveguide, and oven cavity must all be carefully shielded. The door of the oven requires particular attention; besides close mechanical tolerances, the joint around the door usually employs RF absorbing material and a $\lambda/4$ choke flange to reduce leakage to an acceptable level.

Energy Transfer

Electrical power transmission lines are a very efficient and convenient way to transfer energy from one point to another, as they have relatively low loss and initial costs, and can be easily routed. There are applications, however, where it is inconvenient or impossible to use such power lines. In such cases it is conceivable that electrical power can be transmitted without wires by a well-focused microwave beam [7].

One example is the solar satellite power station, where it has been proposed that electricity be generated in space by a large orbiting array of solar cells, and transmitted to a receiving station on earth by a microwave beam. We would thus be provided with a virtually inexhaustible source of electricity. Placing the solar arrays in space has the advantage of power delivery uninterrupted by darkness, clouds, or precipitation, which are problems encountered with earth-based solar arrays.

To be economically competitive with other sources, the solar power satellite station would have to be very large. The present proposal [5] involves a solar array about 5×10 km in size, feeding a 1 km diameter phased array antenna. The power output on earth would be on the order of 5 GW. Such a project is extremely large in terms of cost and complexity. Also of legitimate concern is the operational safety of such a scheme, both in terms of radiation hazards associated with the system when it is operating as designed, as well as the risks involved with a malfunction of the system. These considerations, as well as the political and philosophical ramifications of such a large, centralized power system, have made the future of the solar power satellite station doubtful.

Similar in concept, but on a much smaller scale, is the transmission of electrical power from earth to a vehicle such as a small drone helicopter or airplane. The advantages are that such an aircraft could run indefinitely, and very quietly, at least over a limited area. Battlefield surveillance and weather prediction would be some possible applications. The concept has been demonstrated with several projects involving small pilotless aircraft.

A very high power pulsed microwave source and a high-gain antenna can be used to deliver an intense burst of energy to a target, and thus used as a weapon. The pulse may be intense enough to do physical damage to the target, or it may act to overload and destroy sensitive electronic systems.

Electronic Warfare

Military radar and communication systems may be limited or prevented from performing their intended function by deliberate means, such as interference, jamming, and other countermeasures. We will give a brief outline of this subject, which generally goes under the title of *electronic warfare*. Most of the discussion applies to radar systems, although some techniques, such as jamming, apply to communication systems as well. Electronic warfare can be divided into three major headings.

- 1. Electronic support measures (ESM). These involve the use of "threat warning" receivers to detect the presence of a search or tracking radar, or the presence of a jamming signal. Such receivers are used on aircraft, ships, and ground vehicles, and usually give characteristics of the threat radar such as operating frequency, pulse waveform, and direction.
- 2. *Electronic countermeasures* (ECM). These include the use of both active and passive techniques to either confuse or deceive a radar or communication system.
- **3.** Electronic counter-countermeasures (ECCM). Any radar or communication system can be rendered ineffective by a forceful and determined ECM effect. The purpose of ECCM is to make such a goal too costly to be achieved.

We will discuss electronic countermeasures in more detail. Passive countermeasures are methods to reduce the probability that a target will be detected by a radar, or to present false targets to the radar. Passive countermeasures do not involve the emission of any signals or noise. Examples include chaff, which are clouds of thin foil strips ejected by an aircraft to present a false target to a radar; decoys, which are dummy vehicles that appear to a radar as an aircraft or missile; radar cross section (RCS) reduction, by proper shaping of the vehicle, or by coating it with lossy materials.

Active countermeasures involve the radiation of a signal to confuse or deceive the radar or communication system, and generally take the form of jammers. A noise jammer radiates a relatively large power in the operating band of the radar, to mask the true target return, or to overload the receiver front end. An effective ECCM strategy here is to design the receiver with overload protection, and to provide frequency agility, whereby the radar changes its operating frequency over a large range in a pseudo-random manner. A barrage jammer radiates noise signals over a wide band of frequencies, to encompass the tuning range of a frequency agile system. The jamming power over a given radar bandwidth is then less than that of the noise jammer operating at one frequency. If the jamming signal is being transmitted from a location different than the target, its effect can be reduced by using an adaptive array antenna on the radar to form a pattern with a null in the direction of the jammer. A repeater jammer receives the radar pulse, delays it (or otherwise modifies it), and retransmits it to give the radar false range or velocity information.

It is relatively straightforward to use the radar equation and Friis power transmission formula to quantitatively describe some of the most common jamming scenarios. In the *self-screening jammer* (SSJ) case, the interogating radar is jammed by a jammer at the target itself, as illustrated in Figure 12.22a. The objective of the jammer is to radiate enough power at the radar to overwhelm the target return. The jammer has the advantage of only a one-way path loss, but it generally must radiate over a broader bandwidth and



FIGURE 12.22 Two common jamming scenarios. (a) Self-screening jammer. (b) Stand-off jammer.

larger angular sector than the radar, to be sure of blanking its signal. The important parameter is the jammer-to-signal (J/S) power ratio; effective jamming requires a J/S ratio of 0 to 20 dB, depending on the radar characteristics. From the radar equation of (12.27), the received power at the radar due to reflection from the target is

$$P_r = P_t \frac{G_r^2 \lambda^2 \sigma}{(4\pi)^3 R^4},$$
 12.39

where P_t is the radar transmitter power, G_r is the radar antenna gain (monostatic case assumed), and σ is the radar cross section of the target. Using the Friis power transmission formula of (12.15) gives the jammer noise power at the radar as

$$P_n = P_j \frac{G_j G_r \lambda^2}{(4\pi R)^2} \left(\frac{B_r}{B_j}\right),$$
12.40

where P_j is the radiated power of the jammer, and G_j is the gain of the jammer antenna. The factor B_r/B_j is the ratio of the radar bandwidth to the bandwidth of the jammer, and accounts for the fact that the jammer must radiate power over a frequency band large enough to account for an uncertainty in the radar frequency. Using these two results gives the J/S ratio as,

$$\frac{J}{S} = \frac{P_n}{P_r} = \frac{P_j}{P_t} \frac{4\pi R^2 G_j}{\sigma G_\tau} \left(\frac{B_r}{B_j}\right) \qquad (SSJ).$$

In a *stand-off jammer* (SOJ) scenario, the jammer is at a different location than the target. The target lies in the main beam of the radar antenna, but the jammer radiates power into the sidelobes, as illustrated in Figure 12.22b. The radar return from the target is still given by (12.39), but the noise power received from the jammer is now given by

$$P_n = P_j \frac{G_j G_r^{\text{SL}} \lambda^2}{(4\pi R_j)^2} \left(\frac{B_r}{B_j}\right), \qquad 12.42$$

where G_r^{SL} is the gain in the sidelobe (SL) region of the radar antenna, and R_j is the distance from the jammer to the radar. Then the J/S ratio for this case is

$$\frac{J}{S} = \frac{P_n}{P_r} = \frac{P_j}{P_t} \frac{4\pi R^4 G_j G_r^{SL}}{R_j^2 G_r^2 \sigma} \left(\frac{B_r}{B_j}\right)$$
(SOJ). 12.43

EXAMPLE 12.5 Electronic Countermeasures

A 10 GHz tracking radar has a transmitter power of 100 kW, an antenna gain of 35 dB, and an IF bandwidth of 1 MHz. It is attempting to track an aircraft with a cross section of 4 m² at a range of 3 km. A self-screening jammer aboard the aircraft has a transmitter power of 1 kW over a 20 MHz bandwidth, with an antenna gain of 10 dB. Find the resulting J/S ratio, and compare with J/S for the same jammer used in a stand-off configuration at a distance of 10 km from the radar. Assume a 20 dB sidelobe level for the radar antenna.

Solution

The radar antenna gain is

$$G_r = 10^{35/10} = 3162.$$

and the jammer antenna gain is

$$G_i = 10^{10/10} = 10.$$

The wavelength is $\lambda = 0.03$ m. Then from (12.41), the J/S ratio for the SSJ case is

$$\frac{J}{S} = \frac{P_j}{P_t} \frac{4\pi R^2 G_j}{\sigma G_r} \left(\frac{B_r}{B_j}\right) = \frac{(1000)(4\pi)(3000)^2(10)(1)}{(10^5)(4)(3162)(20)} = 45 = 16.5 \text{ dB} \quad (SSJ),$$

The radar antenna gain in its sidelobe region is

$$G_r^{\rm SL} = 10^{(35-20)/10} = 31.6.$$

so the J/S ratio for the SOJ case can be found from (12.43) as

$$\frac{J}{S} = \frac{P_j}{P_l} \frac{4\pi R^4 G_j G_r^{SL}}{R_j^2 G_r^2 \sigma} \left(\frac{B_r}{B_j}\right) = \frac{(1000)(3000)^4 (10)(31.6)(1)}{(10^5)(10.000)^2 (3162)^2 (20)}$$
$$= 0.013 = -19 \text{ dB} \qquad (SOJ).$$

So we see that the self-screening jammer should be reasonably effective, but the stand-off jammer will not be very useful.

Biological Effects and Safety

The proven dangers of exposure to microwave radiation are due to thermal effects. The body absorbs RF and microwave energy and converts it to heat; as in the case of a microwave oven, this heating occurs within the body, and may not be felt at low levels. Such heating is most dangerous in the brain, the eye, the genitals, and the stomach organs. Excessive radiation can lead to cataracts, sterility, or cancer. Thus it is important to determine a safe radiation level standard, so that users of microwave equipment will not be exposed to harmful power levels.

The most recent U.S. safety standard for human exposure to electromagnetic fields is given by IEEE Standard C95.1-1991. In the RF-microwave frequency range of 100 MHz to 300 GHz, exposure limits are set on the power density (W/cm²) as a function of frequency, as shown in Figure 12.23. The recommended safe power density limit is as low as 0.2 mW/cm² at the lower end of this frequency range, because fields penetrate the body more deeply at lower frequencies. At frequencies above 15 GHz the power density limit rises to 10 mW/cm², since most of the power absorption at such frequencies occurs near the skin surface. By comparison, the sun radiates a power density of about 100 mW/cm² on a clear day, but the effect of this radiation is much less severe than a corresponding level of lower-frequency microwave radiation because the sun heats the outside of the body, with much of the generated heat reabsorbed by the air, while microwave power heats from inside the body. At frequencies below 100 MHz electric and



FIGURE 12.23 IEEE Standard C95.1-1991 recommended power density limits for human exposure to RF and microwave electromagnetic fields.

magnetic fields interact with the body differently than higher frequency electromagnetic fields, and so separate limits are given for field components at these lower frequencies.

Other countries have different standards for microwave exposure limits, some of which are lower than the U.S. limits. Some of these standards are a function of exposure time, with lower power density limits for prolonged exposure. A separate standard applies to microwave ovens in the United States: law requires that all ovens be tested to ensure that the power level at 5 cm from any point on the oven does not exceed 1 mW/cm².

Most experts feel that the above limits represent safe levels, with a reasonable margin. Some researchers, however, feel that health hazards may occur due to nonthermal effects of long-term exposure to even low levels of microwave radiation.



EXAMPLE 12.6 Power Density in the Vicinity of a Microwave Radio Link

A 6 GHz common-carrier microwave communications link uses a tower-mounted antenna with a gain of 40 dB, and a transmitter power of 5 W. To evaluate the radiation hazard of this system, calculate the power density at a distance of 20 m from the antenna. Do this for a position in the main beam of the antenna, and for a position in the sidelobe region of the antenna. Assume a worst-case sidelobe level of -10 dB.

Solution

The numerical gain of the antenna is

$$G = 10^{40/10} = 10^4$$

Then from (12.10) and (12.11), the power density in the main beam of the antenna at a distance of R = 20 m is

$$S = \frac{P_{\text{in}}G}{4\pi R^2} = \frac{5 \times 10^4}{4\pi (20)^2} = 10. \text{ W/m}^2 = 1.0 \text{ mW/cm}^2.$$

The worst-case power density in the sidelobe region would be 10 dB below this, or 0.10 mW/cm^2 .

Thus we see that the power density in the main beam at 20 m is below the United States standard. The power density in the sidelobe region is well below this limit. These power densities will diminish rapidly with increasing distance, due to the $1/r^2$ dependence.

REFERENCES

- [1] C. A. Balanis. Antenna Theory: Analysis and Design, Harper and Row, N. Y., 1982.
- [2] F. T. Ulaby, R. K. Moore, and A. K. Fung, Microwave Remote Sensing: Active and Passive, Volume I. Microwave Remote Sensing, Fundamentals and Radiometry. Addison-Wesley, Reading, Mass., 1981.
- [3] K. Miyauchi, "Millimeter-Wave Communications," in Infrared and Millimeter Waves, vol. 9, K. J. Button, Ed., Academic Press, N. Y., 1983.
- [4] M. I. Skolnik, Introduction to Radar Systems, McGraw-Hill, N. Y., 1962.
- [5] F. E. Gardiol. Introduction to Microwaves, Artech House, Dedham, Mass., 1984.

- [6] E. C. Okress, Microwave Power Engineering, Academic Press, N. Y., 1968.
- [7] W. C. Brown, "The History of Power Transmission by Radio Waves," *IEEE Trans. Microwave Theory and Techniques*, vol. MTT-32, pp. 1230–1242 September 1984.

PROBLEMS

- 12.1 At a distance of 5 km from an antenna, the radiated power density in the main beam is measured to be $1.27 \times 10^{-6} \text{ W/m}^2$. If the input power to the antenna is 100 W, what is the gain of the antenna?
- 12.2 A short monopole antenna on a ground plane has a far-zone electric field given by

$$E_{\theta} = \frac{C}{r} \sin \theta e^{-jk_0 r}, \quad \text{for } 0 \le \theta \le 90^\circ.$$

and is zero elsewhere (i.e., below the ground plane). Determine the directivity of this antenna.

- 12.3 A plane wave with a power density of 2×10^{-6} W/m² is incident in the main beam direction of an antenna having a gain of 12 dB. If the frequency is 8 GHz, what is the power delivered to a matched load at the antenna terminals?
- 12.4 When a certain antenna, having a physical temperature of 295 K, is aimed at the overhead sky an equivalent noise temperature of 34.5 K is measured. If we assume that the antenna has enough gain so that it sees an essentially uniform brightness temperature of 5 K, what is the efficiency of the antenna?
- 12.5 A microwave radio link at 4.9 GHz uses transmit and receive antennas with gains of 30 dB. If the distance between transmitter and receiver is 27 km, and it is desired to have a minimum received power level of -60 dBm, what is the required transmitter power?
- 12.6 A microwave relay transmitter has a power of 10 W. If the antenna has a gain of 30 dB, and we assume a worst-case sidelobe level of -15 dB (below the main beam), what are the power densities in the main beam and sidelobe regions of the antenna, at a distance of 30 m?
- 12.7 Derive the radar equation for the bistatic case, where the transmit and receive antennas have gains of G_t and G_r , and are at distances R_t and R_r from the target, respectively.
- 12.8 A pulse radar has a pulse repetition frequency $f_r = 1/T_r$ Determine the maximum unambiguous range of the radar. (Range ambiguity occurs when the round-trip time of a return pulse is greater than the pulse repetition time, so it becomes unclear as to whether a given return pulse belongs to the last transmitted pulse, or some earlier transmitted pulse.)
- **12.9** A doppler radar operating at 12 GHz is intended to detect target velocities ranging from 1 m/sec to 20 m/sec. What is the required passband of the doppler filter?
- **12.10** A pulse radar operates at 2 GHz and has a per-pulse power of 1 kW. If it is to be used to detect a target with $\sigma = 20 \text{ m}^2$ at a range of 10 km, what should be the minimum isolation between the transmitter and receiver, so that the leakage signal from the transmitter is at least 10 dB below the received signal? Assume an antenna gain of 30 dB
- **12.11** An antenna, having a gain G, is shorted at its terminals. What is the minimum monostatic radar cross section in the direction of the main beam?
- 12.12 A self-screening jammer has an output power of 100 W, an antenna gain of 15 dB, and a bandwidth of 10 MHz. It is used against a radar having a transmitter power of 10 kW, with an antenna gain of 30 dB and an IF bandwidth of 0.5 MHz. If the operating frequency is 12 GHz and the target RCS is 5 m², compute the minimum range for which J/S = 1.
- 12.13 On a clear day, with the sun directly overhead, the received power density from sunlight is about 1300 W/m². If we make the simplifying assumption that this power is transmitted via a single-frequency plane wave, find the resulting amplitude of the incident electric and magnetic fields.

Appendices

- Appendix: A Prefixes
- Appendix: B Vector Analysis
- Appendix: C Bessel Functions
- Appendix: D Other Mathematical Results
- Appendix: E Physical Constants
- Appendix: F Conductivities for Some Materials
- Appendix: G Dielectric Constants and Loss Tangents for Some Materials
- Appendix: H Properties of Some Microwave Ferrite Materials
- Appendix: I Standard Rectangular Waveguide Data
- Appendix: J Standard Coaxial Cable Data

Appendix A PREFIXES

Multiplying Factor	Prefix	Symbol
10 ¹²	lera	<u>т</u>
10 ⁹	giga	G
10 ⁶	mega	М
10 ³	kilo	k
10 ²	hecto	h
10 ¹	deka	da
10^{-1}	deci	d
10^{-2}	centi	с
10-3	milli	m
10 ⁻⁶	micro	μ
10 ⁻⁹	nano	ß
10^{-12}	pico	р
10 ⁻¹⁵	femto	f

Appendix B VECTOR ANALYSIS Coordinate Transformations

Rectangular to cylindrical	:	\hat{x}		\hat{y}	ź
	ρ	$\cos\phi$	-	$\sin \phi$	0
	$\hat{\phi}$	$-\sin\phi$	ç	$\cos \phi$	0
-	£	0	()	1
Rectangular to spherical:	$ \hat{x} \\ \frac{\sin\theta\cos\phi}{\cos\phi} \\ -\sin\phi $		$\vec{\mathbf{y}}$ $\sin\theta\sin\phi$ $\cos\theta\sin\phi$ $\cos\phi$		$ \begin{array}{c} $
г О Ф					
Cylindrical to spherical:		ĝ	 φ	ź	
	ŕ	$\sin \theta$	0	$\cos \theta$	-
	в д	$\begin{array}{c} \cos\theta\\ 0\end{array}$	0 1	— sin 0	Ø
		· · ·			

These tables can be used to transform unit vectors as well as vector components; e.g., $\hat{\rho} = \pm \cos \phi + \hat{y} \sin \phi$

 $A_{\rho} = A_x \cos \phi + A_y \sin \phi$

.

Vector Differential Operators

Rectangular coordinates:

$$\nabla f = \hat{x} \frac{\partial f}{\partial x} + \hat{y} \frac{\partial f}{\partial y} + \hat{z} \frac{\partial f}{\partial z}$$

$$\nabla \cdot \bar{A} = \frac{\partial A_x}{\partial x} + \frac{\partial A_y}{\partial y} + \frac{\partial A_z}{\partial z}$$

$$\nabla \times \bar{A} = \hat{x} \left(\frac{\partial A_z}{\partial y} - \frac{\partial A_y}{\partial z} \right) + \hat{y} \left(\frac{\partial A_x}{\partial z} - \frac{\partial A_z}{\partial x} \right) + \hat{z} \left(\frac{\partial A_y}{\partial x} - \frac{\partial A_x}{\partial y} \right)$$

$$\nabla^2 f = \frac{\partial^2 f}{\partial x^2} + \frac{\partial^2 f}{\partial y^2} + \frac{\partial^2 f}{\partial z^2}$$

$$\nabla^2 \bar{A} = \hat{x} \nabla^2 A_x + \hat{y} \nabla^2 A_y + \hat{z} \nabla^2 A_z$$

Cylindrica' coordinates:

$$\nabla f = \hat{\rho} \frac{\partial f}{\partial \rho} + \hat{\phi} \frac{1}{\rho} \frac{\partial f}{\partial \phi} + \hat{z} \frac{\partial f}{\partial z}$$

$$\nabla \cdot \bar{A} = \frac{1}{\rho} \frac{\partial}{\partial \rho} (\rho A_{\rho}) + \frac{1}{\rho} \frac{\partial A_{\phi}}{\partial \phi} + \frac{\partial A_{z}}{\partial z}$$

$$\nabla \times \bar{A} = \hat{\rho} \left(\frac{1}{\rho} \frac{\partial A_{z}}{\partial \phi} - \frac{\partial A_{\phi}}{\partial z} \right) + \hat{\phi} \left(\frac{\partial A_{\rho}}{\partial z} - \frac{\partial A_{z}}{\partial \rho} \right) + \hat{z} \frac{1}{\rho} \left[\frac{\partial (\rho A_{\phi})}{\partial \rho} - \frac{\partial A_{\rho}}{\partial \phi} \right]$$

$$\nabla^{2} f = \frac{1}{\rho} \frac{\partial}{\partial \rho} \left(\rho \frac{\partial f}{\partial \rho} \right) + \frac{1}{\rho^{2}} \frac{\partial^{2} f}{\partial \phi^{2}} + \frac{\partial^{2} f}{\partial z^{2}}$$

$$\nabla^{2} \bar{A} = \nabla (\nabla \cdot \bar{A}) - \nabla \times \nabla \times \bar{A}$$

Spherical coordinates:

•

$$\nabla f = \hat{r} \frac{\partial f}{\partial r} + \hat{\theta} \frac{1}{r} \frac{\partial f}{\partial \theta} + \frac{\hat{\phi}}{r \sin \theta} \frac{\partial f}{\partial \phi}$$

$$\nabla \cdot \bar{A} = \frac{1}{r^2} \frac{\partial}{\partial r} (r^2 A_r) + \frac{1}{r \sin \theta} \frac{\partial}{\partial \theta} (\sin \theta A_\theta) + \frac{1}{r \sin \theta} \frac{\partial A_\phi}{\partial \phi}$$

$$\nabla \times \bar{A} = \frac{\hat{r}}{r \sin \theta} \left[\frac{\partial}{\partial \theta} (A_\phi \sin \theta) - \frac{\partial A_\theta}{\partial \phi} \right] + \frac{\hat{\theta}}{r} \left[\frac{1}{\sin \theta} \frac{\partial A_r}{\partial \phi} - \frac{\partial}{\partial r} (rA_\phi) \right]$$

$$+ \frac{\hat{\phi}}{r} \left[\frac{\partial}{\partial r} (rA_\theta) - \frac{\partial A_r}{\partial \theta} \right]$$

$$\nabla^2 f = \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial f}{\partial r} \right) + \frac{1}{r^2} \frac{\partial}{\sin \theta} \frac{\partial}{\partial \theta} \left(\sin \theta \frac{\partial f}{\partial \theta} \right) + \frac{1}{r^2 \sin^2 \theta} \frac{\partial^2 f}{\partial \phi^2}$$

$$\nabla^2 \bar{A} = \nabla \nabla \cdot \bar{A} - \nabla \times \nabla \times \bar{A}$$

Vector identities:

$$\vec{A} \cdot \vec{B} = |A| |B| \cos \theta, \quad \text{where } \theta \text{ is the angle between } \vec{A} \text{ and } \vec{B} \qquad B.1$$
$$|\vec{A} \times \vec{B}| = |A| |B| \sin \theta, \quad \text{where } \theta \text{ is the angle between } \vec{A} \text{ and } \vec{B}, \qquad B.2$$
$$\vec{A} \cdot \vec{B} \times \vec{C} = \vec{A} \times \vec{B} \cdot \vec{C} = \vec{C} \times \vec{A} \cdot \vec{B} \qquad B.3$$
$$\vec{A} \times \vec{B} = -\vec{B} \times \vec{A} \qquad B.4$$

$$\vec{A} \times (\vec{B} \times \vec{C}) = (\vec{A} \cdot \vec{C})\vec{B} - (\vec{A} \cdot \vec{B})\vec{C}$$
B.5

$$\nabla(fg) = g\nabla f + f\nabla g \tag{B.6}$$

$$\nabla \cdot (f\tilde{A}) = \tilde{A} \cdot \nabla f + f \nabla \cdot \tilde{A}$$
B.7

$$\nabla \cdot (\vec{A} \times \vec{B}) = (\nabla \times \vec{A}) \cdot \vec{B} - (\nabla \times \vec{B}) \cdot \vec{A}$$
 B.8

$$\nabla \times (f\bar{A}) = (\nabla f) \times \bar{A} + f \nabla \times \bar{A}$$
B.9

$$\nabla \times (\vec{A} \times \vec{B}) = \vec{A} \nabla \cdot \vec{B} - \vec{B} \nabla \cdot \vec{A} + (\vec{B} \cdot \nabla) \vec{A} - (\vec{A} \cdot \nabla) \vec{B}$$
B.10

$$\nabla \cdot (\vec{A} \cdot \vec{B}) = (\vec{A} \cdot \nabla)\vec{B} + (\vec{B} \cdot \nabla)\vec{A} + A \times (\nabla \times \vec{B}) + \vec{B} \times (\nabla \times \vec{A})$$
 B.11

$$\nabla \cdot \nabla \times \bar{A} = 0$$
 B.12

$$\nabla \times (\nabla f) = 0 \tag{B.13}$$

$$\nabla \times \nabla \times \vec{A} = \nabla \nabla \cdot \vec{A} - \nabla^2 \vec{A}$$
 B.14

Note: the term $\nabla^2 \hat{A}$ has meaning only for rectangular components of \hat{A} .

$$\int_{V} \nabla \cdot \vec{A} \, dv = \oint_{S} \vec{A} \cdot d\vec{s} \qquad \text{(divergence theorem)} \qquad B.15$$

$$\int_{\mathcal{S}} (\nabla \times \tilde{A}) \cdot d\tilde{s} = \oint_{C} \bar{A} \cdot d\tilde{\ell} \qquad \text{(Stokes' theorem)} \qquad B.16$$

Appendix C BESSEL FUNCTIONS

Bessel functions are solutions to the differential equation,

$$\frac{1}{\rho}\frac{d}{d\rho}\left(\rho\frac{df}{d\rho}\right) + \left(k^2 - \frac{n^2}{\rho^2}\right)f = 0 \qquad C.1$$

where k^2 is real and n is an integer. The two independent solutions to this equation are called ordinary Bessel functions of the first and second kind, written as $J_n(k\rho)$ and $Y_n(k\rho)$, and so the general solution to (C.1) is

$$f(\rho) = AJ_n(k\rho) + BY_n(k\rho)$$
C.2

where A and B are arbitrary constants to be determined from boundary conditions.

These functions can be written in series form as

$$J_n(x) = \sum_{m=0}^{\infty} \frac{(-1)^m (x/2)^{n+2m}}{m!(n+m)!}$$
C.3

$$Y_n(x) = \frac{2}{\pi} \left(\gamma + \ln \frac{x}{2} \right) J_n(x) - \frac{1}{\pi} \sum_{m=0}^{n-1} \frac{(n-m-1)!}{m!} \left(\frac{2}{x} \right)^{n-2m} - \frac{1}{\pi} \sum_{m=0}^{\infty} \frac{(-1)^m (x/2)^{n+2m}}{m!(n+m)!} \\ \times \left(1 + \frac{1}{2} + \frac{1}{3} + \dots + \frac{1}{m} + 1 + \frac{1}{2} + \dots + \frac{1}{n+m} \right)$$
C.4

where $\gamma = 0.5772...$ is Euler's constant, and $x = k\rho$. Note that Y_n becomes infinite at x = 0, due to the ln term. From these series expressions, small argument formulas can be obtained as

$$J_n(\pi) \sim \frac{1}{n!} \left(\frac{x}{2}\right)^n \qquad C.5$$

$$Y_0(x) \sim \frac{2}{\pi} \ln x \qquad \qquad C.6$$

$$Y_n(x) \sim \frac{-1}{\pi} (n-1)! \left(\frac{x}{2}\right)^n, \qquad n > 0$$
 C.7

Large argument formulas can be derived as

$$J_n(x) \sim \sqrt{\frac{2}{\pi x}} \cos\left(x - \frac{\pi}{4} - \frac{n\pi}{2}\right)$$
 C.8

$$Y_n(x) \sim \sqrt{\frac{2}{\pi x}} \sin\left(x - \frac{\pi}{4} - \frac{n\pi}{2}\right)$$
 C.9

Figure C.1 shows graphs of a few of the lowest order Bessel functions of each type.

Recurrence formulas relate Bessel functions of different orders:

$$Z_{n+1}(x) = \frac{2n}{x} Z_n(x) - Z_{n-1}(x)$$
 C.10

$$Z'_{n}(x) = \frac{-n}{x} Z_{n}(x) + Z_{n-1}(x)$$
 C.11

$$Z'_{n}(x) = \frac{n}{x} Z_{n}(x) - Z_{n+1}(x)$$
 C.12

$$Z'_{n}(x) = \frac{1}{2} [Z_{n-1}(x) - Z_{n+1}(x)]$$
 C.13

where $Z_n = J_n$ or Y_n . The following integral relations involving Bessel functions are useful:

$$\int_0^x Z_m^2(kx)x \, dx = \frac{x^2}{2} \left[Z_n^{\prime 2}(kx) + \left(1 - \frac{n^2}{k^2 x^2} \right) Z_n^2(kx) \right]$$
 C.14

$$\int_0^x Z_n(kx) Z_n(\ell x) x \, dx = \frac{x}{k^2 - \ell^2} [k Z_n(\ell x) Z_{n+1}(kx) - \ell Z_n(kx) Z_{n+1}(\ell x)]$$
 C.15

$$\int_{0}^{p_{nm}} \left[J_{n}^{\prime 2}(x) + \frac{n^{2}}{x^{2}} J_{n}^{2}(x) \right] x \, dx = \frac{p_{nm}^{2}}{2} J_{n}^{\prime 2}(p_{nm}) \tag{C.16}$$

$$\int_{0}^{p'_{nm}} \left[J_{n}^{'2}(x) + \frac{n^{2}}{x^{2}} J_{n}^{2}(x) \right] x \, dx = \frac{(p'_{nm})^{2}}{2} \left(1 - \frac{n^{2}}{(p'_{nm})^{2}} \right) J_{n}^{2}(p'_{nm}) \qquad \text{C.17}$$

where $J_n(p_{nm}) = 0$, and $J'_n(p'_{nm}) = 0$. The zeros of $J_n(x)$ and $J'_n(x)$ are on pages 702 and 703.



I



IOT U <	< x < 12			
n	ł	2	3	4
0	2.4048	5.5200	8.6537	11.7951
1	3.8317	7.0155	10.1743	
2	5.1356	8.4172	11.6198	
3	6.3801	9.7610		
4	7.5883	11.0647		
5	8.7714			
6	9.9361			
7	11.0863			
	-			

Zeros of Bessel functions of first kind: $J_n(x) = 0$ for 0 < x < 12

n	1	2	3	4
0	3.8317	7.0156	10.1735	13.3237
1	1.8412	5.3314	8.5363	11.7060
2	3.0542	6.7061	9.9695	
3	4.2012	8.0152	11.3459	
4	5.3175	9.2824		
5	6.4156	10.5199		
6	7.5013	11.7349		
7	8.5778			
8	9,6474			1
9	10.7114			
10	11.7709			<u> </u>
				Í
				<u>}</u>

Extrema of Bessel functions of first kind: $dI_{-}(\pi)/d\pi = 0$ for $0 < \infty < 12$

Appendix D OTHER MATHEMATICAL RESULTS

Useful Integrals

$$\int_0^a \cos^2 \frac{n\pi x}{a} dx = \int_0^a \sin^2 \frac{n\pi x}{a} dx = \frac{a}{2}, \qquad \text{for } n \ge 1 \qquad D.1$$

$$\int_{0}^{a} \cos \frac{m\pi x}{a} \cos \frac{n\pi x}{a} dx = \int_{0}^{a} \sin \frac{m\pi x}{a} \sin \frac{n\pi x}{a} dx = 0, \quad \text{for } m \neq n \quad D.2$$

$$\int_{0}^{a} \cos \frac{m\pi x}{a} \sin \frac{n\pi x}{a} dx = 0$$
 D.3

Taylor Series

$$f(x) = f(x_0) + (x - x_0) \frac{df}{dx} \bigg|_{x = x_0} + \frac{(x - x_0)^2}{2!} \left| \frac{d^2 f}{dx^2} \bigg|_{x = x_0} + \cdots \right| D.4$$

$$e^{x} = 1 + x + \frac{x^2}{2!} + \frac{x^3}{3!} + \cdots$$
 D.5

$$\frac{1}{1-x} = 1 + x + x^2 + x^3 + \cdots, \qquad \text{for } |x| < 1 \qquad D.6$$

$$\sqrt{1+x} = 1 + \frac{x}{2} - \frac{x^2}{8} + \cdots,$$
 for $|x| < 1$ D.7

$$\ln x = 2\left(\frac{x-1}{x+1}\right) + \frac{2}{3}\left(\frac{x-1}{x+1}\right)^3 + \cdots, \quad \text{for } x > 0 \qquad D.8$$

$$\sin x = x - \frac{x^3}{3!} + \frac{x^5}{5!} + \cdots$$
D.9

$$\cos x = 1 - \frac{x^2}{2!} + \frac{x^3}{4!} + \cdots$$
 D.10

Appendix E PHYSICAL CONSTANTS

- Permittivity of free-space = $\epsilon_0 = 8.854 \times 10^{-12}$ F/m
- Permeability of free-space = $\mu_0 = 4\pi \times 10^{-7}$ H/m

14.45

- Impedance of free-space = $\eta_0 = 376.7\Omega$
- Velocity of light in free-space $= c = 2.998 \times 10^8$ m/s
- Charge of electron = $q = 1.602 \times 10^{-19}$ C
- Mass of electron $= m = 9.107 \times 10^{-31}$ kg
- Boltzmann's constant = $k = 1.380 \times 10^{-23}$ J/°K
- Planck's constant = $\hbar = 1.054 \times 10^{-34}$ J-s
- Gyromagnetic ratio = $\gamma = 1.759 \times 10^{11}$ C/Kg (for g = 2)

Appendix F CONDUCTIVITIES FOR SOME MATERIALS

Material	Conductivity S/m (20°C)	Material	Conductivity S/m (20°C)
	3.816×10 ⁷	Nichrome	1.0×10 ⁶
Brass	2.564×10^{7}	Nickel	1.449×10^{7}
Bronze	1.00×10^{7}	Platinum	9.52×10^{6}
Chromium	3.846×10^{7}	Sea water	3-5
Conter	5.813×10^{7}	Silicon	4.4×10^{-4}
Distilled water	2×10^{-4}	Silver	6.173×10^{7}
Germanium	2.2×10^{6}	Steel (silicon)	2×10^{6}
Gold	4.098×10^{7}	Steel (stainless)	1.1×10^{6}
Graobite	7.0×10^4	Solder	7.0×10^{6}
Iron	1.03×10^{7}	Tungsten	1.825×10^{7}
Mercury	1.04×10^{6}	Zinc	1.67×10^{7}
Lead	4.56×10^{6}		
Appendix G DIELECTRIC CONSTANTS AND LOSS TANGENTS FOR SOME MATERIALS

Material	Frequency	Er .	tan δ (25°C)
Alumina (99.5%)	10 GHz	9.5–10.	0.0003
Barium tetratitanate	6 GHz	37± 5%	0.0005
Beeswax	10 GHz	2.35	0.005
Beryllia	10 GHz	6.4	0.0003
Ceramic (A-35)	3 GHz	5.60	0,0041
Fused quartz	10 GHz	3.78	1000.0
Gallium arsenide	10 GHz	13.	0.006
Glass (pyrex)	3 GHz	4.82	0.0054
Glazed ceramic	10 GHz	7.2	0.008
Lucite	10 GHz	2.56	0.005
Nylon (610)	3 GHz	2.84	0.012
Parafin	10 GHz	2.24	0.0002
Plexiglass	3 GHz	2.60	0.0057
Polyethylene	10 GHz	2.25	0.0004
Polystyrene	10 GHz	2.54	0.00033
Porcelain (dry process)	100 MHz	5.04	0.0078
Rexolite (1422)	3 GHz	2.54	0.00048
Silicon	10 GHz	11.9	0.004
Styrofoam (103.7)	3 GHz	1.03	0.0001
Teflon	10 GHz	2.08	0.0004
Titania (D-100)	6 GHz	96±5%	0.001
Vaseline	10 GHz	2.16	0.001
Water (distilled)	3 GHz	76.7	0.157

<u>Appendix H</u> PROPERTIES OF SOME MICROWAVE FERRITE MATERIALS

Material	Trans-Tech Number	$4\pi Ms$ G	ΔH Oe	€r	$ an \delta$	$\overset{T_c}{^{\circ}\mathbf{C}}$	$4\pi Mr$ G
Magnesium ferrite	TT1-105	1750	225	12.2	0.00025	225	1220
Magnesium ferrite	TT1-390	2150	540	12.7	0.00025	320	1288
Magnesium ferrite	TT1-3000	3000	190	12.9	0.0005	240	2000
Nickel ferrite	TT2-101	3000	350	12.8	0.0025	\$85	1853
Nickel ferrite	TT2-113	500	150	9.0	0.0008	120	140
Nickel ferrite	TT2-125	2100	460	12.6	0.001	560	1426
Lithium ferrite	TT73-1700	1700	<400	16.1	0.0025	460	1139
Lithium ferrite	TT73-2200	2200	<450	15.8	0.0025	\$20	1474
Yttrium gamet	G-113	1780	45	15.0	0.0002	280	1277
Aluminum garnet	G-610	680	40	14.5	0.0002	185	515

.

Appendix I STANDARD RECTANGULAR WAVEGUIDE DATA

Band [*]	Recommended Frequency Range (GHz)	TE ₁₀ Cutoff Frequency (GHz)	EIA Designation WR-XX	Inside Dimensions Inches (cm)	Outside Dimensions Inches (cm)
L	1.12-1.70	0.908	WR-650	6.500×3.250	6.660 × 3.410
				(16.51×8.255)	(16.916×8.661
R	1.70-2.60	1.372	WR-430	4.300×2,150	4.460×2.310
				(10.922×5.461)	(11.328×5.867
\$	2.60-3.95	2.078	WR-284	2.840×1.340	3.000×1.500
				(7.214×3.404)	(7.620×3.810)
H (G)	3.95-5.85	3.152	WR-187	1.872×0.872	2.000×1.000
				(4.755×2.215)	(5.080×2.540)
C (J)	5.85-8.20	4.301	WR-137	1.372×0.622	1.500×0.750
				(3.485×1.580)	(3.810×1.905)
W (H)	7.05-10.0	5.259	WR-112	1.122×0.497	1.250×0.625
				(2.850×1.262)	(3.175×1.587)
X	8.20-12.4	6.557	WR-90	0.900×0.400	1.000×0.500
				(2.286×1.016)	(2.540×1.270)
Ku (P)	12.4-18.0	9.486	WR-62	0.622×0.311	0.702×0.391
				(1.580×0.790)	(1.783×0.993)
Ķ	18.0-26.5	14.047	WR-42	0.420×0.170	0.500×0.250
				(1.07 ×0.43)	(1.27 ×0.635)
Ka (R)	26.5-40.0	21.081	WR-28	0.280×0.140	0.360×0.220
				(0.711×0.356)	(0.914×0.5 59)
Q	33.0-50.5	26.342	WR-22	0.224×0.112	0.304×0.192
				(0.57 ×0.28)	(0.772×0.488)
U	40.0-60.0	31.357	WR-19	0.188×0.094	0.268×0.174
				(0.48 ×0.24)	(0.681×0.442)
V	50.0-75.0	39.863	WR-15	0.148×0.074	0.228×0.154
				(0.38 ×0.19)	(0.579×0.391)
B	60.0-90.0	48.350	WR-12	0.122×0.061	0.202×0.141
				(0.31 ×0.015)	(0.513×0. 356)
W	75.0-110.0	59.010	WR-10	0.100×0.050	0.180×0.130
				(0.254×0.127)	(0,458×0.330)
F	90.0-140.0	73.840	WR-8	0.080×0.040	0.160×0.120
				(0.203×0.102)	(0.406×0.305)
Ð	10.0-170.0	90.854	WR-6	0.065×0.0325	0.145×0.1125
	· · _ · · · · · · ·	•	··· ···	(0.170×0.083)	(0.368×0.2858
Ġ	140.0-220.0	115.750	WR-5	0.051×0.0255	0.131×0.1055
æ			-1 -2	(0.130×0.0648)	$(0.333 \times .2680)$

* Letters in parentheses denote alternative designations.

Cable Type	Impedance (Ω)	Dielectric Material [†]	Overall Diameter (In.)	Dielectric Diameter (ln.)	Maximum Operating Voltage
RG-8A/U	52	P	0.405	0.285	5000
RG9B/U	50	P	0.425	0.285	5000
RG55/U	54	P	0.216	0.116	1900
RG-58/U	50	P	0.195	0.116	1900
RG-59/U	75	P	0.242	0.146	2300
RG-141/U	50	Ŧ	0.190	0.116	1900
RG142/U	50	Ť	0.206	0.116	1900
RG-174/U	50		0.100	0.060	1500
RG178/U	50	Т	0.075	0.036	750
RG-180/U	95	T	0.137	0.103	750
RG-187/U	75	Ť	0.110	0.060	1200
RG-188/U	50	_	0.110	0.060	
RG-195/U	95	Т	0.155	0.102	1500
RG213/U	50	P	0.405	0.285	5000
RG-214/U	50	P	0.425	0.285	5000
RG-223/U	50	P	0.216	0.116	1900
RG-401	50	T	0.250	0.208	
RG-402	50	Т	0.141	0.118	
RG-405	50	r	0.087	0.066	

[†] P: Polyethylene, T: Tefton



FIGURE J.1 Attenuation data for coaxial cables.

708



ABCD parameters, 206-209 table for basic circuits, 208 table for conversions, 211 Admittance inverters, 468–469 Admittance matrix, 191–194 table for conversions, 211 AM modulation, 563, 666-667 Ampere's law, 7 Amplifier design, 618–641 balanced, 632-635 distributed, 635-640 low-noise, 628-632 maximum gain, 618-621 specified gain, 622-627 stability, 612-618 Anisotropic media, 11–12 Antenna basic characteristics, 655-662 temperature, 662 types, 656-658 Aperture coupling, 237-244, 337-339 Attenuation atmospheric, 686~687 transmission line, 90-98 Attenuation constant for circular waveguide, 136, 141 coaxial line, 92 dielectric loss, 111-112 microstrip line, 163 barallel plate waveguide, 116-117, 119 plane wave in lossy dielectric, 18-19 rectangular waveguide, 125, 128 stripline, 156 Attenuator, 198-199

Balanced amplifiers, 632-635 Bandpass filters coupled line, 474-485 coupled resonator, 486-494 lumped element, 457-462 **Bandstop** filters coupled resonator, 486-490 lumped element, 457-462 Bessel functions, 700-703 zeroes of, 135, 137, 702-703 Bethe hole coupler, 369-374 Bethe small hole coupling theory, 237-244 Binomial coefficients, 279 Binomial filter response, 444-445 Binomial matching transformer. 278-282 Biological effects, 694-695 Black body, 679 Bloch impedance, 426 Bode-Fano criterion, 295-297 Boltzmann's constant, 550 Boundary conditions, 12-16 Branch line coupler, 379-383 Brewster angle, 41 Brightness temperature, 680 Cascade connection, 207 Cavity resonators cylindrical cavity, 318-323 dielectric resonator, 323-328 rectangular cavity, 313-318 Characteristic impedance, 58-59 coaxial line, 65 microstrip line, 162

Characteristic impedance (cont.) parallel plate line, 113 stripline, 155-156 Chebyshev filter response, 445 matching transformers, 282-288 polynomials, 283-285 Chip capacitor, resistor, 257-258 Choke bias, 577. 603 flange. 131-132 Circular polarization, 25-26, 504-506 Circular waveguide attenuation, 136, 139 cutoff frequency, 135, 137 field patterns, 140 propagation constant, 134, 137 table for, 139 TE and TM mades, 133-138 Circulator ferrite junction, 537-543 general properties, 353, 535-537 Coaxial connectors, 146-147 Coaxial line attenuation constant, 92, 95-96 characteristic impedance, 65 data for standard lines, 707 distributed line parameters, 61-62 field analysis. 63-65, 141-145 higher-order modes, 143-145 voltage breakdown, 176-177 Commensurate line filters, 463-468 Commensurate line transformers, 277 Communication systems, 662-672 Composite filters, 440-443 Compression point. 549 Computer aided design (CAD), 222 Conductivity, 9 table for metals, 704 Conductor loss, 28-30 Conjugate matching, 89-90 Connectors, coaxial, 146-147 Constant gain circles, 622-625 Constant-k filters, 433-436, 442 Constant noise figure circles, 628-632 Conversion loss, mixer, 567 Coplanar waveguide, 175-176 Coupled lines, 384-388 characteristic impedances, 386-388

couplers, 389-397 filters, 474-485 Couplers (see Directional couplers) Coupling aperture, 237-244 coefficient, 333-334 critical, 332-334 resonator, 332-339 Cross guide coupler, 412 Current displacement, 7, 11 electric, magnetic, 6, 9 Cutoff frequency circular waveguide, 135, 137 parallel plate waveguide, 115, 118 rectangular waveguide, 123, 126 Cutoff wavelength, 115, 119, 128, 139 DC block. 577, 683-694 Decibel notation, 72-73 Demagnetization factor, 508-510 Detector, 559-565 sensitivity, 563 Dicke radiometer, 684-685 Dielectric constant, table, 705 Dielectric loaded waveguide, 130-131, 169-170 Dielectric loss, 29 Dielectric loss tangent, table, 705 Dielectric resonator oscillators, 648-651 Dielectric resonators, 323-328 Dielectric strength for air, 176-177 Dielectric waveguide, 175 Diode detectors, 559-565 I-V curve, 560-561 Directional couplers, 354-357 aperture coupled, 369-379 Bethe hole, 369-374 branch fine, 379-383 coupled line, 389-397 Lange, 398-401 Moreno cross guide, 412 multihole waveguide, 374-389 Riblet short slot, 413-414 ring hybrid, 401-407 Schwinger reversed phase, 413 Directivity antenna, 658-660 coupler, 356-359

Index

Discontinuities, 222-230 microstrip, 224, 229-230 waveguide, 223 Dispersion. 92, 170-173 Distortionless line, 92-93 Double sideband modulation, 567 Dynamic range, 548-549 Effective permittivity, microstrip, 161–162 Electric energy, 27 Electric field, 6 Electric flux density, 6 Electric polarizability, 239 Electric potential, 108 Electric susceptibility, 9 Electric wall, 15 Electromagnetic spectrum, 2 Electron gun, 593 Electronic warfate, 691-694 Elliptic filter, 445–446 Emissivity, 680 Energy, electric, magnetic, 27 Energy transmission, 690 E plane T-junction. 359 Equal ripple filter response, 445 Equivalent voltages and currents, 183-187 Even-odd mode analysis for branch line coupler. 380-382 coupled line coupler, 389-394 coupled line filter, 474-485 ring hybrid. 401-407 tapered coupled line hybrid, 407-411 Wilkinson divider, 363-366 Even-odd mode characteristic impedance for coupled microstrip lines, 388 coupled striplines, 387 Exponential tapered line, 290-291 Extraordinary wave, 516-517 Fabry-Perot resonator, 328-332 Faraday rotation, 512-515 Faraday's law, 7 Fernte devices circulators, 537-543 gyrator, 535 isolators, 523-530 loaded waveguide, 518-523 phase shifters, 530-535 Ferrites, 497 loss in, 506-508

microscopic model for, 498-500 permeability tensor for, 503 plane wave propagation in, 511-517 table of properties, 705 Fiber optics, 665 Field displacement isolator, 527-530 Filters bandpass, 457-462, 474-485, 486-494 bandstop, 457-461, 486-490 composite. 440-443 constant-k. 433-436 elliptic, 445-446 high pass, 442, 455-456 implementation, 462-469 linear phase, 446, 451-452 low pass, 433-443, 447-452 m-derived, 436-440 scaling, 454-456 transformations, 457-462 Flanges, waveguide . 131-132 Flow graph, 213-217 Frequency bands, 2 Friis power transmission formula, 663-665 Gain (see Power gain) amplifier, 610-611 antenna, 662 two-port power, 606-610 Group delay, 446 Group velocity, 170–173 for periodic structures, 429 for waveguide, 173 Gunn diode, 589-591 Gyrator, 535 Gyromagnetic ratio, 498 Gyrotropic medium (see Ferrites) Helmholtz equations, 16 Hertz, H., 3-4, 5 High pass filters constant-k, 436, 442 m-derived, 442 transformation to, 455 High-Z, Low-Z filters, 470-473 History, of microwave engineering, 3-5 H plane T-junction, 359 Hybrid junctions coupled line, 389-397 quadrature, 379-383 ring (rat-race), 402-407

711

Hybrid junctions (cont.) scattering matrix. 355-356 tapered coupled line, 407-411 waveguide magic-T, 411 Image impedance, 431-433 Image parameters, filter design using, 433-443 Image theory, 47-49 IMPATT diode, 589-591 Impedance characteristic, 58-59 concept of, 187-190 image, 431-433 intrinsic, 17 wave, 17, 19, 108, 109, 110 Impedance inverters, 468-469 Impedance matching, 251-252 Bode-Fano criterion, 295-297 double stub 266-271 L-section, 252-257 multisection transformer, 277-278 quarter wave transformer, 83-87, 276-277 single stub. 258-266 tapered line, 288-295 Impedance matrix, 191-196 table for conversions, 211 Impedance transformers (see Impedance matching) Incremental inductance rule, Wheeler, 96-98 Insertion loss, 72 Insertion loss method for filter design, 443-452 Intermodulation distortion, 574-575 Inverters, admittance, impedance, 468-469 Iris. waveguide, 223 Isolators field displacement, 527-530 resonance, 523-527

Jammer. 691–694 Junction circulator, 537–543 Junctions (see Hybrid junctions)

Kittel's equation, 510 Klopfenstein tapered line, 291–293 Klystron, 594 Kuroda identities, 464–468

Lange coupler, 398-401 Linearly polarized plane waves, 16-21 Linear phase filter, 446, 451-452 Line parameters (per unit length), 60-62

Line width, gyromagnetic resonance, 507-508 Loaded lines (see Periodic structures) Loaded Q, 306 Loaded waveguide dielectric loading, 130-131, 169-170 ferrite loading, 518-523 Loop coupling in waveguide, 236-237 Loss (see also Attenuation constant) conductor, 28-30 dielectric, 28-29 femite. 506-508 insertion, 72 return, 67 Loss tangent, 11 table, 705 Lossy transmission lines, 90-98 Low pass filters constant-k, 433-436, 442 Mederica, 436-440, 442 prototype, 447-452 L-section matching. 252-257 Magic-T. 411 Magnetic energy, 27 Magnetic field, 6 Magnetic flux density, 6 Magnetic polarizability, 239 Magnetic susceptibility, 11 Magnetic wall, 15 Matched line, 66 Matching (see Impedance matching) Material constants table of conductivities, 704 table of dielectric constants and loss tangents, 705 table of ferrite properties. 705 Maximally flat filter response, 444-445 Maximum power capacity, 176-177 Maxwell, J., 3-5 Maxwell's equations, 5-6, 8-9 m-derived filters, 436-440, 442 Microstrip approximate analysis, 164-167 attenuation, 163 characteristic impedance, 162 coupled. 388 effective permittivity, 161-162 field pattern, 161 propagation constant, 161 Microstrip discontinuities, 224, 229-230 Microwave heating, 689-690

713

Microwave integrated circuits (MIC), 583 hybrid, 584 monolithic (MMIC), 584-587 Microwave oven, 689-690 Microwave sources, 588-596 Gunn diode, 589-591 IMPATT diode, 589-591 oscillators, 641-652 tubes, 593-596 Microwave tubes, 593-596 backward wave oscillator, 594 crossed-field amplifier, 594-595 extended interaction oscillator, 594 gyratron, 595 klystron, 594 magnetron, 595 traveling wave tube, 594 Millimeter waves, 1 Mixers. 559, 565-573 antiparallel diode, 571 balanced. 568-571 conversion loss, 567 double balanced, 571-572 image rejection, 572-573 image response, 567 single-ended. 565-568 Modal analysis, 222-229 Mode chart, for cavity, 320 Modes cavity modes, 313-315, 318-320 circular waveguide, 132-139 parallel plate waveguide, 112-120 rectangular waveguide, 120-128 Modulation, 563-565 Monolithic integrated circuits, 583-587 Multiple reflections, on quarter wave transformer, 85-87 Multiple section coupled line couplers, 394-397 Multiple section matching transformers, 277-288 Multiplexers, 670-672 Negative resistance oscillators, 641-644 Neper, 72-73 Network analyzer, 205-206 Noise, 548-550 figure, 555-559 sources, 548-550, 552-553 temperature, 550-553 Noise figure, 555-559 of cascade, 557-559

of lossy line, 556 of transistor amplifier, 628-632 Ohm's law for fields, 9-10 Open circuit stub, impedance, 69 Oscillators dielectric resonator, 648-651 negative resistance, 641-644 transistor, 644-647 Parallel plate waveguide attenuation, 116-119 bouncing plane wave interpretation, 116 characteristic impedance (TEM), 113 table for, 119 Periodic structures analysis, 424–426 Bloch impedance, 426 k-3 diagram, 428-430 phase and group velocities, 428-429 propagation constant, 426 Permanent magnets, 510-511 Permeability, 6, 11–12 tensor, for ferrite, 503 Permittivity, 6, 9-11 of atmosphere, 685-686 Perturbation theory for attenuation, 94-96 cavity resonance. 340-346 ferrite loaded waveguide, 521 Phase constant (see Propagation constant) Phase matching, 40 Phase shifters Faraday rotation, 533-534 loaded line, 581-582 reflection, 582-583 Reggia-Spencer, 534-535 remanent (latching), 530-533 switched line, 580-581 Phase velocity coaxial line, 59 plane wave, 17 waveguide, 113, 115, 119, 124, 170 Phasor notation, 8 Physical constants, table, 704 PIN diodes, 576-577 phase shifters, 580-583 switches, 577-580 Plane waves, 16-26 in conducting media, 19-20

Plane wayes (cont.) in ferrites, 511-517 in lossless dielectric, 16-18 Plasma, 688–689 Polarizability, 239 Polarization, wave, 25-26 Power, 26-30 loss in conductor, 28-30, 36-38 toss in dielectric, 29 toss in transmission line, 94 Power capacity of transmission line, 176-177 Power divider (see also Directional coupler) resistive, 361-362 T-junction, 359-362 Wilkinson, 363-368 Power gain, 606-611 available. 606-608 transducer, 606-609 unilateral transducer. 606-609 Poynting's theorem, 27-28 Poynting vector, 28 Precession, magnetic dipole, 499-500, 501-502 Probe coupling, 234-236 Propagation atmospheric effects, 685-687 ground effects, 687-688 plasma effects. 688-689 Propagation constant for circular waveguide, 134, 137 coaxia] line, 64 microstrip line, 161 parallel plate guide, 113, 114, 118 plane waves in a good conductor, 19, 21 plane waves in lossless dielectric, 16, 21 rectangular waveguide, 123, 126 stripline, 155 TEM modes, 107 TM or TE modes, 109-110

Q. 302, 306
for circular cavity, 320-322
for dielectric resonator, 327
for Fabry-Perot resonator, 330
loaded, 306
for rectangular cavity, 315-317
for RLC circuit, 302, 305, 307
for transmission line resonator, 308-309, 311, 312
Quadrature hybrid, 379-383
Quarter-wave transformers
binomial, 278-282

Chebyshev, 282-288 multiple reflection viewpoint, 85-87 multisection, 277-278 single-section. 83-87, 276-277 Radar systems, 672-679 doppler, 677-678 equation, 673-675 pulse. 574--677 Radar cross section, 678-679 Radiation condition, 15-16 Radiation hazards, 694-695 Radiometer. Dicke switched, 684-685 total power, 681-684 Radiometry, 679-685 Rat-race (ring hybrid), 402-407 Reciprocal networks, 193-194, 199-200 Reciprocity theorem, 45-47 Rectangular cavity (see Cavity resonators) Rectangular waveguide attenuation, 125, 127-128 cutoff frequency, 123, 126 field patterns, 129 group velocity, 173 maximum power capacity, 176-177 phase velocity, 124, 128 propagation constant, 123-124, 126 table for, 128 table of standard sizes, 706 Rectification, 559, 563 Reflection coefficient, 66-68 Reflection of plane waves normal incidence, 30-39 oblique incidence at a dielectric, 39-45 Reflectometer, 414-416 Refraction of plane waves, 39-43 by atmosphere, 685-686 Remanent magnetization, 531 Resonance isolator, 523-527 Resonant circuits, 300-307 Return loss, 67 Richards' transformation, 462-463 Ridge waveguide, 174-175 Root-finding algorithms, 152-153 Saturation magnetization, 500 Scattering matrix, 196-205 for circulator, 353, 536-537 for directional coupler, 354-356

715

generalized, 204-205 for gyrator, 535 for quadrature hybrid, 379 for reciprocal or lossless networks, 199-200 for ring hybrid, 402 shift in reference planes, 202-204 table for conversions, 211 Schwinger reversed phase coupler, 413 Separation of variables method, 21-22, 121-122, 133-134, 142-143 Short circuit stub impedance, 68-69 Signal flow graphs, 213-217 Single sideband modulation, 567 Skin depth, 19–21 Slot line, 175 Slotted line, 79-82 SMA connectors, 146 Small reflection theory, 276-278 Smith chart, 73-79 Snell's law, 40 Sources solid state, 589-593 tubes, 593-596 S parameters (see Scattering matrix) Stability amplifier, 612-618 circles, 613-614 open resonator, 330-332 Standing wave ratio (SWR), 67 Stepped impedance filters, 470-473 Stripline approximate analysis, 157–160 attenuation, 156 characteristic impedance, 155-156 coupled, 387 field pattern, 154 propagation constant, 155 Surface current, 9-10, 14-15 Surface impedance, 36-38 Surface resistance, 30, 37 Surface waves at dielectric interface, 43-45 of dielectric slab, 147-152 Switches, PIN diode, 577-580 Tapered coupled line hybrid, 407-411 Tapered transmission lines exponential taper, 290-291 Klopfenstein taper, 291-293 triangular taper, 291

Telegrapher equations, 57 TEM waves and modes attenuation due to dielectric loss, 111-112 plane waves. 16-26 propagation constant, 107 transmission lines, 56-65 wave impedance, 65, 108 TE. TM modes attenuation due to dielectric loss, 111-112 propagation constant, 109-110 wave impedance, 109-110 Terminated transmission line, 65-72 reflection coefficient, 66-68 SWR. 67 voltage maxima and minima, 67 Third-order intercept, 574-575 T-junction, 351-352, 359-362 Total reflection, plane wave, 43-45 Transducer power gain, 606-608 Transmission coefficient, 72 Transmission line equations, 57-58 input impedance, 68--69 junctions, 70–72 parameters, 60-62 Transmission line resonators, 306-313 Transmission lines coaxial, 61-65, 141-145 microstrip. 160-167 parallel plate, 62, 112-120 stripline, 153-160 two-wire, 62 Transverse resonance method, 167–170 Traveling wave amplifier (see Amplifier design) Traveling waves plane waves, 17 on transmission lines, 58 TRL calibration, 217-221 Two-port networks, equivalent circuits for, 210-213 Two-port power gains (see Power gain) Type N connector, 146 Unilateral device, 611 Unilateral figure of merit, 623 Unilateral transducer power gain, 609 Uniqueness theorem, 49-51 Unitary matrix, 201 Unit element, 464

Unit matrix, 199

Unloaded Q, 306

716 Index

Varactor-tuned oscillator, 591–592 Velocity (*see* Wave velocities) Voltage standing wave ratio (*see* Standing wave ratio)

Wave equation, 16–18, 21
Waveguide (*see* Rectangular waveguide; Circular waveguide; Loaded waveguide)
Waveguide components, 121
directional couplers, 368–379
discontinuities, 223
isolators, 523–530
magic-T, 403
phase shifters, 530–535
T-junctions, 359
Waveguide excitation by apertures, 237–244
arbitrary sources, 232–237

current sheets, 230–232 Waveguide flanges, 131–132 Waveguide impedance, 109–110 Wavelength in free-space, 17 on transmission line, 59 for waveguide, 115, 119, 123, 128 Wave velocities group, 170–173, 429 phase, 17, 113, 115, 119, 170, 428 Wheeler incremental inductance rule, 96–98 Wilkinson power divider, 363–368 YIG-tuned oscillator, 591–592

Y-parameters (see Admittance matrix)

Z-parameters (see Impedance matrix)



VECTOR ANALYSIS

Coordinate Transformations

Rectangular to cylindrical:

	ส์	\hat{y}	ź
$\hat{\rho}$	$\cos\phi$	sin ø	0
$\hat{\phi}$	$-\sin\phi$	$\cos\phi$	0
ź	U	0	1

Rectangular to spherical:

	Î	ŷ	ź
î	$\sin\theta\cos\phi$	$\sin \theta \sin \phi$	$\cos\theta$
θ	$\cos\theta\cos\phi$	$\cos heta\sin\phi$	$-\sin \theta$
ô.	– sin <i>o</i>	$\cos\phi$	0

Cylindrical to spherical:

	Â	$\hat{\phi}$	2
ŕ	sin $ heta$	0	$\cos\theta$
Û	$\cos \theta$	Ø	$-\sin \theta$
ô	0	1	0

These tables can be used to transform unit vectors as well as vector components; e.g.

$$\hat{
ho} = \hat{x}\cos\phi + \hat{y}\sin\phi$$

 $A_{
ho} = A_x\cos\phi + A_y\sin\phi$

Vector Differential Operators

Rectangular coordinates:

$$\nabla f = \hat{x}\frac{\partial f}{\partial x} + \hat{y}\frac{\partial f}{\partial y} + \hat{z}\frac{\partial f}{\partial z}$$
$$\nabla \cdot \bar{A} = \frac{\partial A_x}{\partial x} + \frac{\partial A_y}{\partial y} + \frac{\partial A_z}{\partial z}$$
$$\nabla \times \bar{A} = \hat{x}\left(\frac{\partial A_z}{\partial y} - \frac{\partial A_y}{\partial z}\right) + \hat{y}\left(\frac{\partial A_x}{\partial z} - \frac{\partial A_z}{\partial x}\right) + \hat{z}\left(\frac{\partial A_y}{\partial x} - \frac{\partial A_z}{\partial y}\right)$$
$$\nabla^2 f = \frac{\partial^2 f}{\partial x^2} + \frac{\partial^2 f}{\partial y^2} + \frac{\partial^2 f}{\partial z^2}$$
$$\nabla^2 \tilde{A} = \hat{x}\nabla^2 A_x + \hat{y}\nabla^2 A_y + \hat{z}\nabla^2 A_z$$

Cylindrical coordinates:

$$\nabla f = \hat{\rho} \frac{\partial f}{\partial \rho} + \hat{\phi} \frac{1}{\rho} \frac{\partial f}{\partial \phi} + \hat{z} \frac{\partial f}{\partial z}$$
$$\nabla \cdot \tilde{A} = \frac{1}{\rho} \frac{\partial}{\partial \rho} (\rho A_{\rho}) + \frac{1}{\rho} \frac{\partial A_{\phi}}{\partial \phi} + \frac{\partial A_{z}}{\partial z}$$
$$\nabla \times \tilde{A} = \hat{\rho} \left(\frac{1}{\rho} \frac{\partial A_{z}}{\partial \phi} - \frac{\partial A_{\phi}}{\partial z} \right) + \hat{\phi} \left(\frac{\partial A_{\rho}}{\partial z} - \frac{\partial A_{z}}{\partial \rho} \right) + \hat{z} \frac{1}{\rho} \left[\frac{\partial (\rho A_{\phi})}{\partial \rho} - \frac{\partial A_{\rho}}{\partial \phi} \right]$$
$$\nabla^{2} f = \frac{1}{\rho} \frac{\partial}{\partial \rho} \left(\rho \frac{\partial f}{\partial \rho} \right) + \frac{1}{\rho^{2}} \frac{\partial^{2} f}{\partial \phi^{2}} + \frac{\partial^{2} f}{\partial z^{2}}$$
$$\nabla^{2} \hat{A} = \nabla (\nabla \cdot \bar{A}) - \nabla \times \nabla \times \hat{A}$$

Spherical coordinates:

$$\nabla f = \hat{r} \frac{\partial f}{\partial r} + \hat{\theta} \frac{1}{r} \frac{\partial f}{\partial \theta} + \frac{\hat{\theta}}{r \sin \theta} \frac{\partial f}{\partial \phi}$$
$$\nabla \cdot \bar{A} = \frac{1}{r^2} \frac{\partial}{\partial r} (r^2 A_r) + \frac{1}{r \sin \theta} \frac{\partial}{\partial \theta} (\sin \theta A_\theta) + \frac{1}{r \sin \theta} \frac{\partial A_\phi}{\partial \phi}$$
$$\nabla \times \bar{A} = \frac{\hat{r}}{r \sin \theta} \left[\frac{\partial}{\partial \theta} (A_\phi \sin \theta) - \frac{\partial A_\theta}{\partial \phi} \right] + \frac{\hat{\theta}}{r} \left[\frac{1}{\sin \theta} \frac{\partial A_r}{\partial \phi} - \frac{\partial}{\partial r} (rA_\phi) \right] + \frac{\hat{\theta}}{r} \left[\frac{\partial}{\partial r} (rA_\theta) - \frac{\partial A_r}{\partial \theta} \right]$$
$$\nabla^2 f = \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial f}{\partial r} \right) + \frac{1}{r^2 \sin \theta} \frac{\partial}{\partial \theta} \left(\sin \theta \frac{\partial f}{\partial \theta} \right) + \frac{1}{r^2 \sin^2 \theta} \frac{\partial^2 f}{\partial \phi^2}$$
$$\nabla^2 \tilde{A} = \nabla \nabla \cdot \bar{A} - \nabla \times \nabla \times \bar{A}$$



Maxwell's equations:

$$\nabla \times \bar{E} = -j\omega\mu \bar{H} - \bar{M} \qquad \nabla \cdot \bar{D} = \rho$$
$$\nabla \times \bar{H} = j\omega\epsilon \bar{E} + \bar{J} \qquad \nabla \cdot \bar{B} = 0$$

Surface resistance and skin depth:

$$\hat{R}_s = \sqrt{rac{\omega\mu}{2\sigma}} \qquad \delta_s = \sqrt{rac{2}{\omega\mu\sigma}}$$

Input impedance of terminated lossless transmission lines:

$$Z_{in} = Z_0 \frac{Z_L + jZ_0 \tan \beta \ell}{Z_0 + jZ_L \tan \beta \ell}$$
 (arbitrary load)

$$Z_{in} = jZ_0 \tan \beta \ell$$
 (short-circuited line)

$$Z_{in} = -jZ_0 \cot \beta \ell$$
 (open-circuited line)

Relations between load impedance and reflection coefficient:

$$\Gamma = \frac{Z_L - Z_0}{Z_L + Z_0} \qquad Z_L = Z_0 \frac{1 + \Gamma}{1 - \Gamma}$$

Definitions of return loss, insertion loss and SWR:

$$RL = -20 \log |\Gamma|, \qquad IL \approx -20 \log |T|, \qquad \text{SWR} = \frac{1+|\Gamma|}{1-|\Gamma|}$$

Conversion between dB and nepers:

Elements of the ferrite permeability tensor:

$$\mu = \mu_0 \left(1 + \frac{\omega_0 \omega_m}{\omega_0^2 - \omega^2} \right) \qquad \omega_0 = \mu_0 \gamma H_0$$

$$\kappa = \mu_0 \frac{\omega \omega_m}{\omega_0^2 - \omega^2} \qquad \omega_m = \mu_0 \gamma M_s$$

(or 2.8 MHz/Oersted)

Conversion between some values of reflection coefficient, SWR, and return loss:

$ \Gamma $	0.024	0.032	0.048	0.050	0.056	0.10	0.178	0.200	0.316	0.33
SWR	1.05	1.07	1.10	1.11	1,12	1.22	1.43	1.50	1.92	2.00
$RL(d\mathbf{B})$	32.3	30.0	26.4	26.0	25.0	20.0	15.0	14.0	10.0	9.6

Circuit	ABCD Parameters		
c	$A = 1$ $\mathbf{c}^* = 0$	B = Z $D = 1$	
	A = C = Y	B = 0 $D = 1$	
οο Ζ _{ιι} , β	$A = \cos \beta l$ $C = jY_0 \sin \beta l$	$B = fZ_0 \sin \beta t$ $D = \cos \beta t$	
	A = N $C = 0$	$B = 0$ $D = \frac{l}{N}$	
· · · · · · · · · · · · · · · · · · ·	$A = 1 + \frac{Y_2}{Y_3}$ $C = Y_1 + Y_2 + \frac{Y_1Y_2}{Y_3}$	$B = \frac{1}{Y_3}$ $D = 1 + \frac{Y_1}{Y_3}$	
	$A = i + \frac{Z_1}{Z_3}$ $C = \frac{1}{Z_3}$	$B = Z_1 + Z_2 + \frac{Z_1 Z_2}{Z_3}$ $D = 1 + \frac{Z_2}{Z_3}.$	

The ABCD Parameters of Some Useful Two-Port Circuits.